# State Route 37 Sears Point to Mare Island Improvement Project

State Route 37 from State Route 121 to Mare Island Napa, Sonoma, and Solano Counties, California 04-SON-SR 37 (PM 2.9/6.2); 04-SOL-SR 37 (PM 0.0/R7.4); 04-SON-121 (PM 0.0/0.2) EA - 04-1Q761; EFIS - 0419000255

# **Final Environmental Impact Report/Environmental Assessment with Finding** of No Significant Impact **VOLUME 1**



Prepared by the State of California, Department of Transportation

The environmental review, consultation, and any other actions required by applicable Federal environmental laws for this project are being, or have been, carried out by Caltrans pursuant to 23 USC 327 and the Memorandum of Understanding dated May 27, 2022, and executed by FHWA and Caltrans.



February 2023

This page intentionally left blank

## **General Information about this Document**

#### What's in this document:

The California Department of Transportation (Caltrans), as assigned by the Federal Highway Administration (FHWA), has prepared this Final Environmental Impact Report (EIR)/Environmental Assessment (EA), which examines the potential environmental impacts of the proposed project in Sonoma, Napa and Solano counties, California. Caltrans is the lead agency under the National Environmental Policy Act and the California Environmental Quality Act. The document explains why the project is being proposed, what alternatives have been considered for the project, and how the existing environment could be affected by the project. It also describes the potential impacts of each of the alternatives, and proposed avoidance, minimization, and/or mitigation measures. The Draft Environmental Impact Report/Environmental Assessment was circulated to the public for 45 days between January 13 and February 28, 2022. Comments received during this period are included in Appendix K (under separate cover). Elsewhere throughout this document, a vertical line in the margin indicates a change made since the draft document circulation. Minor editorial changes and clarifications have not been so indicated. Additional copies of this document and the related technical studies are available for review at District 4, 111 Grand Avenue, Oakland, CA 94612. This document may be downloaded at the following website at https://dot.ca.gov/caltrans-near-me/district-4/d4-popular-links/d4environmental-docs#district-wide.

#### Alternative Formats:

For individuals with sensory disabilities, this document can be made available in Braille, in large print, on audiocassette, or on computer disk. To obtain a copy in one of these alternate formats, please call or write to Department of Transportation, Attn: Jeff Weiss, P.O. Box 23660 MS 8B, Oakland, CA, 94623-0660, e-mail <u>stateroute37@dot.ca.gov</u>, or at (510) 715-8770 (Voice), or use California Relay Service 1 (800) 735-2929 (TTY to Voice), 1 (800) 735-2922 (Voice to TTY), 1 (800) 855-3000 (Spanish TTY to Voice and Voice to TTY), 1-800-854-7784 (Spanish and English Speech-to-Speech) or 711. This page intentionally left blank

SCH Number: 2020070226 04-SON-37-PM 2.9/6.2; 04-SOL-37-PM 0.0/R7.4; 04-Son-121 PM 0.0/0.2 EA No. 04-1Q761 Project No. 0419000255

## State Route 37 Sears Point to Mare Island Improvement Project

Final Environmental Impact Report/Environmental Assessment with Finding of No Significant Impact

Submitted Pursuant to: (State) Division 13, California Public Resources Code (Federal) 42 USC 4332(2)(C) and 49 USC 303

### THE STATE OF CALIFORNIA Department of Transportation

Responsible Agencies under CEQA: California Transportation Commission, San Francisco Bay Conservation and Development Commission, California Department of Fish and Wildlife, San Francisco Bay Regional Water Quality Control Board

02/08/2023

Date

Dina El-Tawansy

Dina A. El-Tawansy District Director, District 4 – Bay Area CEQA and NEPA Lead Agency

The following persons may be contacted for more information about this document:

Cody Ericksen, Acting Senior Environmental Scientist California Department of Transportation, District 4 P.O. Box 23660, MS 8B Oakland, CA 94623-0660 Cody.Ericksen@dot.ca.gov (510) 506-9678 (cell) This page intentionally left blank.

## CALIFORNIA DEPARTMENT OF TRANSPORTATION FINDING OF NO SIGNIFICANT IMPACT (FONSI)

FOR

State Route 37 Sears Point to Mare Island Improvement Project

The California Department of Transportation (Caltrans) has determined that identification of the preferred alternative (Alternative 3B) would have no significant impact on the human environment. This FONSI is based on the attached Environmental Assessment (EA), which has been independently evaluated by Caltrans and determined to adequately and accurately discuss the need, environmental issues, and impacts of the proposed project and appropriate mitigation measures. It provides sufficient evidence and analysis for determining that an Environmental Impact Statement is not required. Caltrans takes full responsibility for the accuracy, scope, and content of the attached EA.

The environmental review, consultation, and any other actions required by applicable federal environmental laws for this project are being, or have been, carried out by Caltrans pursuant to 23 USC 327 and the Memorandum of Understanding dated May 27, 2022, and executed by FHWA and Caltrans.

Dina Cl-Tawansy

Dina El-Tawansy District Director Caltrans District 4 CEQA/NEPA Lead Agency

02/08/2023

Date

This page intentionally left blank.

## Summary

### **NEPA Assignment**

California participated in the "Surface Transportation Project Delivery Pilot Program" (Pilot Program) pursuant to 23 United States Code (USC) 327, for more than 5 years, beginning July 1, 2007, and ending September 30, 2012. MAP-21 (Public Law 112-141), signed by President Obama on July 6, 2012, amended 23 USC 327 to establish a permanent Surface Transportation Project Delivery Program. As a result, the California Department of Transportation (Caltrans) entered into a Memorandum of Understanding (MOU) pursuant to 23 USC 327 (National Environmental Policy Act [NEPA] Assignment MOU) with the Federal Highway Administration (FHWA). The NEPA Assignment MOU became effective October 1, 2012, and was renewed on May 27, 2022, for a term of 10 years. In summary, Caltrans continues to assume FHWA responsibilities under NEPA and other federal environmental laws in the same manner as was assigned under the Pilot Program, with minor changes. With NEPA Assignment, FHWA assigned and Caltrans assumed all of the United States Department of Transportation Secretary's responsibilities under NEPA. This assignment includes projects on the State Highway System and Local Assistance Projects off the State Highway System in the State of California, except for certain categorical exclusions that FHWA assigned to the Department under the 23 USC 326 CE Assignment MOU; projects excluded by definition; and specific project exclusions.

### Joint NEPA/CEQA Document

The proposed project is a joint project by Caltrans and FHWA, and is subject to state and federal environmental review requirements. Project documentation, therefore, has been prepared in compliance with both the California Environmental Quality Act (CEQA) and NEPA. Caltrans is the lead agency under NEPA. Caltrans is the lead agency under CEQA. In addition, FHWA's responsibility for environmental review, consultation, and any other actions required by applicable federal environmental laws for this project are being, or have been, carried out by Caltrans pursuant to 23 United States Code Section 327 (23 USC 327) and the Memorandum of Understanding, dated May 27, 2022, and executed by FHWA and Caltrans.

Some impacts determined to be significant under CEQA may not lead to a determination of significance under NEPA. Because NEPA is concerned with the significance of the project as a whole, a "lower level" document is often prepared for NEPA. One of the most common joint document types is an Environmental Impact Report/Environmental Assessment (EIR/EA).

Caltrans prepared a draft EIR/EA that was circulated to the public from January 13, 2022, to February 28, 2022, for review and comment. This Final EIR/EA was prepared after receiving comments from the public and reviewing agencies. Written comments from individuals, organizations, and public agencies received during the circulation period are included as an appendix. The identified preferred alternative in this Final

EIR/EA was presented as Build Alternative 3B, Widen to Four Lanes, with 8-Foot Shoulders; and as Widen Sonoma Creek Bridge in the Draft EIR/EA, and includes some modifications from what was presented in the Draft EIR/EA. The rationale and identification of the preferred alternative is described in Section 1.4.3.1.

Caltrans has decided to issue a Finding of No Significant Impact (FONSI) for compliance with NEPA. A Notice of Determination (NOD) will be published for compliance with CEQA. A Notice of Availability (NOA) of the FONSI will be sent to the affected units of federal, state, and local government, and to the State Clearinghouse in compliance with Executive Order 12372.

#### Introduction

The California Department of Transportation (Caltrans)—in cooperation with the Metropolitan Transportation Commission and Sonoma County Transportation Authority, Solano Transportation Authority, and Napa Valley Transportation Authority—is proposing improvements to the portion of Highway 37 between Sears Point and Mare Island. Highway 37 is referred to as State Route (SR) 37 throughout the remainder of this document.

The proposed section of SR 37 for improvement is a two-lane conventional highway (one lane in each direction) between SR 121 and the Mare Island Interchange (approximately 10 miles). East of the Mare Island, Interchange SR 37 is a four-lane freeway facility between Interstate 80 and Mare Island (approximately 4.5 miles). West of SR 121, it is a four-lane conventional highway between SR 121 and U.S. Highway 101 (approximately 7.3 miles).

The purpose of the project is to improve traffic flow and peak travel times, and increase vehicle occupancy (the number of people moved per vehicle). The project is needed to address reoccurring congestion in the near term on SR 37, where the highway narrows to one lane in each direction between SR 37/SR 121.

Caltrans proposes to reconfigure the existing SR 37 highway lanes from west of the SR 121 intersection to the Walnut Avenue Overcrossing at Mare Island. The project's preferred alternative would widen the Tolay Creek Bridge; have four lanes, with two full-time lanes in each direction; have 8-foot shoulders (except at Sonoma Creek Bridge, which would have 4-foot shoulders); and would require widening the Sonoma Creek Bridge. The preferred alternative would introduce the eastbound high-occupancy vehicle (HOV) lane at the SR 121/SR 37 intersection vicinity using Scenario 3, with the HOV lane on the left side, about 0.6 mile west of SR 121, and extend the eastbound left-turn lane approximately 0.5 mile west.

The added lanes would be for general purpose use (one new lane in each direction between SR 121 and the Walnut Avenue Overcrossing at Mare Island). The project includes the installation and operation of Open Road Tolling on the new general purpose lanes, either one-way or two-way tolling. The existing lanes in this same segment would be converted to HOV lanes, and operate as HOV lanes 24 hours per day, 7 days a week.

### **Project Impacts**

Table S-1 summarizes and compares the effects of all of the alternatives considered at the time of the Draft EIR/EA: Alternatives 1, 2, 3A, 3B, and the No Build Alternative. As explained in Section 1.4.3.1, Alternative 3B was identified as the preferred alternative following completion of review of the Draft EIR/EA, and consideration of the comments received. The proposed avoidance, minimization, and/or mitigation measures to reduce the effects of the Build Alternatives are also presented. A complete description of potential effects and recommended measures is provided in Chapter 2.

February 2023

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Existing and Future Land Use	None	Alternative 1 would require temporary construction easements (TCEs) in the Napa-Sonoma Marshes Wildlife Area (NSMWA) and San Pablo Bay National Wildlife Refuge (Refuge). However, the affected parcels would be restored following construction, and no permanent change to any land use would result.	Same as Alternative 1	Alternative 3A would require TCEs, in addition to permanent right-of-way acquisition at the Refuge (1.65 acres) and NSMWA. The partial acquisitions would occur along the edge of the Refuge and NSMWA where these lands are bisected by the existing roadway and provide limited recreational value. Temporary or permanent impacts would not affect existing land uses of the rest of the Refuge or NSMWA areas. Therefore, project construction and operation would not result in major changes to the land use or zoning of any parcels in the project area.	Same as Alternative 3A, but total permanent acquisition would include additional parcel acquisition. At the NSMWA, temporary acquisition would be 0.29 acre and permanent 1.43 acres. At the Refuge, permanent acquisition would be 2.76 acres. The partial acquisitions and temporary easements would occur along the edge of the NSMWA and Refuge, where it is bisected by the existing roadway and provides limited recreational value and would not affect existing land uses of the rest of the Refuge area.	Acquisitions and TCEs would require compensation. Because the TCEs are on public lands, the temporary use of these lands and any compensation would be defined in coordination with the Refuge.

## Table S-1 Summary of Impacts and Avoidance, Minimization, and/or Mitigation Measures

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Consistency with State, Regional, and Local Plans and Programs	The No Build Alternative would not be consistent with <i>Plan Bay Area</i> <i>2050</i> because it would not encourage ridesharing, car- pooling, or vanpooling. It would also not be consistent with Solano, Napa, and Sonoma Counties because it would not involve traffic management improvements along SR 37 to reduce congestion and/or transportation improvements that make intra-city travel easier. The existing lane reductions in both directions would remain, and would result in congestion, backups, and delays.	Alternative 1 would be consistent with most applicable plans and policies. It would be consistent with plans, programs, and policies related to bicycle facilities because it would continue to provide accessible shoulders for bicycle access.	Alternative 2 would be consistent with most applicable plans and policies. It would not be consistent with plans, programs, and policies related to bicycle facilities because it would not accommodate bicyclists when the shoulder running lane is open to traffic during the peak hours in the peak direction. Legislation to prohibit bicycle use along the corridor would be proposed.	Alternative 3A would be consistent with most applicable plans and policies. It would not be consistent with plans, programs, and policies related to bicycle facilities. Shoulders would be narrower (4 feet) but would not continue at the Sonoma Creek Bridge. Legislation to prohibit bicycle use along the corridor would be proposed.	Alternative 3B would be consistent with most applicable plans and policies. It would be consistent with plans, programs, and policies related to bicycle facilities. It would provide an 8-foot-wide shoulder in both directions along the entire route—except at the Sonoma Creek Bridge, which would have a 4-foot-wide shoulder. The shoulders would be open to bicycle use along the corridor in both directions.	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Parks and Recreation Facilities	None	Alternative 1 would require TCEs in the NSMWA and Refuge but would not require the permanent use of any publicly owned park or recreational facility.	Same as Alternative 1.	Alternative 3A would require TCEs and permanent acquisition of sliver takes in the NSMWA and Refuge. It would also include permanent use of areas in the Refuge lands totaling 1.65 acres.	Same as Alternative 3B, but at the NSMWA, temporary acquisition would be 0.29 acre, and permanent 1.43 acres. At the Refuge, permanent acquisition would be 2.76 acres.	Land acquisition at the Refuge would require compensation.
Farmlands	None	None	None	None	None	None
Growth		Build Alternative 1 would increase the capacity of SR 37 in the project area but would not change overall land use or provide access to previously undeveloped land. It would accommodate planned growth but would not affect land use decisions in a way that would encourage growth beyond reasonably foreseeable levels.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Community Character and Cohesion	None	The project would generally improve access to adjacent and nearby land uses by reducing congestion. Alternative 1 would not displace or relocate any residents or encourage more people to move to the surrounding areas.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None
Relocations and Real Property Acquisition	None	Alternative 1 would not require any full property acquisitions and would not require any home or business relocations. TCEs would be required. However, Build Alternative 1 would require TCEs of approximately 0.12 acre of the 15,200-acre NSMWA.	Alternative 2 would not require any full property acquisitions and would not require any home or business relocations. TCEs would be required. Build Alternative 2 would require TCEs of approximately 0.16 acre of the NSMWA and approximately 0.44 acre of the 19,000-acre Refuge.	Alternative 3A would require TCEs and permanent property acquisition of areas in the Refuge and NSMWA. It would not require any home or business relocations. TCEs would be required. However, Build Alternative 3A would require TCEs of approximately 0.15 acre of the NSMWA and 0.03 acre of the Refuge, in addition to permanent use of approximately 1.65 acres of the Refuge.	Alternative 3B would require TCEs and permanent property acquisition of areas in the Refuge and NSMWA. Build Alternative 3B would require TCEs of approximately 0.29 acre and permanent acquisition of 1.43 acres of the NSMWA. Alternative 3B would also acquire permanent use of 2.76 acres of the Refuge.	None. Acquisitions and TCEs would require compensation.
Environmental Justice	None	None	None	None	None	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Utilities/ Emergency Services	None	The relocation of electrical facilities may result in temporary interruptions of service. It would not result in long-term effects to utilities or emergency services.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None
Traffic and Transportation	The No Build Alternative would not improve traffic conditions along SR 37.	Alternative 1 would improve traffic conditions compared to the No Build conditions but would not eliminate all congestion, due to the limited hours of operation for HOV lanes. Daily Vehicle miles traveled (VMT) was evaluated across all nine Bay Area counties. VMT would increase by 6,346 in 2025 and 31,729 in 2045 (in comparison to the No Build), without the implementation of tolling. VMT would decrease compared to the No Build with tolling implemented.	Same as Alternative 1	The project is expected to improve traffic conditions along SR 37, and to improve the traffic flow and travel times in the peak direction. Alternative 3A would eliminate the lane reductions in each direction at all times and shows benefits for the general-purpose lane as well as the HOV lane in each direction. Delays are improved compared to the No Build Alternative. VMT, as evaluated across all nine Bay Area counties, would increase by 9,599 in 2025 and 47,992 in 2045 in comparison to the No Build, without the implementation of tolling.	Same as Alternative 3A	VMT is an impact criteria under CEQA. With tolling implemented as proposed with the project, there would be no adverse impact requiring mitigation. VMT-1: Reductions in VMT can be achieved by measures in addition to tolling. These include project support of bus service within the SR 37 corridor that would use the HOV lanes. Support of additional park and ride facilities and ride sharing services that serve SR 37 commuters. The performance measure will be the difference between the VMT impact without mitigation, as identified for each alternative, and with mitigation. For

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
		Vehicle hours of delay and vehicle hours traveled would decrease with or without tolling.		Tolling is a proposed project component and would be applied to the new lane in either one or both directions. With tolling, VMT be lower than the No Build for the preferred Alternative 3B, and no adverse VMT impact is identified. Vehicle hours of delay and vehicle hours traveled would decrease with or without tolling.		Alternatives 1 and 2, without the implementation of tolling, VMT would increase by 6,346 in 2025 and 31,729 in 2045 (in comparison to the No Build) without mitigation. For Alternatives 3A/3B, without the implementation of tolling, VMT would increase by 9,599 in 2025 and 47,992 in 2045 in comparison to the No Build without mitigation. VMT is influenced by tolling rates. If necessary, toll rates would be increased as needed to reduce VMT in the proposed toll lane in either one or both directions. In the HOV lane in each direction, HOV enforcement would be increased to reduce HOV violations. Either or both of these strategies would be applied if necessary to meet VMT performance measures.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Visual/ Aesthetics	None. The existing 32-inch-high median barrier will remain. There are no existing outside barrier rail systems.	Alternative 1 would have moderate visual impacts in combination with the Midwest Guard Rail System outside barrier, but would not have substantial adverse effects. With the Type 85B outside barrier, Alternative 1 would create moderate- high to high visual impacts because of its potential to obstruct views of scenic landscapes on the other side of the highway for most motorists.	Alternative 2 would create substantial adverse effects to a scenic vista because the 42-inch median barrier would interfere with views of low-lying scenic landscapes on the other side of the highway for many drivers. The Type 85B outside barrier would add to the adverse impacts by reducing views on the same side of the highway.	Same as Alternative 2.	The DED circulated for public review and comment considered the visual impacts for Alternative 3B to be the same as Alternative 2. Further review of the median barrier during identification of the preferred alternative and preparation of the Final EIR/EA determined it could be reduced from 42 inches to 36 inches high. This would reduce but not eliminate the height change from the existing 32-inch-high barrier. Further review also identified that Midwest Guard Rails System would be appropriate for Alternative 3B, further reducing potential impacts on Bay views to travelers on SR 37.	VIS-01: Limit Light Pollution. For permanent impacts, lighting on new ramps, at intersections, in advance of tolling gantries, and at CHP enforcement areas will be designed to limit light pollution and have minimum impact on the surrounding environment. All light fixtures will have light-emitting diodes configured at the minimum necessary number of bulbs, optimal mounting height, mast-arm length, and angle to restrict light to the roadways. Where applicable, shields on the fixtures to prevent light trespass to adjacent properties will be considered during the detailed design phase.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Cultural Resources	None	Ground-disturbing activities during construction of the project could affect unknown buried cultural resources in areas adjacent to SR 37.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None. Standard project features would be implemented during construction to address potential discovery of previously unidentified cultural resources, or human remains.
Hydrology and Floodplain	None	The fill placed for the proposed widening would have minimal effects on the floodplain. Most improvements in the project would be within the existing impervious area and would not change the 100-year floodplain. The amount of new impervious surface area added would not have an impact to the flows within the project's limits.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Water Quality and Storm Water Runoff	None	Alternative 1 would disturb 44.86 acres of soil area, add 12.17 acres of net new impervious area, and replace 11.57 acres of disturbed impervious area. The project would comply with standard practices to reduce impacts to water quality and would be in compliance with National Pollutant Discharge Elimination System (NPDES).	Alternative 2 would disturb 44.86 acres of soil area, add 19.75 acres of net new impervious area, and replace 20.42 acres of disturbed impervious area. The project would comply with standard practices to reduce impacts to water quality and would be in compliance with NPDES. The project would implement best management practices.	Alternative 3A would disturb 79.88 acres of soil area, add 21.19 acres of net new impervious area, and replace 21.11 acres of disturbed impervious area. The project would comply with standard practices to reduce impacts to water quality and would be in compliance with NPDES.	Alternative 3B would disturb 87.42 acres of soil area, add 28.25 acres of net new impervious area, and replace 21.27 acres of disturbed impervious area. The project would comply with standard practices to reduce impacts to water quality and would be in compliance with NPDES.	Standard project features would be implemented during design and construction to address stormwater runoff in compliance with Caltrans' municipal separate storm sewer system and NPDES permits. These may include construction site BMPs, design pollution prevention BMPs, treatment BMPs including hydromodification and treatment BMPs to minimize temporary and permanent impact for maximum extent practicable (MEP) approach.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Geology/Soils/ Seismic	The No Build Alternative would be subject to the existing geologic, soils, and seismic hazards.	Earthmoving activities such as grading, excavation, and trenching have the potential to result in soil erosion and loss of topsoil, especially in areas where there are steeper slopes. Hazards related to landslides, lateral spreading, and liquefaction susceptibility are considered low or low to moderate. Work would be done to widen Sonoma Creek Bridge and Tolay Creek Bridge. The median barrier would also be removed and replaced.	Same as Alternative 1	Impacts will be similar to those under Alternative 1. The difference is that a retaining wall would be implemented west of the SR 121 interchange, which would require ground disturbances.	Same as Alternative 3A	None. Standard project features would be implemented during design and construction to address potential geological and seismic hazards.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Paleontology	None	No known resources were identified at the project. Three fossil localities are potentially in the regional vicinity of the project and there are areas within the project limits that have potential sensitivity. Although there are areas of potential sensitivity, there is low potential of impact due to the implementation of project features.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None
Hazardous Waste/ Materials	None	Construction of the project could result in the potential disturbance of hazardous materials in the soil and groundwater. No long- term impacts are expected to occur.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None. Standard project features would be implemented during design and construction to address potential contamination hazards.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Air Quality	None	When compared to the No Build Alternative, regional emissions under Build Alternative 1 would be the same or slightly higher, given that this alternative slightly increases VMT in the region and provides some increases in travel time. Over time, emissions associated with improvements in vehicle tailpipe emissions would decrease.	Same as Alternative 1	Build Alternative 3A would have regional emissions slightly lower or approximately the same as the No Build Alternative. This is due to a greater reduction in regional travel time (i.e., higher travel speeds) despite an increase in regional VMT. Over time, emissions associated with improvements in vehicle tailpipe emissions would decrease.	Same as Alternative 3A	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Noise	Existing noise levels will increase by 0 to 2 A-weighted decibels (dBA) by 2045.	Noise levels would increase by 1 to 2 dBA. Noise levels would approach or exceed noise abatement criteria at six locations, but would not achieve a minimum abatement reduction. One location at Sonoma Creek would achieve a minimum reduction, but the barrier did not meet cost reasonableness criteria. Construction would have temporary noise impacts, including impacts from pile driving.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1.	None. Standard project features would be required of the contractor to control construction noise. Following public and agency review, the final noise abatement decision is that no soundwalls met the minimum criteria, and none are proposed for the project.
Energy	None	Alternative 1 would have small direct and indirect energy increases compared to the No Build Alternative. Operational daily fuel consumption would have a 0.03 percent change from the No Build Alternative in both 2025 and 2045.	Same as Alternative 1	Alternative 3A would have small direct and indirect energy increases compared to the No Build Alternative. Operational daily fuel consumption would have 0.02 percent change from the No Build Alternative in both 2025 and 2045.	Same as Alternative 3A	None

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Natural Communities	None	Alternative 1 would have substantial direct and indirect impacts to natural communities by impacting approximately 1.91 acres of sensitive marsh and wetland communities (i.e., salt marsh bulrush marshes, pickleweed mats, and California cordgrass marsh), and would have negligible impacts to trees and valley oaks.	Alternative 2 would have substantial direct and indirect impacts to natural communities by impacting approximately 3.29 acres of sensitive marsh and wetland communities (i.e., salt marsh bulrush marshes, pickleweed mats, and California cordgrass marsh), and would have negligible impacts to trees and valley oaks.	Alternative 3A would have substantial direct and indirect impacts to natural communities by impacting approximately 3.85 acres of sensitive marsh and wetland communities (i.e., salt marsh bulrush marshes, pickleweed mats, and California cordgrass marsh), and would have negligible impacts to trees and valley oaks.	Alternative 3B would have the most substantial impact to natural communities, impacting approximately 7.55 acres of sensitive marsh and wetland communities (i.e., salt marsh bulrush marshes, pickleweed mats, and California cordgrass marsh), and would have negligible impacts to trees and valley oaks.	BIO-01: Wetlands Protection – Invasive Plants BIO-02: Wetland Protection BIO-03: Tree Replacement, Landscaping, and Revegetation Plan
Wetlands and Other Water	None	Alternative 1 would have substantial direct impacts resulting in approximately 2.03 acres of permanent loss of wetlands and other waters in the project area, primarily from fill in wetlands where roadway expansion is planned. Approximately 6.34 acres of temporary impacts to wetlands and other	Alternative 2 would have substantial direct impacts resulting in approximately 3.49 acres of permanent loss of wetlands and other waters in the project area, primarily from fill in wetlands where roadway expansion is planned. Approximately 10.02 acres of temporary impacts to wetlands and other	Alternative 3A would have substantial direct impacts (similar to Alternative 2) resulting in approximately 4.28 acres of permanent loss of wetlands and other waters in the project area, primarily from fill in wetlands where roadway expansion is planned. Approximately 10.35 acres of temporary impacts to	Alternative 3B would have the greatest impact on wetlands and other waters, resulting in approximately 9.02 acres of permanent loss of wetlands and other waters in the project area, primarily from fill in wetlands where roadway expansion is planned, and a relatively minor amount for bridge widening at Tolay Creek and Sonoma Creek bridges. An additional	BIO-01: Wetlands Protection – Invasive Plants BIO-02: Wetland Protection BIO-04: Estuarine Dewatering Work Window BIO-05: Turbidity Control BIO-07: Wetlands and Other Waters Compensation

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
		waters, primarily from temporary construction access, would occur and would be restored to pre-project conditions or better. Indirect impacts from soil disturbance and stormwater run-off would be minimal under all alternatives.	waters, primarily from temporary construction access, would occur and would be restored to pre-project conditions or better. Indirect impacts from soil disturbance and stormwater run-off would be minimal under all alternatives.	wetlands and other waters, primarily from temporary construction access, would occur and would be restored to pre-project conditions or better. Indirect impacts from soil disturbance and stormwater run-off would be minimal under all alternatives.	0.7 acre of wetlands and other waters would be permanently shaded by the Tolay Creek and Sonoma Creek Bridge widenings. Approximately 7.02 acres of temporary impacts to wetlands and other waters, primarily from temporary construction access, would occur and would be restored to pre-project conditions or better Indirect impacts from soil disturbance and stormwater run-off would be minimal under all alternatives.	

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Plant Species	None	Alternative 1 is unlikely to have any direct or indirect impacts to special-status plant species. However, additional floristic surveys and monitoring during construction are proposed to confirm the presence or absence of all potential special- status species. Invasive plant control is proposed as a standard measure to limit potential impacts to special-status plant species.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	<b>BIO-08:</b> Targeted Pre- Construction Plant Survey <b>BIO-09:</b> Special-Status Plant Monitoring
Animal Species: State Special Concern	None	Alternative 1 has potential to impact non- listed special-status species habitat during vegetation removal, including the pallid bat, San Pablo song sparrow, and nesting bird species protect by the Migratory Bird Treaty Act (MBTA). project impacts would be negligible for these species. Project features and proposed nesting bird protection measures (BIO-10)	Same as Alternative 1 for pallid bat, San Pablo song sparrow, nesting bird species protect by the MBTA, and the Western Burrowing Owl. Suisun shrew habitat impacts would be permanently and temporarily impacted. Impacts would be the same as those described for salt marsh harvest mouse for this alternative. Salt marsh harvest mouse	Same as Alternative 1 for pallid bat, San Pablo song sparrow, nesting bird species protect by the MBTA, and the Western Burrowing Owl. Suisun shrew habitat impacts would be permanently and temporarily impacted. Impacts would be the same as those described for salt marsh harvest mouse for this alternative. Salt marsh harvest mouse	Same as Alternative 1 for San Pablo song sparrow, nesting bird species protect by the MBTA, and the Western Burrowing Owl. Pallid bat: in addition to BIO-10, measures BIO-11 and BIO-12 are proposed to address potential impacts to bats and roosting habitat that may occur on bridges that would be widened under Alternative 3b.	BIO-10: Nesting Bird Protection BIO-11: Pre-Construction Bat Surveys and Avoidance Measures BIO-12: Bat Monitoring Protocols BIO-13: Western Burrowing Owl Pre- Construction Surveys BIO-14: Western Burrowing Owl Nest Avoidance

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
		would serve to avoid impacts completely or to make impacts negligible. Suisun shrew habitat impacts would be permanently and temporarily impacted the same as those described for salt marsh harvest mouse for Alternative 1. Salt marsh harvest mouse measures would fully address impacts to the shrew, and no additional measures are proposed here. Western Burrowing Owl is not anticipated to be nesting in the project area, but has potential to overwinter or graze in adjacent grasslands outside of the project footprint. With implementation of proposed surveys and monitoring as appropriate (BIO-13 and BIO-14), no impacts are anticipated.	proposed for threatened and endangered fish species would fully address impacts to Sacramento Splittail, and no additional measures are proposed here. measures would fully address impacts to the shrew, and no additional measures are proposed here. Sacramento Splittail habitat would be permanently and temporarily impacted. Habitat impacts would be the same as those described for threatened and endangered anadromous fish species for this alternative. Measures	measures would fully address impacts to the shrew, and no additional measures are proposed here. Sacramento Splittail habitat would be permanently and temporarily impacted. Habitat impacts would be the same as those described for threatened and endangered anadromous fish species for this alternative. Measures proposed for threatened and endangered fish species would fully address impacts to Sacramento Splittail, and no additional measures are proposed here.	Sacramento Splittail habitat would be permanently and temporarily impacted. Habitat impacts would be the same as those described for threatened and endangered anadromous fish species for this alternative. Measures for threatened and endangered fish species, as well as measures for pile driving, would fully address impacts to Sacramento Splittail, and no additional measures are proposed here.	

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
		Sacramento Splittail habitat would be permanently and temporarily impacted. Habitat impacts are the same as those described for threatened and endangered anadromous fish species for this alternative. Those measures would fully address impacts to Sacramento Splittail, and no additional measures are proposed here.				
Threatened and Endangered Species	None	Alternative 1 would have substantial permanent and temporary impacts on habitat for listed Chinook salmon, Delta smelt, steelhead, green sturgeon, longfin smelt, Ridgway's rail, California black rail; salt marsh harvest mouse, and California red-legged frog habitat, respectively. Environmental commitments to address permanent	Alternative 2 would have relatively similar, but slightly greater, area of impacts to the same listed species habitat as Alternative 1. With implementation of proposed measures, direct impacts to all listed species with potential to occur would be limited to no greater than harassment during construction.	Alternative 3A would have relatively similar area of impacts to the same listed species habitat as Alternative 2. With implementation of proposed measures, direct impacts to all listed species with potential to occur would be limited to no greater than harassment during construction.	Alternative 3B would have the greatest area of impacts to the same listed species habitat as Alternatives 1, 2, and 3B. Alternative 3B has potential to result in harm or mortality for green sturgeon during in-water construction to widen the Sonoma Creek Bridge. With proposed measures for Alternative 3 BIO-36, BIO-37. and BIO-38, potential impacts to this	<ul> <li>BIO-08: Targeted Pre- Construction Plant Survey</li> <li>BIO-09: Special-Status</li> <li>Plant Monitoring</li> <li>BIO-18: Wildlife Species</li> <li>Relocation</li> <li>BIO-19: Construction</li> <li>Noise</li> <li>BIO-20: California Red- Legged Frog Habitat Work</li> <li>Window</li> <li>BIO-21: California Red- Legged Frog Pre- Construction Surveys</li> </ul>

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
		and temporary habitat impacts to threatened and endangered species are proposed for Alternative 1 (BIO-18 through 30, BIO-35, and BIO-39). Soft bird's-beak, a federally listed plant species, is not likely to be directly or indirectly impacted. However, measures to avoid take of this species where it has potentially suitable habitat are proposed for all project alternatives (BIO-08 and BIO-09). With implementation of proposed measures, direct impacts to listed species would be limited to no greater than harassment during construction.			species would be greatly reduced. With implementation of proposed measures, direct impacts to other listed species with potential to occur would be limited to no greater than harassment during construction.	<ul> <li>BIO-22: California Red- Legged Frog Monitoring Protocols</li> <li>BIO-23: Compensation for California Red-Legged Frog Habitat Effects</li> <li>BIO-24: Ridgway's Rail and California Black Rail Pre-Construction Survey</li> <li>BIO-25: Ridgway's Rail and California Black Rail Protocol-Level Surveys and Avoidance and Minimization Measures.</li> <li>BIO-26: Compensation for Ridgway's Rail Habitat Effects</li> <li>BIO-27: Salt Marsh Harvest Mouse Pre- Construction Surveys</li> <li>BIO-28: Salt Marsh Harvest Mouse Exclusion Fencing</li> <li>BIO-29: Salt Marsh Harvest Mouse Monitoring Protocols</li> <li>BIO-30: Compensation for Salt Marsh Harvest Mouse and California Black Rail Habitat Effects</li> <li>BIO-31: Vibratory Pile Driving</li> </ul>

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
						<b>BIO-32:</b> In-Water Sheet Pile Fish Entrapment Avoidance
						BIO-33: Fish Monitoring
						BIO-34: Fish Relocation
						<b>BIO-35:</b> Compensation for Chinook salmon, steelhead, green sturgeon, longfin smelt and Delta smelt habitat
						<b>BIO-36:</b> In-Water Impact Pile Driving Work Window
						<b>BIO-37:</b> In-Water Impact Pile Driving Attenuation
						<b>BIO-38:</b> Hydroacoustic Monitoring
						<b>BIO-40:</b> Swainson's Hawk Pre-Construction Surveys

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
Invasive Species	None	Previously described measures for plant species and sensitive habitat would effectively avoid and minimize effects from nonnative invasive plants (NNIPs). With implementation of already proposed measures, no substantial impacts from NNIP species is anticipated, and no new measures are proposed.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	<b>BIO-01:</b> Wetlands Protection – Invasive Plants
Wildfire	Existing wildfire hazards are low in the project area. The No Build Alternative would not change fire risk conditions.	The project would not change fire risk conditions and it would not change the alignment of SR 37.	Same as Alternative 1	Same as Alternative 1	Same as Alternative 1	None
Climate Change	The No Build Alternative would have lower greenhouse gas (GHG) emissions than existing conditions. Existing GHG gas emissions from vehicle use on SR 37 would occur.	Alternative 1 would have lower GHG emissions than existing conditions. Modeling shows that Build Alternative 1 would have slightly lower annual GHG emissions than the No Build Alterative in 2025, 2040, and	Same as Alternative 1	Alternative 3A would have lower GHG emissions than existing conditions. Modeling shows that Alternative 3A would have slightly lower GHG emissions than No Build conditions in 2025. In 2040 and 2045, Alternative 3A	Same as Alternative 3A	None.

Affected Resource	Potential Impact: No Build Alternative	Potential Impact: Alternative 1	Potential Impact: Alternative 2	Potential Impact: Alternative 3A	Potential Impact: Alternative 3B (Preferred Alternative)	Avoidance, Minimization, or Mitigation Measures
	GHG emissions for base year 2020 are 24,555,199 million metric tons of carbon dioxide equivalent. Traffic congestion would not be improved, and GHG emissions due to idling vehicles will continue to occur. As traffic worsens due to regional growth, GHG would increase but would be at least partially offset by driver conversion to lower emission vehicles over time (the conversion to lower emission vehicles is not a result of implementing the No Build Alternative, however).	2045, Reducing queues and stop-and- go traffic would reduce idling and therefore reduce GHG emissions due to idling. Therefore, factoring these benefits in will result in additional GHG emission reduction benefits compared to the No Build Alternative. GHG emissions will be emitted during construction. The transition by drivers to lower emission vehicles over time will lower GHG emissions the same as No Build (this is not a result of this alternative).		would be higher than No Build conditions. Reducing queues and stop-and-go traffic would reduce idling and therefore reduce GHG emissions due to idling. Therefore, factoring these benefits in will result in additional GHG emission reduction benefits compared to the No Build Alternative. GHG emissions will be emitted during construction. The transition by drivers to lower emission vehicles over time will lower GHG emissions same as No Build (this is not a result of this alternative).		

This page intentionally left blank

## **Table of Contents**

Chapte	r 1 Prop	osed Pr	oject	1-1
1.1	Introduc	ction		1-1
	1.1.1	History a	and Other Corridor Projects	1-5
1.2	Purpose	e and Ne	ed	1-7
	1.2.1	Purpose	of the Project	1-7
	1.2.2	Project I	Need	1-8
		1.2.2.1	Existing Congestion and Transportation Demand	1-8
		1.2.2.2	Legislation	1-10
	1.2.3	Indepen	dent Utility and Logical Termini	1-11
1.3	Project	Descripti	on	1-13
1.4	Alternat	ives		1-14
	1.4.1	Preferre	d Alternative	1-14
		1.4.1.1	Alternative 3B, Widen to Four Lanes, with Eight-Foot	
			Shoulders, and Widen Sonoma Creek Bridge	1-14
		1.4.1.2	HOV Scenarios	1-17
		1.4.1.3	Design Features of the Preferred Alternative	1-18
		1.4.1.4	Iransportation System Management and Transportation	1 05
		4 4 4 5	Demand Management Alternatives	1-20
		1.4.1.0	Access to Nevigeble Divers	1-20
	112	1.4.1.0 Compor	icon of Altornativos	1-20
	1.4.2	1 / 2 1	No Build Alternative	1 26
		1.4.2.1	Alternative 1 Three Lane Contra Flow with Moveable	1-20
		1.4.2.2	Median Barrier and HOV/Lane during Peak Periods	1_27
		1423	Alternative 2 – Convert Existing Outside Shoulders to	1-21
		1.4.2.0	HOV Use during Peak Periods (Part-Time Use Lane)	1-30
		1424	Alternative 3A – Convert Existing Outside Shoulders to	1 00
		1.7.2.7	HOV in Each Direction (Four-Lane Facility)	1-32
		1425	Alternative 3B – Widen Highway to Add a Full-time HOV	1 02
		1.1.2.0	lane in each Direction (Four-Lane Facility), 8-Foot	
			Outside Shoulders, Widen Sonoma Creek Bridge	1-34
	1.4.3	Alternati	ve Selection Process	1-35
		1.4.3.1	Identification of the Preferred Alternative	1-36
		1.4.3.2	Alternatives Evaluated in the Draft EIR/EA but	
		-	Eliminated from Further Consideration following Public	
			and Agency Review and Comment	1-38
		1.4.3.3	Alternatives Considered but Eliminated During Early	
			Development of the Project	1-41
1.5	Other P	roject Fe	atures	1-43
1.6	Permits	and App	provals Needed	1-43
Chapte	r 2 Affe	cted Env	vironment: Environmental Consequences: and	
Avoida	nce, Mir	nimizatio	on, and/or Mitigation Measures	2-1

i

2.1	Topics	Consider	red but Determined Not to be Relevant	2-1				
2.2	Human	Environ	ment	2-1				
	2.2.1	Existing	and Future Land Use	2-1				
		2.2.1.1	Affected Environment	2-1				
		2.2.1.2	Environmental Consequences	2-2				
		2.2.1.3	Avoidance, Minimization, and/or Mitigation Measures	2-3				
	2.2.2	Consistency with State, Regional, and Local Plans and						
		Program	ns	2-3				
		2.2.2.1	Affected Environment	2-3				
		2.2.2.2	Environmental Consequences	2-7				
		2.2.2.3	Avoidance, Minimization, and/or Mitigation Measures	2-8				
	2.2.3	Coastal	Zone	2-16				
		2.2.3.1	Regulatory Setting	2-16				
		2.2.3.2	Affected Environment	2-16				
		2.2.3.3	Environmental Consequences	2-24				
		2.2.3.4	Avoidance, Minimization, and/or Mitigation Measures	2-24				
	2.2.4	Parks a	nd Recreational Facilities	2-24				
		2.2.4.1	Regulatory Setting	2-24				
		2.2.4.2	Affected Environment	2-25				
		2.2.4.3	Environmental Consequences	2-25				
		2.2.4.4	Avoidance, Minimization, and/or Mitigation Measures	2-26				
	2.2.5	Farmlar	nds	2-26				
		2.2.5.1	Regulatory Setting	2-26				
		2.2.5.2	Affected Environment	2-27				
		2.2.5.3	Environmental Consequences	2-28				
		2.2.5.4	Avoidance, Minimization, and/or Mitigation Measures	2-28				
	2.2.6	Growth.		2-28				
		2.2.6.1	Regulatory Setting	2-28				
		2.2.6.2	Affected Environment	2-28				
		2.2.6.3	Environmental Consequences	2-30				
		2.2.6.4	Avoidance, Minimization, and/or Mitigation Measures	2-30				
	2.2.7	Commu	nity Character and Cohesion	2-30				
		2.2.7.1	Regulatory Setting	2-30				
		2.2.7.2	Affected Environment	2-31				
		2.2.7.3	Environmental Consequences	2-33				
		2.2.7.4	Avoidance, Minimization, and/or Mitigation Measures	2-34				
	2.2.8	Relocat	ions and Real Property Acquisition	2-34				
		2.2.8.1	Regulatory Setting	2-34				
		2.2.8.2	Affected Environment	2-35				
		2.2.8.3	Environmental Consequences	2-35				
		2.2.8.4	Avoidance, Minimization, and/or Mitigation Measures	2-35				
	2.2.9	Environ	mental Justice	2-39				
		2.2.9.1	Regulatory Setting	2-39				
		2.2.9.2	Affected Environment	2-39				
		2.2.9.3	Environmental Consequences	2-40				
		2.2.9.4	Avoidance, Minimization, and/or Mitigation Measures	2-41				
	2.2.10	Utilities/Emergency Services	2-41					
-----	---------	--	-------					
		2.2.10.1 Environmental Consequences	2-42					
		2.2.10.2 Avoidance, Minimization, and/or Mitigation Measures	2-42					
	2.2.11	Traffic and Transportation/Pedestrian and Bicycle Facilities	2-42					
		2.2.11.1 Regulatory Setting	2-42					
		2.2.11.2 Affected Environment	2-43					
		2.2.11.3 Environmental Consequences	2-48					
		2.2.11.4 Avoidance, Minimization, and/or Mitigation Measures	2-55					
	2.2.12	Visual/Aesthetics	2-55					
		2.2.12.1 Regulatory Setting	2-55					
		2.2.12.2 Affected Environment	2-56					
		2.2.12.3 Environmental Consequences	2-58					
		2.2.12.4 Avoidance. Minimization. and/or Mitigation Measures	2-76					
	2.2.13	Cultural Resources	2-77					
		2.2.13.1 Regulatory Setting	2-77					
		2 2 13 2 Affected Environment	2-78					
		2 2 13 3 Environmental Consequences	2-80					
		2.2.13.4 Avoidance Minimization and/or Mitigation Measures	2-80					
23	Physica	al Environment	2-80					
2.0	2.3.1	Hydrology and Floodplain	2-80					
	2.0.1	2.3.1.1 Regulatory Setting	2-80					
		2312 Affected Environment	2-81					
		2.3.1.3 Environmental Consequences	2_82					
		2.3.1.4 Avoidance Minimization and/or Mitigation Measures	2_83					
	232	Water Quality and Stormwater Runoff	2-83					
	2.0.2	2.3.2.1 Regulatory Setting	2_83					
		2.3.2.1 Affected Environment	2 87					
		2.3.2.2 Allected Environmental Consequences	2 02					
		2.3.2.4 Avoidance Minimization and/or Mitigation Measures	2-93					
	222	2.5.2.4 Avoluance, Minimization, and/or Miligation Measures	2-97					
	2.3.3	2.2.2.1 Degulatory Sotting	2-97					
		2.3.3.1 Regulatory Setting						
		2.3.3.2 Allected Environmental Canada Superson						
		2.3.3.3 Environmental Consequences						
	004	2.3.3.4 Avoidance, Minimization, and/or Miligation Measures						
	2.3.4	Paleontology						
		2.3.4.2 Affected Environment	2-101					
		2.3.4.3 Environmental Consequences	2-101					
		2.3.4.4 Avoidance, Minimization, and/or Mitigation Measures	2-102					
	2.3.5	Hazardous Waste/Materials	2-102					
		2.3.5.1 Regulatory Setting	2-102					
		2.3.5.2 Affected Environment	2-103					
		2.3.5.3 Environmental Consequences	2-108					
		2.3.5.4 Avoidance, Minimization, and/or Mitigation Measures	2-109					
	2.3.6	Air Quality	2-109					
		2.3.6.1 Regulatory Setting	2-109					

		2.3.6.2	Affected Environment	2-111
		2.3.6.3	Environmental Consequences	2-114
		2.3.6.4	Avoidance, Minimization, and/or Mitigation Measures	2-123
		2.3.6.5	Climate Change	2-123
	2.3.7	Noise ar	nd Vibration	2-123
		2.3.7.1	Regulatory Setting	2-123
		2.3.7.2	Affected Environment	2-126
		2.3.7.3	Environmental Consequences	2-133
		2.3.7.4	Avoidance, Minimization, and/or Mitigation Measures	2-140
	2.3.8	Energy.	-	2-140
		2.3.8.1	Regulatory Setting	2-140
		2.3.8.2	Affected Environment	2-140
		2.3.8.3	Environmental Consequences	2-141
		2.3.8.4	Avoidance, Minimization, and/or Mitigation Measures	2-145
2.4	Biologio	cal Enviro	onment	2-145
	2.4.1	Natural	Communities	2-145
		2.4.1.1	Affected Environment	2-147
		2.4.1.2	Environmental Consequences	2-148
		2.4.1.3	Avoidance, Minimization, and/or Mitigation Measures	2-151
	2.4.2	Wetland	s and Other Waters	2-152
		2.4.2.1	Regulatory Setting	2-152
		2.4.2.2	Affected Environment	2-154
		2.4.2.3	Environmental Consequences	2-155
		2.4.2.4	Wetlands Only Practicable Alternative Finding	2-157
		2.4.2.5	Avoidance, Minimization, and/or Mitigation Measures	2-158
	2.4.3	Plant Sp	pecies	2-159
		2.4.3.1	Regulatory Setting	2-159
		2.4.3.2	Affected Environment	2-160
		2.4.3.3	Environmental Consequences	2-163
		2.4.3.4	Avoidance, Minimization, and/or Mitigation Measures.	2-163
	2.4.4	Animal S	Species	2-163
		2.4.4.1	Regulatory Setting	2-163
		2.4.4.2	Affected Environment	2-164
		2.4.4.3	Environmental Consequences	2-165
		2.4.4.4	Avoidance, Minimization, and/or Mitigation Measures	2-170
	2.4.5	Threater	ned and Endangered Species	2-173
		2.4.5.1	Regulatory Setting	2-173
		2.4.5.2	Summary of Affected Environment and Effect	
			Determinations	2-1/4
		2.4.5.3	Affected Environment: California Red Legged Frog	2-1/6
		2.4.5.4	Environmental Consequences: California Red-Legged	0 4 7 0
		0 4 5 5	Frog	2-178
		2.4.5.5	Avoidance, Minimization, and/or Mitigation Measures:	0.470
		0450		
		2.4.5.6	Affected Environment: Ridgway's Rail	2-183
		2.4.5.7	Environmental Consequences: Ridgway's Rail	2-184

	2.4.5.8	Avoidance, Minimization, and/or Compensation: Ridoway's Rail	2-185
	2459	Affected Environment: Salt Marsh Harvest Mouse	2-187
	2.4.5.10	Environmental Consequences: Salt Marsh Harvest	2 101
		Mouse	2-188
	2.4.5.11	Avoidance, Minimization, and/or Compensation: Salt	
		Marsh Harvest Mouse	2-190
	2.4.5.12	Affected Environment: Chinook Salmon	2-193
	2.4.5.13	Environmental Consequences: Chinook Salmon	2-194
	2.4.5.14	Avoidance, Minimization, and/or Compensation:	-
		Chinook Salmon	2-197
	2.4.5.15	Affected Environment: Steelhead Central California	
		Coast and California Central Valley Distinct Population	
		Segments	2-201
	2.4.5.16	Environmental Consequences: Steelhead	2-202
	2.4.5.17	Avoidance, Minimization, and/or Compensation:	
		Steelhead	2-202
	2.4.5.18	Affected Environment: North American Green Sturgeon	2-202
	2.4.5.19	Environmental Consequences: Green Sturgeon	2-203
	2.4.5.20	Avoidance, Minimization, and/or Compensation: Green	
		Sturgeon	2-203
	2.4.5.21	Affected Environment: Delta Smelt	2-203
	2.4.5.22	Environmental Consequences: Delta Smelt	2-204
	2.4.5.23	Avoidance, Minimization, and/or Compensation: Delta	
		Smelt	2-204
	2.4.5.24	Affected Environment: Soft Bird's-Beak	2-204
	2.4.5.25	Environmental Consequences: Soft Bird's-Beak	2-204
	2.4.5.26	Avoidance, Minimization, and/or Compensation: Soft	
		Bird's-Beak	2-205
	2.4.5.27	Affected Environment: Longfin Smelt	2-205
	2.4.5.28	Environmental Consequences: Longfin Smelt	2-205
	2.4.5.29	Avoidance Minimization and/or Compensation: Longfin	
	0 4 5 00		2-205
	2.4.5.30	Affected Environment California Black Rail	2-206
	2.4.5.31	Environmental Consequences: California Black Rail	2-206
	2.4.5.32	Avoidance, Minimization, and/or Compensation:	0 000
	0 4 5 00		2-206
	2.4.5.33	Affected Environment: Swainson's Hawk	2-206
	2.4.5.34	Environmental Consequences: Swainson's Hawk	2-207
	2.4.5.35	Avoidance, Minimization, and/or Miligation Measures.	2 207
246	Facantia	J Fish Habitat	2-207
2.4.0		n Fish nabilal Pegulatony Setting	2-200
	2.4.0.1 2162	Affected Environment	2-200 2_208
	2.4.U.Z	Environmental Consequences	2-200 2_208
	2.4.0.J 2161	Avoidance Minimization and/or Mitigation Maggures	2-200
	2.4.0.4	Avoluance, withinization, and/or withgation weasures	2-209

	2.4.7	Invasive Species	.2-209
		2.4.7.1 Regulatory Setting	.2-209
		2.4.7.2 Affected Environment	.2-209
		2.4.7.3 Environmental Consequences	.2-211
		2.4.7.4 Avoidance. Minimization. and/or Mitigation Measures	.2-211
2.5	Cumula	ative Impacts	2-212
	251	Regulatory Setting	2-212
	252	Cumulative Impact Analysis	2-212
	253	Resource Areas with No Contribution to Cumulative Effects	2-214
	254	Resources Considered for Contribution to Cumulative Effects	2-214
	2.0.1	2.5.4.1 Biological Resources	2-214
		2542 Visual/Aesthetics	2-216
		2543 Transportation	2-217
			. 2 2 17
Chapte	r 3 Cali	fornia Environmental Quality Act Evaluation	3-1
3.1	Determ	ining Significance under CEQA	3-1
3.2	Significa	ant Irreversible Environmental Changes	3-1
3.3	CEQA E	Environmental Checklist	3-2
	3.3.1	Aesthetics	3-3
		3.3.1.1 CEQA Significance Determinations for Aesthetics	3-3
	3.3.2	Agriculture and Forest Resources	3-5
		3.3.2.1 CEQA Significance Determinations for Agriculture and	
		Forest Resources	3-6
	3.3.3	Air Quality	3-7
		3.3.3.1 CEQA Significance Determinations for Air Quality	3-7
	3.3.4	Biological Resources	3-9
		3.3.4.1 CEQA Significance Determinations for Biological	
		Resources	3-9
	3.3.5	Cultural Resources	3-19
		3.3.5.1 CEQA Significance Determinations for Cultural	
		Resources	3-19
	3.3.6	Energy	3-20
		3.3.6.1 CEQA Significance Determinations for Energy	3-20
	3.3.7	Geology and Soils	3-21
		3.3.7.1 CEQA Significance Determinations for Geology and	
		Soils	3-22
	3.3.8	Greenhouse Gas Emissions	3-24
		3.3.8.1 CEQA Significance Determinations for Greenhouse Ga	S
		Emissions	3-24
	3.3.9	Hazards and Hazardous Materials	3-26
		3.3.9.1 CEQA Significance Determinations for Hazards and	
		Hazardous Materials	3-27
	3.3.10	Hydrology and Water Quality	3-29
		3.3.10.1 CEQA Significance Determinations for Hydrology and	
		Water Quality	3-30
	3.3.11	Land Use and Planning	3-33

		3.3.11.1 CEQA Significance Determinations for Land Use and	
		Planning	3
	3.3.12	Mineral Resources	4
		3.3.12.1 CEQA Significance Determinations for Mineral	
		Resources	4
	3.3.13		4
	2211	3.3.13.1 CEQA Significance Determinations for Noise	с С
	3.3.14	3 3 1/ 1 CEOA Significance Determinations for Population and	C
		Housing 3-3	6
	3 3 15	Public Services 3-3	7
	010110	3.3.15.1 CEQA Significance Determinations for Public Services 3-3	7
	3.3.16	Recreation	З
		3.3.16.1 CEQA Significance Determinations for Recreation	3
	3.3.17	Transportation	9
		3.3.17.1 CEQA Significance Determinations for Transportation3-3	9
	3.3.18	Tribal Cultural Resources	)
		3.3.18.1 CEQA Significance Determinations for Tribal Cultural	~
	0 0 4 0	Resources	)
	3.3.19	Utilities and Service Systems	I
		Service Systems	1
	3 3 20	Wildfire 3-5	2
	0.0.20	3.3.20.1 CEQA Significance Determinations for Wildfire	2
	3.3.21	Mandatory Findings of Significance	4
		3.3.21.1 CEQA Significance Determinations for Mandatory	
		Findings of Significance	5
3.4	Climate	Change	6
	3.4.1	Regulatory Setting	5
		3.4.1.1 Federal	5
	040	3.4.1.2 State	1
	3.4.Z	Environmental Setting	ป ว
	3.4.3	3/31 Operational Emissions	2 1
		3432 Construction Emissions 3-6	R
		3.4.3.3 CEQA Conclusion	B
	3.4.4	Greenhouse Gas Reduction Strategies	9
		3.4.4.1 Statewide Efforts	9
		3.4.4.2 Caltrans Activities	1
	3.4.5	Adaptation	2
		3.4.5.1 Federal Efforts	2
		3.4.5.2 State Efforts	3
		3.4.5.3 Caltrans Adaptation Efforts	С
Chapte	r 4 Com	ments and Coordination4-	I
4.1	Corrido	-4-	1
4.2	Public S	Scoping Process4-	1

4.3	Project	and Stak	ceholder Coordination	4-2
4.4	Consul	tation and	d Coordination with Public Agencies	4-3
	4.4.1	Federal	Highway Administration	4-3
	4.4.2	United S	States Army Corps of Engineers	4-4
	4.4.3	United S	States Fish and Wildlife Service and National Marine	
		Fisherie	s Service	4-4
	4.4.4	Tribal E	ntities	4-5
	4.4.5	State Ag	gencies	4-6
		4.4.5.1	State Historic Preservation Officer	4-6
		4.4.5.2	California Department of Fish and Wildlife	4-6
	4.4.6	Regiona	al Agencies	4-6
		4.4.6.1	Metropolitan Transportation Commission (Air Quality	
			Conformity)	4-6
		4.4.6.2	San Francisco Bay RWQCB	4-7
		4.4.6.3	Bay Conservation Development Commission	4-7
4.5	Circula	tion, Rev	iew, and Comments on the Draft Environmental	
	Docum	ent		4-7
Chapte	er 5 List	of Prepa	arers	5-1
Chapte	er 6 Dist	tribution	List	6-1
Chapte	er 7 Ref	erences .		7-1
•				

# Appendices (included in Volume 2)

# Appendix A. Map Book

# Appendix B. Section 4(f)

# Appendix C. Title VI Policy Statement

# Appendix D. Consultation and Coordination

 PM<sub>2.5</sub> Interagency Consultation, MTC Air Quality Conformity Task Force Determination
SHPO Concurrence
Native American Heritage Commission Coordination

# Appendix E. Environmental Commitments Record (ECR)

# Appendix F. List of Acronyms and Abbreviations

# Appendix G. Notice of Preparation and Scoping Summary

# Appendix H. Intersection Levels of Service

# Appendix I. USFWS and NOAA Fisheries Service Species Lists

# Appendix J. List of Technical Studies

**Appendix K. Comments and Responses** (under separate cover, included in Volume 3)

viii

# Figures

Figure 1-1	Project Location and Vicinity Map	1-3
Figure 1-2	Typical Cross Section Alternative 3B	. 1-15
Figure 1-3	Typical Cross Section of Alternative 1	. 1-27
Figure 1-4	Typical Cross Section of Alternative 2	. 1-30
Figure 2-1	Existing Conditions from KV-1	. 2-60
Figure 2-2	Simulated Condition for Alternative 1 from KV-1	. 2-61
Figure 2-3	Simulated Condition for Alternatives 2 or 3A from KV-1 with the	
U	Type 85B Outside Barrier	. 2-63
Figure 2-4	Simulated Condition for Alternative 3B from KV-1	2-64
Figure 2-5	Existing Condition at KV-2	
Figure 2-6	Simulated Condition KV-2 – Alternative 1	
Figure 2-7	Simulated Condition KV-2 – Alternatives 2, 3A, and 3B	
Figure 2-8	Existing Condition KV-3	. 2-68
Figure 2-9	Simulated Condition KV-3. Alternative 1 with MGS Outside Barrier	
Figure 2-10	Simulated Condition KV-3. Alternatives 2 or 3A with MGS Outside	
5	Barrier	
Figure 2-11	Simulated Condition KV-3. Alternative 3B with Type 85B Outside	
5	Barrier	
Figure 2-12	Existing Condition KV-4	
Figure 2-13	Existing Condition KV-5	.2-73
Figure 2-14	Simulated View KV-5 – Alternative 1	
Figure 2-15	Simulated View KV-5 – Alternatives 2 or 3A	2-75
Figure 2-16	Simulated View KV-5 – Alternative 3B	
Figure 2-17	Site Geology	.2-89
Figure 2-18	Noise Levels of Common Activities	2-125
Figure 2-19	Noise Measurements and Evaluated Barriers	2-129
Figure 3-1	SR 37 Potential Express Bus Service Connecting Transit Services.	
Figure 3-2	Existing Park-and-Ride Lots in Vicinity of SR 37	.3-46
Figure 3-3	Park-and-Ride Opportunities in Vicinity of SR 37	3-47
Figure 3-4	U.S. 2019 Greenhouse Gas Emissions	3-60
Figure 3-5	California 2018 Greenhouse Gas Emissions by Economic Sector	3-61
Figure 3-6	Change in California GDP. Population, and GHG Emissions Since	
i igui e e e	2000	3-62
Figure 3-7	Possible Use of Traffic Operation Strategies in Reducing On-road	
rigare e r	CO <sub>2</sub> Emissions	3-65
Figure 3-8	California Climate Strategy	3-70
Figure 3-9	Areas of Flood Impact to Highway Under Existing Conditions with	
r igure e e	a 1-Year Storm Scenario	3-78
Figure 3-10	Areas of Flood Impact to Highway Under Existing Conditions with	
rigare e re	a 10-Year Storm Scenario	3-79
Figure 3-11	Areas of Flood Impact to Highway Under Existing Conditions with	
. igaio o TT	a 50-Year Storm Scenario	3-80
Figure 3-12	Areas of Flood Impact to Highway Under Existing Conditions with	
	a 100-Year Storm Scenario	3-81

Figure 3-13	Areas of Flood Impact to Highway Under 1.9 Feet of SLR	
	Conditions with a 1-Year Storm Scenario	3-83
Figure 3-14	Areas of Flood Impact to Highway Under 1.9 Feet of SLR	
-	Conditions with a 10-Year Storm Scenario	3-84
Figure 3-15	Areas of Flood Impact to Highway Under 1.9 Feet of SLR	
-	Conditions with a 50-Year Storm Scenario	3-85

# Tables

Table S-1	Summary of Impacts and Avoidance, Minimization, and/or	
	Mitigation Measures	S-4
Table 1-1	Other Studies or Proposed Projects within the SR 37 Corridor	1-7
Table 1-2	SR 37 Peak Direction Travel Time Comparisons (Between	
	U.S. 101 and SR 29) – AM Peak Period	1-10
Table 1-3	SR 37 Peak Direction Travel Time Comparisons (Between	
	U.S. 101 and SR 29) – PM Peak Period	1-10
Table 1-4	Other Project Features	1-44
Table 1-5	Project Permits and Approvals	1-52
Table 2-1	Current and Proposed Planned Developments within One Mile of	
	the Project Area	2-2
Table 2-2	Consistency of No Build and Build Alternatives with General	
	Plans, Regional Plans, and Transportation Plans	2-9
Table 2-3	BCDC Bay Plan Maps 1 and 2 Policies	2-18
Table 2-4	Consistency with BCDC Bay Plan Transportation Policies by	
	Alternative	2-19
Table 2-5	Publicly Owned Parks and Recreation Facilities	2-25
Table 2-6	Population Change in the Project Area	2-29
Table 2-7	Temporary Construction Easements	2-36
Table 2-8	Permanent Partial Acquisition	2-38
Table 2-9	Summary of Race, Ethnicity, and Poverty Status in the Study Area	
	and Reference Areas	2-40
Table 2-10	Intersection LOS Thresholds	2-45
Table 2-11	VMT, VHD, and VHT for 2020 (Existing Conditions)	2-52
Table 2-12	VMT, VHD, and VHT Estimates of No Build and Build Alternatives	
	for 2025 (no tolling)	2-52
Table 2-13	VMT, VHD, and VHT Estimates of No Build and Build Alternatives	
	for 2045 (no tolling)	2-53
Table 2-14	VMT Estimates of No Build and Build Alternatives with Tolling	2-54
Table 2-15	Summary of Key View Narrative Ratings	2-76
Table 2-16	Beneficial Uses for the Receiving Waters in the Project Area	2-91
Table 2-17	Existing and Potential Beneficial Uses in Groundwater Basins	2-91
Table 2-18	CWA 303(d) List Impairments and TMDLs at Water Bodies in the	
	Project Area	2-92
Table 2-19	Disturbed Soil and Added Impervious Areas by Build Alternative	2-94
Table 2-20	State and Federal Attainment Status	2-112

Table 2-21	Air Quality Concentrations for 2015-2019	.2-115
Table 2-22	Construction Emissions (Per Day or Phase)	.2-119
Table 2-23	Construction Emissions (Total Tons)	.2-119
Table 2-24	Summary of Comparative Emissions Analysis in Pounds Per Day.	.2-120
Table 2-25	Summary of Comparative MSAT Emissions Analysis	.2-122
Table 2-26	Noise Abatement Criteria	.2-124
Table 2-27	Summary of Long-Term Noise Measurements	.2-127
Table 2-28	Summary of Short-Term Noise Measurements	.2-128
Table 2-29	Calculated Noise Measurements by Alternative	.2-135
Table 2-30	Calculated Noise Measurements by Phase	.2-137
Table 2-31	Vibration Source Levels for Construction Equipment	.2-139
Table 2-32	Fossil Fuel Consumption in California for the Transportation	
	Sector (2018)	.2-141
Table 2-33	Direct Energy Factors for Energy Consumption	.2-142
Table 2-34	Operational Daily Fuel Consumption for the Study Area	.2-143
Table 2-35	Daily Vehicle Hours of Delay in the Traffic Study Area	.2-144
Table 2-36	Construction CO <sub>2</sub> Emissions/Energy Usage	.2-145
Table 2-37	Crossing Information in the PAD	. 2-147
Table 2-38	Estimated Permanent Effects on Sensitive Marsh and Wetland	
	Communities	.2-149
Table 2-39	Preliminary Estimated Permanent Impacts to Wetland and Other	
	Waters	.2-156
Table 2-40	Preliminary Estimated Temporary Impacts to Wetlands and Other	
	Waters	.2-156
Table 2-41	Plant Survey Dates	.2-160
Table 2-42	Estimated Permanent Fill Impacts to California Red-Legged Frog	
	Habitat in the BSA	.2-178
Table 2-43	Estimated Temporary Impacts to California Red-Legged Frog	
	Habitat in the BSA	.2-178
Table 2-44	Estimated Permanent Impacts to Ridgway's Rail Habitat in the	
	BSA	.2-184
Table 2-45	Estimated Temporary Impact Areas to Ridgway's Rail Habitat in	~
T 1 0 10		.2-184
Table 2-46	Alternative 3B Shading Impacts to Ridgway's Rail Habitat in the	0 405
T I I 0 47		.2-185
1 able 2-47	Estimated Permanent Impacts to Salt Marsh Harvest Mouse	0 4 0 0
	Habitat In the BSA	.2-189
Table 2-48	Estimated Temporary Impact Areas to Salt Marsh Harvest Mouse	0 400
Table 2.40	Habilal In the BSA	.2-189
Table 2-49	Alternative 3B Shading Impacts to Sait Marsh Harvest Mouse	2 100
Table 2 50	Estimated Dermanant Impacts to Anadromous Fish Lightet in the	. 2-190
1 able 2-30		2 105
Table 2 51	Estimated Temporary Impact Areas to Anadromous Eish Habitat	. 2-190
	in the BSA	2_105
		. 2-130

Table 2-52	Alternative 3B Estimated Underwater Sound Pressure Effects on Fish Habitat from Impact Pile Driving 2-196	5
Table 2-53	Alternative 3B Shading Impacts to Anadromous Fish Habitat in the BSA	7
Table 2-54	Cal-IPC High- and Moderate-Rated Species Observed in the BSA2-210	)
Table 2-55	Current and Proposed Planned Developments within One Mile of	
	the Project Area2-213	3
Table 2-56	Current and Proposed Planned Developments within One Mile of	
	the Project Area2-214	1
Table 3-1	Estimated Permanent Habitat Impacts to Listed Species Habitats 3-10	)
Table 3-2	Estimated Temporary Impacts Areas in Listed Species Habitats 3-10	)
Table 3-3	Summary of VMT and Vehicle Trip Increases by Alternative (no	
	tolling)	)
Table 3-4	VMT Reduction with the Preferred Alternative (HOV Lanes and	
	Tolling)	1
Table 3-5	Regional and Local Greenhouse Gas Reduction Plans	3
Table 3-6	Modeled Annual CO <sub>2</sub> e Emissions and VMT by Alternative	7
Table 3-7	Summary of State of California SLR Projections for San Francisco	
	Bay	3
Table 4-1	Overall Topics Raised in Public Scoping Comments	2

# Chapter 1 Proposed Project

#### 1.1 Introduction

The California Department of Transportation (Caltrans)—in cooperation with the Metropolitan Transportation Commission (MTC) and the north bay partner agencies of Sonoma County Transportation Authority (SCTA), Solano Transportation Authority (STA), and Napa Valley Transportation Authority (NVTA)—is proposing improvements to the portion of State Route (SR) 37 that is one lane in each direction. Caltrans is the lead agency under CEQA and NEPA.

The section of SR 37 proposed for widening is a two-lane conventional highway between SR 121 and the Mare Island Interchange (approximately 10 miles). East of the Mare Island, Interchange SR 37 is a four-lane freeway facility between Interstate 80 (I-80) and Mare Island (approximately 4.5 miles). West of SR 121, it is a four-lane conventional highway between SR 121 and United States Highway 101 (U.S. 101) (approximately 7.3 miles). The highway allows local road and driveway access from adjoining properties, has acceleration and deceleration lanes at some local intersections, and has an existing median barrier along most of the route. Bridge crossings are at Sonoma and Tolay creeks.

The project is within or adjoins the counties of Sonoma, Solano, and Napa and would add a high-occupancy vehicle (HOV) lane(s) from just west of SR 121 to the Mare Island (Walnut Avenue) Overcrossing, within the portion of SR 37 that is currently served by only one lane in each direction. Four alternatives were considered; Alternative 3B was identified as the preferred alternative following circulation of the Draft Environmental Impact Report (EIR)/Environmental Assessment (EA), and consideration of comments received. Each of the four Build Alternatives considered would reconfigure the existing SR 37 highway lanes from west of the SR 121 intersection to the Walnut Avenue Overcrossing at Mare Island. Each alternative would involve widening the existing bridge over Tolay Creek.

The overarching design and traffic differences between the four alternatives are presented below:

- Alternative 3B (preferred alternative) would provide four lanes, with two full-time lanes in each direction, and include the construction of 8-foot outside shoulders between Mare Island and SR 121, and widening of the Sonoma Creek Bridge. The existing lane in each direction between SR 121 and Mare Island would convert to HOV lanes. The added lane in each direction would be for general purpose use, and would be tolled.
  - Three HOV scenarios were considered for Alternative 3B, and are described below. Scenario 3 was the selected scenario under the preferred alternative. Scenario 3 introduces the HOV lane on the left side, about 0.6 mile west of SR 121, and extends the eastbound left-turn lane approximately 0.5 mile west.

- Alternative 1 would have provided three lanes during the peak traffic periods—two in the peak period flow direction and one in the non-peak flow direction—using a movable center median barrier from the Noble Road intersection to just west of Mare Island interchange. The additional lane would be an HOV lane. During nonpeak periods the highway would remain as one lane in each direction within this section. SR 37 from SR 121 to Noble Road would be permanently widened from one lane in each direction to two lanes in each direction with a solid median barrier. Alternative 1 was not carried forward following identification of Alternative 3B as the preferred alternative.
- Alternative 2 would have provided three lanes in the peak period only by allowing traffic to access the outside shoulders for use as a traffic lane in the peak direction, but with a fixed median barrier. During non-peak periods the highway would remain as one lane in each direction. The additional lane would be an HOV lane. Alternative 2 was carried forward following identification of Alternative 3B as the preferred alternative.
- Alternative 3A would provide four lanes, with two full-time lanes in each direction. Alternative 3A would include the construction of 4-foot outside shoulders between Mare Island and SR 121, except at the Sonoma Creek Bridge, where existing shoulder widths would be reduced in both directions to avoid bridge widening. The HOV scenarios described below were considered for this alternative. Alternative 3A was not carried forward following identification of Alternative 3B as the preferred alternative.
  - Scenario 1 introduces the HOV lane on the right side, approximately 0.6 mile west of SR 121.
  - Scenario 2 introduces the HOV lane on the left side east of SR 121, in the vicinity of the Tolay Creek Bridge.
  - Scenario 3 introduces the HOV lane on the left side, about 0.6 mile west of SR 121, and extends the eastbound left-turn lane approximately 0.5 mile west.

All alternatives would also involve installation of advance warning signs to alert drivers approaching the proposed HOV and toll lanes. The preferred alternative, Alternative 3B, proposes the installation and operation of Open Road Tolling (ORT) on the proposed general purpose lanes. Figure 1-1 shows a view of the overall project vicinity and study limits.



SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

# FIGURE 1-1: PROJECT LOCATION

This page intentionally left blank

February 2023

HOV lanes, also known as carpool or diamond lanes, are a traffic management strategy used to promote and encourage ridesharing, which helps alleviate congestion and maximizes the people-carrying capacity of a highway. HOV or carpool vehicles may be configured as HOV 2+ (two or more occupants), HOV3+ (three or more occupants), or other. The option of using tolling allows for differential pricing that can further encourage the use of multi-occupant vehicles, such as through a discounted cost or no-charge cost for HOVs. A discounted means-based (income based) toll is also proposed. The application of tolling would require California Transportation Commission (CTC) approval, and is evaluated in this environmental document as a component of the project.

Plan Bay Area 2050, the regional transportation plan (RTP), includes a Freeway Performance Program (RTP ID 21-T06-035) for SR 37, which was used for developing a corridor plan and funding preliminary studies to identify projects that address transportation congestion, sea-level rise (SLR) adaptation, and flooding. This proposed project was identified as the Interim Project to address traffic congestion that occurs where SR 37 narrows to one lane in each direction. The project would be included in a Transportation Improvement Program (TIP) for adoption by the MTC.

The RTP can be found on the following website: https://www.planbayarea.org/final plan2050. The project listing can be found on the following website, at page 4 of 14 (RTP ID # 21-T06-035): https://www.planbayarea.org/sites/default/files/documents/ Plan\_Bay\_Area\_2050\_Transportation\_Project\_List\_October\_2021.pdf. The TIP ID #VAR210004 can be found on the following website: https://fms.mtc.ca.gov/fms/ showProjectDetailPrepare.ds?projectVersionSeq=34482. The TIP listing is also included in Appendix D, with the MTC Air Quality Conformity Task Force Determination.

# 1.1.1 History and Other Corridor Projects

The SR 37 corridor has been the subject of several studies related to SLR and traffic congestion. These include the Highway 37 Stewardship Study (July 2013); the State Route 37 Integrated Traffic, Infrastructure, and Sea Level Rise Analysis (UC Davis Study, February 2016); and the Caltrans Transportation Concept Report (2015). These studies and the SR 37 Corridor Plan identified SR 37 between SR 121 and the Mare Island Interchange as a priority segment for capacity enhancement to address bottlenecks that extend at either end due to lane drops between the four-lane segments; and to address the vulnerability and risks associated with SLR and flooding, public safety, transit feasibility, recreational activities, economic impacts on commuters, and transport of goods.

In December 2019, a Project Study Report-Project Development Support (PSR-PDS) was approved that included the following alternatives:

- No Build
- Interim Build Alternatives (previously referred to as the SR 37 Congestion Relief Project and now the Highway 37 Sears Point to Mare Island Improvement Project)
  Three lane movable median barrier
  - $\circ$   $\,$  Three-lane movable median barrier

- Shoulder conversion to travel lane during peak hour
- Ultimate Build Alternatives (hereafter called the Ultimate Project)
  - Hybrid Section
  - Causeway

The interim and Ultimate projects were identified in the PSR-PDS report as separate projects based on their individual purpose and need, priority, and lead time to gain funding to proceed. The purpose of the proposed project is to address existing recurring congestion on SR 37 where the highway narrows to one lane in each direction between SR 121 and Mare Island. This will be accomplished by improving traffic flow and peak travel times, and increasing vehicle occupancy (the number of people moved per vehicle).

The purpose of the future Ultimate Project is to maintain traffic conditions, address the resiliency of the corridor with respect to SLR and flooding, and provide ecological and hydrologic enhancements. The future Ultimate Project to address SLR would require a substantial long-term investment to fund the reconstruction of the highway's vertical profile and improve hydrologic connectivity with the Bay. The anticipated years of construction needed to build a project that addresses SLR means that the SR 37 highway must remain in place and intact until completion of any improvements that raise the highway, so that transportation access along this corridor is not substantially interrupted during long-term construction.

A number of other transportation studies or projects are planned or anticipated along SR 37 that address the need for separate safety improvements, roadway or bridge repair and maintenance, drainage/flooding, and traffic operations (Table 1-1). There are two highway safety and operation projects, the SR 37/SR 121 intersection improvement (Caltrans project EA 1Q480) and SR 37 eastbound lane extension and railroad crossing at Tolay Creek (EA 2Q200), that have been combined into one single project, EA 04-2Q20U. These projects were proposed to reduce vehicle hours of delay at the SR 37/SR 121 intersection. These improvements fall within the footprint of the Sears Point to Mare Island project. This environmental document addresses the potential environmental impacts of these projects, which may be constructed either as an independent operational improvement, or incorporated into the Sears Point to Mare Island project. If these projects were to proceed to construction independent of the Sears Point to Mare Island project, this environmental document addresses their potential environmental impacts. For further information related to the corridor, please visit the Caltrans website at https://dot.ca.gov/caltrans-near-me/district-4/d4-projects/d4-37-corridor-projects.

Project Name/Description	Expenditure Authorization (EA) Number	Project Type	Post Miles	Sponsor	Project Status or Delivery Goal
SR 37 Pavement Rehabilitation – Capital Preventive Maintenance	2K740	Maintenance	R11.2/14.6	Caltrans	Delivery 2024
SR 37 Flood Reduction Project (U.S. 101 to Atherton Avenue Undercrossing)	4Q320	Flood Response/ Resiliency	R11.2/13.7	Caltrans	Delivery 2025
SR 37 Ultimate SLR Resilience Design Alternatives Assessment (U.S. 101 to SR 121)	Not Identified	Traffic Improvement/ Resiliency	R11.2/14.6; 0.0/3.9	MTC	Feasibility Study
SR 37 Petaluma River Bridge Preservation	2Q500	Maintenance	14.5/15.0	Caltrans	Delivery 2024
Reconstruct Intersection of SR 37 and SR 121 (combined into EA 2Q20U)	1Q480	Traffic Improvement	3.8/4.3	Caltrans	Delivery 2024
SR 37 Lane Extension and Railroad Crossing at Tolay Creek (combined into EA 2Q20U)	2Q200	Traffic Improvement	3.8/4.3	Caltrans	Delivery 2024
SR 37 SLR Resilience Project (SR 121 to Mare Island)	1Q762	Traffic Improvement	3.9/6.2; 0.0/R7.4	MTC/CTAs	Feasibility Study
SR 37 Alternatives Assessment for the Ultimate Project (SR 121 to Mare Island Interchange) (completed)	Not Identified	Project and Corridor Planning	3.5/6.2; 0.0/R7.4	MTC	Feasibility Study
SR 37 Pedestrian Enhancements at Wilson Avenue and Fairgrounds Drive (EA)	0P760	Pedestrian/ Bicycle Improvements	Various	Caltrans	Delivery 2024
Fairgrounds Drive Interchange Improvements	4A441	Traffic Improvement	10.6/11.2	STA	Final Design
SR 37 Corridor SLR and Complete Streets (U.S. 101 to SR 29)	4Q960	Project and Corridor Planning	R11.2/14.6; 0.0/6.2; 0.0/ R9.6	Caltrans	Feasibility Study
SR 37 Corridor Planning and Environmental Linkages (PEL) Study (U.S. 101 to I-80)	Not Identified	Project and Corridor Planning	R11.2/14.6; 0.0/6.2; 0.0/ R11.4	Caltrans	Pre- Planning

Source: Caltrans 2020e; SR 37 Policy Committee January 2021

# 1.2 Purpose and Need

#### **1.2.1** Purpose of the Project

The project is intended to address existing recurring congestion on SR 37 where the highway narrows to one lane in each direction between SR 121 and Mare Island. The purpose of the project is to:

1-7

• Improve traffic flow and peak travel times, and

• Increase vehicle occupancy (the number of people moved per vehicle).

# 1.2.2 Project Need

The project is needed to address reoccurring congestion in the near term on SR 37, where the highway narrows to one lane in each direction between SR 37/SR 121. Presently capacity and merging constraints result in traffic queueing at the SR 37/SR 121 intersection. Current and anticipated future transportation demand contribute to the need for this project. The following sections discuss congestion problems in more detail.

# 1.2.2.1 Existing Congestion and Transportation Demand

The information in this section is based on the *Traffic Operations Analysis Report* prepared in December 2021 (AECOM 2021a). Based on traffic observations performed in 2019, westbound SR 37 traffic typically experiences congestion approaching the lane drop west of the Mare Island Interchange for about 6 hours during the weekday AM peak period. Eastbound SR 37 congestion occurs approaching the lane drop east of the SR 121 intersection for about 6 hours during the weekday PM peak period. On typical weekdays, the maximum westbound delay in the AM peak period is about 50 minutes, and the maximum eastbound delay in the PM peak period is about 68 minutes from U.S. 101 to SR 29. Future traffic forecast conditions indicate that the traffic congestion would continue to worsen.

The lane drops and merges on SR 37 affect the operations of the highway where it narrows from two lanes to one lane in each direction (just east of SR 121, and at Mare Island just west of the Napa River Bridge). Lane merges on SR 37 are the major causes of congestion along this segment of the corridor. Within the project limits, faster vehicles cannot pass slower vehicles because there is only one lane in each direction, and no passing is allowed. These capacity and merging constraints result in traffic queueing at the SR 37/SR 121 intersection. In addition, roadway settlement causes traffic to slow near the railroad crossing near SR 121.

# **Existing Conditions**

AM Peak Period: During the weekday AM peak period, a bottleneck occurs on westbound SR 37 between the lane drop west of the Mare Island Interchange and the SR 121 intersection, forming about 5 AM and dissipating about 11 AM. Data and observations collected in 2019 indicated that the longest queue associated with this bottleneck extended approximately 1.2 miles east to the Wilson Avenue Interchange. The bottleneck section for westbound SR 37 had a mainline throughput traffic volume between 1,200 and 1,300 vehicles per hour (vph) per lane, which is below the expected one-way capacity (approximately 1,400 or more vph per lane) for a similar conventional highway. The maximum travel time between SR 29 and U.S. 101 is observed to be approximately 50 minutes during the 6 to 7 AM hour; the minimum travel time is observed to be approximately 25 minutes during the 10 to 11 AM hour. This is a 25-minute difference between the maximum and minimum travel times.

In the AM peak period, vehicles with two or more passengers account for approximately 20 percent of the total vehicle composition in the eastbound direction, and 23 percent in the westbound direction. In the AM peak period, truck volumes account for 10 percent in the eastbound direction and 6 percent in the westbound direction.

PM Peak Period: During the weekday PM peak period, there is a substantial bottleneck on eastbound SR 37 beginning at the lane drop just east of the SR 121 intersection. The mainline queue approaching this bottleneck was observed to extend east of the Petaluma River Bridge, which is approximately 4 miles west of the SR 121 intersection, forming about 2 PM and dissipating about 8 PM. On a typical weekday, the mainline bottleneck throughput for the single eastbound lane peaks at Noble Road at approximately 1,300 vph at 2:00 PM and was observed to be as low as 1,100 vph, compared to a typical capacity of 1,400 vph for a similar conventional highway. The maximum travel time between U.S. 101 and SR 29 is observed to be approximately 68 minutes during the 4 to 5 PM hour; the minimum travel time is observed to be approximately 22 minutes during the 8 to 9 PM hour. This is a 46-minute difference between the maximum and minimum travel times.

During the PM peak period, vehicles with two or more occupants accounted for 23 percent of the total vehicle composition in the eastbound direction, and 18 percent in the westbound direction. In the PM peak period, truck volumes account for 3 percent in the eastbound direction and 4 percent in the westbound direction.

# Method for Forecasting Traffic Volumes

Forecast volumes for the years 2025 and 2045 for the study area were developed using the most current Travel Model One (TM1) V6 model that was developed and is maintained by MTC. The traffic demand inputs for the model were developed using traffic volume data from 2019 counts collected by AECOM. The year 2020 was approved for use as the base year (existing) model which is the closest to the existing condition. The final future traffic forecasts were generated by adding the demand growth estimated by TM1 to existing traffic data (per National Cooperative Research Program Report 255). From 2025 to 2045, the average annual growth rate in the study area is projected to be approximately 0.8 percent per year, as discussed in Section 2.2.6.

These forecasted conditions are based on traffic counts and historic growth rates developed prior to the economic and travel conditions experienced beginning in spring 2020 and are therefore representative of historic commuter demand. These traffic conditions would represent economic recovery in the future based on existing land uses that experience modest growth.

# SR 37 Travel Time Calibration and Validation

The SR 37 Corridor Plan analyzed the traffic operations of the study corridor. The traffic demand inputs for the model were developed using traffic volume data from 2019 counts collected by AECOM. A traffic model called VISSIM was developed for the study limits between U.S. 101 and SR 29 and calibrated to replicate 2019 field conditions collected for the SR 37 Corridor Plan. Once calibrated, the VISSIM model was then applied to evaluate future conditions.

Peak direction travel times for 2019 are summarized in Table 1-2 and Table 1-3.

# Table 1-2 SR 37 Peak Direction Travel Time Comparisons (Between U.S. 101 and<br/>SR 29) – AM Peak Period

Time Interval	Eastbound Field Travel Times	Westbound Travel Times
5 AM	21.4	38.9
6 AM	21.7	49.5
7 AM	22.1	44.4
8 AM	21.9	41.6
9 AM	21.0	28.5
10 AM	21.7	25.0
Average	21.6	37.9

Source: AECOM 2021a

Average of travel time runs collected on October 8, 2019, and October 16, 2019

# Table 1-3 SR 37 Peak Direction Travel Time Comparisons (Between U.S. 101 and<br/>SR 29) – PM Peak Period

Time Interval	Eastbound Field Travel Times in Minutes	Westbound Travel Times in Minutes
2 PM	40.8	22.4
3 PM	56.9	22.7
4 PM	67.8	23.1
5 PM	62.6	22.0
6 PM	49.0	22.6
7 PM	34.4	21.7
8 PM	22.0	21.3
Average	47.6	22.3

Source: AECOM 2021a

Average of travel time runs collected on October 8, 2019, and October 16, 2019

#### 1.2.2.2 Legislation

The following legislation (and proposed legislation) relates to this project:

- Draft legislation was introduced in early 2020 to place SR 37 under the direction of the Bay Area Toll Authority (BATA), an agency that controls tariffs on regional bridges. The intent of this legislation, if approved, would be to raise funding that would be available for long-term improvements on SR 37.
- Assembly Bill (AB) 194 (approved in 2015), Chapter 687 (amending Section 149.7 and 149.12) of the California Streets and Highways Code allows regional transportation agencies and Caltrans to develop and operate express lanes or other tolling facilities. The legislation removes the prior limits on the number of facilities and the approval deadline. The legislation created the

Highway Toll Account in the State Transportation Fund for the management of funds received for toll facilities operated by Caltrans. This legislation allows BATA/Caltrans to designate and operate a toll lane on SR 37 (if it receives other approvals).

- 23 United States Code (USC) 166 provides rules for operation of HOV facilities and HOV facilities by a public authority (such as Caltrans/BATA).
- 23 USC 166 provides a public authority the ability to define the number of occupants in a qualified HOV facility. For SR 37, this can allow or restrict users, such as qualified multi-occupant vehicles, motorcycles, and clean air vehicles, consistent with applicable rules for the State Highway System.
- Senate Bill 1050 MTC SR 37 Toll Bridge Act was proposed in 2022 but did not pass. This bill would have authorized MTC to apply to the CTC for high-occupancy toll lanes or other toll facilities on existing and proposed lanes on SR 37, and create an SR 37 Toll Authority. The proposed project described in this document differs from this original legislative concept by proposing one HOV lane and one new lane in each direction. Tolling will apply to the new tolled lane.

# 1.2.3 Independent Utility and Logical Termini

Federal Highway Administration (FHWA) regulations (23 Code of Federal Regulations [CFR] 771.111[f]) require that the project:

- Have rational end points for a transportation improvement and be of sufficient length for environmental issues to be adequately addressed;
- Be useable and require a reasonable expenditure even if no additional transportation improvements in the area are made; and
- Not restrict consideration of other foreseeable transportation improvements.

The following discussion clarifies how the project meets the above-listed federal regulation requirements.

**Logical Termini:** The existing highway between U.S. 101 and SR 121 (4.5 miles long) and between I-80 and Mare Island (7.3 miles long) is four lanes (two lanes in each direction), while the 9-mile-long segment between SR 121 and Mare Island is two lanes (one lane in each direction). Congestion begins during peak periods, starting before the highway transitions from two lanes in each direction to one lane in each direction. The limits of the project were selected to optimize operational efficiency, and to include proposed features necessary to manage traffic:

• The western terminus of the planned pavement widening begins approximately 0.5 mile west of the SR 121 intersection. This limit was chosen to introduce or start the HOV lane going eastbound west of the SR 121 intersection. This would

allow vehicles to transition into the HOV lane efficiently and safely prior to the SR 121 intersection and Tolay Creek Bridge.

• The eastern terminus where widening begins would be near the Mare Island westbound off-ramp. Currently, the on-ramp transitions quickly, requiring a merge into westbound SR 37. With the project, the merging lane entering from the on-ramp would be extended 0.2 to 0.3 mile farther west to provide a transition zone for vehicles to enter or exit the right-hand lane. The third eastbound lane would merge in this transition zone, and two lanes would continue west (one HOV lane and one general-purpose lane).

The overall environmental study limits are from east of Lakeville Highway to the eastern side of the Napa River Bridge. These study limits extend beyond the proposed widening to allow for the placement of advance roadway signs notifying drivers of the upcoming HOV lane and the option of installing tolling equipment.

The proposed limits allow for consideration of design alternatives that can serve peakperiod traffic, and provide a more consistent lane configuration that matches the existing four-lane facility west of SR 121 and east of Mare Island. These limits allow for consideration of alternatives that address the existing segment of SR 37, where it is limited to one lane in each direction, and would serve the most heavily congested portion of SR 37.

The rationale for the endpoints outlined above provide sufficient basis for analysis of the potential environmental impacts of the proposed project, which are thoroughly discussed in Chapter 2 of this document.

**Independent Utility:** The proposed project would require no other improvements to provide time-saving benefits to HOV drivers within the project limits, while meeting the purpose and need of the project. It would improve traffic flow and peak travel times, and increase vehicle occupancy in the travel corridor between Mare Island and SR 121. The proposed project considered (and does not preclude) a full range of options, including HOV designation in the peak direction only (Alternative 1), HOV lane designation during the peak period only in both directions (Alternative 2), or a full-time lane that can also be designated for HOV use during peak periods (Alternative 3A) or HOV lanes used 24 hours per day (Alternative 3B).

The proposed project does not restrict consideration of nor depend upon other foreseeable transportation improvements. These are listed in Table 1-1, and include proposed improvements for traffic operations (intersection, interchange, and lane improvements), flood protection, bridge preservation or protection, pedestrian improvements, and SLR adaptation. Those projects are not reliant on the proposed SR 37 Sears Point to Mare Island Project, and could be carried out independently. An independent eastbound lane extension (referred to as EA 2Q200) and SR 37/SR 121 intersection (1Q480) improvement projects have been proposed. The eastbound lane extension would add an eastbound lane on the right-hand side of the highway from the SR 121/SR 37 intersection to east of the railroad tracks and Tolay Creek Bridge. The SR 37/121 intersection project would modify the turning lanes between SR 121 and

SR 37. The EA 2Q200 and 1Q480 projects were combined under EA 2Q20U and are proposed to relieve congestion, and may proceed in the event that the SR 37 Sears Point to Mare Island project is not constructed. This environmental document addresses all potential impacts of both the lane extension and the SR 37/121 intersection improvements. Neither of these projects are needed if the SR 37 Sears Point to Mare Island project is advanced for construction.

This SR 37 project is being designed at the existing roadway elevation and does not fully address SLR. Caltrans, in cooperation with the MTC and the North Bay partner agencies of SCTA, STA, and NVTA, are involved in separate, long-term efforts under the Planning and Environmental Linkages process to address projected year 2130 SLR in the corridor from US 101 to I-80. Caltrans and its partners are proposing that the urgency of the congestion and vehicle hours of delay (VHD) from Sears Point to Mare Island justify proposed interim improvements. This congestion relief project would not conflict or restrict the consideration of future projects that address SLR, because those projects would likely require evaluation of alternative alignments that accommodate multi-year construction of a raised or elevated highway while maintaining access on the existing SR 37 during construction of such a project. This project would also not conflict with other SR 37 projects listed in Table 1-1 in Section 1.1.1, above, which address maintenance and non-capacity-increasing operational improvements elsewhere in the corridor.

# 1.3 Project Description

SR 37 extends from U.S. 101 in Novato, Marin County, to I-80 in Vallejo, Solano County (Figure 1-1). It is an important regional connection linking the north, east, and west San Francisco Bay sub-regions. Additionally, the highway is a parallel route north of the Richmond-San Rafael Bridge (Interstate 580 [I-580]) and is part of the Interregional Roads System between U.S. 101 and I-80. SR 37 connects job markets and housing in Marin, Sonoma, Napa, and Solano counties and provides access to the popular wine-growing regions of Napa and Sonoma counties, the San Pablo Bay National Wildlife Refuge (Refuge), Sonoma Raceway in Sonoma County, Six Flags Discovery Kingdom, and Mare Island in Solano County. Between U.S. 101 and I-80, SR 37 connects with Lakeville Highway, SR 121, and SR 29. The commute, freight movement, and recreational functions of SR 37 require efficient traffic management on both weekdays and weekends.

This project focuses on the portion of SR 37 that has a traffic capacity need where it transitions from four to two lanes between approximately SR 121 and the Mare Island Interchange. The project would relieve traffic congestion by improving traffic flow, reducing peak travel times, and increasing vehicle occupancy in the travel corridor between Mare Island and SR 121. SR 37 narrows from two lanes in each direction to one lane in each direction within the project limits. The highway has acceleration and deceleration lanes at some local intersections, and an existing median barrier along most of the route.

Caltrans proposes to reconfigure the existing SR 37 highway lanes from west of the SR 121 intersection to the Walnut Avenue Overcrossing at Mare Island. The project's

preferred alternative, Alternative 3B, would widen the Tolay Creek Bridge; have four lanes, with two full-time lanes in each direction; have 8-foot shoulders (except at Sonoma Creek Bridge, which would have 4-foot shoulders); and would require widening the Sonoma Creek Bridge. The preferred alternative would introduce the eastbound HOV lane at the SR 121/SR 37 intersection vicinity using Scenario 3, with the HOV lane on the left side, about 0.6 mile west of SR 121, and extend the eastbound left-turn lane approximately 0.5 mile west.

The added lane in each direction would be for general purpose use, subject to tolling in one or both directions. The existing general purpose lane in each direction would convert to HOV use and operate 24 hours per day, 7 days a week, in both directions. The project also would install advance signs to alert drivers approaching the proposed HOV lanes. The project includes the installation and operation of Open Road Tolling on the new general purpose lane, with either one-way or two-way tolling.

# 1.4 Alternatives

The proposed action evaluated in the Draft EIR/EA included five alternatives, with four build alternatives and a no-build alternative, between SR 121 and the Mare Island interchange (project limits). This Final EIR/EA includes the identification of the preferred alternative: Alternative 3B, widen to four lanes with 8-foot shoulders, and widen Sonoma Creek Bridge, which is described in Section 1.4.1. Alternatives 1, 2, and 3A have been rejected after the circulation of the Draft EIR/EA, and are described in Section 1.4.3. Section 1.4.3.1 describes the decision and reasons for the identification of the preferred alternative.

# 1.4.1 Preferred Alternative

Alternative 3B was selected as the preferred alternative. Alternative 3B would have four lanes, with two full-time lanes in each direction. The alternative would reconfigure the existing SR 37 highway lanes within the project limits, and would widen the Tolay Creek and Sonoma Bridges. The alternative would also involve installation of advance signs to alert drivers approaching the proposed HOV lanes. Alternative 3B is described in more detail below.

# 1.4.1.1 Alternative 3B, Widen to Four Lanes, with Eight-Foot Shoulders, and Widen Sonoma Creek Bridge

The preferred Alternative 3B will widen the highway to provide four full-time lanes: two in each direction of SR 37. The inside lane (left-side lane) would function as HOV use 24 hours per day, 7 days a week, in both directions, to incentivize a mode shift from single-occupancy vehicles (SOVs) in both directions. Static signs are proposed to manage the lanes. Figure 1-2 shows a typical cross section of Alternative 3B.



Figure 1-2 Typical Cross Section Alternative 3B

The preferred Alternative 3B includes the following:

- In each direction of SR 37, there will be a 2-foot inside (left) shoulder, an 11-foot inside lane, a 12-foot outside lane, and an 8-foot outside (right) shoulder (except at Sonoma Creek Bridge [see the following bullet]), for a total roadway minimum width of 68 feet. The westbound and eastbound lanes would be separated by a 6-foot-wide median with concrete barrier and 2-foot inside (left) shoulders on either side.
- The Sonoma Creek Bridge will be widened to accommodate two 11-foot-wide lanes, a 2-foot-wide concrete median barrier, 1-foot-wide inside (left) shoulders and 4-foot-wide outside (right) shoulders in both directions, for a total roadway width of 56 feet. This bridge would be widened on both sides. The existing 32-inch concrete median barrier along the bridge would be replaced with a 36-inch-high concrete barrier. A Type 836 barrier would be placed on the Sonoma Creek Bridge in both directions as outside bridge railing. At Sonoma Creek Bridge, an enhanced advisory sign package including an advisory speed limit of 45 miles per hour (mph) has been implemented. Also, at Sonoma Creek Bridge, bicycle/pedestrian detection will be implemented. A Variable Speed Limit Sign (VSLS) system will be implemented in both directions at the Sonoma Creek Bridge and approaches, to enhance safety and to accommodate pedestrians and bicycles across Sonoma Creek Bridge. The VSLS system signs will display the regulatory posted speed limit of 40 mph when pedestrians and/or bicycles are present on the bridge.
- Temporary construction access will be necessary at and beneath the Sonoma Creek Bridge for access to the bridge railings and underside of the bridge. A temporary barge may be needed in Sonoma Creek for this work.

- Both sides of Tolay Creek Bridge will be widened to accommodate one 12-foot-wide lane, one 11-foot-wide lane, and an 8-foot-wide outside (right) shoulder in the westbound direction; and two 12-foot-wide lanes, one 11-foot-wide lane, an 8-foot-wide outside (right) shoulder in the eastbound direction, and a 6-foot-wide median with a new 2-foot-wide concrete barrier, for a total roadway width of 80 feet.
- There will be two lanes in each direction of SR 37 during all hours; however, the inside lane (left-side lane) in each direction will be restricted to HOV use.
- The eastbound HOV lane will begin approximately 0.6 mile west of the SR 121 intersection. The westbound HOV lane will begin immediately west of the Mare Island on-ramp.
- To accommodate merging, two general-purpose lanes and one HOV lane in the eastbound direction will continue east of the SR 121 Intersection. From there, one general-purpose lane would drop, and merge approximately 0.3 mile east of this intersection.
- The eastbound direction west of SR 121 will be widened for approximately 0.8 mile to accommodate the eastbound HOV lane, and extend the left-turn lane to the west.
- The existing 32-inch-high (Type 50) median barrier will be replaced with a 36-inch-high concrete Type 60MS barrier between the SR 121 intersection to Mare Island Interchange for approximately 9.2 miles.
- SR 37/Skaggs Island Road intersection will remain as an at-grade tee intersection, and be stop-controlled at Skaggs Island Road. SR 37 will be widened along the eastbound side to accommodate an approximately 130-footlong left-turn pocket. Southbound traffic on Skaggs Island Road will access eastbound SR 37 by crossing two lanes of westbound SR 37 traffic, and use an approximately 100-foot-long acceleration lane. Along westbound SR 37, an approximately 50-foot-long right-turn deceleration lane will be constructed to access Skaggs Island Road, similar to the existing condition.
- The Cullinan Ranch Restoration Area driveway on westbound SR 37 will remain a right-in and right-out only driveway.
- The Sonoma-Marin Area Rail Transit (SMART) Railroad crossing signals and arms will be reconstructed to accommodate the wider roadway section.
- Approximately 9.1 miles of existing outside shoulder will be reconstructed and converted to a travel lane pavement section and shoulder pavement section in each direction of SR 37.
- The SR 37/Noble Road Intersection will be signalized.

- Approximately 18.2 miles of Midwest Guard Rail System (MGS) will be placed at the edge of the outside shoulder (9.1 miles in each direction).
- The public access driveways on each side of Sonoma Creek, the existing intersection access at SR 121/Sears Point Road/Tolay Creek Road, the driveway to the San Pablo Bay National Wildlife Refuge office, and other private gated driveway access points will be maintained.
- This alternative proposes a 1:1 side slope where widening is required to minimize right-of-way and environmental impacts. Between SR 121 and Sonoma Creek Bridge, sheet pile walls are proposed along the edge of the roadway for pavement confinement, and to minimize settlement. Between Sonoma Creek Bridge and Mare Island, the proposed roadway widening will be on embankment.

The preferred alternative will provide 8-foot outside (right) shoulders between SR 121 and Mare Island, similar to the existing shoulder widths, except at Sonoma Creek Bridge, which will have 4-foot outside (right) shoulders. The preferred alternative will be able to accommodate bicycles across the entire segment of the project limits, including on Sonoma Creek Bridge.

# 1.4.1.2 HOV Scenarios

Alternative 3B introduces the eastbound HOV lane in the vicinity of the SR 121/SR 37 intersection. Three eastbound HOV lane scenarios and the SR 121/SR 37 intersection improvements were considered. Based on discussions with the Project Development Team (PDT), consensus was reached that Scenario 3 was the most appropriate scenario for implementation on this project. Rejected HOV Scenarios 1 and 2 are discussed in Section 1.4.3.2.

Alternative 3B would introduce the eastbound HOV lane on the left side about 0.6 mile west of SR 121, and extend the eastbound left-turn lanes to northbound SR 121 by approximately 0.5 mile to the west. The eastbound left-turn lane would begin approximately 0.7 mile west of the intersection, and would widen to two left-turn lanes approximately 0.1 mile west of the intersection. The second eastbound left-turn lane would be approximately 400 feet long, and terminate at SR 121. The eastbound direction would be widened to accommodate two left-turn lanes, one HOV lane, and two general-purpose lanes with 8-foot outside shoulders. The westbound lane configuration would be maintained with two general-purpose lanes west of the intersection. East of the intersection, the westbound approach would consist of two general-purpose lanes and one left-turn lane. The westbound off-ramp to northbound SR 121 diverges from the outer lane just west of the railroad crossing. Southbound SR 121 would be widened to include a through lane to access Tolay Creek Road on the opposite side of SR 37.

# 1.4.1.3 Design Features of the Preferred Alternative

#### **High-Occupancy Vehicle Lane**

For Alternative 3B, there would be an HOV lane in each direction adjacent to the median, in addition to the new general-purpose lane that will be a tolled lane. The HOV lane use will be designated as 24 hours, 7 days per week (full time HOV lane).

#### Tolling

Implementation of ORT on SR 37 between SR 121 and the Mare Island Interchange is proposed as part of this project. Tolling is proposed on the new general purpose lanes. The project sponsors would request the addition of tolling on the new general purpose lane(s) through the CTC.

The existing travel lane will be converted to an HOV lane in each direction, and would be designated for HOV use 24 hours per day (a full-time HOV lane), but the HOV lanes would not be tolled. Tolling infrastructure, including two toll gantries, would be constructed as part of this project.

Two methods of toll collection are being considered. The first is westbound-only tolling in the project corridor, similar to the seven state-owned tolled bridges in the Bay Area (such as the Richmond-San Rafael Bridge). The second is eastbound and westbound tolling at half of the toll rate for each direction. Only the new general purpose lanes would be tolled; the existing lanes in each direction will be converted to HOV use and would not be subject to tolling. Means-based toll discounts will be evaluated as part of the tolling plan within the overall program parameters reviewed for the environmental analysis. Tolls would be collected through ORT and All-Electronic Tolling; therefore, toll booths would not be required. Tolls would be collected electronically using transponders carried in the vehicles, and vehicles without transponders would be billed by identifying the owner of the vehicle through images of the license plates. The traffic analysis and forecasting for tolling was completed to a level of detail to support this environmental document, and understand and report the effects of tolling. Additional analysis will be conducted to refine the analysis before tolling is implemented on SR 37.

Up to two overhead gantries will be needed for tolling. Tolls would be collected electronically when vehicles using the general purpose lane cross beneath the toll gantry(s). Tolls would not be collected, or apply to vehicles using the HOV lane, in either the westbound or eastbound direction of travel. An overhead gantry would be installed on SR 37, spanning both directions approximately 0.2 mile west of the Mare Island overcrossing. If final design determines that a second gantry is needed, it would span both directions just east of Tolay Creek Bridge and east of the SR 121 intersection. Locations of the gantries would be installed on the gantries to read vehicle transponders and photograph vehicle license plates. For more details and visual simulations of the gantries, please see Section 2.2.12.

# **Outside Safety Barriers**

The existing highway has metal beam guardrail barriers at the approaches to local street intersections and the Sonoma Creek and Tolay Creek Bridges. With all alternatives, additional outside barriers would be needed to meet current traffic safety requirements. These barriers would be either metal beam guardrails or a Type 85 concrete see-through barrier that allows some visibility through the barrier.

# Signs and Lighting

New roadside and/or overhead signs would be placed along SR 37 in each direction, in advance of the beginning of the HOV lanes, to inform drivers of the upcoming toll zone. The types of new signs would include:

- Signs along the side of the highway would notify drivers of the upcoming HOV lane. These signs would include information on the number of occupants for a qualifying HOV user, the hours of operation of the HOV lane, and penalties for SOVs using the HOV lane.
- Overhead and roadside signs would be installed to notify and inform drivers of the upcoming tolling zone and the applicable toll, and penalties for enforcement of the toll.
- Roadside signs would be place indicating the upcoming exit ramps (these already exist along SR 37).

Overhead signs would require subsurface foundations in the median or alongside the highway. Subsurface excavation for the overhead signs may be up to 60 feet in vertical depth, depending on the subsurface conditions.

Lighting would be added along the corridor in advance of the tolling gantries and at California Highway Patrol (CHP) observational areas. Lighting may also be added at local road intersections to improve safety for vehicles entering or exiting the highway.

# CHP Observational Areas

Observational areas for CHP vehicles to park, monitor, and enforce compliance with the HOV lanes and tolling would be installed at the beginning of the HOV lane and toll lane, at or near the toll gantries. Enforcement areas would be developed in consultation with the CHP.

# **HOV Enforcement**

The CHP will be the responsible agency for conducting HOV lane enforcement. The project proposes to facilitate enforcement by providing the following features:

• CHP observation areas will be placed ahead of the toll gantries. The observation areas will allow sufficient room for an officer to park a vehicle and observe vehicles that are traveling in the HOV lane approaching the toll gantry. Similar to the toll bridges in the Bay Area, the officer would conduct an

February 2023

observation of the vehicle approaching the toll gantry to determine whether that vehicle qualifies to use the HOV lane. Should there be a violator, the enforcement could be conducted either at the Cullinan Ranch Staging Area or the Sonoma Creek Viewing Area for westbound enforcement, or at the California Department of Fish and Wildlife (CDFW) Tubbs Island staging area or the Sonoma Creek Viewing Area in the eastbound direction.

- Similar to other toll bridges in the Bay Area, it is expected that all users on SR 37 within the limits of this project will be required to carry a FasTrak® transponder. It is an enforceable violation to use a toll lane without a toll tag (California Vehicle Code Section 23302). The reading of the transponder would allow the number of declared occupants in a vehicle to be displayed on a LED panel mounted on the toll gantry that can be easily seen by a CHP officer stationed at the observation area. The display of the declared number of occupants will aid the CHP with enforcement.
- HOV-eligible traffic approaching the toll gantries would be funneled through a HOV lane that is introduced ahead of the CHP enforcement observation areas and toll gantries. This would segregate SOV from HOV for tolling and enforcement purposes.
- Advanced overhead HOV signs would be placed, providing HOV requirements, hours of operation, and minimum violation rates.
- HOV signs mounted on median barriers will be placed throughout the corridor, providing HOV requirements, hours of operation, and minimum violation rates.
- HOV pavement markings will be placed on the HOV lane.
- Placement of a solid white strips separating the HOV lane from the general purpose lane will be considered.

# Pullout Areas

Because Alternative 3B was selected as the preferred alternative, 8-foot-wide side shoulders would be provided, and pullout areas would no longer be needed as part of the project.

# **HOV Lane Transition**

Alternative 3B includes transition lanes where the HOV lanes begin. At the eastern end of the project, there would be three lanes in the westbound direction: one HOV lane and two general-purpose lanes from westbound SR 37. The westbound on-ramp from Walnut Avenue merges into the outside general-purpose lane just as the westbound HOV lane begins about 700 feet west of Napa River Bridge. The westbound outside lane would then drop and merge with the inside mixed-flow lane approximately 0.25 mile west of the westbound HOV lane commencement. This area would allow HOV users to change lanes and transition to the HOV lane. The two lanes would continue west (one HOV lane and one general-purpose tolled lane). Under Alternative 3B, the eastbound HOV lane would be added approximately 0.6 mile west of the SR 121 intersection. The HOV lane and two general-purpose lanes would continue east of the SR 121 intersection. The second, outside general-purpose lane would drop, and merge with the other general-purpose lane approximately 0.3 west of the intersection. Two lanes would continue east (one HOV lane and one general-purpose tolled lane).

# **Slope Protection and Reinforcement**

Portions of SR 37 were originally constructed on fill, and there is recurring settlement in some areas. Where settlement has occurred or minor widening of the existing cross section of the highway is needed to accommodate the proposed improvements, reinforcement of the highway section would be performed. Design measures would include driving sheet pile along the edges of the highway shoulder area to help stabilize the roadway and slopes. Sheet piles typically consist of metal sheeting and are vibrated into the earth to form a subsurface wall that would help support the roadbed and help prevent or reduce uneven settlement. Once installed, the sheet pile would not be exposed, or would be minimally exposed where it is functioning as a retaining wall. In addition to sheet piles, rock slope protection may be added or reinforced, or engineered slopes would be installed. All of these measures would be designed to help correct existing recurring deformation of the SR 37 roadway structural section, and to allow for minimal widening of the roadbed to accommodate the proposed new lanes and improvements.

# **Tolay Creek and Sonoma Creek Bridges**

The project limits include two bridge crossings, one at Sonoma Creek and the other at Tolay Creek. The Sonoma Creek Bridge has been previously widened for seismic strengthening and placement of a concrete median barrier. As described earlier, the Sonoma Creek Bridge would require widening for Alternative 3B, involving temporary staging and equipment at the creek. Sonoma Creek Bridge is a United States Coast Guard (USCG) permitted bridge, and widening it would require a Bridge Permit or amendment from the agency.

The Tolay Creek Bridge is a single-span bridge and would be widened on one or both sides to accommodate the additional lanes. The existing abutments would be widened. The existing Tolay Creek channel would remain the same width, and no work is proposed in the channel except potential temporary construction access.

Requests to lengthen Tolay Creek Bridge were submitted as part of public comments on the Draft EIR/EA, with the goal of enhancing tidal prism (volume) currently constrained by the bridge. Lengthening the bridge would result in a wider channel, enhance wetlands and other waters in the project area, benefit special status species that utilize these habitats, and enable upstream Baylands restoration identified as part of the Sonoma Creek Baylands Restoration Strategy (May 2020). Caltrans and the project sponsors will examine Tolay Creek Bridge lengthening/replacement potentially as a separate effort, including any necessary environmental impact evaluation and

1-21

public review. A bridge lengthening/replacement will be determined in coordination with regulatory agencies as part of project permits.

#### Local Road Intersections

SR 37 is a conventional highway, with connecting cross-roads and driveways. These include access to Tolay Creek Road/Sears Point Road, Skaggs Island Road, Noble Road (providing access to Vallejo Flood and Wastewater District and Wing and Barrel Ranch), unnamed access roads, vista points and trail heads, and parking areas. The local road connections are summarized as follows:

- At Noble Road, a traffic signal would be added. This is a lightly traveled road, and the signal would only activate when a vehicle approaches the SR 37 Noble Road intersection.
- Skaggs Island Road is a gated road and is stop-controlled to enter SR 37. Vehicles would continue to be permitted to cross opposing traffic to make a left turn from SR 37 into Skaggs Island Road, and from Skaggs Island Road onto eastbound SR 37.
- Cullinan Ranch Restoration Area driveway would remain right-in and right-out only.

Other existing roadway and driveway access would be maintained, including the public access driveways on each side of Sonoma Creek, the existing intersection access at SR 121/Sears Point Road/Tolay Creek Road, the driveway to the Refuge office, and other private gated driveway access points.

# SMART Railroad (Northwestern Pacific Railroad)

This railroad line crosses SR 37 at grade between Tolay Creek and the SR 121 intersection. It is an active railroad, and there are crossing signals and swing arm barriers that activate when a train is approaching. The crossing signals and arms would need to be reconstructed to accommodate the additional proposed lanes.

#### **Drainage and Culverts**

Roadway widening would be minimized, and the existing drainage systems would be maintained to the extent feasible. However, all four Build Alternatives propose adding impervious areas, which will increase stormwater flows going to the existing waterways. Treatment of this additional runoff would be incorporated along the highway where space permits, but because of the constrained nature of the roadway, offsite treatment options would be needed.

No changes to the existing drainage patterns are anticipated, other than the addition of pavement along the corridor. Runoff from the roadway primarily sheet flows off site or is collected along asphalt concrete dikes and drained to roadside ditches or waterways. At superelevation locations, there are median storm drain systems that outlet to roadside ditches and waterways. The project would require extending or replacing existing cross culvert, culvert crossings at cross-drives, median drainage

systems, and existing treatment media filters and biofiltration trenches. Existing asphalt concrete dikes would be removed, and rock slope protection would be added to side slopes in certain areas. The project would preserve existing drainage patterns to the extent feasible.

# **Right-of-Way**

Temporary construction easements (TCEs) may be needed for the roadway work at SR 121, Tolay Creek Bridge, Noble Road, the Cullinan Ranch public access intersection, and other private access driveways to provide construction access. The duration of the TCEs is expected to be one construction season.

Permanent right-of-way acquisition would be needed at the Refuge, Napa-Sonoma Marshes Wildlife Area (NSMWA), and State-owned parcels, which are adjacent to a portion of the highway under Alternatives 3A and 3B. This would consist of acquisition of a portion of the NSMWA, State-owned and Refuge parcels on the southern side, between approximately across the Cullinan Ranch public access driveway (PM SOL 3.88) and slightly greater than 0.9 mile (5,000 feet) east of the Mare Island Overcrossing (PM SOL R6.20), for a total length of 2.3 miles (12,200 feet). Another partial right-of-way acquisition would be required for Alternative 3B on the southern side of SR 37, approximately 0.6 mile west of SR 121, for a length of 0.09 mile (500 feet). This parcel is understood to be part of the Refuge. Another partial right-of-way acquisition would be required for Alternative 3B on the northern side of SR 37, immediately east of the Sonoma Creek Bridge westbound viewing area, for a total length of slightly greater than 0.1 mile (700 feet). This parcel is understood to be managed by CDFW.

# **Construction Staging**

SR 37 traffic must be maintained during construction, and construction staging areas would be needed along or near the route for equipment and materials. Construction staging areas are determined during final project design, but one potential location on private land has been preliminarily identified. The private land parcel would involve using a portion of the Wing and Barrel Ranch land adjacent to SR 37 off Noble Road; this would require agreement with the ranch and restoration of the site following completion of construction.

# **Other Construction Activities and Requirements**

The construction contractor would be required to follow all standard requirements and procedures to be included during detailed design, specifications, and permits or other authorizations.

# Transportation Management Plan

As part of standard practices, a transportation management plan (TMP) would be prepared during the design phase of the project to address traffic disruptions from project construction. The TMP would include outreach to inform the agencies and the public of the times and locations of upcoming construction, construction signs in and approaching the project area, and incident management for traffic control in the vicinity of construction activities. Access would be maintained for emergency response vehicles.

#### Executive Order 13112

Executive Order (EO) on Invasive Species, EO 13112, is a standard practice that Caltrans adheres to for all projects. In compliance with EO 13112 and subsequent guidance from the FHWA, the landscaping and erosion control included in the project would use species that are not listed as noxious weeds. The following methods would be used in accordance with standard construction practices:

- No disposal of soil and plant materials would be allowed from areas that support invasive species to areas dominated by native vegetation.
- Construction workers would be educated on weed identification and the importance of controlling and preventing the spread of identified invasive nonnative species.
- Gravel and/or fill material to be placed in relatively weed-free areas would come from weed-free sources. Certified weed-free imported materials (or rice straw in upland areas) would be used.

#### **Erosion Control and Construction Discharges**

The following standard practices would be part of the project for erosion control and construction discharges:

- As part of construction, no debris, soil, silt, sand, bark, slash, sawdust, cement, concrete, washings, petroleum products, or other organic or earthen material shall be allowed to enter into or be placed where it may be washed by rainfall or runoff into waters of the United States or drainages. No discharges of excessively turbid water would be allowed, and all equipment would be well-maintained and free of leaks.
- A water pollution control program (WPCP) and erosion control best management practices (BMPs) would be developed and implemented to minimize any wind or water-related material discharges, in compliance with the requirements of the Regional Water Quality Control Board (RWQCB) as well as the 2018 Caltrans Standard Specifications, Section 13-2. The WPCP would provide water pollution control practices to limit stormwater and non-stormwater discharges; temporary construction BMPs would be used to the maximum extent necessary.
- Erosion control methods may include silt fencing, straw wattles, straw bales, coir blankets, sediment traps, and other protective methods to limit the potential for erosion of sediment beyond the work area.

# **Estimated Cost and Schedule**

The project is currently funded through the project approval and environmental document phase. The estimated total cost of the project is between \$250 and \$430 million. The proposed schedule includes completion of preliminary design and environmental review in early 2023, detailed design from 2023 to 2025, and completion of construction (open to use) in 2027.

# 1.4.1.4 Transportation System Management and Transportation Demand Management Alternatives

Traffic Systems Management (TSM) strategies increase the efficiency of existing facilities by accommodating a greater number of vehicle trips on a facility without increasing the number of through lanes. Traffic Demand Management focuses on regional means of reducing the number of vehicle trips and vehicle miles traveled (VMT) as well as increasing vehicle occupancy. TSM encourages transit use and ridesharing, which the proposed project would encourage through the installation of HOV lanes. Although TSM measures alone could not satisfy the purpose and need of the project, the following TSM measures have been incorporated into the Build Alternatives evaluated for this project: vehicle detection systems to monitor traffic speed and density, enforcement, incident management, and other subsystems to maintain acceptable traffic flow, which would benefit transit and other HOVs using the SR 37 corridor.

This project will toll general purpose lanes. Congestion or toll pricing will be used to provide an incentive for HOVs and transit, which are considered a TSM and Transportation Demand Management (TDM) measure. This will be achieved through transit and HOV access into the proposed HOV lanes, which will have no tolls, less congestion, and offer faster travel times than the general purpose lanes. Toll pricing is also a TSM/TDM measure. Other TSM/TDM measures include the proposed funding for additional carpool parking areas, bus transit support, and ridesharing programs.

TSM practices that will or are already applied to SR 37 include incident response, where contingency response planning between agencies (CHP, local law enforcement, and Caltrans) is coordinated to provide emergency response and minimize delays associated with accidents or other events.

# 1.4.1.5 Reversible Lanes

In 2016, California's Governor signed AB 2542 into law, requiring Caltrans to demonstrate that reversible lanes have been considered for any project that would increase capacity or realign a highway. Reversible lanes are lanes that can be used in either direction, depending on the flow of traffic and congestion patterns. The movable barrier considered under Alternative 1 met the requirement for considering reversible lanes, consistent with AB 2542, but was eliminated from further consideration for the reasons given in Section 1.4.3.2.

# 1.4.1.6 Access to Navigable Rivers

The proposed alternatives would not construct a new bridge across a navigable water. The project is not subject to California Streets and Highway Code Section 84.5.

# 1.4.2 Comparison of Alternatives

The effects of the Build Alternatives in comparison with the No Build Alternative are summarized in Table S-1 in the Executive Summary. The complete evaluation of alternatives is provided in Chapter 2. The following summarizes the comparison of the alternatives that was presented in the Draft EIR/EA circulated for public review. Alternatives 1, 2, and 3A were considered, but rejected after the circulation of the Draft EIR/EA.

# 1.4.2.1 No Build Alternative

The No Build Alternative assumes that no project improvements would be constructed. SR 37 transitions from a four-lane facility (two lanes on each side) to a two-lane facility (one lane on each side) within the project limits. The lanes are 12 feet wide, and the shoulders are 8.75 feet wide. There is a 10-foot median with a concrete barrier 36 inches high. There are no bicycle or pedestrian facilities on SR 37 within the project limits. However, bicyclists are permitted on the shoulders of SR 37. The existing bottleneck conditions caused by the lane reduction (from two lanes to one lane) in the westbound direction near the Walnut Avenue Overcrossing and in the eastbound direction near the SR 121 intersection would continue to delay peak-period traffic. Traffic congestion caused by these bottlenecks would continue to deteriorate in the foreseeable future as north Bay Area traffic demand increases. The No Build Alternative provided a basis of comparison with the Build Alternatives.

SR 37 would retain the existing lane and shoulder configuration under the No Build Alternative scenario. This section of highway would provide one lane in each direction at all times, and the road shoulders would remain at their existing widths. The substantial traffic backups and queuing that currently occur where the lanes merge from two to one in each direction would continue to form during the morning and afternoon/evening peak periods, and would lengthen in distance and duration as traffic demand is expected to increase in future years. These backups occur during peak travel periods in the westbound direction starting from Mare Island and extend east toward Vallejo, and in the eastbound direction from the SR 121 intersection toward Lakeville Highway.

The No Build Alternative would avoid construction costs (no capital expenditure). It would avoid impacts from construction activities on environmental resources that are anticipated under the build alternatives related to widening the highway, and temporary impacts. However, the No Build Alternative would not meet the project purpose and need. The existing substantial traffic backups would worsen over time, leading to longer vehicle delays and travel times.
### 1.4.2.2 Alternative 1 – Three-Lane Contra-Flow with Moveable Median Barrier and HOV Lane during Peak Periods

This alternative proposed to convert the existing two-lane highway to a three-lane highway, mostly within the existing roadway prism, with a movable median barrier separating the two directions of traffic. The movable median barrier would provide for two lanes during the peak period in the peak direction and a single lane in the non-peak direction. The additional lane would be a HOV lane to provide an incentive for mode shift from single-occupancy vehicles (SOVs).

The movable median barrier would require daily transfer, requiring a dedicated work crew and equipment. This introduces new, relatively high operational costs, and requires storage building(s) for the equipment and a structure for crew quarters, which have not been defined or designed but would likely be located along the SR 37 corridor near SR 121 and Tolay Creek. Existing subsidence of the roadway occurs that could affect the operation of the barrier. Although the project would be designed to account for this settlement, substantial subsidence and settlement may affect the way that this movable barrier functions over time. Figure 1-3 shows a typical cross section of Alternative 1, in both the AM and PM peak period locations (the barrier would be in only one location at a time).



Figure 1-3 Typical Cross Section of Alternative 1

Alternative 1 would have included the following:

- From just east of SR 121 and the SMART at-grade crossing to Noble Road, SR 37 would be widened to create a four-lane facility. This section would have 4- to 6-foot outside shoulders with approximately 1.4 miles of fixed median barrier to separate eastbound and westbound traffic.
- The HOV lane would be on the left side of the highway (adjacent to the median) and open only during the peak period in the peak flow direction. In the

eastbound direction, the HOV lane would begin east of the SR 121 intersection. The westbound lane would begin west of the Mare Island on-ramp.

- From Noble Road to approximately 0.3 mile west of the Walnut Avenue Overcrossing, approximately 7.6 miles of movable barrier would replace the existing median concrete. This section would consist of three 12-foot lanes directionally divided by the movable barrier, with no inside shoulder and 8-foot outside shoulders that would provide for shared bicycle use. When there are two lanes open in one direction during the peak period, the movable inside lane would be an HOV lane.
- Storage of a barrier transfer machine is anticipated to be immediately west of Noble Road and along the median, approximately 0.3 mile west of the Walnut Avenue Overcrossing structure.
- The median barrier would be moved at least twice per day to accommodate typical peak period directional flow traffic.
- There would be around 4 feet of widening along the corridor, for a total roadway width of 54 feet; under the No Build Conditions, the width is 50.75 feet.
- Both sides of Tolay Creek Bridge (Bridge No. 20-0090) would be widened.
- Sonoma Creek Bridge (Bridge No. 23-0063) would be widened on the westbound side by about 4 to 5 feet unless a design exception is approved. If Sonoma Creek Bridge is not widened, the bridge section would consist of three 12-foot lanes directionally divided by the movable barrier with no inside shoulder, a 4-foot outside shoulder on one side and 8-foot outside shoulders on the other side.

Alternative 1 would serve HOV vehicles in the peak period, peak flow direction only. During non-peak periods the highway between Noble Road and the Walnut Avenue overhead would remain as two lanes (one lane in each direction). Alternative 1 would require buildings for the barrier transfer machine and its backup equipment, maintenance/operator crew facilities, and equipment storage facilities. Employee parking and access driveways for the barrier operation and maintenance crew would also be necessary. These facilities have not been defined to date. The total project construction cost is approximately \$256 million (this does not include operations and maintenance of the movable barrier, estimated at just over \$2 million/year).

Alternative 1 would maintain 6- to 8-foot shoulders. These shoulders would be narrower than existing but accessible to bikes and disabled vehicles, including over Sonoma Creek Bridge. The reduced shoulders may require a design exception, otherwise widening of Sonoma Creek Bridge would be necessary.

The traffic benefits, which are the purpose of the project, are limited to peak hour periods only, in the peak travel direction only. This alternative would not benefit non-peak periods or in the non-peak travel direction (during non-peak periods SR 37 would have one lane in each direction and during peak periods would have only one lane

within the project limits in the non-peak travel direction). When the barrier transfer machine is in operation, there would also be only one lane open in each direction. Changes in travel patterns such as during holidays or special events at the Sonoma Raceway (located near the SR 37/SR 121 intersection) may not be served by the movable barrier if traffic demands are high in both directions of travel at the same time.

Widening would be necessary to accommodate the movable median barrier and the reversible lane, and the addition of one lane in each direction between Noble Road and SR 121. Widening of 4 to 5 feet of Sonoma Creek Bridge would be necessary on the westbound side to meet design requirements for shoulder and median widths unless a design exception is approved. If the Sonoma Creek Bridge is not widened, the highway would have three 12-foot wide lanes, no inside shoulder, and 4- to 8-feet wide outside shoulders.

Alternative 1 represents the narrowest or least amount of widening of the Build Alternatives. However, the movable barrier does not avoid the need for roadway widening and impacts to habitat and wetlands. Additional width is required to accommodate the additional lane and transfer operations, and for widening to two lanes in each direction between Noble Road and SR 121. Approximately 2 acres of wetlands would be permanently impacted, requiring mitigation, with additional impacts to sensitive habitat.

The operation and maintenance of the movable barrier is a major consideration for this alternative that is not required with the other build alternatives. These considerations include:

- Slight deformations of the roadway sections can cause issues over time with the barrier transfer operation. The barrier has to be slightly lifted and moved across the pavement during the transfer operation, and settlement or uneven pavement conditions has the potential to interfere with effective transfer movement of the barrier. Failure to be able to move the barrier would cause adverse traffic congestion as only one lane in each direction would be operable until the pavement condition can be corrected.
- The movable barrier would require approximately 1.5 hours to change travel direction in the center lane, based on a 7-mile-long barrier (Lindsay Transportation Solutions 2022). This transfer operation would occur at least twice daily.
- Multiple barrier transfer machines and redundancy of crew are needed to
  ensure ability to move the barrier 7 days a week, or at special events or
  changes in traffic flow. The crew necessary to operate and maintain the
  movable barrier represents a long-term financial commitment over the life of the
  project (twenty of more years) to fund the required workforce, training, and all
  costs associated with staffing for this operation.
- Maintenance and storage of the barrier transfer machine is necessary on-site. Storage of the barrier transfer machine is anticipated at two locations, in the

median immediately east of Noble Road and approximately 1,500 feet west of the Walnut Avenue overcrossing. At least one maintenance building would be required for storage of equipment. Crew quarters would also be necessary on or near site. There are no similar structures along SR 37 between the Napa River and SR 121, and these modern buildings would be highly visible and contrast with the existing rural nature of the route.

### 1.4.2.3 Alternative 2 – Convert Existing Outside Shoulders to HOV Use during Peak Periods (Part-Time Use Lane)

This alternative would have used the existing highway shoulders to provide a traffic lane during the peak periods in the peak direction. During peak hours in the peak direction, the outside shoulder was proposed to act as an HOV lane, and act as a shoulder in the non-peak direction. The outside lane would be for HOV use during peak periods to provide an incentive for mode shift from SOVs. Static signs were proposed to manage the part-time lanes. Figure 1-4 shows a typical cross section of Alternative 2.



Figure 1-4 Typical Cross Section of Alternative 2

Alternative 2 would have included the following:

- There would be two 11-foot inside lanes separated by a fixed median barrier with a 2-foot inside shoulder (6-foot median), two 12-foot outside lanes, and a 4-foot dedicated shoulder, for a total roadway minimum width of 60 feet.
- During the peak period, there would be two lanes in the peak direction. The inside lane would be for general-purpose use only. The outside lane would be for HOV use during peak periods only. The peak time periods are 5 AM to

11 AM and PM peak time periods are 2 PM to 9 PM. During the non-peak period, there would be only one lane in the non-peak direction; it would be a general-purpose lane (open to all vehicles), with the outside lane reverting to a shoulder, amounting to a 16-foot shoulder.

- The eastbound HOV lane would begin east of the SR 121 intersection to the Walnut Avenue Overcrossing. The westbound lane would begin west of the Mare Island on-ramp.
- Approximately 9.09 miles of existing outside shoulder would be reconstructed and converted to a travel lane pavement section in each direction.
- The existing 32-inch-high concrete median barrier may need to be replaced with a new standard 42-inch-high concrete barrier for approximately 9.3 miles.
- Approximately 25 vehicle pullout areas would be constructed.
- Both sides of Tolay Creek Bridge would be widened.

The existing Sonoma Creek Bridge can accommodate the proposed lane configuration except for the 4-foot outside shoulder. This alternative would not accommodate bicycles because the Sonoma Creek Bridge would be too narrow to maintain an adequate shoulder for safe passage. The shoulders would be available to disabled vehicles and for bicycle use during non-peak periods, but would not be available during peak periods in the peak direction. During non-peak periods, the highway remains as two lanes (one lane in each direction), and similar to existing conditions there would be a lane drop in each direction at SR 121 and at Mare Island where the two lanes merge to one lane and the highway would remain one lane in each direction between SR 121 and Mare Island. Pullouts would be added at periodic locations for disabled vehicles and enforcement. Additional signs would be installed to notify and inform motorists of when the HOV lane is available or closed.

This alternative has been identified as having the potential for driver and enforcement confusion with respect to when the shoulders are available for use as a through lane, versus when the shoulder is restricted to disabled vehicle use only. Because the shoulder is open to use only during certain time periods, drivers have to pay attention to and understand the times when the shoulder is open to through traffic in the peak period, peak direction only. There is a potential for a driver to use the shoulder as if it was open for use (when it is not), and result in a collision with a disabled vehicle that is stationary in the shoulder.

Bicyclists using the shoulders during non-peak periods would have to be aware that a return trip may not be possible during the peak period, peak direction when the shoulder is open to through traffic only (no bikes allowed in the peak direction, peak period of travel). Because of the lack of shoulders during the peak travel period, legislation to prohibit bicycle and pedestrian use along this corridor will be proposed.

The total pavement width and widening required would be about 4 to 6 feet more than Alternative 1, and would be similar to Alternative 3A. Construction of this alternative would be generally similar to Alternative 3A. Widening and reconstruction of the shoulders would be necessary to provide a road foundation and width adequate to support traffic using the shoulders during the peak periods.

A total of 3.5 acres of wetlands would be permanently impacted, requiring mitigation, with additional impacts to sensitive habitat. Total project cost is estimated at about \$306 million.

The cost of constructing this alternative, with the required widening, is relatively similar to Alternative 3A but this alternative does not provide the benefit of an additional full time HOV lane in each direction that is gained with Alternative 3A and 3B. The travel benefits of Alternative 2 are considered limited with respect to the investment necessary to add this part-time lane.

### 1.4.2.4 Alternative 3A – Convert Existing Outside Shoulders to HOV in Each Direction (Four-Lane Facility)

This alternative would have widened the highway to provide four lanes, two in each direction. All four lanes would be general-purpose lanes during non-peak periods. The inside lane (left-side lane) would then be changed over for HOV use during peak periods to provide an incentive for mode shift from SOVs. Static signs are proposed to manage the lanes. Three HOV scenarios were considered for Alternative 3A, and are described in Sections 1.4.1.2 and 1.4.3.2. Alternative 3A has the same cross section as Alternative 3B, as illustrated in Figure 1-2.

Alternative 3A would have included the following:

- In each direction, there would be a 2-foot inside shoulder, an 11-foot inside lane, a 12-foot outside lane, and a 4-foot outside shoulder, for a total roadway minimum width of 60 feet. The westbound and eastbound lanes would be separated by a 6-foot-wide median with concrete barrier and 2-foot shoulders on either side.
- There would be two lanes in each direction of SR 37 during all hours; however, during the peak period, the inside lane (left-side lane) in each direction would be restricted to HOV use.
- The eastbound HOV lane would begin approximately 0.6 mile west of the SR 121 intersection. The westbound HOV lane would begin immediately west of the Mare Island on-ramp.
- To accommodate merging, two general-purpose lanes and one HOV lane in the eastbound direction would continue east of the SR 121 Intersection. Then, one general-purpose lane would drop and merge approximately 0.3 mile east of this intersection.

- The eastbound direction west of SR 121 for would be widened for approximately 0.8 mile) to accommodate the eastbound HOV lane and extend the left-turn lane to the west.
- Approximately 9.1 miles of existing outside shoulder would be reconstructed and converted to a travel lane pavement section and smaller shoulder in each direction. A Type 85 or MGS would be placed at the edge of the outside shoulder in each direction.
- The existing 32-inch-high (Type 50) concrete median barrier would be replaced with a standard 36-inch-high (Type 60MS) concrete barrier for approximately 9.3 miles.
- Approximately 25 vehicle pullout areas would be constructed.
- Both sides of Tolay Creek Bridge would be widened.
- The SR 37/Noble Road intersection would be signalized.

The existing Sonoma Creek Bridge can accommodate the proposed lane configuration except for the 4-foot outside shoulder. This alternative cannot accommodate bicycles because the Sonoma Creek Bridge would be too narrow to maintain an adequate shoulder for safe passage. Because this alternative has nonstandard inside and outside shoulders, additional design measures were proposed.

Similar to Alternative 2, the center median would have a fixed concrete barrier similar to existing, but 42 inches tall (existing barrier is 32 inches high). The left lanes would be for HOV use during peak periods. Approximately 96,000 feet (18 miles) of Type 85 barrier would be installed on the edge of the outside shoulders. Total project cost is estimated at about \$325 million.

This alternative (and Alternative 3B) eliminates the existing lane reductions in both directions that cause the bottlenecks and congestion points starting near the SR 121 interchange in the eastbound direction, and starting at Mare Island in the westbound direction. With this alternative, SR 37 would have at least two continuous full-time lanes in both directions at all times between Vallejo and U.S. 101. For drivers it provides predictability that the lanes are available and continuous at all times, limited to the HOV restrictions during peak periods. Alternative 3A (and 3B) provide the maximum time savings advantage for HOVs of all alternatives considered, which is one of the primary elements of the project's purpose and need.

Compared to the existing highway, the shoulders would be reduced from approximately 8 feet wide to 4 feet wide in most locations. At the Sonoma Creek Bridge, the structure is wide enough to accommodate the additional full time travel lanes in each direction but would require design exceptions for lane width, inside shoulder, and no outside shoulder. With the lack of an outside shoulder bicycles would not be able to cross over the Sonoma Creek Bridge. The shoulders would be 4 feet or wider in all other portions of the project, with periodic pullouts installed to accommodate disabled vehicles. Because of the changes in shoulder widths, legislation to prohibit bicycle and pedestrian use along this corridor will be proposed.

Alternative 3A would require permanent acquisition of approximately 1.65 acres of right-of-way from the United States Fish and Wildlife Service (USFWS) Refuge south of Cullinan Ranch, and permanent acquisition and temporary construction easements at the NSMWA (this is a Section 4(f) impact). Permanent wetland impacts are estimated at 4.3 acres.

Alternative 3A (and 3B) include three possible design variations for the eastbound HOV lane at the SR 37/121 intersection. Scenario 1 introduces the HOV lane on the right side of SR 37, west of SR 121. Scenario 2 introduces the HOV lane on the left side of SR 37, in the vicinity of the Tolay Creek Bridge. Scenario 3 introduces the HOV lane on the left side of SR 37 about 3,000 feet west of SR 121, extends the eastbound left turn lanes leading to SR 121, and widens the left turn lanes to two approximately 500 feet west of the intersection.

Because Alternative 3A has non-standard outside shoulder widths, it has additional design requirements that are under consideration. These include the implementation of an Incident Management Plan, introduction of legislation for prohibiting bicycle and pedestrian access along the corridor (along the highway), providing low-cost or free bicycle shuttle service with intermediate stops, accommodating future bus/transit service that can carry bicycles, placement of approximately 25 vehicle pullout areas, median barrier emergency gates, additional lighting at four curve locations to enhance visibility, and outside concrete barrier railings.

### 1.4.2.5 Alternative 3B – Widen Highway to Add a Full-time HOV lane in each Direction (Four-Lane Facility), 8-Foot Outside Shoulders, Widen Sonoma Creek Bridge.

Alternative 3B is the identified preferred alternative (see Section 1.4.3.1), and is described here to compare with the other evaluated alternatives. Alternative 3B is the same as Alternative 3A in that it would provide a new full time lane in each direction for HOV use during peak travel periods, and would be open to all vehicles during non-peak periods. However, Alternative 3B would maintain 8-foot-wide outside shoulders between Mare Island and SR 121 instead of the 4-foot road shoulders considered in Alternative 3A.

Similar to Alternative 3A, the lane configuration in each direction would be 11-footwide inside lanes, and 12-foot outside lanes, a 6-foot-wide median, and the outside shoulders would be 8 feet wide (Alternative 3A outside shoulders would be 4-foot wide, with no shoulders over the Sonoma Creek Bridge). The total minimum width of pavement would be 68 feet. Similar to Alternatives 2 and 3A, the center median would have a fixed concrete barrier similar to existing, but 36 inches tall (existing barrier is 32 inches high). The left lanes would be for HOV use and operate 24 hours per day in both directions. 96,000 feet (18 miles) of Type 85 or MGS Guardrail would be installed on the edge of the outside shoulders. Total project cost is estimated at about \$430 million, and is the highest of all build alternatives, adding approximately \$105 to \$175 million to the cost of the project compared to the other build alternatives due to the wider highway width and the widening of Sonoma Creek Bridge.

The 8-foot-wide shoulders would maintain access for bikes over the Sonoma Creek Bridge. Disabled vehicles would also be able to use the shoulders anywhere along the route, compared to the pullout areas proposed for Alternative 3A.

Alternative 3B would widen the Sonoma Creek Bridge on both sides. The median and lanes on the bridge would be shifted to align with the widened structure. The bridge widening adds additional cost, time, and complexity to the project. It would contribute to a greater amount of potential impacts to natural resources associated with both bridge and road shoulder widening, and is anticipated to require a substantially greater amount of mitigation to offset impacts relative to other build alternatives. Temporary access during construction to the wetlands and waters directly below and adjacent to the Sonoma Creek Bridge is anticipated for installation of a debris catchment system. Caltrans would use vehicles and man lifts to reach the underside of the bridge. Access through wetland vegetation would use mats to minimize impacts on existing vegetation. Work on the underside of the bridge directly above waters would be conducted from a barge-mounted lift. Alternative 3B previously considered a temporary falsework trestle adjacent to Sonoma Creek Bridge over 1 to 3 years that has been removed from the project. The navigational channel of Sonoma Creek would remain open during construction consistent with the existing USCG permit for this bridge.

Alternative 3B includes the same eastbound HOV lane extensions on SR 37 as described for Alternative 3A.

Alternative 3B would require permanent acquisition of sliver takes of right-of-way and temporary construction easements at the CDFW NSMWA and permanent sliver takes of right-of-way at the USFWS Refuge south of Cullinan Ranch (this is a Section 4(f) impact). Permanent wetland impacts are estimated at 9.0 acres.

### 1.4.3 Alternative Selection Process

Caltrans has identified a preferred alternative and final determination of the project's effect on the environment. Under the California Environmental Quality Act (CEQA), Caltrans certifies that the project complies with CEQA; has identified significant impacts; and identified mitigation for impacts. Caltrans will file a Notice of Determination with the State Clearinghouse after project approval, and identify whether the project would have significant impacts; and identified mitigation measures included as conditions of project approval. Caltrans, as assigned by the FHWA, has determined that the National Environmental Policy Act (NEPA) action does not adversely impact the environment, and has issued a Finding of No Significant Impact (FONSI).

### 1.4.3.1 Identification of the Preferred Alternative

The PDT identified Alternative 3B as the preferred alternative. This decision was made after a series of PDT meetings conducted in March and April 2022 that included a summary of the environmental impacts associated with each alternative; safety; design criteria; traffic benefits and performance of the alternatives; maintenance; and public and agency comments and concerns. The PDT identified Alternative 3B as the preferred alternative at a March 25, 2022 meeting, recognizing that there may be further potential to reduce in-water work and environmental impacts based on construction impacts and widening at the Sonoma Creek Bridge. Caltrans will continue to examine opportunities to further refine the project's construction footprint (the area of construction) during the project's design phase with the intention of further reducing environmental impacts of the project. Along with the environmental impacts, we considered operational concerns regarding maintenance worker exposure and safety, driver safety concerns raised by CHP. Additional PDT meetings were held on April 8 and 14, 2022, and a final meeting on April 21 confirmed the PDT's decision to select Alternative 3B as the preferred alternative. During the course of project development, the PDT considered the information in the technical studies completed for the project (listed in Appendix J), the information in the Draft EIR/EA, comments received during the public scoping period and public review period for the Draft EIR/EA, and the public scoping meeting and the public meeting held after release of the Draft EIR/EA.

#### Alternative 3B (Preferred Alternative) – Full-time HOV Lane in each Direction, 8-Foot Outside Shoulders with 4-Foot Outside Shoulders at Sonoma Creek Bridge

Alternative 3B would provide maximum traffic benefits similar to Alternative 3A, but more importantly, the preferred alternative would provide outside shoulders that would be 8 feet wide, except at the Sonoma Creek Bridge, where the outside shoulders would be 4 feet wide. Eight feet outside shoulders are considered standard, and safer for a heavily traveled highway such as SR 37, because it provides important safety refuge for motorists that have disabled vehicles, and safety for enforcement and emergency responders needing to use the shoulders to stay out of harm's way of travel lanes. Input contributing to this decision came from feedback received during scoping and the public review period from the CHP and Caltrans subject-matter experts, who supported the importance of maintaining the shoulder width that is most consistent with the existing conditions. The PDT also considered that providing two full-time lanes on SR 37 provides the ability to use SR 37 to evacuate communities in the event of an emergency (e.g., wildfire or earthquake), as has been the case in the northern county area.

Alternative 3B would provide the maximum traffic benefits, best meeting a purpose and need based on the Traffic Operations Analysis Report. Alternative 3B would provide HOV lanes in both directions of travel (as opposed to Alternatives 1 and 2, which would have an HOV lane in only one direction of travel). Having the HOV lane in both directions best addresses traffic growth in the future, as travel demand increases in both directions during both the AM and PM peak periods. Alternative 3B would maintain access on SR 37 for bicyclists, compared to Alternatives 2 and 3A, which proposed elimination of bicycle access due to the reduced outside shoulder widths. Maintaining bicycle access eliminates the identified condition (under Alternatives 2 and 3A) for legislative approval to exclude bicycle access on SR 37. Alternative 3B would align with Caltrans Complete Streets policy. In addition, many agencies and bicycle coalition groups expressed support during the public review period for the alternative that would best maintain bicycle access.

The PDT also considered and included improvements to Alternative 3B at the Sonoma Creek Bridge to reduce environmental impacts. As originally proposed in the Draft EIR/EA, Alternative 3B would have required the placement of additional supporting columns in the banks and in Sonoma Creek to adequately support the originally proposed 8-foot-wide outside shoulders. The design for the preferred alternative was modified to reduce the outside shoulders, on Sonoma Creek Bridge only, to 4 feet wide, which could be constructed without the need for supporting columns in the creek channel and banks. Installing these piers would have required impact pile driving equipment in the channel and banks, producing noise impacts associated with impact pile driving. The impact pile driving would have further complicated construction and associated environmental impacts, especially to fish and tidal habitat in Sonoma Creek. Although temporary construction work and access would still be necessary under the Sonoma Creek Bridge, the construction period and intensity of construction activity disturbances would be substantially minimized. Under these improvements, the shoulder widths on the Sonoma Creek bridge would be reduced but still allow bicycle access along the bridge.

The PDT reviewed and discussed the potential environmental impacts of the project throughout the development of the project alternatives. All four build alternatives would result in impacts to VMT, visual, water quality, wetlands, and biological habitat, and each alternative had varying degrees of impacts. Alternative 3B represents the maximum environmental footprint, because of the added lanes in each direction and full shoulder widths. However, the other alternatives were developed to consider a narrower width of the highway that did not adequately meet the purpose and need of the project, and/or were considered unacceptable for the reasons described in Section 1.4.3.1.

The PDT considered three HOV scenarios to include for the HOV lane transition at the SR 37/SR 121 intersection for Alternative 3B, as described in Sections 1.4.1.2 and 1.4.3.2. The PDT identified Scenario 3, the Eastbound HOV lane segment at SR 37/121, to be optimal in terms of safety and operational performance. It was identified for inclusion in the preferred alternative because it would provide a predictable left-side HOV configuration. HOV vehicles would be able to move to the HOV lane west of the SR 121 intersection, and minimize the queueing in the non-HOV lane.

### 1.4.3.2 Alternatives Evaluated in the Draft EIR/EA but Eliminated from Further Consideration following Public and Agency Review and Comment

In light of the analysis and the identification of the preferred alternative, the following alternatives were eliminated from further consideration based on analysis in the Draft EIR/EA, consideration of public comments, and PDT discussion. Alternatives 1, 2, and 3A were all developed and evaluated with the specific intent to minimize harm to wetlands and protected habitat, as well as reduce the need for off-setting mitigation, and reduce project costs. Alternative 1 would employ a movable barrier to alternate the directional flow of the center lane of the highway, using the pavement space to emphasize use for the peak direction of travel as it switches in direction twice a day. Alternative 2 was conceived to consider using the shoulders as travel lanes during the peak periods, and then revert the paved space back to shoulders during the non-peak periods. Alternative 3A was designed to reduce the shoulders to a width of 4 feet (the shoulders are currently 8 feet wide on SR 37, which is a design standard). Although these alternatives provided reduced impact to wetlands and habitat alongside SR 37, after careful consideration by the project team, and after receiving public and agency input, the PDT ultimately made the decision to eliminate them from further consideration for the reasons explained in the following sections.

### Alternative 1 (Eliminated from Further Consideration) – Three Lanes with Moveable Median Barrier

Alternative 1 was not carried forward for several reasons. Adding one additional lane would provide some traffic congestion relief, but not one as substantial over time as the preferred alternative. This alternative required an obligation by Caltrans to operate and maintain a movable barrier for the duration of the project's lifespan. Other factors that contributed to this decision are discussed in Section 1.4.3.2.

This alternative partially met the project's purpose and need. However, the traffic benefits would be limited to the peak period in one direction only, and would provide no traffic benefits in the non-peak direction of travel with respect to congestion reduction, improvement in travel time, or reduction of queuing. Comparing the peak period travel times shows a substantial reduction in delays achieved by the preferred Alternative 3B, Scenario 3, compared to Alternative 1 Movable Barrier:

- PM Peak: In 2045, maximum eastbound travel time for Alternative 1 would be approximately 156 minutes for SOVs and 149 minutes for HOVs. A shorter (faster) travel time would be achieved by Alternative 3B Scenario 3, at 67 minutes for SOVs and 54 minutes for HOVs. Comparison of the peak hours shows Alternative 1 would continue to have queuing that backs up to the SR 37 westbound ramps and Walnut Avenue/Main Gate, with LOS F conditions for all hours between 5 AM and 10 AM. Alternative 3B would reduce the queuing and provide an LOS of C or better for all hours except between 6 AM and 8 AM.
- AM Peak: In 2045, the maximum westbound travel time for Alternative 1 would be 137 minutes for SOVs and 97 minutes for HOVs. This compares to

101 minutes for SOVs and 97 minutes for HOVs with the preferred Alternative 3B, Scenario 3. Peak hour delays at the intersections would be substantially lower with Alternative 3B than with Alternative 1 at all intersections studied (Lakeville Highway, SR 121, Noble Road, Skaggs Island Road, and the Walnut Avenue ramps).

Alternative 1, by having a single lane in the non-peak direction, reduces the ability of the highway to accommodate changes in travel patterns associated with special events such as fluctuations in travel on holidays or Sonoma Raceway events, when travel demand may be in both directions. Alternative 3B, with two lanes in each direction, would provide more accommodation of changes and growth in traffic in both directions.

Input received during the review period of the Draft EIR/EA was an important factor in dropping Alternative 1 from further consideration. CHP provided input that the movable barrier would introduce an unwanted risk to Caltrans personnel and contractors, and to emergency response times during transfer operations. CHP and local landowners emphasized the need for openings in the median barrier at Skaggs Island Road and Noble Road to allow for access and left turns. This would be critical to the CHP and emergency responders to make U-turns at these midway locations and allow for quicker response times. Alternative 1 would require travel to each end of the continuous movable barrier before being able to reverse directions to respond to an emergency or enforcement event in the opposite direction of travel. Segmenting the movable barrier could be achieved (providing gaps or openings at the local roads), but it would complicate the barrier transfer machine when crossing these openings through local intersection cross traffic and reconnecting to the next barrier segment to continue the barrier transfer operation.

Caltrans maintenance was not supportive of the Alternative 1 Movable Barrier. The barrier would require maintaining an operational and maintenance crew daily to move the barrier at least twice per day. This represented a commitment over the lifespan of the project that would have to be funded for personnel and capital costs. As noted in the Draft EIR/EA, this would include the operational and maintenance crews, housing the crew at or very near the project location, hiring and training, and providing for storage of equipment at the corridor. The environmental sensitivity of the corridor area (much of it within protected habitat and a designated wildlife refuge) restricts the availability of areas where additional structures for equipment storage and maintenance can be located.

### Alternative 2 (Eliminated from Further Consideration) – Convert Existing Outside Shoulders to HOV Use during Peak Periods (Part-Time Use Lane)

Alternative 2 would add only one lane in the peak direction in the peak hour, which would provide a traffic congestion relief benefit, but not one as substantial as the preferred alternative. In addition, this alternative was identified as introducing unwanted potential driver confusion and misuse of the extra wide shoulders during

non-peak periods, a lack of shoulder space during peak travel periods for the safety of emergency and enforcement responders, and lack of bicycle access.

This alternative partially meets the purpose and need of the project. During the peak period, the shoulder would be available as an HOV lane during its hours of operation, but would not be available during non-peak periods, and not available in the non-peak direction of travel. Although this would help alleviate congestion during the peak period, it would only provide this benefit in one direction of travel. This design and operation would require a substantial investment in widening the highway, but would not maximize the use of the widened highway.

The outside shoulders would appear unusually wide during non-peak periods and in the non-peak direction of travel. Instead of typical 8-foot-wide shoulders, they would be 16 feet wide with signs identifying them as restricted from use during non-peak periods. Drivers would have to be aware or familiar with the restriction of hours of use during non-peak periods to avoid a violation, and the potential for frequent violations along the corridor would likely increase law enforcement efforts. This design was therefore seen as having the potential for driver confusion, and increased misuse by drivers using the wide shoulders as a passing lane on the right side to get around slower vehicles.

The closure of the shoulders to bicyclists would be necessary (for safety of the bicyclists) during peak periods when the outside shoulders are only 4 feet wide, which was determined not to be acceptable. Having the shoulders closed during peak periods, but open during non-peak periods, was also determined not practicable or safe because it might prevent bicyclists from making a necessary return trip if the shoulder was closed at the time they attempt to return.

# Alternative 3A (Eliminated from Further Consideration) – Full-time HOV Lane in each Direction, 4-Foot Outside Shoulders, No Shoulders or Widening at Sonoma Creek Bridge

Alternative 3A was not carried forward for safety reasons. The proposed 4-foot-wide outside shoulders along the majority of the route, with minimal to no shoulders on the Sonoma Creek Bridge, was determined unacceptable for safety of motorists needing emergency use of the shoulders. The four foot space was not considered adequate to width to provide room for disabled vehicle refuge (ability for a vehicle to be safely outside of the travel lanes). Caltrans also received concern from the CHP about safety of motorists, safety of enforcement and emergency response providers, the potential reduced safe space and access for CHP enforcement activities, and safety of maintenance vehicles that require safe access along the highway shoulders. The project team considered the use of pullouts, where periodic areas of paved outside shoulder up to 12 feet would be provided, but this was dropped from consideration because the pullouts would not provide the same level of safety as the continuous 8-foot outside shoulder proposed for Alternative 3B.

Similar to Alternative 2, the necessary closure of the shoulders to bicyclists for Alternative 3A was determined not acceptable, based on feedback from public and agency review comments.

### HOV Lane Transition Scenarios Considered but Rejected

- Scenario 1 would introduce the HOV lane on the right side approximately 0.6 mile west of SR 121. This scenario was not selected as the preferred option because of the unconventional nature of the HOV lane location and the potential for driver confusion. The eastbound direction would be widened to accommodate two left-turn lanes, two general-purpose lanes, and one HOV lane with an 8-foot-wide outside shoulder. The westbound lane configuration would be maintained with two general-purpose lanes west of the intersection. East of the intersection, the westbound approach would consist of two general-purpose lanes and one left-turn lane. The westbound off-ramp to northbound SR 121 would diverge from the outer lane just west of the railroad crossing. Southbound SR 121 would be widened to include a through lane to access Tolay Creek Road on the opposite side of SR 37.
- Scenario 2 would introduce the HOV lane on the left side east of SR 121, in the vicinity of the Tolay Creek Bridge. This scenario was not selected as the preferred option because it provided the least traffic operational benefits with regard to travel time compared to Scenarios 1 and 3.

### 1.4.3.3 Alternatives Considered but Eliminated During Early Development of the Project

Additional alternatives were considered during the early stages of project development but were eliminated because they did not meet the project's purpose and need and would have had greater environmental effects compared to other alternatives. The following describe these alternatives and why they were not advanced for further evaluation.

### Four-Lane Standard Section

This alternative proposes widening the current two-lane SR 37 to a full standard fourlane facility at the existing elevation. The proposed addition of one lane in each direction would provide sufficient capacity to prevent the existing bottle necks between within the project limits from developing and would relieve traffic congestion experienced in this corridor. Per Caltrans conventional highway standards, the proposed lane configuration includes two 12-foot lanes, a 5-foot left shoulder, and a 10-foot right shoulder. This alternative proposes to widen the existing 50-foot roadway section to a 74-foot section. This alternative also proposes to widen both the Sonoma Creek and Tolay Creek Bridges. The area of environmental disturbance would mean a greater commitment of mitigation and necessary approvals from various regulatory agencies. This alternative was not considered further for the following reasons:

- The Four-Lane Standard Section would have one of the highest environmental impacts and widening costs of all alternatives, to achieve 12-foot wide lanes and wider inside shoulders (adjacent to the median). The total roadway width would be 76 feet, compared to next widest alternative (Alternative 3B at 68 feet wide).
- The relatively wider widening requirement could be avoided by Alternatives 3A or 3B, which also achieve two-through full-time lanes in each direction with less environmental impact.

#### Fixed Barrier and Three-Lane Reversible Lane Section

This alternative proposes a fixed barrier and a separated reversible lane section consisting of a 12-foot reversible lane for peak directional traffic, 2-foot left shoulders, two fixed permanent barriers on each side of the reversible lane, and 8-foot right shoulders for the general-purpose lanes and reversible lane. Therefore, the outside lanes would serve east and west bound traffic, while the center lane could be reversed to serve the peak flow direction during the peak hour. This operation would be similar to Alternative 1, except there would be two solid non-movable barriers with a reversible lane between them. It proposes widening into environmentally sensitive areas that are along the existing corridor (this alternative would have the widest footprints and environmental impact of the alternatives considered, greater than Alternative 3B). In addition to the widening, this alternative would have ongoing operational and maintenance costs for the reversible lane operations.

This alternative was not carried forward past the PSR-PDS for the following reasons:

- The solid barriers would require at least one wide (8-foot) shoulder to allow for emergency vehicles and for disabled vehicles to pull out of the traveled lane. This increases the overall width of the highway.
- Reversing the center lane would require a commitment to substantial maintenance and operation costs, similar to Alternative 1. This would include a dedicated maintenance and operation crew available to operate the reversible lane, with redundancy of crew and equipment to ensure at least twice daily operation of the lane.
- Each end of the barrier where vehicles enter or exit would require movable gates to allow closure of the lane. A maintenance crew and possibly a patrol vehicle would have to check that the lane is cleared of all vehicles before the lane could be opened in the reverse direction.
- The lane has the potential for driver confusion because lane operation reverses at least twice a day. There would be the potential for an inattentive or confused driver to mistakenly attempt to enter the lane in the wrong direction.

This alternative was not further advanced because it did not offer any advantages beyond those of the other alternatives considered. It would mandate a commitment to

a high operation and maintenance commitment over the lifetime of the project. The creation of a walled-in lane in the center of SR 37 would have a high adverse visual impact.

### 1.5 Other Project Features

This project contains a number of standardized project features that are employed on most, if not all, Caltrans projects in accordance with standard specifications, state and federal laws, and anticipated standard environmental permit conditions, and were not developed in response to any specific environmental impact resulting from the proposed project. Project features are separated out from avoidance, minimization, and minimization measures (AMMs), which directly relate to the impacts resulting from the proposed project. AMMs and other measures are discussed separately in each environmental section.

A summary of these project features is presented in Table 1-4.

### **1.6 Permits and Approvals Needed**

Table 1-5 summarizes the permits, reviews, and approvals that would be required for project construction. Most permit applications would be submitted during the design phase. Further discussion about the project's status with respect to permits and approvals can be found in Chapter 4.

Resource	Feature Number	Description
Air Quality	PF-AIR-01	Construction Best Practices for Dust
		The following are BMPs from Mitigation Measure 2.2 2 in the Addendum to Final Environmental Impact Report Plan Bay Area 2050 (ABAG and MTC 2021a). These measures control dust during any construction period that involves ground disturbance.
		• All exposed surfaces (e.g., parking areas, staging areas, soil piles, graded areas, and unpaved access roads) shall be watered two times per day. For projects more than 5 acres in size, soil moisture should be maintained at a minimum of 12 percent. Moisture content can be verified by laboratory samples or a moisture probe.
		• All haul trucks transporting soil, sand, or other loose material offsite shall be covered.
		• All visible mud or dirt track-out onto adjacent public roads shall be removed using wet power vacuum street sweepers at least once per day. Dry power sweeping should only be performed in conjunction with thorough watering of the subject roads.
		<ul> <li>All vehicle speeds on unpaved roads and surfaces shall be limited to 15 mph.</li> </ul>
		• All roadway, driveway, and sidewalk paving shall be completed as soon as possible. Building pads shall be paved as soon as possible after grading.
		• All construction sites shall provide a posted sign visible to the public, with the telephone number and person to contact at the Lead Agency regarding dust complaints. The recommended response time for corrective action shall be within 48 hours. The Bay Area Air Quality Management District's complaint line (1 800 334 6367) shall also be included on posted signs to ensure compliance with applicable regulations.
		• All excavation, grading, and/or demolition activities shall be suspended when average wind speeds exceed 20 mph.
		• Wind breaks (e.g., trees or fences) shall be installed on the windward side(s) of actively disturbed areas of construction. Wind breaks should have at maximum 50 percent air porosity.
		• Vegetative ground cover (e.g., fast-germinating native grass seed) shall be planted in disturbed areas as soon as possible and watered appropriately until vegetation is established.
		• The simultaneous occurrence of excavation, grading, and ground- disturbing construction activities on the same area at any one time shall be limited. Activities shall be phased to reduce the amount of disturbed surfaces at any one time.
		• All trucks and equipment, including their tires, shall be washed off before leaving the site.
		• Site access from the paved road to a distance of 100 feet shall be treated with a 6- to 12-inch compacted layer of wood chips, mulch, or gravel.

### **Table 1-4 Other Project Features**

Resource	Feature Number	Description
Air Quality	PF-AIR-02	<b>Construction Best Practices for Exhaust.</b> The following are BMPs from Mitigation Measure 2.2 2 in the Addendum to Final Environmental Impact Report Plan Bay Area 2050 (ABAG and MTC 2021b). These measures control exhaust during any construction period that involves ground disturbance.
		<ul> <li>Road equipment greater than 25 horsepower that would be operated for more than 20 hours over the entire duration of construction will include the following requirements: (1) be zero emissions; OR (2) have engines that meet or exceed either United States Environmental Protection Agency (U.S. EPA) or California Air Resources Board (CARB) Tier 2 off- road emission standards; AND (3) have engines that are retrofitted with an CARB Level 3 Verified Diesel Emissions Control Strategy (VDECS), if one is available for the equipment being used. Equipment with engines that meet Tier 4 Interim or Tier 4 Final emission standards automatically meet this requirement; therefore, a VDECS would not be required.</li> </ul>
		• Idling time of diesel-powered construction equipment and trucks shall be limited to no more than 2 minutes. Clear signage of this idling restriction shall be provided for construction workers at all access points.
		<ul> <li>All construction equipment shall be maintained and properly tuned in accordance with the manufacturers' specifications.</li> </ul>
		• Portable diesel generators shall be prohibited. Grid power electricity should be used to provide power at construction sites; propane and natural gas generators may be used when grid power electricity is not feasible.
Biological	PF-BIO-01	<b>Environmentally Sensitive Area Delineation.</b> Before the start of construction, environmentally sensitive areas, including wetlands and habitats suitable for sensitive species, will be shown on the project plans. The bid solicitation package special provisions will specify acceptable fencing material and prohibited construction-related activities in these areas. Prior to construction in or near environmentally sensitive areas, a project biologist will delineate them in the field using signage, flagging, wildlife exclusion fencing (WEF), or other site markers as appropriate.
Biological	PF-BIO-02	<b>Wildlife Exclusion Fencing.</b> Before ground-disturbing activities commence, high-visibility WEF (suitable for amphibian and small mammal exclusion) will be installed along environmentally sensitive area boundaries to protect special-status animal species and to keep them from entering the project footprint. Maintenance of the WEF shall happen regularly and as requested by the project biologist in coordination with the Resident Engineer. Repair and maintenance costs for the fence shall be a bid item in the project contract.
Biological	PF-BIO-04	<b>Site Restoration.</b> All temporarily disturbed areas and staging areas will be cleaned up and recontoured to original grade or designed contours. All construction-related materials will be removed after construction, site clean-up, and restoration activities are complete. Temporarily impacted areas where vegetation was removed will be revegetated within one growing season of completion of project activities.

Resource	Feature Number	Description
Biological	PF-BIO-05	Landscaping and Revegetation Plan. Vegetation and trees removed by construction operations within the project limits will be replaced according to Caltrans policy. Appropriate native species will be used to the maximum extent possible, and trees, shrubs, and groundcover will be selected for drought tolerance and disease resistance. Mulch will be applied to planted areas to reduce weed growth, conserve moisture, and minimize maintenance operations. A 3-year plant establishment period will be included in the final revegetation plan. Caltrans will develop and implement a 5- to 10-year post-construction vegetation monitoring plan for planted areas.
Biological	PF-BIO-07	<b>Approved Project Biologist.</b> Prior to initiation of the construction, the qualifications of the biological monitor(s) will be submitted to USFWS, the National Marine Fisheries Service (NMFS), and CDFW, as appropriate for the respective jurisdictions, for approval. Such approved biologists are hereafter referred to as the project biologist(s).
Biological	PF-BIO-08	<b>Biological Monitoring.</b> The project biologist(s) will be on site during initial ground-disturbing activities in previously undisturbed areas, during work that occurs in wetlands or in waters below mean higher high water (MHHW) elevation, and thereafter as needed to fulfill the role of the project biologist as specified in project permits. The project biologist(s) will keep copies of applicable permits in their possession when on site. Prior to any initial ground-disturbing activity, the project biologist(s) will conduct work site surveys for the presence of special-status plant and animal species no less than 48 hours before work. The project biologist(s) will implement appropriate avoidance measures in the field and in coordination with the Resident Engineer to ensure that any identified special-status species or environmentally sensitive areas are clearly marked for avoidance.
Biological	PF-BIO-09	<b>Staging Areas.</b> Vehicle, barge, and equipment staging will be restricted to the areas reviewed, analyzed, and considered during the environmental review process. If new staging areas are required, they will require their own environmental review for potential impacts and may require additional regulatory action.
Biological	PF-BIO-10	<ul> <li>Construction Site Best Management. The following site restrictions will be implemented to avoid or minimize potential effects on listed species and their habitats, pursuant to Caltrans Standard Specifications and Special Provisions.</li> <li>Speed Limit. Vehicles will not exceed 15 mph in the project footprint, to reduce dust and excessive soil disturbance.</li> </ul>
		<ul> <li>Trash Control. Food and food-related trash items will be secured in sealed trash containers and removed from the site at the end of each day.</li> <li>Pets. Pets will be prohibited from entering the project limits during construction.</li> </ul>
		<ul> <li>Firearms. Firearms will be prohibited within the project limits, except for those carried by authorized security personnel or local, state, or federal law enforcement officials.</li> </ul>

Resource	Feature Number	Description	
Biological	PF-BIO-11	<b>Vegetation Removal.</b> Native vegetation will be cleared only when necessary and will be cut above soil level, except in areas that will be excavated. A truck with a chipper will be used for chipping the removed trees. All vegetation will be conducted within appropriate species protection work windows.	
Biological	PF-BIO-12	<b>Tree Protection.</b> Only trees that require removal will be removed. Whenever possible, trees will be trimmed rather than removed. Retained trees will be safeguarded during construction through the following measures:	
		<ul> <li>Protected trees will be fenced around the drip line to limit construction impacts to the root zone.</li> </ul>	
		<ul> <li>No construction equipment, vehicles, or materials will be stored, parked, or staged within the tree dripline.</li> </ul>	
		• Work will not be performed within the dripline of the remaining trees without consultation with the project biologist. If trees are damaged during construction and become unhealthy or die, the damaged tree(s) will be removed and replaced.	
Biological	PF-BIO-13	<ul> <li>construction site in accordance with Caltrans' Highway Design Manual Topic 110.5, "Control of Noxious Weeds – Exotic and Invasive Species," and EO 13112 (Invasive Species), and by methods approved by a Caltrans' landscape architect or vegetation control specialist.</li> <li>To minimize the spread of nonnative invasive plant (NNIP), any borrow material, erosion-control material (i.e., fiber rolls), and seed mixtures for erosion control will meet the following Caltrans (2018) specifications as they relate to NNIP species, including:</li> <li>Eiber roll must be a premanufactured and roll filled with rise or wheet</li> </ul>	
		<ul> <li>Fiber roll must be a premanufactured and roll-filled with rice or wheat straw, wood excelsior, or coconut fiber. Fiber roll must be covered with biodegradable jute, sisal, or coir fiber netting secured tightly at each end. Fiber rolls must be certified to be free of prohibited noxious weeds (those Rated "A" by California Department of Food and Agriculture [CDFA]).</li> <li>Imported topsoil must be free from deleterious substances such as litter, refuse, toxic waste, stones larger than 1 inch in size, coarse sand, heavy</li> </ul>	
		or stiff clay, brush, sticks, grasses, roots, noxious weed seed, weeds, and other substances detrimental to plant, animal, and human health.	
		<ul> <li>Seed must not contain any prohibited noxious weed seed, or more than 1.0 percent total weed seed by weight.</li> </ul>	
		<ul> <li>All equipment brought into work areas will be free of soil and plant matter.</li> </ul>	
		<ul> <li>In work areas where CDFA-listed noxious weeds or California Invasive Plant Council Moderate- or High-Rated NNIP species occur in fruit or flower and may spread seed as a result of the project, these NNIP species will be removed to an approved offsite disposal location.</li> </ul>	
Biological	PF-BIO-14	<b>Erosion Control Matting.</b> Plastic monofilament netting or similar material will not be used. Acceptable substitutes would include coconut coir matting or tackifying hydroseeding compounds.	

Resource	Feature Number	Description
Biological	PF-BIO-15	<b>Construction Lighting and Signage.</b> Construction area lighting will be used only where necessary for safety and signage. Downcast lighting and shielding to minimize lighting of natural areas will be used throughout the project footprint.
Biological	PF-BIO-16	<b>Prevent Wildlife Entrapment.</b> To prevent inadvertent entrapment of special status animal species during construction, excavated holes or trenches more than 1 foot deep with walls steeper than 30 degrees will be covered at the close of each working day by plywood or similar materials. Alternatively, an additional 4-foot-high vertical barrier, independent of exclusionary fences, will be used to further prevent the inadvertent animal entrapment. If it is not feasible to cover an excavation or provide an additional 4-foot-high vertical barrier, independent of exclusionary fences, one or more escape ramps constructed of earthen fill or wooden planks will be installed. Before such holes or trenches are filled, they will be thoroughly inspected for trapped animals. If at any time a trapped animal is discovered, the project biologist will be contacted, and they or their designee will immediately place escape ramps or other appropriate structures to allow the animal to escape, or USFWS and/or CDFW will be contacted by telephone for guidance as appropriate.
Cultural Resources	PF-CULT- 01	<b>Discovery of Human Remains.</b> If human remains are discovered, all work within 60 feet of the remains will be halted, and the Caltrans' Office of Cultural Resource Studies will be notified. Caltrans OCRS will assess the remains, and if determined to be human, will contact the County Coroner, in accordance with California Health and Safety Code (H&SC) Section 5097.98, 5097.99, and 7050.5 If the remains are thought by the coroner to be Native American, the coroner will notify the Native American Heritage Commission (NAHC), who, pursuant to PRC Section 5097.98, will then notify the Most Likely Descendent (MLD). Caltrans will consult with the MLD on the respectful treatment and disposition of the remains. Further provisions of PRC 5097.98 are to be followed as applicable.
Cultural Resources	PF-CULT- 02	<b>Discovery of Archeological Materials.</b> If cultural materials are discovered during construction, all earth-moving activity within 60 feet of the discovery will be diverted until a qualified archaeologist can assess the nature and significance of the find.
Geology/ Soils/Seismic/ Topography	PF-GEO-01	<b>Geotech Investigations.</b> A geotechnical investigation will be performed during final design for any proposed new earthwork or new structure within the project limits, including retaining walls, overhead signs, embankments, bridges, and sound walls; it will address geologic hazards, including liquefaction, cracking, differential compaction, ground shaking, and shrink swell.
Geology/ Soils/Seismic/ Topography	PF-GEO-02	Seismic Standards. Caltrans' design and construction guidelines incorporate engineering standards that address seismic risks. project elements will be designed and constructed to meet seismic design requirements for ground shaking and ground motions, as determined for the project vicinity and site conditions.

Resource	Feature Number	Description
Hazardous Materials	PF-HAZ-01	<b>Site Investigations.</b> A preliminary site investigation (PSI) for aerially deposited lead, agricultural chemicals, and unexploded ordnance (UXO) would be conducted during the project design phase. A PSI will be performed to investigate hazardous materials concerns related to soil, groundwater, and building materials within the project limits. Caltrans will prepare a work plan for the PSI. The findings of the PSI will be used to evaluate soil and groundwater handling practices, construction worker health and safety concerns, and soil and groundwater reuse and disposal options. If hazardous materials are identified during the PSI, additional investigation would be required to their full evaluation. All environmental investigations for the project will be provided to project contractors so the findings can be incorporated into their Health and Safety and Hazard Communication Programs.
Hazardous Materials	PF-HAZ-02	<b>Health and Safety Management Plan.</b> A Health and Safety Management Plan would be prepared to outline procedures if UXO, mustard gas, or similar military hazards are encountered within the project limits.
Hazardous Materials	PF-HAZ-03	<b>Soil and Groundwater Management Plan.</b> A Soil and Groundwater Management Plan would be prepared to properly manage any impacted soil or groundwater discovered during ground-disturbing activities within the project limits.
Hazardous Materials	PF-HAZ-04	<b>Hazardous Structure Material Surveys.</b> Hazardous Structure Material Surveys would be conducted for asbestos-containing material, lead-based paint, treated-wood waste, and polychlorinated biphenyls (PCBs).
Noise	PF-NOI-01	<b>Construction Noise.</b> The Caltrans 2018 Standard Specifications, Section 14 8.02, requires that the Maximum Sound Level not exceed 86 A-weighted decibels at 50 feet from the job site, from 9:00 p.m. to 6:00 a.m. Construction noise would not exceed thresholds or Caltrans' standards. Construction noise control measures would be required of the contractor. These include control measures for equipment and operating hours such as:
		All construction equipment shall conform to Section 14 8.02, Noise Control, of the latest Standard Specifications.
		Noise-generating construction activities shall be restricted to between 7:00 a.m. and 7:00 p.m. on weekdays, with no construction occurring on weekends or holidays. If work is necessary outside of these hours, Caltrans shall require the contractor to implement a construction noise monitoring program and provide additional noise controls where practical and feasible. All internal-combustion-engine-driven equipment shall be equipped with manufacturer-recommended intake and exhaust mufflers that are in good condition and appropriate for the equipment.
		Unnecessary idling of internal combustion engines within 100 feet of residences shall be strictly prohibited.
		receptors when sensitive receptors adjoin or are near the construction project area.
		"Quiet" air compressors and other "quiet" equipment shall be used where such technology exists.

Resource	Feature Number	Description	
Paleontology	PF-PAL-01	<b>Discovery of Paleontological Resources</b> . If paleontological resources are discovered during construction, all earth-moving activity within and around the immediate discovery area will be diverted until a qualified paleontologist can assess the nature and significance of the find.	
Paleontology	PF-PAL-02	<b>Paleontological Mitigation Plan</b> . During the project design phase, Caltrans will determine whether or not a Paleontological Mitigation Plan (PMP) is needed.	
Water Quality	PF-WQ-01	Water Quality Best Management Practices. The contractor will adhere to the instructions, protocols, and specifications outlined in the most current Caltrans Construction Site Best Management Practices Manual and Caltrans Standard Specifications. At a minimum, protective measures will include the following:	
		<ul> <li>The discharging of pollutants from vehicle and equipment cleaning into storm drains or watercourses will be disallowed.</li> </ul>	
		<ul> <li>Storing or servicing vehicles and construction equipment including fueling, cleaning and maintenance, will be performed at least 50 feet from aquatic habitat unless separated by a topographic or drainage barrier.</li> </ul>	
		• Equipment will be maintained to prevent the leakage of vehicle fluids such as gasoline, oils, or solvents, and a spill response plan will be developed. Hazardous materials such as fuels, oils, or solvents, will be stored in sealable containers in a designated location that is at least 50 feet from aquatic habitats.	
		<ul> <li>Concrete wastes and water from curing operations will be collected and disposed of in appropriate washouts at least 50 feet from watercourses.</li> </ul>	
		<ul> <li>Temporary stockpiles will be covered.</li> </ul>	
		<ul> <li>Coir rolls or straw wattles will be installed along or at the base of slopes during construction to capture sediment.</li> </ul>	
		<ul> <li>Graded areas will be protected from erosion using a combination of silt fences, fiber rolls, and erosion control netting (jute or coir), as appropriate.</li> </ul>	
Water Quality	PF-WQ-02	<b>Temporary Dewatering Activities.</b> Groundwater extracted from temporary dewatering activities will be managed based on the groundwater quality in the project area. Clean groundwater could be used for dust control, collected on site using desilting basins and/or tanks prior to discharging to receiving waters, and transported to a publicly owned treatment works.	
Water Quality	PF-WQ-03	<b>Groundwater Treatment.</b> If the project area contains contaminated groundwater or groundwater that may release contaminated plumes when disturbed, applicable permits and authorizations from the RWQCB would be obtained during the project's final design phase. An active treatment system will be implemented, as necessary and appropriate to treat contaminated groundwater exposed during excavation activities. Dewatering requirements and design of any necessary active treatment system would be determined during the project's final design or during construction.	

Resource	Feature Number	Description	
Water Quality	PF-WQ-04	Inclement Weather Restriction. No new ground-disturbing work will occur during or within 24 hours of a rain event exceeding 0.2 inch, as measured by the National Oceanic and Atmospheric Administration Weather Service for Novato/Gnoss Field, California KDVO (NWS/FAA-MTR) base station available at: https://www.wrh.noaa.gov/mesowest/getobext.php?wfo=mtr&sid=KDVO#=72&raw=0. Approval from the state and/or federal agencies, as required in project permits to continue work during or within 24 hours of a rain event, will be considered on a case-by-case basis.	
Wildfire	PR-WF-01	Project Features for Minimizing Fire Risks. BMPs would be incorporated such as clearing vegetation from the work area, prohibiting the use of highly flammable chemicals, following locally changing meteorological conditions, and maintaining awareness of the possibility of increased fire danger during the time work is in progress.	

Agency	Permit/Approval/Acknowledgement	Status
FHWA	Concurrence with project's conformity to the Transportation Improvement Program (TIP) under the federal Clean Air Act and other requirements. The TIP is a supporting document of the MTC 2050 Regional Transportation Plan.	<ul> <li>FHWA concurred on November 1, 2022, with Caltrans' air quality conformity determination, and that the project conforms with the State Implementation Plan (SIP).</li> <li>The Interagency Air Quality Conformity Task Force determined that the project is not a Project of Air Quality Concern (POAQC) at their May 22, 2021, meeting.</li> </ul>
USFWS	<ul> <li>Endangered Species Act Section 7 consultation for threatened and endangered species (terrestrial)</li> <li>United States Department of Transportation (USDOT) Section 4(f) Use Concurrence</li> </ul>	<ul> <li>Caltrans will continue Section 7 consultation, including submittal of a Biological Assessment to USFWS. Consultation will continue into the design phase.</li> <li>Based on the preferred alternative, USFWS will issue either a letter of concurrence with the findings of effect in the Biological Assessment, or a biological opinion which may authorize take of federally listed species to Caltrans.</li> <li>Right-of-way acquisition and TCEs will be sought during the design phase. Sliver acquisitions and TCEs are necessary to accommodate widening and temporary construction in refuge lands. USFWS has concurred with the Section 4(f) de minimis use</li> </ul>
NMFS	<ul> <li>Endangered Species Act Section 7 consultation for threatened and endangered species (fish)</li> <li>Magnuson-Stevens Fishery Conservation Management Act consultation for Essential Fish Habitat</li> </ul>	<ul> <li>Caltrans has initiated consultation with NMFS, including submittal of a Biological Assessment. Consultation will continue into the design phase. The Biological Assessment includes analysis of and request for consultation for Essential Fish Habitat impacts, as appropriate.</li> <li>Based on the preferred alternative, NMFS will issue either a letter of concurrence with the findings of effect in the Biological Assessment, or a Biological Opinion allowing take of federally listed species to Caltrans.</li> </ul>

Table 1-5 FIDJECL FEITING and Approval	Table 1-5	Project	Permits	and	Approvals
--	-----------	---------	---------	-----	-----------

Agency	Permit/Approval/Acknowledgement	Status
State Historic Preservation Officer (SHPO)	National Historic Preservation Act (NHPA) Section 106 consultation	• Caltrans' consultation on identification of cultural resources eligible for the National Register of Historic Places was completed on December 23, 2016.
		<ul> <li>Results of studies were submitted to SHPO. No comments were received, and consultation was completed.</li> </ul>
United States Army Corps of Engineers	<ul> <li>Preliminary Jurisdictional Determination for jurisdictional wetlands and waters of the United</li> </ul>	<ul> <li>A wetland delineation was completed and will be submitted to USACE for concurrence.</li> </ul>
(USACE)	<ul> <li>States</li> <li>Clean Water Act Section 404 and Rivers and Harbors Act Section 10 permit for placement of fill in waters of the United States</li> </ul>	<ul> <li>A permit application will be submitted during the project design phase.</li> </ul>
USCG	USCG Bridge Permit Amendment	<ul> <li>Alternative 3B has been modified to minimize in-water work and would require only temporary access along the Sonoma Creek shoreline and channel area. A Bridge Permit is not anticipated for permanent changes. Caltrans will consult with USCG during final design to verify that no vertical or horizontal clearances impair the navigation channel, and for any agreements needed for temporary construction activities.</li> </ul>
Interagency Air Quality Conformity Task Force	<ul> <li>Concurrence that the project is not a POAQC</li> </ul>	<ul> <li>MTC's Air Quality Control Task Force determined the project is not a POAQC on May 22, 2021.</li> </ul>
San Francisco Bay RWQCB	<ul> <li>Section 401 Water Quality Certification or waiver and or/Porter Cologne Act Waste Discharge Requirements</li> <li>Compliance with the NPDES Construction General Permit</li> </ul>	<ul> <li>A joint "Application for 401 Water Quality Certification" and/or "Report of Waste Discharge" will be submitted during the project design phase.</li> <li>A statewide NPDES permit for construction and operations will be in effect for the project. Compliance regime will take place during the</li> </ul>
		review will take place during the design phase.

Agency	Permit/Approval/Acknowledgement	Status
CDFW	<ul> <li>California Fish and Game Code (CFGC) Section 1602 Lake and Streambed Alteration Notification for affects to nontidal channels and water bodies</li> <li>CFGC Section 2081 Incidental Take Permit if take of state-listed species is anticipated</li> <li>USDOT Section 4(f) Use Agreement/ Concurrence</li> </ul>	<ul> <li>An application for a CFGC Section 1602 Lake and Streambed Alteration Agreement will be submitted during the project design phase.</li> <li>An application for a CFGC Section 2081 Incidental Take Permit will be submitted during the project design phase, if required.</li> <li>Right-of-way acquisition and TCEs would be sought during the design phase. Sliver acquisitions and TCEs are necessary to accommodate widening and temporary construction within NSMWA lands. CDFW has concurred with the Section 4(f) de minimis use.</li> </ul>
San Francisco Bay Conservation and Development Commission (BCDC)	<ul> <li>A BCDC permit would be required for development within BCDC jurisdictions.</li> </ul>	The BCDC permit application will be submitted during the design phase.
SMART (Northwestern Pacific Railroad line)	<ul> <li>A railroad agreement may be required for work at the crossing near Tolay Creek Bridge.</li> <li>General Order 88-B Approval for Modification of an Existing Rail Crossing</li> </ul>	An application will be submitted during the design phase.
California Transportation Commission	Approval to implement tolling	This approval will be requested after the environmental review phase is completed.

## Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

### 2.1 Topics Considered but Determined Not to be Relevant

As part of the scoping and environmental analysis carried out for the project, the following environmental issues were considered, but no adverse impacts were identified. These include Timberlands, Mineral Resources, and Wild and Scenic Rivers. These topics were eliminated because these resources are not present within or near the project area. As a result, there is no further discussion about these issues in this document.

### 2.2 Human Environment

### 2.2.1 Existing and Future Land Use

### 2.2.1.1 Affected Environment

The following section is based on the *Community Impact Assessment* prepared for the project (AECOM 2021f).

The project would occur primarily within Caltrans' right-of-way along SR 37. This corridor crosses through the City of Vallejo and the counties of Solano, Napa, and Sonoma. This section describes existing land uses within the project limits and general vicinity, and designated land uses as shown in each local or regional land use plan. In general, existing land uses adjacent to SR 37 consist primarily of natural resource and open space areas.

In Vallejo, existing land uses adjacent to SR 37 include public facilities and parks, recreation and open space. In Solano County, adjacent land uses mainly consist of marsh designated areas (County of Solano 2008). In Napa County, adjacent land uses consist of agriculture, watershed, and open space. In Sonoma County, adjacent land uses consist of agriculture and recreation/visitor-serving commercial. SR 37 is also next to the Refuge and the NSMWA. Parks and community uses are discussed further in Section 2.2.4, Parks and Recreation Facilities.

Future planned developments within 1 mile of the project area are described below in Table 2-1. The information in Table 2-1 was obtained from CEQAnet (2020) and the planning departments for the counties of Solano, Napa, and Sonoma and the City of Vallejo. The land uses in Solano, Napa, and Sonoma counties adjacent to SR 37 are generally not designated for development, so development proposals are limited to the City of Vallejo and Mare Island at the eastern extent of the project corridor. For this reason, only current and planned development in City of Vallejo and Mare Island are presented in Table 2-1.

### Table 2-1 Current and Proposed Planned Developments within One Mile of the<br/>Project Area

Name	Proposed Uses	Location (Approximate Distance from Project Area)
City of Vallejo/Waterfront Project	175 single-family detached residences, commercial areas, and two parks. EIR certified in 2005.	Between Mare Island Way and Mare Island Causeway (1 mile southeast)
City of Vallejo/North Mare Island	Film production, wine and beverage manufacturing, office, and retail use.	Adjacent to the south side of SR 37
City of Vallejo/Mare Island	Mixed-use development.	South of G Street (1 mile south)

Sources: City of Vallejo 2019, 2020

### 2.2.1.2 Environmental Consequences

#### No Build Alternative

Because the No Build Alternative would not result in changes to SR 37, it would not conflict with any existing land use designations or preclude the development of any proposed projects.

### **Build Alternatives**

The project would require both temporary use and partial acquisition of areas along SR 37, including portions of parcels in the Refuge and NSMWA. Potential property acquisitions are described in Section 2.2.8. No full parcel acquisition would be required, and the partial acquisitions would not affect existing land uses in the project area.

Each of the Build Alternatives would require TCEs for construction of the project. TCEs may be needed for construction at SR 121, Tolay Creek Bridge, Noble Road, the Cullinan Ranch public access intersection, and other private access driveways, including areas in the NSMWA and/or the Refuge, to provide construction access. The TCEs required construction of the Build Alternatives would not result in changes to existing land uses on any of the affected parcels, because only a limited work area would be required for a limited period of time (one construction season). Following completion of construction, the affected TCE parcels would be restored to pre-project conditions. Construction would have no effect on the zoning and land use designations of the TCE parcels. Because the affected parcels would be restored, no permanent change to any land use would result.

As described in Section 2.2.8, Build Alternatives 3A and 3B would also require permanent right-of-way acquisition at the Refuge to facilitate widening of the existing roadway. The partial acquisitions would occur along the edge of the Refuge where it is bisected by the existing roadway and provides limited recreational value. Implementation of Build Alternatives 3A or 3B would not affect the existing land uses of the rest of the Refuge area. Therefore, project construction and operation would not result in major changes to the land use or zoning of any parcels in the project area.

The Build Alternatives may provide modified access along the SR 37 corridor but would not provide additional access points. As a result, the project would not open new areas to development. The project would not conflict with any existing or planned land use, or preclude the development of the proposed development projects listed above in Table 2-1.

### 2.2.1.3 Avoidance, Minimization, and/or Mitigation Measures

Acquisitions and TCEs would require compensation. Caltrans would follow the process for required under the federal Uniform Relocation Assistance Program, which allows for the uniform, fair, and equitable treatment of persons and agencies whose real property is acquired in connection with federally funded projects.

### 2.2.2 Consistency with State, Regional, and Local Plans and Programs

### 2.2.2.1 Affected Environment

There are several community, regional, and transportation plans that apply to the project study area. The following types of plans were considered and are discussed below:

- Transportation plans/programs
- Regional growth plans
- General plans and related plans
- Habitat conservation plans (HCPs)
- Other planning influences

### Transportation Plans/Programs

*Plan Bay Area 2050* is the RTP for the nine-county San Francisco Bay Area (ABAG and MTC 2021a; RTP ID 21-T06-035). The RTP lists projects of local and regional importance based on factors such as local support and need, ridership, and potential cost and funding. These factors provide direction on how anticipated federal, state, and local transportation funds would be spent in the Bay Area during the next 20 years. The project is included in the Plan Bay Area 2050.

Caltrans' statewide *Toward an Active California, State Bicycle and Pedestrian Plan* supports travel by bicyclists and pedestrians through objectives, strategies, and actions to facilitate the provision of robust multimodal transportation options in the state (Caltrans 2017c). The plan aims to achieve the goals in the Caltrans Strategic Management Plan—specifically related to increasing pedestrian and bicycle trips and safety, and the promotion of complete streets—through four plan objectives related to safety, mobility, preservation, and social equity. These four plan objectives are supported by fifteen strategies and sixty actions specific to the active transportation modes of walking and biking.

The Caltrans Complete Streets Program was implemented pursuant to Caltrans Deputy Directive 64-R2 (Caltrans 2014b). As provided for in that directive, "The

Department provides for the needs of travelers of all ages and abilities in all planning, programming, design, construction, operations, and maintenance activities and products on the State Highway System." The Caltrans Complete Streets Program describes a complete street as a transportation facility that is planned, designed, operated, and maintained to provide safe mobility for all users, including bicyclists, pedestrians, transit vehicles, truckers, and motorists, appropriate to the function and context of the facility. A complete street is designed according to its context, community preferences, the types of road users, and their needs.

The Contextual Guidance for Bike Facilities memorandum dated March 11, 2020, provides guidance for the design and selection of bicycle facilities based on placetype, speed, and volume (Caltrans 2020d). In accordance with that memorandum, routes traversing rural areas with posted speeds of greater than 45 mph warrant development of a Class III bikeway (a right-of-way on-street or off-street, designated by signs or permanent markings and shared with pedestrians and motorists), and routes in urban or suburban areas warrant a Class I bikeway (a completely separated right-of-way designated for the exclusive use of bicycles and pedestrians with crossflows by motorists minimized) or Class IV bikeway (separated bikeways which promote active transportation and provides a right-of-way designated exclusively for bicycle travel adjacent to a roadway and which are separated from vehicular traffic) (SHC Section 890.4).

The Sonoma County Comprehensive Transportation Plan identifies SR 37 as the fourth worst congested area in Sonoma County in 2007 (SCTA 2016). This plan recognizes the need for highway widening as well as addressing the threat of long-term SLR. The Napa Countywide Transportation Plan recognizes SR 37 as a principal thoroughfare that is not adequate for current traffic volumes, causes drivers to use alternative more circuitous routes, and increases VMT (NVTA 2015). The Solano County Comprehensive Transportation Plan includes similar mention of the conditions on SR 37, and the need for regional funding of improvements (STA 2020).

### **Regional Growth Plans**

Plan Bay Area 2050 (ABAG and MTC 2021a) also functions as a regional growth plan for the nine-county San Francisco Bay Area. Plan Bay Area 2050 designates priority development areas (PDAs), which are areas in existing communities that have been identified and approved by a local city or county for future growth because of proximity to transit, jobs, shopping, and other services. Promoting compact development in PDAs is intended to take development pressure off the region's open space and agricultural lands (ABAG and MTC 2021a).

There are two PDAs within 1 mile of the project area (ABAG 2020). The Mare Island PDA is adjacent to the project area, and the Waterfront and Downtown PDA is approximately 0.85 mile from the project area.

• The Mare Island PDA is an 811-acre area adjacent to the project area south of the Mare Island Interchange. It contains mainly industrial and residential uses,

and limited office uses. The Mare Island PDA is not currently served by any public transportation except for one ferry terminal.

• The Waterfront and Downtown PDA is a 196-acre area south of SR 37 and west of Mare Island Way in Vallejo. The PDA includes retail, restaurants, office, and residential uses. A portion of the PDA is included in the Waterfront Project that would develop the area as a mixed-use community on the city's waterfront. The PDA is served by the SolTrans bus service.

### General and Local Plans

General and local plans were reviewed for the jurisdictions in the project area, including Solano, Napa, and Sonoma counties and the City of Vallejo. The planning documents listed below were also considered. However, these plans do not include any objectives, goals, or policies that are applicable to the proposed project and the proposed project does not include any features that are within the jurisdiction of these plans. Therefore, the following documents were reviewed, but they were not further evaluated in this EIR/EA:

- Mare Island Specific Plan (City of Vallejo 2013)
- Napa-Sonoma Marshes Wildlife Area Land Management Plan (CDFW 2011)
- Napa County General Plan (County of Napa 2008)

### Solano County General Plan

The Solano County General Plan (adopted in 2008) covers 910 square miles, including 830 square miles of land and 80 square miles of water. The Solano County General Plan extends from Yolo County to the north, Sacramento metropolitan regions to the east, San Francisco metropolitan regions to the south, and San Pablo Bay and Napa County to the west. The Solano County General Plan contains the following goal that relates to the project (County of Solano 2008):

• Goal: TC.G-2: Promote coordinated approaches to creating, maintaining, and improving transportation corridors and facilities by working with other jurisdictions and transportation agencies in funding and implementing projects.

### City of Vallejo General Plan

The *City of Vallejo General Plan* (adopted in 2017) covers 50 square miles of western Solano County, from unincorporated Solano County to the northeast, Carquinez Strait to the south, and Napa/Sonoma Marshes to the west.

The City of Vallejo General Plan contains the following policy that relates to the project (City of Vallejo 2017):

• Action MTC-3.1A: Work with Caltrans, Solano County, Soltrans, and STA to identify and seek funding for improvements that make intra-city travel easier, including for transit, bicycles, and pedestrians.

### Sonoma County General Plan

The *Sonoma County General Plan 2020* (adopted in 2008) covers 1,500 square miles, bordered by the Solano, Napa, and Lake Counties to the east, Marin County and San Pablo Bay to the south, the Pacific Ocean to the west, and Mendocino County to the north.

The Sonoma County General Plan contains goals and policies that relate to the project (County of Sonoma 2020) such as ones that encourage:

- ridesharing;
- HOV Lanes;
- bikeway network linkages among cities, communities, and major activity centers;
- development and enhancement of new and existing bikeways consistent with appropriate standards;
- coordination with Caltrans on traffic management improvements along Highway 37 to reduce congestion consistent with the designated road classifications; and
- expansion of the County's zero net fill requirements within the 100-year Federal Emergency Management Act (FEMA) special flood hazard area (SFHA).

### San Francisco Bay Conservation and Development Commission (BCDC)

Most of the project is within BCDC jurisdiction, excepting the project area east of Tolay Creek, and on Tubb's Island between Sonoma Creek Bridge and the Tolay Creek wetlands/Tubbs Island Trailhead parking area south of SR 37. Project alternatives may potentially intersect with the BCDC's Bay, shoreline band, salt ponds, managed wetlands, and certain waterways (i.e., Tolay Creek and Sonoma Creek) jurisdictions. Fill within BCDC jurisdiction would require that the project demonstrate its consistency with BCDC's laws, regulations, and policies for fill in the Bay to obtain a BCDC permit. Any development project in the shoreline band is required to provide "maximum feasible public access consistent with the proposed project" to obtain a BCDC permit.

The BCDC San Francisco Bay Plan (referred to hereafter as Bay Plan) is a comprehensive and enforceable plan for the conservation of the water of the Bay and the development of its shoreline. BCDC developed Bay Plan policies and maps for the Bay. SR 37 is defined as a scenic drive and is adjacent to areas of priority use as wildlife refuge, tidal marsh, and salt pond or managed wetlands in BCDC's Bay Plan Map 1 – San Pablo Bay, and Bay Plan Map 2 – Carquinez Strait.

### **Habitat Conservation Plans**

Solano County is in the process of finalizing the *Solano County Multispecies HCP*, which overlaps with a portion of the project area in Solano County. The draft *Solano* 

*County Multispecies HCP* proposes the conservation of 36 special status species across approximately 585,000 acres in Solano County and Yolo County (LSA Associates, Inc. 2012). The proposed project would require TCEs in areas that are expected to be subject to the HCP once it is finalized. However, the proposed project would not preclude implementation of the proposed HCP or its conservation goals.

The *Pacific Gas and Electric Company (PG&E) Bay Area Operations and Maintenance HCP* also overlaps with a portion of the project area. However, because the plan is specific to PG&E operation and maintenance activities, it does not contain policies or goals related to the proposed project (USFWS 2017a).

### State Scenic Highway Program

Though SR 37 is an Eligible State Scenic Highway in Solano and Sonoma counties, it is not an Officially Designated State Scenic Highway. The visual impacts of the project are addressed in Section 2.2.12.

### **Bicycle Plans**

The proposed improvements along the SR 37 corridor project coincide with the planned alignment of the San Francisco Bay Trail. The Association of Bay Area Governments (ABAG) adopted the Bay Trail Plan in 1989 to provide a 500-mile walking and bicycling path around the Bay. Since adoption of the plan, substantial portions of the planned Bay Trail alignment have been implemented, including in the vicinity of the proposed project. Various policies in the plan guide the design and construction of the trail promote: access to various modes of travel (Policy 12), physical separation of new trails from streets (Policy 13), and trail alignments sufficiently wide to reduce conflicts among trail users (Policy 14).

A portion of SR 37 in Solano County is included in the *Solano Countywide Bicycle Transportation Plan* (STA 2011). The portion of SR 37 from just east of the Mare Island Interchange to the Solano County/Sonoma County line is identified as a Class III bicycle route, which overlaps with the project area.

A portion of SR 37 in Sonoma County is also included in the *2010 Sonoma County Bicycle and Pedestrian Plan* (County of Sonoma 2010). The portion of SR 37 from the Solano County/Sonoma County line to just west of the SR 121 intersection is identified as a proposed Class II bicycle route, which overlaps with the project area.

Although the two above-mentioned Sonoma County plans include SR 37, the route is a State highway. Design of the highway and designation of uses are ultimately the responsibility of Caltrans.

### 2.2.2.2 Environmental Consequences

The project is almost entirely within Caltrans' right-of-way (a small amount of federal lands would be acquired for construction). The project is not within local jurisdictions and is not subject to city or county general plan policies. The proposed project is not included in Sonoma County's transportation plan, Napa County's transportation plan,

or Solano County's transportation plan. However, the project was included in the RTP Plan Bay Area 2040 for a multi-county evaluation of feasibility options to address congestion and future SLR, and these regional counties participated in the development of the project alternatives. The project is also included in the current Plan Bay Area 2050. City and county general plan policies have been included in this section to show that Caltrans has reviewed and considered these policies, and that Caltrans is partnering with local agencies to best satisfy their policies.

Table 2-2 summarizes the consistency of the No Build and Build Alternatives with applicable local plan policies. The Build Alternatives would support local policies promoting ride sharing and transit. Related to Sonoma County's General Plan policies on flooding, the project adds fill to a 100-year flood hazard area; however, the project would add fill outside of Sonoma County's jurisdiction and would not exacerbate existing conditions related to flooding in or outside of Caltrans' right-of-way, as discussed in the *Location Hydrology Report* for this project.

Build Alternatives 1 and 3B would continue to allow passage of bicycles along the shoulder. The No Build Alternative would also continue to allow bicycle access along the shoulder of the highway. Alternatives 2 and 3A would have narrower shoulders, including at the segment of the highway at Sonoma Creek Bridge, where shoulders would not be available. Legislation to prohibit bicycle use along the corridor would be proposed. Neither the No Build nor Build Alternatives propose new bicycle lanes.

### 2.2.2.3 Avoidance, Minimization, and/or Mitigation Measures

The existing SR 37 shoulders can accommodate bicycle use, consistent with a conventional highway that has shoulders. The No Build and Build Alternatives 1 and 3B would not substantially change this situation. Alternatives 2 and 3A are partially inconsistent, with narrowing of shoulders, specifically at the Sonoma Creek Bridge, due to a lack of width that can accommodate a shoulder. Measures are not available for Alternatives 2 and 3A that can change the availability of shoulder space at the Sonoma Creek Bridge.
# Table 2-2 Consistency of No Build and Build Alternatives with General Plans, Regional Plans, and TransportationPlans

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Solano County General Plan	Goal: TC.G-2: Promote coordinated approaches to creating, maintaining, and improving transportation corridors and facilities by working with other jurisdictions and transportation agencies in funding and implementing projects.	This alternative would not involve transportation improvements.	The transportation improvements associated with this alternative would be coordinated between Solano County and MTC.	The transportation improvements associated with this alternative would be coordinated between Solano County and MTC.	The transportation improvements associated with this alternative would be coordinated between Solano County and MTC.	The transportation improvements associated with this alternative would be coordinated between Solano County and MTC.
Solano County General Plan	Action MTC-3.1A: Work with Caltrans and local transportation sponsors to identify and seek funding for improvements that make intra-city travel easier, including for transit, bicycles, and pedestrians.	This alternative would not involve transportation improvements.	This alternative would improve travel time on SR 37 between cities.	This alternative would improve travel time on SR 37 between cities. Shoulders on Sonoma Creek Bridge would not be available at peak periods in the peak direction.	This alternative would improve travel time on SR 37 between cities. Shoulders on Sonoma Creek Bridge would not be available in either direction.	This alternative would improve travel time on SR 37 between cities.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Sonoma County General Plan	Policy CT-2t: Encourage measures that increase the average occupancy of vehicles, including: (1) Vanpools or carpools, ridesharing programs for employees, preferential parking, parking subsidies for rideshare vehicles, and transportation coordinator positions, and Circulation and Transit Element Footnote* Mitigating Policy Page CT-21 (2) Preferential parking space and fees for rideshare vehicles, flexibility in parking requirements, HOV lanes on freeways, and residential parking permit restrictions around major traffic generators.	This alternative would not modify SR 37 in such a way that average vehicle occupancy would increase.	This alternative would add an additional peak hour HOV lane to SR 37 to provide an incentive for a shift from SOVs.	This alternative would convert the existing shoulders to HOV lanes during peak periods to provide an incentive for a shift from SOVs.	This alternative would widen the highway and add an HOV lane in each direction to provide an incentive for a shift from SOVs.	This alternative would widen the highway and add an HOV lane in each direction to provide an incentive for a shift from SOVs.
Sonoma County General Plan	Objective CT-3.1: Design, construct and maintain a comprehensive Bikeways Network that links the County's cities, unincorporated communities, and other major activity centers including, but not limited to, schools, public facilities, commercial centers, recreational areas, and employment centers.	This alternative would not alter existing accommodations for bicyclists.	This alternative could accommodate bicyclists in the shoulder in both directions.	This alternative would not accommodate bicyclists at Sonoma Creek Bridge when during peak travel times in the peak flow direction.	This alternative would not accommodate bicyclists at Sonoma Creek Bridge in the shoulders at Sonoma Creek Bridge.	This alternative would accommodate bicyclists in the shoulder.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Sonoma County General Plan	Policy CT-3n: Use the following criteria to determine consistency of public and private projects with the Bikeways Plan: (1) Development of lands traversed or adjoined by an existing or future Class I bikeway shall not preclude establishment of the bikeway, nor conflict with use and operation of the bikeway or adversely affect long-term maintenance and safety of the facility. (2) Construction, widening, or maintenance of roads with designated bikeways meets the design and maintenance standards for the appropriate class of bikeway s Plan.	This alternative would not alter existing accommodations for bicyclists. The existing roadway has 8-foot- shoulders along most of the route but is not designated as a Class II bikeway except for a short westbound portion at the Cullinan Ranch Road intersection.	This alternative would not accommodate a Class II bicycle lane because the roadway shoulder would continue to be shared with motorists, including emergency vehicles.	This alternative would not accommodate a Class II bicycle lane because the roadway shoulder would continue to be shared with motorists, including emergency vehicles. The shoulder would not be available to bicycles at Sonoma Creek Bridge.	This alternative would not accommodate a Class II bicycle lane because the roadway shoulder would continue to be shared with motorists, including emergency vehicles. The shoulder would not be available to bicycles at Sonoma Creek Bridge.	This alternative would allow bicycles access in both directions within the entire project limits. There will be 8-foot-shoulders along the entire route (same as existing) except at Sonoma Creek Bridge which would have 4-foot- wide shoulders, and the 8- and 4-foot-wide shoulders would be open to bicycle use.
Sonoma County General Plan	Policy CT-7ii/ss: Work with Caltrans in considering turning lanes, access controls, and other traffic management improvements along Highway 37 to reduce congestion, provided that the improvements are consistent with the designated road classifications.	This alternative would not provide transportation improvements.	This alternative would increase the efficiency of the transportation system, with a goal of reducing traffic congestion and increasing person throughput on the corridor.	This alternative would increase the efficiency of the transportation system, with a goal of reducing traffic congestion and increasing person throughput on the corridor.	This alternative would increase the efficiency of the transportation system, with a goal of reducing traffic congestion and increasing person throughput on the corridor.	This alternative would increase the efficiency of the transportation system, with a goal of reducing traffic congestion and increasing person throughput on the corridor.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Sonoma County General Plan	Policy PS-2e: Expand the County's zero net fill requirements to address all areas of the unincorporated County that are located within the 100-year FEMA SFHA.	SR 37 is entirely in Caltrans' right-of-way and would not increase fill within Sonoma County's jurisdiction.	Alternative 1 is in Caltrans' right-of-way and would not add fill to Sonoma County's jurisdiction. Furthermore, the project would not create a flooding issue within the County's jurisdiction.	Same as Alternative 1.	Same as Alternative 1.	Same as Alternative 1.
Sonoma County General Plan	Policy PS-2i: Each discretionary project located in watersheds with major flood problems shall analyze drainage and flooding impacts and include feasible and appropriate mitigation measures to reduce flood hazards.	The No Build Alternative is existing, and is not a discretionary project.	This alternative is not a discretionary project subject to oversight of the County. However, flooding issues have been reviewed and would be addressed during the design phase of the project.	Same as Alternative 1.	Same as Alternative 1.	Same as Alternative 1.
Sonoma County General Plan	Policy PS-2I: On-site and off-site flood related hazards shall be reviewed for all projects located within areas subject to known flood hazards.	The No Build Alternative would not involve any changes subject to review.	Flood hazards have been reviewed and are discussed in this document.	Same as Alternative 1.	Same as Alternative 1.	Same as Alternative 1.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Bicycle Plans and Programs	Caltrans State Bicycle and Pedestrian Plan: Improve Public Safety and Security Develop local and regional networks of high-quality bicycle and pedestrian facilities Integrate bicycle and pedestrian needs in planning and design of multimodal transportation systems and services	The No Build Alternative would continue to allow for bicycle passage through the SR 37 corridor in the study area. However, this alternative would not promote or support the plan strategies and actions intended to achieve the goals related to safety and mobility.	This alternative would allow for continued bicycle passage through the SR 37 corridor in the study area, but would not fully support the relevant strategies and actions to promote safety and mobility.	This alternative would preclude bicyclists during peak travel periods in the peak direction. It would not provide adequate space for bicyclists on the Sonoma Creek Bridge in one direction at the peak period. It would conflict with the relevant strategies and actions intended to support safety and mobility goals.	This alternative would accommodate bicyclists in both directions except at the Sonoma Creek Bridge. It would conflict with the relevant plan policies intended to support safety and mobility.	This alternative would provide a continuous shoulder along the corridor in both directions, similar to existing conditions (except at the Sonoma Creek bridge, where the shoulder would be narrowed to 4 feet in each direction, but remain accessible to bicycles). It would not promote or support the plan strategies and actions intended to achieve the goals related to safety and mobility.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Bicycle Plans and Programs	Caltrans Complete Streets Program	This alternative would continue to provide shoulders that are generally accessible by bicycles and would not conflict with the guidance provided in the complete streets programs.	This alternative would continue to provide shoulders that are generally accessible by bicycles and would not conflict with the guidance provided in the complete streets programs.	This alternative would preclude bicycles during the peak hour in the peak direction and would not provide an accessible shoulder over the Sonoma Creek Bridge during peak travel periods. Legislation to prohibit bicycle use along the corridor would be proposed.	This alternative would narrow the existing shoulders to 4 feet wide and eliminate shoulders at the Sonoma Creek Bridge. Legislation to prohibit bicycle use along the corridor would be proposed.	This alternative would continue to provide shoulders that are accessible by bicycles and would not conflict with the guidance provided in the complete streets programs.
Bicycle Plans and Programs	San Francisco Bay Trail	Although this alternative would not conflict with the Bay Trail plan, it would not support implementation of a Bay Trail segment in the study area.	Improved bay shoreline access is being considered as part of the project. Final options would be defined in consultation with BCDC.	Improved bay shoreline access is being considered as part of the project. Final options would be defined in consultation with BCDC.	Improved bay shoreline access is being considered as part of the project. Final options would be defined in consultation with BCDC.	Improved bay shoreline access is being considered as part of the project. Final options would be defined in consultation with BCDC.

Plan	Policy/Goal/Objective/ Action	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
Bicycle Plans and Programs	Solano Countywide Bicycle Transportation Plan and Sonoma County Bicycle and Pedestrian Plan	This alternative would not alter existing accommodations for bicyclists.	This alternative could accommodate bicyclists in the shoulder, consistent with existing conditions.	This alternative would preclude bicycles during the peak hour in the peak direction and would not provide an accessible shoulder over the Sonoma Creek Bridge during peak travel periods. Legislation to prohibit bicycle use along the corridor would be proposed.	This alternative would narrow the existing shoulders to 4 feet wide and eliminate shoulders at the Sonoma Creek Bridge. Legislation to prohibit bicycle use along the corridor would be proposed.	This alternative would continue to provide shoulders that are accessible by bicycles.

Source: County of Solano 2008; City of Vallejo 2017; County of Napa 2008; Sonoma County 2020

# 2.2.3 Coastal Zone

# 2.2.3.1 Regulatory Setting

BCDC, created prior to the California Coastal Act, retains oversight and planning responsibilities for development and conservation of coastal resources in the San Francisco Bay Area. The regulatory authority for BCDC is the McAteer-Petris Act and the Suisun Marsh Protection Act. BCDC uses its federally approved Management Program for the San Francisco Bay Segment of the California Coastal Zone (Management Program) to exercise its federal consistency authority under the Coastal Zone Management Act (CZMA). The BCDC exercises authority under Section 307 CZMA (16 USC section 1456) over federal activities and development projects and nonfederal projects that require a federal permit or license or are supported by federal funding. The consistency provisions of section 307 of the CZMA provide that any federal activity, including a federal development project, that affects any land or water use or natural resource of the BCDC's coastal zone, must be conducted in a manner that is "consistent to the maximum extent practicable" with the enforceable policies of the BCDC's federally approved coastal management program. Similarly, any nonfederal activity that requires either a federal permit or license or is supported by federal financial assistance that affects the BCDC's coastal zone must be conducted in a manner that is fully consistent with the enforceable policies of the BCDC's federally approved coastal management program.

The management program also provides that BCDC would generally follow its procedures for processing a permit application when it reviews a consistency determination for a federal project or activity, or a consistency certification for a nonfederal project subject to consistency review. Since the Bay is subject to BCDC jurisdiction a BCDC permit would be required. BCDC is also the federal delegate for the CZMA and would issue that consistency determination as part of their permit. The Bay is outside of the California Coastal Commission jurisdictional area and no Coastal Development Permit is required.

# 2.2.3.2 Affected Environment

## McAteer Petris Act Jurisdictional Areas

Caltrans has completed a preliminary delineation of BCDC's jurisdictional areas as defined in the McAteer Petris Act Section 66610. In accordance with these definitions, Caltrans understands that the project alternatives would intersect with the BCDC bay, shoreline band, salt ponds, managed wetlands, and certain waterways (i.e., Tolay Creek and Sonoma Creek). The definitions of BCDC's jurisdictions are available at https://www.bcdc.ca.gov/plans/mcateer\_petris.html#ch2 and are summarized here.

BCDC Jurisdictional Areas include:

• **Bay Jurisdiction:** San Francisco Bay, being all areas that are subject to tidal action from the southern end of the Bay to the Golden Gate (Point Bonita-Point Lobos) and to the Sacramento River line (a line between Stake Point and

Simmons Point, extended northeasterly to the mouth of Marshall Cut), including all sloughs and, specifically, the marshlands lying between mean high tide and 5 feet above mean sea level; tidelands (land lying between mean high tide and mean low tide); and submerged lands (land lying below mean low tide).

- Shoreline Band Jurisdiction: A shoreline band consisting of all territory between the shoreline of San Francisco Bay and a line 100 feet landward of and parallel with that line, but excluding any portions of such territory which are within Bay, salt pond, managed wetland, or certain waterway jurisdictions; provided that the commission may, by resolution, exclude from its area of jurisdiction any area within the shoreline band that it finds and declares is of no regional importance to the Bay.
- **Saltpond Jurisdiction:** Saltponds consisting of all areas which have been diked off from the Bay and have been used for the solar evaporation of Bay water in the course of salt production.
- **Managed Wetland Jurisdiction:** Managed wetlands consisting of all areas which have been diked off from the Bay as a duck hunting preserve, game refuge or for agriculture.
- **Certain Waterway Jurisdiction:** Certain waterways, consisting of all areas that are subject to tidal action, including submerged lands, tidelands, and marshlands up to 5 feet above mean sea level, on, or tributary to, the listed portions of the following waterways in the project area:
  - Tolay Creek in Sonoma County, to the northerly line of Sears Point Road (SR 37)
  - o Napa River, to the northernmost point of Bull Island
  - Sonoma Creek, to its confluence with Second Napa Slough

McAteer Petris Act Section 66605 requires that any new fill in BCDC's Bay Jurisdiction be limited to "water-oriented uses" such as ports, water-related industry, airports, bridges, wildlife refuges, water-oriented recreation and public assembly, water intake and discharge lines for desalinization plants and power generating plants requiring large amounts of water for cooling purposes, or minor fill for improving shoreline appearance or public access to the Bay. Fill in the Bay jurisdiction would require that the project demonstrate that the public benefit is greater than the impact from the fill; that the fill is a water-oriented use or a small amount of fill to improve shoreline appearance; that there is no possible alternative to filling the Bay; that the fill is the minimum amount necessary, is located to minimize impacts to the greatest extent possible, is constructed to safety standards for the structure, and would establish a permanent shoreline; and that the applicant has valid title to the property.

Development in the shoreline band does not require that fill meet the definition of "water-oriented use" and may be permitted by BCDC. McAteer Petris Act Section 66632.4 requires that any development project in the shoreline band provide "maximum feasible public access consistent with the proposed project" to obtain a BCDC permit and requires BCDC to consult with CDFW to determine whether proposed public access is compatible with wildlife protection in the Bay.

McAteer Petris Act Section 66654 allows for the continuation of existing use in the shoreline band, salt ponds, and managed wetland jurisdictions, provided that no substantial change in use is made. Although the existing SR 37 and SR 121 roadways are existing uses where they intersect those jurisdictions, work would occur in and beyond the existing developed roadway, and would require a BCDC permit. A permit from BCDC would be obtained prior to construction.

## BCDC Bay Plan, Bay Plan Maps, and Policies

Section 66603 requires and Chapter 5 of the McAteer Petris Act (Sections 66651 through 66663.1) defines BCDC's establishment and procedures for an enforceable planning document which defines land use priorities in San Francisco Bay. The Bay Plan is a comprehensive and enforceable plan for the conservation of the water of the Bay and the development of its shoreline. BCDC developed Bay Plan policies and maps for the Bay. SR 37 is defined as a Scenic Drive and adjacent to areas of Priority Use as Wildlife Refuge, Tidal Marsh, and Salt Pond or Managed Wetlands in Bay Plan Map 1 – San Pablo Bay and Bay Plan Map 2 – Carquinez Strait. Policies found in Bay Plan Map 1 and Map 2 are summarized in Table 2-3.

Bay Plan Map Number	Bay Plan Map Policy Number	Policy Statement
1	12	Route 37 – Evaluate design options if and when travel demand warrants. Provide public access in a manner protective of sensitive wildlife. Provide opportunities for wildlife compatible activities, such as wildlife observation and fishing.
2	2	Route 37 – Evaluate design options if and when travel demand warrants. Provide public access in a manner protective of sensitive wildlife. Provide opportunities for wildlife compatible activities, such as wildlife observation and fishing.

Table 2-3 BCDC Bay Plan Maps 1 and 2 Policies

The Bay Plan developed specific policies for actions that would occur in its Bay and Shoreline jurisdictions that are applicable to the project. These policies would be considered in the final design and permitting phase as part of the project's BCDC permit application. BCDC provides Bay Plan Policies on several topics for development within its jurisdiction. Recently, BCDC updated its Bay Plan to include specific climate change policies, with consideration of SLR incorporated into other existing Bay Plan Policies. SLR for the proposed project is discussed in detail under Section 3.4.5. Specific BCDC Bay Plan policies would be analyzed for the selected alternative during the project's permitting phase. This document focuses on the Transportation Policies in the Bay Plan (Table 2-4).

Policy Number	Policy	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
1	Because of the continuing vulnerability of the Bay to filling for transportation projects, BCDC should continue to take an active role in Bay Area regional transportation and related land use planning affecting the Bay, particularly to encourage alternative methods of transportation and land use planning efforts that support transit and that do not require fill. MTC, Caltrans, the California Transportation Commission, FHWA, county congestion management agencies, and other public and private transportation authorities should avoid planning or funding roads that would require fill in the Bay and certain waterways.	SR 37 was constructed on fill prior to the existence of BCDC. The No Build alternative would maintain SR 37 consistent with standard operations and maintenance procedures and is not anticipated to have fill impacts.	Partially Consistent. Caltrans began coordinating with BCDC on this project during early scoping and planning. Alternative 1 would include the unavoidable placement of fill material to widen the roadway in areas that are anticipated to be partially or entirely within BCDC Bay and shoreline band. This would include 0.03 acre in tidal waters and 1.52 acres in tidal wetlands. Site constraints provide no upland alternative to the relatively minor amount of proposed fill.	Partially Consistent. Caltrans began coordinating with BCDC on this project during early scoping and planning. Alternative 2 would include unavoidable filling to widen the roadway in areas that are anticipated to be partially or entirely within BCDC Bay and shoreline band. This would include 0.16 acre in tidal waters and 3.2 acres in tidal wetlands. Site constraints provide no upland alternative to the relatively minor amount of proposed fill.	Partially Consistent. Caltrans began coordinating with BCDC on this project during early scoping and planning. Alternative 3A would be very similar to Alternative 2 and would include unavoidable filling to widen the roadway in areas that are anticipated to be partially or entirely within BCDC Bay and shoreline band. This would include 0.24 acre in tidal waters and 3.75 acres in tidal wetlands. Site constraints provide no upland alternative to the relatively minor amount of proposed fill.	Partially Consistent. Caltrans began coordinating with BCDC on this project during early scoping and planning. Alternative 3B would include unavoidable filling to widen the roadway in areas that are anticipated to be partially or entirely within BCDC Bay and shoreline band. This would include 1.23 acres in tidal waters and 7.33 acres in tidal wetlands. Site constraints are the same as all alternatives. The increase in permanent fill in tidal waters is primarily from bridge widening at Sonoma Creek, which is consistent with Bay Plan Policies for fill.

#### Table 2-4 Consistency with BCDC Bay Plan Transportation Policies by Alternative

Policy Number	Policy	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
2	If any additional bridge is proposed across the Bay, adequate research and testing should determine whether a feasible alternative route, transportation mode, or operational improvement could overcome the particular congestion problem without placing an additional route in the Bay and, if not, whether a tunnel beneath the Bay is a feasible alternative.	Not Applicable. Existing bridges would be maintained as built.	Not applicable. Existing bridges and route would be maintained as built. Tolay Creek Bridge would be widened at its current location.	Not applicable. Existing bridges and route would be maintained as built. Tolay Creek Bridge would be widened at its current location.	Not applicable. Existing bridges and route would be widened. No new bridges proposed. Tolay Creek Bridge would be widened at its current location.	Not applicable. Existing bridges would be widened. No new bridges proposed. Tolay Creek and Sonoma Creek Bridges would be widened at their current locations.
3a	If a route must be located across the Bay or a certain waterway, the crossing should be placed on a bridge or in a tunnel, not on solid fill.	Not Applicable. The roadway was constructed prior to BCDC existence.	Not applicable. Existing bridges would be maintained as built. Tolay Creek Bridge would be widened at its current location	Not applicable. Existing bridges would be maintained as built. Tolay Creek Bridge would be widened at its current location	Not applicable. Existing bridges would be widened. No new bridges proposed. Tolay Creek Bridge would be widened at its current location	Not applicable. Existing bridges at Tolay Creek and Sonoma Creek would be widened. No new bridges proposed.
3b	If a route must be located across the Bay or a certain waterway, bridges should provide adequate clearance for vessels that normally navigate the waterway beneath the bridge.	Not applicable. Existing bridges would be maintained as built.	Not applicable. Existing bridges would be maintained as built. Tolay Creek Bridge would be widened at its current location	Not applicable. Existing bridges would be maintained as built. Tolay Creek Bridge would be widened at its current location	Not applicable. Existing bridges would be widened. Tolay Creek Bridge would be widened at its current location	Sonoma Creek is a navigable channel. Temporary construction at Sonoma Creek would be necessary, but the navigable channel would be maintained.

Policy Number	Policy	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
3с	If a route must be located across the Bay or a certain waterway, toll plazas, service yards, or similar facilities should not be constructed on new fill and should be far enough from the Bay shoreline to provide adequate space for maximum feasible public access along the shoreline.	Not applicable. No toll plaza would be placed.	Partially Consistent. Tolling is proposed as a potential project element that would include fully automated tolling facilities, if implemented, within the SR 37 corridor. Alternative 1 would require new maintenance worker facilities and storage areas along SR 37 for a movable barrier.	Consistent. Tolling is proposed as a potential project element that would include fully automated tolling facilities, if implemented, within the SR 37 corridor.	Consistent. Tolling is proposed as a potential project element that would include fully automated tolling facilities, if implemented, within the SR 37 corridor.	Consistent. Tolling is proposed as a project element that would include fully automated tolling facilities within the SR 37 corridor.
3d	If a route must be located across the Bay or a certain waterway to reduce the need for future Bay crossings, any new Bay crossing should be designed to move the largest number of travelers possible by employing technology and operations that increase the efficiency and capacity of the infrastructure, accommodating nonmotorized transportation and, where feasible, providing public transit facilities.	Not applicable. No new Bay crossing is proposed.	Not applicable. No new Bay crossing is proposed.	Not applicable. No new Bay crossing is proposed.	Not applicable. No new Bay crossing is proposed.	Not applicable. No new Bay crossing is proposed.

Policy Number	Policy	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
4	Transportation projects on the Bay shoreline and bridges over the Bay or certain waterways should include pedestrian and bicycle paths that would either be a part of the Bay Trail or connect the Bay Trail or connect the Bay Trail with other regional and community trails. Transportation projects should be designed to maintain and enhance visual and physical access to the Bay and along the Bay shoreline.	Consistent. Existing bicycle access would be maintained. Bay views would be retained. The existing median barrier is 32 inches high.	Partially consistent. Existing bicycle access would be maintained. Bay views would be similar to existing views, depending on height of movable barrier. New visible maintenance and operations buildings would be necessary at one or both ends of movable barrier.	Partially consistent. Bicycle use along SR 37 shoulders would be restricted by legislation due to narrow shoulders, especially at Sonoma Creek Bridge. Bay views would be reduced by the increased median barrier height. New barriers along the shoulder may be necessary.	Partially consistent. Bicycle use along SR 37 shoulders would be restricted by legislation due to narrow shoulders, especially at Sonoma Creek Bridge. Bay views would be reduced by the increased median barrier height. New barriers along the shoulder would be necessary.	Consistent. Bicycle access would be maintained similar to current conditions. Additionally, public access would be addressed in the final project development. This may include funding to construct a bike path as part of the Bay Trail. Bay views would be reduced by the increased median barrier height. New barriers along the shoulder may be necessary. During consideration and review to identify the preferred alternative, the project determined that a 36-inch-high median barrier (as opposed to the 42-inch-high median barrier considered in the draft environmental document) would be acceptable, and shoulder barriers would primarily use a Midwest Guardrail System along most of the corridor segment. Both revised elements would reduce impacts to Bay views for SR 37 travelers for the identified preferred alternative.

Policy Number	Policy	No Build Alternative	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
5	Ferry terminals should be sited at locations that are near navigable channels, would not rapidly fill with sediment, and would not significantly impact tidal marshes, tidal flats, or other valuable wildlife habitat. Wherever possible, terminals should be near higher density, mixed-use development served by public transit. Terminal parking facilities should be set back from the shoreline to allow for public access and enjoyment of the Bay.	Not applicable.	Not applicable.	Not applicable.	Not applicable.	Not applicable.

# 2.2.3.3 Environmental Consequences

All build alternatives would cause temporary and permanent impacts to areas within BCDC's jurisdictions. Temporary impacts for all Build Alternatives include vegetation clearing, temporary construction access, installation of piles, staging and laydown areas, and construction noise and air quality emissions. All alternatives would involve work at Tolay Creek to widen the bridge, including work in the creek channel and banks. Alternative 3B would involve additional impacts at Sonoma Creek for the widening of the bridge, staging areas for equipment and access to the bridge, and potential use of barges in the creek for construction.

Permanent impacts for all Build Alternatives include placement of sheet piles adjacent to the roadway, roadway expansion, in-water and upland fill (pile) and shading from the Tolay Creek Bridge expansion (suspended fill), and existing roadway reconfigurations. Alternative 3B would require additional width of the bridge over Sonoma Creek (shading). Current estimates for project impacts in BCDC jurisdiction are preliminary and would be refined after identification of the preferred alternative and in the final design. Temporary and permanent impacts in BCDC jurisdictions are anticipated in each project alternative. Final areas would be determined during the project's final design and permitting phase. A summary of the project's consistency with BCDC's Bay Plan Policies on Transportation is shown in Table 2-4.

To meet BCDC's regulatory mandate to provide maximum feasible access consistent with the proposed project for development projects in their shoreline band jurisdiction, Caltrans and MTC would develop a public access proposal that is reasonable, feasible, and appropriate for the selected alternative as part of the BCDC permitting process to be carried out during the project's final design phase. Meeting this regulatory requirement is part of the BCDC permitting process.

## 2.2.3.4 Avoidance, Minimization, and/or Mitigation Measures

Measures proposed for temporary and permanent impacts to wetlands and other waters as detailed in Section 2.4.2 would also offset any anticipated impacts within BCDC's jurisdiction. Upland development in BCDC's jurisdiction that is outside of the existing roadway would require authorization from BCDC and would be addressed as part of the permitting process.

## 2.2.4 Parks and Recreational Facilities

## 2.2.4.1 Regulatory Setting

The Park Preservation Act (California Public Resources Code [PRC] Sections 5400-5409) prohibits local and state agencies from acquiring any property which is in use as a public park at the time of acquisition unless the acquiring agency pays sufficient compensation or land, or both, to enable the operator of the park to replace the park land and any park facilities on that land.

# 2.2.4.2 Affected Environment

Publicly owned parks and recreation areas within 0.75 mile of the project area are listed in Table 2-5. Table 2-5 includes descriptions of the location of each park or recreation area in relation to the project area.

Jurisdiction	Name	Description
City of Vallejo	River Park	City-owned park with open space and trails.
City of Vallejo	Terrace Park	City-owned park with play area, picnic areas, barbeque pits, and ball field.
Solano County/Napa County/Sonoma County	Refuge	The 19,000-acre refuge overlaps the counties of Solano, Napa, and Sonoma adjacent to SR 37. The refuge in the project area includes pedestrian/bicycle trails, boating, hunting, fishing, and wildlife viewing.
Solano County/Napa County/Sonoma County	NSMWA	The 15,200-acre wildlife area consists of Baylands, tidal sloughs, and wetland habitat and overlaps the counties of Solano, Napa, and Sonoma adjacent to SR 37. The wildlife area includes pedestrian trails, boating, hunting, fishing, and wildlife viewing.
City of Vallejo/Solano County/Napa County/ Sonoma County	San Francisco Bay Trail (existing and planned)	The San Francisco Bay Trail is a planned 500-mile trail for pedestrian and cyclists that would encompass all nine Bay Area counties, 47 cities, and cross seven toll bridges.
Sonoma County	Vista Point (5000 Sears Point Road)	Vista point for wildlife viewing.

 Table 2-5 Publicly Owned Parks and Recreation Facilities

Sources: City of Vallejo 2020; USFWS 2016, 2017b, 2021; CDFW 2020; ABAG and MTC 2020b

## 2.2.4.3 Environmental Consequences

## No Build Alternative

The No Build Alternative would not affect recreation facilities near the project area.

# **Build Alternatives**

Each of the Build Alternatives would require TCEs in the NSMWA and the Refuge. Build Alternatives 3A and 3B would also require permanent use and acquisition of areas in the NSMWA and Refuge. The NSMWA and the Refuge are park and recreational facilities protected by Section 4(f) of the Department of Transportation Act of 1966. This project would result in a "use" of those facilities as defined by Section 4(f). Please see Appendix B, Section 4(f), for additional details.

Build Alternatives 1 and 2 would not require the permanent use of any publicly owned park or recreational facility because they would be constructed largely within Caltrans' right-of-way. All build alternatives would require TCEs (all area quantities provided are approximate). Build Alternative 1 would require a TCE of 0.12 acre within the 15,200-acre NSMWA. Build Alternative 2 would require TCEs of approximately 0.16 acre within the NSMWA, and 0.44 acre within the 19,000-acre Refuge. Build

Alternative 3A would require TCEs of 0.15 acre within the NSMWA, 0.03 acre in the Refuge, in addition to permanent use of approximately 1.65 acres of the Refuge. Build Alternative 3B would require TCEs of 0.29 acre, permanent acquisition of 1.43 acres within the NSMWA, and permanent use of 2.76 acres of the Refuge.

The temporary use of the properties would be needed for sufficient space to accommodate construction activities. Given the limited area that would be occupied and the short-term duration, Build Alternatives 1 and 2 would not preclude or substantially impede the use of any parks or recreation facilities during construction. Construction activities would not require closure or substantial alteration of the recreational facilities listed above in Table 2-5. Furthermore, Build Alternatives 1 and 2 would not have long-term effects to any publicly owned parks or recreation facilities.

Permanent use of areas in the Refuge would be required for Build Alternatives 3A and 3B to facilitate widening of the existing roadway. The proposed sliver acquisitions along SR 37 would not affect recreational access to the Refuge or NSMWA, and would affect areas only alongside SR 37. Therefore, implementation of Build Alternatives 3A and 3B would not represent a major adverse change to the NSMWA or the Refuge.

Construction activities would take place primarily within Caltrans' right-of-way and would not require closure or substantial alteration of the recreational facilities listed in Table 2-5. However, Alternative 3B could require a partial closure of a parking lot during construction. The parking lot provides access to recreational activities. Because portions of the parking lot would remain open during construction, recreational activities would not be impacted. Construction impacts would be short term, and the minimal area of TCEs in adjacent recreational facilities would be used for a limited period of one or two construction seasons. The Build Alternatives would not have long-term effects to any publicly owned parks or recreation facilities.

Appendix B provides the Section 4(f) analysis. Caltrans has determined that no change would occur in the values, accessibility, or attributes of Section 4(f) uses as a result of the project. USFWS and CDFW have concurred with this determination, and their concurrence letters are included in the Final EIR/EA, Volume 2, Appendix B.

# 2.2.4.4 Avoidance, Minimization, and/or Mitigation Measures

Property acquisition at the CDFW and Refuge lands would require compensation. This would be identified in consultation with the these agencies, and be either monetary compensation or provision of new land area that could be incorporated into the CDFW wildlife area and USFWS Refuge; this would be worked out during the design phase.

# 2.2.5 Farmlands

# 2.2.5.1 Regulatory Setting

NEPA and the Farmland Protection Policy Act (FPPA, 7 USC 4201-4209; and its regulations, 7 CFR Part 658) require federal agencies, such as the FHWA, to coordinate with the Natural Resources Conservation Service if their activities may

irreversibly convert farmland (directly or indirectly) to nonagricultural use. For purposes of the FPPA, farmland includes prime farmland, unique farmland, and land of statewide or local importance.

CEQA requires the review of projects that would convert Williamson Act contract land to nonagricultural uses. The main purposes of the Williamson Act are to preserve agricultural land and to encourage open space preservation and efficient urban growth. The Williamson Act provides incentives to landowners through reduced property taxes to discourage the early conversion of agricultural and open space lands to other uses.

# 2.2.5.2 Affected Environment

The information provided in this section is from the Community Impact Assessment prepared in June 2021 (AECOM 2021f). Farmland is classified and mapped by the California Department of Conservation, Division of Land Resource Protection for the purposes of tracking farmland development throughout the state. The Farmland Mapping and Monitoring Program classifies farmland according to five types:

- Prime Farmland: Farmland with the best combination of physical and chemical features able to sustain long-term agricultural production
- Farmland of Statewide Importance: Farmland similar to Prime Farmland but with minor shortcomings, such as greater slopes or less ability to store soil moisture
- Unique Farmland: Farmland of lesser quality soils, used for the production of the State's leading agricultural crops
- Farmland of Local Importance: Land of importance to the local agricultural economy, as determined by each county's board of supervisors and a local advisory committee
- Grazing Land: Land on which the existing vegetation is suited to the grazing of livestock (CDC 2010)

The California Land Conservation Act of 1965, which is commonly referred to as the Williamson Act, enables local governments to enter into contracts with private landowners for the purpose of restricting specific parcels of land to agricultural or related open space uses. In return, the landowners receive property tax assessments that are much lower than would otherwise be the case because the taxes are based on property value assessments that assume farming and open space uses in contrast to potential market rate development.

The project is adjacent to Farmland of Local Importance and Grazing Land as well as Non-Prime Agricultural parcels with Williamson Act contracts located between the SR 37/SR 121 interchange and the Sonoma Bridge in Sonoma County.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### 2.2.5.3 Environmental Consequences

The project would not convert Prime Farmland, Farmland of Statewide Importance, Unique Farmland, or Farmland of Local Importance away from farmland use. The project would require TCEs; however, the easements would not affect the continued use of the properties for agricultural use. None of the Build Alternatives are anticipated to require permanent property acquisitions. Additionally, there would be no permanent acquisition of Williamson Act properties. The project would not modify, nullify, or require changes to the Williamson Act contracts on the properties. Therefore, there would be no impact.

#### 2.2.5.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required.

#### 2.2.6 Growth

#### 2.2.6.1 Regulatory Setting

The Council on Environmental Quality regulations, which established the steps necessary to comply with NEPA, require evaluation of the potential environmental effects of all proposed federal activities and programs. This provision includes a requirement to examine indirect effects, which may occur in areas beyond the immediate influence of a proposed action and at some time in the future. The Council on Environmental Quality regulations (40 CFR 1508.8) refer to these consequences as indirect impacts. Indirect impacts may include changes in land use, economic vitality, and population density, which are all elements of growth.

CEQA also requires the analysis of a project's potential to induce growth. The CEQA Guidelines (Section 15126.2[d]) require that environmental documents "...discuss the ways in which the proposed project could foster economic or population growth, or the construction of additional housing, either directly or indirectly, in the surrounding environment..."

#### 2.2.6.2 Affected Environment

The following section is based on the *Community Impact Assessment* prepared for the project (AECOM 2021f).

The project area is in North Bay region of the Bay Area on SR 37 along the northern edge of San Pablo Bay. The project extends from the western landing of SR 37 at the Napa River Bridge across the Napa River, in Solano County, to 1.15 miles west of the SR 37/SR 121 Interchange in Sears Point, Sonoma County. The project area is in an unincorporated community but connects various moderately growing communities in Sonoma, Napa, and Solano Counties. Table 2-6 displays the increase in population in the Bay Area from 2015 to 2050, according to Plan Bay Area 2050 population projections.

Jurisdiction	2015 Population*	2030 Population	2040 Population	2050 Population	Percent increase 2015-2050
Bay Area (nine counties)	7,660,000	8,560,000	9,490,000	10,330,000	+34.9%

Table 2-6 Population Change in the Project Area

Notes: \*A tabulation from the 2010 pre-run microdata, designed to approximate (but may still differ from) Census 2010 counts.

Source: Plan Bay Area 2050 - Projections 2050

The projections indicate that recent population growth in the Bay Area is expected to continue. It is projected that there would be a 34.9 percent increase in population in 2050 from 2015.

Based on MTC's regional travel demand model, traffic growth is estimated at 0.8 percent per year through 2040 and is expected to result in increased peak period congestion and longer travel times. This regional travel demand model indicates that traffic along the SR 37 corridor would continue to get more congested as the region's population increases. The average annual daily trips on SR 37 at the Sonoma/Solano County border was 35,000 in 2017. According to the SCTA, trips are projected to increase up to 58,000 by 2040 (Caltrans 2017b).

This project is designed to address existing traffic demand along this corridor. The current traffic congestion on SR 37 between the Napa River Bridge and the SR 37/SR 121 Interchange in Sears Point is primarily due to the reduction from two lanes to one lane in each direction, causing a bottleneck situation that backs up traffic on either side. The segment of SR 37 from U.S. 101 to the SR 37/SR 121 interchange was upgraded to a four-lane express way in 1957 to support the rising traffic demand on SR 37 due to the Mare Island Naval Base (Tom 2020).

The segment of SR 37 extending east from the Napa River Bridge was upgraded to a four-lane expressway when the state rebuilt the bridge in 1967. In 2004 the two-lane section of SR 37 between Napa River Bridge and the SR 37/SR 121 Interchange in Sears Point had a cross-over barrier installed to prevent the increasingly high rate of fatal collisions along the two-lane corridor (which had led to it being coined the "Blood Alley of San Pablo Bay") (Tom 2020). The upgrades that were historically made on SR 37 to provide more capacity on SR 37, coupled with the continued increase in population in the north bay counties, ultimately created an eastbound and westbound bottleneck of traffic congestion on SR 37.

As previously discussed, the project is designed to accommodate current traffic demands to reduce traffic congestion along this section of the SR 37 corridor. The proposed alternatives each include an HOV/express lane to facilitate a shift from SOVs, thereby reducing traffic demand and decreasing corridor congestion while

increasing person throughput. Although this design would help alleviate traffic along this corridor, it would not eliminate the traffic congestion completely.

## 2.2.6.3 Environmental Consequences

## No Build Alternative

The No Build Alternative would not increase capacity for the project and would not foster or accommodate economic or population growth in the project area.

# **Build Alternatives**

The Build Alternatives would increase SR 37 capacity by incorporating a movable barrier (Alternative 1) to provide an additional lane, converting the existing outside shoulder lane during peak hours (Alternative 2), or by converting the outside shoulder lanes to provide a four-lane facility (Alternatives 3A and 3B). This additional capacity is anticipated to alleviate congestion and improve travel time reliability along SR 37. The project would not provide new access to previously undeveloped land. Although the project would accommodate planned growth, it would not result in reasonably foreseeable changes to planned land uses both adjacent to and in the vicinity of the project study area.

The project would increase roadway capacity in the study area. Specifically, the capacity increase on SR 37 would result in increased throughput of vehicles and persons. Although the project may facilitate transportation access through the corridor, any such development would have to occur in accordance with those uses designated in the applicable general plan and subject to the zoning of the affected jurisdiction. The preferred Alternative 3B would convert the existing lanes to HOV lanes, and the added lanes would be for general purpose use. HOV lanes would be operational 24 hours per day, and encourage carpools and bus use. In addition, much of the SR 37 corridor is protected as wildlife refuge lands and unavailable to local development. Thus, the project would help alleviate traffic congestion and would not induce unplanned growth. If planned growth increases in existing communities served by SR 37, the precise location and type of such growth is not reasonably foreseeable; therefore, further analysis of the effects associated with an increased rate of growth is considered speculative.

# 2.2.6.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required.

# 2.2.7 Community Character and Cohesion

# 2.2.7.1 Regulatory Setting

NEPA, as amended, established that the federal government must use all practicable means to ensure for all Americans safe, healthful, productive, and aesthetically and culturally pleasing surroundings (42 USC 4331[b][2]). FHWA, in its implementation of

NEPA (23 USC 109[h]), directs that final decisions on projects are to be made in the best overall public interest. This requires taking into account adverse environmental impacts, such as destruction or disruption of human-made resources, community cohesion, and the availability of public facilities and services.

Under CEQA, an economic or social change by itself is not to be considered a significant effect on the environment. However, if a social or economic change is related to a physical change, then social or economic change may be considered in determining whether the physical change is significant. Because this project would result in physical change to the environment, it is appropriate to consider changes to community character and cohesion in assessing the significance of the project's effects.

## 2.2.7.2 Affected Environment

The following section is based on the *Community Impact Assessment* prepared for the project (AECOM 2021f).

## Population

In recent years, the populations of Solano and Napa Counties both grew. Solano County saw a 0.259 percent increase between 2017 and 2018, and Napa County saw a 0.129 percent increase between 2016 and 2017. In contrast, Sonoma County declined from 504,217 to 499,942 between 2017 and 2018, a -0.848 percent decrease (Data USA 2020).

The study area is predominately White and Hispanic or Latino. The population of Solano County is 37.3 percent White, 26.9 percent Hispanic or Latino, and 15.2 percent Asian. The population of Sonoma County is 62.8 percent White, 27.2 percent Hispanic or Latino, and 4.17 percent Asian. Similarly, Napa County's population is 53.2 percent White, 33.7 percent Hispanic or Latino, and 7.76 percent Asian (Data USA 2020).

The median household income in Solano County is \$84,395, in Sonoma County is \$81,395, and in Napa County is \$79,637. Solano County's poverty rate is 11.5 percent, Sonoma County's is 10.7 percent, and Napa County's is 8.21 percent (Data USA 2020).

## Neighborhoods/Communities/Community Character

Residential land uses in the project area are described in Section 2.2.1; as noted in that section, no existing or planned residential uses are adjacent to or near SR 37. Parks and recreational facilities are described in Section 2.2.4. Businesses in the project area and study area are discussed below. There are no activity centers such as childcare centers, banks, churches, or grocery stores in the immediate project area.

There are several local roads with direct access to SR 37 that lead to rural residences and businesses. These roads include Tolay Creek Road, Noble Road, and Skaggs Island Road.

## Housing

In 2018, Solano County had 62.9 percent homeowners and a median property value of \$442,700; Sonoma had 61.5 percent homeowners and a median property value of \$655,200; and Napa had 63.1 percent homeowners and a median property value of \$603,700. Each county has rate of homeowners similar to the national average of 63.9 percent (Data USA 2020). According to the California Association of Realtors, the median price of a home in the Bay Area is around \$900,000, nearly four times the national average.

In the nine-county Bay Area, the largest job clusters are in Santa Clara County (916,000), Alameda County (700,000), and San Francisco (591,000). Most residents work in their county of residence. However, employees in Solano County have a longer commute time (33.3 minutes) than the average U.S. worker (25.7 minutes). Additionally, 9.16 percent of the workforce in Solano County experience "super commutes" greater than 90 minutes (Data USA 2020). On the other hand, Sonoma County employees have a shorter commute time (23.5 minutes) compared to the average U.S. worker, and only 3.84 percent of the Sonoma County workforce experience "super commutes" (Data USA 2020). Napa County residents have the shortest commutes in the study area, with an average of 22.8 minutes.

# **Regional/Local Economy**

Solano County employs 216,000 people, Sonoma County employs 260,000 people, and Napa County employs 71,100. The largest industries in Solano County are health care, and social assistance (34,197 people), Retail Trade (24,584 people), and Construction (20,092 people). The highest-paying industries are Utilities (\$101,222), Mining, Quarrying, and Oil and Gas Extraction (\$91,495), and Public Administration (\$74,894). The Solano County Chamber of Commerce indicates that the largest employer in the county is Travis Airbase, followed by Vallejo Kaiser Permanente, and then Solano County.

The largest industries in Sonoma County are Health Care and Social Assistance (34,113 people), Retail Trade (30,715 people), and Construction (24,425 people). The highest-paying industries are Utilities (\$100,179), Public Administration (\$80,762), and Information (\$75,782).

The largest industries in Napa County are Health Care and Social Assistance (9,719 people), Manufacturing (9,196 people), and Accommodation and Food Services (7,699 people). The highest-paying industries are Public Administration (\$75,877), Mining, Quarrying, and Oil and Gas Extraction (\$73,464), and Utilities (\$71,838) (Data USA 2020).

Businesses directly adjacent to the project area include Speedway Sonoma LLC (commonly known as Sonoma Raceway), Paradise Vineyards, Gold Coast Vineyards, and The Wing and Barrel Ranch/Kenwood-DPSC Hunt Club. Paradise Vineyards and Gold Coast Vineyards have access to SR 37 by Tolay Creek Road/Sears Point Road at the traffic light intersection of SR 37 and SR 121.

The Wing and Barrel Ranch/Kenwood-DPSC Hunt Club is accessed by Noble Road, which has a turning lane in the eastbound direction of SR 37 but does not currently have a traffic signal. Sonoma Raceway is one of the largest businesses in the vicinity of the project, though it does not have direct access from SR 37. The main entrance to Sonoma Raceway is accessed by SR 121, approximately 0.5 mile north from the SR 37/SR 121 interchange; there is also a private entrance off Lakeville Highway, which is opened to the public for large events. The raceway houses a motorsports industrial park of more than 70 businesses in 104 shops (Speedway Motorsports, LLC 2020). Sonoma Raceway hosts many events throughout the year, including the annual NASCAR race, which holds a multi-day event with thousands of visitors. These large annual racing events pose traffic challenges along SR 37 and SR 121, which bring approximately 100,000 people and 5,000 campers to the annual NASCAR event (Sonoma Index Tribune 2019).

# 2.2.7.3 Environmental Consequences

## No Build Alternative

The No Build Alternative would not change access to the existing community and would not affect neighborhoods or the local economy. It would have no effect on existing community cohesion.

# **Build Alternatives**

## Population

As described in Section 2.2.6, the increased capacity on SR 37 proposed by the Build Alternatives is not expected to encourage more people or employers to move to Sonoma, Solano, or Napa Counties. The project would primarily benefit the current population of commuters in the study area and at the regional level.

## Neighborhoods/Communities/Community Character

The project would not change any existing community boundaries or physically divide an established community. The project is not anticipated to influence growth patterns for the reasons described in Section 2.2.6, and as a result would not change the existing character of the communities in the study area.

Local residents and businesses could experience temporary access impacts from the construction closures. Property access would be maintained throughout project construction. No full closures of SR 37 or SR 121 are anticipated. Implementation of a TMP would minimize the potential for short-term construction impacts.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

# Housing

The project would not displace or relocate any residents or encourage more people to move to unincorporated Sonoma, Solano, or Napa Counties, including the surrounding areas. The proposed project would not create additional land availability or affect housing stock in the study area or at a regional level. The additional capacity provided by the project would shift some traffic to the SR 37 corridor, but is not expected to increase growth, because SR 37 is an existing highway within land uses largely protected from growth. Although potential changes resulting from increased housing demand and associated population increase may occur in the North Bay Area, the relatively incremental change to traffic patterns that would result from the proposed project in the context of larger Bay Area is not expected to be substantial.

# Regional/Local Economy

The project would not directly affect the employment rates in the study area, nor would it impede the accessibility to the adjacent businesses, as described in Section 2.2.7.2. Project operations may increase the accessibility of adjacent and nearby businesses by reducing congestion, but—as noted previously—no unplanned development is anticipated that would substantially alter employment in the study area.

The project would add a traffic signal at Noble Road and SR 37, which would only activate when a vehicle approaches the SR 37 Noble Road intersection. This would have no effect on accessibility to the business.

Project construction would have to minimize or avoid lane reductions or closures that would overlap with events at local land uses, primarily the Sonoma Raceway. Project construction would be subject to the TMP described under Section 1.4.1.3. Given the limited duration of project construction activities that would affect the Sonoma Raceway and the implementation of the TMP, construction activities would not substantially affect access or operation of the raceway.

The project would generally improve access to adjacent and nearby land uses by reducing congestion. Therefore, the project is not expected to adversely affect the values of properties along the frontage roads.

# 2.2.7.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation measures are required.

# 2.2.8 Relocations and Real Property Acquisition

# 2.2.8.1 Regulatory Setting

The Caltrans Relocation Assistance Program (RAP) is based on the Federal Uniform Relocation Assistance and Real Property Acquisition Policies Act of 1970, as amended, and 49 CFR 24. The purpose of the RAP is to ensure that persons displaced as a result of a transportation project are treated fairly, consistently, and

equitably so that such persons would not suffer disproportionate injuries as a result of projects designed for the benefit of the public as a whole.

All relocation services and benefits are administered without regard to race, color, national origin, persons with disabilities, religion, age, or sex. Please see Appendix C for a copy of the Caltrans Title VI Policy Statement.

# 2.2.8.2 Affected Environment

The project is primarily within Caltrans' right-of-way along SR 37; this corridor crosses through the City of Vallejo and the counties of Solano, Napa, and Sonoma. The Refuge and NSMWA are adjacent to a portion of the route.

# 2.2.8.3 Environmental Consequences

Most of the project would be constructed within Caltrans' right-of-way, and no relocations of homes or businesses are necessary. However, to provide construction access—in addition to widening required for Build Alternatives 3A and 3B; and work at the SR 121 intersection, Tolay Creek Bridge, Noble Road, the Cullinan Ranch public access intersection, and other private access driveways—the project would result in partial property acquisitions and TCEs on several properties. Table 2-7 below identifies the potentially proposed TCEs. Permanent partial property acquisition would occur at the Refuge, as listed in Table 2-8. The actual impacts to properties would be determined during detailed project design.

The TCEs required for the temporary construction activities for the Build Alternatives would not result in substantial changes to existing land uses on any of the affected parcels. Only a limited work area would be required for a limited period time (one construction season). Following construction, the affected TCE parcels would be restored to pre-project conditions. Construction would have no effect related to the zoning and land use designations of the TCE parcels. Because the affected parcels would be result. The project would not require any full property acquisitions and would not relocate any residences or businesses. The permanent partial property acquisitions would not affect the continued function of the Refuge. The project would not result in the conversion of any parcels to a new land use or otherwise interfere with the continued use of parcels for their existing purpose.

Property access would be maintained throughout project construction, and no full closures of SR 37 or SR 121 are anticipated. Property owners whose access may be temporarily affected by project construction would be notified in advance.

# 2.2.8.4 Avoidance, Minimization, and/or Mitigation Measures

No relocations are required. Temporary easements and partial property acquisitions would be compensated, which would be determined during the right-of-way phase of the project.

Assessor's Parcel Number (APN)	Street Address	Owner	Alternative 1 TCE (Acres)	Alternative 2 TCE (Acres)	Alternative 3A TCE (Acres)	Alternative 3B TCE (Acres)	Existing Parcel Use: Project Impact
068-170-002	29730 Tolay Creek Road Sonoma, CA 95476	Private	_	0.003	0.003	0.003	Residence: Temporary construction – construct motor vehicle pullout for sign structure
068-190-H2O	Tolay Creek	Creek	0.005	0.005	0.005	0.005	Miscellaneous: Temporary access road from staging area
068-190-015	6600 Noble Road Sonoma, CA 95476	Wing and Barrel Ranch	0.09	0.09	0.09	0.09	Miscellaneous: Temporary access road from staging area
068-190-017	Unincorporated Sonoma County	Wing and Barrel Ranch	2.30	2.30	2.30	2.30	Field crops: Temporary construction staging area and access road
068-190-008	5400 Sears Point Road Sonoma, CA 95476	Black Point Game Bird Club	0.06	0.06	0.06	0.05	Field crops: Intersection reconstruction at Noble Road
068-190-032	None	Sonoma Marin Area Rail Transit	0.15	0.15	0.15	0.15	Railroad grade crossing and railroad right-of-way: Modify grade crossing panels to accommodate roadway widening. Relocate railroad signal and cabinets.
067-02-0010	None	State of California – CDFW	0.006	_	0.07	0.18	Miscellaneous: Temporary construction – viewpoint intersection modifications/driveway
068-180-020	5000 Sears Point Road, Unincorporated Sonoma County	Vallejo Sanitation and Flood Control District	_		_	_	Miscellaneous: Reconstruct Noble Road driveway

 Table 2-7 Temporary Construction Easements

Assessor's Parcel Number (APN)	Street Address	Owner	Alternative 1 TCE (Acres)	Alternative 2 TCE (Acres)	Alternative 3A TCE (Acres)	Alternative 3B TCE (Acres)	Existing Parcel Use: Project Impact
067-02-0100	None	State of California – CDFW	0.11	0.10	0.09	0.10	NSMWA: Temporary construction – Cullinan Ranch Road intersection modifications
067-02-0110	None	State of California	_	0.004	_	0.009	Miscellaneous: Temporary construction
067-02-0140	None	Solano County		_	0.002	0.012	Miscellaneous: Temporary construction
067-03-0060	None	State of California	_	_	0.005	_	Miscellaneous: Temporary construction
067-04-0050	None	State of California			0.020		Miscellaneous: Temporary construction
067-04-0120	None	USFWS		0.438	_		Miscellaneous: Temporary construction
Totals (Acres)	—	—	2.712	3.220	2.795	2.899	_

APN	Street Address	Owner	Alternative 1 (Acres)	Alternative 2 (Acres)	Alternative 3A (Acres)	Alternative 3B (Acres)	Existing Parcel Use: Project Impact
068-160-006	2100 Highway 37, Sonoma, CA 95476	US Government	_	_	0.141	0.141	USFWS Managed – Agricultural/Dairy Part of San Pablo Bay National Wildlife Refuge Partial Take for Roadway Fill
067-02-0010	None	State of California – CDFW	_	—	_	0.1236	Napa-Sonoma Marshes Wildlife Area Partial take for roadway fill
067-02-0110	None	State of California - CDFW	_		_	0.016	Napa-Sonoma Marshes Wildlife Area Partial take for roadway fill
067-03-0110	None	State of California – CDFW	_	—	_	0.006	State Lands Commission - CDFW Managed Partial Take for roadway fill
067-03-0060	None	State of California – CDFW	_	_	_	0.72	State Lands Commission – CDFW Managed Partial take for roadway fill
067-04-0050	None	State of California - CDFW	_	_	_	0.56	State Lands Commission – CDFW Managed Partial take for roadway fill
067-04-0120	None	USFWS	_	_	1.65	2.62	Refuge: Partial take for roadway fill
Totals	—	—	0	0	1.791	4.187	—

**Table 2-8 Permanent Partial Acquisition** 

# 2.2.9 Environmental Justice

# 2.2.9.1 Regulatory Setting

All projects involving a federal action (funding, permit, or land) must comply with EO 12898, Federal Actions to Address Environmental Justice in Minority Populations and Low-Income Populations, signed by President William J. Clinton on February 11, 1994. This EO directs federal agencies to take the appropriate and necessary steps to identify and address disproportionately high and adverse effects of federal projects on the health or environment of minority and low-income populations to the greatest extent practicable and permitted by law. Low income is defined based on the Department of Health and Human Services poverty guidelines. For 2017, this was \$24,600 for a family of four.

All considerations under Title VI of the Civil Rights Act of 1964, and related statutes, have also been included in this project. The Department's commitment to upholding the mandates of Title VI is demonstrated by its Title VI Policy Statement, signed by the Director, which can be found in Appendix C of this document.

## 2.2.9.2 Affected Environment

The environmental justice analysis for this project included review of the Census Block Groups that border the project area. Block groups are divisions of census tracts that are delineated by local or regional organizations and usually consist of a cluster of several blocks. For the environmental justice analysis completed for this project, the study area block groups were compared to the county overall. Data for the analysis were derived from the United States Census Bureau's 2013-2017 American Community Survey 5-Year Estimates (United States Census Bureau 2017).

Caltrans identifies a community as an environmental justice community of concern if it meets one or both of the following criteria:

- The minority population exceeds 50 percent or is meaningfully greater (e.g., more than 10 percentage points) than the minority population percentage in the general population or other appropriate unit of geographic analysis (e.g., the counties overlapping the study area).
- The low-income population comprises more than 25 percent of the census block group or tract.

The environmental justice analysis includes the census tracts 1506.12, 1501, 2518.02, 2508.01, 2518.03, 2517.01, 2517.02, and 2011.02 (tracts from east of SR 29 and Broadway in Vallejo, to the Sonoma/Napa County line west of Lakeville Highway). All tracts are immediately adjacent to the project area. The census tracts are compared to Sonoma County, Solano County, and Napa County, which serve as the reference area for the project. California data are also provided for context. There are eight census block groups that border the project area. Table 2-9 describes the minority and low-income populations in the selected tracts in each county.

## 2.2.9.3 Environmental Consequences

The No Build Alternative would not change operations on SR 37 nor have the potential to disproportionally effect a minority or low-income community.

The Build Alternatives would add HOV lanes to improve traffic conditions along SR 37 and encourage carpooling. As shown in Table 2-9, four of the census tracts adjacent to the project area meet the criteria of an environmental justice community due to a minority population exceeding a meaningfully greater proportion than the minority population percentage in Solano County. None of the census tracts adjacent to the project contain more than 25 percent low-income populations.

Table 2-9	Summary of Race,	Ethnicity,	and Poverty	Status	in the S	Study	Area a	and
	-	Refere	nce Areas			-		

Geography	Black	Native American	Asian	Native Hawaiian or Other Pacific Islander	Minority*	Hispanic	Below Poverty Level
California	5.8%	0.7%	14.1%	0.4%	21%	38.8%	15.1%
Sonoma County	1.6%	1.1%	3.9%	0.3%	6.9%	26.4%	10.7%
Census Tract 1506.12	0.4%	0.2%	6.2%	0.0%	6.26%	10.1%	4%
Census Tract 1501	0.0%	0.3%	0.0%	0.0%	0.3%	12.7%	7.2%
Solano County	14.2%	0.5%	15.3%	0.9%	30.9%	25.8%	11.5%
Census Tract 2518.02	23.6%	0.0%	23.4%	0.08%	47.08%	34.6%	4.3%
Census Tract 2508.01	28.9%	0.3%	13.4%	4.2%	46.8%	18.8%	19.4%
Census Tract 2518.03	17.5%	1.1%	23.8%	0.2%	42.6%	42.3%	5%
Census Tract 2517.01	25.9%	0.1%	11.3%	0.2%	37.5%	21.7%	21.6%
Census Tract 2517.02	31.2%	0.0%	28.5%	4.7%	64.4%	13.4%	9.6%
Napa County	2.1%	0.9%	7.9%	0.2%	11.1%	33.7%	8.2%
Census Tract 2011.02	0.0%	0.0%	3.0%	0.0%	3%	8.6%	3.4%

Notes:

\* Minority is the sum of United States Census Bureau reported Black, Native American, Asian, and Native Hawaiian or Other Pacific Islander

Shaded Cells indicate a reference population. Bold cells indicate an Environmental Justice Community. Source: United States Census Bureau 2017

The proposed work in these areas would be limited to pavement widening and resurfacing for each Build Alternative to accommodate an additional HOV lane, installation of electronic tolling equipment, replacement of the concrete median barrier,

and the addition of signs and lighting. During project construction, air quality and noise impacts would occur, but these impacts would be short in duration. These temporary impacts are not expected to be substantial and would only occur during a short duration. Impacts from construction would not be more impactful to environmental justice communities than non-environmental-justice communities.

During operation of the project, traffic conditions are expected to improve, resulting in reduced air pollution. This would be a benefit to environmental justice communities.

## Toll Lanes

Caltrans is proposing the installation and operation of tolling for the SR 37 general purpose lane. The HOV lane would not be tolled in either direction. It is recognized that tolls paid on a frequent basis could represent a greater economic burden to low-income travelers than to middle- and high-income travelers. Although none of the Census Tracts adjacent to the project contain more than 25 percent low-income populations, there still are low-income individuals represented in these census tracts who could be impacted. Tolling is described in the project description, Section 1.4.1.3, including that a means-based toll discount would be implemented. The means-based discount would be available based on income, through a program that would prequalify the motorists that would be eligible for the discount. The program would be developed as part of the tolling plan for the corridor. The project would also promote and encourage ridesharing, which helps alleviate congestion and maximize the people-carrying capacity of a highway. Any user of the highway in a multiple-occupant vehicle would gain a time savings and toll-free advantage in the HOV lane, and this advantage is available to all multi-occupant drivers and passengers.

Based on the above discussion and analysis, the Build Alternatives would not cause disproportionately high and adverse effects on any minority or low-income populations, in accordance with the provisions of EO 12898.

# 2.2.9.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, and mitigation measures (AMMs) are required.

# 2.2.10 Utilities/Emergency Services

The following section is based on the *Community Impact Assessment* prepared for the project (AECOM 2021f). There are power, gas, telecommunication (fiber optic), and water utilities in the study area. PG&E provides gas and electricity service, and American Telephone & Telegraph Company (AT&T) provides telecommunication service. Sonoma Water, Sonoma Valley County Sanitation District, Solano County Water Agency, and Vallejo Flood and Wastewater District manage water utilities within the project limits.

Police protection and traffic enforcement services in the study area are provided by the Sonoma County Sheriff's Department, Solano County Sheriff's Department, and the City of Vallejo Police Department. The CHP has jurisdiction over SR 37 for matters involving traffic violations and emergency services. Fire protection services in the study area are provided by Vallejo Fire Department, Sonoma County Fire Department, and Sonoma Valley Fire Department.

## 2.2.10.1 Environmental Consequences

#### No Build Alternative

Because the No Build Alternative would not result in changes to SR 37, it would not require utility relocations or construction activities that could interfere with the provision of emergency services.

## **Build Alternatives**

The Build Alternatives would require relocation of some PG&E overhead electrical distribution lines. Some wooden poles would be relocated due to the construction and widening of the roadway in certain sections.

The relocation of electrical facilities may result in temporary interruptions of service. Final verifications of utilities would be performed during the project's detailed design phase, and any needed relocations would be coordinated with the affected utility owner. No disruption to electrical power is anticipated. No impacts to water service are anticipated.

Law enforcement, fire, and emergency services would be maintained during project construction. During final design, a TMP would be developed for the project to minimize construction-related delays and inconvenience to project area residents and the traveling public. The TMP would include notification to emergency service providers and the public of lane closures and detours; coordination with CHP and local law enforcement on contingency plans; and using portable Changeable Message Signs, CHP Construction Zone Enhanced Enforcement Program, and Freeway Service Patrol where possible to minimize delays. The TMP would be implemented to ensure that emergency services would not be affect during the construction of the project. Therefore, the project is not expected to result in decreased response times.

For these reasons, the Build Alternatives would not result in long-term effects on utilities or emergency services.

# 2.2.10.2 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required.

# 2.2.11 Traffic and Transportation/Pedestrian and Bicycle Facilities

# 2.2.11.1 Regulatory Setting

The Department, as assigned by the FHWA, directs that full consideration should be given to the safe accommodation of pedestrians and bicyclists during the development of federal-aid highway projects (see 23 CFR 652). It further directs that the special needs of the elderly and the disabled must be considered in all federal-aid projects that include pedestrian facilities. When current or anticipated pedestrian and/or bicycle

traffic presents a potential conflict with motor vehicle traffic, every effort must be made to minimize the detrimental effects on all highway users who share the facility.

In July 1999, USDOT issued an Accessibility Policy Statement pledging a fully accessible multimodal transportation system. Accessibility in federally assisted programs is governed by the USDOT regulations (49 CFR 27) implementing Section 504 of the Rehabilitation Act (29 USC 794). The FHWA has enacted regulations for the implementation of the 1990 Americans with Disabilities Act (ADA), including a commitment to build transportation facilities that provide equal access for all persons. These regulations require application of the ADA requirements to federal-aid projects, including Transportation Enhancement Activities.

# 2.2.11.2 Affected Environment

The information in this section is based on the *Traffic Operations Analysis Report* prepared in August 2021 (AECOM 2021a). The operational analysis evaluated existing and future conditions. Existing conditions represent the year 2020. Future conditions are projected for the years 2025 (Opening Year) and 2045 (Design Year). The majority of SR 37 is a four-lane facility with two lanes in the eastbound and westbound directions. The segment of SR 37 that makes up the project limits (between SR 121 and the Mare Island Interchange) narrows to a two-lane facility with one lane in each direction.

Currently, there are significant recurring traffic congestion/delay issues experienced by commuters at the bottlenecks within the traffic study limits during the peak hours, due to traffic demands exceeding capacity in segments between SR 121 and Mare Island, where the existing two lanes merge into one lane in both the eastbound and westbound directions. Traffic conditions are expected to worsen with continual developments in the region and within and adjacent to the SR 37 corridor. Growth in travel demand on SR 37 is expected to result in longer periods of travel times during the AM and PM peak periods.

The traffic study included the segment of SR 37 between SR 121 and Mare Island, and several mainline and intersections extending from the SR 29 interchange in the City of Vallejo to the U.S. 101 interchange in the City of Novato. The traffic study area extended farther than the physical project limits to allow for full analysis of traffic conditions. The study mainline segments and intersections include:

# Mainline Segments

- SR 37 between U.S. 101 Interchange and SR 121 Intersection
- SR 37 between SR 121 Intersection and Mare Island Interchange
- SR 37 between Mare Island Interchange to SR 29 Interchange

## Intersections

- SR 37 and Lakeville Hwy (Signal)
- SR 37 and SR 121 (Signal)
- SR 37 and Noble Road (Two-Way Stop Control)

- SR 37 and Skaggs Island Road (Two-Way Stop Control)
- SR 37 Westbound Ramps and Walnut Avenue/Main Gate (Two-Way Stop Control)
- SR 37 Eastbound Ramps and Walnut Avenue/Main Gate (Two-Way Stop Control)

#### Pedestrian and Bicycle Facilities

There are no pedestrian facilities on SR 37 in the project study area, except along the Napa River bridge and the approaches to that bridge. Bicyclists are permitted on the shoulders of SR 37.

## Traffic Operations Analysis Study Area and Methods

The traffic study area is described in the preceding section. This section describes and compares overall performance of the No Build and Build future conditions, by factors such as delay, total travel time, and speed, and VMT. The traffic study analyzed peak period conditions defined as 5 AM to 11 AM (AM peak) and 2 PM to 9 PM (PM peak). These conditions represent the most congested periods of the day and are used to define the peak hour for purposes of the impact analysis.

The operational analysis for the project was conducted using the VISSIM simulation modeling program. VISSIM is a simulation model capable of analyzing the vehicle-to-vehicle interaction along the highway mainline, HOV lane facilities, ramps, and intersections. Existing conditions AM and PM peak period VISSIM models were developed and calibrated to replicate observed field condition bottlenecks and queues for the hours specified above.

The traffic volumes dataset for the SR 37 corridor was derived from the project-specific data collections, including 2019 SR 37 Highway Vehicle Classification Counts; 2019 SR 37 Highway HOV Occupancy Survey; 2019 SR 37 Highway and Ramp Segment Counts; and 2019 Study Intersection Turning Movement Counts. The 2019 data collected were used to establish the 2020 baseline year.

Level of service (LOS) is used to describe the ability of a roadway to accommodate prevailing traffic volumes at the critical intersections based on the physical characteristics of the roadway. LOS ranges from "A" (representing uncongested free-flow conditions) to "F" (representing total breakdown with stop-and-go operation). Table 2-10 shows the LOS designation and corresponding delay thresholds for both signalized and unsignalized intersections.

VMT is a measure of the number of miles traveled by vehicles in a roadway network. It is the calculation of every trip taken or estimated, multiplied by the length of the trip. An increase or decrease in VMT indicates more or fewer trips, a longer or shorter average length of trips, or a combination of both a change in trips and a change in the average distance traveled. VMT is generally evaluated by comparing alternatives to indicate the overall difference between the alternatives.
LOS	Signalized Intersection Delay Thresholds	Unsignalized Intersections Delay Thresholds	Description
А	delay ≤ 10.0	delay ≤ 10.0	Little or no traffic delay
В	10.0 < delay ≤ 20.0	10.0 < delay ≤ 15.0	Minimal traffic delay
С	20.0 < delay ≤ 35.0	15.0 < delay ≤ 25.0	Average traffic delay
D	35.0 < delay ≤ 55.0	25.0 < delay ≤ 35.0	Long traffic delay
E	55.0 < delay ≤ 80.0	35.0 < delay ≤ 50.0	Very long traffic delay
F	delay > 80.0	delay > 50.0	Extreme traffic delay

Table 2-10	Intersection LOS	Thresholds

# Method Used to Estimate VMT

The Caltrans Transportation Analysis Framework (TAF) (Caltrans 2020c) provides guidance on the methods for estimating VMT, including induced VMT associated with a lane addition. Two methods are available for the VMT analysis. The first is the National Center for Sustainable Transportation (NCST) Induced Travel Calculator (NCST calculator) (NCST 2019); second is the use of a regional travel demand model (TDM), which is the MTC model given the multi-county location of this project. Induced demand refers to a change in travel behavior, such as longer trips that avoid slower areas of congestion, new trips because a travel mode offers a lower cost, quicker, or more convenient route by car, or land use changes that might attract trips that are longer in distance. Because the NCST calculator is not an application for the evaluation of a tolled facility, the MTC model was selected for the VMT analysis.

The TAF includes a series of steps for evaluation of the VMT induced demand analysis. This includes a checklist that was completed regarding the sensitivity of the model to land use inputs, the sensitivity of trip making behavior on travel times and travel costs, the sufficiency of the model network detail and study area, model network assignment, and model calibration and validation. Potential deficiencies of the TDM model were identified for specific varied land uses and level of detail of the roadway network, and therefore a calibration or benchmark comparison was made between the MTC travel model and the NCST VMT calculator method, consistent with the TAF guidance.

The NCST calculator was used as an off-model tool to inform and benchmark the results and any necessary adjustments to the MTC model. The following overall model comparison and evaluation was used to determine the VMT analysis for the project and location:

- Apply the NCST calculator for a travel lane addition and identify induced VMT.
- Add the proposed HOV lane(s) to the MTC model and run the MTC model to determine induced VMT.
- Compare VMT results between the NCST calculator and the MTC travel model.

- If the MTC model results are within 20 percent of the value provided by the NCST Calculator, the MTC model results are used. The 20 percent threshold was established by Caltrans' TAF (Caltrans 2020c).
- If the MTC model results differ from that of the NCST calculator by more than 20 percent, then the NCST calculator results are used. In some cases the MTC model results may be used with specific quantitative evidence explaining this variation provided.

# **Benchmark/Model Validation Process**

For the travel lane addition in the same location as the project (regardless of HOV or general-purpose lanes), the NCST calculator estimated an induced annual VMT of approximately 40.1 million, based on approximately 5.4 directional lane miles in Sonoma County and 13.4 directional lane miles in Solano County. These lane miles represent the addition of a single lane in each direction, consistent with the project description and the proposed new lane miles in those two counties. The project is in Sonoma and Solano Counties, in Metropolitan Statistical Areas (MSAs) with a Class 2 facility (as defined by the TAF method for projects on the State Highway System). The TAF specifies that the analysis for a project involving a lane addition should apply a travel demand model or county and or TDM estimate benchmarked against the NCST Calculator, which was the method followed. For the purposes of the benchmark/ validation process, the MTC model projected an induced annual VMT with proposed HOV lanes in each direction of approximately 17.8 million. This is with the addition of the HOV lanes only, without the implementation of tolling for general purpose lanes. Tolling reduces VMT in the travel demand model, and therefore the initial analysis of VMT without tolling is considered a worst-case or maximum estimate of induced demand with the proposed lane additions.

Because the proposed project is an HOV lane addition, use of the lane is limited only to eligible carpool vehicles during the peak commute periods, with vehicular volumes less than those a general-purpose lane. Existing HOV use accounts for 21 percent of all vehicles based on existing field-collected data. This is projected to increase to approximately 25 percent with the proposed project, because the HOV lane would incentivize more travelers to carpool rather than driving alone. The existing SR 37 route at the project location has only one general-purpose lane in each direction, with an estimated hourly total vehicular traffic demand in the order of 2,000 vph. The HOV volumes are estimated to be in the order of 500 vph during the peak hours; this is significantly below the capacity of the existing general-purpose lane, which is approximately 1,650 vph (according to the MTC model for a conventional highway traffic lane).

The MTC model's projected VMT increase with an HOV lane in each direction is 44 percent compared to the NCST calculator's estimation of a travel lane addition. Although this exceeds the 20 percent difference tolerance, it is expected that the induced travel from the proposed HOV lane in each direction on SR 37 would be significantly lower than the induced travel from a general-purpose lane addition. This is because the project encourages a mode shift away from driving alone to carpooling.

Moreover, the HOV lanes would be restricted to eligible carpool vehicles at all times. The MTC model's projected VMT increase due to the proposed HOV lane demonstrated that the model is more sensitive and conservative (higher VMT) in estimating potential induced travel than the NCST Calculator.

To further assess the MTC model's ability to estimate induced travel demand, an additional scenario run was performed by adding a general-purpose lane in the same location as the project. The model then projected an annual VMT increase of approximately 48.7 million. This is a more conservative (higher VMT) result than the NCST calculator's projected VMT increase of 40.1 million.

For the reasons listed above, the MTC model was deemed appropriate and conservative (higher estimated VMT) as the selected tool to perform the VMT analysis for assessment of impacts for the proposed project alternatives.

# **Existing Conditions**

Based on field observations, westbound is the peak direction during the AM peak period. The westbound SR 37 bottleneck starts at the Mare Island lane drop and the queue extends to Wilson Avenue interchange during the AM peak hours. The maximum flow from this bottleneck is approximately 1,250 vph. The maximum travel time between SR 29 and southbound U.S. 101 is 50 minutes, as observed from 6 to 7 AM; and the minimum travel time is 25 minutes, as observed during the 10 to 11 AM hour.

During the AM peak period, the intersection of the SR 37 westbound ramps/Walnut Avenue intersection operates at LOS E/F during 5 to 8 AM hours. The intersection of SR 37 eastbound ramps/Walnut Avenue intersection operates at LOS E/F between 6 and 8 AM. The congestion forms at these intersections because demand exceeds capacity for SR 37.

During the PM peak period, eastbound is the peak direction. The eastbound bottleneck starts at the lane drop east of the SR 121 Intersection, and the queue extends to Railroad Avenue. The maximum flow from this bottleneck is approximately 1,250 vph. The maximum travel time between northbound U.S. 101 and SR 29 is 68 minutes, as observed during the 4 to 5 PM hour; and the minimum travel time is 22 minutes, as observed during the 8 to 9 PM hour.

In the PM peak period, the intersection of SR 37/SR 121 operates at LOS F during from 2 to 8 PM, and the intersection of SR 37/Lakeville Highway operates at LOS E/F from 3 to 7 PM (evening commute). Eastbound traffic typically becomes congested from the bottleneck east of the SR 121 Intersection to the Lakeville Highway intersection during evening commute hours.

There are several study intersections currently operating at LOS (E/F) conditions in one or more hours during the AM and PM peak periods. Unsignalized intersections at Noble Road and Skaggs Island Road experience delays in general as the drivers exiting from these side streets have difficulties finding gaps in the heavily used SR 37.

# HOV Volumes

HOV volumes were extracted from the 2019 vehicle occupancy data collected manually at Noble Road for both eastbound and westbound SR 37 for the project limits.

### **Truck Volumes**

In the project limits during the AM peak period, truck volumes account for 12 percent of the of vehicle composition in the eastbound direction and 9 percent in the westbound direction. During the PM peak period, truck volumes account for 5 percent of the vehicle composition in both eastbound and westbound directions.

### 2.2.11.3 Environmental Consequences

This section summarizes the traffic conditions for the opening year 2025 and design year 2045 for the No Build and Build Alternatives, as reported in the December 2021 Traffic Operations Analysis Report (TOAR). LOS were modeled for key intersections and are included in Appendix H. Travel times (the modeled average time to travel a segment of the corridor) were also predicted and are discussed in the following sections. The SR 37/Noble Road intersection was evaluated and reported in the TOAR without a signal, but a project decision was made to include a signal at this intersection. With the signal, operations at SR 37/Noble Road would be expected to function as or better than as described in this section and the TOAR for all alternatives. Overall, the project's Build Alternatives would provide for additional traffic capacity during the peak periods and, in many cases, improve travel time.

# Intersection LOS 2025

Intersection LOS results were obtained from the VISSIM operational analysis and are summarized in intersection LOS tables in Appendix H.

Similar to the existing conditions, under 2025 No Build Conditions, the results indicate that out of six study intersections, three in the AM (SR 37/Noble Road, SR 37 westbound ramps/Walnut Avenue, and SR 37 eastbound ramps/Walnut Avenue) would operate at LOS E or F for one or more peak period hours. The alternatives would have the following results:

 Alternatives 1 and 2: In the AM peak period, the SR 37/Noble Road would generally improve from LOS E (No Build) and operate with Alternatives 1 and 2 at no worse than LOS D. But the intersections at SR 37/Walnut Avenue in the westbound and eastbound directions would continue to operate at LOS F and delay times would generally worsen, beginning as early as 5 AM. During the 5 AM peak hour, the delay increases from 57 seconds delay to 150 seconds; and during the 6 AM peak hour, the delay increases from 89 to 101 seconds.

In the PM peak period, with the No Build Alternative, the SR 37 intersections at Lakeville Highway, SR 121, Noble Road, and Skaggs Island Road operate at LOS E or F, with substantial delays of up to 3 to 6 minutes. These intersections

would still operate at LOS E and F with Alternatives 1 and 2 but would have comparatively reduced delay times of less than 1 minute to 5 minutes.

 Alternatives 3A and 3B: In the AM peak period, these alternatives would improve intersection operations compared to the No Build Alternative at the SR 37/Walnut Avenue intersection, and would be similar to the No Build except at one location/one peak hour: SR 37/Noble Road at the 8:00 AM peak hour (Scenario 1) would decrease from LOS E to LOS F, and delay would increase from 43 to 57 seconds. All other intersections would function at LOS D or better. The Alternative 3A/3B variations S1, S2, and S3 at the SR 37/SR 121 intersection would all function at LOS D or better (except for SR 37/Noble Road at 8 AM, Scenario 1, as noted above), and the differences in delay would be similar (the differences in delay would be within a range of 0 to 3 seconds).

In the PM peak period, Alternatives 3A and 3B would improve most study intersections to LOS D or better. The SR 37/Noble Road intersection would function at conditions of LOS E and F during the afternoon hours of 2 to 5 PM, but the delay would be less than the No Build Alternative. The SR 37/SR 121 intersection would have substantial improvement compared to the No Build (and Alternatives 1 and 2).

# Intersection LOS 2045

Under 2045 No Build conditions, the results indicate that out of six study intersections, three intersections in the AM period and five intersections in the PM period operate at LOS E or F in one or more hours during the AM and PM peak period. The intersections affected with the No Build and Build Alternatives are similar to the 2025 year, but the delays are longer due to growth in traffic. Delays are most severe for the No Build Alternative, and Alternatives 1 and 2.

- Alternatives 1 and 2: In the AM peak period at the SR 37 ramps/Walnut Avenue intersection, delays increase compared to the No Build Alternative, especially at the 5 AM peak period hour. The remaining intersections would operate at LOS A to C, except for the 8 AM hour at SR 37/Noble Road, which would function at LOS E (although this would be an improvement over the No Build, which would function during the 8 AM hour at LOS F).
- In the PM period, delays with the No Build Alternative increase substantially at the Lakeville Highway, SR 121, Noble Road, Skaggs Island Road, and the westbound SR 37/Walnut Avenue ramps. At the Noble Road intersection, delays are as high as 53 minutes at the 6 PM peak hour and range from 3 to 5 minutes delay at the Lakeville Highway, SR 121, Skaggs Island Road, and Walnut Avenue intersections. With Alternatives 1 and 2, these delays decrease at these same intersections but still remain at 1 to 10 minutes of delay.
- Alternatives 3A and 3B: In the AM peak period, intersection operations would function at LOS A to C at the Lakeville Highway, SR 121, and Skaggs Island intersections, similar to the No Build and Alternatives 1 and 2. The Noble Road

and Walnut Avenue intersections would still operate at LOS E and F, but delays would be reduced.

• In the PM peak period, LOS would improve at most of the intersections, especially at Lakeville Highway, Noble Road, Skaggs Island Road, and the westbound ramps at Walnut Avenue.

# 2025 Traffic Operations Analysis

# Eastbound PM Peak Period

The following paragraphs compare 2025 traffic conditions of the No Build Alternative and Build Alternatives for SOVs and HOVs during the PM peak period in the eastbound direction. All times provided below are approximate.

- Under the No Build conditions, the maximum travel time is 100 minutes for SOV and HOVs because there is no HOV lane provision.
- Under Alternatives 1 and 2, the maximum travel time is 53 minutes for SOVs and 46 minutes for HOVs. The travel time savings for SOV and HOV is 47 minutes and 54 minutes, respectively, compared to the No Build Alternative.
- Under Alternatives 3A and 3B, the maximum travel time is 26 minutes for SOVs and 23 minutes for HOVs. The travel time savings for SOVs and HOVs is 74 minutes and 77 minutes, respectively, compared to the No Build Alternative.

# Westbound AM Peak Period

The following paragraphs compare 2025 traffic conditions of the No Build Alternative and Build Alternatives for SOVs and HOVs during the AM peak period in the westbound direction. All times provided below are approximate.

- Under the No Build conditions, the maximum travel time is 65 minutes for SOVs and HOVs because there is no HOV lane provision.
- Under Alternatives 1 and 2 conditions, the maximum travel time is 40 minutes for SOVs and 37 minutes for HOVs. The travel time savings for SOVs and HOVs is 25 minutes and 28 minutes compared to the No Build Alternative.
- Under Alternatives 3A and 3b conditions, the maximum travel time is 40 minutes for SOVs and 37 minutes for HOVs. The travel time savings for SOVs and HOVs is 25 minutes and 28 minutes, respectively, compared to the No Build Alternative.

# 2045 Traffic Operations Analysis

The following paragraphs compare 2045 traffic conditions of the No Build Alternative and Build Alternatives for SOVs and HOVs during the PM and AM peak periods.

### Eastbound PM Peak Period

- Under the No Build conditions, the eastbound queues extend 11 miles beyond the U.S. 101 interchange (outside of the project study limits), compared to up to Harbor Drive in the 2025 No Build Conditions.
- Under Alternatives 1 and 2, PM peak-period queues would extend up to 7.5 miles beyond the US 101 interchange due to the HOV operational hours, congestion at the SR 121 intersection, and queuing associated with the general-purpose lanes.
- Under Alternatives 3A and 3B, queues extend close to Railroad Avenue or Harbor Drive due to the proposed lane geometry conditions at the SR 37/ SR 121 intersection.

### Westbound AM Peak Period

- Under the No Build conditions, the westbound queues extend up to 19 miles beyond the SR 29 interchange (outside of the project study limits), compared to up to the SR 29 interchange in 2025 No Build conditions during the AM peak period.
- Under Alternatives 1 and 2, queues extend up to 9.5 miles beyond the SR 29 Interchange Conditions due to limited HOV operational hours.
- Under Alternatives 3A and 3B, queues extend up to just beyond the SR 29 Interchange, similar to existing conditions.

Along the SR 37 corridor, the project would remove an existing constraint in the westbound and eastbound directions. In the westbound direction during AM peak period, because the Build Alternatives improve the throughput traveling toward U.S. 101, there would be additional traffic reaching the interchange of SR 37 with the ramps at U.S. 101 in Marin County. In addition, the westbound traffic would disperse onto other roadway links, including SR 121 and Lakeville Highway. The freeway ramps at the SR 37/U.S. 101 interchange constrain traffic getting onto U.S. 101; therefore, freeway operations on U.S. 101 are not anticipated to worsen during the AM peak period. In the eastbound direction, additional traffic able to flow eastbound toward I-80 and SR 29. The currently constrained traffic is not expected to substantially affect these downstream roadways.

# Total Vehicle Miles Traveled, Vehicle Hours Traveled, and Vehicle Hours of Delay

As discussed in the Section 2.2.11.2, the MTC model was used to evaluate the differences in measures of travel and delay for the No Build Alternative and Build Alternatives for the years 2020 (representing existing), 2025, and 2045. Daily VMT indicates the total miles of all vehicle trips, measured by the distance traveled. VHD indicates the total hours that vehicles are delayed as a result of congestion. Vehicle hours traveled (VHT) is the total hours traveled for all vehicle trips. These measures

for 2020, 2025, and 2045 are shown and compared in Table 2-11, Table 2-12, and Table 2-13, respectively.

The VMT was modeled for all nine counties of the Bay Area, which is why the daily VMT quantities listed in Table 2-11 and Table 2-12 are relatively large. Modeling the VMT for all Bay Area counties captured the regional changes between alternatives that might occur when trips change or divert because of changes in congestion. To evaluate the influence of tolling on traffic demand, VMT, VHD, and VHT were forecast for "no tolling" in 2025 and 2045 (Table 2-12 and Table 2-13) and with tolling (Table 2-14; tolling is proposed for the preferred Alternative 3B).

 Table 2-11
 VMT, VHD, and VHT for 2020 (Existing Conditions)

Description	No Build	
Daily VMT	149,948,925	
Daily VHD	5,523,543	
Daily VHT	8,783,953	

Table 2-12	VMT, VHD, and VHT Estimates of No Build and Build Alternatives
	for 2025 (no tolling)

Description	No Build	Alternatives 1/2	Alternatives 3a/3b	
Daily VMT	156,255,326	156,261,672	156,264,925	
VMT Difference from No Build	Not Applicable 6,346		9,599	
Daily VHD	6,501,187	6,499,000	6,498,000	
VHD Difference from No Build	Not Applicable	-2,187	-3,187	
Daily VHT	9,893,225	9,892,534	9,892,538	
VHT Difference from No Build	Not Applicable	-691	-687	

Description	No Build	Alternatives 1/2	Alternatives 3a/3b	
Daily VMT	181,480,934	181,512,664	181,528,926	
VMT Difference from No Build	Not Applicable	31,729	47,992	
Daily VHD	10,411,762	10,400,827	10,395,826	
VHD Difference from No Build	Not Applicable	-10,935	-15,936	
Daily VHT	14,330,313	14,326,857	14,326,880	
VHT Difference from No Build	Not Applicable	-3,456	-3,443	

Table 2-13	VMT, VHD, and VHT Estimates of No Build and Build Alternatives
	for 2045 (no tolling)

Without tolling, daily VMT in 2025 and 2045 is projected to increase for all Build Alternatives compared to the No Build conditions. Based on regional model results showing an average trip length of 53.4 miles, the increase in trips is estimated at 180 vehicle trips per day. In 2045, the increase in trips would be 899 vehicles per day. As described earlier, this increase includes all estimated new trips, including induced trip generation. Although VMT would increase, VHD and VHT would reduce for all Build Alternatives compared to the No Build Alternative.

# Application of Tolling

For this project, tolling is proposed for the general purpose lanes. To incentivize the use of carpooling, HOVs would not be tolled. That is, tolling would apply to the new general purpose lanes, but the HOV lane in both directions would not be tolled. Tolling scenarios were evaluated using the MTC travel model to estimate VMT changes with the project. The toll rate is assumed to be consistent with the other state-owned tolled bridges in the region, such as the Richmond-San Rafael Bridge. An income means-based discount would also be available for eligible motorists, which is described in Section 3.3.17.1.

For evaluation of traffic and VMT, tolling was modeled for either one direction only or both directions for the study years 2025 and 2045. In 2025, the preferred alternative with tolling would result in reduced daily VMT by 16,668 for westbound-only tolls, and by 11,166 when both directions are tolled. In 2045, the preferred alternative with tolling in the westbound direction would result in a net daily VMT reduction of approximately 83,340 from No Build conditions, and a daily reduction of 55,831 with two-way tolling, compared to No Build conditions. The results are shown in Table 2-14, below.

Description	No Build	Alternative 3B (one-way tolling)	Alternative 3B (two-way tolling)
2025 Daily VMT	156,255,326	156,238,658	156,244,160
Difference from No Build (2025)	Not Applicable	(-16,668)	(-11,166)
2045 Daily VMT	181,480,934	181,397,594	181,425,103
Difference from No Build (2045)	Not Applicable	(-83,340)	(-55,831)

Table 2-14 VMT Estimates of No Build and Build Alternatives with Tollin
---

# **Bicycle Access**

Bicyclists are allowed on the outside shoulders for the No Build Alternative, Alternative 1, and Alternative 3B along SR 37. Bicyclists would be prohibited on the outside shoulders for Alternatives 2 and 3A due to reduced shoulders, especially at Sonoma Creek Bridge. As discussed in Section 1.4, existing shoulders are 8.75 feet wide. Under the Build Alternatives, shoulders widths may be reduced depending on the chosen alternative. Under Alternative 1, shoulders would be 8 feet wide along the project alignment, except for an at-grade crossing at Noble Road, where shoulders would be 4 to 6 feet wide; and on Sonoma Creek Bridge, where shoulders would be 4 feet wide. Alternative 2 would have 4-foot shoulders within the project limits. Alternative 3A would have 4-foot shoulders within the project limits, except at Sonoma Creek Bridge, where there would be no shoulders. Therefore, for Alternative 3A, bicyclists would not have use of the shoulders in this segment (Sonoma Creek Bridge). Alternative 3B would provide for 8-foot-wide shoulders within the project limits, except at the Sonoma Creek Bridge, where the shoulders would be 4 feet wide. Therefore, due to varying widths of each project alternative, access for bicyclists would either be maintained (Alternatives 1 and 3B) or prohibited (Alternative 2 and 3A).

# **Effect Summary**

The project is expected to improve traffic conditions along SR 37 by adding HOV lanes in both directions, which would improve the person-carrying capacity of the corridor, as well as improve the traffic flow and travel times in the peak direction. The project also supports reducing VMT by encouraging carpools and fewer vehicle trips.

Alternatives 1 and 2 would add one HOV lane in the peak direction during the peak periods. Alternatives 3A and 3B would widen SR 37 within the project limits to four full-time lanes, two lanes in each direction. Alternatives 1 and 2 would improve traffic conditions compared to the No Build conditions but would not eliminate the entire congestion, due to limited hours of operations for HOV lane; during non HOV hours, there would be only one lane, similar to the existing situation. Alternatives 3A and 3B would eliminate the lane reductions in each direction at all times, and shows benefits for the general-purpose lane as well as the HOV lane in each direction. The queues—slowdowns caused when an approaching car is within one car length of a stopped vehicle—are improved in the eastbound direction compared to the No Build conditions.

2-54

Additionally, intersection traffic would be improved under all the Build Alternatives during most peak period hours.

Compared to the No Build, VMT is expected to decrease with the combined HOV lane and toll lane (Table 2-14). VHD and VHT would also decrease with the preferred alternative compared to the No Build Alternative.

Related to bicycle access, the Build Alternatives have varying widths of shoulders, and would not restrict or prevent bicycle access except for Alternatives 2 and 3A at Sonoma Creek Bridge, where shoulders would be eliminated.

# **Construction Impacts**

Although SR 37 traffic would be maintained during construction, temporary traffic impacts may occur during construction. Traffic speeds would be reduced for vehicles approaching construction on the shoulder. Construction staging areas would be needed along or near the route for equipment and materials. A TMP would be prepared during the design phase of the project. A TMP would be incorporated as part of standardized project measures to address traffic disruptions from project construction during structural replacements and widening work. Access would be maintained for all emergency response vehicles. Periodic delays may occur if lane closures are necessary, but these would be minimized and planned during nonpeak periods. A temporary shuttle service for bicyclists would be evaluated during the design phase and development of construction staging because the highway shoulders may not be available at times. Effects on traffic during project construction would be temporary, and traffic access along SR 37 would be maintained during peak travel periods.

# 2.2.11.4 Avoidance, Minimization, and/or Mitigation Measures

With the HOV lane and new tolled general purpose lane in each direction, VMT would decrease. The proposed improvements add sufficient capacity to improve traffic conditions along SR 37. Neither the U.S. EPA nor FHWA require VMT analysis. FHWA emphasizes concepts of sustainability in highway planning, project development, design, operations, and maintenance. Because there have been requirements set forth in California legislation and EOs on VMT reduction, the issue is addressed in the CEQA transportation chapter of this document (Section 3.3.17). The CEQA analysis may be used to inform the NEPA determination for the project.

# 2.2.12 Visual/Aesthetics

# 2.2.12.1 Regulatory Setting

NEPA, as amended, establishes that the federal government use all practicable means to ensure all Americans safe, healthful, productive, and aesthetically (emphasis added) and culturally pleasing surroundings (42 USC 4331[b][2]). To further emphasize this point, FHWA, in its implementation of NEPA (23 USC 109[h]), directs that final decisions on projects are to be made in the best overall public interest taking

into account adverse environmental impacts, including among others, the destruction or disruption of aesthetic values.

CEQA establishes that it is the policy of the state to take all action necessary to provide the people of the state "with...enjoyment of aesthetic, natural, scenic and historic environmental qualities" (California PRC Section 21001[b]).

California Streets and Highways Code Section 92.3 directs Caltrans to use drought resistant landscaping and recycled water when feasible and incorporate native wildflowers and native and climate-appropriate vegetation into the planting design when appropriate.

# 2.2.12.2 Affected Environment

The following discussion is based on the *Visual Impact Assessment* completed for this project in July 2021 (AECOM 2021h).

The project corridor is a two-lane highway along the northern shoreline of San Pablo Bay, from the northern end of Mare Island in the east to Sears Point in the west. The project corridor crosses low-lying coastal marshland and is at points no more than 2 feet above typical daily tide levels.

The eastern and central part of the project corridor (from the eastern terminus to Sonoma Creek) passes primarily through publicly owned marshland (the Refuge and NSMWA). This area is a patchwork of meandering sloughs, open water, mud flat, tidal marsh, estuary, and seasonal wetlands. The wildlife refuges contain a recreational area at Cullinan Ranch and some trails, as discussed below. The western end of the project corridor, from Sonoma Creek to the western terminus, is mostly privately owned land that has been converted to agriculture or managed for waterfowl hunting and fishing. Oat hay farms and vineyards are the primary agricultural use. The area west of Tolay Creek rises up toward Sears Point, the southernmost peak of the Sonoma Mountains.

The eastern and central parts of the project corridor are undeveloped. The area west of Sonoma Creek contains a little development associated with agriculture. Sonoma Raceway, a professional racetrack with elevated stands, is just upslope from the western end of the project corridor on Sears Point. Other development includes Wing and Barrel Ranch, a membership-based hunting club accessible from the project corridor that offers hunting and fishing grounds. Thirty-Seven Winery is 0.1 mile from the intersection of SR 37 and SR 121. The winery's vineyards cascade down a hillside visible from the western end of the project corridor.

Designated scenic areas, roadways, corridors, vistas, and trails in and near the project corridor include the following:

• Eligible State Scenic Highway. The project corridor is on a stretch of SR 37 that is classified as eligible for the State Scenic Highway designation but it is not officially designated. Sonoma County has indicated its intention to pursue

an official state scenic highway designation for the portion of the project corridor within its borders in its General Plan (page OS-57) (Sonoma County 2020). The closest Officially Designated State Scenic Highway is a stretch of SR 12, 3 miles north of the project corridor.

- Sonoma County Scenic Corridor and Scenic Landscape Unit. SR 37, within the project corridor, is a designated Sonoma County Scenic Corridor (Sonoma County 2020). In addition, the parts of Sonoma County through which the project corridor passes (Napa-Sonoma Marsh and Sears Point) are county-designated Scenic Landscape Units.
- Solano County Scenic Roadway. SR 37, within the project corridor, is a designated Solano County Scenic Roadway (County of Solano 2008).
- **Refuge/Cullinan Ranch.** The east, central, and westernmost parts of the project corridor pass through or are adjacent to the Refuge, a 13,190-acre refuge established to protect migratory birds, wetland habitat, and endangered species. Cullinan Ranch is a 1,500-acre recreational area in the refuge that is accessible from the project corridor. It contains open-water areas accessible to boaters and fishers and an approximately 1-mile walking trail. Cullinan Ranch features an observation pier, interpretive panels, and a gangway to a dock. Waterfowl hunters use the boat launch to access hunting areas.
- San Francisco Bay Trail. The San Francisco Bay Trail is a planned 500-mile walking and cycling shoreline path around the entire San Francisco Bay. More than 350 miles of the trail is already in place. A planned segment of the San Francisco Bay Trail runs alongside SR 37, from the project corridor's eastern terminus to the western side of Sonoma Creek within the project corridor, where the planned trail turns south and follows the shoreline of Tubb's Island. Where the Tubb's Island shoreline meets Lower Tubbs Island, the planned path intersects a finished section of the San Francisco Bay Trail, which follows Tolay Creek back to the Tolay Creek/Tubbs Island Trailhead beside SR 37. From this point, the planned San Francisco Bay Trail continues alongside SR 37 to the western terminus of the project corridor.
- Wildlife Viewing Sites. There are wildlife viewing areas with walkways and interpretive panels on either side of Sonoma Creek, accessible from the project corridor.
- **Tolay Creek/Tubbs Island Trail.** Popular with bird watchers, the 8-mile Tolay Creek/Tubbs Island Trail is open to hikers and cyclists and offers views of San Pablo and Greater San Francisco Bays and surrounding ridgelines and peaks. The trail starts along the project corridor near the intersection of SR 121 and SR 37 near Sears Point on the southern side of SR 37, follows Tolay Creek, and splits for a loop around Lower Tubbs Island.

Visual impacts analyze existing visual resources, resource change created by the project, and viewer response to that change. Visual resource change is assessed by

evaluating the visual character and the visual quality of the visual resources that comprise the project area before and after construction of the project. Changes in visual character can be identified by how visually compatible a proposed project would be with the existing condition, using visual character attributes as an indicator. The project corridor is a rural highway that offers widespread views of marshlands, San Pablo Bay, agricultural lands, and the surrounding hills. Visual quality is evaluated by identifying the vividness, intactness, and unity present in the project area. Public attitudes validate the assessed level of quality and predict how changes to the project area can affect these attitudes.

Viewer response is a measure or prediction of the viewer's reaction to changes in the visual environment and has two dimensions, as previously mentioned: viewer exposure and viewer sensitivity. Overall, motorists would have moderately high to high viewer response to project-related changes, depending on specific location within the project corridor.

It was not feasible to analyze all the views in which the proposed project would be seen; consequently, it was necessary to select a number of key views (KVs) that would most clearly demonstrate the change in the project's visual resources. Five KVs were selected to represent project-related changes and to highlight changes to certain areas. The KV locations also represent the viewer groups that have the highest potential to be affected by the project, considering exposure and sensitivity.

These views are described below:

- KV-1: Looking down the westbound side of the project corridor near eastern terminus; represents motorist views of a potential new toll gantry
- KV-2: Looking toward the project corridor from Cullinan Ranch; represents views of recreationists
- KV-3: Looking down the eastbound side of the project corridor near the Refuge; represents motorist views of surrounding landscape
- KV-4: Looking at Sonoma Creek Bridge from a wildlife viewing area adjacent to the project corridor; represents views of recreationists
- KV-5: Looking down the eastbound side of the project corridor toward Tolay Creek Bridge; represents views of motorists at a potential new toll gantry

# 2.2.12.3 Environmental Consequences

# Short-Term Construction Impacts

There would be short-term and temporary impacts to visual resources during the construction of the project. Because of the flat terrain of the project area, staging areas for construction would be visible from SR 37 or near SR 37 for viewers, which include motorist, nearby workers, and recreationists. Temporary impacts include removal of existing vegetation and visibility of dust, construction equipment, materials, and the construction site. Additionally, some night-time work would be required, and construction lighting would be used to illuminate construction areas in an otherwise dark environment. Alternative 1 would involve construction of maintenance and

operation buildings near the highway. Alternative 3B would require construction at the Sonoma Creek Bridge.

# Long-Term Operational Impacts

Alternatives 2 and 3A are similar from a visual perspective, so they are considered together for this analysis. Alternatives 1 and 3B are discussed separately from the other Alternatives because they have different visual impacts. In addition to the alternatives, two outside barrier design options are considered: Type 85B and MGS Guard Rail. Type 85B is a partially see-through barrier consisting of concrete vertical posts and horizontal beams. The Type 85 barrier is 36 inches high with posts spaced every 10 feet. Type 85B outside barriers are typically used in scenic areas as a replacement to a solid barrier. MGS is a partially see-through barrier consisting of steel horizontal rails (also referred to as w-shaped beams) and wood or steel posts spaced about every 6 feet. The MGS barrier height is 31 inches. VIS-01 would be implemented to limit permanent light pollution from proposed light on new ramps, at intersections and on toll gantries, as described in more detail in Section 2.2.12.4.

During the process of reviewing all project alternatives and identifying Alternative 3B as the preferred alternative, the PDT further examined the median barrier height and shoulder barrier type selection for the preferred alternative. The PDT determined that a 36-inch-high median barrier would be acceptable, and this change is shown in the simulations for Alternative 3B in the Final EIR/EA. The change in median barrier height for Alternatives 2 and 3A may have been acceptable as well, but the shoulder barrier would likely have remained as presented for those alternatives. Alternative 1 with the movable barrier would not be affected by this decision. Simulations for Alternatives 2 and 3A, which were not identified as the preferred alternative, are not updated in the Final EIR/EA, and remain as presented in the draft document.

# Key View 1 – Looking Westbound Down the Project Corridor near the Eastern Terminus

Figure 2-1 presents the view looking westbound along the project corridor from, KV-1, near its eastern terminus. The existing median barrier is 36 inches high, and there is no outside barrier except for a short stretch of MGS several yards ahead on the right shoulder. Long-distance views are available on both sides of the highway across undeveloped marshland. From this vantage point, water is visible on both sides of the highway, depending on the season and tidal activity. Inundated marshland is currently visible on the left side of the highway and extends outside of the range of view. A few hundred yards ahead of this viewpoint, a section of the Bay is visible on the left side of the distance across the horizon line, depending on atmospheric conditions. A few signs and a streetlight are the only encroaching elements that detract from the natural qualities of the view.





#### KV-1 Proposed Condition – Alternative 1

An overhead toll gantry has been constructed across the highway. In this stretch of highway, the existing 32-inch-high median barrier is likely to be replaced by a 42-inch-high barrier (the Alternative 1 32-inch moveable barrier starts west of this location, outside of this view). An MGS outside barrier would be constructed along this stretch of the project corridor for Alternative 1, visible on the right side starting at the gantry (Figure 2-2).



Figure 2-2 Simulated Condition for Alternative 1 from KV-1

The new toll gantry would the tallest built element in view. The gantry may attract attention by creating a new tall, wide structure in a primarily low-lying natural setting. However, its gray color, slim lines, and lattice form would reduce its obtrusiveness. The gantry would only be visible at close range for several seconds. From distant views, it would be partially visible, with the open lattice structure helping reduce its visibility. The gantry would not block water views on either side of the highway.

The new, taller median barrier partially blocks views of the landscape on the left side of the highway. In the existing condition, most highway travelers can see above the median barrier. In the proposed condition, the added 6 inches to the barrier would affect some motorists' views over the barrier, depending on the height of the vehicle. Views from taller vehicles, such as trucks, are not likely to be altered. With this median barrier in place, views of the wetlands and the Bay south of SR 37 are partially or no longer available in the westbound direction from lower profile vehicles. However, views of the landscape north of the highway would be preserved under Alternative 1, as would views of distant hills. The combination of the gantry and the higher median barrier would add to the built character or features visible along this route.

Additionally, an MGS outside barrier is proposed and would be visible from KV-1. The new outside barrier starts at approximately the same location as the existing metal beam guard railing and continues into the distance. The barrier can be seen through at oblique angles. However, in frontal views down the highway, the barrier appears more

solid. The barrier would not be tall enough substantially to block views, but it would block some views of the adjacent vegetation. However, coupled with the new medium barrier, the outside barrier would have moderate to high resource change because it would stand out as a new feature in this rural setting.

The movable barrier would require new structure(s) or buildings alongside the highway for maintenance and worker crews. These have not been defined but would be present at either end of the highway. These would introduce a new building(s) where none exist along this corridor. This is considered a high visual impact.

The project at this location would have a moderate to high viewer response and a moderate to high resource change. Combined with a moderately high level of viewer response, Alternative 1 would create a moderately high to high level of visual impact from KV-1.

### KV-1 Proposed Condition – Alternatives 2 or 3A

An additional lane has been added to each side of the highway; three westbound lanes are now merging into two. A new toll gantry, a higher median barrier, and outside barrier would also be constructed. The visual impacts of the proposed gantry and the median barrier would be the same as those discussed for KV-1 Alternative 1 (see discussion above). The median barrier shown on Figure 2-3 is the same height as the median barrier pictured in Alternative 1 (42 inches). However, it appears to be lower in this simulation because the viewer is farther away. From the middle or left lane, the median barrier would partially block views of the wetlands and Bay on the southern side of the highway. Views of distant hills are preserved under Alternatives 2 or 3A.

Under Alternatives 2 or 3A, either MGS or Type 85B outside barriers would be considered. The Type 85B outside barrier option is simulated on Figure 2-3 because it represents the taller option under consideration. The new outside barrier starts beyond the existing MGS barrier and continues into the distance. The Type 85B would be made of concrete and slightly taller than the MGS. However, visual impacts would be similar. The barrier can be seen through at oblique angles. However, in frontal views down the highway, the barrier appears more solid and would block views of adjacent vegetation. Coupled with the new medium barrier, the outside barrier would have moderate to high resource change because it would stand out as a new feature in this rural setting.



# Figure 2-3 Simulated Condition for Alternatives 2 or 3A from KV-1 with the Type 85B Outside Barrier

The project at this location would have a moderate to high viewer response and a moderate to high resource change. Combined with a moderately high level of viewer response, Alternatives 2 or 3A would create a moderately high level of visual impact from KV-1.

Figure 2-4 represents visual changes that would be created by Alternative 3B. Alternative 3B would have visual changes similar to those of Alternatives 2 and 3A, discussed above. An additional lane has been added to each side of the highway and, at KV-1, three westbound lanes are now merging into two. A new toll gantry, a higher median barrier, and an outside barrier have also been constructed. These features would present a moderate to high resource change.

#### KV-1 Proposed Condition – Alternative 3B



Figure 2-4 Simulated Condition for Alternative 3B from KV-1

The difference is that under Alternative 3B, an MGS outside barrier would be constructed, as shown on Figure 2-3. The new outside barrier starts at approximately the same location as the existing metal beam guard railing and continues into the distance. The wooden legs are spaced apart, which provides areas were viewers could see through at oblique angles. The barrier would not be tall enough substantially to block views, but it would block some views of the adjacent vegetation. Coupled with the new medium barrier, the outside barrier would have moderate to high resource change, because it would stand out as a new feature in this rural setting.

The combination of the proposed gantry, 36-inch median barrier, and additional lane would add to the built character or features visible along this route. Because views of the coastline on the left side of the view would be blocked (depending on the height of the vehicle), long-term resource change would be moderately high from KV-1 for Alternative 3B. Combined with a moderately high level of viewer response, Alternative 3B would create a moderately high level of visual impact from KV-1.

### Key View 2 – View from Cullinan Ranch Toward Project Corridor

KV-2 looks south toward the project corridor from the Cullinan Ranch parking lot (Figure 2-5). In this view, the existing 32-inch median barrier largely blocks views of the wetlands and Bay south of the highway. Only the tops certain shrubs and trees are visible. Depending on weather conditions, hills on the other side of the Bay are partially visible across the horizon above the median barrier.



Figure 2-5 Existing Condition at KV-2

# KV-2 Proposed Condition – Alternative 1, 2, 3A, and 3B

Alternatives 1, 2, 3A and 3B are grouped in this analysis because they have the same or very similar visual impacts at this location. The median barrier height as seen from the Cullinan Ranch parking lot is very similar for all three alternatives. On the right and left side of the simulations, the guard rails are MGS (Alternatives 1 and 3B) or Type 85B (Alternatives 2 and 3A). As Figure 2-5 shows, at this location, the visual change from existing conditions would be minor.

For Alternative 1, the existing 36-inch median barrier has been replaced by a 42-inch barrier, and an outside barrier has been constructed. An MGS outside barrier is proposed. Although the combination of new median barrier and outside barrier block more of the vegetation and distant hills, the project corridor appears substantially similar to the existing condition (Figure 2-6). Visual character and quality would not be substantially altered. Long-term resource change for Alternative 1 would be very low from KV-2. The view is substantially similar to existing conditions in terms of visual character and quality. Long-term resource change for Alternative 1 would be very low from KV-2.



Figure 2-6 Simulated Condition KV-2 – Alternative 1

Alternatives 2, 3A, and 3B at location KV-2 would have similar visual impacts as Alternative 1. The existing 32-inch median barrier has been replaced by a 36-inch barrier, and an outside barrier has been constructed. The new, taller median barrier blocks more of the vegetation and the distant hills beyond the highway. On Figure 2-7, the view is simulated with an MGS outside barrier. Either MGS or a Type 85B outside barrier is proposed at this location. Both barrier types would have a low resource change. At this location, the project corridor appears substantially similar to the existing condition. The view is substantially similar to existing conditions in terms of visual character and quality. The long-term resource change for Alternatives 2, 3A, and 3B would be very low from KV-2.



KV-2 Proposed Condition – Alternatives 2, 3A, and 3B

Figure 2-7 Simulated Condition KV-2 – Alternatives 2, 3A, and 3B

# Key View 3 – Looking Eastbound Down Project Corridor Near Cullinan Ranch

Figure 2-8 presents a view from KV-3 looking east from the project corridor along the Refuge and NSMWA. This stretch of highway is undeveloped and offers long-distance views of wetlands and open water as well as views of hills in the distance. North of the highway (left side) are views of open water and marshland in Cullinan Ranch recreational area. South of the highway (right side) are views of seasonally and tidally inundated marshlands that form the coastline of San Pablo Bay.



Figure 2-8 Existing Condition KV-3

### KV-3 Proposed Condition – Alternative 1

Under Alternative 1, the highway would be widened by approximately 4 feet to construct a three-lane highway with a 32-inch movable median barrier separating the two directions of traffic. The new median barrier would be the same height as the existing barrier. An outside barrier would also be installed. Figure 2-9 simulates the movable median barrier and MGS outside barrier design option.

The additional lane and outside barrier somewhat alter the character of the highway. On the right side of the highway, the addition of the MGS creates a low visual barrier that somewhat reduces the expansiveness of the view, but the MGS barrier is not tall enough to substantially block views of the landscape south of the highway. The MGS would have a moderate to low resource change. On the left side of the highway, the combination of new median barrier and the new outside MGS barrier would decrease visibility of wetlands and open water north of the highway from most vehicles but would not entirely block it. Views of distant hills across the horizon line are not substantially affected.

On the whole, Alternative 1 with the MGS outside barrier design option would largely preserve views of adjacent landscape on both sides of the highway, but add an outside barrier. The project would create a moderately low level of adverse resource change.



Figure 2-9 Simulated Condition KV-3. Alternative 1 with MGS Outside Barrier

KV-3 Proposed Condition – Alternatives 2 or 3A with MGS Outside Barrier

Figure 2-10 shows a simulated view of Alternatives 2 or 3A with an MGS outside barrier. Under Alternatives 2 or 3A, the highway has been widened to accommodate an additional lane in each direction and 4-foot outside shoulders. Furthermore, the existing 32-inch median barrier has been replaced with a 42-inch median barrier.

The widened freeway and new barriers give the roadway a more enclosed character. An MGS outside barrier is simulated on Figure 2-10, although the Type 85B design is also being considered. On the right side, the MGS forms a new low barrier but is not tall enough to substantially alter views of the adjacent landscape. However, coupled with the new medium barrier, the outside barrier would have a high resource change. The new median barrier blocks views of the adjacent landscape north (left side) of the highway from many vehicles. In Figure 2-10, all though the tops of shrubs adjacent to the highway can still be seen, but marshland is no longer visible. With this median barrier, views of the marshland north of SR 37, including Cullinan Ranch, can no longer be readily seen in the eastbound direction from most lower profile vehicles. Distant hills are still visible in the background, but portions of the hills are slightly blocked by the median barrier.



Figure 2-10 Simulated Condition KV-3. Alternatives 2 or 3A with MGS Outside Barrier

Alternatives 2 and 3A with the MGS outside barrier option would create a high resource change because views of open water in the Refuge on the northern side of the highway would be blocked from most motorists.

### KV-3 Proposed Condition – Alternative 3B with MGS Outside Barrier

Figure 2-11 shows a simulated view of Alternative 3B with an MGS outside barrier. Under Alternative 3B, the highway has been widened to accommodate an additional lane in each direction, and 8-foot outside shoulders. The existing 32-inch median barrier has been replaced with a 36-inch median barrier.

The view and changes are similar to Alternatives 2 and 3A, with the exception that the median barrier is shown 6 inches lower; and the outside shoulder is 8 feet wide (Alternatives 2 and 3A): under Alternative 3B, it is 4 feet wide. The evaluation is the same as for Alternatives 2 and 3A. Alternative 3B with the MGS outside barrier option would create a high resource change because views of open water in the Refuge on the northern side of the highway would be blocked from most motorists. However, mountains and hills in the distance can still be seen, so scenic views would not be completely impaired.



Figure 2-11 Simulated Condition KV-3. Alternative 3B with Type 85B Outside Barrier

### Key View 4 – Looking at Project Corridor From Wildlife Viewing Area

Figure 2-12 presents the existing view from a wildlife viewing area accessible on the westbound side of the project corridor. It features views of the Sonoma Creek Bridge and surrounding marshlands, with Sears Point in the background. A series of lattice steel electric transmission towers is also visible.

No simulation was created for this viewpoint because project-related changes to Sonoma Creek Bridge would be so minor that they would not alter the character or quality of the view from this viewpoint. For Alternative 1, Approximately 4 to 5 feet of widening on the westbound side of the road is proposed, but the Sonoma Creek Bridge would not be widened. The road widening in advance of the bridge would not create a change from the current view. Additionally, under Alternative 1, the existing median barrier would be replaced with a barrier that is the same height. This change in median barrier is not considered noticeable.



Figure 2-12 Existing Condition KV-4

Alternative 3B would widen the Sonoma Creek Bridge on both sides. From the perspective at KV-4, the impact would be low to moderate because the bridge would be widened, but once completed, would not appear noticeably different from the existing structure.

For Alternatives 2 and 3A, no changes would be made to the bridge width or outside barriers. No new signage or lighting is proposed. Under Alternatives 2 and 3A, the median barrier would be replaced with one that is 4 to 6 inches taller. Neither change in median barrier height would be very noticeable from this point because the visual mass or shape of the bridge would not be altered.

### Key View 5 – Looking Eastbound Down Project Corridor Toward Tolay Creek Bridge

Figure 2-13 presents a view from KV-5 looking eastbound toward Tolay Creek Bridge near the western terminus of the project corridor. The Tolay Creek Bridge is not very visible and is only marked by a section of concrete outside the barrier, beyond the railroad tracks. Just ahead are marshlands associated with Tolay Creek. Behind the bridge are the hills of Sears Point. This view is of farmland typical of the western end of the project corridor. This particular view has encroaching elements such as rail crossing infrastructure and electric transmission lines.



Figure 2-13 Existing Condition KV-5

### KV-5 Proposed Condition – Alternative 1

In Figure 2-13, Alternative 1 is simulated with a Type 85B outside barrier; it is the taller barrier option under consideration. However, the MGS design is also being considered. Under Alternative 1, the highway has been widened to two lanes in each direction. New median barrier and outside barriers have been added to the roadway. In the distance, a proposed toll gantry is visible.

The addition of an extra lane in each direction adds additional structures to this relatively rural highway. KV-5 is near the western terminus of the project corridor, which slopes gradually up to Sears Point (Figure 2-14). KV-5 shows the highway gently sloping downhill. From this vantage point, the new median and outside barriers do not block views of the surrounding landscape. The new toll gantry is visible but not an entirely new element in the view because its open lattice design and gray color help blend into the horizon line, and is similar to the railroad gates. The most prominent parts of the gantry are its side poles. However, these are of a form and weight similar to many other utility and railroad poles in the landscape. The gantry does not block views of the surrounding landscape or of distant hills. The project improvements would blend in with the existing railroad structure and signage. Therefore, Alternative 1 would create a moderate resource change from this vantage point.



Figure 2-14 Simulated View KV-5 – Alternative 1

### KV-5 Proposed Condition – Alternatives 2 or 3A

In Figure 2-15, Alternatives 2 and 3A are simulated with a Type 85B outside barrier as the taller barrier option under consideration. However, the MGS design is also being considered. Under Alternatives 2 or 3A, the highway has been widened. The westbound side of the highway now has two lanes and the eastbound side three, including an HOV lane. This HOV lane merges into the second lane just past Tolay Creek Bridge. The eastbound lanes would taper into two lanes toward the east. New median and outside barriers have been added to the roadway. In the distance, a new toll gantry is visible.

Alternatives 2 and 3A would create a wider roadway than Alternative 1 from KV-5 because of the eastbound HOV transition lane that merges a few hundred yards ahead of the viewpoint and because of the pullout. However, other aspects of the alternative are similar to those discussed for Alternative 1 and would create a similar level of resource change. Alternatives 2 or 3A would create a moderate resource change from this vantage point.



Figure 2-15 Simulated View KV-5 – Alternatives 2 or 3A

# KV-5 Proposed Condition – Alternative 3B

In Figure 2-16, Alternative 3B is simulated with a 36-inch median barrier and an MGS outside barrier design. It appears similar to Alternatives 2 and 3A, except that it has an MGS barrier instead of the Type 85 outside barrier. All other conditions are the same. Alternative 3B would create a moderate resource change from this vantage point.



Figure 2-16 Simulated View KV-5 – Alternative 3B

Table 2-15 summarizes the differences between alternatives with respect to the visual impact findings.

Key View	Alternative	Outside Barrier Design Option	Viewer Response	Resource Change	Visual Impact
1	1	MGS	Moderate-High	Moderate-High	Moderate-High
1	2/3A	Either	Moderate-High	Moderate-High	Moderate-High
1	3B	MGS	Moderate-High	Moderate-High	Moderate-High
2	1	MGS	Moderate-High	Very Low	Moderate-Low to Moderate
2	2/3A	Either	Moderate-High	Very Low	Moderate-Low to Moderate
2	3B	MGS	Moderate-High	Very Low	Moderate-Low to Moderate
3	1	MGS	High	Moderate-Low	Moderate
3	2/3A	MGS	High	High	High
3	2/3A	Type 85B	High	High	High
3	3B	MGS	High	High	High
4	1/2/3A1/ 2/3A	N/A*	Moderate	Very Low	Low
4	3B	N/A*	Moderate	Very Low	Moderate
5	1	Either	Moderate	Moderate	Moderate
5	2/3A	Either	Moderate	Moderate	Moderate
5	3B	MGS	Moderate	Moderate	Moderate

 Table 2-15
 Summary of Key View Narrative Ratings

# 2.2.12.4 Avoidance, Minimization, and/or Mitigation Measures

The following measure would be implemented to limit impacts from light pollution for all proposed build alternatives:

• VIS-01: Limit Light Pollution. For permanent impacts, lighting on new ramps, at intersections, in advance of tolling gantries, and at CHP enforcement areas would be designed to limit light pollution and have minimum impact on the surrounding environment. All light fixtures would have light-emitting diodes configured at the minimum necessary number of bulbs, optimal mounting height, mast-arm length, and angle to restrict light to the roadways. Where applicable, shields on the fixtures to prevent light trespass to adjacent properties would be considered during the detailed design phase.

Final design and permitting will involve providing public access amenities. Section 2.2.3.3 notes that public access would be addressed in the final project development. This may include funding to construct a bike path as part of the Bay Trail, which would provide access opportunities for views of the Bay and the shoreline.

# 2.2.13 Cultural Resources

# 2.2.13.1 Regulatory Setting

The term "cultural resources," as used in this document, refers to the "built environment" (e.g., structures, bridges, railroads, and water conveyance systems); places of traditional or cultural importance; and archaeological sites (both prehistoric and historic), regardless of significance. Under federal and state laws, cultural resources that meet certain criteria of significance are referred to by various terms, including "historic properties," "historic sites," "historical resources," and "tribal cultural resources." Laws and regulations dealing with cultural resources include the NHPA and CEQA.

The NHPA, as amended, sets forth national policy and procedures for historic properties, defined as districts, sites, buildings, structures, and objects included in or eligible for listing in the National Register of Historic Places (NRHP). Section 106 of the NHPA requires federal agencies to take into account the effects of their undertakings on historic properties and to allow the Advisory Council on Historic Preservation (ACHP) the opportunity to comment on those undertakings, following regulations issued by the ACHP (36 CFR 800). On January 1, 2014, the First Amended Section 106 Programmatic Agreement (PA) among FHWA, the ACHP, the California SHPO, and Caltrans went into effect for Department projects, both state and local, with FHWA involvement. The PA implements the ACHP's regulations, 36 CFR 800, streamlining the Section 106 process and delegating certain responsibilities to Caltrans. FHWA's responsibilities under the PA have been assigned to Caltrans as part of the Surface Transportation Project Delivery Program (23 USC 327).

CEQA requires the consideration of cultural resources that are historical resources and tribal cultural resources, as well as "unique" archaeological resources. California PRC Section 5024.1 established the California Register of Historical Resources (CRHR) and outlined the necessary criteria for a cultural resource to be considered eligible for listing in the CRHR and, therefore, a historical resource. Historical resources are defined in PRC Section 5020.1(j). In 2014, AB 52 added the term "tribal cultural resources" to CEQA, and AB 52 is commonly referenced instead of CEQA when discussing the process to identify tribal cultural resources (as well as identifying measures to avoid, preserve, or mitigate effects to them). Defined in PRC Section 21074(a), a tribal cultural resource is a CRHR or local register-eligible site, feature, place, cultural landscape, or object which has a cultural value to a California Native American tribe. Tribal cultural resources must also meet the definition of a historical resource. Unique archaeological resources are referenced in PRC Section 21083.2.

PRC Section 5024 requires state agencies to identify and protect state-owned historical resources that meet the NRHP listing criteria. It further requires Caltrans to inventory state-owned structures in its rights-of-way. Sections 5024(f) and 5024.5 require state agencies to provide notice to and consult with the SHPO before altering, transferring, relocating, or demolishing state-owned historical resources that are listed

on or are eligible for inclusion in the NRHP or are registered or eligible for registration as California Historical Landmarks. Procedures for compliance with PRC Section 5024 are outlined in a Memorandum of Understanding between Caltrans and SHPO, effective January 1, 2015. For most federal-aid projects on the State Highway System, compliance with the Section 106 PA would satisfy the requirements of PRC Section 5024.

# 2.2.13.2 Affected Environment

The following section is based on information from the *Archeological Survey Report* (AECOM 2021c); *the Extended Phase I Report* (AECOM 2021d); and the *Historic Property Survey Report* (AECOM 2021e) for the proposed project. These reports were completed in April 2021.

The Area of Potential Effects (APE) defined for the Cultural Resources study encompasses all areas in the physical footprint of improvements proposed for Alternatives 1, 2, 3A, and 3B. The project's APE encompasses nearly the entirety of Caltrans right-of-way and extends from approximately Lakeville Highway to the west, the Sacramento Street Overcrossing in Vallejo to the east, and approximately 0.2 mile north on SR 121 in Sonoma County. A project staging area is proposed on the northern side of SR 37, east of SR 121, in a portion of APN 068-190- 017. A project staging area is proposed on the northern side of SR 37, east of SR 121, in a portion of APN 068-190- 017. The APE was developed to assess the project's potential effects on cultural resources.

# **Records and Archival Review**

A cultural resources records search was conducted by AECOM Senior Archeologist, Karin Beck, at the Northwest Information Center of the California Historical Resources Information System, Sonoma State University, on October 15, 2019. Site records and previous studies were accessed for the APE and a 0.5-mile radius of the APE. The records search identified one prehistoric archaeological site and four historic-era resources mapped in or adjacent to the APE. The background search also found that almost the entire APE is mapped as moderately sensitive for submerged archaeology; however, small portions of the APE are mapped as high or highest sensitivity for submerged archaeology.

# Field Survey and Subsurface Testing Results

On October 5 and October 6, 2020, AECOM conducted geoarchaeological testing to determine whether archaeological deposits were associated with one previously recorded archaeological site and whether any unrecorded archeological sites were contained in the APE. A drill rig was used to excavate a series of exploratory bores. A total of six Geoprobe bores were excavated. No artifacts, features, or culturally sensitive soils were identified in any of the bores. Additionally, no previously unrecorded archaeological sites were identified.

### **Native American Consultation**

The NAHC was contacted in October 2019 to request a search of the Sacred Lands File for sacred lands or other cultural properties of significance to Native Americans in or near the APE. The NAHC replied via email on October 25, 2019, stating that a search of the file had been completed and was negative for cultural resources. The NAHC also provided a list of Native American individuals who may have information related to cultural resources in the APE, and/or concerns about the project. On November 1, 2019, Caltrans requested AECOM to send letters describing the proposed project to the Native American individuals specified by the NAHC and requested any information or concerns regarding the APE. No responses were received to these letters.

AECOM followed up with individuals via phone on December 23, 2019. A call was received from the United Auburn Indian Community (UAIC), Anna Starkey. She was provided via email with the project map, the NAHC response listing the UAIC as a potential interested party, and AECOM's letter of notification. On December 30, Ms. Starkey responded that the APE was outside the Tribe's boundary, and that the Tribe did not have information on potentially sensitive cultural areas and would not be requesting consultation for the project.

An email was received from Federated Indians Graton Rancheria (FIGR) Tribal Historic Preservation Officer, Buffy McQuillen, on January 2, 2020. Ms. McQuillen stated that the Tribe was concerned that the project could impact cultural resources and requested consultation and the opportunity to comment on the project as it develops. Caltrans would consult with FIGR moving forward.

### Historic Architectural and Archaeological Resources

The background search identified one prehistoric archaeological site, designated as Nelson Mound, and four historic-era resources mapped in or adjacent to the APE. Historic-era resources in the APE include:

- The Mare Island Naval Shipyard is a California Historic Landmark and National Historic Landmark District. It was established as a NRHP property in 1975. Its boundaries increased in 1996. The 1996 expanded boundaries of the historic district terminate at G Street, approximately 0.70 mile south of the SR 37 APE.
- A small portion (0.12 acre) of the Tubbs Island levee is in the APE. The Tubbs Island Levee does not meet the criteria for listing in the NRHP or the CRHR, nor does it retain historic integrity.
- An unrecorded segment of the Northwestern Pacific Railroad transects the APE near the western end of the APE at Sears Point. It does not appear to be eligible for listing in the NRHP or CRHR as a historic district.
- This pile of historical artifacts was first recorded by Nels Nelson in 1907 and designated as Nelson Mound. It was described as a "shellheap...cut by the railroad" with a few feet of material still intact on a knoll west of the track. In

2005, two historic-era artifacts—a ceramic plate fragment with a partial makers mark, and a yellowware fragment—were observed on the side of the knoll, halfway down to the railroad tracks. However, no definitive evidence suggested that any portion of the prehistoric components of this site as originally recorded by Nelson remained. During AECOM's filed investigations, no archaeological deposits related to the site or a previously unrecorded archaeological resource were identified in either of the bores.

# 2.2.13.3 Environmental Consequences

Most project construction would occur in Caltrans' right-of-way, which consists of areas that are paved or have had previous ground disturbances. Ground-disturbing activities during construction of the project could affect unknown buried cultural resources in areas adjacent to SR 37. The background research and literature review conducted for this project identified one previously recorded prehistoric archaeological site, designated as Nelson Mound. During the field survey and subsurface testing, no artifacts, features, or culturally sensitive soils were identified in any of the bores. The conclusion of investigations at this location found that an archaeological site is not present at this location.

The conclusion of the Historic Properties Survey Report (AECOM 2021e) was that historic properties present within the APE were either exempt from evaluation (the Northwestern Pacific Railroad), were determined not eligible for listing in the NRHP (three bridges and an overcrossing), and that the Mare Island Naval Shipyard District is eligible for NRHP nomination, but that no resources associated with that property are adjacent or within the APE, and would not be affected. For this project, no historic properties and no Section 4(f) historic sites would be affected (nonhistoric Section 4(f) properties are discussed in Appendix B). Therefore, there would be no impact to cultural resources.

# 2.2.13.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-CULT-01. Discovery of Human Remains; and
- PF-CULT-02. Discovery of Archeological Materials.

# 2.3 Physical Environment

# 2.3.1 Hydrology and Floodplain

# 2.3.1.1 Regulatory Setting

EO 11988 (Floodplain Management) directs all federal agencies to refrain from conducting, supporting, or allowing actions in floodplains unless it is the only practicable alternative. FHWA requirements for compliance are outlined in 23 CFR 650 Subpart A.

2-80
To comply, the following must be analyzed:

- The practicability of alternatives to any longitudinal encroachments.
- Risks of the action.
- Impacts on natural and beneficial floodplain values.
- Support of incompatible floodplain development.
- Measures to minimize floodplain impacts and to preserve/restore any beneficial floodplain values affected by the project.

The base floodplain is defined as "the area subject to flooding by the flood or tide having a 1 percent chance of being exceeded in any given year." An encroachment is defined as "an action within the limits of the base floodplain."

# 2.3.1.2 Affected Environment

The following discussion is based on the SR 37 *Draft Water Quality Assessment Report* (WRECO 2021) and Location Hydraulic Study Report (WRECO 2021).

The project is adjacent to San Pablo Bay, which connects to San Francisco Bay. The project crosses over three water bodies: Tolay Creek, Sonoma Creek, and the Napa River, all of which drain into San Pablo Bay. The Napa Slough is a surrounding water body near the project location, north of SR 37, that is not a receiving water (i.e., it does not receive treated or untreated wastewaters [e.g., stormwater runoff] or effluent [e.g., liquid waste or sewage]). The Napa Slough drains south of the NSMWA and is part of the Refuge. The slough then flows into Sonoma Creek before draining south toward San Pablo Bay. East of the project limits, Dutchman Slough joins the Napa River and drains to San Pablo Bay near Mare Island.

#### Floodplains

As determined from FEMA Flood Insurance Rate Maps (FIRMs), most of the project is in SFHA Zone AE. This zone represents the base floodplain with areas subject to flooding by the 100-year flood event, where base floodplain elevations are provided. In these areas, the 100-year flood elevation is approximately 10 to 11 feet North American Vertical Datum of 1988 (NAVD88).

Portions of the project area adjacent to Tubbs Island (between Tolay Creek and Sonoma Creek) are in SFHA Zone VE, which are coastal areas subject to coastal high-hazard flooding and to inundation by the 1-percent-annual-chance flood event, with additional hazards due to storm-induced velocity wave action. In these project areas, the 100-year flood elevation is approximately 11 to 12 feet NAVD88.

A small portion of the project area just east of the Tolay Creek crossing is in a shaded Zone X area. Zone X represents areas of moderate flood hazard, usually depicted on FIRMs as between the limits of the base and 500-year floods.

# Natural and Beneficial Floodplain Values

The surrounding floodplain in the area surrounding the project consists of waterways, open space, and agricultural lands, as defined in the General Land Use Plans of Napa, Solano, and Sonoma Counties (County of Napa 2008; County of Solano 2008; Sonoma County 2020).

#### Incompatible Floodplain Development

The project would follow the existing SR 37 roadway within the project limits and would not create new access to developed or undeveloped land in the flood zone. Therefore, the project would not support incompatible floodplain development.

#### 2.3.1.3 Environmental Consequences

#### No Build Alternative

The No Build Alternative would not affect the floodplains within the project limits. None of the existing floodplain values in or adjacent to the project would be altered under the No Build Alternative.

#### **Build Alternative**

Under all Build Alternatives, the project would not change the overall land use in the watershed basin and would not add substantial amounts of impervious area to the watershed. Although the Build Alternatives would require different amounts of permanent fill to widen the highway, most improvements in the project would re-work existing impervious areas. The proposed fill placed in the floodplain is relatively minor in the context of the greater floodplain area and is not anticipated to impede flood waters, affect bay level floodplains, or substantially reduce the area available to convey floodwaters.

23 CFR 650.105 defines a significant floodplain encroachment of a highway as: (1) a significant potential for interruption or termination of a transportation facility which is needed for emergency vehicles or provides a community's only evacuation route; (2) a significant risk; or (3) a significant adverse impact on natural and beneficial floodplain values. During a 100-year flood event, flood waters could cause the low-lying sections of SR 37 to close, which could affect travel, evacuation access. However, the project would not cause any changes to these existing flood events or the frequency of their occurrence. The following routes provide practicable detours around the project area in the case of traffic interruptions from potential flooding conditions:

- For westbound traffic: SR 29 south to I-80, west to Richmond Parkway to I-580, west to Richmond Bridge to U.S. 101
- For eastbound traffic: U.S. 101 to I-580 east to Richmond Bridge to Richmond Parkway or I-80

There is the potential for temporary closures, but not a significant potential for longterm interruption or termination of SR 37, because these events would not occur often. Closures would require drivers to temporarily use the alternative routes identified. Potential short-term adverse effects to the natural and beneficial floodplain values during the construction of the new bridge may include the following: (1) the loss of vegetation during construction activity; and (2) temporary disturbance of wildlife and aquatic habitat. Construction would not create a hazard to life during the service life of the highway, nor would it result in adverse effects to natural and beneficial flood-plain values. Biological compensation described in Section 2.4 would be provided as appropriate where natural and beneficial floodplain values are potentially impacted within the project area.

None of the existing floodplain values in or adjacent to the project would be altered under any of the Build scenarios, and the project would not create a significant risk to property. The project would not result in a significant encroachment into the base 100-year floodplain.

# 2.3.1.4 Avoidance, Minimization, and/or Mitigation Measures

The amount of new impervious surface area added would not have an impact to the flows within the project's limits. Therefore, no floodplain avoidance or minimization measures are proposed, and no mitigation measures are required for this project under the Build Alternatives.

# 2.3.2 Water Quality and Stormwater Runoff

# 2.3.2.1 Regulatory Setting

# Federal Requirements: Clean Water Act

In 1972, Congress amended the Federal Water Pollution Control Act, making the addition of pollutants to the waters of the United States (U.S.) from any point source unlawful unless the discharge is in compliance with a National Pollutant Discharge Elimination System (NPDES) permit. This act and its amendments are known today as the Clean Water Act (CWA). Congress has amended the act several times. In the 1987 amendments, Congress directed dischargers of stormwater from municipal and industrial/construction point sources to comply with the NPDES permit scheme. The following are important CWA sections:

- Sections 303 and 304 require states to issue water quality standards, criteria, and guidelines.
- Section 401 requires an applicant for a federal license or permit to conduct any activity that may result in a discharge to waters of the U.S. to obtain certification from the state that the discharge would comply with other provisions of the act. This is most frequently required in tandem with a Section 404 permit request (see below).
- Section 402 establishes the NPDES, a permitting system for the discharges (except for dredge or fill material) of any pollutant into waters of the U.S. RWQCBs administer this permitting program in California. Section 402(p)

requires permits for discharges of stormwater from industrial/construction and municipal separate storm sewer systems (MS4s).

 Section 404 establishes a permit program for the discharge of dredge or fill material into waters of the U.S. This permit program is administered by USACE.

The goal of the CWA is "to restore and maintain the chemical, physical, and biological integrity of the Nation's waters."

The USACE issues two types of 404 permits: General and Individual. There are two types of General permits: Regional and Nationwide. Regional permits are issued for a general category of activities when they are similar in nature and cause minimal environmental effect. Nationwide permits are issued to allow a variety of minor project activities with no more than minimal effects.

Ordinarily, projects that do not meet the criteria for a Regional or Nationwide Permit may be permitted under one of the USACE's individual permits. There are two types of individual permits: standard permits and letters of permission. For individual permits, the USACE decision to approve is based on compliance with U.S. EPA's Section 404 (b)(1) Guidelines (40 CFR Part 230), and whether the permit approval is in the public interest. The Section 404(b)(1) Guidelines (Guidelines) were developed by the U.S. EPA in conjunction with the USACE, and allow the discharge of dredged or fill material into the aquatic system (waters of the U.S.) only if there is no practicable alternative which would have less adverse effects. The Guidelines state that the USACE may not issue a permit if there is a least environmentally damaging practicable alternative (LEDPA) to the proposed discharge that would have lesser effects on waters of the U.S. and not have any other significant adverse environmental consequences. According to the Guidelines, documentation is needed that a sequence of avoidance, minimization, and compensation measures has been followed, in that order. The Guidelines also restrict permitting activities that violate water quality or toxic effluent standards, jeopardize the continued existence of listed species, violate marine sanctuary protections, or cause "significant degradation" to waters of the U.S. In addition, every permit from the USACE, even if not subject to the Section 404(b)(1) Guidelines, must meet general requirements. See 33 CFR 320.4. A discussion of the LEDPA determination, if any, for the document is included in the Wetlands and Other Waters section.

# State Requirements: Porter-Cologne Water Quality Control Act

California's Porter-Cologne Act, enacted in 1969, provides the legal basis for water quality regulation in California. This act requires a "Report of Waste Discharge" for any discharge of waste (liquid, solid, or gaseous) to land or surface waters that may impair beneficial uses for surface and/or groundwater of the state. It predates the CWA and regulates discharges to waters of the state. Waters of the State include more than just waters of the U.S., like groundwater and surface waters not considered waters of the U.S. Additionally, it prohibits discharges of "waste" as defined, and this definition is broader than the CWA definition of "pollutant." Discharges under the Porter-Cologne

Act are permitted by Waste Discharge Requirements (WDRs) and may be required even when the discharge is already permitted or exempt under the CWA.

The State Water Resources Control Board (SWRCB) and RWQCBs are responsible for establishing the water quality standards (objectives and beneficial uses) required by the CWA and regulating discharges to ensure compliance with the water quality standards. Details about water quality standards in a project area are included in the applicable RWQCB Basin Plan. In California, RWQCBs designate beneficial uses for all water body segments in their jurisdictions and then set criteria necessary to protect those uses. As a result, the water quality standards developed for particular water segments are based on the designated use and vary depending on that use. In addition, the SWRCB identifies waters failing to meet standards for specific pollutants. These waters are then state-listed in accordance with CWA Section 303(d). If a state determines that waters are impaired for one or more constituents and the standards cannot be met through point source or nonpoint source controls (NPDES permits or WDRs), the CWA requires the establishment of total maximum daily loads (TMDLs). TMDLs specify allowable pollutant loads from all sources (point, nonpoint, and natural) for a given watershed.

# SWRCB and RWQCBs

The SWRCB administers water rights, sets water pollution control policy, and issues water board orders on matters of statewide application, and oversees water quality functions throughout the state by approving Basin Plans, TMDLs, and NPDES permits. RWCQBs are responsible for protecting beneficial uses of water resources within their regional jurisdiction using planning, permitting, and enforcement authorities to meet this responsibility.

# **NPDES Program**

# MS4s

Section 402(p) of the CWA requires the issuance of NPDES permits for five categories of stormwater discharges, including MS4s. An MS4 is defined as "any conveyance or system of conveyances (roads with drainage systems, municipal streets, catch basins, curbs, gutters, ditches, human-made channels, and storm drains) owned or operated by a state, city, town, county, or other public body having jurisdiction over stormwater, that is designed or used for collecting or conveying stormwater." The SWRCB has identified Caltrans as an owner/operator of an MS4 under federal regulations. The Department's MS4 permit covers all Department rights-of-way, properties, facilities, and activities in the state. The SWRCB or the RWQCB issues NPDES permits for 5 years, and permit requirements remain active until a new permit has been adopted.

The Department's MS4 Permit, Order No. 2012-0011-DWQ (adopted on September 19, 2012, and effective on July 1, 2013), as amended by Order No. 2014-0006-EXEC (effective January 17, 2014), Order No. 2014-0077-DWQ (effective May 20, 2014) and Order No. 2015-0036-EXEC (conformed and effective April 7, 2015), has three basic requirements: The Department must comply with the requirements of the Construction General Permit (see below).

The Department must implement a year-round program in all parts of the state to effectively control stormwater and non-stormwater discharges.

The Department stormwater discharges must meet water quality standards through implementation of permanent and temporary (construction) BMPs, to the maximum extent practicable, and other measures as the SWRCB determines to be necessary to meet the water quality standards.

To comply with the permit, Caltrans developed the Statewide Storm Water Management Plan (SWMP) to address stormwater pollution controls related to highway planning, design, construction, and maintenance activities throughout California. The SWMP assigns responsibilities in Caltrans for implementing stormwater management procedures and practices as well as training, public education and participation, monitoring and research, program evaluation, and reporting activities. The SWMP describes the minimum procedures and practices Caltrans uses to reduce pollutants in stormwater and non-stormwater discharges. It outlines procedures and responsibilities for protecting water quality, including the selection and implementation of BMPs. The proposed project would be programmed to follow the guidelines and procedures outlined in the latest SWMP to address stormwater runoff.

# **Construction General Permit**

Construction General Permit, Order No. 2009-0009-DWQ (adopted on September 2, 2009 and effective on July 1, 2010), as amended by Order No. 2010-0014-DWQ (effective February 14, 2011) and Order No. 2012-0006-DWQ (effective on July 17, 2012), regulates stormwater discharges from construction sites that result in a disturbed soil area (DSA) of 1 acre or greater, and/or are smaller sites that are part of a larger common plan of development. By law, all stormwater discharges associated with construction activity where clearing, grading, and excavation result in soil disturbance of at least 1 acre must comply with the provisions of the General Construction Permit. Construction activity that results in soil disturbances of less than 1 acre is subject to this Construction General Permit if there is potential for significant water quality impairment resulting from the activity, as determined by the RWQCB. Operators of regulated construction sites are required to develop Storm Water Pollution Prevention Plans (SWPPPs); to implement sediment, erosion, and pollution prevention control measures; and to obtain coverage under the Construction General Permit.

The Construction General Permit separates projects into risk levels 1, 2, or 3. Risk levels are determined during the planning and design phases, and are based on potential erosion and transport to receiving waters. Requirements apply according to the risk level determined. For example, a risk level 3 (highest risk) project would require compulsory stormwater runoff pH and turbidity monitoring, and before-construction and after-construction aquatic biological assessments during specified seasonal windows. For all projects subject to the permit, applicants are required to develop and implement

an effective SWPPP. In accordance with Caltrans' SWMP and Standard Specifications, a WPCP is necessary for projects with a DSA of less than 1 acre.

# Section 401 Permitting

Under Section 401 of the CWA, any project requiring a federal license or permit that may result in a discharge to a water of the U.S. must obtain a 401 Certification, which certifies that the project would be in compliance with state water quality standards. The most common federal permits triggering 401 Certification are CWA Section 404 permits issued by the USACE. The 401 permit certifications are obtained from the appropriate RWQCB, depending on the project location, and are required before the USACE issues a 404 permit.

In some cases, the RWQCB may have specific concerns with discharges associated with a project. As a result, the RWQCB may issue a set of requirements known as WDRs under the State Water Code (Porter-Cologne Act) that define activities, such as the inclusion of specific features, effluent limitations, monitoring, and plan submittals that are to be implemented for protecting or benefiting water quality. WDRs can be issued to address both permanent and temporary discharges of a project.

# 2.3.2.2 Affected Environment

The following discussion is based on the SR 37 *Draft Water Quality Assessment Report* (WRECO 2021).

# Surface Water Resources

The project lies along San Pablo Bay, which is connected to San Francisco Bay. The project limits are within the San Pablo hydrologic unit, in three unidentified hydrologic subareas (HSA) of the Petaluma, San Pablo, and Napa Rivers (HSAs 206.30, 206.40, and 206.50). The overall drainage pattern of the area is from the north to south, with portions of the project area in watersheds that drain to the Petaluma River hydrologic area, to the Sonoma Creek hydrologic area, and to the Napa River hydrologic area.

# **Receiving Waters**

The project's receiving waters are Tolay Creek, Sonoma Creek, Napa River, and San Pablo Bay. The creeks flow north to south and drain into San Pablo Bay (Figure 2-17, on the second page following).

# Tolay Creek

Tolay Creek is one of the surface waters to cross the project; the creek crosses SR 37 at PM SON 4.04, near Sears Point, California, and drains into San Pablo Bay. Tolay Creek is a natural meandering channel with rock slope protection near the Tolay Creek Bridge. Water surface elevations in Tolay Creek are controlled by San Pablo Bay well upstream of SR 37.

# Sonoma Creek

Sonoma Creek crosses the project at PM SOL R0.01. The creek lies between Sonoma and Solano Counties. Sonoma Creek is one of three principal streams draining the subbasin, Sonoma Valley, to the mouth of San Pablo Bay. The creek is less than 1 mile wide and composed of unconfined gravel sand, silts, clays, and peat, according to the FEMA Flood Insurance Study (FEMA 2017). The channel is described in the Sonoma Creek Bridge Inspection Report as a broad, flat tidal channel with Bay Mud that flows north to south. Water surface elevations in Sonoma Creek are controlled by San Pablo Bay well upstream of SR 37.

# Napa River

The eastern project limits end prior to the Napa River crossing, adjacent to the Mare Island interchange. The Napa River basin drains north to south and is joined by various tributaries. The river ends in broad tidal sloughs approximately 9 miles north of San Pablo Bay, as described in the FEMA Flood Insurance Study (FEMA 2016).

# San Pablo Bay

San Pablo Bay is a receiving water body of the SR 37 corridor. All creek crossings and surrounding water bodies in the project vicinity drain into San Pablo Bay. San Pablo Bay is north of San Francisco Bay.

The surface waterways in or near the project areas provide beneficial uses defined in the RWQCB Water Quality Control Plan (Basin Plan) for San Francisco Bay, including industrial service supply; commercial and sports fishing; shellfish harvesting; cold freshwater habitat; estuarine habitat; fish migration; preservation of rare and endangered species; fish spawning; warm freshwater habitat; wildlife habitat; water contact recreation; noncontact water recreation; and navigation (SFRWQCB 2019). Table 2-16 provides a summary of these existing beneficial uses at each receiving water.

# Groundwater Resources

The project area is in the San Francisco Bay Hydrologic Region and spans three groundwater basins: the Petaluma Valley Basin (2-001), and the Napa-Sonoma Valley-Sonoma Valley subbasin (2-002.02) and Napa-Sonoma Valley-Napa-Sonoma Lowlands subbasin (2-002.03) of the Napa- Sonoma Valley Basin. Table 2-17 summarizes the groundwater basins and subbasins present within the project limits, as well as the existing and potential beneficial uses listed in the Basin Plan (SFRWQCB 2019).

# CWA 303(d) List

The 2014/2016 Integrated Report, CWA Section 303(d) List/305(b) Report, (SWRCB 2017) lists the water quality impairments and TMDL for Sonoma Creek, Napa River, and San Pablo Bay. Tolay Creek is not listed as having any water quality impairments or TMDLs. Although Caltrans is a named stakeholder for the sediment TMDL at Napa River and Sonoma Creek, the tidal portions are not listed for sediment. Table 2-18 summarizes all TMDLs for each receiving water in the project area.



# FIGURE 2-17:SITE GEOLOGY



# FIGURE 2-17: SITE GEOLOGY

Beneficial Use	Tolay Creek	Sonoma Creek	Napa River	San Pablo Bay, Mare Island Strait and Tidal Waters
Industrial service supply	NO	NO	NO	YES
Commercial and sports fishing	NO	YES	YES	YES
Shellfish harvesting	NO	NO	NO	YES
Cold freshwater habitat	NO	YES	NO	NO
Estuarine habitat	NO	NO	YES	YES
Fish migration	NO	YES	YES	YES
Preservation of rare and endangered species	YES	YES	YES	YES
Fish spawning	NO	YES	YES	YES
Warm freshwater habitat	YES	YES	NO	NO
Wildlife habitat	YES	YES	YES	YES
Water contact recreation	YES	YES	YES	YES
Noncontact water recreation	NO	YES	YES	YES
Navigation	NO	NO	YES	YES

 Table 2-16
 Beneficial Uses for the Receiving Waters in the Project Area

Source: Summarized from SFRWQCB 2019

#### Table 2-17 Existing and Potential Beneficial Uses in Groundwater Basins

Use Status	Beneficial Use	Petaluma Valley Groundwater Basin; Petaluma Valley Subbasin	Napa Valley- Sonoma Basin; Sonoma Valley Subbasin	Napa Valley- Sonoma Basin; Napa-Sonoma Lowlands Subbasin
Existing	Municipal and Domestic Water Supply	Present	Present	Present
Existing	Industrial process water supply	Not Present	Present	Present
Existing	Industrial service water supply	Not Present	Present	Present
Existing	Agricultural water supply	Present	Present	Present
Potential	Industrial process water supply	Present		
Potential	Industrial service water supply	Present	_	

Source: SFRWQCB 2019

# Table 2-18CWA 303(d) List Impairments and TMDLs at Water Bodies in the<br/>Project Area

Receiving Water	Pollutant Category	Pollutant	Source
Sonoma Creek, Tidal	Nutrients	Nutrients	Agriculture Onsite Wastewater Systems (Septic Tanks)
Sonoma Creek, Tidal	Pathogens	Pathogens	Onsite Wastewater Systems (Septic Tanks)
Napa River, Mare Island Strait	Toxic Organics	PCBs	Unknown
Napa River, Mare Island Strait	Pesticides	Chlordane	Unknown
Napa River, Mare Island Strait	Pesticides	Dieldrin	Unknown
Napa River, Mare Island Strait	Pesticides	Total dichlorodiphenyltrichloroethane (DDT) (sum of 4,4'- and 2,4'- isomers of DDT, DDE, and DDD)	Unknown
Napa River, Mare Island Strait	Metals	Mercury	Unknown
Napa River, Tidal	Nutrients	Nutrients	Agriculture Onsite Wastewater Systems (Septic Tanks)
Napa River, Tidal	Pathogens	Pathogens	Agriculture Onsite Wastewater Systems (Septic Tanks)
San Pablo Bay	Pesticides	Dieldrin	Unknown
San Pablo Bay	Toxic Organics	PCBs (dioxin-like)	Unknown
San Pablo Bay	Toxic Organics	Dioxin compounds (including 2,3,7,8-TCDD)	Unknown
San Pablo Bay	Pesticides	Chlordane	Unknown
San Pablo Bay	Toxic Organics	Furan Compounds	Unknown
San Pablo Bay	Other Cause	Invasive Species	Unknown
San Pablo Bay	Toxic Organics	PCBs	Unknown
San Pablo Bay	Metals	Selenium	Unknown
San Pablo Bay	Pesticides	DDT	Unknown
San Pablo Bay	Metals	Mercury	Unknown

Source: SWRCB 2017

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### MS4s

The project is entirely within Caltrans' right-of-way and, therefore, is not subject to other MS4 permits. However, the Memorandum of Caltrans Post-Construction Stormwater and Hydromodification Standards (SFRWQCB 2008) states that Caltrans District 4 projects that are subject to 401 Water Quality Certification are required to design biofiltration devices and implement hydromodification assessment in accordance with the local city/county stormwater design criteria.

The project is adjacent to MS4 districts, including Sonoma, Solano, and Napa Counties, and all alternatives considered would comply with county stormwater design criteria.

#### 2.3.2.3 Environmental Consequences

#### No Build Alternative

No short-term water quality impacts would occur with the No Build Alternative because it would not require any construction activities.

#### **Build Alternatives**

Under all Build Alternatives, there would be no anticipated impacts to the following water quality parameters:

- Current, circulation or drainage patterns
- Flood control functions
- Hydromodification (erosion and accretion patterns)
- Recreational or commercial fisheries
- Water related recreation

These parameters are not discussed further in this document.

Impacts to biological parameters, including wetlands, wildlife, special aquatic sites, and other biological characteristics of the aquatic environment, are discussed later in this document under Section 2.4, Biological Environment.

#### Disturbed Soil Areas and Impervious Areas

Temporary water quality impacts can result from sediment discharge from DSAs and construction near water resources or drainage facilities that discharge to water bodies.

Permanent impacts to water quality result from the addition of impervious area; this additional impervious area prevents runoff from naturally dispersing and infiltrating into the ground, resulting in increased concentrated flow. The project would not remove any impervious areas under any alternative. Each alternative would rework and replace existing impervious surfaces; these impacts would be a maintenance of existing roadway and would not be considered a new permanent impact. Preliminary estimates for DSAs and added impervious area for each Build Alternatives are listed in

Table 2-19. The DSA and impervious area values would be further refined during the design phase once the limits of grading, construction staging locations, and other areas of improvement have been further developed.

Alternative	DSA (acres)	Added Impervious Area (acres)	Replaced Impervious Area (acres)	New Impervious Surfaces: Added + Replaced (acres)
1	44.86	12.17	11.57	23.74
2	44.86	19.75	20.42	40.17
3A	79.88	21.19	21.11	42.41
3B	87.42	28.25	21.27	49.52

Table 2-19	<b>Disturbed Soil and</b>	Added Impervious Ar	reas by Build Alternative
------------	---------------------------	---------------------	---------------------------

Implementation of water quality project features is required for all construction projects, in compliance with federal, state, and local requirements, and would minimize the potential for water quality impacts from runoff entering storm drains.

The Caltrans MS4 permits (SWRCB 2013) require this project to implement treatment BMPs in the Caltrans' right-of-way because the proposed improvements result in the creation or replacement of more than 1 acre of impervious area. Because SR 37 is bound on most sides by wetlands and waters, with a high groundwater table in much of the Caltrans' right-of-way, the potential locations for onsite treatment are limited. There would be a treatment deficit of approximately 25 to 40 acres, which must be treated off site.

# Suspended Particulates (Turbidity)

The additional impervious areas described above would increase the amount of runoff and sediment-laden flow directly discharging to receiving water bodies. Within the limits of the project, existing drainage facilities are expected to be modified or removed, capped, or abandoned, and new drainage features installed to convey runoff. Rock slope protection would be placed at locations where necessary, and existing culverts would be maintained or extended where shoulder widening is necessary. These drainage design features would limit increases in suspended sediment in storm drain systems and receiving water bodies.

This project would be required to implement treatment BMPs because the proposed improvements result in the creation or replacement of more than 1 acre of impervious area. The treatment BMP strategy for areas within Caltrans' right-of-way would comply with the Caltrans MS4 Permit and the *Memorandum of California Department of Transportation Post-Construction Stormwater and Hydromodification Standards* (SFRWQCB 2008).

Permanent erosion control measures would be applied to all exposed areas once grading or soil disturbance work is completed, as a permanent measure to achieve final slope stabilization. These measures may include hydraulically applying a combination of hydroseed with native seed mix, hydro-mulch, straw, tackifier, and compost to promote vegetation establishment, and installing fiber rolls to prevent sheet flow from concentrating and causing gullies. For steeper slopes or areas that may be difficult for vegetation to establish, measures such as netting, blankets, or slope paving could be considered to provide stabilization. The project would consider bioretention swales and biofiltration strips within Caltrans' right-of way.

Given that the project area has shallow groundwater, other conventional treatment measures that capture and treat stormwater runoff may need to be considered; these devices could include basins or media filters. Due to the presence of environmentally sensitive areas and limited available treatment area for BMPs, the project may need to consider offsite stormwater treatment options. WQ-01: Offsite Stormwater Treatment would be implemented as an AMM since a deficit of approximately 25 to 40 acres requires off-site treatment. On-site BMPs may address some portion of this deficit in the final design, and preliminary discussions with the City of Vallejo and the City of Sonoma have identified potential off-site mitigation projects. The final drainage design, selection of treatment BMP types and locations, and determination of impervious area treated would be refined during the final design. Permanent project features would include measures to address permanent erosion control, drainage facilities, and onsite treatment.

# **Oil Grease and Chemical Pollutants**

Heavy metals associated with vehicle tire and break wear, oil and grease, and exhaust emissions are the primary pollutants associated with transportation corridors. Generally, roadway stormwater runoff has the following pollutants: total suspended solids, nitrate nitrogen, total Kjeldahl nitrogen, phosphorus, ortho-phosphate, copper, lead, and zinc. The pollutants are dispersed from tree leaves that have been exposed through aerial deposition, combustion products from fossil fuels, and the wearing of brake pads and tires. The project is expected to ease congestion leading to less deposition of particulates from exhaust and heavy metals and braking. the project would implement treatment BMPs to remove pollutants, including trash, mercury, and PCBs, from stormwater runoff before discharging into the receiving waters. The project would maintain the goal of treating new impervious services as summarized in Table 2-19, and with implementation of BMPs is anticipated to completely avoid impacts to water quality from oil, grease, and chemical pollutants.

# Trash

According to Caltrans District 4's Regional Board 2 Trash General Map Application (2020), the project contains a low trash density area highway and medium trash density ramps. Therefore, the project would be required to implement trash capture devices at the ramps. Additional trash device locations may be required due to the Section 401 Water Quality Certification criteria. The project environmental study is inclusive of trash capture locations; however, the final design, selection of trash capture devices and locations, and determination of impervious area treated would be refined during the selected alternative's final design.

# Aquifer Recharge and Groundwater

There are no aquifers within the project limits, and the majority of impervious fills would be placed in areas underlain by saturated soils. Permanent impacts to groundwater are not anticipated for all build alternatives.

# Temporary Impacts to Water Quality

Temporary impacts to water quality during construction are anticipated for all build alternatives. Soil disturbing activities, stockpiling, and unclean equipment during construction may cause sediment to runoff and enter storm drainage facilities or directly discharge into the receiving water bodies, increasing the turbidity, decreasing the clarity, and potentially impacting the beneficial uses of the receiving water bodies. Fueling or maintenance of construction vehicles may pose risk of accidental spills or releases of fuels, oils, or other potentially toxic materials that would threaten water quality if contaminants enter storm drains, open channels, or receiving water bodies.

Because the project involves more than 1 acre of DSA under all considered alternatives, it must comply with the SWRCB's Construction General Permit (NPDES No. CAS000002, SWRCB Order No. 2009-0009-DWQ as amended). The permit regulates stormwater discharges from construction sites that result in a DSA of 1 acre or greater, and/or are smaller sites that are part of a larger common plan of development. For all projects subject to the Construction General Permit, the applicant is required to hire a qualified SWPPP developer to develop and implement an effective SWPPP.

The project would require dewatering for excavation activities for work in Tolay Creek. Dewatering activities would comply with *Caltrans' Field Guide to Construction Site Dewatering* (2014a) and *Caltrans' Standard Specifications* (2020), and, if needed, a separate dewatering permit would be obtained prior to the start of construction. Dewatering may also be required along the sideslopes through the length of the corridor.

Temporary construction BMPs would be developed in the contractor's SWPPP that would meet Caltrans standard stormwater treatment requirements, MS4 requirements, and any additional requirements provided by regulatory agencies in issued permits to the project or as provided by the RWQCB. Temporary BMP measures during construction would address soil stabilization, sediment control, tracking control, non-stormwater management (e.g., dewatering operations), waste management, pollutions control, and job site management.

# **Operations and Maintenance Impacts**

The added impervious area summarized in Table 2-19 would have a minimal increase to stormwater pollution effects. Runoff from project activities would be treated with stormwater treatment facilities and diverted into modified drainage systems to maximize infiltration. Pollution and runoff sources are not expected to change. These impacts would be reduced through the implementation of stormwater treatment BMPs.

# 2.3.2.4 Avoidance, Minimization, and/or Mitigation Measures

Implementation of the following project features as described Table 1-4 in Section 1.5 are anticipated to avoid and minimize impacts to water quality from temporary dewatering and stormwater runoff:

- PF-WQ-01. Water Quality Best Management Practices;
- PF-WQ-02. Temporary Dewatering Activities;
- PF-WQ-03. Groundwater Treatment;
- PF-WQ-04. Inclement Weather Restriction

The following AMM would be implemented:

• WQ-01: Offsite Stormwater Treatment. Offsite treatment to address the site's limited onsite stormwater treatment capacity would be coordinated with appropriate mitigation project proponents and the RWQCB during the project's final design phase. The project would be programmed to meet the requirements of Caltrans' current MS4 and NPDES permits, (SWRCB 2013) following the guidelines and procedures outlined in Caltrans' latest Statewide SWMP to address stormwater runoff; and in accordance with *Memorandum of Caltrans Post-Construction Stormwater and Hydromodification Standards* (SFRWQCB 2008).

# 2.3.3 Geology/Soils/Seismic/Topography

# 2.3.3.1 Regulatory Setting

For geologic and topographic features, the key federal law is the Historic Sites Act of 1935, which establishes a national registry of natural landmarks and protects "outstanding examples of major geological features." Topographic and geologic features are also protected under CEQA.

This section also discusses geology, soils, and seismic concerns as they relate to public safety and project design. Earthquakes are prime considerations in the design and retrofit of structures. Structures are designed using the Department's Seismic Design Criteria (SDC). The SDC provides the minimum seismic requirements for highway bridges designed in California. A bridge's category and classification would determine its seismic performance level and which methods are used for estimating the seismic demands and structural capabilities. For more information, please see the Department's Division of Engineering Services, Office of Earthquake Engineering, Seismic Design Criteria at https://dot.ca.gov/programs/engineering-services.

# 2.3.3.2 Affected Environment

The information in this section is summarized from the *Preliminary Geotechnical Design Report* (AECOM 2021j).

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### Site Geology

The project limits consist of artificial fill (af), artificial levee fill (alf), early Pliocene and late Miocene Petaluma Formation (Tps), late Pleistocene alluvial fan deposits (Qpf), Holocene alluvial fan deposits (Qhf), and Holocene Bay Mud deposits (Qhbm) deposits. Figure 2-17 shows site geology along the project limits.

The bedrock northwest of the site near Sears Point is mapped as Cretaceous to Jurassic-age mélange and metagraywacke of the Franciscan Complex; serpentinite of the Great Valley Complex, overlain by late Miocene-age Donall Ranch mafic and rhyolitic volcanics; and early Pliocene and late Miocene-age claystone and mudstone of the Petaluma Formation, interfingered with Pliocene and late Miocene-age andesite and rhyolite flows of the Sonoma Volcanics. The bedrock north of the site near Highway 12 consists of the early Pleistocene and Pliocene volcanoclastic conglomerate of the Huichica Formation, which overlies Eocene-age Domengine sandstone. To the southeast on Mare Island and Vallejo, Late Cretaceous-age undivided sandstone, siltstone, and shale of the Great Valley Complex underlie the surficial soils (Graymer et al. 2002).

#### Seismic Hazards

The project limits are in a seismically active area and lie between known active and potentially active geologic faults. Hayward/Rodgers Creek, Green Valley/Concord, San Andreas, and Calaveras faults are all within 62 miles of the project limits. These faults are expected to represent the highest potential hazard to the project area due to their proximity and the probability that they would experience one or more earthquakes over a magnitude of 6.7 in the next 30 years. Other potentially active faults that could affect the project alignment include the West Napa fault, less than 4 miles to the east; Franklin fault, adjacent to the project alignment to the vest; and Lakeview fault, approximately 1 mile from the project alignment to the west. Additionally, recent studies suggest that the Rodgers Creek fault connects with the Hayward fault under San Pablo Bay, in which case the fault would pass through the western end of the project limits, near SR 121. However, the project alignment is not in the Alquist-Priolo Earthquake Fault Zone.

#### **Other Geological Hazards**

**Expansive Soils.** Near-surface fills are expected to have low expansion potential; the underlying native Bay Mud has a Plasticity Index of 50 or more and is expected to exhibit high shrink/swell behavior.

**Landslides and Erosion.** The project alignment is relatively flat, exclusive of the bridge approach embankment. The likelihood of landslides affecting the proposed project is considered low. Fills used to construct existing bridge approach embankments are generally described as clayey and would have low erosion potential, as would the underlying Bay Mud.

**Subsidence.** The ground surface along the project alignment is subject to long-term settlement due to consolidation of underlying soft Bay Mud. Areas directly below the roadway embankment have undergone long-term settlement.

**Liquefaction.** The project alignment is underlain by soils with low to moderate liquefaction susceptibility. The near-surface clayey sands that overlie the Bay Mud are present as thin and discontinuous lenses and would have low potential for liquefaction. The sands encountered below the Bay Mud are very dense and also not potentially liquefiable.

**Lateral Spreading.** The potential for lateral spreading is expected to be low, based on conditions revealed in historic borings.

# 2.3.3.3 Environmental Consequences

The following discussion pertains to both the Build and No Build Alternatives because seismic and geologic hazards on the project alignment are present under the existing condition and would be present under both the No Build and Build Alternatives. The proposed improvements would not increase existing seismic or other geological hazards.

# Seismic Hazards

The proposed project would not exacerbate the potential for seismic shaking; the intensity of the earthquake ground motion at the site would depend on the characteristics of the generating fault, the distance to the earthquake epicenter, the magnitude and duration of the earthquake, and specific site geologic conditions. Caltrans' design and construction guidelines incorporate engineering standards that address seismic risks, including ground failure related to liquefaction, landslides, and lateral spreading. Project elements would be designed and constructed to meet seismic design requirements for ground shaking and ground motions, as determined for the project vicinity and site conditions. Caltrans also requires that additional geotechnical subsurface and design investigations be performed during the final project design and engineering phase. These standards and requirements would avoid the potential for adverse impacts related to seismic activity. Furthermore, Caltrans would implement standard measures described in Table 1-4. Therefore, the project would have no impact.

# **Other Geological Hazards**

During construction of the project, earthmoving activities such as grading, excavation, and trenching have the potential to result in soil erosion and loss of topsoil, especially in areas where there are steeper slopes. Because the project alignment is relatively flat and a large portion of the project lies on artificial fill (clayey soils with a low erosion potential), there would be lower potential for substantial soil erosion to occur. The embankment of the bridge has greater slopes but is on soils with a low erosion potential. Nonetheless, BMPs would be implemented to reduce erosional impacts during construction activities, such as stabilization by paving, rock slope protection, and erosion control.

The project alignment is relatively flat, exclusive of the bridge approach embankment. The likelihood that landslides would affect the proposed project is considered low. The potential for lateral spreading is expected to be low, based on conditions revealed in historic borings. The project alignment is underlain by soils with low to moderate liquefaction susceptibility. Subsidence and settlement have the potential to occur, but the project would be designed to account for this settlement. Additionally, Caltrans' design and construction guidelines incorporate engineering standards that address expansive soils. Caltrans would implement project features described in Table 1-4, and no project related effects on geological resources are anticipated.

# 2.3.3.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-GEO-01. Geotech Investigations;
- PF-GEO-02. Seismic Standards; and
- PF-GEO-03. Embankment Design.

# 2.3.4 Paleontology

# 2.3.4.1 Regulatory Setting

Paleontology is a natural science focused on the study of ancient animal and plant life as it is preserved in the geologic record as fossils. Several federal statutes specifically address paleontological resources, their treatment, and funding for mitigation as a part of federally authorized projects.

16 USC 431-433 (the "Antiquities Act") prohibits appropriating, excavating, injuring, or destroying any object of antiquity situated on federal land without the permission of the Secretary of the Department of Government having jurisdiction over the land. Fossils are considered "objects of antiquity" by the Bureau of Land Management, the National Park Service, the Forest Service, and other federal agencies.

16 USC 470aaa (the Paleontological Resources Preservation Act) prohibits the excavation, removal, or damage of any paleontological resources located on federal land under the jurisdiction of the Secretaries of the Interior or Agriculture without first obtaining an appropriate permit. The statute establishes criminal and civil penalties for fossil theft and vandalism on federal lands.

23 USC 1.9(a) requires that the use of Federal-aid funds must be in conformity with all federal and state laws.

Under California law, paleontological resources are protected by CEQA.

# 2.3.4.2 Affected Environment

The information in this section is summarized from the *Paleontological Identification Report/Paleontological Evaluation Report* (PIR/PER) prepared in October 2020 (AECOM 2020a). The area studied for paleontological resources was nearly the entirety of Caltrans right-of-way and extends from approximately Lakeville Highway to the west, the Sacramento Street Overcrossing in Vallejo to the east, and approximately 0.2 mile north on SR 121 in Sonoma County. A project staging area is proposed on the northern side of SR 37, east of SR 121, in a portion of APN 068-190- 017. This PIR/PER presents the results of identification efforts, consistent with Volume 1, Chapter 8, of the Caltrans Standard Environmental Reference (Caltrans 2020b), and guidelines from the Society of Vertebrate Paleontology (1996, 2010).

Figure 2-17 in Section 2.3.2.2 show site geology along the project limits, which contain artificial fill (af), artificial levee fill (alf), early Pliocene and late Miocene Petaluma Formation (Tps), late Pleistocene alluvial fan deposits (Qpf), Holocene alluvial fan deposits (Qhf) and Holocene Bay Mud deposits (Qhbm).

A paleontological locality search of the University of California Museum of Paleontology (UCMP) records indicated that no paleontological resources have previously been recorded within the project limits. However, the UCMP listed three fossil localities in the vicinity of the project limits. These fossil localities include:

- UCMP locality V5231 is approximately 2 miles north of the western end of the PDA and was reported to contain a single horse tooth of likely Pliocene age from the Petaluma Formation.
- UCMP locality IP8549 (formerly United Sates Geological Survey [USGS] Cenozoic locality M4234) is about 0.5 mile west of Sears Point. The UCMP online database did not have any specimens currently cataloged at this location but noted it contained invertebrates. It was noted that mollusks were accepted from this locality from the USGS in Menlo Park.
- UCMP locality D7294 was the site of the excavation of a canal for the Leslie Salt work ponds along SR 37 west of Vallejo in 1975. The UCMP online database contained 29 listings of Pleistocene age invertebrate (mollusks) from this location.

A field study was conducted on May 23, 2020. Limitations related to parking and locked gates prevented surveyors from surveying the surrounding lands. No fossils were observed during the field study.

# 2.3.4.3 Environmental Consequences

Impacts to paleontological resources are rated in accordance with the sensitivity ratings of the geological rock units impacted. Below is a summary of these ratings for the geological rock units that occur within the project limits:

# High Potential Sensitivity (Direct impacts to high sensitivity rock units)

- Late Pleistocene Alluvial fan deposits (Qpf)
- Early Pliocene and late Miocene Petaluma Formation (Tps)

# No Potential (Direct impacts to zero sensitivity rock units)

- Artificial fill (af)
- Artificial levee fill (alf)
- Holocene Alluvial fan deposits (Qhf)
- Holocene Bay Mud deposits (Qhbm)

No paleontological resources have previously been recorded within the project limits and none were found during the field study. However, there is a potential to encounter unknown paleontological resources during project construction, given that there were three fossil localities in the vicinity of the project limits and there are areas within the project limits that have high potential sensitivity.

Direct impacts to paleontological resources could occur during project construction involving earthmoving activities such as grading, excavation, and boring. Direct impacts are the destruction of the fossil remains and the geographic, geologic, phylogenetic, and taphonomic information associated with them. There is greater potential for direct impacts to occur in the two high geological rock units, Qpf and Tps. These areas are at the western portion of the project limits. It is highly unlikely that any impacts would occur in geological rock units af, alf, Qhf, and Qhbm, which make most of the project limits. PF-PAL-01 and PF-PAL-02 described in Table 1-4 in Section 1.5 would be implemented to avoid any impacts to paleontological resources during construction activities. Therefore, no effect is expected to occur.

# 2.3.4.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-PAL-01. Discovery of Paleontological Resources
- PF-PAL-02. During design phase, Caltrans would determine whether a Paleontology Mitigation Plan is needed.

# 2.3.5 Hazardous Waste/Materials

# 2.3.5.1 Regulatory Setting

Hazardous materials, including hazardous substances and wastes, are regulated by many state and federal laws. Statutes govern the generation, treatment, storage and disposal of hazardous materials, substances, and waste, and also the investigation and mitigation of waste releases, air and water quality, human health, and land use.

The primary federal laws regulating hazardous wastes/materials are the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA)

of 1980, and the Resource Conservation and Recovery Act (RCRA) of 1976. The purpose of CERCLA, often referred to as "Superfund," is to identify and cleanup abandoned contaminated sites so that public health and welfare are not compromised. The RCRA provides for "cradle to grave" regulation of hazardous waste generated by operating entities. Other federal laws include:

- Community Environmental Response Facilitation Act of 1992
- CWA
- Clean Air Act
- Safe Drinking Water Act
- Occupational Safety and Health Act (OSHA)
- Atomic Energy Act
- Toxic Substances Control Act
- Federal Insecticide, Fungicide, and Rodenticide Act

In addition to the acts listed above, EO 12088, Federal Compliance with Pollution Control Standards, mandates that necessary actions be taken to prevent and control environmental pollution when federal activities or federal facilities are involved.

California regulates hazardous materials, waste, and substances under the authority of the California State Health and Safety Code (available at https://leginfo.legislature.ca. gov/faces/codesTOCSelected.xhtml?tocCode=HSC&tocTitle=+Health+and+Safety +Code+-+HSC) and is also authorized by the federal government to implement RCRA in the state. California law also addresses specific handling, storage, transportation, disposal, treatment, reduction, cleanup, and emergency planning of hazardous waste. The Porter-Cologne Water Quality Control Act also restricts disposal of wastes and requires cleanup of wastes that are below hazardous waste concentrations but could impact ground and surface water quality. California regulations that address waste management and prevention and cleanup of contamination include Title 22 Division 4.5 Environmental Health Standards for the Management of Hazardous Waste, Title 23 Waters, and Title 27 Environmental Protection.

Worker and public health and safety are key issues when addressing hazardous materials that may affect human health and the environment. Proper management and disposal of hazardous material is vital if it is found, disturbed, or generated during project construction.

# 2.3.5.2 Affected Environment

Information in this section is summarized from the Initial Site Assessment (ISA) prepared on July 30, 2020, for this project (AECOM 2020b). The purpose of the ISA was to identify potential hazardous materials in soil, groundwater, and/or building materials that could be disturbed during project construction and maintenance activities. The assessment included a review of environmental records and investigations of hazardous materials release sites within 0.5 mile of the project limits.

# Hazardous Materials Sites and/or Investigations

A review of the Department of Toxic Substances Control's (DTSC's) EnviroStor database included four state response/voluntary cleanup sites within 0.5 mile of the project limits. All four of the cases were completed, and no further action is required. A review of the SWRCB's GeoTracker database contained 15 potential contaminated sites and/or investigated sites within a 0.5-mile radius of the project limits. All but three of the sites were at Mare Island, Skaggs Island, and Tubbs Island, which were previously used for military activities. Thirteen of these sites are listed as closed cases, which indicates that hazardous materials have been addressed to the satisfaction of the applicable regulatory authority, and a formal closure decision document has been issued.

Of the 19 sites listed in the regulatory databases, one was determined to have a greater potential to impact the proposed project: the former Tubbs Island Gunnery Range on the Coast of San Pablo Bay in Petaluma. The former gunnery range historical use may present hazards in soil within the project limits, as described below in the Recognized Environmental Conditions section.

# **Recognized Environmental Conditions**

A Recognized Environmental Condition (REC) is defined by the presence or likely presence of any hazardous substances or petroleum products in, on, or at a property (1) due to any release to the environment; (2) under conditions indicative of a release to the environment; or (3) under conditions that pose a material threat of a future release to the environment.

Two RECs have been identified within 0.5 mile of the project area and are described below.

# *Tubbs Island Gunnery Range, Southern Tip of Sonoma County, on the Coast of San Pablo Bay, Petaluma (Approximately 0.5 Mile South of Project Limits, 358 Acres of Land)*

The former Tubbs Island Gunnery Range site is a flat reclaimed tidal marshland situated approximately 1 foot below sea level. According to USACE, the United States Army built the Tubbs Island Gunnery Range to support gunnery training at Hamilton Army Airfield in 1941. The range was used as a combination of an air-to-ground strafing range using machine guns, cannons, practice bombs, and sub-caliber aircraft rockets. It is suspected that around 1945 chemical spray missions were conducted. The range was closed in October 1945. The land is currently used for farming wheat, oat, and hay (Parsons 2008). No evidence of hazardous, toxic, or radiological waste was found during site inspections by USACE personnel. In March 2013, the DTSC concurred with a finding of no Department of Defense actions indicated for formerly used defense sites in California (DTSC 2013). Although no evidence of hazardous materials were discovered in the 2012 investigation, this investigation was only conducted on the southern part of the island and not within the project limits. Based on the proximity, historical site use, and a suggestion by a DTSC case worker noted in a work plan, the presence of UXO, mustard gas, or other military hazards cannot be ruled out.

#### Sears Point Farming Company, 5400 Sears Point Road, Sonoma, California (north of SR 37 at Sonoma County mile marker 5.4); Paradise Vineyards, 3000 Sears Point Road, Sonoma, California (south of SR 37)

Paradise Vineyards and Sears Point Farming Company have used lands adjacent to the project area for farming. Agricultural chemicals are presumed to have been used for farming processes and may be found in soils within the project limits.

# **Historical Recognized Environmental Conditions**

Historical Recognized Environmental Conditions (HRECs) are a past release of any hazardous substances or petroleum products that occurred in connection with the property and has been addressed to the satisfaction of the applicable regulatory authority, or meets unrestricted use criteria established by a regulatory authority, without subjecting the property to any required controls.

Three HRECs have been identified within 0.5 mile of the project area and are described below.

# Bondurant School of Driving, 29355 Arnold Drive, Sonoma (Approximately 0.4 mile North of Project Limits)

Potential contaminants of concern included waste oil, motor oil, hydraulic oil, and lubricating media in soil. Because it is not known whether contaminants have been removed from this site, it is not known whether contamination remains in the ground and has migrated to the project area. Although detailed information is missing regarding the scope, it can be assumed that closure was granted to this site by the RWQCB based on an in-depth investigation.

# *Riverview Estates Wilson & Lighthouse, Vallejo (Approximately 0.5 Mile East of Project Limits)*

On September 18, 2000, a heating oil/Bunker C underground storage tank (UST) was removed. Soil impacts were reported, and soil was excavated and transported off site for disposal. The case was granted closure by the RWQCB in April of 2002 (Solano County Environmental Health Services 2002).

# Skaggs Island Naval Security Group Activity (NSGA), 8 Miles Northwest of Vallejo (Approximately 0.8 Mile North of Project Limits)

NSGA Skaggs Island operated between 1942 and 1993. Its primary mission was operating and maintaining a system of high frequency, direction-finding antenna; providing communication computer support; and training the US Navy and other parts of the Department of Defense. Skaggs Island comprises 4,390 acres, and the US Navy owned approximately 3,310 of those acres. A series of environmental studies and remedial efforts have been previously conducted at Skaggs Island. Investigations have included the former pistol range, landfills, gas stations, antenna areas, and sludge-drying beds. In October of 2010, the RWQCB issued case closure on all associated cases with this former military facility. In February 2011 the DTSC concurred with a No Further Action determination, with no restrictions and approval the land for use as a wetland. The land was transferred to the Refuge.

# **Controlled Recognized Environmental Conditions**

Controlled Recognized Environmental Conditions (CRECs) are RECs resulting from a past release of hazardous substances or petroleum products that has been addressed to the satisfaction of the applicable regulatory authority, with hazardous substances or petroleum products allowed to remain in place subject to the implementation of required controls.

Four CRECs have been identified within 0.5 mile of the project limits and are described below.

#### Mare Island Naval Shipyard (Vallejo) – Mare Island Naval Shipyard USTs 993-1, 993-2 and 993-3, Mare Island Naval Shipyard, Vallejo, Solano County, California (Approximately 0.5 South of Project Limits)

In June 1999, the RWQCB issued a case closure letter for three 12,000-gallon gasoline USTs formerly located at the facility address above, with the stipulation that if a change in land use is proposed, the owner must notify the RWQCB.

#### Mare Island Naval Shipyard (Vallejo) – Mare Island Naval Shipyard UST 993-4, Mare Island Naval Shipyard, Vallejo, Solano County, California (Approximately 0.5 Mile South of Project Limits at Solano County Mile Marker 7.2)

Building 993 was the former base fuel station between Walnut Avenue and Railroad Avenue. Four USTs were historically used at Building 993. USTs 993-1, 993-2, and 993-3 received closure status on January 9, 1999. UST 993-4 was a 500-gallon waste oil tank removed on July 18, 1990. The Navy removed contaminated soil outside of the existing building and conducted soil and groundwater investigation inside and around the building. In 2016, the RWQCB issued no further action, but the site has residual petroleum hydrocarbons at concentrations greater than the unrestricted standards for groundwater. The site can only be used for commercial/industrial use.

#### Mare Island Naval Shipyard (Vallejo) – Mare Island Naval Shipyard 577S, Railroad Ave, Mare Island, CA 94592, Solano County, Military UST Site (Approximately 0.5 Mile South of Project Limits at Solano County Mile Marker 7.3)

A 2,000-gallon waste oil tank and a 500-gallon heating oil tank were removed from this facility sometime in the late 1990s. A site investigation was completed in 1997. Based on the current land use, the RWQCB issued two no further action letters for the locations of the former two tanks. If there is a change in land use, the owner must notify the RWQCB.

#### Mare Island Lennar Investigation Area A3, 900 Walnut Avenue, Quarters D, Vallejo, CA 94592, (Approximately 0.4 Mile South of Project Limits at Sonoma County Mile Marker 6.9, 50 Acres of Land)

Investigation Area A3 was used as a small arms firing range from approximately 1917 to 1940. Previous investigations in the area have indicated PCBs contamination in soils. Some remediation efforts have been completed in the area. Two land use covenants were executed and recorded with Solano County in 2003 and 2004. A

covenant to restrict use of property was executed between Lennar Mare Island, LLC and DTSC in 2003. The covenant restricts the use of Investigation Area A3 to commercial/industrial activities and prohibits the building of residences, hospitals, daycares, or schools for persons under 18 years old. A subsequent covenant to restrict use of property was executed between Qualified Investment Opportunities, Inc. and DTSC in 2004. This covenant prohibits the use of the land for residences, hospitals for humans, public or private schools for person under 18 years old, or daycares for children.

# **Additional Finding**

Although reports indicated that biosolids were being spread and tilled at the time of inspection, proper setbacks were maintained and no violations or areas of concern were observed. Tubbs Island Sludge Disposal at 5400 Sears Point in Sonoma affects or has the potential to affect water quality in the project limits.

#### **Natural Occurring Arsenic**

Based on previous studies indicting elevated concentrations of naturally occurring arsenic in the Bay Area, naturally occurring arsenic is anticipated to be present in soils in the project footprint.

# **Asbestos-Containing Materials**

Asbestos-containing material (ACM) is commonly found in many man-made products. In the State Highway System, ACM is commonly found in the shims used under aluminum bridge barrier rails and in asbestos-containing cement pipe, and has been found, on rare occasions, in Portland cement concrete. Therefore, ACM might be present in the project footprint.

# Aerially Deposited Lead

Leaded gasoline was used from early 1920s through the 1970s. Because SR 37 has been in operation since the use of lead in gasoline, it is anticipated that aerially deposited lead (ADL) would be observed in soils next to SR 37.

#### Lead-Containing Paint

Lead and other potentially toxic substances could be present in traffic stripes, pavement markings, and paint used on concrete and steel bridges.

#### **Treated-Wood Waste**

The preservatives used to treat the wood can include one or more of the following constituents: arsenic, chromium, copper, pentachlorophenol, or creosote. When the treated wood has reached the end of its service life, it is regarded as treated-wood waste.

# 2.3.5.3 Environmental Consequences

#### No Build Alternative

The No Build Alternative would not affect potential hazardous material sites in the project area.

#### Build Alternatives 1, 2, 3A, and 3B

#### Short-Term Construction Impacts

Project construction activities are expected to involve the temporary transport, use, and disposal of typical construction hazardous materials (e.g., fuels, paints, asphalt, and lubricants). These materials could pose a threat to human health or the environment if not properly managed.

Standardized measures, as required by state and federal regulations, would be implemented during project construction. Hazardous materials must be transported in accordance with RCRA and USDOT regulations and disposed of in accordance with RCRA and the California Code of Regulations at a facility that is permitted to accept the waste. Workers who handle hazardous materials are required to adhere to OSHA and Cal/OSHA health and safety requirements.

In accordance with the SWRCB, a SWPPP must be prepared and implemented during construction for coverage under the Construction General Permit. The SWPPP requires implementation of BMPs for hazardous materials storage and soil stockpiles, inspections, maintenance, training of employees, and containment of releases to prevent runoff into existing stormwater collection systems or waterways. In addition, BMPs would be incorporated, such as performing fueling and maintenance operations of vehicles and equipment at least 50 feet away from watercourses.

ADL from the historical use of leaded gasoline, exists along roadways throughout California. There is the likely presence of soils with elevated concentrations of lead as a result of ADL on the State Highway System right-of-way within the limits of the project alternatives. Soil determined to contain lead concentrations exceeding stipulated thresholds must be managed under the July 1, 2016, ADL Agreement between Caltrans and the California DTSC. This ADL Agreement allows such soils to be safely reused within the project limits as long as all requirements of the ADL Agreement are met.

Adherence to federal and state regulations during project construction reduces the risk of exposure to hazardous materials and accidental hazardous materials releases. Compliance with existing regulations is mandatory. Therefore, construction of the proposed project is not expected to create a hazard to construction workers, the public, or the environment through the routine transport, use, disposal, or accidental release of hazardous materials.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### Disturbance of Hazardous Materials

Construction of the project could result in the potential disturbance of hazardous materials in the soil and groundwater. If there is contamination in the project area, ground-disturbing activities during construction, such as drilling, excavation, grading, and trenching, could potentially expose construction workers and the public to hazardous conditions. Grading activities could also result in accidental mobilization of contaminants from the soil to groundwater or air. As described in the Affected Environment, there is a potential for natural occurring arsenic, ACM, ADL, and treated-wood waste to be encountered during soil excavation activities.

A review of regulatory databases Geotracker and EnviroStor identified sites that were investigated for chemical releases. Seventeen of the 19 sites have been addressed to the satisfaction of the applicable regulatory authority. Therefore, the majority of these sites are unlikely to result in impacts during construction and maintenance activities for the proposed project. However, hazardous material at the former Tubbs Island Gunnery Range has the potential to have impacts.

To avoid impacts related to disturbing possible contamination in the soil and groundwater, Caltrans would implement project features, as described in Table 1-4. With the incorporation of these measures, impacts from hazardous materials are not expected to be substantial.

#### Long-Term Operation Impacts

Following construction, no long-term impacts are expected to occur related to hazardous waste and materials. Maintenance work would be required periodically over the life of the of SR 37 and may require the use of hazardous materials. However, with adherence to federal and state regulations regarding the use of hazardous material there would be no long-term impacts.

# 2.3.5.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-HAZ-01. Site Investigations;
- PF-HAZ-02. Health and Safety Management Plan;
- PF-HAZ-03. Soil and Groundwater Management Plan; and
- PF-HAZ-04. Hazardous Structure Material Surveys.

# 2.3.6 Air Quality

# 2.3.6.1 Regulatory Setting

The Federal Clean Air Act (FCAA), as amended, is the primary federal law that governs air quality while the California Clean Air Act is its companion state law. These laws, and related regulations by the U.S. EPA and CARB, set standards for the concentration of pollutants in the air. At the federal level, these standards are called

National Ambient Air Quality Standards (NAAQS). NAAQS and state ambient air quality standards have been established for six criteria pollutants that have been linked to potential health concerns: carbon monoxide (CO), nitrogen dioxide (NO<sub>2</sub>), ozone (O<sub>3</sub>), particulate matter (PM)—which is broken down for regulatory purposes into particles of 10 micrometers or smaller (PM<sub>10</sub>) and particles of 2.5 micrometers and smaller (PM<sub>2.5</sub>), lead, and sulfur dioxide (SO<sub>2</sub>). In addition, state standards exist for visibility reducing particles, sulfates, hydrogen sulfide, and vinyl chloride. The NAAQS and state standards are set at levels that protect public health with a margin of safety and are subject to periodic review and revision. Both state and federal regulatory schemes also cover toxic air contaminants (air toxics); some criteria pollutants are also air toxics or may include certain air toxics in their general definition.

Federal air quality standards and regulations provide the basic scheme for projectlevel air quality analysis under NEPA. In addition to this environmental analysis, a parallel "Conformity" requirement under the FCAA also applies.

# Conformity

The conformity requirement is based on FCAA Section 176(c), which prohibits the USDOT and other federal agencies from funding, authorizing, or approving plans, programs, or projects that do not conform to SIP for attaining the NAAQS. "Transportation Conformity" applies to highway and transit projects and takes place on two levels: the regional (or planning and programming) level and the project level. The proposed project must conform at both levels to be approved.

Conformity requirements apply only in nonattainment and "maintenance" (former nonattainment) areas for the NAAQS, and only for the specific NAAQS that are or were violated. U.S. EPA regulations at 40 CFR 93 govern the conformity process. Conformity requirements do not apply in unclassifiable/attainment areas for NAAQS and do not apply at all for state standards regardless of the status of the area.

Regional conformity is concerned with how well the regional transportation system supports plans for attaining the NAAQS for CO, NO<sub>2</sub>, O<sub>3</sub>, particulate matter (PM<sub>10</sub> and PM<sub>2.5</sub>), and in some areas (although not in California), SO<sub>2</sub>. California has nonattainment or maintenance areas for all of these transportation-related "criteria pollutants" except SO<sub>2</sub>, and also has a nonattainment area for lead; however, lead is not currently required by the FCAA to be covered in transportation conformity analysis. Regional conformity is based on emission analysis of RTPs and Federal Transportation Improvement Programs (FTIPs) that include all transportation projects planned for a region over a period of at least 20 years (for the RTP) and 4 years (for the FTIP).

RTP and FTIP conformity uses travel demand and emission models to determine whether or not the implementation of those projects would conform to emission budgets or other tests at various analysis years showing that requirements of the FCAA and the SIP are met. If the conformity analysis is successful, the Metropolitan Planning Organization (MPO), FHWA, and Federal Transit Administration make the determinations that the RTP and FTIP are in conformity with the SIP for achieving the goals of the FCAA. Otherwise, the projects in the RTP and/or FTIP must be modified until conformity is attained. If the design concept and scope and the "open-to-traffic" schedule of a proposed transportation project are the same as described in the RTP and FTIP, then the proposed project meets regional conformity requirements for purposes of project-level analysis.

Project-level conformity is achieved by demonstrating that the project comes from a conforming RTP and TIP; the project has a design concept and scope that has not changed significantly from those in the RTP and TIP; project analyses have used the latest planning assumptions and U.S. EPA-approved emissions models; and in PM areas, the project complies with any control measures in the SIP. Furthermore, additional analyses (known as hot-spot analyses) may be required for projects in CO and PM nonattainment or maintenance areas to examine localized air quality impacts.

# 2.3.6.2 Affected Environment

The analysis summarized in this section is from the *Air Quality Report* prepared in September 2021 (AECOM 2021g). The project area is in the San Francisco Bay Area Air Basin (SFBAAB). Air quality regulation in SFBAAB is administered by the Bay Area Air Quality Management District (BAAQMD).

# Climate, Meteorology, and Topography

Weather and terrain can influence air quality. Certain weather parameters are highly correlated to air quality, including temperature, the amount of sunlight, and the type of winds at the surface and above the surface. Winds can transport O3 and O3 precursors (i.e., reactive organic gases [ROG]) from one region to another, contributing to air quality problems downwind of source regions. Furthermore, mountains can act as a barrier that prevents O3 from dispersing.

The Gnoss Field climatological station near Novato, California, is maintained by the National Weather Service and is the nearest station at approximately 6 miles from the western end of the project site. The climate of the project area is generally Mediterranean in character, with cool winters (average 24-hour temperature of 50 degrees Fahrenheit in January) and warm, dry summers (average 24-hour temperature of 64 degrees Fahrenheit in July). SR 37 traverses one of the Bay Area's largest remaining tidal marsh environments, known as the San Pablo Bay lands. As a result, San Francisco Bay and the coastal mountains have a significant influence on the climate of the project area. Annual average rainfall is 23.6 inches (at Gnoss Field), mainly falling during the winter months.

# **Regional Air Quality Conformity**

State and federal government set standards for criteria air pollutants, including O<sub>3</sub>; CO; PM<sub>10</sub>; PM<sub>2.5</sub>; NO<sub>2</sub>; SO<sub>2</sub>; and lead. The BAAQMD monitors these pollutants of concern and air quality conditions throughout the SFBAAB. Under current designations of the Air Basin, the area is in nonattainment for California Ambient Air Quality Standards (CAAQS) for O<sub>3</sub>, PM<sub>10</sub>, and PM<sub>2.5</sub>, and NAAQS nonattainment for O<sub>3</sub> and PM<sub>2.5</sub>. The area is in unclassifiable/attainment for PM<sub>10</sub>, NO<sub>2</sub> and SO<sub>2</sub>. Table 2-20 shows the state and federal attainment status for each pollutant.

Pollutant	Averaging Time	State Standard <sup>1</sup>	Federal Standard <sup>2</sup>	State Project Attainment Status	Federal Project Area Attainment Status
O <sub>3</sub> <sup>3</sup>	1 hour	0.09 ppm (parts per million)	N/A	N	N/A
O <sub>3</sub>	8 hours	0.070 ppm 0.070 ppm (4th highest 3 years)		Ν	N (Marginal)
CO <sup>4</sup>	1 hour	20 ppm	35 ppm	А	A
CO	8 hours	9.0 ppm	9 ppm	А	A
PM10 <sup>5</sup>	24 hours	50 micrograms per cubic meter (μg/m³)	150 μg/m <sup>3</sup> (expected number of days above standard < or equal to 1)	N	U
PM <sub>10</sub>	Annual	20 µg/m³	20 µg/m <sup>3</sup> N/A		N/A
PM <sub>2.5</sub> <sup>6</sup>	24 hours	N/A	35 µg/m <sup>3 6</sup>	N/A	N
PM <sub>2.5</sub>	Annual	12 µg/m³	12.0 µg/m³	N	U, A
NO <sub>2</sub>	1 hour	0.18 ppm	0.100 ppm <sup>7</sup>	А	U
NO <sub>2</sub>	Annual	0.030 ppm	0.053 ppm	N/A	А
SO2 8	1 hour	0.25 ppm	0.075 ppm (99th percentile over 3 years)	А	N/A
SO <sub>2</sub>	3 hours	N/A	0.5 ppm <sup>9</sup>	N/A	N/A
SO <sub>2</sub>	24 hours	0.04 ppm	0.14 ppm (for certain areas)	А	U
SO <sub>2</sub>	Annual	N/A	0.030 ppm (for certain areas)	N/A	U
Pb <sup>10</sup>	Monthly	1.5 µg/m³	N/A	N/A	А
Pb	Calendar Quarter	N/A	1.5 μg/m <sup>3</sup> (for certain areas)	N/A	A
Pb	Rolling 3-month average	N/A	0.15 µg/m <sup>3 11</sup>	N/A	N/A
Sulfates	24 hours	25 µg/m³	N/A	А	N/A
H <sub>2</sub> S	1 hour	0.03 ppm	N/A	U	N/A
Visibility Reducing Particles (VRP) <sup>12</sup>	8 hours	Visibility of 10 miles or more (Tahoe: 30 miles) at relative humidity less than 70 %	N/A	U	N/A

Table 2-20 State and Federal Attainment Stati	e 2-20	e and Federal Attainment St	atus
---	--------	-----------------------------	------

Pollutant	Averaging Time	State Standard <sup>1</sup>	Federal Standard <sup>2</sup>	State Project Attainment Status	Federal Project Area Attainment Status
Vinyl Chloride <sup>12</sup>	24 hours	0.01 ppm	N/A	N/A	N/A

Source: CARB Air Quality Standards chart accessed June 29, 2021. Greenhouse gases (GHGs) do not have concentration standards; conformity requirements do not apply to GHGs and therefore are not listed.

Notes:

A = Attainment

N = Nonattainment

U = Unclassified

N/A = Not Applicable or No Information

- <sup>1</sup> California standards for O<sub>3</sub>, CO (except 8-hour Lake Tahoe), SO<sub>2</sub> (1 and 24 hour), NO<sub>2</sub>, and particulate matter (PM<sub>10</sub>, PM<sub>2.5</sub>, and visibility reducing particles), are values that are not to be exceeded. All others are not to be equaled or exceeded. CAAQS are listed in the Table of Standards in Section 70200 of Title 17 of the California Code of Regulations.
- $^2$  Federal standards (other than O<sub>3</sub>, PM, and those based on annual arithmetic mean) are not to be exceeded more than once a year. The O<sub>3</sub> standard is attained when the fourth highest 8-hour concentration measured at each site in a year, averaged over three years, is equal to or less than the standard. For PM<sub>10</sub>, the 24-hour standard is attained when the expected number of days per calendar year with a 24-hour average concentration above 150  $\mu$ g/m<sup>3</sup> is equal to or less than one. For PM<sub>2.5</sub>, the 24-hour standard is attained when 98 percent of the daily concentrations, averaged over three years, are equal to or less than the standard. Contact the U.S. EPA for further clarification and current national policies.
- <sup>3</sup> On October 1, 2015, the national 8-hour O<sub>3</sub> primary and secondary standards were lowered from 0.075 to 0.070 ppm. Transportation conformity applies in newly designated nonattainment areas for the 2015 national 8-hour O<sub>3</sub> primary and secondary standards on and after August 4th, 2019 (see Transportation Conformity Guidance for 2015 O<sub>3</sub> NAAQS Nonattainment Areas).
- <sup>4</sup> Transportation conformity requirements for CO no longer apply after June 1, 2018 for the following California CO Maintenance Areas (see U.S. EPA CO Maintenance Letter).
- <sup>5</sup> On December 14, 2012, the national annual PM<sub>2.5</sub> primary standard was lowered from 15 μg/m<sup>3</sup> to 12 μg/m<sup>3</sup>. The existing national 24-hour PM<sub>2.5</sub> standards (primary and secondary) were retained at 35 μg/m<sup>3</sup>, as was the annual secondary standard of 15 μg/m<sup>3</sup>. The existing 24-hour PM<sub>10</sub> standards (primary and secondary) of 150 μg/m<sup>3</sup> also were retained. The form of the annual primary and secondary standards is the annual mean, averaged over 3 years.
- <sup>6</sup> The 65 μg/m<sup>3</sup> PM<sub>2.5</sub> (24-hr.) NAAQS was not revoked when the 35 μg/m<sup>3</sup> NAAQS was promulgated in 2006. The 15 μg/m<sup>3</sup> annual PM<sub>2.5</sub> standard was not revoked when the 12 μg/m<sup>3</sup> standard was promulgated in 2012. Therefore, for areas designated nonattainment or nonattainment/maintenance for the 1997 and or 2006 PM<sub>2.5</sub> NAAQS, conformity requirements still apply until the NAAQS are fully revoked.
- <sup>7</sup> Final 1-hour NO<sub>2</sub> NAAQS published in the Federal Register on 2/9/2010, effective 3/9/2010. Initial area designation for California (2012) was attainment/unclassifiable throughout. Project-level hot spot analysis requirements do not currently exist. Near-road monitoring starting in 2013 may cause re-designation to nonattainment in some areas after 2016.
- <sup>8</sup> On June 2, 2010, a new 1-hour SO<sub>2</sub> standard was established and the existing 24-hour and annual primary standards were revoked. To attain the 1-hour national standard, the 3-year average of the annual 99th percentile of the 1-hour daily maximum concentrations at each site must not exceed 75 ppb. The 1971 SO<sub>2</sub> national standards (24-hour and annual) remain in effect until one year after an area is designated for the 2010 standard, except that in areas designated nonattainment for the 1971 standards, the 1971 standards remain in effect until implementation plans to attain or maintain the 2010 standards are approved.
- <sup>9</sup> Secondary standard, the levels of air quality necessary to protect the public welfare from any known or anticipated adverse effects of a pollutant rather than health. Conformity and environmental analysis address both primary and secondary NAAQS.
- <sup>10</sup> The CARB has identified vinyl chloride and the PM fraction of diesel exhaust as toxic air contaminants. Diesel exhaust PM is part of PM<sub>10</sub> and, in larger proportion, PM<sub>2.5</sub>. Both the CARB and U.S. EPA have identified lead and various organic compounds that are precursors to O<sub>3</sub> and PM<sub>2.5</sub> as toxic air contaminants. There are no exposure criteria for adverse health effect due to toxic air contaminants, and control requirements may apply at ambient concentrations below any criteria levels specified above for these pollutants or the general categories of pollutants to which they belong.
- <sup>11</sup> Lead NAAQS are not considered in Transportation Conformity analysis.
- <sup>12</sup> In 1989, the CARB converted both the general statewide 10-mile visibility standard and the Lake Tahoe 30-mile visibility standard to instrumental equivalents, which are "extinction of 0.23 per kilometer" and "extinction of 0.07 per kilometer" for the statewide and Lake Tahoe Air Basin standards, respectively.

# Local Ambient Air Quality

The BAAQMD operates a network of air monitoring sites throughout the SFBAAB. Air quality monitoring stations collect air quality data and monitor these pollutants of concern. The closest operating air quality monitoring site to the project is the Vallejo-Tuolumne Street site, approximately 2.8 miles southeast of SR 37 at 304 Tuolumne Street in Vallejo. However, the Station does not collect PM<sub>10</sub> data. Therefore, the next closest stations that do collect PM<sub>10</sub> data were selected. These stations are the North College Parking Station in Napa, California; and the 534 4th Street Station in San Rafael. Table 2-21 shows the air quality concentrations recorded at these three stations for the past 5 years, or 2015 through 2019.

# Mobile Source Air Toxics

In addition to the criteria air pollutants for which standards exist, U.S. EPA also regulates air toxics. Most air toxics originate from human-made sources, including onroad mobile sources. Mobile source air toxics (MSATs) are a subset of the air toxics defined by the Clean Air Act. Some toxic compounds are present in fuel and are emitted to the air when the fuel evaporates or passes through the engine unburned. Other toxics are emitted from the incomplete combustion of fuel or as secondary combustion products. Metal air toxics also result from engine wear or impurities in oil or gasoline.

Vehicles that travel along SR 37 are the largest source of MSATs affecting sensitive receptors in the project area. Vehicle traffic in the area is generated by commuters during the weekdays and by recreational travelers on the weekends.

# 2.3.6.3 Environmental Consequences

# No Build Alternative

The No Build Alternative would not change air quality in the project area.

# Build Alternatives 1, 2, 3A, and 3B

# Regional Air Quality Conformity

The project would remove an existing bottleneck between Mare Island and SR 121 by constructing either one HOV lane or two HOV/multipurpose lanes. Each of the Build Alternatives would improve operations, reduce congestion, and increase vehicle occupancy within the travel corridor. However, because the project would add lanes, this would be considered a capacity-increasing project. Therefore, the Build Alternatives would not be considered exempt from either regional or project-level conformity requirements. Consequently, the project requires a separate listing in the RTP (i.e., Plan Bay Area 2050), MTC's financially constrained 2021 TIP, and their associated regional emissions analyses to demonstrate regional conformity.

Pollutant	Monitoring Location	Criteria	Standard	2015	2016	2017	2018	2019
O <sub>3</sub>	Vallejo	Maximum 1-hour concentration	-	0.085 ppm	0.097 ppm	0.105 ppm	0.070 ppm	0.092 ppm
O <sub>3</sub>	Vallejo	Number of days exceeded: CAAQS	0.09 ppm	0	1	1	0	0
O <sub>3</sub>	Vallejo	Maximum 8-hour concentration	-	0.071 ppm	0.072 ppm	0.089 ppm	0.056 ppm	0.076 ppm
O <sub>3</sub>	Vallejo	Number of days exceeded: CAAQS	0.070 ppm	1	1	2	0	1
O3	Vallejo	NAAQS	0.070 ppm	0	1	2	0	1
со	Vallejo	Maximum 1-hour concentration	-	2.4 ppm	2.1 ppm	3.1 ppm	2.8 ppm	2.0 ppm
со	Vallejo	Number of days exceeded: CAAQS	20 ppm	0	0	0	0	0
со	Vallejo	NAAQS	35 ppm	0	0	0	0	0
со	Vallejo	Maximum 8-hour concentration	-	1.9 ppm	1.8 ppm	2.1 ppm	2.4 ppm	1.5 ppm

Table 2-21 Air Quality Concentrations for 2015-2019

Pollutant	Monitoring Location	Criteria	Standard	2015	2016	2017	2018	2019
со	Vallejo	Number of days exceeded: CAAQS	9.0 ppm	0	0	0	0	0
со	Vallejo	NAAQS	9 ppm	0	0	0	0	0
PM10	San Rafael/ Napa	Maximum 24-hour concentration	-	51.5 µg/m³	33.0 µg/m³	94.0 µg/m³	26 µg/m³	39 µg/m³
PM <sub>10</sub>	San Rafael/Napa	Number of days exceeded: CAAQS	50 µg/m³	6.1	0	2	0	0
PM <sub>10</sub>	San Rafael/Napa	NAAQS	150 µg/m³	0	0	0	0	0
PM <sub>10</sub>	San Rafael/Napa	Maximum annual concentration	-	18.7 µg/m³	16.6 µg/m³	17.7 µg/m³	19.0 µg/m³	14.2 µg/m³
PM <sub>10</sub>	San Rafael/Napa	Standard exceeded: CAAQS	20 µg/m³	No	No	No	No	No
PM2.5	Vallejo	Maximum 24-hour concentration	-	41.8 µg/m <sup>3</sup>	23.0 µg/m <sup>3</sup>	101.9 µg/m <sup>3</sup>	197.2 µg/m³	30.6 µg/m <sup>3</sup>
PM2.5	Vallejo	Number of days exceeded: NAAQS	35 µg/m³	3.0	0	9.3	16.4	0
Pollutant	Monitoring Location	Criteria	Standard	2015	2016	2017	2018	2019
-------------------	------------------------	--------------------------------------	------------	-----------	-----------	------------	------------	------------
PM <sub>2.5</sub>	Vallejo	Maximum annual concentration	-	9.6 µg/m³	7.3 µg/m³	11.6 µg/m³	13.3 µg/m³	8.8 µg/m³
PM <sub>2.5</sub>	Vallejo	Standard exceeded: CAAQS	12 µg/m³	No	No	No	Yes	No
PM <sub>2.5</sub>	Vallejo	NAAQS	12.0 µg/m³	No	No	No	Yes	No
NO <sub>2</sub>	Vallejo	Maximum 1-hour concentration	-	0.044 ppm	0.043 ppm	0.049 ppm	0.051 ppm	0.053 ppm
NO <sub>2</sub>	Vallejo	Number of days exceeded: CAAQS	0.18 ppm	0	0	0	0	0
NO <sub>2</sub>	Vallejo	NAAQS	0.10 ppm	0	0	0	0	0
NO <sub>2</sub>	Vallejo	Maximum annual concentration	-	0.008 ppm	0.007 ppm	0.008 ppm	0.008 ppm	0.0071 ppm
NO <sub>2</sub>	Vallejo	Number of days exceeded: CAAQS	0.030 ppm	N/A	N/A	N/A	N/A	N/A
NO <sub>2</sub>	Vallejo	NAAQS	0.053 ppm	N/A	N/A	N/A	N/A	N/A

Source: CARB 2020d; BAAQMD 2020

The proposed project is listed in the Plan Bay Area 2050 financially constrained RTP, which was found to conform by MTC on October 21, 2021. The FHWA and FTA has made a regional conformity determination finding on the RTP. The project is also included in the MTC financially constrained 2021 TIP (TIP ID VAR210004). MTC's 2021 TIP also received a conformity determination issued by the FHWA and FTA. The design concept and scope of the proposed project is consistent with the project description in the 2050 RTP, 2021 TIP, and the open to traffic assumptions of the MTC regional emissions analysis.

# Project-Level Conformity

The project is in an attainment/maintenance area for CO and a nonattainment area for PM<sub>2.5</sub>. Therefore, a project-level conformity analysis applies to the project for both pollutants under 40 CFR 93.109. However, current guidance from FHWA and Caltrans states that a project-level CO hot-spot analysis is no longer required to demonstrate project-level conformity, hot-spot analysis for PM<sub>2.5</sub> is only required for projects found to meet the definition of a POAQC by the MPO's Air Quality Conformity Task Force (AQCTF). The project was found not to be a POAQC by MTC's AQCTF at their May 27, 2021, consultation meeting. Therefore, a PM<sub>2.5</sub> hot-spot analysis is not required.

40 CFR 93.123(c)(5) states: "CO, PM<sub>10</sub>, and PM<sub>2.5</sub> hot-spot analyses are not required to consider construction-related activities which cause temporary increases in emissions. Each site which is affected by construction-related activities shall be considered separately, using established 'Guideline' methods. Temporary increases are defined as those which occur only during the construction phase and last 5 years or less at any individual site." Because construction of the project is expected to last less than 5 years, an evaluation of CO, PM<sub>10</sub>, and PM<sub>2.5</sub> emissions during project construction is not required for project-level conformity determination.

# Short-Term Construction Impacts

There would be temporary and short-term air quality impacts during construction activities, such as excavation, grading, hauling, cut and fill, paving, and removing and replacing pavement of existing roadways. These activities are expected to release particulate emissions such as airborne dust into the atmosphere. Emissions from construction equipment and on-road vehicles powered by gasoline and diesel engines are also anticipated and would include CO, oxides of nitrogen (NO<sub>X</sub>), ROG, directly emitted PM<sub>10</sub> and PM<sub>2.5</sub>, and toxic air contaminants such as diesel exhaust PM. These activities would also result in temporary VMT increases associated with worker and construction vehicle trips. Construction activities in the area may temporarily increase traffic congestion and slow the speed of traffic, resulting in a temporary increase in onroad emissions. These emissions would be limited to the immediate area impacted by construction-related traffic.

Construction activities would not last for more than 5 years at one general location, so construction-related emissions do not need to be included in the regional and project-level conformity analysis (40 CFR 93.123(c)(5)). Construction emissions were estimated using the latest Sacramento Metropolitan Air Quality Management District's Road Construction Model (RCEM) version 9.0. Table 2-22 and Table 2-23 shows the construction emissions associated with the project.

Stage/Emissions Rate	Phase/Activity	ROG <sup>*</sup> (pounds/ day)	CO (pounds/ day)	NO <sub>x</sub> (pounds/ day)	Exhaust PM <sub>10</sub> (pounds/ day)	Exhaust PM₂.₅ (pounds/ day)	CO <sub>2</sub> Equivalent (metric tons/ phase)
Roadway	Grubbing/Land Clearing	2.13	19.12	17.96	0.90	0.71	97.5
Roadway	Grading/ Excavation	5.66	51.79	56.45	2.43	2.07	987.8
Roadway	Drainage/ Utilities/ Sub-Grade	3.18	36.65	34.21	1.55	1.33	542.7
Roadway	Paving	2.30	25.80	20.86	1.06	0.83	173.8
Structures (Flyover Ramp, Overcrossing, etc.)	Grubbing/Land Clearing	0.94	10.25	9.94	0.45	0.38	28.0
Structures (Flyover Ramp, Overcrossing, etc.)	Grading/ Excavation	7.60	63.33	80.29	3.25	2.95	311.5
Structures (Flyover Ramp, Overcrossing, etc.)	Drainage/ Utilities/ Sub-Grade	5.37	46.20	56.09	2.27	2.07	217.1
Structures (Flyover Ramp, Overcrossing, etc.)	Paving	0.94	13.25	9.89	0.51	0.43	27.5
Average Workday Emissions (pounds/day)	(Based on 396 Workdays)	5.70	52.82	55.75	2.42	2.12	1,590 MT/ year

#### Table 2-23 Construction Emissions (Total Tons)

Stage/Emissions Rate	ROG <sup>*</sup> (tons)	CO (tons)	NO <sub>x</sub> (tons)	Exhaust PM₁₀ (tons)	Exhaust PM <sub>2.5</sub> (tons)	CO <sub>2</sub> Equivalent (metric tons)
Roadway Construction	0.82	7.79	7.82	0.35	0.30	1,802
Structures Construction	0.31	2.67	3.22	0.13	0.12	584
Total Construction	1.13	10.46	11.04	0.48	0.42	2,386

Notes:

\* ROG is reactive organic gases, which is a subset of total organic gases.

Source: Illingworth & Rodkin using RCEM version 9.0.0, 2021 and off model application of SAFE Rule adjustment factors.

Short-term air quality impacts would not be substantial and are expected to be localized around construction activities. AMMs described in Section 2.3.6.4 would be implemented during construction activities and would reduce or eliminate construction-related air quality impacts. Therefore, the impacts would be minimal.

## Long-Term Operational Impacts

Daily operational emissions estimates were developed using Caltrans' CT-EMFAC2017 emissions model, which is based on CARB's EMFAC2017 emissions model. Emissions were estimated for the baseline year (2019), opening year (2025), RTP horizon year (2040), and design year (2045) for the No Build and Build Alternatives. Overall, emissions in the future would decrease as older vehicles are replaced by newer vehicles with more stringent emissions and fuel economy standards. Based on the operational period emission data in Table 2-24, all of the Build Alternatives would have emissions similar to those for No-Build conditions for each study year.

Scenario/ Analysis Year	CO (pounds/day)	PM <sub>2.5</sub> (pounds/day)	PM₁₀ (pounds/day)	ROG (pounds/day)	NO <sub>x</sub> (surrogate for NO <sub>2</sub> ) (pounds/day)
Baseline (Existing Conditions) 2019	414,585	16,190	64,132	53,053	164,013
No Build Alternative 2025	255,774	14,675	63,596	36,159	88,581
Build Alternatives 1 and 2 2025	255,763	14,675	63,598	36,165	88,568
Build Alternatives 3A and 3B 2025	255,767	14,675	63,600	36,165	88,570
No Build Alternative 2040	202,379	16,459	73,451	25,512	92,381
Build Alternatives 1 and 2 2040	202,319	16,461	73,462	25,509	92,293
Build Alternatives 3A and 3B 2040	202,339	16,462	73,467	25,512	92,308
No Build Alternative 2045	206,653	16,612	73,843	25,251	89,551
Build Alternatives 1 and 2 2045	206,548	16,614	73,855	25,228	89,420
Build Alternatives 3A and 3B 2045	206,555	16,679	74,220	25,231	90,761

Table 2-24	Summary of C	omparative	Emissions	Analysis in	Pounds	Per Day
------------	--------------	------------	-----------	-------------	--------	---------

Source: Illingworth & Rodkin using CT-EMFAC2017 Version 1.0.2, 2021.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

## MSATs

The project would not change the traffic mix nor move major roadways closer to sensitive receptors. Sensitive receptors are children, elderly, asthmatics, and others whose are at a heightened risk of negative health outcomes due to exposure to air pollution. Sensitive receptors are typically located near schools, hospitals, nursing homes or residential communities where sensitive receptors typically occur. The amount of MSAT emitted is expected to be proportional to VMT, assuming other variables such as fleet mix remain the same.

CT-EMFAC2017, released in January 2019, was used to estimate the emissions of nine MSAT pollutants: acetaldehyde, benzene, ethylbenzene, 1,3-butadiene, formaldehyde, acrolein, naphthalene, diesel PM, and polycyclic organic matter (POM). VMT were estimated for the baseline year (2019), opening year (2025), horizon year (2040), and design year (2045), and applied to the CT-EMFAC2017 emission factors. Table 2-25 shows the MSAT emissions estimated for the baseline, No Build Alternative, and Build Alternatives for all analysis years.

## Conclusions

- MSAT emissions notably decrease for all Build Alternatives compared to existing conditions, but are relatively similar between No Build and Build Alternatives for each study year and alternative.
- CO emissions notably decrease for all Build Alternatives compared to existing conditions, but are relatively similar between No Build and Build Alternatives for each study year and alternative.
- PM<sub>2.5</sub> and PM<sub>10</sub> emissions are slightly lower for all alternatives at the opening year when compared to baseline conditions, but then slightly increase above Baseline as VMT increases in the future. A slight increase in PM<sub>2.5</sub> and PM<sub>10</sub> emissions is computed for all Build Alternatives in each of the analysis years when compared to the No Build Alternative. This is due directly to an increase in regional VMT between the Build Alternatives and the No Build Alternative.
- A decrease in ROG emissions is computed for all alternatives when compared to baseline conditions. All of the build alternatives have a slightly higher ROG emissions in the opening year (2025) when compared to the No-Build Alternative. However, by the design year (2045), ROG emissions would be lower than the No-Build Alternative.
- A decrease in NOx emissions is computed for all alternatives when compared to baseline conditions. All of the build alternatives were estimated to have emissions lower than the No-Build for all years, with the exception of Alternatives 3A and 3B in 2045. NO<sub>X</sub> emissions are higher than the No-Build Alternative for Alternatives 3A and 3B in 2045.

Scenario/Analysis Year	1,3-butadiene (pounds/day)	Acetaldehyde (pounds/day)	Acrolein (pounds/day)	Benzene (pounds/day)	Diesel PM (pounds/day)	Ethylbenzene (pounds/day)	Formaldehyd e (pounds/day)	Naphthalene (pounds/day)	POM (pounds/ day)
Baseline (Existing Conditions) 2020	134.40	705.00	27.05	983.45	2,322.43	741.16	1,605.56	62.90	37.32
No Build Alternative 2025	74.98	173.85	16.52	596.17	460.26	554.28	464.12	45.41	13.57
Build Alternatives 1 and 2 2025	74.97	173.81	16.52	596.19	460.23	554.37	464.03	45.42	13.55
Build Alternatives 3A and 3B 2025	74.97	173.81	16.52	596.20	460.24	554.38	464.04	45.42	13.57
No Build Alternative 2040	57.61	172.40	12.38	437.50	428.92	386.76	428.94	32.77	9.60
Build Alternatives 1 and 2 2040	57.57	172.23	12.37	437.33	428.91	386.73	428.55	32.76	9.50
Build Alternatives 3A and 3B 2040	57.58	172.27	12.37	437.40	429.00	386.79	428.63	32.77	9.50
No Build Alternative 2045	60.08	174.39	12.90	445.12	396.83	387.06	436.55	32.83	9.76
Build Alternatives 1 and 2 2045	60.01	174.09	12.88	444.65	396.89	386.74	435.85	32.80	9.53
Build Alternatives 3A and 3B 2045	60.03	175.00	12.89	444.73	402.41	386.73	437.44	32.79	9.54

Table 2-25	Summary of Comparative MSAT Emissions Analysi
------------	---

Source: Illingworth & Rodkin using CT-EMFAC2017 Version 1.0.2, 2021

## 2.3.6.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-AIR-01. Construction Best Practices for Dust; and
- PF-AIR-02. Construction Best Practices for Exhaust.

#### 2.3.6.5 Climate Change

Neither U.S. EPA nor FHWA has issued explicit guidance or methods to conduct project-level GHG analysis. FHWA emphasizes concepts of resilience and sustainability in highway planning, project development, design, operations, and maintenance. Because there have been requirements set forth in California legislation and EOs on climate change, the issue is addressed in the CEQA chapter of this document. The CEQA analysis may be used to inform the NEPA determination for the project.

#### 2.3.7 Noise and Vibration

## 2.3.7.1 Regulatory Setting

NEPA and CEQA provide the broad basis for analyzing and abating highway traffic noise effects. The intent of these laws is to promote the general welfare and to foster a healthy environment. The requirements for noise analysis and consideration of noise abatement and/or mitigation, however, differ between NEPA and CEQA.

#### California Environmental Quality Act

CEQA requires a strictly baseline versus build analysis to assess whether a proposed project would have a noise impact. If a proposed project is determined to have a significant noise impact under CEQA, then CEQA dictates that mitigation measures must be incorporated into the project unless those measures are not feasible. The rest of this section would focus on the NEPA/23 CFR 772 noise analysis; please see Chapter 3 of this document for further information on noise analysis under CEQA.

#### National Environmental Policy Act and 23 CFR 772

For highway transportation projects with FHWA involvement (and the Department, as assigned), the Federal-Aid Highway Act of 1970 and its implementing regulations (23 CFR 772) govern the analysis and abatement of traffic noise impacts. The regulations require that potential noise impacts in areas of frequent human use be identified during the planning and design of a highway project. The regulations include noise abatement criteria (NAC) that are used to determine when a noise impact would occur. The NAC differ depending on the type of land use under analysis. For example, the NAC for residences (67 A-weighted decibels [dBA]) is lower than the NAC for commercial areas (72 dBA). Table 2-26 lists the NAC for use in the NEPA/23 CFR 772 analysis.

Figure 2-18 lists the noise levels of common activities to enable readers to compare the actual and predicted highway noise levels discussed in this section with common activities.

Activity Category	NAC, Hourly A- Weighted Noise Level, Hourly Equivalent Sound Level (∟eq[h])	Description of Activity Category
A	57 (Exterior)	Lands on which serenity and quiet are of extraordinary significance and serve an important public need and where the preservation of those qualities is essential if the area is to continue to serve its intended purpose.
B <sup>1</sup>	67 (Exterior)	Residential
C <sup>1</sup>	67 (Exterior)	Active sport areas, amphitheaters, auditoriums, campgrounds, cemeteries, day care centers, hospitals, libraries, medical facilities, parks, picnic areas, places of worship, playgrounds, public meeting rooms, public or nonprofit institutional structures, radio studios, recording studios, recreation areas, Section 4(f) sites, schools, television studios, trails, and trail crossings.
D	52 (Interior)	Auditoriums, day care centers, hospitals, libraries, medical facilities, places of worship, public meeting rooms, public or nonprofit institutional structures, radio studios, recording studios, schools, and television studios.
E	72 (Exterior)	Hotels, motels, offices, restaurants/bars, and other developed lands, properties, or activities not included in A–D or F.
F	No NAC—reporting only	Agriculture, airports, bus yards, emergency services, industrial, logging, maintenance facilities, manufacturing, mining, rail yards, retail facilities, shipyards, utilities (water resources, water treatment, electrical, etc.), and warehousing.
G	No NAC—reporting only	Undeveloped lands that are not permitted

**Noise Abatement Criteria** Table 2-26

Notes:

<sup>1</sup> Includes undeveloped lands permitted for this activity group.

Common Outdoor Activities	Noise Level (dBA)	Common Indoor Activities
Jet Fly-over at 300m (1000 ft)	110	Rock Band
Gas Lawn Mower at 1 m (3 ft)		
Diesel Truck at 15 m (50 ft),	90	Food Blender at 1 m (3 ft)
Noisy Urban Area, Daytime	80	Garbage Disposal at 1 m (3 π)
Gas Lawn Mower, 30 m (100 ft) Commercial Area	70	Vacuum Cleaner at 3 m (10 ft) Normal Speech at 1 m (3 ft)
Heavy Traffic at 90 m (300 ft)	60	Large Business Office
Quiet Urban Daytime	50	Dishwasher Next Room
Quiet Urban Nighttime Quiet Suburban Nighttime	40	Theater, Large Conference Room (Background)
Quiet Rural Nighttime	30	Library Bedroom at Night,
	20	Concert Hall (Background) Broadcast/Recording Studio
	10	
Lowest Threshold of Human Hearing	0	Lowest Threshold of Human Hearing

Figure 2-18 Noise Levels of Common Activities

According to the Department's *Traffic Noise Analysis Protocol for New Highway Construction and Reconstruction Projects, May 2011*, a noise impact occurs when the predicted future noise level with the project substantially exceeds the existing noise level (defined as a 12 dBA or more) or when the future noise level with the project approaches or exceeds the NAC. A noise level is considered to approach the NAC if it is within 1 dBA of the NAC.

If it is determined that the project would have noise impacts, then potential abatement measures must be considered. Noise abatement measures that are determined to be reasonable and feasible at the time of final design are incorporated into the project plans and specifications. This document discusses noise abatement measures that would likely be incorporated in the project.

The Department's Traffic Noise Analysis Protocol sets forth the criteria for determining when an abatement measure is reasonable and feasible. Feasibility of noise abatement is basically an engineering concern. Noise abatement must be predicted to reduce noise by at least 5 decibels (dB) at an impacted receptor to be considered feasible from an acoustical perspective. It must also be possible to design and construct the noise abatement measure for it to be considered feasible. Factors that affect the design and constructability of noise abatement include, but are not limited to, safety, barrier height, topography, drainage, access requirements for driveways, presence of local cross streets, underground utilities, other noise sources in the area, and maintenance of the abatement measure. The overall reasonableness of noise abatement is determined by the following three factors: (1) the noise reduction design goal of 7 dB at one or more impacted receptors; (2) the cost of noise abatement; and (3) the viewpoints of benefited receptors).

# 2.3.7.2 Affected Environment

The analysis summarized in this section is from the Noise Study Report (Illingworth & Rodkin, Inc. 2021) completed for the proposed project. The CEQA baseline for this section is 2019-2020, when the traffic and noise measurements were conducted. The noise study and report were completed in 2021. The NEPA baseline for comparing environmental impacts is the No Build Alternative.

The noise study area encompasses all developed and undeveloped land uses surrounding the project limits, with a focus on noise-sensitive land uses. In general, noise-sensitive land uses include areas where serenity and quiet are of extraordinary significance, such as residential land uses and other community uses such as hospitals, schools, cemeteries, and parks.

The existing noise environment throughout the project limits varies by location, depending on site characteristics such as proximity of receptors to U.S. 101, major roadways, or other sources of noise in the area; the relative base elevations of roadways and receptors; and the presence of any intervening structures or barriers. Noise receptor locations in the project area were identified through a review of project mapping, aerial photographs, and field reconnaissance. The noise analysis focused on locations with defined outdoor activity areas, including residential backyards, parks,

trailheads, and active sport areas. There are no other noise-sensitive noise receptors, such as libraries, churches, hospitals, in the project area. The noise-sensitive land uses in or along the project corridor include Activity Categories B and C.

Noise-sensitive land uses in the project area include single-family and multi-family residences (Activity Category B); active sport areas, day care centers, hospitals, medical facilities, parks, playgrounds, recreation areas, schools, and trails (Activity Category C); day care centers, hospitals, medical facilities, and schools (Activity Category D); and hotels, motels, offices, restaurants, and bars (Category E). These land uses vary in their sensitivity to freeway and road noise and are ranked by activity category in Table 2-26. The noise receptor locations are shown on Figure 2-19 on the second page following.

## Noise Study

Long- and short-term noise measurements were made in November 2020. Long-term reference noise measurements were made at three locations in the project vicinity to quantify the diurnal trend in noise levels and to establish the peak traffic noise hour. These reference noise measurements included a site near the Northwestern Pacific Railroad track, just south of SR 37 along Tolay Creek Road (L1); a site just north of SR 37 along Noble Road, at 5400 Sears Point Road (L2); and a site just south of SR 37, at 984 Fortune Street in Vallejo. Nineteen short-term noise measurements (S1 through S19) were made in the project vicinity in concurrent time intervals with the data collected at the long-term reference measurement sites. This method facilitates a direct comparison between both the short-term and long-term noise measurements; it also allows for the identification of the loudest-hour noise levels at land uses in the project vicinity where long-term noise measurements were not made, but where both short-term and long-term measurements are exposed to the same primary noise source. The results of the short and long-term measurements are listed in Table 2-27 and Table 2-28.

Traffic counts and speed observations were made along SR 37 during the short-term noise measurements, for model calibration purposes. Traffic volumes were classified into five vehicle types: (1) light-duty automobiles and trucks, (2) medium-duty trucks (typically trucks with two axles and more than four wheels), (3) heavy-duty trucks (typically trucks with more than two axles), (4) buses, and (5) motorcycles.

Handheld weather meters were used to collect weather data at noise measurement locations during the field noise investigation. Noise monitoring did not occur if weather conditions consisted of rain or high winds (i.e., greater than 11 mph).

Receptor ID	Location (Figure 2-19)	Date	Loudest Hour(s)	Measured Loudest-Hour L <sub>eq[h]</sub> , dBA
L1	Tolay Creek Road	11/19/2020	7:00 a.m., 10:00 a.m.	72
L2	5400 Sears Point Road	11/19/2020	10:00 a.m.	76
L3	984 Fortune Street	11/19/2020	6:00 a.m.	71

 Table 2-27
 Summary of Long-Term Noise Measurements

Receptor ID	Location (Figure 2-19)	Date	Start Time	10-minute L <sub>eq</sub> , dBA
64	2100 Secre Daint Baad	11/10/2020	10:00 a.m.	72
51	2100 Sears Point Road	11/19/2020	10:10 a.m.	72
60	20700 Taley Creak Bood	11/10/2020	10:00 a.m.	64
52	29790 Tolay Creek Road	11/19/2020	10:10 a.m.	63
62	Tubbo Jolond Troilbood	11/10/2020	10:00 a.m.	69
		11/19/2020	10:10 a.m.	69
<u>84</u>	Nabla Dood	11/10/2020	10:40 a.m.	72
54		11/19/2020	10:50 a.m.	72
S.F.	5000 Seere Deint Beed	11/10/2020	10:40 a.m.	64
55	5000 Sears Point Road	11/19/2020	10:50 a.m.	64
56	Sonoma Crook (North)	11/19/2020	11:00 a.m.	67
50	Sonoma Creek (North)	11/18/2020	11:10 a.m.	68
67	Samara Grack (South)	11/10/2020	11:30 a.m.	57 <sup>1</sup>
57	Sonoma Creek (South)	11/18/2020	11:40 a.m.	57 <sup>1</sup>
60	Skagge Jelend Deed	11/19/2020	10:30 a.m.	71
50	Skaggs Island Road	11/10/2020	10:40 a.m.	71
80	Cullinan Ranch Trailhead	11/18/2020 -	10:00 a.m.	67
59			10:10 a.m.	67
<u>810</u>	Outline Maria	11/10/2020	11:20 a.m.	66
510	Sylvan way	11/19/2020	11:30 a.m.	66
<u>811</u>	Povlanda Viata Daint	11/19/2020 -	11:20 a.m.	64
511	Baylands Vista Point		11:30 a.m.	65
S10	North Diar Streat	11/10/2020	11:20 a.m.	59
512	North Pier Street	11/19/2020	11:40 a.m.	60
<u>612</u>	121 Lighthouse Drive	11/10/2020	12:10 p.m.	64
515	131 Lighthouse Drive	11/19/2020	12:20 p.m.	63
<u>814</u>	297 Lighthouse Drive	11/10/2020	12:10 p.m.	63
514	287 Lighthouse Drive	11/19/2020	12:20 p.m.	62
<u>8</u> 15	125 Company Court	11/10/2020	12:10 p.m.	65
515	125 Compass Court	11/19/2020	12:20 p.m.	64
S16	Federal Terrace Elementary	11/10/2020	12:50 p.m.	48 <sup>2</sup>
510	School	11/19/2020	1:00 p.m.	47 <sup>2</sup>
617	1016 Fortune Street	11/10/2020	12:50 p.m.	65
517	To to Follune Street	11/19/2020	1:00 p.m.	65
C10	094 Fortuno Stroot	11/10/2020	12:50 p.m.	65
310	904 FUILUITE SUIEEL	11/19/2020	1:00 p.m.	65
<b>S</b> 10	5400 Soore Doint Bood	11/10/2020	10:30 a.m.	75
019		11/19/2020	10:40 a.m.	75

Table 2-28	Summary	/ of Short-Term	Noise	Measurements
------------	---------	-----------------	-------	--------------

Notes:

1

Noise data acquired at measurement S7 is unreliable because weather effects contaminated results. Hourly average noise levels at measurement S16 were skewed by local traffic events. L<sub>50</sub> noise levels are used to best represent noise originating from SR 37. 2





SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

Measurement Point ---- Modeled Barrier



Source: AECOM, 2020; Solano County, 2019; Sonoma County, 2018; ESRI, 2016 (roads)

FIGURE 2-19: NOISE MEASUREMENTS AND EVALUATED BARRIERS Page 1 of 4





SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

Measurement Point - Modeled Barrier



Source: AECOM, 2020; Solano County, 2019; Sonoma County, 2018; ESRI, 2016 (roads)

FIGURE 2-19: NOISE MEASUREMENTS AND EVALUATED BARRIERS Page 2 of 4





Feet

Ñ

Measurement Point - Modeled Barrier



FIGURE 2-19: NOISE MEASUREMENTS AND EVALUATED BARRIERS Page 3 of 4

EIR \Figure 2-19 - Noise.mx







Measurement Point ---- Modeled Barrier



Source: AECOM, 2020; Solano County, 2019; Sonoma County, 2018; ESRI, 2016 (roads)

FIGURE 2-19: NOISE MEASUREMENTS AND EVALUATED BARRIERS Page 4 of 4

## Future Undeveloped Land Uses

Lists of planned and approved projects in Sonoma, Solano, and Napa Counties and in the City of Vallejo in the vicinity of the project were reviewed to identify undeveloped lands for which development is planned, designed, and programmed, so that those proposed developments may be considered approved (or a part of the existing conditions). According to the Traffic Noise Analysis Protocol, future development would be considered planned, designed, and programmed once it receives final development approval. The review focused on projects within approximately 500 feet of the project limits, where traffic noise levels from the improved project roadways could dominate the noise environment. Projects beyond this distance were excluded from further analysis. No noise-sensitive projects that are proposed within 500 feet of the project alignment were identified.

# 2.3.7.3 Environmental Consequences

The proposed project would add through traffic lanes and was therefore determined to be a Type I project in accordance with 23 CFR Part 772, requiring a traffic noise analysis.

## **Traffic Noise Modeling Results**

Traffic noise levels were predicted using the FHWA Traffic Noise Model Version 2.5 (TNM 2.5). TNM 2.5 is a computer model based on two FHWA reports: FHWA-PD-96-009 and FHWA-PD-96-010 (FHWA 1998a, 1998b). Key inputs to the traffic noise model were the locations of roadways, traffic mix and speed, shielding features (e.g., topography and buildings), noise barriers, ground type, and receptors. Three-dimensional representations of these inputs were developed using CAD drawings, aerials, and topographic contours.

Traffic noise was evaluated under existing conditions, design-year no-project conditions, and design-year conditions with the project alternatives. Loudest-hour traffic volumes, vehicle classification percentages, and traffic speeds under existing and design-year (2045) conditions were input into the traffic noise model.

To validate the accuracy of the model calculations, TNM 2.5 was used to compare measured traffic noise levels to modeled noise levels at field measurement locations. For each receptor, traffic volumes counted during the short-term measurement periods were normalized to 1-hour volumes. These normalized volumes were assigned to the corresponding project area roadways to simulate the noise source strength at the roadways during the actual measurement period. Modeled and measured sound levels were then compared to determine the accuracy of the model and whether additional adjustment of the model was necessary.

Traffic noise modeling results and predicted traffic noise impacts (rounded to the nearest dB) for existing and design year conditions are shown in Table 2-29. In this table, 2045 Build traffic noise levels are compared to existing conditions and to 2045 No Build conditions. The comparison to existing conditions is included in the analysis to identify traffic noise impacts as defined under 23 CFR 772. The comparison

between 2045 Build and 2045 No Build conditions indicates the direct effect of the project.

As shown in Table 2-29, the loudest-hour noise levels at Category B land uses are calculated to range from 50 to 66 dBA  $L_{eq}(h)$  under existing conditions, from 51 to 66 dBA  $L_{eq}(h)$  under 2045 No Build conditions, and from 52 to 67 dBA  $L_{eq}(h)$  under 2045 Build conditions. The loudest-hour noise levels at Category C land uses are calculated to range from 48 to 72 dBA  $L_{eq}(h)$  under Existing conditions, from 49 to 73 dBA  $L_{eq}(h)$  under 2045 No Build conditions, and from 49 to 73 dBA  $L_{eq}(h)$  under 2045 No Build conditions, and from 49 to 73 dBA  $L_{eq}(h)$  under 2045 Build conditions. The loudest-hour noise level at the Category D land use was calculated to reach 63 dBA  $L_{eq}(h)$  under Existing conditions and 2045 No Build conditions, and 64 dBA  $L_{eq}(h)$  under 2045 Build conditions. 2045 Build traffic noise levels are predicted to approach or exceed the NAC at one Category B receptor at 29790 Tolay Creek Road (S2). 2045 Build traffic noise levels are predicted to approach or exceed the NAC at one Category B receptor at 29790 Tolay Creek Road (S2). 2045 Build traffic noise levels are predicted to approach or exceed the NAC at Six Category C receptors, including the Tubbs Island Trailhead (S3), 5000 Sears Point Road (S5), Sonoma Creek Trail (S6), Sonoma Creek Wildlife Viewing Point (S7), Skaggs Island Road Trailhead (S8), and the Cullinan Ranch Trailhead (S9).

The noise level at the worst-case exterior façades of the Category D land use identified at the Refuge Headquarters (R1) was calculated to reach 64 dBA  $L_{eq}(h)$ . Based on observations made and photographs taken at measurement location S1, approximately 180 feet from the structure, the Refuge Headquarters building nearest SR 37 and represented by R1 is equipped with mechanical ventilation, allowing occupants the option of closing windows to control noise. Assuming a minimum exterior-to-interior noise reduction of 20 dB, interior noise levels in the structure are not anticipated to exceed 52 dBA  $L_{eq}(h)$ .

Noise levels would increase by up to 2 dBA over existing conditions under 2045 No Build conditions. Under Build Alternatives 1, 2, 3A, and 3B, noise levels would increase by 1 to 2 dBA when compared to existing conditions. The project would not result in noise level increases that would be considered substantial according to the Caltrans threshold of 12 dBA.

# Short-Term Construction Noise Impacts

Project construction is anticipated to occur over a period of 2 years and would include grubbing/land clearing, grading/excavation/foundation/sheet pile walls, drainage/ utilities/subgrade, and paving. Pile driving is anticipated to be used as a method of construction along segments of the project alignment for structure foundation. Construction noise would primarily result from the operation of heavy construction equipment and arrival and departure of heavy-duty trucks. The types of equipment needed to complete the construction may include, but are not limited to, crawler tractors, excavators, signal boards, cranes, grades, rollers, rubber-tired loaders, scrapers, backhoes, bore/drill rigs, cement and mortar mixers, air compressors, generator sets, plate compactors, pumps, rough terrain forklifts, pavers, and paving equipment.

		Lou	dest-Ho	our Noise Leve	ls, L <sub>eq(h)</sub> dBA	Increase Over Existing, dBA			Increase Over 2045 No Build, dBA		Impact <sup>1</sup>	
	Activity		2045	2 B	045 uild	2045	2 B	045 suild	2045 Build		2045 Build	
Receptor ID	Category (NAC)	Exist	No Build	Alternatives 1 and 2	Alternatives 3A and 3B	No Build	Alternatives 1 and 2	Alternatives 3A and 3B	Alternatives 1 and 2	Alternatives 3A and 3B	Alternatives 1 and 2	Alternatives 3A and 3B
S1	Reference <sup>2</sup>	70	71	71	71	1	1	1	0	0	N/A	N/A
S2	B(67)	66	66	67	67	0	1	1	1	1	A/E	A/E
S3	C(67)	71	72	72	72	1	1	1	0	0	A/E	A/E
S4	Reference <sup>2</sup>	73	74	75	75	1	2	2	1	1	N/A	N/A
S5	C(67)	68	69	69	69	1	1	1	0	0	A/E	A/E
S6	C(67)	67	67	68	68	0	1	1	1	1	A/E	A/E
S7	C(67)	68	69	69	69	1	1	1	0	0	A/E	A/E
S8	C(67)	72	73	73	73	1	1	1	0	0	A/E	A/E
S9	C(67)	67	68	69	69	1	2	2	1	1	A/E	A/E
S10	G <sup>2</sup>	66	67	68	68	1	2	2	1	1	N/A	N/A
S11	C(67)	64	65	65	65	1	1	1	0	0	None	None
S12	G <sup>2</sup>	58	59	60	60	1	2	2	1	1	N/A	N/A
S13	B(67)	63	64	64	64	1	1	1	0	0	None	None
S14	B(67)	62	64	64	64	2	2	2	0	0	None	None
S15	Reference <sup>2</sup>	64	66	66	66	2	2	2	0	0	N/A	N/A
S16	C(67)	48	49	49	49	1	1	1	0	0	None	None
S17	B(67)	62	64	64	64	2	2	2	0	0	None	None
S18	B(67)	61	63	63	63	2	2	2	0	0	None	None
S19	Reference <sup>2</sup>	74	75	76	76	1	2	2	1	1	N/A	N/A
R1	D(52) <sup>3</sup>	63	63	64	64	0	1	1	1	1	None	None
R2	E(72)	65	66	66	66	0	1	1	1	1	None	None
R3	B(67)	62	63	63	63	1	1	1	0	0	None	None
R4	B(67)	50	51	52	52	1	2	2	1	1	None	None

Table 2-29 Calculated Noise M	leasurements by Alternative
-------------------------------	-----------------------------

		Loudest-Hour Noise Levels, L <sub>eq(h)</sub> dBA			Increase Over Existing, dBA			Increase Over 2045 No Build, dBA		Impact <sup>1</sup>		
Activity		2045		2045 Build		2045 2045 Build		2045 Build		2045 Build		
Receptor ID	Category (NAC)	Exist	No Build	Alternatives 1 and 2	Alternatives 3A and 3B	No Build	Alternatives 1 and 2	Alternatives 3A and 3B	Alternatives 1 and 2	Alternatives 3A and 3B	Alternatives 1 and 2	Alternatives 3A and 3B
R5	B(67)	60	61	62	62	1	2	2	1	1	None	None
R6	B(67)	61	62	63	63	1	2	2	1	1	None	None
R7	B(67)	56	57	57	58	1	1	2	0	1	None	None
R8	B(67)	54	55	56	56	1	2	2	1	1	None	None
R9	B(67)	61	62	63	63	1	2	2	1	1	None	None
R10	B(67)	55	56	57	57	1	2	2	1	1	None	None
R11	B(67)	61	62	63	63	1	2	2	1	1	None	None

Notes:

<sup>1</sup> Impact Type: A/E = Approach or Exceed NAC; None = increase is less than 12 dB and noise levels do not approach or exceed the NAC; N/A = not applicable to reference measurements and measurements not in areas of frequent human use.

<sup>2</sup> Measurement point was selected based on accessibility constraints but was not representative of an area of frequent human use.

<sup>3</sup> R1 is at the northwestern façade of Refuge Headquarters building. This location does not include any exterior areas of frequent human use and is considered a Category D land use only. Exterior noise levels are presented in the table.

Although the overall construction schedule is anticipated to occur over a period of 2 years, roadway construction activities typically occur for relatively short periods of time in any specific location as construction proceeds along the project's alignment. Construction noise would mostly be of concern in areas where heavy construction would be concentrated for extended periods of time in areas adjacent to noise-sensitive receptors, where noise levels from individual pieces of equipment are substantially higher than ambient conditions, or when construction activities would occur during noise-sensitive time periods, such as early morning, evening, or nighttime hours. Noise-sensitive receptors typically include residences, hospitals, schools, guest lodging, libraries, and churches.

As indicated through comparison of Table 2-30, most construction phases would generate average noise levels that would exceed ambient daytime noise levels at adjacent land uses by 15 to 20 dBA  $L_{eq}(h)$ . Except for short periods of pile driving (if used as a method of construction), heavy demolition, and site preparation, construction noise levels would not be expected to exceed the quantitative noise limits established by Caltrans.

Construction		Maximu Le (L <sub>max</sub> ,	m Noise vel dBA)	Hourly Average Noise Level (L <sub>eq[h]</sub> , dBA)	
Туре	Construction Phase	50 feet	100 feet	50 feet	100 feet
	Grubbing/Land Clearing	84	78	83	77
<b>_</b>	Grading/Excavation	85	79	90	84
Roadway	Grading/Excavation/Sheet Pile Walls	101	95	96	90
Construction	Drainage/Utilities/Sub-Grade	85	79	90	84
	Paving	90	84	86	80
	Grubbing/Land Clearing	84	78	82	76
Bridge	Grading/Excavation/Foundation	85	79	88	82
Construction	Drainage/Utilities/Sub-Grade	85	79	88	82
	Paving	90	84	85	79
Impact Pile Driving Alone	_	101	95	94	88

 Table 2-30
 Calculated Noise Measurements by Phase

# Preliminary Noise Abatement Analysis

Noise abatement is considered where noise impacts are predicted in areas of frequent human use that would benefit from a lowered noise level. Noise abatement must be predicted to provide at least a 5 dB minimum reduction at an impacted receptor to be considered feasible by Caltrans (i.e., the barrier would provide a noticeable noise reduction). Additionally, the Traffic Noise Analysis Protocol's acoustical design goal states that the noise barrier must provide at least 7 dB of noise reduction at one or more benefited receptors. Noise abatement measures that provide noise reduction of more than 5 dB are encouraged, as long as they meet the reasonableness guidelines. Noise barriers were considered as noise abatement for exterior land uses in the project area, as shown in Figure 2-19. Each noise barrier has been evaluated for feasibility based on achievable noise reduction. Once a noise barrier achieved the minimum of a 5 dB reduction at a given receptor and achieved the 7 dB acoustical design goal for at least one receptor, the reasonable allowance was determined. Impacted receptors including S2, S3, S8, and S9 are situated along railways that cross SR 37 or roads that are accessed directly from a turn off of the SR 37 mainline. It is not possible to construct one continuous barrier at these receptors. Instead, sets of eastern and western barriers on either side of the railway or road were evaluated, with an estimated reasonable space remaining in between to allow for adequate sightlines. Due to the gap between the eastern and western barriers, little noise reduction was possible. Evaluated barriers 1, 2, 5, 6, and 7 would not feasibly abate traffic noise at any height and/or would not meet the minimum 7 dBA reduction goal, and were not considered further. Barrier 3 would feasibly abate traffic noise at a height of 10 feet and above, but would not meet the 7 dBA minimum reduction goal at any height and was also not considered further.

Barrier 4 would feasibly abate traffic noise at receptor S6 at heights of 10 feet and above and would meet the 7 dB acoustical design goal at heights of 14 feet and above. This barrier would have to be on the edge of the Sonoma Creek Bridge, in the westbound direction, to abate noise at the adjacent trail and parking lot located near Sonoma Creek. For any noise acoustically feasible barrier to be considered reasonable from a cost perspective, the estimated cost of the barrier should be equal to or less than the total cost allowance calculated for the barrier. The cost calculations of the noise barrier must include all items appropriate and necessary for construction of the barrier, such as traffic control, drainage modification, retaining walls, landscaping for graffiti abatement, and right-of-way costs. The reasonableness allowance was calculated for the noise barriers that were determined to be acoustically feasible and to meet the Caltrans acoustical design goal. For each noise barrier found to be acoustically feasible, reasonable cost allowances were calculated by multiplying the number of benefited receptors by \$107,000.

The engineering estimate for construction of barrier 4 included the costs of adding a soundwall to the westbound edge of the Sonoma Creek Bridge. The cost estimate included the construction of a wall on the bridge, crew mobilization, and other factors specific to this location. The total cost was estimated at approximately \$2 million, which is substantially higher than the allowance of \$107,000. The relatively high cost is due to the necessary location of the wall on an existing bridge structure. Because the wall exceeded the cost allowance, it was not considered further and not included in the project design.

## **Final Noise Abatement Decision**

The 45-day public review period for the Draft EIR/EA was from January 13 through February 28, 2022, and a virtual public meeting was held on February 2. Comments from the public and agencies, and responses, are included in Appendix K. There were no comments received regarding noise abatement or the evaluation of the potential noise barriers described in this section. Several general comments were made that construction noise should be evaluated and minimized, and this is addressed in this document. The final decision on noise abatement for this project is that of the potential barriers evaluated, only one was identified as acoustically feasible (Barrier 4), but Barrier 4 was determined not reasonable because the estimated cost of the wall on the Sonoma Creek Bridge exceeded the cost allowance. Therefore, no traffic-related noise abatement is required or proposed for this project. Construction noise minimization measures are specified in Table 1-5.

#### Vibration Analysis

Construction activities would include grubbing/land clearing, grading/excavation/ foundation/sheet pile walls, drainage/utilities, and paving. Pile driving would be used as a method of construction for structure foundation. Blasting, which has the potential to result in high levels of vibration, would not be used. Traffic, including heavy trucks traveling on a highway, rarely generates vibration amplitudes high enough to cause structural or cosmetic damage.

Construction activities with the greatest potential of generating perceptible vibration levels would include the removal of pavement and soil, the dropping of heavy objects, and the movement of heavy tracked equipment. Table 2-31 presents typical vibration levels that could be expected from representative construction equipment at a reference distance of 25 feet and calculated vibration levels at distances representative of the setbacks from the project to the nearest structures. Vibration levels are highest close to the source, and then attenuate with increasing distance depending on soil conditions.

		PPV at 25 feet	Vibration Level at Distance Representative of Setbacks to Nearest Structures (in/sec PPV) <sup>1</sup>				
Equi	oment	(in/sec)	80 feet	190 feet	260 feet		
Pile driver	Upper range	1.158	0.322	0.124	0.088		
(impact)	Typical	0.644	0.179	0.069	0.049		
Pile driver	Upper range	0.734	0.204	0.079	0.056		
(sonic)	Typical	0.170	0.047	0.018	0.013		
Clam shovel drop		0.202	0.056 0.022		0.015		
Hydromill	In soil	0.022	0.002	0.001	0.001		
(slurry wall)	In rock	0.047	0.005	0.002	0.001		
Vibratory roller		0.210	0.058	0.023	0.016		
Hoe ram		0.089	0.025	0.025 0.010			
Large bulldoze	r	0.089	0.025 0.010		0.007		
Caisson drilling	]	0.089	0.025	0.010	0.007		
Loaded trucks		0.076	0.021	0.008	0.006		
Jackhammer		0.035	0.010	0.004	0.003		
Small bulldoze	r	0.003	0.001	0.000	0.000		

Table 2-31	Vibration Source Levels for Construction Equipment

Source: Illingworth & Rodkin, Inc. 2021

Caltrans identifies a vibration limit of 0.5 inch per second (in/sec) peak particle velocity (PPV) as the threshold at which there is a potential risk of damage to new residential and modern commercial/industrial structures, 0.3 in/sec PPV for older residential structures, and a conservative limit of 0.25 in/sec PPV for historic and some old buildings. These limits would only be exceeded at the nearest structures during upper range impact pile driving. The nearest structure represented by the 80-foot setback distance is a building at the Refuge Headquarters (receptor S1). According to project plans, no pile driving would be required west of the SMART railroad crossing. The Refuge Headquarters building is approximately 1 mile west of the crossing, and therefore there are no structures that would be exposed to construction-generated vibration which would have the potential to exceed Caltrans limits. No construction vibration minimization measures would be necessary.

## 2.3.7.4 Avoidance, Minimization, and/or Mitigation Measures

No avoidance, minimization, or mitigation is required. The following project features would be implemented as described Table 1-4 in Section 1.5:

- PF-AIR-01. Construction Best Practices for Dust; and
- PF-AIR-02. Construction Best Practices for Exhaust.

# 2.3.8 Energy

# 2.3.8.1 Regulatory Setting

NEPA (42 USC Part 4332) requires the identification of all potentially significant impacts to the environment, including energy impacts.

CEQA Guidelines section 15126.2(b) and CEQA's Appendix G, Energy Conservation, require an analysis of a project's energy use to determine if the project may result in significant environmental effects due to wasteful, inefficient, or unnecessary use of energy, or wasteful use of energy resources.

# 2.3.8.2 Affected Environment

This section summarizes the *Energy Technical Report* prepared in November 2021 for the project (AECOM 2021b). The NEPA baseline for comparing environmental impacts is the No Build Alternative.

The transportation sector is the top consumer of energy in California, comprising nearly 40 percent of energy consumption in 2018 (EIA 2018). The State of California relies on both nonrenewable and renewable energy sources. Nonrenewable energy resources used in California include petroleum, natural gas, and nuclear power; renewable energy resources include hydroelectric, biomass, wind, solar, and geothermal heat (heat given off by the Earth). A total of 36 percent of California's electricity comes from renewable sources, and 42 percent of that renewable energy comes from solar, the state's top renewable energy source. Fossil fuels have been the leading transportation fuels in the country and state. Gasoline is the most consumed fuel in California, at approximately 55.79 percent of total fossil fuel consumption for the state's transportation sector. Table 2-32 shows fossil fuel consumption in California for the transportation sector. The amount of fuel used is expressed by British Thermal Unit (Btu). A Btu is the amount of energy required to raise the temperature of 1 pound of water 1-degree Fahrenheit.

Fuel Type	Trillion Btu	Percent of Total California Consumption
Natural Gas	44.8	1.42
Aviation Gasoline	2.2	0.07
Distillate Fuel Oil	483.8	15.30
HGL	0.7	0.02
Jet Fuel	684.8	21.65
Lubricants	13.2	0.42
Motor Gasoline	1,764.4	55.79
Residual Fuel Oil	168.8	5.34
Total	3,162.7	100.00

# Table 2-32Fossil Fuel Consumption in California for the Transportation Sector<br/>(2018)

Source: EIA 2018

Alternatives to fossil fuels for transportation have helped decrease the dependence on gasoline and other fossil fuels. In addition to traditional petroleum fuels, California currently uses the following "alternative" fuels and energy sources:

- Compressed natural gas (CNG)
- Electric (EVC)
- Ethanol, 85 percent (E85)
- Hydrogen
- Liquefied natural gas (LNG)
- Liquefied petroleum gas (LPG)

## 2.3.8.3 Environmental Consequences

#### **Direct Energy Impacts**

The direct energy impacts were evaluated based on VMT and average travel speeds.

**VMT.** Table 2-12 and Table 2-13 in Section 2.2.11 show the estimated VMT for 2025 and 2045. VMT is expected to increase with all the project alternatives (i.e., Build and No Build, relative to the base year conditions [2020]). The increase in VMT is attributed to increased travel demand on SR 37, regardless of the chosen alternative. Additionally, VMT is expected to slightly increase with the Build Alternatives compared to the No Build Alternative. In 2025, daily VMT would increase by 6,346 from No Build Conditions for Alternatives 1 and 2, and by 9,599 for Alternatives 3a and 3b. In 2045,

Daily VMT would increase by 31,729 for Alternatives 1 and 2 and by 47,992 in 2045 for Alternatives 3A and 3B.

**Energy Use Related to VMT.** Energy use factors were calculated as a statistical average to estimate fuel consumption in gallons per mile. To calculate and project the vehicle fuel used by the proposed project, the total VMT (in miles per day) for typical on-road vehicles and the total amount of vehicle fuel (in gallons per day) used in the Bay Area region were obtained from the CARB EMFAC2017 model. EMFAC2011 vehicle categories and aggregated model year and speed were used for the analysis.

Results from the model were calculated for the Base Year, 2025 Opening Year, and the 2045 Design Year for both gasoline and diesel fuel types. Average gallons per mile were then calculated for each year and each fuel type. Table 2-33 shows the results of calculating the direct energy factors. Though the projected VMT appears to increase over the years, the total gallons consumed per day decrease, which is associated with better energy efficiency and standards that apply as older vehicles are replaced over time by increasingly more fuel-efficient cars and trucks.

Fuel Types	Year	Fuel-Specific VMT (miles/day)	Fuel Consumption (Gallons/Day)	Average Gallons/Mile	Percent Share VMT
Gasoline	2020 Base Year	160,072,121.51	6,334,019.09	0.0396	93.22%
Gasoline	2025 Opening Year	165,408,573.25	5,653,255.92	0.0342	92.67
Gasoline	2045 Design Year	187,133,077.19	4,985,132.81	0.0266	91.78
Diesel	2020 Base Year	11,634,424.06	1,234,967.26	0.1061	6.78%
Diesel	2025 Opening Year	13,085,992.54	1,231,711.07	0.0941	7.33
Diesel	2045 Design Year	16,752,299.19	1,256,414.43	0.0750	8.22

 Table 2-33
 Direct Energy Factors for Energy Consumption

Source: EMFAC 2017

The average gallons per mile derived in Table 2-33 were multiplied by operational daily VMT to estimate direct energy consumption in gallons of gasoline and diesel consumed by vehicles for each analysis year. Gasoline automobiles were assumed to contribute 93.22, 92.67, and 91.78 percent of daily operational VMT for 2020, 2025, and 2045, respectively. To convert to direct energy consumption in Btu, it is assumed that a gallon of gasoline has an energy content of 120,941 Btu, and a gallon of diesel has 137,320 Btu. Table 2-34 shows the operational fuel consumption for the study area.

Table 2-34 shows that overall energy consumption is anticipated to decrease over time relative to the base year, regardless of the chosen alternative. This is associated with better energy efficiency and standards. Total energy consumption is similar for the Build and No Build Alternatives, with a slight increase in energy consumption for the Build Alternatives. This correlates to the slight increase in VMT anticipated for the Build Alternatives.

Project Alternatives	Energy Consumption: Automobile- Gasoline (gallons)	Energy Consumption : Truck- Diesel (gallons)	Total Energy Consumption (100,000 Btu)	Change from Base Year (1000,000 Btu)	Percent Change from Base Year	Change from No Build (100,000 Btu)	Percent Change from No Build
2020 Base Year	5,531,410.29	1,078,479.64	8,170,711.16	—	—	—	_
2025 No Build	4,948,897.70	1,078,248.04	7,465,896.57	-704,814.59	-8.63	—	_
2025 Alternatives 1 and 2	4,949,098.69	1,078,291.83	7,466,199.79	-704,511.38	-8.62	303.21	0.00
2025 Alternatives 3A and 3B	4,949,201.72	1,078,314.27	7,466,355.21	-704,355.95	-8.62	458.64	0.01
2045 No Build	4,437,329.32	1,118,350.27	6,902,269.04	-1,268,442.13	-15.52	_	_
2045 Alternatives 1 and 2	4,438,105.14	1,118,545.80	6,903,475.82	-1,267,235.34	-15.51	1,206.79	0.02
2045 Alternatives 3A and 3B	4,438,502.75	1,118,646.01	6,904,094.32	-1,266,616.84	-15.50	1,825.28	0.03

Source: AECOM 2021b and EMFAC 2017.

Evaluation of Energy Consumption Based on Traffic Conditions. In addition to analyzing direct energy consumption based on VMT, direct energy consumption may be inferred from traffic flow for No Build and Build Alternatives. Higher speeds correlate to better fuel economy, and stop-and-go traffic conditions lead to increased fuel consumption. Fuel efficiency for midsize conventional gasoline- and dieselpowered vehicles tends to peak between 35 and 55 mph, after which efficiency steadily declines (United States Department of Energy 2020). VHD are anticipated to increase over time, regardless of the chosen alternative. However, all Build Alternatives are anticipated to reduce VHD relative to the No Build Alternative. Build Alternatives 1 and 2 are anticipated to reduce daily VHD by 2,187 and 10,935 in 2025 and 2045, respectively, when compared to the No Build Alternative (Table 2-35). Alternatives 3a and 3b are anticipated to reduce daily VHD by 3,187 and 15,936 in 2025 and 2045, respectively, when compared to the No Build Alternative. The savings in VHD are representative of improved traffic conditions for the Build Alternatives. Therefore, all Build Alternatives would reduce the potential for wasteful energy use as a consequence of stop-and-go traffic conditions.

Project Alternatives	Daily VHD	Change from Base Year	Percent Change from Base Year	Change from No Build	Percent Difference – Build Versus No Build
2020 Base Year	5,523,543	—	—	—	—
2025 No Build	6,501,187	977,644	0.18	—	—
2025 Alternatives 1 and 2	6,499,000	975,457	0.18	2,187	0.03
2025 Alternatives 3A and 3B	6,498,000	974,457	0.18	3,187	0.05
2045 No Build	10,411,762	4,888,219	0.89	—	—
2045 Alternatives 1 and 2	10,400,827	4,877,284	0.89	10,935	0.11
2045 Alternatives 3A and 3B	10,395,826	4,872,283	0.89	15,936	0.15

 Table 2-35
 Daily Vehicle Hours of Delay in the Traffic Study Area

MTC Travel Model One (TM1) V6, AECOM 2021b

**Energy Use for Construction.** Project construction would be a temporary commitment of energy needed for any infrastructure improvement project. Energy consumption during construction would be conserved and minimized to the maximum extent feasible. Energy conservation in construction activities is assumed because the construction contractor would have a financial incentive and statutory mandate to minimize waste and externalities, respectively. Regulations that stipulate the reduction of energy-related externalities include CARB Title 13, Section 2485 of California Code of Regulations. This regulation limits the idling time of diesel construction equipment to 5 minutes.

Energy use for construction was calculated based on the project-specific results of the RCEM, as reported in the project's Air Quality Report (AECOM 2021g). The project would involve standard construction techniques and require large-scale construction equipment and labor-intensive activities. Project construction is anticipated to begin in 2023 and last 3 years, with several overlapping phases. Emission factors for construction equipment for the project were based on OFFROAD2011 and EMFAC2017. The analysis conservatively assumes that 100 percent diesel equipment would be used for construction (Caltrans 2021c). The RCEM results were used to determine short-term energy usage for construction, by converting construction-related CO<sub>2</sub> emissions to gallons of diesel consumed, and subsequently converting gallons of diesel to Btu.

The analysis of direct energy use for construction accounted for the Safer Affordable Fuel-Efficient (SAFE) Vehicles rule, by applying the appropriate EMFAC2017 off-model adjustment factor issued by CARB in their June 26, 2020, memorandum (CARB 2020c). Based on a construction year of 2023, an adjustment factor of 1.0126 was applied to projected  $CO_2$  emissions from worker commute vehicles, which were output by the RCEM. The SAFE rule in this context applies to light duty vehicles, such as those driven by construction crews to commute to the job site. The SAFE-adjusted  $CO_2$  emissions were converted to gallons of gasoline consumed by using the

U.S. EPA conversion factor of 8,887 grams of CO<sub>2</sub> per gallon of gasoline consumed (U.S. EPA 2021a). Other CO<sub>2</sub> emissions were converted to gallons of diesel consumed by using the U.S. EPA conversion factor of 10,180 grams of CO<sub>2</sub> emissions per gallon of diesel consumed. Gallons of diesel consumed were then converted to energy use in Btu, by the assumption that a gallon of gasoline has an energy content of 120,941 Btu and a gallon of diesel has 137,320 Btu.

Construction emissions and energy use are summarized in Table 2-36.

Emissions Scenario	CO <sub>2</sub> (Tons)	SAFE Adjusted <sup>1</sup> CO <sub>2</sub> (Metric Tonnes)	Fuel Consumed <sup>2</sup> (Gallons)	Energy Usage (100,000 Btu)
Build Alternative (Gasoline)	4,207.18	3,816.69	429,469.04	519,404.15
Build Alternative (Diesel)	2,295.90	2,082.81	204,597.84	280,953.76
Total	6,503.08	5,899.50	634,066.88	800,357.91

 Table 2-36
 Construction CO<sub>2</sub> Emissions/Energy Usage

Source: AECOM 2021b, RCEM 9.0.0.

Notes:

<sup>1</sup> Uses an adjustment factor of 1.0126.

<sup>2</sup> 10,180 grams of CO<sub>2</sub>/gallon of diesel =  $10.180 \times 10^{-3}$  metric tons CO<sub>2</sub>/gallon of diesel. 8,887 grams of CO<sub>2</sub>/gallon of gasoline =  $8.887 \times 10^{-3}$  metric tons CO<sub>2</sub>/gallon of gasoline.

Through this analysis, it is anticipated that construction of the Build Alternative would require a one-time energy commitment of more than 80 billion Btu.

# Indirect Energy Impacts

Indirect energy use is primarily associated with project maintenance, such as fuel use by equipment for periodic maintenance of the system. Energy use from maintenance would be periodic and is expected to be minimal.

# 2.3.8.4 Avoidance, Minimization, and/or Mitigation Measures

Aside from construction BMPs, no avoidance, minimization, or mitigation is proposed. The proposed project would not lead to impacts stemming from energy use. Therefore, no avoidance, minimization, or mitigation is required.

# 2.4 Biological Environment

# 2.4.1 Natural Communities

The analysis summarized in this section is from the *Natural Environment Study* (NES) prepared in September 2021.

This section of the document discusses natural communities of concern. The focus of this section is on biological communities, not individual plant or animal species. This section also includes information on wildlife corridors, fish passage and habitat

fragmentation. Wildlife corridors are areas of habitat used by wildlife for seasonal or daily migration. Habitat fragmentation involves the potential for dividing sensitive habitat and thereby lessening its biological value. Sensitive natural communities considered here include specific marsh and wetland vegetation communities (salt marsh bulrush marshes, pickleweed mats, and California cordgrass marsh), valley oaks (*Quercus lobata*), fish passage at previously unassessed crossings, and trees.

Regulations relevant to the natural communities discussed include EO 11990 Protection of Wetlands, Senate Bill (SB) 857, and CFGC Section 5981, which provide fish passage protections for anadromous streams (i.e., streams that support fish migration upstream from the sea to spawn); and State Senate Concurrent Resolution No. 17 – Oak Woodlands.

Habitat areas that have been designated as critical habitat under the Federal Endangered Species Act (FESA) are discussed below in Section 2.4.5, Threatened and Endangered Species. Wetlands and other waters are discussed in Section 2.4.2. A discussion on fish passage as it pertains to listed species is included under Section 2.4.1, Natural Communities. Landcover types and biological environmentally sensitive areas are shown in the map book included as Appendix A.

The Biological Study Area (BSA) shown in the Map Set (Appendix A) represents the area that was studied to evaluate all potential direct and indirect project effects on existing sensitive natural resources (such as noise, lighting, and downstream water quality). It includes the physical limits of proposed construction, including all temporary and permanent impact areas; i.e., all construction access and staging areas, traffic staging, parking areas, turnouts, borrow/disposal/stockpile sites, and utility relocation areas. The BSA includes the area of project impacts and a 50-foot buffer from those impacts. The BSA was expanded beyond the project footprint to include areas that were biologically relevant or may be used by Caltrans in the future for restoration purposes (including tree planting).

Natural communities and other land cover types were mapped to listed-species-specific habitats and vegetation communities to quantify potential impacts to biological resources in technical studies and for use in consultation with resource agencies. For this environmental document, landcover types are generalized in the project mapbook to show marsh and wetland communities, riparian communities, surface waters, upland forested communities, upland grassland/shrubland communities, and other land cover types, including upland disturbed, and landscaped vegetation communities (Appendix A).

The classification system of vegetation referred to when discussing plant communities was created by the California Native Plant Society (CNPS), based on alliance qualifications (CNPS 2020). Each alliance is given a global (G) and state (S) ranking. The G ranking indicates the alliance's rarity and threat globally, whereas the S ranking indicates the alliance's rarity and threat in California (CNPS 2020). Alliances marked with G1 through G3 codes indicate rare and threatened throughout their ranges (CNPS 2020). A G3 S3 ranking describes a sensitive natural community with 21 to 100 viable occurrences worldwide and statewide, and/or more than 2,590 to 12,950 hectares (CNPS

2020). In the BSA, the sensitive natural communities that are considered vulnerable (G3 or S3) include ashy ryegrass – creeping ryegrass turfs; California cordgrass marsh; pickleweed mats; salt marsh bulrush marshes; and valley oak woodland.

## 2.4.1.1 Affected Environment

## Sensitive Marsh and Wetlands

Marsh and wetland communities in the BSA were defined using the botanical nomenclature provided in the *Manual of California Vegetation* online database (CNPS 2020). The manual categorizes natural communities based on plant species composition. Sensitive marsh and wetland communities in the BSA include salt marsh bulrush marshes (G4S3), pickleweed mats (G4S3), and California cordgrass marsh (G3S3.2). These sensitive marsh and wetland communities are all associated with either brackish tidal waters or seasonally inundated areas with saline soils.

Wetland communities are spread across the length of the project area, and impacts would occur in narrow strips adjacent to existing disturbed road shoulders and developed roadways. At least one of these natural communities is present in nearly every segment of the BSA, including (from west to east): Tolay Creek channel, Upper Tolay Lagoon, the western halves of Lower and Upper Tubbs Island, Sonoma Creek channel, the Refuge, West End Land Club, Detjen-Fleishhacker Club, NSMWA Ponds 1 and 1A, Strip Marsh, Cullinan Ranch, Cullinan Ranch East, and Mare Island Strait Interchange. The only segments of the BSA lacking any of these communities are upland areas and wetlands with only freshwater inputs; they include the SR 121 Interchange, Sonoma Raceway, Paradise Vineyards, the eastern halves of Lower and Upper Tubbs Island, and Kenwood-BPSC Hunt Club.

#### **Fish Passage**

The proposed project must comply with Section 156.3 of California Streets and Highways Code. The proposed project footprint overlaps three stream crossings, all of which have entries in the CDFW Passage Assessment Database (PAD). In addition to the crossings that overlap with the BSA, CDFW requested in a comment letter on the Notice of Preparation (NOP) (dated August 20, 2020) that a fish passage assessment also be completed for PAD ID 732818 (38.137781, 122.470225). This crossing is outside of both the project footprint and BSA for the project. Relevant information from the PAD on each crossing assessed for the proposed project is included in Table 2-37.

PAD ID	Stream Name	Site Name	Barrier Status	Coordinates
732818*	Unnamed	Culvert SR 37	Unknown	38.137781, -122.470225
761446	Water Tank Cattle Pass	SR 37 Crossing	Unassessed	38.144704, -122.45881
762954	Tolay Creek	SR 37 Crossing	Not a Barrier	38.151707, -122.446944
761318	Sonoma Creek	SR 37 Crossing	Not a Barrier	38.155632, 122.406787

in the PAD
İ

Notes

\* PAD ID 732818 is outside of the project footprint and BSA and is included here at the request of CDFW

During field surveys for the proposed project, the preliminary assessment of the unknown and unassessed barriers was conducted. None of the above structures provided a barrier to anadromous fish, since they are either absent or can pass freely through the structure. A detailed discussion of each of these crossings and the potential for fish passage can be found in Section 2.4.4.2.

# Trees

A total of 329 trees were recorded in the BSA. Trees were mapped if they occurred in or immediately adjacent to the BSA (to account for trees that may have root systems extending into the BSA). The project area is dominated by landscaped California native and nonnative ornamentals. The most common species are Monterey pine (*Pinus radiata*) and red gum (*Eucalyptus camaldulensis*). Nine tree species observed in the BSA are native to California; however, many of the California-native species are not native to the counties where the project is located, including Monterey cypress (*Hesperocyparis macrocarpa*) and Monterey pine. A total of 32 of the surveyed trees are native riparian trees (occurring adjacent to riverine features or wetlands).

## Valley Oaks

State Senate Concurrent Resolution No. 17 – Oak Woodlands is legislation that requests state agencies having land use planning duties and responsibilities to assess and determine the effects of their decisions or actions in any oak woodlands containing Blue, Englemann, Valley, or Coast Live Oak. The measure requests those state agencies to preserve and protect native oak woodlands to the maximum extent feasible or provide replacement plantings where designated oak species are removed from oak woodlands.

Presence of valley oaks in the project area consists of four scattered valley oaks on roadsides of Lower Tubbs Island.

# 2.4.1.2 Environmental Consequences

## Sensitive Marsh and Wetlands

The project would have direct permanent effects on sensitive marsh and wetland communities (Table 2-38) through ground disturbance during construction. Temporary effects on these communities would include direct and indirect impacts, such as vegetation removal and soil disturbance. All temporarily disturbed areas would be restored, such as through hydroseeding and replanting. Some areas would not be replanted because of conversion of natural vegetation communities to paved road. Vegetation removal would be determined during final design.

Community Type	Global and State Ranking	Alternative 1 (acres)	Alternative 2 (acres)	Alternative 3A (acres)	Alternative 3B (acres)
Salt marsh bulrush marshes	G4S3	0.03	0.04	0.04	0.11
Pickleweed mats	G4S3	1.65	3.04	3.59	7.19
California cordgrass marsh	G3S3.2	0.23	0.21	0.22	0.29
Total		1.91	3.29	3.85	7.59

Table 2-38	Estimated Permanent Effects on Sensitive Marsh and Wetland
	Communities

All sensitive marsh and wetland vegetation communities are mostly within state and federally regulated waters jurisdictions. Further discussion of impacts to wetlands and other waters of the U.S. and state is provided in Section 2.4.2.

#### Fish Passage

Culvert SR 37 on Unnamed Drainage (PAD ID 732818) and SR 37 Crossing on Water Tank Cattle Pass (PAD ID 761446)

There are no California Natural Diversity Database (CNDDB) occurrences of anadromous fish in either of the drainages that lead to Culvert SR 37 on Unnamed Drainage (PAD ID 732818) and SR 37 Crossing on Water Tank Cattle Pass (PAD ID 761446), nor are there historical or current records of anadromous fish in these drainages in published literature (Leidy et al. 2005, SRCD 2015, NMFS 2016b). In addition, there is no mention of steelhead or other anadromous fish currently or historically occurring in these drainages in the Coastal Conservancy land acquisition recommendation for the North Point Joint Venture Acquisition, in which the drainages are located, despite the recommendation listing special-status species found on the property (Coastal Conservancy 2004). The parcel that drains to the crossings was acquired by the Sonoma Land Trust in 2005 and is now called Sears Point Ranch. The Sonoma Land Trust describes Sears Point Ranch as containing wetlands and seasonal creeks and discusses other special-status species occurrences but does not mention historical or current anadromous fish occurrences in Sears Point Ranch (Sonoma Land Trust 2021).

Based on the lack of any historical or current occurrence records of anadromous fish in the drainages in which Culvert SR 37 on Unnamed Drainage (PAD ID 732818) and SR 37 Crossing on Water Tank Cattle Pass (PAD ID 761446) are located, there is no evidence that anadromous fish are, or historically were, found in these drainages. Therefore, SB 857 does not require Caltrans to complete a fish passage assessment at these crossings.

# SR 37 Crossing on Tolay Creek (PAD ID 762954)

The SR 37 Crossing on Tolay Creek (PAD ID 762954) is not a barrier; therefore, an additional fish passage assessment at this crossing is not necessary.

## SR 37 Crossing on Sonoma Creek (PAD ID 761318)

A site visit was conducted at the Sonoma Creek crossing (PAD ID 761318) on September 3, 2020. Based on this site visit and a review of aerial imagery, a greengray-red first-phase fish passage evaluation was completed following the methods described in Part IX: Fish Passage Evaluation at Stream Crossings of the California Salmonid Stream Habitat Restoration Manual (Taylor and Love 2003).

There appears to be fine substrate throughout the channel under the bridge. The active channel width is approximately 395 feet, measured using aerial imagery approximately 2,000 feet downstream of the bridge. The inlet width (i.e., bridge length) is approximately 1,830 feet. Because there is streambed substrate throughout the channel under the bridge and the inlet width is greater than the active channel width, the first-phase fish passage evaluation filter results in a green classification, meaning it is considered passable for all salmonid life stages.

A Fish Passage Incidental Report (First Pass Data Sheet) for SR 37 Crossing on Sonoma Creek would be submitted by Caltrans to the PAD.

## Trees

The project would have direct and indirect permanent effects on 125 trees (118 nonnative and seven native trees) including removal of some trees. An area with a radius of approximately 10 feet may be impacted around each tree that would be removed. Any paving in the critical root zone of the tree would be considered a permanent impact to the tree because this action may result in tree mortality. Temporary effects would result from minor tree trimming or staging of equipment in the critical root zone.

Areas subject to paving would not be replanted and some trees would not be replanted because of the need to maintain a clear recovery zone near the shoulders for driver safety. All trees in areas that are permanently impacted presumably would be removed; however, trees in and adjacent to temporarily impacted areas may not need to be removed. The exact number of trees to be removed would depend on field conditions, such as the geology of the area where cut slopes are excavated, condition of trees, location of supporting roots, and other considerations to ensure the post-construction stability of permanent structures. Tree removal would be determined during final design.

## Valley Oaks

Four valley oak trees (*Quercus lobata*) occur along the eastbound roadside of SR 37 on Tubbs Island. Project activities may cause direct and indirect impacts on valley oaks in the project footprint. Temporary impacts include pruning of less than 30 percent of the canopy, removal of less than 25 percent of the roots (within the drip line of the tree), or soil compaction to less than 30 percent of the critical root zone. Any paving in the critical root zone of the tree would be considered a permanent effect on the tree. Temporary effects would result from minor tree trimming or staging of equipment in the critical root zone.

Trees in and adjacent to temporarily affected areas may not need to be removed. The exact number of trees to be removed would depend on field conditions and would be determined during final design phase.

#### 2.4.1.3 Avoidance, Minimization, and/or Mitigation Measures

#### Sensitive Marsh Wetlands

General and specific AMMs would be implemented for all considered Build Alternatives to avoid and minimize potential impacts to sensitive marsh and wetlands. Additional specific measures proposed to address potential impacts to water quality in Section 2.3.2 and to wetlands and other waters in Section 2.4.2 would also serve to avoid and minimize impacts to sensitive marsh wetlands. Measures that would avoid and minimize impacts to sensitive marsh wetlands include:

**BIO-01: Wetlands Protection – Invasive Plants.** To prevent the introduction of NNIP species such as smooth cordgrass (*Spartina alterniflora* and hybrids), stinkwort (*Dittrichia graveolens*), and prickly Russian thistle (*Salsola tragus*) into areas of tidal vegetation during construction and restoration activities, the following measures will be implemented:

- The project biologist will conduct a NNIP assessment of areas subject to construction activities and will recommend specific measures to minimize the spread of NNIP species.
- Wetland areas that are temporarily disturbed will be monitored. All NNIP infestations discovered in the project area in wetland habitats will be controlled and removed upon discovery.
- A long-term (5 years after project completion) vegetation monitoring plan for post-disturbance impacts in wetlands will be developed in coordination with USFWS and CDFW and implemented by Caltrans.

**BIO-02: Wetland Protection.** The following measures will be implemented in and adjacent to delineated wetland environmentally sensitive areas in the project area:

- Work in and adjacent to delineated wetlands where flooding has potential to occur will be scheduled outside of the wet-weather season.
- Work in and adjacent to delineated tidal wetlands will not occur within 2 hours before or after extreme high-tide events (6.5 feet above mean lower low water elevation or greater, as determined from the nearest National Oceanic and Atmospheric Administration tidal gage station to the activity) when the marsh plain is inundated.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### Fish Passage

All crossings assessed in the BSA and the additional crossing assessed outside of the proposed project area do not present barriers to fish passage. No measures or mitigation for fish passage are proposed.

#### Trees

**BIO-03: Tree Replacement, Landscaping, and Revegetation Plan.** During final design, Caltrans will develop a landscaping plan that will identify the location and number of trees that will be replanted in the right-of-way. Locally appropriate native species will be used to the maximum extent possible, and trees, shrubs, and groundcover will be selected for drought tolerance and disease resistance. Mulch will be applied to planted areas to reduce weed growth, conserve moisture, and minimize maintenance operations. A 3-year plant establishment period will be included in the final revegetation plan. Caltrans will develop and implement a 5- to 10-year post-construction vegetation monitoring plan for planted areas.

#### Valley Oaks

The same measures proposed for trees would also protect valley oak trees in the project area, and no additional measures or mitigation are proposed.

#### 2.4.2 Wetlands and Other Waters

## 2.4.2.1 Regulatory Setting

Wetlands and other waters are protected under several laws and regulations. At the federal level, the Federal Water Pollution Control Act, more commonly referred to as the CWA (33 USC 1344), is the primary law regulating wetlands and surface waters. One purpose of the CWA is to regulate the discharge of dredged or fill material into waters of the U.S., including wetlands. Waters of the U.S. include navigable waters, interstate waters, territorial seas, and other waters that may be used in interstate or foreign commerce. The lateral limits of jurisdiction over nontidal water bodies extend to the ordinary high water mark (OHWM), in the absence of adjacent wetlands. When adjacent wetlands are present, CWA jurisdiction extends beyond the OHWM to the limits of the adjacent wetlands. To classify wetlands for the purposes of the CWA, a three-parameter approach is used that includes the presence of hydrophytic (water-loving) vegetation, wetland hydrology, and hydric soils (soils formed during saturation/inundation). All three parameters must be present, under normal circumstances, for an area to be designated as a jurisdictional wetland under the CWA.

Section 404 of the CWA establishes a regulatory program that provides that discharge of dredged or fill material cannot be permitted if a practicable alternative exists that is less damaging to the aquatic environment or if the nation's waters would be significantly degraded. The Section 404 permit program is run by USACE with oversight by U.S. EPA.
The USACE issues two types of 404 permits: General and Individual. There are two types of General permits: Regional and Nationwide. Regional permits are issued for a general category of activities when they are similar in nature and cause minimal environmental effect. Nationwide permits are issued to allow a variety of minor project activities with no more than minimal effects.

Ordinarily, projects that do not meet the criteria for a Regional or Nationwide Permit may be permitted under one of USACE's Individual permits. There are two types of Individual permits: Standard permits and Letters of Permission. For Individual permits, the USACE decision to approve is based on compliance with U.S. EPA's Section 404(b)(1) Guidelines (40 CFR Part 230 [available at https://www.epa.gov/cwa-404/cwa-section-404b1-guidelines-40-cfr-230]), and whether permit approval is in the public interest. The Section 404 (b)(1) Guidelines (Guidelines) were developed by the U.S. EPA in conjunction with the USACE, and allow the discharge of dredged or fill material into the aquatic system (waters of the U.S.) only if there is no practicable alternative which would have less adverse effects. The guidelines state that the USACE may not issue a permit if there is a LEDPA to the proposed discharge that would have lesser effects on waters of the U.S., and not have any other significant adverse environmental consequences.

The Executive Order for the Protection of Wetlands (EO 11990) also regulates the activities of federal agencies with regard to wetlands. Essentially, EO 11990 states that a federal agency, such as FHWA and/or the Department, as assigned, cannot undertake or provide assistance for new construction in wetlands unless the head of the agency finds: (1) that there is no practicable alternative to the construction and (2) the proposed project includes all practicable measures to minimize harm. A Wetlands Only Practicable Alternative Finding must be made.

At the state level, wetlands and waters are regulated primarily by the SWRCB, the RWQCBs and CDFW. In certain circumstances, the Coastal Commission (or BCDC or the Tahoe Regional Planning Agency) may also be involved. Sections 1600-1607 of CFGC require any agency that proposes a project that would substantially divert or obstruct the natural flow of or substantially change the bed or bank of a river, stream, or lake to notify CDFW before beginning construction. If CDFW determines that the project may substantially and adversely affect fish or wildlife resources, a Lake or Streambed Alteration Agreement would be required. CDFW jurisdictional limits are usually defined by the tops of the stream or lake banks, or the outer edge of riparian vegetation, whichever is wider. Wetlands under jurisdiction of the USACE may or may not be included in the area covered by a Streambed Alteration Agreement obtained from the CDFW.

The RWQCBs were established under the Porter-Cologne Water Quality Control Act to oversee water quality. Discharges under the Porter-Cologne Act are permitted by WDRs and may be required even when the discharge is already permitted or exempt under the CWA. In compliance with Section 401 of the CWA, the RWQCBs also issue water quality certifications for activities which may result in a discharge to waters of the

U.S. This is most frequently required in tandem with a Section 404 permit request. Please see the Water Quality and Stormwater Runoff section for more details.

The analysis in this section relies on draft Caltrans Aquatic Resource Delineation Report from July 2021 and draft NES (AECOM 2021i) from June 2021 to determine the extent of jurisdictional waters and wetlands and potential impacts in the project BSA for each considered alternative.

## 2.4.2.2 Affected Environment

The analysis summarized in this section is from the NES prepared in September 2021.

An estimated total of 75.17 acres of tidal wetlands, tidal other-waters, and nontidal wetlands and nontidal other-waters were identified in the BSA. Wetlands and waters that occur in the project's BSA are shown generally in Appendix A. Wetlands observed in the BSA include brackish marsh, tidal marsh, freshwater emergent marsh, forested/ shrub wetland, and seasonal wetland. Other waters in the BSA include intermittent streams, perennial streams, tidal creeks, tidal ponds, tidal sloughs, and tidal ponds.

Most of the BSA has been constructed over historical tidal wetlands, which are defined as wetlands and aquatic areas open to tidal influence below the topographic contour that corresponds to the maximum possible extent of the tides; the tidal boundary that would be observed during the highest tide of the current tidal epoch, if there were no levees, dikes, flood gates, or other unnatural obstructions to the landward flow of tidal water (SFEI 2011). Types of tidal wetlands in the BSA include brackish marsh, salt flats, and mud flats. In tidal wetlands, brackish open-water features were observed, including brackish tidal ponds and salt ponds. Open water features were delineated as other waters of the United States. Freshwater emergent wetlands observed in the BSA include cattail channels and freshwater marsh, isolated to agricultural areas. Freshwater marshes are nontidal, flooded, depressional wetlands. Vegetation dominating these areas include annual and perennial emergent species, such as cattails, bulrush, and sedges.

Named creeks and wetland features in the BSA include Tolay Creek and Sonoma Creek. Named creeks and wetland features adjacent to the BSA include the Napa River as it turns into the Mare Island Strait. All of these waterways flow south toward San Pablo Bay. Named confluences that occur north of the BSA and flow generally east and west, depending on tidal influence, are Dutchman Slough, South Slough, China Slough, and Napa Slough. Unnamed sloughs meander naturally through the marshes along the southern side of SR 37; unnamed marsh sloughs along the northern side of SR 37 are all artificially altered by dikes, levees, and excavated channels.

Riparian scrub and forest (or mixed riparian forest) habitat is a multi-alliance assemblage of wetland and riparian trees and shrubs that narrowly line the western edge of the marsh depression of the historic Tolay Creek channel, running north along railroad tracks parallel to SR 121. This community also occurs along several other riverine features in the BSA. The willow riparian habitat is characterized by willows (*Salix* sp.) in the shrub or tree canopy (CNPS 2020). These vegetation communities occur along stream banks and benches, slope seeps, and stringers along drainages with plants such as coyote brush (*Baccharis pilularis*) and black elderberry (*Sambucus nigra*) (CNPS 2020). These communities are state-jurisdictional under Section 401 of the CWA and are considered sensitive habitat by CDFW. The sensitive habitat status is due to the high value of riparian habitat to wildlife and the relatively limited (and declining) distribution of this habitat at the local and statewide level.

Ten potentially jurisdictional culverted waters of the U.S. were observed during field surveys and from aerial imagery. These features are potentially jurisdictional due to connectivity to jurisdictional wetlands and other waters. There are no anticipated impacts to these features, and they are not discussed further in this analysis.

## 2.4.2.3 Environmental Consequences

The project is anticipated to have permanent impacts from placement of permanent fill for road widening, retaining walls, sheet piles, rock slope protection, and placement of guard rails. Under Alternative 3B, additional permanent impacts to wetlands and waters would be realized through bridge widening work at Sonoma Creek and would permanently shade additional areas below the widened bridge. The primary permanent impact under all scenarios is associated with road widening; the alternatives with greater shoulder widths added directly correlate to an increase in permanent impacts on wetlands and waters. Temporary impacts are associated with construction access, staging areas, and temporary dewatering activities with potential to temporarily impact wetlands and waters.

The wetlands and other waters that are potentially impacted under all build alternatives are spread across the length of the project area, and most impacts would occur in narrow strips adjacent to existing disturbed road shoulders and developed roadways. The potentially impacted areas border thousands of acres of tidal waters and wetlands, and are anticipated to be relatively low when considered in this greater context. Alternative 3B has a substantially greater area of impact because of the substantially wider roadway proposed under that alternative. Although the project would cause temporary and permanent impacts to the intertidal areas, it has been designed to minimize fill and turbidity to the greatest extent feasible. Preliminary estimates of permanent and temporary fill impacts for all alternatives, as well as shading impacts from Alternative 3B, are shown in Table 2-39 and Table 2-40, respectively.

Habitat Type	Alternative 1 Permanent fill (Acres)	Alternative 2 Permanent fill (Acres)	Alternative 3A Permanent fill (Acres)	Alternative 3 B Permanent fill (Acres)	Alternative 3B Shading (Acres)
Tidal wetlands	1.52	3.2	3.75	7.33	0.52
Nontidal wetlands	0.45	0.11	0.27	0.44	0.00
Subtotal Wetlands	1.97	3.31	4.02	7.77	0.52
Tidal Other-Waters	0.03	0.16	0.24	1.27	0.17
Nontidal Other-Waters	0.03	0.02	0.02	0.01	0.01
Subtotal Other-Waters	0.06	0.18	0.26	1.28	0.18
Total Wetlands and Other Waters	2.03	3.49	4.28	9.05	0.70

# Table 2-39Preliminary Estimated Permanent Impacts to Wetland and OtherWaters

Note: All values were rounded up to the nearest 0.01 acre.

# Table 2-40Preliminary Estimated Temporary Impacts to Wetlands and Other<br/>Waters

Habitat Type	Alternative 1 Temporary Impacts (Acres)	Alternative 2 Temporary Impacts (Acres)	Alternative 3A Temporary Impacts (Acres)	Alternative 3B Temporary Impacts (Acres)	Alternative 3 B Shading (Acres)
Tidal wetlands	3.3	5.59	6.29	7.28	0.87
Nontidal wetlands	1.19	0.97	0.85	0.97	0
Subtotal Wetlands	4.49	6.56	7.14	8.25	0.87
Tidal Other-Waters	1.43	3.35	3.09	4.63	0.87
Nontidal Other-Waters	0.42	0.11	0.12	0.20	0.02
Subtotal Other-Waters	1.85	3.46	3.21	4.83	0.89
Total Wetlands and Other Waters	6.34	10.02	10.35	13.08	1.76

Note: All values were rounded up to the nearest 0.01 acre.

# 2.4.2.4 Wetlands Only Practicable Alternative Finding

As summarized in Section 2.4.2.1, in compliance with EO 11990, Caltrans cannot undertake construction in wetlands unless there are no practicable alternatives to the construction; the proposed project includes all practicable measures to minimize harm.

The Caltrans PDT has determined that a build alternative is required to meet the project's purpose and need to address traffic congestion on SR 37 due to capacity constraints where the highway narrows down to one lane in each direction. The Caltrans PDT carefully weighed the benefits and drawbacks of each build alternative. They evaluated the extent of environmental impacts of each alternative, and how each one met the project's purpose.

As discussed in Section 1.4.3.1, Identification of a Preferred Alternative, Alternative 3B has been identified as the preferred alternative. Alternative 3B was determined to best serve the project's purpose of relieving traffic congestion by adding a permanent lane on each side of the highway, while also including necessary 8-foot shoulders to be used by CHP, emergency vehicles, breakdowns, and bicyclists. A design exception is being made at the Sonoma Creek Bridge, where the shoulders will be 4 feet wide, which still allows for bicycle access, but reduces the extent of construction at this bridge. The other build alternatives considered by the PDT (1, 2, and 3A) did not best meet the project's purpose and need (Alternatives 1 and 2), and/or did not meet safety or introduced operational challenges. These included that Alternatives 2 and 3A would not provide acceptable safety due to narrow shoulders; and following input from the CHP and consideration of Caltrans design standards for safety, were determined not acceptable. Alternative 1 would require necessary openings along the median barrier for emergency service access (as needed by CHP and emergency responders), with the openings complicating the movement of the barrier and potentially conflicting with traffic turning at the openings during barrier transfer operations. The operation and maintenance requirements of the movable barrier was determined not acceptable by Caltrans maintenance and operations. Additional explanation is provided in Section 1.4.3.1.

Alternative 3B has the greatest temporary and permanent wetland impacts compared to the other considered build alternatives, as shown in Table 2-38 and Table 2-39. The other build alternatives had fewer acres of wetland impacts, but as noted above and in Section 1.4.3.1, were not determined practicable, and/or did not meet the project's traffic improvements as well as Alternative 3B. The project design has been refined to minimize impacts to wetlands to the maximum extent possible, including minimizing the originally proposed work in Sonoma Creek by eliminating the need for any new permanent piers (as was originally proposed in the Draft EIR/EA for Alternative 3B). Measures have also been included to minimize the effects to wetlands during construction; these are described in Section 2.4.2.5.

Based on the above considerations, it is determined that no practicable alternative exists to the proposed construction in wetlands, and that the proposed action includes all practicable measures to minimize harm to wetlands that may result from such use.

### 2.4.2.5 Avoidance, Minimization, and/or Mitigation Measures

All proposed measures under Section 2.4.1 for Sensitive Marsh and Wetland Communities (**BIO-01: Wetlands Protection – Invasive Plants**; and **BIO-02: Wetland Protection**) would also serve to protect wetlands and other waters for all Build Alternatives considered. All measures proposed for Water Quality in Section 2.3.2.4 would also serve to protect wetlands and other waters.

In addition to those measures, the following measures and compensation are proposed:

**BIO-04: Estuarine Dewatering Work Window.** In-water work requiring dewatering in tidal waters will be scheduled to occur between June 1 and November 30. Other work below MHHW (excluding impact pile driving) may be done year-round.

**BIO-05:** Turbidity Control. During the expansion of the Tolay Creek Bridge abutments and at other locations where ground disturbance would be conducted below MHHW, a silt-curtain, sheet pile or gravel-bag cofferdam, or other equivalent means will be installed as needed to minimize the generation of turbidity plumes in nearby tidal waters. Such cofferdams would be installed when there is no surface water present (i.e., at low tide). This requirement does not apply to in-water pile driving.

# Proposed Compensation to Offset Loss of Wetlands and Other Waters

Caltrans proposes to compensate for permanent losses to wetlands and waters for the selected alternative at a 3:1 restoration/enhancement to impact ratio, or at a ratio determined appropriate in coordination with regulatory agencies with jurisdiction, through a project-specific compensation plan. This would include the use of an available conservation bank to the extent that credits are applicable and available, support of off-site restoration projects and programs, and restoration of onsite resources that are temporarily impacted by project construction.

Near the project area, there is currently one approved conservation bank, Burdell Ranch Wetland Mitigation Bank, with credits available for approximately 0.8 acre of non-tidal wetlands (1 credit=0.10 acre) impacts only (no tidal wetland credits are available). Each Alternative has different estimated permanent impact areas to non-tidal wetlands, and only Alternative 2 (which has the lowest anticipated impact area to non-tidal wetlands) would provide a feasible pathway for compensation through this bank (assuming a 3:1 ratio). Alternatives 1, 3A, and 3B would impact more non-tidal wetlands area than are available credits in the area with the same assumed ratio. No banks for tidal wetlands and tidal other waters are available near the project area. Additionally, there are no approved in-lieu fee programs to compensate for impacts to wetlands and other waters.

Caltrans proposes to offset impacts and losses to wetlands and other waters through a project specific plan that would fund nearby tidal restoration and enhancement efforts within the project's watershed(s). This would be achieved through coordination with specific restoration project owners, state and federal environmental regulators with

jurisdiction to determine appropriate funding targets, define appropriate endowments, and develop an in-lieu-fee program specific to the project. Caltrans has identified several potential projects that could be funded to offset and compensate for loss of wetlands and other waters from the selected alternative. These include efforts in the Refuge (Mare Island, Cullinan Ranch, Strip Marsh, Skaggs Island, or Tolay Lagoon), or efforts being conducted through the Sonoma Creek Baylands Strategy (Sonoma Creek Restoration at Detjen and West End) sponsored by the Sonoma Land Trust. Funding and transfers could be established through a co-operative agreement with the California State Coastal Conservancy.

Caltrans would restore temporarily disturbed wetlands and other water areas to pre-project conditions on site at a 1:1 ratio.

The following measures summarize Caltrans commitment to offset impacts to wetlands and other waters based on the preferred alternative:

**BIO-07: Wetlands and Other Waters Compensation**. Caltrans will pursue efforts to restore, enhance, and/or create new wetlands and waters on-site at Tolay Creek to partially offset permanent loss of waters from the project. To offset any remaining permanent loss and habitat degradation of wetlands and other waters in the project area, Caltrans will provide in-lieu fee-based compensation at a 3:1 restoration/enhancement to impact ratio, or at a ratio determined appropriate in coordination with regulatory agencies with jurisdiction. Compensation will be provided through a project-specific plan that would provide in-lieu funding to a nearby restoration program or restoration project that would create, restore, or enhance resources adversely affected by the project. Appropriate compensation will be determined in coordination with state and federal environmental regulatory agencies with jurisdiction.

Caltrans will offset temporary impacts during construction to wetlands and other waters by restoring disturbed areas to pre-project conditions at a 1:1 ratio.

### 2.4.3 Plant Species

### 2.4.3.1 Regulatory Setting

USFWS and CDFW have regulatory responsibility for the protection of special-status plant species. "Special-status" species are selected for protection because they are rare and/or subject to population and habitat declines. Special status is a general term for species that are provided varying levels of regulatory protection. The highest level of protection is given to threatened and endangered species; these are species that are formally listed or proposed for listing as endangered or threatened under FESA and/or the California Endangered Species Act (CESA). Please see the Threatened and Endangered Species section in this document for detailed information about these species.

This section of the document discusses all other special-status plant species, including CDFW species of special concern (SSC), USFWS candidate species, and CNPS rare and endangered plants.

The regulatory requirements for FESA can be found at 16 USC Section 1531, et seq. See also 50 CFR Part 402. The regulatory requirements for CESA can be found at CFGC, Section 2050, et seq. Department projects are also subject to the Native Plant Protection Act, found at CFGC, Section 1900-1913, and CEQA, found at California PRC, Sections 21000-21177.

## 2.4.3.2 Affected Environment

Information analysis provided in this section is based on the Caltrans September 2021 draft NES to identify protected plant species in the area.

In addition to literature and desktop reviews of potential resources in the project area, Caltrans conducted extensive site surveys to determine species presence and potential to occur in the BSA, as summarized in Table 2-41.

Survey Type	Personnel and Agency/Firm	Survey Dates
Rare Plant Survey and Vegetation Community Mapping	Danny Slakey (AECOM botanist) Sunshine Lopez (AECOM botanist) Saana Deichsel (AECOM ecologist)	September 23, 24, 26, and 27, 2019; December 17, 2019, March 13, 2020; and March 19, 2021
Rare Plant Reference Population Site Visit	Danny Slakey Sunshine Lopez	October 4, 2019

Table 2-41Plant Survey Dates

A complete list of special-status plants considered and a summary of the determinations on their potential occurrence or potential to occur is presented in Appendix I. Three special-status plant species—soft bird's-beak (*Chloropyron molle* ssp. *molle*), San Joaquin spearscale (*Extriplex joaquinana*), and saline clover (*Trifolium hydrophilum*)—were determined to have at least a moderate potential to occur in the BSA. The federally listed soft bird's-beak is also discussed under Section 2.4.5.

Some special-status plants may not have been detected if present in the BSA because floristic surveys were conducted in September and December 2019, after the blooming periods for some potentially occurring special-status plants. The potential for occurrence of these species and potential effects the project may have on them are discussed in this section. Other special-status species were evaluated for their potential to occur in the BSA, as discussed in Appendix I, but are not discussed here because either (1) they would have been identifiable at the time of the rare plant survey and were not detected or (2) they have a very low potential to occur in the BSA, or both. One rare plant species, the holly-leaved ceanothus (*Ceanothus purpureus*), was observed in the BSA. Although holly-leaved ceanothus was tentatively identified in the BSA, it is not treated as a special-status plant because it is a landscape planting outside of its known habitat and geographic range. One special-status plant, small spikerush (*Eleocharis parvula*) was observed in the BSA. Its rare plant ranking indicates that it is a watch list species with a low threat level (less than 20 percent of occurrences threatened). The plant was locally common in pickleweed mats and the upper edges of mud flats over a 1.8-mile stretch of the Cullinan Ranch segment north of SR 37, with more than 1,000,000 plants likely present. Because of its abundance in the BSA, the lack of impacts of the project to this species, its low threat level in California, and its widespread distribution beyond California, it is not further discussed in this analysis. No other natural populations of special-status plants were identified on floristic surveys of the BSA.

### Soft Bird's-Beak

Soft bird's-beak was not observed during rare plant surveys conducted in the BSA in late September 2019. Because its flowering period extends into November, it would likely have been observed during the rare plant survey if there were populations present in the BSA. There are no CNDDB occurrences of this plant in the BSA.

Soft bird's-beak is a federally endangered, California State Rare, and California Rare Plant Rank (CRPR) 1B.2 herbaceous annual plant in the broomrape family (Orobanchaceae) (CNPS 2021). Soft bird's-beak is hemiparasitic (i.e., it gets its water and nutrients from the roots of another plant, but also makes food through photosynthesis) and grows with a broad range of host plants that are actively growing during its flowering and fruiting stages. Known hosts include common pickleweed, salt grass, and fleshy jaumea. Soft bird's-beak is restricted to coastal salt marshes habitats, where it occurs in colonies or subpopulations that can shift from year to year (USFWS 2013a).

The CNDDB occurrences of soft bird's-beak nearest to the BSA are about 1 mile from the BSA, but both are considered to no longer exist at those locations. The nearest extant occurrence is about 5 miles north of the BSA. Although the species is considered absent from nearby locations where it was previously known, coastal salt marsh habitats in the BSA with high cover of suitable host species could support soft bird's-beak. This community occurs in nearly all segments of the BSA east of the SR 121 interchange, including the Tolay Creek channel, Upper Tolay Lagoon, Upper and Lower Tubbs Island, Sonoma Creek channel, West End Land Club, Detjen-Flyshacker, State Land Commission (SLC)-leased Refuge, NSMWA Ponds 1 and 1a, Cullinan Ranch, Cullinan Ranch East, Strip Marsh, and the Mare Island Strait Interchange.

### San Joaquin Spearscale

No occurrence of San Joaquin spearscale was observed in the BSA during floristic surveys, nor are there any recorded CNDDB occurrences of this species in the BSA.

Most suitable habitat for this species in the BSA occurs outside of areas where direct construction impacts would occur

San Joaquin spearscale is a CRPR 1B.2 herbaceous annual plant in the goosefoot family (Chenopodiaceae) (CNPS 2021). This plant typically flowers from April through October and usually occurs in uplands, but it can occasionally occur in wetlands. San Joaquin spearscale grows in areas with alkaline soils, including chenopod scrub, meadows and seeps, playas, and valley and foothill grassland habitats (CNPS 2021).

Two CNDDB occurrences are known from approximately 3 miles north of the BSA. One of these was last observed in 1991 along the upper edges of a pickleweed marsh (CDFW 2019). In the BSA, seasonal wetlands with alkaline soils, and areas transitional between brackish marshes and uplands could support populations of San Joaquin spearscale. Transitional areas between brackish pickleweed marshes and uplands are common throughout the BSA but are often highly disturbed and dominated by nonnative invasive poison hemlock and fennel. Seasonal alkaline wetlands are found in several areas in the BSA, such as at the Mare Island Strait Interchange on the eastern end of the BSA. Species commonly associated with San Joaquin spearscale, such as salt grass, alkali heath, and pickleweed (CDFW 2019) are common in both seasonal alkaline wetlands and marsh/upland transition areas in the BSA.

Based on the presence of potentially suitable habitat in much of the BSA (except areas with only upland ruderal vegetation) San Joaquin spearscale has a moderate potential to be present. Given that rare plant surveys were conducted in late September, toward the end of the plant's known flowering period, San Joaquin spearscale may have been present in the BSA but not detected during the rare plant survey. The small plants, if present, may have been undetectable at the time the survey was conducted.

### Saline Clover

Saline clover was not observed in the BSA during floristic surveys, nor are there any recorded occurrences of this species in the BSA found in the CNDDB. Most suitable habitat in the BSA for this species occurs outside of areas where direct construction impacts would occur. If present, saline clover would not have been detected during botanical surveys, because the surveys were conducted well after the plant's known flowering period (April to June).

Saline clover is a CRPR 1B.2 annual herb in the pea family (Fabaceae) (CNPS 2021). Saline clover occurs in marshes and swamps, vernal pools, and mesic grasslands, and sometimes in areas with alkaline soils (CDFW 2019, CNPS 2021). Most occurrences of this plant are known from seasonally inundated habitats, such as mesic grasslands and vernal pools. Saline clover flowers from April to June (CNPS 2021).

The nearest recorded CNDDB occurrence of saline clover is approximately 1 mile west of the BSA and approximately 250 feet north of SR 37. There are three additional occurrences of saline clover within 5 miles of the BSA. Some existing habitat in the

BSA could support this species, particularly at the upper edge of marshes where they transition to salt grass flats, coyote brush scrub, or ruderal habitats.

## 2.4.3.3 Environmental Consequences

The project is unlikely to have any direct or indirect impacts on San Joaquin spearscale, or Saline clover because most potentially suitable habitat in the BSA for these plants occurs outside of areas likely to be impacted by the project. Botanical surveys were conducted for this project covering all areas proposed for construction and were not detected. There is a small potential for both of these species to occur because there is a low risk they may not have been observed during the floristic surveys conducted for the project or they have the opportunity to occur between the time of surveys and start of construction. Construction activities could have adverse effects in the unlikely event that individual plants of either species are present in the project footprint.

## 2.4.3.4 Avoidance, Minimization, and/or Mitigation Measures

To avoid and minimize potential impacts to special-status plant species, the following measures are proposed:

**BIO-08: Targeted Pre-Construction Plant Survey.** During final project design, an experienced botanist will conduct a final floristic survey in the project area during the appropriate blooming period for all special-status plant species that have potential to occur but were not surveyed for previously. The survey does not need to cover the flowering period for species adequately surveyed for during the September 2019 surveys. Surveys should be conducted following the same protocols from the September 2019 surveys, *Protocols for Surveying and Evaluating Impacts to Special-Status Native Plant Populations and Sensitive Natural Communities,* prepared by CDFW, dated March 20, 2018. If special-status plant species are discovered, they will be included as an environmentally sensitive area in project plans and specifications. If any listed species are discovered that could be impacted by project activities, Caltrans will consult with state and federal regulators with jurisdiction, as appropriate.

**BIO-09: Special-Status Plant Monitoring.** If a special-status plant is discovered during construction monitoring in an area where ground-disturbing activities are proposed, it will be marked or fenced for avoidance with a 10-foot buffer. Ground-disturbing work near special-status plant species will proceed under the supervision of a project biologist.

# 2.4.4 Animal Species

# 2.4.4.1 Regulatory Setting

Many state and federal laws regulate impacts to wildlife. USFWS, NMFS, and CDFW are responsible for implementing these laws. This section discusses potential impacts and permit requirements associated with animals not listed or proposed for listing under the FESA or CESA. Species listed or proposed for listing as threatened or

endangered are discussed in the Threatened and Endangered Species Section 2.4.5 below. All other special-status animal species are discussed here, including CDFW fully protected species and SSCs, and USFWS or NMFS candidate species.

Federal laws and regulations relevant to wildlife include the following:

- NEPA
- Migratory Bird Treaty Act (MBTA)
- Fish and Wildlife Coordination Act
- Marine Mammal Protection Act

State laws and regulations relevant to wildlife include the following:

- CEQA
- Sections 1600 1603 of CFGC
- Sections 4150 and 4152 of CFGC

CDFW has administratively designated certain species that are not CESA-listed with special status for greater consideration during CEQA review. According to CDFW, a SSC is a species, subspecies, or distinct population of an animal (i.e., fish, amphibian, reptile, bird, and mammal) native to California that currently satisfies one or more of the following (not necessarily mutually exclusive) criteria (criteria for fishes are similar except that federally listed taxa are not defined as SSCs):

- is extirpated from the state or, in the case of birds, is extirpated in its primary season or breeding role;
- is listed as federally but not state-threatened or endangered; meets the State definition of threatened or endangered but has not formally been listed;
- is experiencing, or formerly experienced, serious (noncyclical) population declines or range retractions (not reversed) that, if continued or resumed, could qualify it for State threatened or endangered status;
- has naturally small populations exhibiting high susceptibility to risk from any factor(s), that if realized, could lead to declines that would qualify it for State threatened or endangered status.

CDFW SSC that are not listed under FESA are reviewed in this section.

### 2.4.4.2 Affected Environment

The analysis summarized in this section is from the NES.

All special-status wildlife species with potential to occur in the project area are summarized in this section. Although no protocol-level surveys were conducted, biologists conducted assessments to determine the suitability of the habitat in the BSA to host the wildlife species listed in Appendix I, and to confirm their potential to occur determinations. American peregrine falcon, California least tern, northern harrier, western snowy plover, and white-tailed kite are known to occur or have a high potential to occur but do not have potential to nest in the BSA due to lack of suitable nesting habitat. Though they may fly through and/or forage inside or in the vicinity of the BSA, direct and indirect effects on these species from project activities are not anticipated or would be negligible, and are not discussed further in the document. Protected species that are not listed under FESA or CESA, their habitat requirements, and potential impacts of the project are discussed in detail in this section. CESA- and FESA-listed species with potential to occur in the project area are noted here, and potential impacts are discussed in Section 2.4.5.

Six SSCs by CDFW have some potential to occur in the BSA: pallid bat, Saltmarsh Common Yellowthroat, San Pablo song sparrow, Sacramento Splittail, Suisun shrew, and Western Burrowing Owl. California red-legged frog and North American green sturgeon are also state SSCs; however, these three species are also listed under the FESA and are discussed in Section 2.4.5. All remaining species that are known to occur in the region were determined to be absent from the BSA due to the lack of suitable habitat; local range restrictions; regional extirpations; lack of connectivity between areas of suitable or occupied habitat; and/or incompatible land use and habitat degradation or alteration of onsite or adjacent lands.

Marine mammals with potential to occur in the BSA include California sea lion (*Zalophus californianus*), northern elephant seal (*Mirounga angustirostris*), and Pacific harbor seal (*Phoca vitulina*). None of these species are federally or state-listed as threatened or endangered; however, all marine mammals are protected under the Marine Mammal Protection Act of 1972. These species may infrequently occur in or immediately adjacent to the BSA near Sonoma Creek and Tolay Creek.

# 2.4.4.3 Environmental Consequences

# **Migratory Birds**

All migratory birds in the BSA are protected by the MBTA and Sections 3503, 3503.5, and 3513 of CFGC. Many species of migratory birds may inhabit the BSA at any given time and would typically use similar nesting locations. Migratory birds comprise many different bird species, including many common species. Potential nesting locations for migratory birds in the BSA include trees, dense shrubs, vegetated wetlands, grasslands, and human-made structures. Nesting birds near the project limits would likely be tolerant of the disturbances and noise associated with the existing high traffic environment. Migratory birds could nest in the BSA during construction.

# State Species of Special Concern

### Pallid Bat

The pallid bat (*Antrozous pallidus*) is an SSC. It is most common in open, dry habitats such as grassland, shrublands, brushy terrain, rocky canyons, open farmland, desert, nonconiferous woodlands, and mixed coniferous forests with rocky areas for roosting (Bolster 1998; CDFW 1988). Pallid bats most often roost in rock crevices, old

buildings, caves, mines, and hollow trees, and are one of the bat species most predictably associated with bridges (Bolster 1998). This species roosts alone, in small groups (two to 20 bats), or in colonies (hundreds of individuals, which form in March-May and stay together until October (Bolster 1998; CDFW 1988).

No pallid bats were observed roosting or nesting in the BSA. There is one CNDDB occurrence of the pallid bat within 5 miles of the BSA; this occurrence is from a barn. Other CNDDB occurrences within 10 miles are from buildings and on bridges. Pallid bat has moderate to high potential to roost on the Tolay Creek and Sonoma Creek Bridges, as well in crevices in trees in the BSA at West End Land Club, Upper and Lower Tubbs Island, Tolay Creek channel, Raceway Sonoma, Paradise Vineyards, Sears Point Ranch, and the Refuge. Bat urine stains were observed on the underside of Tolay Creek Bridge. This species has no potential to occur in the remainder of the BSA due to a lack of roosting structures and unsuitable surrounding habitat.

Construction-related vegetation removal and replacement of Tolay Creek and Sonoma Creek Bridges may impact roosting pallid bats.

#### Saltmarsh Common Yellowthroat

The Saltmarsh Common Yellowthroat (*Geothlypis trichas sinuosa*) is an SSC whose range is restricted to woody swamps, and brackish and freshwater marshes in San Francisco Bay. It is commonly found in the transition zone between moist and upland habitats. In brackish and saline tidal marsh habitat around San Francisco Bay, yellowthroat abundance is positively associated with a high percent cover of rushes (*Scirpus* spp., *Juncus* spp.) and peppergrass (*Lepidium latifolium*). Nests are typically built near the ground in herbaceous vegetation (cattails, tule, and coyote brush) (Shuford and Gardali 2008). Breeding occurs from mid-March to late July.

There are 36 CNDDB occurrences of this species within a 5-mile radius of the BSA. Three occurrences of documented breeding pairs overlap the BSA. Numerous eBird occurrences have been reported in the vicinity of the BSA, and AECOM biologists observed this species foraging at multiple locations in the BSA during site surveys. Previous surveys in the area have shown that a majority of Saltmarsh Common Yellowthroat territories are in brackish marsh habitat (Jones and Stokes 2004). There is suitable breeding habitat in the vicinity of the BSA at the Mare Island overpass and Cullinan Ranch East. There are portions of moderately suitable breeding and foraging habitat in the BSA at Cullinan Ranch, NSMWA Pond 1A, Detjen-Flyshacker Club, West End Land Club, Upper and Lower Tubbs Island, Tolay Creek channel, SLCleased Refuge, and Strip Marsh.

Vegetation removal may impact nesting Saltmarsh Common Yellowthroat. If a nonbreeding Saltmarsh Common Yellowthroat is present in the areas surrounding the project footprint, it is likely to be habituated to a high level of human disturbance and noise from the traffic on SR 37, and the bird would not likely be affected by the noise level produced by project activities.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### San Pablo Song Sparrow

The San Pablo song sparrow (*Melospiza melodia samuelis*) is an SSC. It is a permanent resident of salt marshes along the northern side of the San Francisco and San Pablo Bays, and on the southern side of San Pablo Bay southwest to San Pablo Point on the Richmond headland (Humple and Geupel 2004). In freshwater and brackish marshes, this species nests in cattails and bulrushes; in tidal slough habitats, it is associated with cordgrass, pickleweed, or gumweed (Grindelia spp.) (Humple and Geupel 2004). Breeding occurs from early March to July.

There are 21 occurrences of this species within a 5-mile radius of the BSA, two of which overlap the BSA. One of these occurrences, at Dutchman Slough, documents breeding behavior and juvenile song sparrows. Numerous eBird occurrences have been reported in the vicinity of the BSA, and AECOM biologists observed this species foraging at multiple locations in the BSA during field surveys. There is suitable breeding habitat in the vicinity of the BSA at the Mare Island Strait Interchange and Cullinan Ranch East. Like the saltmarsh common yellowthroat, there are portions of moderately suitable breeding and foraging habitat in the BSA at Cullinan Ranch, NSMWA Pond 1A, Detjen-Flyshacker Club, West End Land Club, Upper and Lower Tubbs Island, Tolay Creek, SLC-leased SPBNWR, and Strip Marsh.

Construction-related vegetation removal may impact nesting San Pablo song sparrows. Implementation of pre-construction nesting bird surveys during the nesting season (early March to July) would avoid and minimize impacts to breeding birds. If a non-breeding San Pablo song sparrow is present in the areas surrounding the Project footprint, it is likely to be habituated to a high level of human disturbance and noise from the traffic on SR 37, and would likely not be affected by project activities.

#### Western Burrowing Owl

The Western Burrowing Owl (*Athene cunicularia*) is an SSC. Breeding habitat for this species consists of open areas with mammal burrows and includes native prairie, pastures, fallow fields, road and railway rights-of-way, and urban habitats (Klute et al. 2003). Burrowing owls require mammal burrows or natural cavities surrounded by sparse vegetation (Klute et al. 2003). Foraging habitat includes cropland, pastures, fallow fields, and areas with vegetation greater than 3 feet tall.

There are eight CNDDB occurrences of this species within a 5-mile radius of the BSA. The nearest CNDDB occurrence is approximately 0.3 mile north of the BSA. Other CNDDB records document owls foraging on levees and overwintering in grazed annual grassland west of Tolay Creek. Numerous eBird occurrences have documented wintering owls in the vicinity of the BSA. There is suitable nesting, foraging, and overwintering habitat, and grasslands with mammal burrows, adjacent to the BSA. The last confirmed breeding in Sonoma County (on Skaggs Island) was in 1986 (Center for Biological Diversity 2003), though surveys conducted between 2011 and 2016 documented one possible breeding pair near the Pacific Coast (Madrone Audubon Society 2020). There is a moderate potential for owls to be overwintering or foraging in grazed annual grassland habitat immediately adjacent to the BSA west of Tolay Creek during the winter months. Western Burrowing Owl is unlikely to nest in or adjacent to the project footprint. It has the potential to overwinter or forage in grazed annual grasslands and agricultural habitats immediately adjacent to the BSA. If a burrowing owl is present in the areas surrounding the project footprint, it is likely to be habituated to a high level of human disturbance from SR 37, and thus it would not be likely to be affected by the project.

## Suisun Shrew

The Suisun shrew is an SSC. This species is restricted to the tidal and brackish marshes along the northern shore of San Pablo and Suisun Bays, from Sonoma Creek to Grizzly Island (Bolster 1998). They prefer areas of low, dense vegetation which provide adequate cover (Bolster 1998). Most surveys for the species detect individuals at the junction between pickleweed marshes and upland levees vegetated with coyote bush (*Baccharis* spp.) and grasses (Bolster 1998). Because of its restricted range, this species is highly susceptible to habitat fragmentation (Jones and Stokes 2004).

There are eight CNDDB occurrences within a 5-mile radius of the BSA; two occurrences overlap the BSA. These occurrences are at Sonoma Creek and adjacent marshes, and near Cullinan Ranch East. Suisun shrew has often been detected during saltmarsh harvest mouse surveys conducted in the area. There is low- to moderate-quality tidal marsh habitat in Cullinan Ranch, Sonoma Creek channel, the SLC-leased Refuge, Strip Marsh, and NSMWA Intake Ponds 1 and 1A, where pickleweed is present.

As a result of project activities, impacts to foraging habitat in Cullinan Ranch, Sonoma Creek channel, the SLC-leased Refuge, Strip Marsh, and NSMWA Intake Ponds 1 and 1A are expected. Construction along the shoulders of SR 37 would result in the removal of vegetation.

Construction activities might result in the injury or mortality of Suisun shrew because of equipment use (e.g., grading), vehicle traffic, and worker foot traffic. Noise, vibrations, and visual disturbance associated with grading, vegetation removal, and bridge construction activities may disturb Suisun shrew inhabiting adjacent marsh habitat. Individuals that vacate the area because of increased levels of noise and disturbance might be exposed to increased competition from conspecifics already occupying the area to which they were displaced, and may experience increased levels of predation because they are unfamiliar with the new area or lack sufficient cover. If project construction were to occur during a flooding event that inundates the adjacent wetlands, Suisun shrew could potentially take refuge in the upland transitional habitat along the roadway in the project footprint until the flooding recedes. Thus, project activities that occur in this habitat during an extreme flooding event may result in the injury or mortality of individual Suisun shrew due to equipment use (e.g., grading), vehicle traffic, and/or worker foot traffic.

### Sacramento Splittail

The Sacramento Splittail (*Pogonichthys macrolepidotus*), a endemic fish that inhabits brackish and fresh waters, is an SSC. In the San Francisco Bay Area, this species is primarily found in the Sacramento-San Joaquin Delta, Suisun Bay, Suisun Marsh,

lower Napa River, and lower Petaluma River (Moyle et al. 2004). Migration to spawning areas occurs between late November and early January, with breeding occurring if the floodplain maintains appropriate depths and temperatures (Moyle et al. 2004). Juveniles occur throughout the estuary from April to August (Moyle et al. 2004). Sacramento Splittail prefer fresh water, though are tolerant of moderate salinities (USFWS 1996).

There are six CNDDB occurrences of the species within 5 miles of the BSA. Many of these occurrences document Sacramento Splittail using tidal channels. Presence of this species in the BSA would be seasonal because Sacramento Splittail migrate from estuaries to spawn in fresh water. No spawning habitat is present in the BSA. A moderate potential exists for the species to occur in all tidally influenced the open-water portions of the BSA. However, there is a low potential of occurrence due to relatively low seasonal presence at Tolay and Sonoma Creek, low tidal exchange at Tolay Creek, and likely prohibitively high salinities in the tidal channels in the BSA during the in-water work periods.

Potential permanent impacts to suitable foraging habitat for Sacramento Splittail may occur under all Build Alternatives from placement of solid fill below the MHHW of tidal waters for roadway expansion. Roadway fills would result in loss of intertidal marsh or mudflat areas, and the placement of rock slope protection to protect expanded roadways from erosion would change some areas of the intertidal zone from soft-substrate to hard-substrate habitat. Potentially impacted habitat would be limited to shallow intertidal areas, the majority of which are not easily accessible to Sacramento Splittail and are not expected to be frequently used by them. There is potential for temporary increases in turbidity during construction. No impacts from entrapment are anticipated. No spawning habitat would be affected by the project.

Placement of permanent piles in Sonoma Creek for bridge expansion are proposed under Alternative 3B. Placement of piles in the waterway at Sonoma Creek would permanently reduce the amount of available foraging habitat. Underwater sound pressure levels have the potential to alter the behavior of Sacramento Splittail and, if sufficiently intense, can cause temporary shifts in hearing ability or injury to internal organs. The project would include the vibratory installation of steel pipe piles at Tolay Creek and Sonoma Creek Bridge. Vibratory driving would not cause injury or temporary hearing threshold shifts for Sacramento Splittail. Impact pile driving may also be used to place piles in Sonoma Creek to expand the existing bridge and would have potential to cause injury or mortality to individual Sacramento Splittail if the species is present. Shading impacts from bridge widening at Sonoma Creek Bridge under Alternative 3B would be realized as an increase in permanent shading caused by widening of the Sonoma Creek Bridge. Shade impacts are not expected to have any meaningful effect on habitat for this species.

### **Marine Mammals**

Marine mammals with potential to occur in the BSA include California sea lion, northern elephant seal, and Pacific harbor seal. None of these species are federally or state-listed as threatened or endangered; however, all marine mammals are protected under the Marine Mammal Protection Act of 1972. These species may infrequently occur within or immediately adjacent to the BSA near Sonoma Creek and Tolay Creek.

Potential impacts to marine mammals are anticipated under Build Alternative 3B only. Marine mammals exposed to noise from vibratory and/or impact pile driving may experience masking of other environmental noises and change their behaviors in response to the noise, such as moving away from the pile driving activity, startle responses, and changes to underwater vocalizations. Such noise masking and behavioral effects would be temporary, localized, and not expected to result in harm. If vibratory or impact pile driving is included in the selected alternative, Caltrans would complete and submit an application to the NMFS Office of Protected Resources for an Incidental Harassment Authorization (IHA) for the year(s) when in-water construction with potential to harass marine mammals. Under an IHA, no take beyond behavioral harassment is allowed for Marine Mammal Protection Act species. Any impacts that would exceed behavioral harassment would be avoided.

## 2.4.4.4 Avoidance, Minimization, and/or Mitigation Measures

The No Build Alternative would not affect animal species in the BSA.

### Raptors and other Migratory Birds

Under all Build Alternatives, nesting or foraging has the potential to be impacted by the project. Project features summarized in Table 1-4 would also serve to protect birds protected by the MBTA. The following additional measures are proposed to avoid and minimize potential impacts to raptors and other migratory birds. No compensation for raptors and migratory birds is anticipated.

**BIO-10: Nesting Bird Protection.** The following measures would be implemented to reduce and avoid impacts to nesting birds in the project area:

- During the bird nesting season (typically February 1 to August 31; as early as January 1 for raptors), a project biologist will conduct pre-construction surveys for active bird nests no more than 7 days prior to the start of ground or vegetation disturbance events and every 14 days during project activities.
- Tree and/or shrub removal or trimming will be conducted outside of bird nesting season.
- Tree trimming and/or shrub trimming/removal will be performed with hand tools.
- If an active nest is identified during construction that may be impacted by project activities, a no-disturbance buffer of 250 feet for raptors and 50 feet for nonraptors will be established immediately, and the project biologist will be notified. A reduced or enlarged buffer and other protection measures will be implemented in accordance with project permit requirements, defined during final design, or in consultation with the appropriate wildlife agency.

Chapter 2 Affected Environment; Environmental Consequences; and Avoidance, Minimization, and/or Mitigation Measures

#### Pallid Bat

Tree removal under all alternatives would have potential to impact pallid bats. Project features and measures proposed for trees under Section 2.4.1.3 as BIO-03 would also serve to protect pallid bats. No compensation is anticipated under any alternative. The following measures are proposed for all Build Alternatives to avoid and minimize impacts to the pallid bat.

**BIO-11: Pre-Construction Bat Surveys and Avoidance Measures.** The project biologist will conduct a visual inspection and habitat assessment for potentially suitable bat roosting habitat within 200 feet of where planned work on existing structures, tree trimming, or tree removal will occur. Assessments of bridges will include inspection of all open crevices and expansion joints. The pre-construction bat survey must be conducted during one of two time periods, either from March 1 to April 1, or from August 31 to October 15. The results of the survey will guide the following measures:

- If the habitat assessment reveals suitable roosting habitat for bats, then the appropriate exclusionary measures will be implemented prior to construction during the period between March 1 and April 15 or between August 31 and October 15.
- If the habitat assessment reveals suitable bat habitat in trees and tree removal is scheduled from April 16 through August 30 and/or October 16 through February 28, then presence/absence surveys will be conducted 2 to 3 days prior to any tree removal or trimming.
  - If presence/absence surveys are negative, then tree removal may be conducted by following a two-phased tree removal system.
  - If presence/absence surveys indicate bat occupancy, then the occupied trees will only be removed from March 1 through April 15 and/or August 31 through October 15.

Potential avoidance measures for roosting bats will be implemented as determined necessary by the project biologist in coordination with the Resident Engineer. Potential measures include visual monitoring, seasonal avoidance, enticements, and appropriate exclusion measures.

- Avoidance Measures: Avoidance measures may include seasonal avoidance, phased construction, and enticements away from the work area (e.g., providing temporary and/or permanent bat housing nearby).
- Exclusion Measures: Exclusion netting will not be used. Other measures to exclude bats from accessing potential roost sites may be implemented at the direction and with the oversight of the project biologist.

**BIO-12: Bat Monitoring Protocols.** Construction activities will stop within 150 feet of a roosting bat or bat colony that could be harmed until a qualified biologist develops a site-specific bat avoidance plan to implement at the roosting site. Once the plan is

implemented, project activities may recommence with project biologist oversight at that location.

#### Salt Marsh Common Yellowthroat and San Pablo Song Sparrow

The measures proposed for migratory birds would successfully serve to avoid and minimize potential impacts to the salt marsh common yellowthroat and the San Pablo song sparrow. No additional measures are proposed for these species. With implementation of the proposed measures for migratory birds, there are no impacts to these species that would require compensation.

#### Western Burrowing Owl

The measures proposed for migratory birds would successfully serve to avoid and minimize potential impacts to the Western Burrowing Owl. Additional measures are proposed to protect Western Burrowing Owl. With implementation of the proposed measures, there are no impacts to this species that would require compensation.

**BIO-13: Western Burrowing Owl Pre-Construction Surveys.** Pre-construction surveys will be conducted where Western Burrowing Owl nesting habitat has potential to occur within 500 feet of work. Survey protocol will include:

- Conduct four survey visits.
- An initial visit must occur between February 15 and April 15.
- A minimum of three subsequent surveys will be conducted with at least 3 weeks between visits, with at least one visit to occur after June 15.
- Conduct an additional take avoidance survey no less than 14 days prior to initiating ground-disturbing activities where work will occur.

**BIO-14: Western Burrowing Owl Nest Avoidance.** If an active Western Burrowing Owl nest is discovered during pre-construction surveys or biological monitoring, the following initial buffers will be implemented:

- From April 1 to October 15, establish a 660-foot (200-meter) no-work buffer from the active nest site.
- From October 16 to March 31, establish a 164-feet (50-meter) no-work buffer from the active nest site.
- Buffers and minimization measures (e.g. blinds or screens) may be adjusted or implemented after coordination with CDFW.

#### Suisun Shrew

Measures proposed for the FESA-listed and state fully protected salt marsh harvest mouse in Section 2.4.5 would serve to protect the Suisun shrew. With implementation of those measures, no compensation is anticipated for this species.

### Sacramento Splittail

Measures proposed for the FESA- and CESA-listed fish species in Section 2.4.5 would also serve to protect the Sacramento Splittail. With implementation of those measures, no compensation is anticipated for this species.

### Marine Mammals

No impacts are anticipated to marine mammals under Alternatives 1, 2, or 3A. Alternative 3B includes the only scenario with potential to impact marine mammals. Alternative 3B has potential to harass marine mammal species, but impacts would be limited to behavioral effects only, and compensation would not be required.

# 2.4.5 Threatened and Endangered Species

# 2.4.5.1 Regulatory Setting

The primary federal law protecting threatened and endangered species is FESA: 16 USC Section 1531, et seq. See also 50 CFR Part 402. This act and later amendments provide for the conservation of endangered and threatened species and the ecosystems upon which they depend. Under Section 7 of this act, federal agencies, such as the FHWA (and the Department, as assigned), are required to consult with the USFWS and NMFS to ensure that they are not undertaking, funding, permitting, or authorizing actions likely to jeopardize the continued existence of listed species or destroy or adversely modify designated critical habitat. Critical habitat is defined as geographic locations critical to the existence of a threatened or endangered species. The outcome of consultation under Section 7 may include a Biological Opinion with an Incidental Take Statement or a Letter of Concurrence. Section 3 of FESA defines take as "harass, harm, pursue, hunt, shoot, wound, kill, trap, capture or collect or any attempt at such conduct."

California has enacted a similar law at the state level, CESA, CFGC Section 2050, et seq. CESA emphasizes early consultation to avoid potential impacts to rare, endangered, and threatened species and to develop appropriate planning to offset project-caused losses of listed species populations and their essential habitats. CDFW is the agency responsible for implementing CESA. Section 2080 of CFGC prohibits "take" of any species determined to be an endangered species or a threatened species. Take is defined in Section 86 of CFGC as "hunt, pursue, catch, capture, or kill, or attempt to hunt, pursue, catch, capture, or kill." CESA allows for take incidental to otherwise lawful development projects; for these actions an incidental take permit is issued by CDFW. For species listed under both FESA and CESA requiring a Biological Opinion under Section 7 of FESA, the CDFW may also authorize impacts to CESA species by issuing a Consistency Determination under Section 2080.1 of CFGC.

CFGC defining species with a "Fully Protected" status in Sections 3511, 4700, 5050, and 5515 are the state's first attempt to identify and provide additional protection to those animals that were considered rare or faced possible extinction. Lists were created for fish, mammals, birds, amphibians, and reptiles. However, this listing/review process was not as rigorous as required under CESA, and several Fully Protected

species in California are common. Fully Protected species may not be taken or possessed at any time, and no licenses or permits may be issued for their take except for collection associated with scientific research and relocation of bird species to protect livestock. Under the state definition, "take" is an action that directly or indirectly kills species. The state definition does not include the terms "harass" and "harm," which are included in the FESA take definition.

Another federal law, the Magnuson-Stevens Fishery Conservation and Management Act of 1976 (MSA), was established to conserve and manage fishery resources found off the coast, as well as anadromous species and Continental Shelf fishery resources of the United States, by exercising (A) sovereign rights for the purposes of exploring, exploiting, conserving, and managing all fish within the exclusive economic zone established by Presidential Proclamation 5030, dated March 10, 1983, and (B) exclusive fishery management authority beyond the exclusive economic zone over such anadromous species, Continental Shelf fishery resources, and fishery resources in special areas.

Endangered species consultation with USFWS and NMFS is necessary when a project has the potential to affect a federally listed species and/or destroy or adversely modify designated critical habitat.

In July 2012, President Obama signed the Moving Ahead for Progress in the 21st Century Act (MAP-21) into law, as Public Law 112-141. Section 1313 of MAP-21 amends 23 USC 327 to establish a permanent Surface Transportation Project Delivery Program, allow any state to participate, and allow a state to renew its participation in the program. Through the NEPA Assignment Memorandum of Agreement, Caltrans is authorized to handle the FHWA's responsibilities under NEPA and other federal environmental laws, such as the FESA. Therefore, Caltrans is authorized to consult with USFWS and NMFS under section 7 of FESA when a project may affect federally listed species. Caltrans has prepared individual biological assessments pursuant to section 7 for USFWS and NMFS to assess and determine potential effects to listed species for the project. Caltrans included in the NMFS biological assessment an assessment of potential impacts to essential fish habitat (EFH) pursuant to a concurrent consultation under the MSA.

### 2.4.5.2 Summary of Affected Environment and Effect Determinations

The BSA supports bird, mammal, fish, and plant species that are listed under the FESA and/or CESA. Three species—the salt marsh harvest mouse, the Ridgway's rail, and California black rail—are also considered Fully Protected species under CFGC.

Areas within the project's BSA where listed species habitat generally occurs is shown in Appendix A. Specific species habitat is not included with this document to avoid divulging potential locations of species that are targeted by poachers. Species are presented by their listing status (FESA; or FESA and CESA listed species first, then CESA-only species) in this document. The following listed species with potential to occur in the BSA are arranged here in the order that they are presented in this document.

- California red-legged frog (federally listed as threatened [FT]; and state SSC)
- Ridgway's rail (federally listed as endangered [FE]; state-listed as endangered [SE]; and state Fully Protected species [FP])
- Salt marsh harvest mouse (FE; SE; and FP)
- Chinook Salmon Central Valley Spring-Run evolutionarily significant unit (ESU) and Sacramento River Winter-Run ESU (FT; and state-listed as threatened [ST])
- Steelhead Central California Coast and California Central Valley distinct population segments (DPSs) (FT)
- North American Green Sturgeon Southern DPS (FT; SSC)
- Delta smelt (FT; and SE)
- Soft bird's-beak (FE)
- Longfin smelt (ST)
- California black rail (state threatened [ST]; and FP)
- Swainson's hawk (ST)

#### **Biological Assessment and Biological Opinion**

Caltrans prepared draft Biological Assessments for USFWS and NMFS in 2021, and then initiated coordination for technical assistance with both agencies in January 2022. Caltrans and the PDT carefully considered the alternatives evaluated in the Draft EIR/EA, and their impacts and practicability, before making a formal decision on the identification of the preferred alternative (Alternative 3B) in April 2022. Caltrans will submit a final Biological Assessments to both agencies based on the identified preferred alternative. Following identification of the preferred alternative, Caltrans has continued coordination with USFWS and NMFS, and anticipates completion of Section 7 consultations and obtaining the Biological Opinions from both agencies in 2022 or 2023. The Section 7 consultation process has therefore been extended by Caltrans to allow USFWS and NMFS adequate time for review and consultation.<sup>1</sup> The Biological Opinions will therefore be finalized and issued for the project during the design phase. Any new requirements from the USFWS and NMFS as a result of their consultations will be updated in the project's Environmental Commitment Record.

<sup>&</sup>lt;sup>1</sup> Caltrans District 4 extended the Section 7 consultation process consistent with the May 28, 2021 *NEPA Process Improvement Team's Memorandum on Flexibility in Timing of Obtaining Biological Opinions* issued by Caltrans Headquarters, Division of Environmental Analysis.

As the lead federal agency, Caltrans made the following preliminary effects determination in its Biological Assessments for federally threatened and endangered species. USFWS regulated species preliminary effect determinations include:

- The project *may affect, but is not likely to adversely affect*: Delta Smelt and soft bird's-beak.
- The project *may affect, and is likely to adversely affect*: California red-legged frog, California Ridgway's rail, and salt marsh harvest mouse.
- The project will have *no effect* on USFWS-designated critical habitat.

NMFS regulated species preliminary effect determinations include:

- The project *may affect, but is not likely to adversely affect*: steelhead CV DPS; Chinook Salmon CV Spring-Run ESU; and Chinook Salmon SRWR ESU.
- The project *may affect, and is likely to adversely affect* steelhead CCC DPS and Green Sturgeon Southern DPS.
- The project *may affect, and is likely to adversely affect* designated critical habitat for: steelhead CCC DPS; steelhead CV DPS; Chinook Salmon SRWR ESU; and Green Sturgeon Southern DPS. In the context of the San Pablo Bay ecosystem, these adverse effects would be limited in scope and are not expected to result in the adverse modification of the estuarine components of critical habitat for these species.
- With regard to EFH, the project *may adversely affect* EFH as designated under the Pacific Salmon FMP, the Pacific Groundfish FMP, and the Coastal Pelagic FMP; however, such effects are expected to be minor and would not substantially alter the value of EFH in the Action Area.

Caltrans has initiated draft Biological Assessments for use in consultation with USFWS and NMFS for the project. Section 7 consultation would be completed during the project's final design phase.

# 2.4.5.3 Affected Environment: California Red Legged Frog

The analysis summarized in the Threatened and Endangered Species sections are from the NES prepared in September 2021; the draft Biological Assessment for USFWS prepared on October 2021; and the draft Biological Assessment for NMFS prepared in September 2021.

California red-legged frogs breed between November and April in aquatic habitats, such as pools, ponds, marshes, springs, sag ponds, dune ponds, and lagoons, and in artificial impoundments, such as stock ponds (USFWS 2002). Adults prefer "deep (greater than 2 feet) still or slow-moving water and dense, shrubby riparian or emergent vegetation" (USFWS 2002). Such habitat generally includes freshwater

marshes, streams, ponds, and other permanent and temporary water sources dominated by dense riparian scrubby vegetation (e.g., willows, cattails, and bulrushes).

Salinity of the water is also a determinant of whether California red-legged frogs have the potential to occur. The maximum salinity tolerance by adults is near 9 parts per thousand (ppt), and embryonic stages have an even lower tolerance (less than 5 ppt) (Jennings and Hayes 1990). Upland dispersal habitats with dense vegetation may be important sheltering habitat during winter. During the dry season, California red-legged frogs may live in small-mammal burrows and moist leaf litter (Jennings and Hayes 1994), as well as under boulders or rocks, organic debris, and agricultural features (USFWS 2002). During dry periods, this species is seldom found far from water. However, during wet weather, individuals may make overland excursions through upland habitats (Tatarian 2008). The average dispersal distance for California redlegged frog is 1 mile; the maximum dispersal distance is nearly 2 miles (USFWS 2010).

There are two California red-legged frog CNDDB occurrences within 1 mile of the BSA, a distance understood to be the species dispersal distance (USFWS 2005). Both occurrences are in the vicinity of Sears Point, and closest includes one adult California red-legged frog that was observed in 2006 along an incised channel about 0.8 mile north of the BSA. There are several natural and artificial wildlife crossings in the BSA that may be used by dispersing California red-legged frogs to cross under SR 37, including culverts carrying streams and drainage channels under the road.

Habitat assessments were conducted in the field to determine the presence of potential dispersal, foraging, aestivation, and breeding habitat of California red-legged frog in the BSA. A review of previous habitat assessments for nearby projects was also considered. The aquatic habitat in the BSA east of Tolay Creek is too saline to support California red-legged frogs. The segment of the BSA west of SR 121 contains some of the elements necessary to meet the physical habitat requirements for California red-legged frog dispersal and aestivation habitat. The survey results determined that there is no suitable breeding habitat present in the BSA. Several factors make aquatic features in the BSA less suitable for breeding: brackish conditions; extreme dense growth of cattails and other in-channel vegetation; lack of permanent deep water; degraded water quality; and the long history of human disturbance. No California red-legged frogs were observed during the biological surveys that were conducted to assess habitats in the BSA.

Critical habitat for the California red-legged frog was designated by USFWS in April 2006 and revised in March 2010 (USFWS 2010). There is no critical habitat for California red-legged frog in the BSA. The closest critical habitat is on either side of I-80 in Napa and Solano Counties, about 4.1 miles to the northeast.

### 2.4.5.4 Environmental Consequences: California Red-Legged Frog

# **Direct Effects**

All Build Alternatives would directly affect suitable California red-legged frog habitat that occurs west of the railroad line near the intersection of SR 121 and SR 37; no suitable habitat occurs east of that point in the BSA (see Map Book, included before the appendices). Potential impacts to California red-legged frog would include permanent and temporary loss of upland dispersal, foraging, and refugia habitat; and loss of aquatic dispersal habitat. Conversion of habitat from vegetated to developed would result in a permanent loss of habitat. Ground disturbance activities from staging, clearing, grubbing, etc., could temporarily impact potential upland dispersal, foraging, and refugia habitat. The proposed modifications to California red-legged frog habitat may have a potential adverse impact on the behavioral patterns of some individuals of this species, including foraging, migration, and aestivation. Anticipated permanent fill impacts to California red-legged frog habitat for each project alternative are summarized in Table 2-42. Table 2-43 summarizes the estimated temporary impacts to California red-legged frog habitat in the BSA.

# Table 2-42Estimated Permanent Fill Impacts to California Red-Legged FrogHabitat in the BSA

Habitat Type	Alternative 1 and 2 (Acres) <sup>2</sup>	Alternative 3A (Acres) <sup>2</sup>	Alternative 3B (Acres) <sup>2</sup>
Nonbreeding Aquatic Dispersal/Foraging <sup>1</sup>	0	0.17	0.17
Upland Dispersal/Foraging/Refugia	0.01	1.34	1.35
Total	0.01	1.51	1.52

Notes:

<sup>1</sup> Aquatic habitat not suitable for breeding.

<sup>2</sup> Acreages are rounded to the nearest hundredth of an acre before summing.

# Table 2-43Estimated Temporary Impacts to California Red-Legged FrogHabitat in the BSA

Habitat Type	Alternative 1 and 2 (Acres) <sup>2</sup>	Alternative 3A (Acres) <sup>2</sup>	Alternative 3B (Acres) <sup>2</sup>
Nonbreeding Aquatic Dispersal/Foraging <sup>1</sup>	—	0.1	0.1
Upland Dispersal/Foraging/Refugia	0.11	0.65	0.68
Total	0.11	0.75	0.78

Notes:

<sup>1</sup> Aquatic habitat not considered suitable for breeding.

<sup>2</sup> Acreages are rounded to the nearest hundredth of an acre before summing.

Based on nearby occurrences and suitable habitat in the BSA, there is potential for this project to result in the "take" of California red-legged frog as defined by section 7 of the ESA. The project could affect frogs dispersing through the project footprint during construction, where they could come into direct contact with construction equipment and/or personnel and be harmed or harassed. In addition, construction-

related disturbances such as increased levels of human activity and increased noise and vibration levels could result in alteration of normal behavior (such as invoking an escape response or alteration of normal dispersal routes), resulting in increased chances of predation or other harm.

One freshwater wetland complex southwest of the SR 121 intersection where California red-legged frog may be dispersing or foraging. Construction activities there have the potential to harm or harass individuals. Permanent and temporary impacts to this wetland would occur because of shoulder widening. Work in this wetland would occur during the dry season, and no water diversion would be required. Any work in this wetland associated with removing vegetation would be considered a temporary impact to nonbreeding aquatic habitat because the wetland would be returned to preproject conditions.

Activities associated with access, staging, storage, and disposal, in addition to activities associated with construction of the project (e.g., shoulder widening, and cut and fill) have the potential to contribute sediment and increase turbidity in riverine waters above those levels generally found under existing conditions. California red-legged frog would not experience any effects associated with sedimentation and turbidity because grading activities in suitable California red-legged frog habitat would occur in the dry season. The potential for construction activities to increase sedimentation or turbidity in creeks would be minimized. Based on the direct permanent and temporary impacts to upland dispersal habitat, foraging, and refugia habitat, and aquatic nonbreeding dispersal and foraging habitat, there is potential for take of individual California red-legged frogs.

### Indirect Effects

No indirect effects are anticipated for any of the Build Alternatives. No new barriers to California red-legged frog dispersal would occur because of project components, and the areas planned for construction would not be between aquatic, upland, or dispersal habitats. The planned project activities are not anticipated to affect long-term California red-legged frog mobility in the area.

### 2.4.5.5 Avoidance, Minimization, and/or Mitigation Measures: California Red-Legged Frog

Previously described measures, including **BIO-03: Tree Replacement, Landscaping, and Revegetation Plan**, would also serve to avoid and minimize potential impacts to California red-legged frog.

All Build Alternatives are anticipated to have unavoidable impacts that would affect California red-legged frog. General and specific measures are summarized here that would avoid and minimize potential impacts to this species. Compensation to offset impacts to California red-legged frog habitat for the selected Build Alternative are proposed at the end of this section.

### **General Measures for All Threatened and Endangered Species**

The following measures are included for all threatened and endangered species with potential to occur in the project area under all Build Alternatives:

**BIO-15: Stop-Work Authority.** Through the Resident Engineer or their designee, the project biologist(s) shall have the authority to stop project activities to minimize take of listed species or if he/she determines that any permit requirements are not fully implemented. If the project biologist(s) exercises this authority, the appropriate resource regulatory agencies shall be notified by telephone and email within 48 hours.

**BIO-16: Worker Environmental Awareness Training.** Before the onset of construction and within 3 days of any new worker arrival, a project biologist will conduct this training for all construction personnel. At a minimum, the training will include a description of all special-status species and their habitats; the potential occurrence of these species in the project area; an explanation of the status of these species and protection under FESA, CESA, and all other federal, state, and local regulatory requirements; the measures to be implemented to conserve listed species and their habitats as they relate to the work site; and boundaries within which construction may occur. A fact sheet conveying this information will be prepared and distributed to all construction crews and project personnel entering the project footprint. Upon completion of the program, personnel will sign a form stating that they attended the program and understand all AMMs and implications of the FESA, CESA, and all other federal, state, and local regulatory requirements.

**BIO-17: Discovery of Injured or Dead Special-Status Species.** Immediately upon discovery of any dead, injured, or entrapped special-status species regulated by USFWS, NMFS, or CDFW, Caltrans will provide appropriate notifications to agency(s) with jurisdiction.

**BIO-18: Wildlife Species Relocation.** When listed wildlife species (that do not have state fully protected status) are present and it is determined that they could be injured or killed by construction activities, the project biologist in coordination with the appropriate state and federal wildlife agencies will identify appropriate methods for capture, handling, exclusion, and relocation of individuals that could be affected. Where listed species cannot be captured, handled, excluded, or relocated, actions that could injure or kill individuals will be avoided or delayed until the species leaves the affected area. Actions that could harm or kill individual state fully protected species that are in the project area will be avoided or delayed until the species leaves the affected area.

**BIO-19: Construction Noise.** Operation of pile drivers, dozers, large excavators, and other heavy equipment that generates vibration and noise impacts that could harm wildlife will be limited to daylight hours when a project biologist is present.

### California Red-Legged Frog Species-Specific Measures

The following measures are proposed specifically for California red-legged frog for all Build Alternatives.

**BIO-20: California Red-Legged Frog Habitat Work Window.** These work windows are applicable only to those portions of the project area where suitable California red-legged frog habitat occurs.

- Initial ground disturbance in California red-legged frog upland dispersal habitat, as identified by a USFWS-approved project biologist, will be timed to occur between April 15 and October 15.
- All work in suitable aquatic habitat for California red-legged frog, as identified by a USFWS-approved project biologist, will only occur once the aquatic feature no longer holds water, or between June 15 and October 15 after installation of WEF.

**BIO-21: California Red-Legged Frog Pre-Construction Surveys.** Pre-construction surveys for the California red-legged frog will be conducted by the project biologist within 14 calendar days of the initiation of project activities in suitable upland and aquatic habitat prior to ground-disturbing activities, vegetation removal, and WEF installation. Surveys will be conducted as outlined in the 2005 USFWS species survey guidelines for California red-legged frog. Pre-construction surveys will include:

- Conduct foot surveys of potential frog habitat within the project limits and accessible adjacent areas (within at least 50 feet of project limits).
- Investigate potential cover sites (burrows, rocks, soil cracks, vegetation, and other potential refuge habitat); and any areas of disturbed soil for signs of California red-legged frog.
- Native vertebrates found in cover sites within the project limits will be documented and, if handling is allowed, relocated to an adequate cover site in the vicinity. Species that cannot be relocated due to special protection status will be addressed in coordination with the appropriate agency(s) with jurisdiction.

**BIO-22: California Red-Legged Frog Monitoring Protocols.** During construction in and near potential California red-legged frog habitat, the following protocols will be observed by the project biologist during construction monitoring:

- WEF installed in California red-legged frog habitat will be checked regularly for potential frog presence, to ensure that it is functioning as intended, and is appropriately maintained. WEF issues will be reported to the Resident Engineer for immediate resolution.
- Within 24 hours prior to initial ground-disturbing activities, portions of the project footprint where potential California red-legged frogs habitat has been identified will be surveyed by a project biologist(s) to clear the site of frogs moving above ground, or taking refuge in burrow openings or under materials that could provide cover.

- A project biologist(s) will be present during all initial ground-disturbing activities and vegetation removal in suitable refugia habitats for the California red-legged frogs to monitor the removal of the top 12 inches of topsoil.
- If potential aestivation burrows are discovered, the burrows will be flagged for avoidance.
- After a rain event, and prior to construction activities resuming, a qualified biologist will inspect the work area and all equipment/materials for the presence of California red-legged frog.
- Upon discovery of a California red-legged frog individual(s) in an active construction area, all work will cease within a 50-foot radius of the frog. The frog will be allowed to leave the site on its own; if the frog(s) does not leave on its own, it will be relocated within 0.25 mile of the construction site and placed in a natural burrow by a project biologist with the appropriate USFWS 10(a)1(A) handling permit.
- USFWS will be notified by phone and email within one working day of any California red-legged frog discovery in the project area.

## Proposed Compensation to Offset Habitat Loss

Caltrans proposes to mitigate for permanent losses of California Red Legged Frog upland dispersal habitat at a 2:1 restoration to impact ratio; and non-breeding aquatic dispersal/foraging habitat at a 3:1 ratio, or at a ratio determined appropriate in coordination with the USFWS, for the selected alternative. Upland dispersal habitat is not suitable for breeding and is abundant in the surrounding area; therefore, a reduced ratio for this habitat type is proposed.

Caltrans identified four approved mitigation banks in the Bay with available credits for California red-legged frog. These banks include the North Bay Highlands Conservation Bank (Marin County), Ohlone West Conservation Bank (Alameda County), Oursan Ridge Conservation Bank (Contra Costa County), and Ridge Top Ranch Wildlife Conservation Bank (Solano County). There currently are enough available credits to compensate for loss of California red-legged frog through purchase of credits at available banks. Caltrans anticipates enough available credits would remain available before the project is complete.

Caltrans would restore temporarily disturbed California red-legged frog habitat areas to pre-project conditions on site at a 1:1 ratio.

The following measures summarizes Caltrans commitment to offset impacts to California red-legged frog habitat from the selected alternative:

**BIO-23. Compensation for California Red-Legged Frog Habitat Effects.** Caltrans will offset permanent loss of California red-legged frog habitat through the purchase of credits from an approved conservation bank in the project's service area. Credits will be purchased as follows:

- Loss of upland dispersal habitat area will be compensated through credit purchase at a 2:1 ratio, or at a ratio determined appropriate in coordination with USFWS.
- Loss of non-breeding aquatic dispersal/forage habitat will be compensated through credit purchase at a 3:1 ratio, or at a ratio determined appropriate in coordination with USFWS.

Caltrans will offset temporary impacts during construction to California red-legged frog habitat by restoring disturbed areas to pre-project conditions at a 1:1 ratio.

### 2.4.5.6 Affected Environment: Ridgway's Rail

This species occurs primarily in tidal salt and brackish marshes that have consistent tidal flows, and access to tidal channel networks, nesting and cover habitat, and prey supply of invertebrates. This species' current range is restricted to the tidal salt and brackish marshes surrounding San Francisco Bay, which includes the South and Central Bay, San Pablo Bay, and Suisun Marsh. Along the perimeter of San Pablo Bay, rails typically inhabit salt marshes dominated by pickleweed and California cordgrass (USFWS 2013a). Breeding season begins in February, with peak nesting period lasting from April through May (USFWS 2013a). Nesting habitat must include sloughs to provide invertebrate prey items and escape from predators (USFWS 2013a). Nests are built on elevated structures surrounded by vegetative cover at an elevation that does not become completely inundated during high tides. Small tidal channels with dense vegetation are important foraging areas (USFWS 2013a).

A total of 22 CNDDB occurrences of Ridgway's rail have been documented within 5 miles of the BSA; one occurrence overlaps the BSA at Tolay Creek and another at Sonoma Creek, both documented breeding behavior. The species has also been documented at Dutchman Slough and Sonoma Creek. The Ridgway's rail is a year-round resident of tidal marsh in the Refuge. Use of brackish marshes by Ridgway's rail in San Pablo Bay is largely restricted to the major sloughs and rivers (USFWS 2013b). Individuals have been documented in the vicinity of the BSA at Cullinan Ranch, Dutchman Slough, Guadalcanal Village, Upper Tolay Lagoon, Sonoma Creek, and at NSMWA Intake Ponds 1 and 1A, but where these locations overlap the BSA there is no suitable habitat present.

Ridgway's rail has a moderate potential to occur at Tolay Creek, Upper Tolay Lagoon, Sonoma Creek, and in portions of SLC-leased Refuge and Strip Marsh (Marriott, M., per. comm, 2020). These locations contain portions of suitable foraging and dispersal tidal marsh habitat. Dense vegetation coverage is largely absent from the BSA. Most vegetated areas do not provide the higher marsh vegetation necessary for cover, retreat, and breeding; though narrow strips of vegetation are present along the waterward sides of SR 37, these areas likely do not provide enough cover for this cryptic species. Although the area to the east of Sonoma Creek, on the southern side of SR 37, has not been surveyed, the species is assumed to be present in the dense marsh habitat south of SR 37. This species has low to no potential to occur in the remainder of the BSA. Critical habitat for this species has not been designated by USFWS.

#### 2.4.5.7 Environmental Consequences: Ridgway's Rail

#### Direct Effects

All Build Alternatives would directly affect suitable Ridgway's rail habitat along the edge of the project footprint. Permanent effects include fill of brackish marshes at SLC-leased Refuge, Sonoma Creek, Strip Marsh, Tolay Creek, and Upper Tolay Lagoon. Table 2-44 summarizes the anticipated permanent fill impacts to Ridgway's rail habitat for each project alternative. Table 2-45 summarizes temporary impact areas to Ridgway's rail habitat from construction access, staging, vegetation clearing, and temporary soil disturbance.

#### Table 2-44 Estimated Permanent Impacts to Ridgway's Rail Habitat in the BSA

Habitat Type <sup>1</sup>	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>
Marginal Foraging/Dispersing	0.34	0.79	1.02	1.65

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

# Table 2-45Estimated Temporary Impact Areas to Ridgway's Rail Habitat in the<br/>BSA

Habitat Type <sup>1</sup>	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>
Marginal Foraging/Dispersing	0.42	1.90	2.02	3.11

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

Accidental spills or runoff from the project footprint could potentially enter and degrade the brackish marsh habitat. Construction activities in or adjacent to the tidal wetlands may affect this species. Effects on Ridgway's rail occurring in or near the project footprint may include an alarm response on the part of the bird, causing it to flush, run away from the source of disturbance, or wait out the disturbance. Other construction noise that may be generated by the project would be similar in magnitude to ongoing roadway noise, as described in Section 2.3.7. Ridgway's rail is a CFGC fully protected species; no take can be authorized under CESA. Project activities are anticipated to result in harassment only, which would require take authorization under FESA but is not considered take as defined in the CFGC and under CESA.

Under Alternative 3B only, there would be an increase in overwater shading due to the widening of the Sonoma Creek Bridge from its current width of 53.5 feet to 57.5 feet. Permanent shading is not expected to affect the productivity of open waters in the BSA because river flow, tidal, and wind-wave circulation would move water continuously through the narrow area of the daily and seasonally shifting shadow. Currently, there is approximately 20 feet of vertical clearance between the bridge and the water surface.

The widened bridge would have a similar amount of vertical clearance to the existing bridge. Shading may have some effect on the composition of benthic organisms in the affected area, but such changes are not expected to substantially alter the value of habitat for this species in the BSA. The creek would continue to provide hydrology to support the vegetation in the adjacent wetland communities. The pickleweed wetland under and immediately adjacent to the bridge is not dense or tall, and so does not currently provide nesting or high quality foraging habitat due to the lack of cover. Additionally, public use in these wetland habitats is relatively high; foot traffic is common through these areas to access Sonoma Creek for fishing and recreation. The frequent human disturbance and low quality vegetation present at these locations limit the use of the wetlands in the project area for this sensitive marsh-bird species. Estimated impacts from shading are summarized in Table 2-46.

# Table 2-46Alternative 3B Shading Impacts to Ridgway's Rail Habitat in the<br/>BSA

Habitat Type <sup>1</sup>	Temporary Shading (Acres) <sup>2</sup>	Permanent Shading (Acres) <sup>2</sup>
Marginal Foraging/Dispersing	0.86	0.52

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

#### **Indirect Effects**

Under all build alternatives, the project would not create new barriers to dispersal for this species. The planned activities are not anticipated to affect long-term mobility in the area for this species. Therefore, indirect effects on Ridgway's rail are not anticipated and are considered discountable.

#### 2.4.5.8 Avoidance, Minimization, and/or Compensation: Ridgway's Rail

Under all Build Alternatives, the following previously described measures to protect wetland habitat and birds in the project area would also serve to avoid and minimize potential impacts on Ridgway's rail: BIO-01: Wetlands Protection – Invasive Plants; BIO-02: Wetland Protection; BIO-03: Tree Replacement, Landscaping, and Revegetation Plan; BIO-04: Estuarine Dewatering Work Window; BIO-05: Turbidity Control; BIO-10: Nesting Bird Protection; BIO-15: Stop-Work Authority; BIO-16: Worker Environmental Awareness Training; BIO-17: Discovery of Injured or Special-Status Species; and BIO-19: Construction Noise.

The following specific measures are proposed for all Build Alternatives to avoid and minimize potential impacts to Ridgway's rail:

**BIO-24:** Ridgway's Rail and California Black Rail Pre-Construction Survey. If Ridgway's rail or California black rail habitat are present within 700 feet of the immediate project area and work would occur during the rail nesting season (February 1 through August 31), a pre-construction survey by a USFWS 10(a)1(A) permit holder for Ridgway's rail will be conducted to determine whether the species are present. Survey requirements and timing would be determined in consultation with USFWS and CDFW.

If Ridgway's rail and/or California black rail are detected during pre-construction surveys, then project activities will not occur within 700 feet of an identified detection (or smaller distance if approved by USFWS and CDFW) during the rail nesting season. If rail activity is detected within the 700-foot buffer, immediate consultation with USFWS and CDFW is required.

**BIO-25: Ridgway's Rail and California Black Rail Protocol-Level Surveys and Avoidance and Minimization Measures.** The following monitoring protocols, and avoidance and minimization measures for Ridgway's rail and California black rail, will be implemented where appropriate:

- Ridgway's Rail/California Black Rail Protocol-Level Surveys: Protocol-level surveys in and surrounding the project area shall be conducted between January 15 and February 1. A minimum of four surveys is required: each survey shall be 2 to 3 weeks apart, and the final survey shall be completed by March or mid-April to ensure that no Ridgway's rail or California black rail are present during construction. Surveys shall be completed prior to the initiation of construction, with 3 weeks remaining after completion of surveys, and before project initiation, to submit results to CDFW for review. Protocol-level survey requirements shall be followed as recommended in the USFWS Clapper Rail Survey Protocol (USFWS 2015), Secretive Marsh Bird Survey Protocol Comparison in San Francisco Bay (Wood 2014), and USFWS Site-Specific Protocol for Monitoring Marsh Birds (Wood et al. 2017).
- Avoidance and Minimization: If Ridgway's rail or California black rail is detected during protocol surveys, no work activity shall occur from February 1 to August 31 during the Ridgway's rail and California black rail nesting season within suitable habitat for those species. Suitable Ridgway's rail or California black rail habitat includes, but is not limited to, marshes, wetlands, streams, and waterways, and associated upland habitat capable of providing upland refugia habitat as determined by a qualified biologist experienced with these species.
- Avoidance Buffers: If breeding Ridgway's rail or California black rail are determined to be present, activities will not occur within 700 feet of an identified calling center. If the intervening distance across a major slough channel or across a substantial barrier between the Ridgway's rail or California black rail calling center and any activity area is greater than 200 feet, work may proceed at that location during the breeding season in consultation with CDFW.
- High Tide Restrictions: To avoid the loss of any individual Ridgway's rail or California black rail, activities in or adjacent to suitable habitat for the species will not occur within 2 hours before or after extreme high tides (6.5 feet or above, as measured at the Golden Gate Bridge). This is when the marsh plain is inundated and protective cover for Ridgway's rail and California black rail is

limited. Project activities in or adjacent to suitable habitat during and within 2 hours before and after extreme high tide events could prevent Ridgway's rail or California black rail from reaching available cover.

### Proposed Compensation to Offset Ridgway's Rail Habitat Loss

Caltrans proposes to offset permanent losses of Ridgway's rail habitat at a 2:1 restoration/enhancement to impact ratio, or at a ratio determined appropriate in coordination with USFWS and CDFW, through purchase of credits at an approved bank. A relatively lower ratio is proposed because the habitat impacted is of marginal quality (i.e., low vegetation stature and high human disturbance in the affected habitat); the usage of the habitat by this species is limited to foraging only; and the rail's use of the habitat is likely relatively low in comparison to the expansive wetland habitats available to this sensitive marsh bird adjacent to the project area.

There are no current Ridgway's rail approved conservation banks for the project area. However, a conservation bank for Ridgway's rail in Alameda County, Newark Slough Conservation Bank, is in the approval process and is estimated to release credits in 2022. The project is within the anticipated service area for Ridgway's rail for this bank and the bank is expected to have credits available to meet compensatory needs for all project alternatives.

The following measure summarizes Caltrans commitment to offset impacts to Ridgway's rail habitat from the selected alternative:

**BIO-26. Compensation for Ridgway's Rail Habitat Effects**. Caltrans will purchase credits from an approved conservation bank in the project's service area to offset permanent loss and degradation of Ridgway's rail habitat at a 2:1 impact to restoration/enhancement area ratio, or at a ratio determined appropriate in coordination with USFWS and CDFW.

Caltrans will offset temporary impacts during construction to Ridgway's rail habitat by restoring disturbed areas to pre-project conditions at a 1:1 ratio.

### 2.4.5.9 Affected Environment: Salt Marsh Harvest Mouse

The salt marsh harvest mouse (*Reithrodontomys raviventris*) is endemic to the marshes of San Francisco Bay (USFWS 2013a). The northern subspecies of salt marsh harvest mouse is found in marshes in San Pablo and Suisun Bays (USFWS 2013a). Salt marsh harvest mouse usually is restricted to saline or brackish marsh habitat. The largest population of the northern subspecies is found in the tidal marshes along northern San Pablo Bay (Petaluma River to Mare Island Strait). Preferred habitat includes pickleweed-dominated vegetation, as well as high-tide/flood refugia, seasonal use of terrestrial grassland, and mixed salt marsh vegetation habitat (USFWS 2013a). Recent trapping efforts have detected the species in significant numbers in brackish marshes and in marshes dominated by plants other than pickleweed, such as alkali bulrush (*Bolboschoenus maritimus*) and tri-corner bulrush (*Schoenoplectus americanus*) (Smith et al. 2018), in the grasslands adjoining marshes—at least 328 feet from the marsh edge, and in disturbed and diked marshes (Smith and Kelt

2019). As a result, managed wetlands are recognized as important habitat for the persistence of this species (Smith 2019). Activities are primarily nocturnal, with some activity identified at sunset and sunrise. The breeding season of salt marsh harvest mice, based on the female cycle, is from September to December in the San Pablo Bay area (Bias 1994).

A total of 16 occurrences have been documented within a 5-mile radius of the project alignment; one occurrence overlaps the BSA at SLC-leased Refuge and Strip Marsh. There is a BSA-adjacent occurrence under the Sonoma Creek Bridge. The tidal marshes along the northern San Pablo Bay, from the Petaluma River to Mare Island Strait, support the largest population of salt marsh harvest mouse (USFWS 2013a) and includes most of the project footprint. Trapping conducted at Tolay Creek and Cullinan Ranch resulted in capture of salt mash harvest mice in grassy, diked areas and on levees, indicating that salt marsh harvest mice use areas other than thick pickleweed, regardless of tidal flooding causing them to escape upland. Therefore, levees and grassy areas adjacent to pickleweed habitats cannot be dismissed as nonhabitat for the salt marsh harvest mouse (Hulst et al. 2001). There are portions of suitable tidal marsh habitat at Tolay Creek, Detjen-Flyshacker Club, Sonoma Creek, SLC-leased Refuge, Strip Marsh, and Cullinan Ranch.

Salt marsh harvest mice may forage in the pickleweed-dominated marsh habitat present throughout much of the BSA and may forage and seek refuge in most adjacent vegetated habitats. However, the immediate roadside ruderal vegetation in the BSA does not provide suitable cover and foraging opportunities for small mammal species and is not suitable as upland escape habitat for salt marsh harvest mice. The swaths of pickleweed adjacent to the road are less likely to be used than areas farther away from disturbance, unless high tide pushes the mice to these areas. Although mice have been observed in disturbed construction sites inside WEF, this is a very rare occurrence, and the likelihood for them to occur in active construction areas is low.

USFWS has not designated critical habitat for the salt marsh harvest mouse.

# 2.4.5.10 Environmental Consequences: Salt Marsh Harvest Mouse

### **Direct Effects**

All Build Alternatives would fill in suitable salt marsh harvest mouse foraging habitat in the BSA near Tolay Creek, Detjen-Flyshacker Club, Sonoma Creek, the SLC-leased Refuge, Strip Marsh, and Cullinan Ranch. Table 2-47 summarizes anticipated permanent impacts from fill for each project alternative to salt marsh harvest mouse habitat. Table 2-48 summarizes anticipated temporary impacts from construction access, temporary structures, staging, vegetation, and other soil-disturbing activities in salt marsh harvest mouse habitat. Both permanent and temporary impacts would occur in narrow areas spreads out along the project footprint. In most of the of the habitat areas that would be impacted, the habitat quality is marginal due to ongoing disturbance associated with SR 37.
# Table 2-47Estimated Permanent Impacts to Salt Marsh Harvest Mouse Habitatin the BSA

Habitat Type <sup>1</sup>	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>
Foraging	0.37	0.69	0.92	1.67

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

# Table 2-48Estimated Temporary Impact Areas to Salt Marsh Harvest MouseHabitat in the BSA

Habitat Type <sup>1</sup>	Alternative 1	Alternative 2	Alternative 3A	Alternative 3B
	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>	(Acres) <sup>2</sup>
Foraging	0.70	2.01	2.14	3.24

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

Construction along the shoulders of SR 37 would result in the removal of ruderal and wetland vegetation, and ground disturbance. These areas may experience increased erosion, resulting in the deposition of sediments in surrounding wetlands and waters. Any work in paved areas adjacent to wetlands and waters would require implementation of standard Caltrans BMPs to reduce erosion and sedimentation, and to prevent construction debris from entering the wetlands and waters downslope.

Salt marsh harvest mouse is a CFGC fully protected species; the project must completely avoid any take impacts to individual salt marsh harvest mouse greater than harassment; and no take (as defined under the CFGC) can be authorized under CESA for this species. Project activities are anticipated to result in harassment only, which would require take authorization under FESA but is not considered take under CESA.

The project is proposing measures that would ensure that salt marsh harvest mice are not injured during construction. Construction activities would create visual and noise disturbance that might result in the harassment of salt marsh harvest mice. Disturbance may be generated during equipment use (e.g., grading), vehicle traffic, and other work activities in areas adjacent to the species' habitat. Increased levels of noise and disturbance may also force individuals to move into new territories. These individuals might be exposed to increased competition from conspecifics in these new areas and increased levels of predation because of unfamiliarity with the area or lack of sufficient cover.

If project construction were to occur during a flooding event that inundates the adjacent wetlands, salt marsh harvest mice could potentially take refuge in the upland transitional habitat along the roadway in the project footprint until the flooding recedes. Project activities that occur in this habitat during an extreme flooding event may result in the harassment of individual salt marsh harvest mice.

For Alternative 3B, there would be an increase in permanent shading over potential salt marsh harvest mouse forage habitat due to the widening of the Sonoma Creek Bridge and Tolay Creek Bridge (Table 2-49). Shading impacts are not expected to substantially alter the value of habitat in the BSA.

# Table 2-49Alternative 3B Shading Impacts to Salt Marsh Harvest MouseHabitat in the BSA

Habitat Type <sup>1</sup>	Temporary Shading (Acres) <sup>2</sup>	Permanent Shading (Acres) <sup>2</sup>
Foraging	0.86	0.52

Notes:

<sup>1</sup> Ridgway's rail and salt marsh harvest mouse habitats are almost entirely in the same area.

<sup>2</sup> Acreages rounded to the nearest hundredth of an acre.

### Indirect Effects

No new barriers to salt marsh harvest mouse dispersal would occur from project components under all Build Alternatives, and the areas planned for construction are not between patches of pickleweed. The planned activities are not anticipated to affect long-term salt marsh harvest mouse mobility in the area. Therefore, indirect project effects on salt marsh harvest mouse are considered discountable.

# 2.4.5.11 Avoidance, Minimization, and/or Compensation: Salt Marsh Harvest Mouse

Under all Build Alternatives, the following previously described measures to protect wetland habitat and wildlife in the project area would also serve to avoid and minimize potential impacts on salt marsh harvest mouse: BIO-01: Wetlands Protection – Invasive Plants; BIO-02: Wetland Protection; BIO-03: Tree Replacement, Landscaping, and Revegetation Plan; BIO-04: Estuarine Dewatering Work Window; BIO-05: Turbidity Control; BIO-15: Stop-Work Authority; BIO-16: Worker Environmental Awareness Training; BIO-17: Discovery of Injured or Special-Status Species; and BIO-19: Construction Noise.

The following specific measures are proposed for all Build Alternatives to avoid and minimize potential impacts to salt marsh harvest mouse.

**BIO-27: Salt Marsh Harvest Mouse Pre-Construction Surveys.** A USFWS- and CDFW-approved project biologist(s) will conduct pre-construction surveys no less than 7 days prior where suitable or potentially suitable habitat for salt marsh harvest mouse occurs and could be disturbed by construction activities in the project area. If salt marsh harvest mouse is discovered, immediate consultation with USFWS and CDFW is required before work near the discovery can proceed.

**BIO-28: Salt Marsh Harvest Mouse Impact Avoidance and Minimization.** WEF will be implemented in areas of potential habitat prior to ground disturbance as follows:

• All supports for the WEF shall be placed on the inside of the work area to prevent salt marsh harvest mouse from climbing the stakes into the work area.

- The salt marsh harvest mouse-proof WEF shall be at least 2 feet high but no higher than 4 feet.
- The fencing shall be made of a heavy plastic sheeting material that is too smooth for salt marsh harvest mouse to climb.
- The toe of the fence shall be buried approximately 6 or 8 inches in the ground to prevent salt marsh harvest mouse from crawling or burrowing underneath it.
- A 4-foot buffer shall be maintained free of vegetation around the exclusion fencing and work areas.
- The final design and proposed location of the fencing shall be reviewed and approved by USFWS prior to placement.
- WEF is not required where temporary construction mats are placed in marsh vegetation below Sonoma Creek Bridge.

Where temporary construction mats would be placed in potential salt marsh harvest mouse habitat, the following measures will be implemented:

- A project biologist will be on-site during all placement and removal of temporary construction mats.
- The project biologist will work in front of mats to be placed in potential salt marsh harvest mouse habitat immediately before and during placement to determine presence of any salt marsh harvest mouse individuals.
- If a salt marsh harvest mouse is discovered within 50 feet of the work area, work will stop immediately, and the individual(s) will be allowed to leave the work area on their own. Work may proceed only after the project biologist has confirmed that all salt marsh harvest mouse have left within the work area and a 50-foot perimeter.
- During mat removal, the project biologist will inspect locations where mats are removed to confirm that no salt marsh harvest mice are present where mats are placed prior to, during, and immediately after mat removal.

**BIO-29: Salt Marsh Harvest Mouse Monitoring Protocols.** The following protocols will be followed during biological monitoring at project locations where salt marsh harvest mouse identified in pre-construction surveys may occur:

- A project biologist with previous salt marsh harvest mouse experience will be on site during all construction activities.
- Salt marsh harvest mouse is a fully protected species under CFGC and may not be handled or captured at any time.

- If any small mouse is discovered during construction, work will cease in the immediate vicinity of the individual until CDFW and USFWS are contacted or the individual(s) leave the work area on their own.
- The project biologist will oversee installation of WEF for salt marsh harvest mouse.
- Salt marsh harvest mouse WEF will be checked daily to ensure that it has no holes and its base remains buried; the fence will be inspected to ensure that no mice are trapped. If a mouse is trapped by the fence, work will stop within 50 feet of the discovery, and the project biologist will monitor the individual(s) until they move away from the immediate work area.
- During vegetation removal in wetlands covered with pickleweed and/or salt grass (or other potential mouse habitat, as determined by project permits or the project biologist), the project biologist will mark and inspect areas to be cleared immediately prior to vegetation removal, and oversee removal work to ensure that salt marsh harvest mice and nests are clear of the work area.
- All vegetation removal will proceed away from the work area and toward contiguous areas of suitable habitat to allow any salt marsh harvest mice in the exclusion area to passively relocate into adjacent habitat.
- Initial removal of pickleweed, salt-grass, and other vegetation in the marked areas will be done using hand tools exclusively. Initial removal may commence until topsoil is visible.
- After initial removal is complete and once topsoil is visible, mowing with a string trimmer or mower may proceed (if necessary), with the project biologist walking in front of the mower and stopping work as needed to allow mice to relocate.

## Proposed Compensation to Offset Salt Marsh Harvest Mouse Habitat Loss

Caltrans proposes to mitigate for permanent loss of salt marsh harvest mouse habitat at a 3:1 restoration/enhancement to impact area ratio, or at a ratio determined appropriate in coordination with USFWS and CDFW, through a project-specific compensation plan. Salt marsh harvest mouse habitat that would be lost in the project area is limited to forage habitat adjacent to the roadway and fringe salt marsh adjacent to Sonoma Creek that experience consistent human disturbance from foot traffic accessing the creek. Salt marsh harvest mouse habitat adjacent to the project area is relatively high in abundance and experiences less disturbance compared to habitat next to the roadway. The measure proposed here also applies to the California black rail that uses the same habitat as the salt marsh harvest mouse.

There are no conservation banks with salt marsh harvest mouse credits available in the project's service area. Additionally, there are no approved in-lieu fee programs to compensate for impacts to salt marsh harvest mouse. Caltrans proposes to offset impacts and losses to salt marsh harvest mouse and California black rail habitat through a project specific compensation plan that would fund nearby tidal restoration and enhancement efforts within the project's watershed(s). This would be achieved through coordination with specific restoration project owners, USFWS, and CDFW to determine appropriate funding targets, define appropriate endowments, and develop an in-lieu-fee program specific to the project. Caltrans has identified several potential projects that could be funded to offset and compensate for loss of salt marsh habitat from the selected alternative. These include efforts in the Refuge (Mare Island, Cullinan Ranch, Strip Marsh, Skaggs Island, or Tolay Lagoon), or efforts being conducted through the Sonoma Creek Baylands Strategy (Sonoma Creek Restoration at Detjen and West End) sponsored by the Sonoma Land Trust. Funding and transfers could be established through a cooperative agreement with the California State Coastal Conservancy.

Caltrans would restore temporarily disturbed salt marsh harvest mouse/California black rail habitat to pre-project conditions on site at a 1:1 ratio.

The following measures summarizes Caltrans commitment to offset impacts to salt marsh harvest mouse and California black rail habitat from the selected alternative:

**BIO-30.** Compensation for Salt Marsh Harvest Mouse and California Black Rail Habitat Effects. Caltrans will offset permanent loss and degradation of salt marsh harvest mouse and California black rail habitat in the project area at a 3:1 impact to restoration/enhancement ratio, or at a ratio determined appropriate in coordination with USFWS and/or CDFW. Compensation will be provided through a project-specific plan that provides in-lieu funding to a nearby restoration program or restoration project that would create, restore, or enhance resources adversely affected by the project. Appropriate compensation will be determined in coordination with state and federal environmental regulatory agencies with jurisdiction.

Caltrans will offset temporary impacts during construction to salt marsh harvest mouse and California black rail habitat by restoring disturbed areas to pre-project conditions at a 1:1 ratio.

## 2.4.5.12 Affected Environment: Chinook Salmon

The Chinook Salmon Central Valley Spring-Run ESU spawns in the upper Sacramento River and its larger tributaries (NMFS 2014). Adult Central Valley Spring-Run Chinook Salmon return from the ocean to begin their upstream migration in late January and early February (CDFW 1998) and enter the Sacramento River between March and September, primarily in May and June (Moyle 2002; Yoshiyama et al. 1998). Spawning normally occurs between mid-August and early October, peaking in September (Moyle 2002). Spring-run fry emerge from the gravel from November to March (Moyle 2002). Juveniles may reside in freshwater for 12 to 16 months, but some migrate to the ocean as young-of-the-year in the winter or spring months within 8 months of hatching (CALFED 2000). Studies in Butte Creek (Ward et al. 2003) found most spring-run migrants to be moving downstream primarily from December to February, and that these movements appeared to be influenced by flow. The Chinook Salmon Sacramento River Winter-Run ESU adults migrate into San Francisco Bay and up the Sacramento River starting in November (Fukushima and Lesh 1998). They complete their upstream migration by May, with spawning occurring from April to August. Incubation of the eggs is from April to October, with fry emergence in June through October. Juveniles remain in the Delta until they are 5 to 10 months of age. Emigration to the ocean begins as early as November and continues through May (Fisher 1994; Myers et al. 1998).

Adult Chinook salmon of both ESUs may occasionally use San Pablo Bay as a migratory corridor to and from their spawning areas. However, adult Chinook salmon are not likely to occur in Sonoma Creek, Upper Tolay Lagoon, or the other tidal wetlands of the BSA because these waters do not lie along a migratory pathway, and migrating adults typically follow the deeper channels in the estuary during migration. Migrating adults could be seasonally present in the Mare Island Strait (Jones and Stokes 2004), outside of and east of the BSA.

NMFS-designated critical habitat for Chinook Salmon Central Valley Spring-Run ESU (NMFS 2005) and Chinook Salmon Sacramento River Winter-Run ESU (NMFS 1993) in the BSA includes all tidal waters of San Pablo Bay west of the Carquinez Bridge; and all tidal waters of San Francisco Bay north of the San Francisco–Oakland Bay Bridge, from San Pablo Bay to the Golden Gate Bridge.

## 2.4.5.13 Environmental Consequences: Chinook Salmon

## **Direct Effects**

All build scenarios have a low potential to affect Chinook salmon. Chinook are unlikely to occur in the project footprint because in-water work would be limited to the shallow margins of intertidal habitat. Additionally, the project schedule proposed to adhere to an in-water work window that would avoid the primary juvenile out-migration period of April and May. The fill required under all Build Alternatives for the project in tidal wetlands and waters would affect potential foraging habitat for anadromous fish species.

All project Build Alternatives would place fill in foraging habitat that is marginal for this species due to poor water quality resulting from limited tidal exchange and limited accessibility for fish from the Bay. Under Alternative 3B, fill also would be placed to widen the existing bridge in the open waters of Sonoma Creek which provides high quality foraging habitat. However, the BSA is not in the likely migratory pathway for Chinook salmon.

The project Build Alternatives have been designed to reduce potential impacts to intertidal habitats and prevent injury to fish from pile-driving activities, to avoid unnecessary turbidity increases from disturbances to intertidal areas, and to use BMPs to maintain water quality standards during in-water construction activities. Table 2-50 summarizes estimated permanent impacts to habitat that supports Chinook salmon in the BSA. Table 2-51 summarizes estimated temporary impacts to habitat that supports Chinook salmon in the BSA.

Habitat Type	Alternative 1 (Acres)	Alternative 2 (Acres)	Alternative 3A (Acres)	Alternative 3B (Acres)		
Tidal Waters	0.02	0.05	0.06	1.7		
Tidal Wetlands	1.03	1.43	1.49	2.08		
Total	1.05	1.48	1.55	3.78		

# Table 2-50Estimated Permanent Impacts to Anadromous Fish Habitat in the<br/>BSA

Note:

All acreages are rounded up to the nearest 0.01 acre before summing.

# Table 2-51Estimated Temporary Impact Areas to Anadromous Fish Habitat in<br/>the BSA

Habitat Type	Alternative 1 (Acres)	Alternative 2 (Acres)	Alternative 3A (Acres)	Alternative 3B (Acres)
Tidal Waters	1.37	1.93	1.71	3.00
Tidal Wetlands	2.44	2.79	2.17	1.98
Total	3.81	4.72	3.88	4.98

Note:

All acreages are rounded up to the nearest 0.01 acre before summing.

### Direct Effects Exclusive to Alternative 3B: Impact Pile Driving

On July 8, 2008, the Fisheries Hydroacoustic Working Group, whose members include the Southwest and Northwest Divisions of NMFS; the California, Washington, and Oregon Departments of Transportation; CDFW; and FHWA—issued an agreement for establishment of interim threshold criteria to determine the effects of high-intensity sound on fish. Although these criteria are not formal regulatory standards, they are generally accepted as viable criteria for underwater sound pressure effects on fish. The agreed-on threshold criteria for impulse-type sound pressure to harm fish have been set at a 206 dB peak for fish of all weights, 187 dB cumulative sound exposure level (cSEL) for fish greater than 2 grams, and 183 dB cSEL for fish less than 2 grams.

The Fisheries Hydroacoustic Working Group determined that sound pressure at or above the 206 dB peak level can cause barotrauma to auditory tissues, the swim bladder, or other sensitive organs. Sound pressure levels above the accumulated sound exposure level (SEL) may cause temporary hearing threshold shifts in fish. Behavioral effects are not covered under these criteria but could occur at these levels or lower. Behavioral effects may include fleeing and the temporary cessation of feeding or spawning behaviors. There are no formal SEL thresholds established for non-impulsive sound pressure, such as vibratory pile driving because implementation of such methods is an accepted impact minimization measure. Therefore, vibratory driving of sheet piles and steel pipe piles proposed for the project are not considered impacts because the underwater noise generated with that method that would not cause take of listed fish species and are not analyzed here. Under Alternative 3B, pile driving scenarios required at Tolay Creek Bridge would include:

• At Tolay Creek Bridge, impact pile driving may be used to drive steel pipe piles for support of the enlarged bridge abutments. Water levels at this location are assumed to be less than 2 feet, and attenuation using conventional methods is assumed to be effectively infeasible.

For listed fish species in the BSA, exposure to underwater sound that exceeds the 206 dB peak or 187 dB cSEL thresholds during impact pile driving that would be required to do abutment work at Tolay Creek Bridge. Underwater pressure levels would be sufficient to cause injury if fish were to occur near impact pile driving activities. Source information from the project's conceptual design and pressure measurement taken from previous Caltrans pile driving activities were used to estimate distance to fish injury criteria. All values provided are preliminary estimates based on conceptual assumptions; actual distance would be refined if Alternative 3B is the selected Build Alternative. Table 2-52 summarizes preliminary estimated distances for the proposed impact pile driving that would be require for Alternative 3B.

# Table 2-52Alternative 3B Estimated Underwater Sound Pressure Effects on<br/>Fish Habitat from Impact Pile Driving

Assumed Pile and Work Description	Pile Driving Method	Distance To 187 dB cSEL Criterion (feet)	Distance to 206 dB Peak Criterion (feet)
Medium steel pipe piles at Tolay Creek Bridge	Impact	95	10
Pile proofing of small (12-inches or less) steel shell piles for temporary structures	Impact (limited and with attenuation)	15	0

Notes:

All estimated distances are based on conceptual-level assumptions and considered to be preliminary and conservative for use in this effect analysis.

## Direct Effects Exclusive to Alternative 3B: Shading

For Alternative 3B, there would be an increase in overwater shading due to the widening of the Tolay Creek and Sonoma Creek bridges (Table 2-53). The permanent shading is not expected to affect the productivity of open waters in the BSA because river flow, tidal, and wind-wave circulation would move water continuously through the narrow area of daily and seasonally shifting shadow. There is no submerged aquatic vegetation in the vicinity of the Sonoma Creek Bridge or the Tolay Creek Bridge. The widened bridges would have the same vertical clearance to the existing bridges. Shading may have some effect on the composition of benthic organisms in the affected area, but such changes are not expected to substantially alter the value of habitat in the BSA.

# Table 2-53Alternative 3B Shading Impacts to Anadromous Fish Habitat in the<br/>BSA

Habitat Type	Temporary Shading (Acres)	Permanent Shading (Acres)
Tidal Waters	0.29	0.17
Tidal Wetlands	0.09	0.10
Total	0.38	0.27

Note: All acreages are rounded up to the nearest 0.01 acre before summing.

### **Indirect Effects**

No new barriers to dispersal for Chinook salmon would occur because of project components. The planned activities are not anticipated to affect the long-term mobility of Chinook salmon in the area. Therefore, indirect project effects on Chinook salmon are considered discountable.

### 2.4.5.14 Avoidance, Minimization, and/or Compensation: Chinook Salmon

Under all Build Alternatives, the following previously described measures to protect wetland habitat and wildlife in the project area would also serve to avoid and minimize potential impacts on salt marsh harvest mouse: BIO-01: Wetlands Protection – Invasive Plants; BIO-02: Wetland Protection; BIO-04: Estuarine Dewatering Work Window; BIO-05: Turbidity Control;; BIO-15: Stop-Work Authority; BIO-16: Worker Environmental Awareness Training; and BIO-17: Discovery of Injured or Special-Status Species.

With implementation of the measures proposed here, compensation for impacts to Chinook salmon and other anadromous fish species with potential to occur in the project area would be limited to the permanent fill impacts in fish habitat for each Build Alternative. Compensation to offset permanent unavoidable loss of Chinook salmon and other anadromous fish habitat is proposed in this section.

### Measures for all Build Alternatives

The following measures would be implemented to avoid and minimize potential impacts to anadromous fish in the project area under all Build Alternatives:

**BIO-31: Vibratory Pile Driving.** Whenever possible, piles will be installed and removed using a vibratory hammer or direct push methods. All sheet piles will be installed with a vibratory driver or direct-push methods. In upland areas out of waters and wetlands, an impact hammer may be used if the vibratory hammer cannot adequately install the pile.

**BIO-32: In-Water Sheet Pile Fish Entrapment Avoidance.** When sheet piles are installed below MHHW, they will be installed in a way that avoids fish entrapment (e.g., by closing off pile walls during low tide) The NMFS-approved project biologist will be present during any sheet pile installation below MHHW.

**BIO-33: Fish Monitoring.** During dewatering where fish may be present and impact pile-driving work, a NMFS-approved project biologist will be on site to observe work for conformance with permits and authorizations, and to monitor for any potential fish take.

**BIO-34:** Fish Relocation. At least 90 days prior to the start of in-water work with potential to strand or entrap fish, Caltrans will develop a fish relocation plan and submit it to NMFS for approval. If NMFS provides no comments on the proposed plan within 60 days, it will be considered approved and implemented as submitted. All biologists monitoring dewatering actions will be qualified and approved by NMFS to conduct fish collections in a manner that minimizes all potential risks to listed fish. The NMFS-approved project biologist(s) will be on-site to observe dewatering activities and to capture/rescue any fish that are observed in isolated areas during dewatering activities.

### Proposed Compensation to Offset Habitat Loss to Listed Anadromous Fish Species

Caltrans proposes to mitigate for permanent loss of anadromous fish habitat in marginal habitat at a 2:1 restoration/enhancement to impact ratio; and at a 3:1 ratio, or at ratios determined appropriate in coordination with regulatory agencies with jurisdiction, for areas where habitat has greater tidal exchange and accessibility for anadromous fish. A relatively lower ratio is proposed for fish habitat in the project area because it is of marginal quality. Potentially affected marginal habitat occurs primarily at the tidal margins of the Caltrans right of way for SR 37. The project would place fill in mostly shallow intertidal areas that are not easily accessible to anadromous fish. Additionally these areas have limited tidal exchange with the Bay and likely experience higher water temperatures, decreased dissolved oxygen, and greater salinity fluctuations, particularly during the summer months. These factors reduce habitat suitability for anadromous fish species where these conditions occur.

The tidal areas in the project footprint with more consistent tidal exchange and greater accessibility to anadromous fish that would be mitigated at a 3:1 ratio, or at a ratio determined appropriate in coordination with regulatory agencies with jurisdiction, are limited to:

- A small portion of the margin of Upper Tolay Lagoon, on the southern side of SR 37
- Sonoma Creek Channel, where it is crossed by SR 37
- Small portions of the 1.25-mile-long tidal slough that parallels SR 37 to the south, just east of the Sonoma Creek Bridge; and
- The southern margin of the NSMWA Intake Ponds 1 and 1A, and Cullinan Ranch Ponds, to the north of SR 37.

There are no species-specific conservation banks in the project service area with credits available for Chinook, steelhead, green sturgeon, or Delta smelt. One private

conservation bank, Liberty Island Conservation Bank, with credits available for longfin smelt was identified. No approved in-lieu fee programs to compensate for impacts to listed anadromous fish species were identified in the Service Area. Caltrans believes that a single approach to offset habitat impacts for all special status anadromous fish would provide greater benefits to these species than splitting compensation efforts between a project specific plan and purchase of conservation bank credits because the habitat impacted provides some benefit to all anadromous fish species in the project area.

Caltrans proposes to offset habitat impacts and losses to special status anadromous fish species through a project specific compensation plan that would fund nearby tidal restoration and enhancement efforts within the project's watershed(s). This would be achieved through coordination with specific restoration project owners, NMFS, USFWS, and CDFW to determine appropriate funding targets, define appropriate endowments, and develop an in-lieu-fee program specific to the project. Caltrans has identified several potential projects that could be funded to meet compensatory offset for loss of special status anadromous fish habitat from the selected alternative. These include efforts in the Refuge (Mare Island, Cullinan Ranch, Strip Marsh, Skaggs Island, or Tolay Lagoon), or efforts being conducted through the Sonoma Creek Baylands Strategy (Sonoma Creek Restoration at Detjen and West End) sponsored by the Sonoma Land Trust. Funding and transfers could be established through a cooperative agreement with the California State Coastal Conservancy.

The following measure summarizes Caltrans commitment to offset impacts to state and federally listed anadromous fish habitat from the selected alternative:

**BIO-35.** Compensation for Chinook Salmon, Steelhead, Green Sturgeon, Longfin Smelt and Delta Smelt Habitat. Caltrans will offset permanent loss of state and/or federally listed anadromous fish species (Chinook salmon, steelhead, green sturgeon, longfin smelt and Delta smelt) habitat through a 3:1 restoration/enhancement to impact ratio, or at a ratio determined appropriate in coordination with regulatory agencies with jurisdiction, where permanent loss or degradation of habitat occurs at:

- Upper Tolay Lagoon
- Sonoma Creek
- The 1.25-mile-long tidal slough east of Sonoma Creek Bridge that is parallel to and south of SR 37
- NSMWA Intake Ponds 1 and 1A
- Cullinan Ranch Ponds

All other permanent loss or degradation of anadromous fish habitat from the project will be compensated at a 2:1 restoration/enhancement to impact ratio, or at a ratio determined appropriate in coordination with regulatory agencies with jurisdiction.

Caltrans will offset temporary impacts during construction to anadromous fish habitat by restoring disturbed areas to pre-project conditions at a 1:1 ratio.

### Measures for Impact Pile Driving

In-water impact pile driving is anticipated for construction to widen Tolay Creek Bridge, which would require additional measures to avoid and minimize potential impacts to anadromous fish species. These would include

**BIO-36: In-Water Impact Pile-Driving Work Window.** Impact pile driving in wetlands and waters will be limited to June 1 through November 30 during daylight hours; vibratory pile driving will not be limited to a work window.

**BIO-37: In-Water Impact Pile-Driving Attenuation.** All in-water impact pile driving in water depths greater than 2 feet at any time during work will use an underwater sound pressure attenuation system (e.g., a dewatered cofferdam or a bubble curtain system).

**BIO-38: Hydroacoustic Monitoring.** During all impact pile-driving events, Caltrans will monitor in-water sound pressure levels relative to the 187 dB cSEL and 206 dB peak pressure level. A hydroacoustic monitoring plan for impact pile driving will be developed and provided at least 90 days prior to impact pile driving for review and approval by NMFS. If NMFS provides no comments on the proposed plan within 60 days, it will be considered approved and implemented as submitted. Vibratory pile driving will not be monitored.

**BIO-39: Pile Proofing**: Caltrans may propose a minimal amount of attenuated in-water pile proofing to construct temporary structures during fish migration periods (e.g., outside of the proposed impact pile driving work window). Pile proofing outside of the impact pile-driving work window would be consistent with accepted guidance from USACE and NMFS (USACE Proposed Additional Procedures and Criteria for Permitting Projects under a Programmatic Determination of Not Likely to Adversely Affect Select Listed Species in California [the 2018 NLAA Program]) (USACE 2018). Pile proofing, if necessary, will be limited to the following.

- All temporary in-water piles must be driven using vibratory methods to the greatest extent possible.
- Steel pipe piles (or H piles) of 12-inch diameter or less will be used.
- No more than 20 piles per day will be driven.
- A marine attenuation system (e.g., bubble curtain or similarly effective methods) will be used in water depths greater than 2 feet.
- Piles driven in intertidal areas where water is less than 2 feet will only be proofed during low-tide or low-low tide events.
- A hammer that is 3,000 pounds or smaller will be used.
- A plastic or wood cushion block will be used between the hammer and the pile.
- Only a single hammer will be used per day.

Impacts to fish are anticipated to be less than adverse with implementation. If Caltrans elects to implement Alternative 3B and this measure, it will provide a complete analysis and impact assessment for state and federally listed fish species impacts during its final design phase and obtain all necessary permits and authorizations prior to construction.

## 2.4.5.15 Affected Environment: Steelhead Central California Coast and California Central Valley Distinct Population Segments

The range of Central California Coast steelhead is defined by the NMFS as all naturally spawned steelhead populations from the Russian River south to Aptos Creek in Santa Cruz County, including drainages of San Francisco, San Pablo, and Suisun Bays eastward to Chipps Island at the confluence of the Sacramento and San Joaquin Rivers. NMFS defines the range of Central Valley steelhead as all populations of steelhead that spawn in the basins of the Sacramento and San Joaquin Rivers (inclusive of and downstream of the Merced River) in California's Central Valley. Steelhead typically prefers steeper gradient stream reaches, farther upstream and farther up tributaries than Chinook or Coho salmon, and the species can spawn in either the mainstem rivers or farther up into tributaries.

Adult steelhead typically begin returning to San Francisco Bay in late fall, and immigration mostly occurs from December through February. Spawning takes place from January through April in freshwater streams. Juvenile steelhead are found in all habitat types, and habitat preferences change with seasonal changes in stream conditions. Estuaries often are an important rearing area for juvenile steelhead on their way to the ocean. Adult steelhead abundance in San Francisco Bay increases from late fall through February. Juvenile steelhead migrate as smolts to the ocean from January through May, with peak outmigration occurring in March and April (Fukushima and Lesh 1998). Estuarine habitat present in the BSA has limited connectivity to San Pablo Bay, reducing the accessibility of those areas to out-migrating juveniles.

There are two occurrences of adults of this species within 5 miles of the BSA in North Slough, approximately 3.5 miles north of the BSA. The Napa River, Petaluma River, and Sonoma Creek are in the vicinity of the BSA and support Central California Coast steelhead runs (Jones and Stokes 2004). Of these, only Sonoma Creek would be impacted by the project. The Tolay Creek floodplain would be affected the project, but Tolay Creek lacks the hydrology and cold-water habitat characteristics needed for freshwater spawning and rearing.

Adult steelhead are known to use the San Pablo Bay and the Sonoma Creek channel as a migratory corridor to and from their spawning areas (Jones and Stokes 2004) but are likely to quickly move along the deeper channels of these water bodies. Juvenile steelhead may also pass through these areas during outmigration and may spend more time in estuarine waters than adults, foraging as they move to the ocean. Juvenile steelhead also go through the smoltification process (physiological changes to adapt from living in freshwater to living in seawater) during this time and may linger in the Sonoma Creek Channel and San Pablo Bay. In general, adult steelhead pass through San Francisco Bay during the winter months and out-migrating smolts typically pass through during April and May (NMFS 2016a).

Preliminary tagging studies conducted with steelhead smolts found the average residence time between Rio Vista and the Golden Gate to be 8.5 days, but variation is large, with some individuals spending a month in the estuary (Klimey et al. 2010). Deep water and channels can serve as migration corridors (Klimey et al. 2010), but the only portion of the BSA with such features is the Sonoma Creek channel under Sonoma Creek Bridge. Central California Coast steelhead are known to occur seasonally in Sonoma Creek, which supports a spawning population of the species. Elsewhere, a moderate potential exists for the species to occur in the tidally connected estuarine waters of the BSA, including the tidally influenced reaches of sloughs and waters associated with San Pablo Bay. However, the species is not expected to be seasonally present during the estuarine in-water work window.

Critical habitat was designated for these DPSs of steelhead in 2005 (NMFS 2005). Because the designated critical habitat for steelhead includes San Francisco Bay and the tidally influenced reaches of tributaries of the Bay, the BSA is in designated critical habitat. Critical habitat for steelhead includes freshwater spawning areas, freshwater rearing and migration areas, and estuarine rearing and migration areas. All tidally influenced surface waters accessible to steelhead that overlap the BSA are considered to be in critical habitat for these species.

## 2.4.5.16 Environmental Consequences: Steelhead

The environmental consequences for steelhead are the same as those described for Chinook salmon in Section 2.4.5.13.

## 2.4.5.17 Avoidance, Minimization, and/or Compensation: Steelhead

The AMMs and compensation to offset permanent unavoidable effects from the project Build Alternatives are the same as those described for Chinook salmon in Section 2.4.5.14. No additional measures are proposed for steelhead.

## 2.4.5.18 Affected Environment: North American Green Sturgeon

Green sturgeon are nocturnal benthic feeders and in estuaries may feed on amphipods, shrimp, clams, or anchovies. San Francisco Bay serves as an important habitat for all life stages of green sturgeon, supporting rearing and serving as an important migratory/connectivity corridor between the Sacramento River system and nearshore coastal marine waters (Moyle et al. 1992).

Subadult green sturgeon (4 to 15 years old) are known to range along the Pacific Coast and move into estuaries like San Pablo Bay during periods of cold water upwelling off the coast, apparently to avoid the cold water. Juvenile green sturgeons move throughout the Delta and San Francisco Bay during their first 3 to 4 years of life, before they move into the ocean as subadults. During this early life stage, they may be found in the Bay throughout the year. Because of known life-history patterns, the species is assumed to have moderate potential to occur in the tidally influenced openwater portions of the BSA.

Designated critical habitat includes all tidally influenced areas of San Francisco Bay, San Pablo Bay, and Suisun Bay, up to the elevation of the mean high water mark, including but not limited to areas upstream from the head of tide endpoint in numerous creeks (NMFS 2009). All tidally influenced waters and marshland below the mean high water elevation in the BSA are considered to be in critical habitat for this species.

## 2.4.5.19 Environmental Consequences: Green Sturgeon

The environmental consequences for green sturgeon are the same as those described for Chinook salmon in Section 2.4.5.13.

## 2.4.5.20 Avoidance, Minimization, and/or Compensation: Green Sturgeon

The AMMs and compensation to offset permanent unavoidable effects from the project Build Alternatives are the same as those described for Chinook salmon in Section 2.4.5.14. No additional measures are proposed for steelhead.

## 2.4.5.21 Affected Environment: Delta Smelt

Delta smelt (*Hypomesus transpacificus*) is found in open waters of bays, tidal rivers, channels, and sloughs of the San Francisco Bay-Delta (USFWS 1996). Delta smelt typically occupy open surface water habitat with salinities lower than 12 ppt, most often at salinities less than 2 ppt, and they move toward the shallow edge waters and slow-moving sloughs to spawn (CDFW 2019; USFWS 1996). When not spawning, Delta smelt are found where saltwater and freshwater mix. Adults migrate to freshwater areas to spawn between January and July (USFWS 1996; Merz et al. 2011). During their spawning migration, adults move into freshwater channels and sloughs between December and January (USFWS 1996). Spawning occurs between January and July (USFWS 1996). Recent survey data have shown evidence of spawning adults in the Lower Napa River, as well as juveniles and adults in San Pablo Bay (Merz et al. 2011).

There are five occurrences within 5 miles of the BSA, one of which overlaps the BSA. Two occurrences outside of the BSA are in in tidal channels that overlap the BSA. A third occurrence in 1998 was one fish taken from a man-made tidal channel approximately 500 feet south of SR 37. Water column surveys in the open waters of San Pablo Bay have detected all Delta smelt life stages, except for pre-spawning and spawning adults (CDFW 2019; Merz et al. 2011). A moderate potential exists for the species to occur in all tidally influenced open water portions of the BSA. However, during the in-water work window of June 1 through November 30, there is a low potential of occurrence due to relatively low population numbers, low tidal exchange at Tolay Creek and Sonoma Creek, and likely prohibitively high salinities in tidal channels during this time frame (Murphy and Hamilton 2013).

The BSA does not overlap with designated critical habitat for Delta smelt (USFWS 1994) The nearest unit is located 4.8 miles southeast of the BSA in the Carquinez Strait.

## 2.4.5.22 Environmental Consequences: Delta Smelt

The environmental consequences for Delta smelt are the same as those described for Chinook salmon in Section 2.4.5.13. However, there would be no effect to Delta smelt Critical Habitat because it is outside of the BSA.

## 2.4.5.23 Avoidance, Minimization, and/or Compensation: Delta Smelt

The AMMs and compensation to offset permanent unavoidable effects from the project Build Alternatives are the same as those described for Chinook salmon in Section 2.4.5.14. No additional measures are proposed for Delta smelt.

## 2.4.5.24 Affected Environment: Soft Bird's-Beak

Soft bird's-beak is a federally endangered, California State Rare, and CRPR 1B.2 herbaceous annual plant in the broomrape family (Orobanchaceae) (CNPS 2021). Soft bird's-beak is hemiparasitic and grows with a broad range of host plants that are actively growing during its flowering and fruiting stages. Known hosts include many include common pickleweed, salt grass, and fleshy jaumea. Winter annual species and many nonnative annual grasses and forbs are generally not suitable host plants, because they are often dying by the time soft bird's-beak is flowering and fruiting (USFWS 2013a). Soft bird's-beak is restricted to coastal salt marshes habitats, where it occurs in colonies or subpopulations that can shift from year to year (USFWS 2013a). Occurrences are distributed along San Pablo Bay, Grizzly Bay, and the delta regions of their tributaries, including the Napa and Sonoma Rivers. There are currently 27 known occurrences of this plant, with 19 of them presumed to be extant (CDFW 2019). Soft bird's-beak is known to flower from June to November (CNPS 2021).

Soft bird's-beak was not observed during rare plant surveys conducted in the BSA in late September 2019. Because its flowering period extends into November, it would likely have been observed during the rare plant survey if there were populations present in the BSA.

Critical habitat for soft bird's-beak was designated by USFWS in April 2007 (USFWS 2007). There is no critical habitat for this species in the BSA. The closest critical habitat is along Interstate 780 in Solano County, approximately 5 miles southeast.

## 2.4.5.25 Environmental Consequences: Soft Bird's-Beak

The proposed project is unlikely to have any direct or indirect impacts to soft bird'sbeak because most potentially suitable habitat in the BSA for this plant occurs outside of impact areas. Because additional pre-construction surveys would be conducted prior to construction during the plant's blooming period in the BSA, soft bird's-beak, if present, would be detected prior to ground-disturbing activities. No substantial impacts to soft bird's-beak are anticipated.

## 2.4.5.26 Avoidance, Minimization, and/or Compensation: Soft Bird's-Beak

Previously described measures for special-status plant species, particularly **BIO-08**: **Targeted Pre-Construction Plant Survey; and BIO-09 Special Status Plant Monitoring** would serve to avoid and minimize potential impacts to soft bird's-beak.

### 2.4.5.27 Affected Environment: Longfin Smelt

The longfin smelt (*Spirinchus thaleichthys*) is listed as state threatened and is a federal candidate for listing. This species is historically found in the San Francisco Estuary, Sacramento-San Joaquin Delta, Humboldt Bay, and the estuaries of the Eel and Klamath Rivers (USFWS 1996). Adult longfin smelt occur in bays, estuaries, and nearshore coastal areas, and migrate into freshwater rivers to spawn from January through March (USFWS 1996). Adult and juvenile longfin smelt primarily use the middle or bottom of the water column in salt or brackish water; larval smelt concentrate near the surface of brackish waters (USFWS 1996). Spawning takes place in freshwater, with high outflows dispersing larvae and juveniles into rearing habitat in Suisun and San Pablo Bays (USFWS 1996).

There are three CNDDB occurrences of longfin smelt within 5 miles of the BSA. Water column surveys in the open waters of San Pablo Bay and Mare Island Strait have detected longfin smelt (CDFW 2019), and surveys in open water areas north of the BSA have detected all life stages (Merz et al. 2013). Sonoma Creek has deep, cool waters that provide spawning habitat for the species, but individuals are likely to quickly move along the deeper channels. A moderate potential exists for the species to occur in all tidally influenced open water portions of the BSA. However, during the inwater impact pile driving work window of June 1 through November 30, there is a low potential of occurrence due to relatively low seasonal presence at Tolay and Sonoma Creek, low tidal exchange at Tolay Creek, and likely prohibitively high salinities in tidal channels during this time frame (Murphy and Hamilton 2013).

## 2.4.5.28 Environmental Consequences: Longfin Smelt

The environmental consequences for longfin smelt are the same as those described for Chinook salmon in Section 2.4.5.13.

### 2.4.5.29 Avoidance Minimization and/or Compensation: Longfin Smelt

The AMMs and compensation to offset permanent unavoidable effects from the project Build Alternatives are the same as those described for Chinook salmon in Section 2.4.5.14. No additional measures are proposed for longfin smelt.

## 2.4.5.30 Affected Environment California Black Rail

California black rail is a state threatened and state fully protected species. California black rail habitat generally includes salt marshes, freshwater marshes, and wet meadows. Most California populations are nonmigratory, and these habitat types provide areas for breeding, foraging, and overwintering. Near tidal areas, the rails also require a dense cover of upland vegetation to provide protection from predators when the birds must leave marsh habitats during high tides. Typical associated vegetation includes pickleweed and bulrush. California black rail forages in the same habitats that it uses for breeding. This species begins breeding in February; nesting occurs from March to June. Nests often are concealed in dense vegetation, often pickleweed, near the upper limits of tidal flooding.

There are 14 occurrences of the California black rail within a 5-mile radius of the BSA, two of which overlap the BSA at Tolay Creek channel, the SLC-leased Refuge, and Strip Marsh. California black rail has a moderate potential to occur at Tolay Creek, Upper Tolay Lagoon, Sonoma Creek, and in portions of SLC-leased Refuge and Strip Marsh (Marriott, M., per. comm, 2020). These locations contain portions of foraging and dispersal tidal marsh habitat. This species has low to no potential to occur in the remainder of the BSA.

## 2.4.5.31 Environmental Consequences: California Black Rail

The environmental consequences for California black rail are the same as those described for Ridgway's rail in Section 2.4.5.7 because these sensitive marsh birds share the same habitat.

## 2.4.5.32 Avoidance, Minimization, and/or Compensation: California Black Rail

The avoidance and minimization measures proposed for all threatened and endangered species in Section 2.4.5.11, for Ridgway's rail in Section 2.4.5.8, and for the salt marsh harvest mouse in Section 2.4.5.5 would also server to project California black rail. Specific compensation is proposed for loss of potential California black rail habitat as **BIO-30 Compensation for Salt Marsh Harvest Mouse and California Black Rail Habitat Effects**. No additional measures are proposed for California black rail.

## 2.4.5.33 Affected Environment: Swainson's Hawk

The Swainson's hawk (*Buteo swainsoni*) is listed as state threatened. It is breeding resident adapted to open grasslands and prairies (CDFW 2016). Swainson's hawk would also forage in managed wetlands during dry summer months (CDFW 2016). They often nest at the edge of riparian corridors with access to foraging habitat, but lone trees in agricultural fields and roadsides trees are often used as well (CDFW 2016). Valley oak, Fremont cottonwood (*Populus fremontii*), walnut (*Juglans* spp.), and willow are the tree species most commonly used for nesting (CDFW 2016). Breeding generally occurs from late March to late August (CDFW 2016).

There are no CNDDB occurrences within 5 miles of the BSA; however, AECOM biologists observed this species foraging in the BSA during field surveys. There are trees suitable for nesting in and adjacent to the BSA at Upper Tubbs Island and west of Tolay Creek. Swainson's hawk have been observed east of Tubbs Island, but because there is no nesting habitat, they would likely only be using this area for foraging. Swainson's hawk has a moderate potential to nest at Upper Tubbs Island and west of Tolay Creek.

## 2.4.5.34 Environmental Consequences: Swainson's Hawk

Under all Build Alternatives considered, vegetation removal may impact nesting Swainson's hawks. Non-nesting Swainson's hawk individuals may also be present though these individuals are less likely to be directly affected by project activities because of existing ambient conditions that include a high level of traffic and human disturbance along the 37 corridor. Implementation of pre-construction nesting bird surveys during the nesting season (early March to July) would minimize impacts to breeding birds.

## 2.4.5.35 Avoidance, Minimization, and/or Mitigation Measures: Swainson's Hawk

Previously described measures to protect tree and oak woodland communities, and migratory birds, would also serve to substantially avoid and minimize any potential effects to Swainson's hawk. These include BIO-03: Tree Replacement, Landscaping, and Revegetation Plan; BIO-10: Nesting Bird Protection; BIO-15: Stop-Work Authority; BIO-16: Worker Environmental Awareness Training; BIO-17: Discovery of Injured or Special-Status Species; and BIO-19: Construction Noise.

No compensation to offset impacts to Swainson's hawk is anticipated. The following specific measure is proposed for Swainson's hawk to avoid and minimize potential impacts during construction:

**BIO-40:** Swainson's Hawk Pre-Construction Surveys. Pre-construction surveys will be conducted within a 0.25-mile radius of Swainson's hawk nesting or forage habitat during the nesting season of February 1 through August 31. Surveys will be conducted in the following manner:

- Surveys will be conducted in accordance with The Swainson's Hawk Technical Advisory Committee's May 31, 2000, *Recommended Timing and Methodology for Swainson's Hawk Nesting Surveys in California's Central Valley.*
- Caltrans will conduct surveys during two survey periods immediately prior to initiating any project-related construction activity.
- If an active Swainson's hawk nest is discovered during surveys or monitoring, Caltrans will immediately contact CDFW to determine requirements on nest impact avoidance measures and work buffer distances.

## 2.4.6 Essential Fish Habitat

## 2.4.6.1 Regulatory Setting

Section 305(b)(2) of the MSA, as amended by the Sustainable Fisheries Act of 1996 (Public Law 104 267), requires federal agencies to consult with the NMFS on activities that may adversely affect EFH for federally managed fish species. These species include commercial fishes with established Fisheries Management Plans (FMPs) as managed by regional fisheries management councils. EFH includes those waters and substrate necessary for fish spawning, breeding, feeding, or growth to maturity. In the definition of EFH, "waters" include aquatic areas and their associated physical, chemical, and biological properties that are used by fish, and may include aquatic areas historically used by fish where appropriate; "substrate" includes sediment, hard bottom, structures underlying the waters, and associated biological communities; "necessary" means the habitat required to support a sustainable fishery and the managed species contribution to a healthy ecosystem; and "spawning, breeding, feeding, or growth to maturity" covers a species' full life cycle (NMFS 2016b).

## 2.4.6.2 Affected Environment

San Francisco Bay, including tidal areas in the BSA, is classified as EFH under the MSA and serves as habitat for at least 14 species of commercially important fish and sharks that are federally managed under two FMPs: the Pacific Groundfish FMP and the Coastal Pelagic FMP. The Coastal Pelagic FMP is designed to protect habitat for a variety of fish species that are associated with open coastal waters. Fish managed under this plan include plankton-feeding fish and their predators. The Pacific Groundfish FMP is designed to protect habitat for more than 90 species of fish, including rockfish, flatfish, groundfish, some sharks and skates, and other species that associate with both hard and soft substrates.

The entire San Francisco Bay is also classified as EFH for species managed under the Pacific Coast Salmon FMP, which includes Chinook salmon. The Pacific Salmon FMP is designed to protect habitat for commercially important salmonid species. Chinook salmon is the only one of these species that may be seasonally present in the BSA. Although evidence suggests that migrating salmonids move along the deeper channels of San Francisco Bay, migration behavior in estuaries is poorly understood. Outmigrating juveniles may forage in estuaries during migration, and they are more likely than migrating adults to enter shallow tidal areas.

In addition to EFH designations, San Francisco Bay is designated as a Habitat Area of Particular Concern for various fish species in the Pacific Groundfish and Coastal Pelagic FMPs because this estuarine system serves as breeding and rearing grounds important to these fish stocks.

## 2.4.6.3 Environmental Consequences

All Build Alternatives may affect EFH through impacts to water quality, and a the relatively small amount of permanent habitat loss resulting from the placement of fill in

tidal waters and wetlands. The project has been designed to avoid unnecessary turbidity increases from channel bed disturbance, to avoid debris falling into open water, and to use BMPs to maintain water quality standards during in-water construction activities.

Hydroacoustic impacts to widen Tolay Creek Bridge and overwater shading would affect EFH.

The project would impact EFH under all Build Alternatives by placing permanent fill into EFH. The impacted areas would be spread out along the alignment of SR 37, which borders thousands of acres of tidal waters and wetlands. These fill materials would also result in temporary effects to and loss of intertidal marsh or mudflat areas. Placement of rock slope protection may modify some intertidal areas from soft substrate to hard substrate habitat. This change in substrate may improve habitat for groundfish species associated with rocky substrate but decrease habitat suitability for others associated with soft substrate.

## 2.4.6.4 Avoidance, Minimization, and/or Mitigation Measures

Measures proposed to protect special-status fish species would also serve to protect EFH, and no additional measures are proposed. No compensation is required for impacts to EFH.

## 2.4.7 Invasive Species

## 2.4.7.1 Regulatory Setting

On February 3, 1999, President William J. Clinton signed EO 13112 requiring federal agencies to combat the introduction or spread of invasive species in the United States. The order defines invasive species as "any species, including its seeds, eggs, spores, or other biological material capable of propagating that species, that is not native to that ecosystem whose introduction does or is likely to cause economic or environmental harm or harm to human health." FHWA guidance issued August 10, 1999 directs the use of the State's invasive species list, maintained by the California Invasive Species Council to define the invasive species that must be considered as part of the NEPA analysis for a proposed project.

## 2.4.7.2 Affected Environment

A high abundance and diversity of NNIP species were observed throughout the BSA, particularly in upland areas adjacent to the elevated roadway. A total of 52 California Invasive Plant Council (Cal-IPC)-rated NNIP species occur in the BSA (Cal-IPC 2019). Of these, Cal-PIC rates eight species as "high," 23 species are rated "moderate," and 21 are rated "limited." The most abundant moderate and high-rated NNIPs and those with the greatest potential to spread in the BSA are discussed in this section.

All the upland semi-natural communities present in the project's BSA area are dominated by NNIP species. Although NNIP mapping was not conducted for this project, most of the large and dense infestations of high and moderate species are in upland semi-natural communities. Dominant NNIP species are found in each of these semi-natural communities. Smaller and lower-density patches of NNIP species are also found in both upland and wetland communities throughout the BSA. A complete list of moderate- and high-rated species is included in Table 2-54. No CDFA A- through C-rated noxious weeds (species that are considered pest species with known economic or environmental detriment) were observed in the BSA.

Scientific Name	Common Name	Cal-IPC Rating
Atriplex semibaccata	Australian saltbush	Moderate
Avena barbata	slender wild oat	Moderate
Brassica nigra	black mustard	Moderate
Bromus diandrus	ripgut brome	Moderate
Bromus madritensis ssp. rubens	foxtail brome	High
Carduus pycnocephalus	Italian thistle	Moderate
Carpobrotus edulis	iceplant	High
Centaurea calcitrapa	purple star thistle	Moderate
Centaurea solstitialis	yellow star-thistle	High
Cirsium vulgare	bullthistle	Moderate
Conium maculatum	poison hemlock	Moderate
Cortaderia jubata	Andean pampas grass	High
Cotoneaster sp.	unknown cotoneaster	Moderate
Cynara cardunculus	cardoon	Moderate
Cynodon dactylon	Bermuda grass	Moderate
Cynosurus echinatus	dogtail grass	Moderate
Dipsacus sativus	Indian teasel	Moderate
Dittrichia graveolens	stinkwort	Moderate
Festuca perennis	Italian rye grass	Moderate
Foeniculum vulgare	fennel	Moderate
Genista monspessulana	French broom	High
Hedera helix	English ivy	High
Hirschfeldia incana	hoary mustard	Moderate
Holcus lanatus	common velvetgrass	Moderate
Hypochaeris radicata	hairy cats ear	Moderate
Lepidium latifolium	perennial pepperweed	High
Mentha pulegium	pennyroyal	Moderate
Phalaris aquatica	Harding grass	Moderate
Rubus armeniacus	Himalayan blackberry	High
Salsola soda	alkali Russian thistle	Moderate
Torilis arvensis	field hedge parsley	Moderate

Table 2-54 Cal-IPC High- and Moderate-Rated Species Observed in the BSA

Source: Cal-IPC California Invasive Plant Inventory

The most highly invaded areas in the BSA are the flats and sloped banks immediately adjacent to the paved roadway. This NNIP-dominated roadside buffer is at least 3 feet wide in most of the BSA but can extend up to about 10 feet wide in some areas. The roadside NNIPs are mostly restricted to uplands above the MHHW elevation. The most abundant NNIP in these areas is fennel, followed by poison hemlock, black mustard, and hoary mustard (all Cal-IPC Moderate species). Some NNIP grass species are also found in this area along the roadway, including Cal-IPC Moderates slender wild oat and Bermuda grass, as well as foxtail brome (Cal-IPC High). The highly invaded roadsides occur in all segments of the BSA.

Cal-IPC High-rated species are of concern because they can have severe ecological impacts as well as high rates of dispersal and establishment. Most of these species occur in the BSA in small, isolated patches, or as infrequent associates in communities dominated by other species. Three communities dominated by Cal-IPC High-rated species were identified in the BSA: fields dominated by yellow-star-thistle, perennial pepper weed patches, and Himalayan blackberry. Yellow star-thistle and Himalayan blackberry were observed in the areas where the corresponding upland semi-natural communities were mapped. Perennial pepper weed patches were only mapped in one segment of the BSA at the Sonoma Raceway, but perennial pepper weed occurs as a common associate in freshwater wetland areas throughout the BSA, especially at the Sears Point Restoration Project North and South, SR 121 Interchange, and other portions of the Sonoma Raceway.

## 2.4.7.3 Environmental Consequences

NNIPs occurring adjacent to the roadway have the greatest potential to spread in response to construction activities from the project, due to their proximity to work locations. Some NNIP species may be less dominant in the BSA currently but have potential to spread because of disturbance, such as the construction of the project. The project alternatives would expand the width of the highway, and the shoulders and cleared areas next to the roadway would likely transition to more ruderal vegetation similar to the habitat that is currently adjacent to SR 37. The contractor would be subject to construction contract requirements applied to all roadway projects, that minimize spread of invasive species through cleaning of equipment, use of clean fill, and revegetation of cleared areas with appropriate seed and vegetation species.

## 2.4.7.4 Avoidance, Minimization, and/or Mitigation Measures

Previously described measures for plant species and sensitive habitat would effectively avoid and minimize effects from NNIP. With implementation of already proposed measures, no substantial impacts from NNIP species is anticipated, and no new measures are proposed here. Measures that would address NNIP species include **BIO-01: Wetlands Protection – Invasive Plants**; **BIO-03: Tree Replacement**, **Landscaping, and Revegetation Plan**; and **BIO-07: Wetlands and Other Waters Compensation**. No compensation is anticipated to be required for invasive species.

## 2.5 Cumulative Impacts

## 2.5.1 Regulatory Setting

Cumulative impacts are those that result from past, present, and reasonably foreseeable future actions, combined with the potential impacts of the proposed project. A cumulative effect assessment looks at the collective impacts posed by individual land use plans and projects. Cumulative impacts can result from individually minor but collectively substantial impacts taking place over a period of time.

Cumulative impacts to resources in the project area may result from residential, commercial, industrial, and highway development, as well as from agricultural development and the conversion to more intensive agricultural cultivation. These land use activities can degrade habitat and species diversity through consequences such as displacement and fragmentation of habitats and populations, alteration of hydrology, contamination, erosion, sedimentation, disruption of migration corridors, changes in water quality, and introduction or promotion of predators. They can also contribute to potential community impacts identified for the project, such as changes in community character, traffic patterns, housing availability, and employment.

CEQA Guidelines Section 15130 describes when a cumulative impact analysis is necessary and what elements are necessary for an adequate discussion of cumulative impacts. The definition of cumulative impacts under CEQA can be found in Section 15355 of the CEQA Guidelines. A definition of cumulative impacts under NEPA can be found in 40 CFR Section 1508.7.

## 2.5.2 Cumulative Impact Analysis

This cumulative impact analysis determines whether the Build Alternatives, in combination with other past, present, or reasonably foreseeable projects, would result in a cumulative effect and, if so, whether the Build Alternatives' contribution to the cumulative impact would be considerable. Present and reasonably foreseeable future projects include land use developments, infrastructure, and other transportation improvements that are planned and funded and would be near the proposed Build Alternatives' improvements.

The No Build Alternative would not include improvements to SR 37. It would not require construction and would not contribute to cumulative environmental effects in combination with other projects.

Table 2-55 lists transportation corridor projects along or near the SR 37 corridor. These projects are in various stages of project development, from early conceptual planning and feasibility study to projects planned for approval.

Future planned land use developments within approximately 1 mile of the project area are described below in Table 2-56. The information in Table 2-55 was obtained from CEQAnet (2020); the planning departments for the counties of Solano, Napa, and Sonoma; and the City of Vallejo. The land uses in Solano, Napa, and Sonoma Counties adjacent to SR 37 are generally not designated for development; development proposals are therefore limited to the incorporated City of Vallejo and Mare Island at the eastern extent of the project corridor. For this reason, only current and planned development in City of Vallejo and Mare Island are presented in Table 2-56.

# Table 2-55Current and Proposed Planned Developments within One Mile of<br/>the Project Area

Ducie et Norme (Des orietien	Expenditure Authorization	Ocuratio		0	Otatus
Project Name/Description	Number	County	Post Miles	Sponsor	Status
SR 37 Pavement Rehabilitation PM R11.2/ 14.6 – Capital Preventive Maintenance. Pavement rehabilitation along highway mainline and ramps, replace guardrails, upgrade curb ramps.	2K740	Marin	R11.2/14.6	Caltrans	Delivery 2024
SR 37 (U.S. 101 to SR 121) SR 37 Flood Reduction Project (U.S. 101 to SR 121) – to address flooding and SLR between the present and 2050.	4Q320	Marin	R11.2/13.7	Caltrans	Delivery 2027
SR 37 Ultimate SLR Resilience Design Alternatives Assessment (U.S. 101 to SR 121)	Not Identified	Marin/ Sonoma	R11.2/ 14.6; 0.0/3.9	MTC	Feasibility Study
SR 37 Petaluma River Bridge Preservation. Resurface the bridge deck, replace bridge fender system, mitigate bridge scour, and upgrade bridge railings.	2Q500	Marin	14.5/15.0	Caltrans	Delivery 2024
Reconstruct Intersection of SR 37 and SR 121. Considering roundabout and "T" intersection design alternatives. This project has been combined into EA 2Q20U.	1Q480	Sonoma	3.8/4.3	Caltrans	Delivery 2024
SR 37 Lane Extension and Railroad Crossing at Tolay Creek. This project would extend the lane in the eastbound direction in the vicinity of SR 121 to the SMART railway crossing area. This project has been combined into EA 2Q20U.	2Q200	Sonoma	3.8/4.3	Caltrans	Delivery 2024
SR 37 Alternatives Assessment Report for the Ultimate Project (SR 121 to Mare Island Interchange) (completed)	Not Identified	Sonoma/ Solano	3.5/6.2; 0.0/R7.4	MTC	Feasibility Study
SR 37 Corridor SLR and Complete Streets (U.S. 101 to SR 29)	4Q960	Marin/ Sonoma/ Napa/ Solano	R11.2/14.6 ; 0.0/6.2; 0.0/R9.6	Caltrans	Feasibility Study
SR 37 Corridor PEL Study (U.S. 101 to I-80)	Not Identified	Marin/ Sonoma/ Napa/ Solano	R11.2/14.6 ; 0.0/6.2; 0.0/R11.4	Caltrans	Pre- Planning
SR 37 Pedestrian Enhancements at Wilson Avenue and Fairgrounds Drive, and other regional locations. Would include warning beacons, high visibility crosswalk markings, signs.	0P760	Solano	Various	Caltrans	Delivery 2024
Fairgrounds Drive Interchange Improvements	4A441	Solano	10.6/11.2	STA	Delivery 2021

# Table 2-56Current and Proposed Planned Developments within One Mile of<br/>the Project Area

Project Name	Project Description	Location
City of Vallejo/Waterfront Project	175 single-family detached residences, commercial areas, and two parks. EIR certified in 2005.	Between Mare Island Way and Mare Island Causeway (approximately 1 mile southeast)
City of Vallejo/North Mare Island	Film production, wine and beverage manufacturing, office, and retail use.	Adjacent to SR 37 to the south
City of Vallejo/Mare Island	Mixed-use development.	South of G Street (approximately 1 mile south)

## 2.5.3 Resource Areas with No Contribution to Cumulative Effects

The resources considered in the cumulative effects analysis follow Caltrans' Eight Step Guidance for identifying and assessing cumulative impacts (Caltrans 2016). No cumulative effects are anticipated for the following resource areas (there would be no adverse effects from each of these individual resource areas; therefore, no incremental effects would be cumulatively considerable for these topic areas):

- Existing and future land use
- Consistency with state, regional, and local plans and programs
- Parks and recreational facilities
- Growth
- Relocations and real property acquisition
- Environmental justice
- Utilities/emergency services
- Cultural resources
- Geology/soils/seismic/topography
- Paleontology
- Hazardous waste/materials
- Air quality
- Noise
- Energy
- Invasive species
- Hydrology/floodplain
- Water quality

## 2.5.4 Resources Considered for Contribution to Cumulative Effects

### 2.5.4.1 Biological Resources

### Interim

Caltrans is proposing projects that would elevate roadways, replace bridge rails, extend lanes or modify intersections, upgrade culverts, and replace bridges in different

parts of the corridor that could affect similar biological resources and habitat types impacted by the proposed Build Alternatives. These projects include the SR 37 Flood Reduction Project, and the Petaluma Bridge Preservation Project. These projects would involve temporary impacts for construction, and permanent impacts where permanent project features are proposed. The aforementioned projects are at varying stages of the environmental process; therefore, this discussion is not meant to preempt those public documents.

The SR 37/SR 121 intersection improvement project and SR 37 eastbound lane extension project are assumed to be incorporated into this project; they would extend the SR 37 eastbound lane drop from its current location west of the SMART railroad to a point east of the Tolay Creek Bridge. These improvements would be included in the Sears Point to Mare Island Improvement project's plans, specifications, and estimates package, which would reduce the temporary repetitive disturbance along the corridor associated with construction staging and activity. However, if the Sears Point to Mare Island Improvement project and/or the SR 37/121 intersection projects may be constructed as independent operational improvements. Both projects would reduce vehicle hours of delay, if built independently of the Sears Point to Mare Island project. Because these improvements fall within the footprint of the Sears Point to Mare Island project, the analysis of impacts contained in this Final EIR/EA addresses the potential impacts of both the eastbound lane extension and the SR 37/121 intersection, if these two projects are advanced independently.

In the shorter term, through this project and other interim projects listed in Table 1-1, infrastructure needs are being defined to limit environmental impacts. Caltrans is consulting with federal and state agencies on this and other projects in the corridor. It is expected that compensation for permanent biological impacts would be limited to each project's permanent biological impacts and requirements, including those of this project. Consequently, there would be no cumulative impacts to biological resources in conjunction with other future foreseeable projects, and no avoidance, minimization, or mitigation measures are required.

## Long-Term

The contiguous environmental setting of the corridor means that one cannot segment the environmental analysis based on the project limits of any one project, including the Highway 37 Sears Point to Mare Island Improvement Project, especially in light of trends toward SLR. Caltrans' longer-term studies, such as the PEL study and the Design Analysis Assessment, are being used to address SLR and other corridor needs prior to project-level planning and design. During preparation of the PEL study, Caltrans consulted with federal and state resource agencies to ensure that the scope of the environmental setting from U.S. 101 to I-80 is being considered. Although the Design Analysis Assessment studies a portion of the corridor from U.S. 101 to SR 121, Caltrans is likewise including federal and state agencies in its development.

The SR 37 Ultimate SLR project would require substantial investment to address future SLR up to 2130 and beyond. No alignment or alternatives have been identified

for the entire corridor, and it is currently not funded. The Highway 37 Sears Point to Mare Island project considered in this EIR/EA is intended to provide traffic congestion relief as described in Section 1.2. Biological impacts under a long-term, corridor-wide project would therefore not coincide in time or duration with the Highway 37 Sears Point to Mare Island Improvement Project. When the ultimate project is ready for the environmental phase, the impacts and mitigation would be identified in their own project-level environmental document.

## 2.5.4.2 Visual/Aesthetics

The proposed project alternatives would introduce a new movable median barrier (Alternative 1) or a new solid barrier (Alternatives 2, 3A, and 3B), that would be the same height for Alternative 1, and slightly higher (36 inches) than the existing median barrier (32 inches). Outside barriers would be constructed, and some new lighting installed. Tolling may be introduced to the corridor involving overhead signage and overhead toll reading equipment.

The project corridor is a relatively rural highway alongside the scenic areas of the north Bay wetlands and refuge lands. Between I-80 and the Napa River, SR 37 was reconstructed under several projects from the 1990s to early 2000s, from a conventional highway with local intersections and stop lights, to a divided freeway with controlled access at interchange ramps. A concrete median barrier was added in the late 1990s and early 2000s to the highway between Vallejo and SR 121 to reduce severe accidents occurring from traffic crossing over the then undivided highway. These improvements have addressed highway capacity in the Vallejo area and improved safety along project corridor, while adding visible elements such as barriers, overhead signs, lighting, and travel lanes and shoulders.

In addition to visual elements of the proposed project discussed in Section 2.2.12, such as toll gantries, bridge rails, and median barriers, other projects in the corridor would add similar visual elements to SR 37. Elements of other projects would/would not contribute to the visual/aesthetic impacts in the corridor.

Although these projects would add visible elements to the corridor, they would be consistent with the existing roadway features along SR 37.

- Bridge railings and outside safety barriers that would be installed by the proposed project and by the other corridor improvements would include open or see-through designs that would continue to allow partial views of the adjacent marsh lands and Bay waters, while improving motorist safety related to minimizing vehicles running off the road.
- The median barrier would be raised in height, by about 4 inches, between Mare Island and SR 121 to meet current design safety requirements. This would partially interfere with views from lower-profile vehicles across the median barrier. This is a cumulative change or impact over time with respect to the original highway, which had no divided barrier before the existing solid concrete median barrier was installed. This has been a necessary change over time to

protect motorists from more serious injuries, as highway design standards have evolved to address collisions and safety.

Overall, the project would not result in a cumulatively considerable adverse effect to visual resources because it would blend in with similar elements along SR 37 and be consistent with existing roadway features. Therefore, no avoidance, minimization, or mitigation measures are required.

## 2.5.4.3 Transportation

Each of the proposed alternatives would add one or two lanes to the existing highway to help alleviate existing congestion caused by the current lane merges. Other proposed SR 37 projects that are being considered that address traffic conditions are the lane extension and railroad crossing at Tolay Creek, and the potential reconstruction of the SR 37/SR 121 intersection.

The Lane Extension and Railroad Crossing Project (EA 2Q200) at Tolay Creek is intended to relieve congestion in the eastbound direction in the vicinity of the SR 121 intersection. This project would extend the SR 37 eastbound lane drop from its current location west of the SMART railroad to a point east of the Tolay Creek Bridge. The SR 37/SR 121 Intersection Improvement Project (EA 1Q480) is intended to address traffic at the turning movements between the two highways, which are at least partially improved with the proposed Sears Point to Mare Island project, especially in the eastbound direction. These improvements fall within the footprint of the Sears Point to Mare Island project, and they have been combined into a single project (EA 2Q20U). This environmental document addresses the potential environmental impacts of these projects, which may be constructed as independent operational improvements or incorporated into the Sears Point to Mare Island project. If the eastbound lane extension and/or the SR 37/SR 121 intersection projects referred to above proceed to construction independent of the Sears Point to Mare Island project, this environmental document addresses their potential environmental impacts. The lane extension and intersection projects would improve vehicle hours of delay at the SR 37/SR 121 interchange. These projects would not introduce cumulative traffic impacts, as each project is intended to improve traffic movement and reduce current backups that occur at constrained locations on SR 37 at and in the vicinity of the SR 121 intersection.

Other projects proposed within this corridor specifically address SLR/flooding, or pavement and bridge maintenance and restoration and would not change the capacity of the highway, and would not contribute to adverse cumulative effects; therefore, no avoidance, minimization, or mitigation measures are required. The long-term ultimate improvement projects that would address SLR would effectively replace the existing highway, if and when these projects are advanced, including operational improvements that have been approved under the Sears Point to Mare Island Improvement Project.

This page intentionally left blank

# Chapter 3 California Environmental Quality Act Evaluation

## 3.1 Determining Significance under CEQA

The proposed project is a joint project by Caltrans and FHWA and is subject to state and federal environmental review requirements. Project documentation, therefore, has been prepared in compliance with both the CEQA and NEPA. FHWA's responsibility for environmental review, consultation, and any other actions required by applicable federal environmental laws for this project are being, or have been, carried out by Caltrans pursuant to 23 USC 327 and the Memorandum of Understanding dated May 27, 2022, and executed by FHWA and Caltrans. Caltrans is the lead agency under CEQA and NEPA.

One of the primary differences between NEPA and CEQA is the way significance is determined. Under NEPA, significance is used to determine whether an Environmental Impact Statement (EIS), or a lower level of documentation, would be required. NEPA requires that an EIS be prepared when the proposed federal action (project) as a whole has the potential to "significantly affect the quality of the human environment." The determination of significance is based on context and intensity. Some impacts determined to be significant under CEQA may not be of sufficient magnitude to be determined significant under NEPA. Under NEPA, once a decision is made regarding the need for an EIS, it is the magnitude of the impact that is evaluated and no judgment of its individual significance is deemed important for the text. NEPA does not require that a determination of significant impacts be stated in the environmental documents.

CEQA, on the other hand, does require the Department to identify each "significant effect on the environment" resulting from the project and ways to mitigate each significant effect. If the project may have a significant effect on any environmental resource, then an EIR must be prepared. Each and every significant effect on the environment must be disclosed in the EIR and mitigated if feasible. In addition, the CEQA Guidelines list a number of "mandatory findings of significance," which also require the preparation of an EIR. There are no types of actions under NEPA that parallel the findings of mandatory significance of CEQA. This chapter discusses the effects of this project and CEQA significance.

## 3.2 Significant Irreversible Environmental Changes

The proposed alternatives do not represent significant irreversible changes. The alternatives consist of widening and improvements to SR 37 to resolve existing and recurring inefficient traffic congestion. No new lands or resources would be accessible (no change in route or secondary road access to new properties). Construction would use typical construction materials associated with a roadway infrastructure project, but this commitment of resources is not considered a significant change or obligation of limited resources.

The project does represent a commitment of resources to maintaining SR 37 at its present alignment and elevation or profile to alleviate current and future traffic congestion. A future project that addresses SLR would likely have to be on new alignment depending on how construction staging could be achieved, with possible abandonment of the existing alignment. The proposed improvements to the current facility are therefore considered an unavoidable and cumulative investment in the highway to meet the project's purpose and need to alleviate traffic congestion until an ultimate improvement can be funded that addresses both traffic and future SLR.

## 3.3 CEQA Environmental Checklist

This checklist identifies physical, biological, social, and economic factors that might be affected by the proposed project. The first column lists pertinent questions applicable to the resource, and the other four columns include the degree of impact for each of those questions. In many cases, technical studies performed in connection with the project indicate that there are no impacts to a particular resource. A "no impact" answer in the last column reflects this determination. The words "significant" and "significance" used throughout the checklist are related to CEQA impacts. The questions in this form are intended to encourage the thoughtful assessment of impacts and do not represent thresholds of significance. Significance determinations (e.g., no impact, less than significant, potentially significant impact) are responded to for each of the CEQA checklist questions; a "yes" or "no" response is given for each significance determination column in each question row. A "yes" response indicates that this is the significance determination that applies for that question. A "no" response indicates that the significance determination in that column does not apply to that question.

Project features, which can include both design elements of the project, and standardized measures that are applied to all or most Caltrans projects such as BMPs and measures included in the Standard Plans and Specifications (Caltrans 2020a) or as Standard Special Provisions, are considered to be an integral part of the project and have been considered prior to any significance determinations documented below; see Chapter 1 and Chapter 2 for a detailed discussion of these features. The annotations to this checklist are summaries of information contained in Chapter 2 to provide the reader with the rationale for significance determinations; for a more detailed discussion of the nature and extent of impacts, please see Chapter 2. This checklist incorporates by reference the information contained in Chapter 1 and Chapter 2.

## 3.3.1 Aesthetics

Except as provided in Public Resources Code Section 21099, would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Have a substantial adverse effect on a scenic vista?	No	No	Yes	No
b) Substantially damage scenic resources, including, but not limited to, trees, rock outcroppings, and historic buildings within a state scenic highway?	No	No	Yes	No
c) In non-urbanized areas, substantially degrade the existing visual character or quality of public views of the site and its surroundings? (Public views are those that are experienced from a publicly accessible vantage point). If the project is in an urbanized area, would the project conflict with applicable zoning and other regulations governing scenic quality?	No	No	Yes	No
d) Create a new source of substantial light or glare which would adversely affect day or nighttime views in the area?	No	No	No	Yes

## **3.3.1.1 CEQA Significance Determinations for Aesthetics**

## a) Have a substantial adverse effect on a scenic vista?

**Less than Significant Impact**. The project corridor is a designated scenic corridor in both Solano and Sonoma Counties. BCDC considers Bay shoreline views such as those available along much of the project corridor to be scenic views. Trail and recreational areas such as Cullinan Ranch in the Refuge and Tolay Creek/Tubbs Island Trail near the project corridor provide scenic vistas.

As described in Section 2.2.12, the combination of a higher median barrier and outside barrier would have an impact on the scenic views along SR 37. Views from scenic areas toward the project corridor would not be substantially affected. However, depending on the Build Alternative and outside barrier option selected, there would be visual impacts to motorists on the project corridor because views of low-lying adjacent marshlands and waterways would be diminished or blocked. Longer distance views toward adjacent hills would not be affected.

Impacts of project improvements are depicted in the visual simulations for KV-1 through KV-5 in Section 2.2.12.3. At KV-3, the project would have the greatest impacts on scenic views for Alternatives 2, 3A, and 3B because the combination of the center median barrier and outside barrier would impair views of the Bay on both sides of SR 37. However, views of distant mountains and hills would still be seen. For all Build Alternatives, the implementation of a new toll gantry would be the tallest built element in view. However, the gantry would not block water views on either side of the

highway. Project improvements that would have the greatest impacts on scenic views would be the higher median barrier and outside barrier. Build Alternatives could impact the view of a scenic vista for motorists traveling along SR 37 because the 36-inch or 42-inch median barrier would block views of low-lying scenic landscapes on the other side of the highway for many drivers. However, SR 37 has scenic views on both sides of the highway, so views would still be visible on whichever side of the highway motorists are traveling.

For the outside barrier, both MGS and Type 85B designs are being considered, depending on the location. Type 85B barriers are designed to be partially transparent and would allow partial views of the Bay. Visual renderings at key viewpoints with Type 85B barriers in place are presented in Section 2.2.12.3. The MGS would have a metal top rail that would partially impair views of the Bay. However, because the MGS is 5 inches shorter than the Type 85B barrier, people would still have views of the Bay above the barrier. Therefore, MGS (which is shorter) and Type 85B (which has a semi-transparent design) would both allow for partial views of the Bay, and would have similar visual impacts. The installation of a Type 85B barrier on the outside shoulder as part of the project avoids or reduces visual impacts because it provides some maintenance of the existing views through the barrier. Furthermore, outside barriers are only proposed at key locations, so the whole length of the project area would not be affected. Table 2-15 summarizes the differences between alternatives with respect to the visual impact findings. There would be less than a significant impact to scenic views.

# b) Substantially damage scenic resources, including, but not limited to, trees, rock outcroppings, and historic buildings within a state scenic highway?

**Less than Significant Impact**. The project would contribute new built elements to a highway that is potentially eligible for scenic highway status. These include overhead gantries, lighting, a new median barrier, and outside safety barriers. These changes would not substantially damage or alter the highway, and scenic views from the highway would still be available.

#### c) In non-urbanized areas, substantially degrade the existing visual character or quality of public views of the site and its surroundings? (Public views are those that are experienced from a publicly accessible vantage point). If the project is in an urbanized area, would the project conflict with applicable zoning and other regulations governing scenic quality?

**Less than Significant Impact**. The project is in a nonurbanized area. Views from publicly accessible recreational areas were evaluated, and these views would not be substantially affected. The primary visible changes would be from the motorist's viewpoint, from the higher barriers. Views would be partially impaired compared to the existing setting, but would remain available to motorists, and the scenic quality of the corridor would largely remain intact.

# d) Create a new source of substantial light or glare which would adversely affect day or nighttime views in the area?

**No Impact.** The project would not create a new source of substantial light or glare. All permanent lighting installed would be consistent with applicable regulations and with street lighting existing in the project vicinity. During construction, some work would occur at night. Construction lighting would be shielded and directed toward the area of work and would not constitute a substantial new source of light outside the work area. VIS-01. Limit Light Pollution would be implemented to limit light pollution and have minimum impact on the surrounding environment. All light fixtures would have light-emitting diodes configured at the minimum necessary number of bulbs, optimal mounting height, mast-arm length, and angle to restrict light to the roadways. Where applicable, shields on the fixtures to prevent light trespass to adjacent properties would be considered during the detailed design phase. Therefore, the project would not create a substantial amount of light or glare, which could adversely affect day or nighttime views. There would be no impact.

## 3.3.2 Agriculture and Forest Resources

In determining whether impacts to agricultural resources are significant environmental effects, lead agencies may refer to the California Agricultural Land Evaluation and Site Assessment Model (1997) prepared by the California Department of Conservation as an optional model to use in assessing impacts on agriculture and farmland. In determining whether impacts to forest resources, including timberland, are significant environmental effects, lead agencies may refer to information compiled by the California Department of Forestry and Fire Protection regarding the state's inventory of forest land, including the Forest and Range Assessment Project and the Forest Legacy Assessment Project; and the forest carbon measurement methodology provided in Forest Protocols adopted by CARB.

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Convert Prime Farmland, Unique Farmland, or Farmland of Statewide Importance (Farmland), as shown on the maps prepared pursuant to the Farmland Mapping and Monitoring Program of the California Resources Agency, to non- agricultural use?	No	No	No	Yes
b) Conflict with existing zoning for agricultural use, or a Williamson Act contract?	No	No	No	Yes

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
c) Conflict with existing zoning for, or cause rezoning of, forest land (as defined in Public Resources Code section 12220(g)), timberland (as defined by Public Resources Code section 4526), or timberland zoned Timberland Production (as defined by Government Code section 51104(g))?	No	No	No	Yes
d) Result in the loss of forest land or conversion of forest land to non-forest use?	No	No	No	Yes
e) Involve other changes in the existing environment which, due to their location or nature, could result in conversion of Farmland, to non-agricultural use or conversion of forest land to non-forest use?	No	No	No	Yes

# 3.3.2.1 CEQA Significance Determinations for Agriculture and Forest Resources

#### a) and b). Convert Prime Farmland, Unique Farmland, or Farmland of Statewide Importance or conflict with existing zoning for agricultural use, or a Williamson Act contract?

**No Impact.** As described in Section 2.2.5, the project area is adjacent to Farmland of Local Importance and Grazing Land as well as Non-Prime Agricultural parcels with Williamson Act contracts, between the SR 37/SR 121 interchange and the Sonoma Bridge in Sonoma County. The project would not convert Prime Farmland, Farmland of Statewide Importance, Unique Farmland, or Farmland of Local Importance. Although the project would require TCEs; the easements would not affect the continued use of the properties for agricultural use. None of the Build Alternatives are anticipated to require permanent property acquisitions. Additionally, there would be no permanent acquisition of Williamson Act properties. The project would not modify, nullify, or require changes to the Williamson Act contracts on the properties. Therefore, there would be no impact.

#### c) and d) Result in the loss of forest land or conversion of forest land to nonforest use?

**No Impact.** There are no forest lands in or adjacent to the project area. Therefore, no changes are anticipated to forest land, and there would be no impact.
#### e) Involve other changes in the existing environment which, due to their location or nature, could result in conversion of Farmland, to non-agricultural use or conversion of forest land to non-forest use?

**No Impact.** The project would not involve changes that could result in the conversion of farmland and forest land. Therefore, there would be no impact.

#### 3.3.3 Air Quality

Where available, the significance criteria established by the applicable air quality management district or air pollution control district may be relied upon to make the following determinations.

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Conflict with or obstruct implementation of the applicable air quality plan?	No	No	No	Yes
b) Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or state ambient air quality standard?	No	No	Yes	No
c) Expose sensitive receptors to substantial pollutant concentrations?	No	No	No	Yes
d) Result in other emissions (such as those leading to odors) adversely affecting a substantial number of people?	No	No	No	Yes

#### 3.3.3.1 CEQA Significance Determinations for Air Quality

#### a) Conflict with or obstruct implementation of the applicable air quality plan?

**No Impact.** The project site is in the SFBAAB and within the jurisdiction of BAAQMD and CARB. The proposed project would not interfere with any of the control measures set forth in BAAQMD's 2017 Clean Air Plan. The project is a capacity-increasing project, and would be included in the MTC RTP, Plan Bay Area 2050. The project would be listed in the conforming 2021 TIP and the MTC RTP. Therefore, the project would also comply with the SIP. The project would not conflict with or obstruct implementation of applicable air quality plans and there would be no impact.

#### b) Result in a cumulatively considerable net increase of any criteria pollutant for which the project region is non-attainment under an applicable federal or state ambient air quality standard?

**Less than Significant Impact**. As shown in Table 2-21 in Section 2.3.6.2, the area is in nonattainment for CAAQS for  $O_3$ ,  $PM_{10}$  micrometers or smaller, and  $PM_{2.5}$ ; and is in

nonattainment for NAAQS for O<sub>3</sub> and PM<sub>2.5</sub>. During construction, there would be air emissions from the use of construction equipment and vehicles powered by gas and diesel, and dust from earthmoving activities such as trenching and grading. Table 2-22 in Section 2.3.6.3 of the EA shows the total estimated construction-related criteria pollutants for the Build Alternatives. The proposed project would also generate pollutants during construction, even with BMPs implemented. There would be temporary increases in criteria pollutants during construction, but they would be less than significant because of the limited duration of construction and with the implementation of BMPs.

Overall, each of the Build Alternatives would have substantially lower operational emissions when compared to existing/baseline conditions due to reduced congestion and improvements in vehicle fleet emissions over time. All of the Build Alternatives would have similar emissions compared to No-Build conditions for each study year (see Table 2-24 in Section 2.3.6.3). This is due to a greater reduction in regional travel time and vehicle idling despite an increase in regional VMT. Increased regional VMT over time is due to projected regional population growth, described in Section 2.2.6.

#### c) Expose sensitive receptors to substantial pollutant concentrations?

**No Impact**. Sensitive receptors are children, elderly, people with asthma, and others who are at a heightened risk of negative health outcomes due to exposure to air pollution. Sensitive receptors are typically associated with schools, residential dwellings, daycare centers, hospitals, and senior-care facilities. The project is in the San Pablo Bay lands. There is little to no development adjacent to SR 37 between Mare Island and SR 121. Most of the land adjacent to the highway is preserved open space or being used for agricultural purposes. The only sensitive receptors in the project area are two single-family homes on the southern side of SR 37, near the western terminus of the project. Sensitive receptors, including two residences, are within 500 feet of the project.

The Build Alternatives would not exceed existing conditions for criteria pollutants or MSATs, or exceed the BAAQMD's recommended thresholds for construction emissions. Air quality conditions for sensitive receptors is not expected to worsen. The Build Alternatives would not expose sensitive receptors that could occur near the project area to substantial pollutant concentrations. Therefore, there would be no impact.

### d) Result in other emissions (such as those leading to odors) adversely affecting a substantial number of people?

**No Impact.** The project would not introduce odors that are not already associated with existing traffic. Therefore, there would be no impact.

#### 3.3.4 Biological Resources

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations, or by the California Department of Fish and Wildlife, U.S. Fish and Wildlife Service, or NOAA Fisheries?	No	Yes	No	No
b) Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the California Department of Fish and Wildlife or U.S. Fish and Wildlife Service?	No	Yes	No	No
c) Have a substantial adverse effect on state or federally protected wetlands (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means?	No	Yes	No	No
d) Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites?	No	No	Yes	No
e) Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance?	No	No	No	Yes
f) Conflict with the provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan?	No	No	No	Yes

#### 3.3.4.1 CEQA Significance Determinations for Biological Resources

a) Have a substantial adverse effect, either directly or through habitat modifications, on any species identified as a candidate, sensitive, or special status species in local or regional plans, policies, or regulations, or by the California Department of Fish and Wildlife, U.S. Fish and Wildlife Service, or NOAA Fisheries?

**Less than Significant with Mitigation Incorporated.** Without mitigation, the project would have potentially significant direct and indirect impacts to special-status species habitat under all Build Alternatives, as described in Section 2.4.5. Build Alternatives 1, 2, and 3A would have relatively similar levels of permanent impacts to species habitat. Alternative 3B would have substantially greater impacts to special status species and

habitat due to roadway and bridge widening at Sonoma Creek that would require additional fill into species habitat. Table 3-1 summarizes anticipated permanent impacts to state and federally listed species with potential habitat in the BSA. Table 3-2 summarizes areas of potential temporary impacts during construction to listed species habit in the BSA.

Species Habitat	Alternative 1 (Acres)	Alternative 2 (Acres)	Alternative 3A (Acres)	Alternative 3B (Acres)	Alternative 3B Shading (acres)
Chinook salmon, steelhead, green sturgeon, longfin smelt, and Delta smelt foraging habitat	0.02	0.05	0.06	1.7	0.17
Chinook salmon, steelhead, green sturgeon, longfin smelt, and Delta smelt prey production habitat	1.03	1.43	1.49	2.08	0.10
Ridgeway's Rail and California black rail	0.34	0.79	1.02	1.65	0.52
salt marsh harvest mouse	0.37	0.69	0.92	1.67	0
California red-legged frog	0.01	0.01	1.51	1.51	0

Table 3-1 Estimated Permanent Habitat Impacts to Listed Species Habitats

#### Table 3-2 Estimated Temporary Impacts Areas in Listed Species Habitats

Habitat	Alternative 1 (Acres)	Alternative 2 (Acres)	Alternative 3A (Acres)	Alternative 3B (Acres)	Alternative 3B Shading
Chinook salmon, steelhead, green sturgeon, longfin smelt, and Delta smelt foraging habitat	1.37	1.93	1.71	3.00	0.29
Chinook salmon, steelhead, green sturgeon, longfin smelt, and Delta smelt prey production habitat	2.44	2.79	2.17	1.98	0.09
Ridgway's rail and California black rail	0.42	1.90	2.02	3.11	0.86
salt marsh harvest mouse	0.70	2.01	2.14	3.24	0.86
California red-legged frog	0.11	0.11	0.75	0.78	0.00

#### Federally Listed Species and Designated Critical Habitat

Under all considered Build Alternatives, the project may affect the following species: soft bird's-beak, California red-legged frog, Ridgway's rail, Chinook Salmon Sacramento River Winter Run ESU, Chinook Salmon Central Valley Spring Run ESU, Delta smelt, steelhead Central Valley DPS, steelhead Central California Coast DPS, North American Green Sturgeon southern DPS, and salt marsh harvest mouse. Construction-related in-water disturbance and addition of fill materials in the project footprint would directly affect federally listed species. Implementation of AMMs would serve to reduce the likelihood of any indirect impacts (in the form of dust or sedimentation) to suitable habitat outside of the project footprint.

Under all Build Alternatives, direct impacts to individual salt marsh harvest mouse and Ridgway's rail would be limited because these species are also CFGC fully protected species. This means that "take," as defined by CFGC Section 86 as "Hunt, pursue, catch, capture, or kill, or attempt to hunt, pursue, catch, capture, or kill," is not allowed for these species. FESA defines "take" as "Harass, harm, pursue, hunt, shoot, wound, kill, trap, capture, or collect, or to attempt to engage in any such conduct" (16 USC 1532[19]). Impacts beyond harassment are not anticipated for either of these species, and specific measures for these species are proposed in Section 2.4.5 to avoid and minimize impacts to these species.

Under Alternative 3B only, the North American green sturgeon southern DPS has potential for direct take from underwater sound pressure during pile-driving activities.

Caltrans has prepared a draft Biological Assessment for consultation with USFWS and NMFS and has made preliminary effects determinations pursuant to section 7 of FESA. Caltrans would request formal consultation with USFWS and NMFS after selection of a preferred Build Alternative. Caltrans has made the following effects determinations:

- *may affect, and is likely to adversely affect*, California red-legged frog, Ridgway's rail, salt marsh harvest mouse, steelhead Central California Coast DPS, and North American green sturgeon southern DPS
- *may affect, but is not likely to adversely affect* soft bird's-beak, Chinook Salmon Sacramento River Winter-Run ESU, Chinook Salmon Central Valley Spring-Run ESU, Delta smelt, and steelhead Central Valley DPS
- *no effect* on federally designated critical habitat for California red-legged frog, Ridgway's rail, salt marsh harvest mouse, and Delta smelt
- may affect, and is likely to adversely affect, designated critical habitat for steelhead Central California Coast DPS, steelhead Central Valley DPS, Chinook Salmon Central Valley Spring-Run ESU, Chinook Salmon Sacramento River Winter-Run ESU, and green sturgeon Southern DPS

Although the project would have some permanent and temporary impacts, with implementation of the measures proposed for federally listed species in Section 2.4.5, the project would have less than significant impacts on federally listed species under all Build Alternatives.

#### State-Listed Species

Special-status animals protected under the CESA, including California black rail, Swainson's hawk, and longfin smelt, have potential to occur in the BSA. Ridgway's rail, Chinook Salmon Central Valley Spring-Run and Sacramento River Winter-Run ESUs, Delta smelt, and salt marsh harvest mouse are listed under both FESA and CESA and are discussed above.

CESA protects species listed as threatened or endangered from take unless authorized through an incidental take permit. Through implementation of the measures described below, direct impacts that may cause take (as defined by CFGC) of statelisted species would be avoided. Impacts to state-listed species would be limited to habitat impacts. Most habitat that would be impacted occurs immediately adjacent to SR 37, where habitat conditions are generally poor. Due to the lack of key habitat features (such as suitable roosting trees), no impacts to Swainson's hawk habitat are anticipated under the Build Alternatives.

Therefore, the project would have a less than significant impact on state-listed species with mitigation incorporated under all Build Alternatives.

### State Species of Special Concern, Migratory Birds, Essential Fish Habitat, and Marine Mammals

Potential impacts to migratory birds, marine mammals, and CDFW State SSCs are discussed in Section 2.4.4. SSC including pallid bat, Saltmarsh Common Yellowthroat, San Pablo song sparrow, Sacramento Splittail, Suisun shrew, and Western Burrowing Owl have potential to occur in the BSA. California red-legged frog and North American green sturgeon are also SCC that are listed under FESA and are discussed under Section 2.4.5. Marine mammals with potential to occur in the BSA include California sea lion, northern elephant seal, and Pacific harbor seal. None of these species are federally or state-listed as threatened or endangered; however, all marine mammals are protected under the Marine Mammal Protection Act of 1972. These species may infrequently occur in or immediately adjacent to the BSA near Sonoma Creek and Tolay Creek. NMFS-regulated EFH is discussed under Section 2.4.6.

BIO-10 through BIO-40 would be implemented to avoid or minimize potential impacts to migratory birds, marine mammals, and state SSCs under all Build Alternatives.

#### **Proposed Avoidance Minimization and Mitigation**

AMMs proposed for all special-status species are detailed in Chapter 2. Proposed measures are listed here for ease of reference.

### General Avoidance and Minimization Measures Applicable to Special-Status Species

The following general measures would serve to avoid and minimize potential impacts to special-status species with potential to occur in the project area under all Build Alternatives.

- BIO-15: Stop Work Authority
- BIO-16: Worker Environmental Awareness Training
- BIO-17: Discovery of Injured or Dead Special-Status Species

### General Avoidance and Minimization Measures for Special-Status Wildlife Species

The following general measures would serve to avoid and minimize potential impacts to special-status wildlife species with potential to occur in the project area under all Build Alternatives.

- BIO-18: Wildlife Species Relocation
- BIO-19: Construction Noise

#### Special-Status Plant-Specific Measures

The following specific measures would serve to avoid and minimize potential impacts to special-status plant species under all Build Alternatives.

- BIO-03: Tree Replacement, Landscaping, and Revegetation Plan
- BIO-08: Targeted Pre-Construction Plant Survey
- BIO-09: Special-Status Plant Monitoring

#### Special-Status Bird-Specific Measures

In addition to complying with the MBTA and CESA, the following specific measures would be implemented to avoid and minimize potential impacts to special-status bird species to a less than significant level under all Build Alternatives.

- BIO-10: Nesting Bird Protection
- BIO-13: Western Burrowing Owl Pre-Construction Surveys
- BIO-14: Western Burrowing Owl Nest Avoidance
- BIO-24: Ridgway's Rail and California Black Rail Pre-Construction Survey
- BIO-25: Ridgway's Rail and California Black Rail Protocol-Level Surveys and Avoidance and Minimization Measures
- BIO-40: Swainson's Hawk Pre-Construction Surveys

#### Special-Status Mammal-Specific Measures

With the following specific measures, the project would avoid, minimize, and mitigate potential impacts to special-status mammal species to a minimal level under all Build Alternatives.

- BIO-11: Pre-Construction Bat Surveys and Avoidance Measures
- BIO-12: Bat Monitoring Protocols
- BIO-27: Salt Marsh Harvest Mouse Pre-Construction Surveys
- BIO-28: Salt Marsh Harvest Mouse Impact Avoidance and Minimization
- BIO-29: Salt Marsh Harvest Mouse Monitoring Protocols

#### California Red-Legged Frog-Specific Measures

With the following specific measures, the project would avoid, minimize, and mitigate potential impacts to California red-legged frog to a level of less than significant under all Build Alternatives.

- BIO-20: California Red-Legged Frog Habitat Work Window
- BIO-21: California Red-Legged Frog Pre-Construction Surveys
- BIO-22: California Red-Legged Frog Monitoring Protocols
- BIO-23: Compensation for California Red-Legged Frog Habitat Effects

#### Special-Status Fish-Specific Measures

The following specific measures would serve to avoid and minimize potential impacts to special-status fish to less than significant under all Build Alternatives.

- BIO-04: Estuarine Dewatering Work Window
- BIO-05: Turbidity Control
- BIO-31: Vibratory Pile Driving
- BIO-32: In-Water Sheet Pile Fish Entrapment Avoidance
- BIO-33: Fish Monitoring
- BIO-34: Fish Relocation

The following measures would be implemented under Alternative 3B only.

- BIO-35: Compensation for Chinook Salmon, Steelhead, Green Sturgeon, Longfin Smelt and Delta Smelt Habitat
- BIO-36: In-Water Impact Pile Driving Work Window
- BIO-37: In-Water Impact Pile Driving Attenuation
- BIO-38: Hydroacoustic Monitoring
- BIO-39: Pile Proofing

#### Compensation to Offset Impacts to Listed Species Habitat

The following measures are proposed to offset any potential direct and/or indirect impacts to listed species habitat under the Build Alternatives. Alternative 3B is anticipated to have the greatest amount of habitat impacts compared to the other Build Alternatives. The specific habitat and mitigation for the preferred alternative, once selected, would be determined during the project's final design. Caltrans' mitigation proposal may include any one or a combination of the following approaches: offsite mitigation through purchase of credits at an approved conservation bank(s); development of a compensation plan that would provide in-lieu funding to a nearby restoration program or restoration project that would create, restore and/or enhance resources adversely affected by the project. Compensation for temporary impacts to special status species habitat would be achieved through onsite in-kind habitat restoration to pre-disturbance conditions.

- BIO-23: Compensation for California Red-Legged Frog Habitat Effects
- BIO-26: Compensation for Ridgway's Rail Habitat Effects

- BIO-30: Compensation for Salt Marsh Harvest Mouse and California Black Rail Habitat Effects
- BIO-35: Compensation for Chinook Salmon, Steelhead, Green Sturgeon, Longfin Smelt and Delta Smelt Habitat

#### Significance Determination

The project is anticipated to have less than significant impacts with the abovediscussed biological mitigation incorporated under the proposed Build Alternatives.

#### b) Have a substantial adverse effect on any riparian habitat or other sensitive natural community identified in local or regional plans, policies, regulations or by the California Department of Fish and Wildlife or U.S. Fish and Wildlife Service?

Less than Significant with Mitigation Incorporated. Sensitive natural communities that occur in the BSA and have potential to be impacted by the project are discussed in Section 2.4.1. These communities include sensitive marsh and wetland communities, and valley oak. EFH and designated critical habitat for USFWS- and NMFS-listed species is discussed in response to the question above. Impacts to wetlands, including riparian habitat, are included in the response to item c, which follows this response.

All Build Alternatives would have impacts on sensitive marsh species vegetation communities, as summarized in Section 2.4.1; all these communities are also in federally protected wetlands. Permanent impacts from fill in sensitive wetland communities would be the least under Alternative 1 (1.91 acres); moderate under Alternatives 2 (3.29 acres) and 3A (3.65 acres); and the largest under Alternative 3B (7.55 acres).

Four valley oak trees occur in the BSA. Impacts to valley oak are anticipated to be similar under all Build Alternatives and are assumed to be minimal. General avoidance and minimization measures to avoid potential impacts to valley oaks are proposed in Section 2.4.

#### Proposed Avoidance, Minimization, and Mitigation

#### Wetland Marsh Communities

The following specific measures proposed in Chapter 2 would serve to avoid and minimize potential impacts to sensitive marsh vegetation communities under all Build Alternatives.

- BIO-01: Wetlands Protection Invasive Plants
- BIO-02: Wetland Protection
- BIO-07: Wetlands and Other Waters Compensation
- BIO-23: Compensation for California Red-Legged Frog Habitat Effects
- BIO-26: Compensation for Ridgway's Rail Habitat Effects

- BIO-30: Compensation for Salt Marsh Harvest Mouse and California Black Rail Habitat Effects
- BIO-35: Compensation for Chinook Salmon, Steelhead, Green Sturgeon, Longfin Smelt and Delta Smelt Habitat

Impacts to wetland communities would be incurred under all Build Alternatives, and compensation would be realized through implementation of the measures listed above, resulting in minimal impacts with mitigation.

#### Valley Oaks

The following general avoidance and minimization measures proposed in Chapter 2 would serve to protect sensitive valley oaks under all Build Alternatives.

• BIO-03: Tree Replacement, Landscaping, and Revegetation Plan

#### Significance Determination

Under Build Alternatives 1, 2, and 3A, the project would have less than significant impacts with mitigation incorporated for impacts to wetland communities.

Under Build Alternative 3B, the project would have less than significant with mitigation incorporated for impacts to wetland communities. Alternative 3B would likely require substantially greater mitigation to offset for permanent and temporary impacts to wetland marsh communities than all other alternatives.

Under all Build Alternatives, the project would have less than significant impacts to valley oak.

#### c) Have a substantial adverse effect on state or federally protected wetlands (including, but not limited to, marsh, vernal pool, coastal, etc.) through direct removal, filling, hydrological interruption, or other means?

Less than Significant with Mitigation Incorporated. The project is anticipated to have permanent impacts from placement of permanent fill for road widening, retaining walls, sheet piles, rock slope protection, and placement of guard rails. Under Alternative 3B, additional permanent impacts to wetlands and waters would be realized through bridge widening work at Sonoma Creek and would permanently shade additional areas below the widened bridge. The primary permanent impact under all scenarios is associated with road widening, with Alternative 3B having the greatest permanent impact on wetlands and waters. Temporary impacts are associated with construction access, staging areas, and temporary dewatering activities.

Although the project would cause temporary and permanent impacts to the intertidal areas, it has been designed to minimize fill and turbidity. The impact areas would be spread out along the alignment of SR 37, which borders thousands of acres of tidal waters and wetlands. Preliminary estimates of permanent and temporary fill impacts

for all alternatives, as well as shading impacts from Alternative 3B, are shown in Table 2-39 and Table 2-40, respectively.

#### Proposed Avoidance, Minimization, and Mitigation

The following specific measures described in Chapter 2 and proposed for state and federally protected wetlands are summarized here.

- BIO-01: Wetlands Protection Invasive Plants
- BIO-02: Wetland Protection
- BIO-07: Wetlands and Other Waters Compensation

#### Significance Determination

Under Alternatives 1, 2, and 3A, impacts to federally protected wetlands would be **less than significant with mitigation**.

Under Alternative 3B, impacts to federally protected wetlands would be **less than significant with mitigation**. Alternative 3B would have substantially greater impacts to wetlands and waters than any other alternatives considered and is anticipated to require a substantially greater amount of compensation to offset impacts accordingly.

## d) Interfere substantially with the movement of any native resident or migratory fish or wildlife species or with established native resident or migratory wildlife corridors, or impede the use of native wildlife nursery sites?

**Less than Significant.** Two crossings to the east of SR 121 were identified by CDFW in a letter to Caltrans on August 20, 2020, regarding the project. These included an unnamed tributary (Fish Passage Assessment Database ID# 732818) and water tank cattle pass (Fish Passage Assessment Database ID# 761446). Results of the preliminary fish passage assessments for these crossings are discussed in Section 2.4.4.1. None of the structures provided a barrier to anadromous fish, since they are either absent or can pass freely through the structure.

Although there is consistently a high level of traffic on SR 37, SR 37 is not considered a total barrier to above-ground terrestrial wildlife movement. However, due to the existing median barrier, small- to medium-sized wildlife are not expected to regularly cross the road over SR 37. Road widening and median barrier height under all build alternatives would further inhibit crossing by terrestrial wildlife within the project limits. The portion of SR 37 where the project would occur does not cross any likely movement corridors for larger terrestrial animals since there is very little land south of the roadway. There are several waterway crossings that could be used by dispersing wildlife to cross under SR 37, including the Tolay Creek and Sonoma Creek bridges, and a culvert undercrossing west of Sears Point that may provide safe passage for both cattle and terrestrial wildlife to cross under SR 37. There are also several smaller culverts associated with tidal water features that may be used by smaller wildlife.

#### Significance Determination

Under all build alternatives, the project would not substantially interfere with the movement of any native or resident migratory fish, would not impede the use of native wildlife nursery sites, and would have no impact. The project would further inhibit road crossings because of road widening and increased median barrier height; although use is likely greater among birds and marine or aquatic species in the surrounding habitat, there is some potential for terrestrial wildlife to pass through the project limits. Therefore, the project conservatively assumes that it would have less than significant impacts on wildlife corridors.

### e) Conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance?

**No Impact.** The project would not conflict with any local policies or ordinances protecting biological resources, such as a tree preservation policy or ordinance, and would have **no impact**.

#### f) Conflict with the provisions of an adopted Habitat Conservation Plan, Natural Community Conservation Plan, or other approved local, regional, or state habitat conservation plan?

**No Impact.** The project runs through a portion of the proposed draft Solano County Multispecies HCP. The draft Solano County HCP states, "The purpose of the HCP is to promote the conservation of biological diversity and the preservation of endangered species and their habitats consistent with the recognition of private property rights; provide for a healthy economic environment for citizens, agriculture, and industries; and allow for ongoing maintenance and operation of public and private facilities in Solano County." The project would be consistent with the purpose of the Solano County HCP because it is a public facility. The proposed project would impact state and federally protected wetlands and waters, and special-status species habitats under all Build Alternatives. However, the project would not inhibit the continued marshland restoration work that is prioritized by the Solano County HCP in the areas surrounding the project. The project as proposed has been designed to reduce environmental impacts, would include measures to offset impacts to wetlands and federally protected species habitat; therefore, the project would be consistent with the Solano County HCP.

#### Significance Determination

The project under all Build Alternatives would not conflict with an adopted HCP, Natural Community Conservation Plan, or other approved local, regional, or state HCP and would have **no impact**.

#### 3.3.5 Cultural Resources

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Cause a substantial adverse change in the significance of a historical resource pursuant to §15064.5?	No	No	No	Yes
b) Cause a substantial adverse change in the significance of an archaeological resource pursuant to §15064.5?	No	No	No	Yes
c) Disturb any human remains, including those interred outside of dedicated cemeteries?	No	No	No	Yes

#### 3.3.5.1 CEQA Significance Determinations for Cultural Resources

### a) Cause a substantial adverse change in the significance of a historical resource pursuant to §15064.5?

**No Impact.** No historical resources as defined in Section 15064.5 would be impacted by the construction and operation of the project. As described in Section 2.2.13, the Mare Island Naval Shipyard is the only NRHP-listed property adjacent to the APE. It would not be affected by project construction or operation, because there are no contributing or noncontributing resources, such as landscape features, archaeological features, buildings, or structures, associated with the Mare Island Naval Shipyard adjacent to or in the APE. Therefore, there would be no impact.

### b) Cause a substantial adverse change in the significance of an archaeological resource pursuant to §15064.5?

**No Impact.** The project has little or no potential to impact archaeological resources. The background research and literature review conducted for this project identified one previously recorded prehistoric archaeological site, designated as Nelson Mound. During the field survey and subsurface testing, no artifacts, features, or culturally sensitive soils were identified in any of the bores. The project includes AMM CUL-1, as described in Section 2.2.13.4, to avoid impacts to archeological resources if they are found during excavation activities. Therefore, there would be no impact.

### c) Disturb any human remains, including those interred outside of dedicated cemeteries?

**No Impact.** There are no formal cemeteries or known burial sites in the project area. Project construction is not expected to disturb any human remains. The project includes AMM CUL-2, as described in Section 2.2.13.4, to avoid impacts to human remains. Therefore, there would be no impact.

#### 3.3.6 Energy

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Result in potentially significant environmental impact due to wasteful, inefficient, or unnecessary consumption of energy resources, during project construction or operation?	No	No	No	Yes
b) Conflict with or obstruct a state or local plan for renewable energy or energy efficiency?	No	No	No	Yes

#### 3.3.6.1 CEQA Significance Determinations for Energy

## a) Result in potentially significant environmental impact due to wasteful, inefficient, or unnecessary consumption of energy resources, during project construction or operation?

**No Impact.** Project construction and maintenance would be a temporary commitment of energy, necessary for any infrastructure improvement project. Energy in the form of gas and diesel would be consumed during construction and ongoing maintenance activities by construction vehicles and equipment operating on site, trucks delivering equipment and supplies, and construction workers driving to and from the project site. Energy consumption during construction would be conserved and minimized to the maximum extent feasible. Energy conservation in construction activities is assumed because the construction contractor would have a financial incentive and statutory mandate to minimize waste and externalities.

As described in Section 2.3.8, all Build Alternatives would reduce the potential for wasteful energy due to a reduction in stop-and-go traffic conditions.

For the reasons listed above, the project would not result in wasteful, inefficient, or unnecessary consumption of energy resources during project construction or operation. There would be no impact.

### b) Conflict with or obstruct a state or local plan for renewable energy or energy efficiency?

**No Impact**. The State of California Energy Action Plan and the Integrated Energy Policy Report regulate energy conservation throughout the state. The State of California Energy Action Plan was adopted to ensure adequate, reliable, and reasonably priced electrical power and natural gas quantities through policies that are cost-effective and environmentally conscious for California's residents (CEC 2003). California policies influenced by the California Global Warming Solutions Act (AB 32), are demonstrated in the 2007 Integrated Energy Policy Report, which is updated regularly to provide policy recommendations to meeting the state's energy demands while addressing carbon constraints. According to SB 100, the state is targeting 100 percent renewable or carbon-free energy usage by 2045. The California Energy Commission's (CEC's) Clean Transportation Program leverages public and private investments to support adoption of cleaner transportation, powered by alternative and renewable fuels.

The proposed project involves creating HOV lanes to alleviate traffic congestion along SR 37. The Build Alternatives would not conflict with any state or regional Energy Conservation Plans described above because it would not cause wasteful, inefficient, or unnecessary consumption use of energy, or wasteful use of energy resources in the project area or region. Therefore, there would be no impact.

#### 3.3.7 Geology and Soils

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
<ul> <li>a) Directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving:</li> <li>i) Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map issued by the State Geologist for the area or based on other substantial evidence of a known fault? Refer to Division of Mines and Geology Special Publication 42.</li> </ul>	No	No	No	Yes
ii) Strong seismic ground shaking?	No	No	No	Yes
iii) Seismic-related ground failure, including liquefaction?	No	No	No	Yes
iv) Landslides?	No	No	No	Yes
b) Result in substantial soil erosion or the loss of topsoil?	No	No	No	Yes
c) Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse?	No	No	No	Yes
d) Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property?	No	No	No	Yes
e) Have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal systems where sewers are not available for the disposal of wastewater?	No	No	No	Yes
f) Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?	No	No	Yes	No

#### 3.3.7.1 CEQA Significance Determinations for Geology and Soils

a) Directly or indirectly cause potential substantial adverse effects, including the risk of loss, injury, or death involving: i) Rupture of a known earthquake fault, as delineated on the most recent Alquist-Priolo Earthquake Fault Zoning Map; ii) Strong seismic ground shaking?; iii) Seismic-related ground failure, including liquefaction?; or iv) Landslides?

**No Impact.** The project is in a seismically active area but is not in the Alquist-Priolo Earthquake Fault Zone. The proposed project would not exacerbate the potential for seismic shaking; the intensity of the earthquake ground motion at the site would depend on the characteristics of the generating fault, distance to the earthquake epicenter, magnitude, and duration of the earthquake, and specific site geologic conditions. Caltrans' design and construction guidelines incorporate engineering standards that address seismic risks, including ground failure related to liquefaction, landslides, and lateral spreading. Project elements would be designed and constructed to meet seismic design requirements for ground shaking and ground motions, as determined for the project vicinity and site conditions. Caltrans also requires additional geotechnical subsurface and design investigations to be performed during the final project design and engineering phase. These standards and requirements would avoid the potential for adverse impacts related to seismic activity. Furthermore, Caltrans would implement AMMs described in Section 2.3.3.4. Therefore, there would be no impact.

#### b) Result in substantial soil erosion or the loss of topsoil?

**No Impact.** During construction of the project, earthmoving activities such as grading, excavation, and trenching have the potential to result in soil erosion and loss of topsoil, especially in areas where there are steeper slopes. Because the project alignment is relatively flat and a large portion of the project lies on artificial fill (clayey soils with a low erosion potential), there would be lower potential for substantial soil erosion to occur. The embankment of the bridge has greater slopes but is on soils with a low erosion potential. Nonetheless, BMPs such as stabilization by paving, rock slope protection, and erosion control would be implemented to reduce erosional impacts during construction activities. Therefore, there would be no impact related to soil erosion or loss of topsoil.

#### c) Be located on a geologic unit or soil that is unstable, or that would become unstable as a result of the project, and potentially result in on- or off-site landslide, lateral spreading, subsidence, liquefaction, or collapse?

**No Impact.** The project alignment would not be in a geologic unit that is unstable, nor would the project result in geologic units or soils becoming unstable. The project alignment is relatively flat, exclusive of the bridge approach embankment. The likelihood of landslides to affect the proposed project is considered low. The potential for lateral spreading is expected to be low based on conditions revealed in historic borings. The project alignment is underlain by soils with low to moderate liquefaction susceptibility. Subsidence and settlement have the potential to occur, but the project

would be designed to account for this settlement. Furthermore, Caltrans would implement AMMs described in Section 2.3.3.4. Therefore, there would be no impact.

## d) Be located on expansive soil, as defined in Table 18-1-B of the Uniform Building Code (1994), creating substantial direct or indirect risks to life or property?

**No Impact.** The near-surface fill is expected to have low expansion potential; the underlying native Bay Mud has a plasticity index of 50 or more and is expected to exhibit high shrink/swell behavior. Caltrans' design and construction guidelines incorporate engineering standards that address expansive soils. Furthermore, Caltrans would implement project features described in Section 2.3.3.4. Therefore, there would be no impact.

## e) Have soils incapable of adequately supporting the use of septic tanks or alternative wastewater disposal systems where sewers are not available for the disposal of wastewater?

**No Impact.** The proposed project would not involve incorporating septic tanks or other wastewater disposal systems. Therefore, there would be no impact.

### f) Directly or indirectly destroy a unique paleontological resource or site or unique geologic feature?

**Less than Significant.** No paleontological resources have previously been recorded within the project limits and none were found during the field study. However, there is a potential to encounter unknown paleontological resources during project construction because there are three fossil localities in the vicinity of the project limits and there are areas within the project limits that have high potential sensitivity.

Direct impacts to paleontological resources could occur during project construction involving earthmoving activities such as grading, excavation, and boring. Direct impacts are the destruction of the fossil remains and the geographic, geologic, phylogenetic, and taphonomic information associated with them. There is greater potential for direct impacts to occur in the two high geological rock units, Qpf and Tps. These areas are in the western portion of the project limits. It is highly unlikely that any impacts would occur in geological rock units af, alf, Qhf and Qhbm, which make up most of the project limits. As necessary, Caltrans would prepare a Paleontology Mitigation Plan as a standard measure to avoid direct impacts to address and minimize effects in soils that have a higher potential to contain paleontological resources, as described in Section 2.3.4.4. Therefore, impacts are expected to be less than significant.

#### 3.3.8 Greenhouse Gas Emissions

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?	No	No	Yes	No
b) Conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases?	No	No	Yes	No

#### 3.3.8.1 CEQA Significance Determinations for Greenhouse Gas Emissions

### a) Generate greenhouse gas emissions, either directly or indirectly, that may have a significant impact on the environment?

**Less than Significant Impact.** Project construction would result in GHG emissions. This would consist primarily of emissions from equipment exhaust and worker and vendor vehicle trips. Construction emissions were estimated using the latest Sacramento Metropolitan Air Quality Management District's Road Construction Model RCEM version 9.0. Table 2-22 in Section 2.3.6 shows the construction emissions associated with the project for each construction activity, which would amount to 5,994 metric tons. Carbon dioxide equivalent (CO<sub>2</sub>e) emissions from earthmoving activities and use of equipment and vehicles would be similar for all Build Alternatives except Alternative 3B, which involves widening of Sonoma Creek Bridge. This widening would require additional structures (steel, concrete, and other materials) and construction staging, all of which would result in additional GHG emissions.

Long-term operational emissions associated with vehicular traffic in the region were evaluated quantitively. GHG emissions impacts for the No Build and Build Alternatives were computed using CT-EMFAC 2017 for existing conditions (year 2020) and future years (2025 Opening Year, 2040 RTP, and 2045 Design Year). The SAFE vehicle emissions adjustment factors developed by CARB for carbon dioxide ( $CO_2$ ) were applied to the CT-EMFAC2017 emissions factors, and the adjusted emissions factors were applied to the project area VMT estimates.

These results are shown in Table 3-6 in Section 3.4.3. The mobile GHG emissions in the region would decrease from existing conditions compared to any of the Build Alternatives and options in the study years 2025, 2040, and 2045. For CEQA analysis, this is considered a less than significant impact (because future emissions for any of the Build Alternatives are less than existing conditions). This decrease for each of the study years would occur with or without the project, primarily due to improvements in technology, vehicle fleet transition to improved cars, and reformulation of fuels.

The changes in GHG and VMT vary by alternative and study year when compared to the No Build Alternative. The evaluation of VMT is discussed in more detail in

Sections 2.2.11 and 3.3.17, including measures that would reduce VMT. In 2025, vehicle emissions of GHGs decline for all alternatives compared to the No Build Alternative. VMT and GHG results are listed in Table 3-6 for the 2020 baseline conditions, and the 2025 and 2045 study years for all alternatives.

For 2025, GHG emissions will be less than existing baseline conditions for all build alternatives (no significant impact) with the proposed additional lane in each direction and conversion of an existing travel lane to an HOV lane in each direction. GHG emissions in 2025 for all Build Alternatives will also be less than the No Build Alternative. With the proposed implementation of tolling in either one direction or both direction, GHG emissions will be further reduced in comparison to the No Build alternative.

In 2040, GHG emissions will be less than existing baseline conditions for all build alternatives (no significant impact). GHG emissions and VMT decline for Alternatives 1 and 2 below the No Build Alternative. With Alternatives 3A and 3B and no tolling, GHG emissions increase compared to the No Build Alternative, associated with increased VMT and an increase in the number of vehicles at higher speeds. However, with implementation of an HOV lane and tolling of the proposed general purpose lane, both VMT and GHG are below the No Build Alternative for all Build Alternatives.

In 2045, GHG emissions will also be less than existing baseline conditions for all build alternatives (no significant impact). GHG emissions decline for Alternatives 1 and 2, and would be less than the No Build Alternative. GHG emissions increase with Alternatives 3A and 3B, compared to the No Build Alternative, including when tolling is applied.

For all alternatives, the analysis of GHG emissions using the CT-EMFAC2017 model is not sensitive to (does not fully account for) reductions in emissions due to improvements in traffic flow, such as reducing stop-and-go traffic congestion. Reducing queues and stop-and-go traffic with the Build Alternatives would reduce idling and may reduce GHG emissions due to idling in a way that is not fully demonstrated by the analysis. Therefore, factoring these benefits in would result in additional GHG emission reduction benefits compared to the No Build option. Furthermore, GHG reduction measures would be implemented during construction to limit GHG emissions.

Because the Build Alternatives would not contribute to substantial increases in GHG emissions over existing conditions, and GHG reduction measures would be implemented during construction, the impact would be less than significant.

### b) Conflict with an applicable plan, policy or regulation adopted for the purpose of reducing the emissions of greenhouse gases?

**Less than Significant Impact.** The BAAQMD's 2017 clean air plan, *Spare the Air, Cool the Climate,* addresses GHGs in the project region. Additionally, Solano and Napa Counties have climate action plans. The Scoping Plan for Achieving California's 2030 Greenhouse Gas Target identifies reductions in GHG emissions goals.

Caltrans work would comply with all local, state, and federal regulations, ordinances, and statutes that apply to GHG emissions. The project is expected to improve GHG emissions by reducing traffic congestion; with the proposed HOV lane and tolling of the general purpose lanes, VMT would also be reduced, which would reduce GHG emissions. Furthermore, the project would promote and support ridesharing, carpooling, vanpooling, and public transportation, which is consistent with these plans. With implementation of construction GHG-reduction measures, the impact would be less than significant.

3.3.9	Hazards and	Hazardous	Materials

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials?	No	No	No	Yes
b) Create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment?	No	No	Yes	No
c) Emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within one-quarter mile of an existing or proposed school?	No	No	No	Yes
d) Be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would it create a significant hazard to the public or the environment?	No	No	Yes	No
e) For a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard or excessive noise for people residing or working in the project area?	No	No	No	Yes
f) Impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan?	No	No	Yes	No
g) Expose people or structures, either directly or indirectly, to a significant risk of loss, injury or death involving wildland fires?	No	No	No	Yes

#### 3.3.9.1 CEQA Significance Determinations for Hazards and Hazardous Materials

### a) Create a significant hazard to the public or the environment through the routine transport, use, or disposal of hazardous materials?

**No Impact.** The proposed project would involve the transport, use, and disposal of hazardous materials used for construction of the project (e.g., fuels, paints, asphalt, and lubricants). Adherence to federal and state regulations during project construction and maintenance reduces the risk of exposure to hazardous materials and accidental hazardous materials releases. Compliance with existing regulations is mandatory. Therefore, construction of the project is not expected to create a hazard to construction workers, the public, or the environment through the routine transport, use, disposal, or accidental release of hazardous materials. There would be no impact.

#### b) Create a significant hazard to the public or the environment through reasonably foreseeable upset and accident conditions involving the release of hazardous materials into the environment?

**Less than Significant.** During construction, hazardous materials such as fuels, paints, asphalt, and lubricants would be used. These materials could pose a threat to human health or the environment if not properly managed. Adherence to federal and state regulations during project construction and maintenance would reduce the risk of exposure to hazardous materials and accidental releases of hazardous materials. Compliance with existing regulations is mandatory. Therefore, construction of the proposed project is not expected to create a hazard to construction workers, the public, or the environment.

Furthermore, construction of the proposed project could result in the potential disturbance of hazardous materials in soil, groundwater, and building materials in the project area. Shallow soils along SR 37 that would be excavated during construction are likely to contain ADL at concentrations above DTSC-regulated levels. UXO, mustard gas, or similar military hazards may also be encountered. Hazardous materials such as ACM, lead-based paint, treated-wood waste, and PCBs could also be encountered during construction and maintenance activities. The implementation of the project features for hazardous materials summarized in Table 1-4 would avoid and/or minimize impacts associated with hazardous materials. Impacts involving the release of hazardous materials are anticipated to be less than significant.

## c) Emit hazardous emissions or handle hazardous or acutely hazardous materials, substances, or waste within one-quarter mile of an existing or proposed school?

**No Impact.** There are no existing schools or proposed schools within 0.25 mile of the project limits. There would be no impact.

## d) Be located on a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962.5 and, as a result, would it create a significant hazard to the public or the environment?

Less than Significant Impact. The project area is not in a site which is included on a list of hazardous materials sites compiled pursuant to Government Code Section 65962. However, DTSC EnviroStor identified 15 potential contaminated sites and/or investigated sites within a 0.5-mile radius of the project limits. All but three of these sites are at Mare Island, Skaggs Island and Tubbs Island, which were previously used for military activities. Thirteen of these sites are listed as closed cases, which indicates that hazardous materials have been addressed to the satisfaction of the applicable regulatory authority, and a formal closure decision document has been issued. Of the 19 sites listed in the regulatory databases, one site was determined to have a greater potential to impact the proposed project: the former Tubbs Island Gunnery Range on the Coast of San Pablo Bay in Petaluma. The former gunnery range historical use may present hazardous materials in the soil within the project limits. The implementation of project features summarized in Table 1-4 would avoid and/or minimize impacts associated with hazardous materials. Therefore, impacts would be less than significant.

#### e) For a project located within an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project result in a safety hazard or excessive noise for people residing or working in the project area?

**No Impact.** The project area is not in an airport land use plan or within 2 miles of a public airport or public use airport where the project would result in a safety hazard for people residing or working in the project area. Therefore, there would be no impact.

### f) Impair implementation of or physically interfere with an adopted emergency response plan or emergency evacuation plan?

**Less than Significant Impact.** None of the Build Alternatives would impair implementation of an emergency response or emergency evacuation plan. During construction, implementation of the TMP would minimize construction-related delays and would include coordination with CHP and local law enforcement agencies. Even with the implementation of the TMP, there may be slower traffic due to construction activities. Although access would be maintained for emergency response vehicles, they may be required to move around traffic, which could result in slight delays.

During project operation, Build Alternatives 3A and 3B would result in the loss of a wider shoulder for emergency vehicle use, but the presence of two lanes in each direction on SR 37 would allow traffic to move over to the other lane, allowing emergency vehicles to pass them. Allowing traffic to move over may result in slight delays. Therefore, the Build Alternatives would have a less than significant impact.

### g) Expose people or structures, either directly or indirectly, to a significant risk of loss, injury or death involving wildland fires?

**No Impact.** Although all the Build Alternatives involve widening SR 37, none of them would change the alignment of the SR or any adjacent land uses. Section 3.3.20 describes fire hazard conditions in the project area and the reasons why the project alternatives are not anticipated to exacerbate wildfire risks. Project construction and operation would not expose people or structures to significant risks involving wildland fires. There would be no impact.

#### 3.3.10 Hydrology and Water Quality

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or ground water quality?	No	No	Yes	No
b) Substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the project may impede sustainable groundwater management of the basin?	No	No	Yes	No
<ul> <li>c) Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces, in a manner which would: <ul> <li>(i) result in substantial erosion or siltation on-or off-site;</li> </ul> </li> </ul>	No	No	Yes	No
<ul> <li>(ii) substantially increase the rate or amount of surface runoff in a manner which would result in flooding on- or offsite;</li> </ul>	No	No	Yes	No
(iii) create or contribute runoff water which would exceed the capacity of existing or planned stormwater drainage systems or provide substantial additional sources of polluted runoff; or	No	Yes	No	No
(iv) impede or redirect flood flows?	No	No	Yes	No
d) In flood hazard, tsunami, or seiche zones, risk release of pollutants due to project inundation?	No	No	Yes	No
e) Conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan?	No	No	No	Yes

#### 3.3.10.1 CEQA Significance Determinations for Hydrology and Water Quality

### a) Violate any water quality standards or waste discharge requirements or otherwise substantially degrade surface or ground water quality?

Less than Significant Impact. During construction, temporary water quality impacts have the potential to occur from sediment discharge from DSAs; construction activities such as grading and excavation near water sources; use of construction vehicles and equipment; and drainage facilities that discharge to water bodies. Table 2-19 shows the amount in acres of DSA for each Build Alternative. Construction site BMPs for erosion and sediment control and material management are considered project features and would be specified in the SWPPP prior to construction, and monitored during construction. These measures are consistent with the practices required under the statewide Construction General Permit. Permanent impacts to water quality could result from the addition of impervious area, which can prevent runoff from naturally dispersing and infiltrating into the ground. Table 2-19 shows the amount of impervious surface added in acres for each Build Alternative. Permanent erosion control measures would be applied to all exposed areas, once grading or soil disturbance work is completed, as a permanent measure to achieve final slope stabilization. Furthermore, implementation of water quality project features is required for this project, in compliance with federal, state, and local requirements. These features would minimize the potential for water quality impacts from runoff entering storm drains. The project would require a 401 Water Quality Certification from the RWQCB. WQ-2 would implement permanent stormwater treatment measures, as described in Section 2.3.2.4. Therefore, the project would have less than significant impacts to water quality and would not violate any water quality standards.

## b) Substantially decrease groundwater supplies or interfere substantially with groundwater recharge such that the project may impede sustainable groundwater management of the basin?

Less than Significant Impact. The project does not involve pumping and/or using groundwater. However, the added impervious surface from the project has the potential to reduce the availability of unpaved areas where runoff can infiltrate into native soils and recharge aquifers. Table 2-19 shows the amount of impervious surface added in acres for each Build Alternative. Stormwater treatment BMPs would allow for stormwater infiltration to minimize impacts to runoff and groundwater, however there is limited area along the route for new treatment options and off site mitigation for runoff is proposed as part of the project. Long-term dewatering activities are also not anticipated for this project. Therefore, permanent impacts to groundwater are not anticipated, and the impact would be less than significant.

## c) Substantially alter the existing drainage pattern of the site or area, including through the alteration of the course of a stream or river or through the addition of impervious surfaces.

Less than Significant Impact with Mitigation.

- i) Temporary water quality impacts can result from sediment discharge from construction near water resources or drainage facilities that discharge to water bodies. BMPs would be implemented during construction to prevent sediment-laden flows from leaving the construction site, such as such as temporary silt fencing, temporary drainage inlet protection, and street sweeping. A SWPPP under the Construction General Permit would also be developed, which would contain soil erosion and pollution prevention control measures. Permanent erosion control measures would be applied to all exposed areas once grading or soil disturbance work is completed as a permanent measure to achieve final slope stabilization. These measures may include hydraulically applying a combination of hydroseed with native seed mix, hydromulch, straw, tackifier, and compost to promote vegetation establishment, and installing fiber rolls to prevent sheet flow from concentrating and causing gullies. Therefore, the project is expected to result in a less than significant impact related to on- and offsite soil erosion and siltation.
- ii) The project would require temporary dewatering for excavation activities for work in Tolay Creek during construction activities. Dewatering may also be required along the side slopes through the length of the corridor. Dewatering activities would comply with Caltrans' Field Guide to Construction Site Dewatering (2014a) and Caltrans' Standard Specifications (2020), and, if needed, a separate dewatering permit would be obtained prior to the start of construction. However, the existing drainage patterns are not anticipated to be impacted, because the goal is to maintain existing drainage patterns. Furthermore, impervious surface added to the project area would not result in substantially increased runoff, because the amount added is small. Therefore, the project is not expected to result in surface runoff that would cause flooding.
- iii) Given that the project area has shallow groundwater, other conventional treatment measures that capture and treat stormwater runoff may need to be considered; these devices could include basins or media filters. Due to the presence of environmentally sensitive areas and limited available treatment area for BMPs, the project would also include offsite stormwater treatment options as well. Implementation of WQ-01: Offsite Stormwater Treatment would address the site's limited onsite stormwater treatment capacity. Appropriate mitigation and coordination with RWQCB during the project's final design phase. The project would be programmed to meet the requirements of Caltrans' current MS4 and NPDES permits, following the guidelines and procedures outlined in Caltrans' latest Statewide SWMP to address stormwater runoff; and in accordance with Memorandum of Caltrans Post-Construction Stormwater and Hydromodification Standards (SFRWQCB 2008). Therefore, the impacts would be less than significant with mitigation.

iv) Although all the Build Alternatives would require different amounts of permanent fill to widen the highway, most improvements in the project would rework existing impervious areas. According to the project's *Location Hydraulic Study Report*, the amount of new impervious surface area added would not have an impact to the flows within the project's limits and would not impact existing floodplain conditions (WRECO, 2021). The proposed fill placed in the floodplain is relatively minor in the context of the greater floodplain area and is not anticipated to impede flood waters, affect bay level floodplains, or substantially reduce the area available to convey floodwaters. Therefore, there would be a less than significant impact.

### d) In flood hazard, tsunami, or seiche zones, risk release of pollutants due to project inundation?

**Less than Significant Impact.** As discussed in Section 2.3.1.1, the majority of the project is in Zone AE. This zone represents the base floodplain with areas subject to flooding by the 100-year flood event, where base floodplain elevations are provided. Portions of the project area adjacent to Tubbs Island (between Tolay Creek and Sonoma Creek) are in SFHA Zone VE, which consists of coastal areas subject to coastal high-hazard flooding and to inundation by the 1-percent-annual-chance flood event. A small portion of the project area just east of the Tolay Creek crossing is in a shaded Zone X area. Zone X represents areas of moderate flood hazard, usually depicted on FIRMs as between the limits of the base and 500-year floods. As described in more detail in Section 2.3.1, the project would not change the overall land use in the watershed basin under any of the Build Alternatives. Most improvements in the project would be in the existing impervious area and would not significantly change or encroach into the 100-year floodplain. In Zone AE of the SR 37 corridor, widening of the existing lanes to accommodate the proposed roadway improvements would not significantly increase the fill in the floodplain.

The project area is mapped in a Tsunami Inundation Area (CalEMA 2009) and could be subject to flood waters from a tsunami event. Tsunamis are a result of offshore earthquakes, and any wave inundation is generally more of a risk to the California coastal area than to an inland Bay shoreline, but water could still be temporarily elevated in the Bay. An offshore earthquake event may allow a period of time for emergency response along coastal or Bay shorelines. Originating off-shore, elevated water from a tsunami is not expected to reach the project area quickly because of the distance it has to travel. Local, state, and federal agencies work in cooperation to provide notification and identify response strategy for tsunami events (OES 2007). The National Oceanic and Atmospheric Administration National Weather Service is responsible for issuing information about and warnings of possible tsunamis via the Tsunami Warning System (NOAA 2021). This information is delivered to the California Office of Emergency Services Warning Center, which passes this information directly to the Operational Areas via alert systems. For example, areas within a 3-hour tsunami travel time of the epicenter are placed in a Tsunami Warning status, and areas within a 3- to 6-hour tsunami travel time are placed in a Tsunami Watch status (OES 2007).

The CHP, police departments, and sheriff's offices can close a highway to traffic if there is a threat to public health or safety (OES 1999). Because the impacts of flood waters from a tsunami would likely take time to reach the project area, there would be time to restrict and/or close down SR 37. An emergency closure would substantially decrease the potential for risk of exposure of traffic, or a spill or other impact of pollution from such an event were it to occur.

Therefore, the project is not expected to result in significant risk of increased pollution due to flooding or inundation, and this would be a less than significant impact.

### e) Conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan?

**No Impact.** The project is required to adhere to the CWA, the Porter-Cologne Water Quality Control Act, Caltrans' MS4 Permit, and the other laws and regulations described in Section 2.3.2.1. As a result, the project would not conflict with or obstruct implementation of a water quality control plan or sustainable groundwater management plan. Therefore, there would be no impact.

#### 3.3.11 Land Use and Planning

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Physically divide an established community?	No	No	No	Yes
b) Cause a significant environmental impact due to a conflict with any land use plan, policy, or regulation adopted for the purpose of avoiding or mitigating an environmental effect?	No	No	Yes	No

#### 3.3.11.1 CEQA Significance Determinations for Land Use and Planning

#### a) Physically divide an established community?

**No Impact.** The project would be constructed in and along Caltrans' right-of-way. Additionally, the existing SR 37 alignment acts as a physical barrier and separator within the communities it traverses. The project would not physically divide an established community and there would be no impact.

#### b) Cause a significant environmental impact due to a conflict with any land use plan, policy, or regulation adopted for the purpose of avoiding or mitigating an environmental effect?

**Less than Significant Impact.** All Build Alternatives would be generally consistent with the general plans, regional plans, and transportation plans discussed in Section 2.2.2. Alternatives 2 and 3A would not be consistent with Sonoma and Solano Counties' policies, and Caltrans' Bicycle Plans and Programs regarding

accommodating bicyclists as they would reduce the available shoulder space. Alternative 2 would preclude bicyclists in the peak direction during peak hours and Alternative 3A would preclude bicyclists at Sonoma Creek Bridge. However, the project would not substantially change local land use patterns. The environmental impact of the project due to a conflict with any land use plan, policy, or regulation adopted to avoid or mitigate an environmental effect would be less than significant.

#### 3.3.12 Mineral Resources

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Result in the loss of availability of a known mineral resource that would be of value to the region and the residents of the state?	No	No	No	Yes
b) Result in the loss of availability of a locally important mineral resource recovery site delineated on a local general plan, specific plan, or other land use plan?	No	No	No	Yes

#### 3.3.12.1 CEQA Significance Determinations for Mineral Resources

# a) and b) Result in the loss of availability of a mineral resource that would be of value to the region and residents of the state or locally important mineral resource recovery site delineated on a local general plan, specific plan, or other land use plan?

**No Impact.** The proposed project would not result in the loss of availability of mineral resources that would be of value to the region or state or result in the loss of locally important mineral resources. The project area does not overlap with any known mining operations. There would be no impact to mineral resources.

#### 3.3.13 Noise

Would the project result in:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Generation of a substantial temporary or permanent increase in ambient noise levels in the vicinity of the project in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies?	No	No	No	Yes
b) Generation of excessive groundborne vibration or groundborne noise levels?	No	No	Yes	No

Would the project result in:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
c) For a project located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project expose people residing or working in the project area to excessive noise levels?	No	No	No	Yes

#### 3.3.13.1 CEQA Significance Determinations for Noise

a) Generation of a substantial temporary or permanent increase in ambient noise levels in the vicinity of the project in excess of standards established in the local general plan or noise ordinance, or applicable standards of other agencies?

**No Impact.** The project would not increase ambient noise levels by more than 2 dBA. Most short-term construction noise would be similar to existing ambient highway noise levels. However, short-term noise levels would result from construction methods such as pile driving, which would be temporarily higher than existing ambient noise levels. However, these construction noises would be short-term and intermittent. Therefore, there would be no impact.

#### b) Generation of excessive groundborne vibration or groundborne noise levels?

**Less than Significant Impact.** Groundborne vibration was evaluated for the project. Pile driving would generate higher levels for short periods when the pile driving is at the upper limit of activity. Elevated vibration levels during construction would be shortterm and temporary during pile driving activities. Furthermore, vibration levels were evaluated at the nearest sensitive receptor, the Refuge Headquarters, and it was determined that it would not exceed Caltrans criteria for vibration impacts. Because vibration would be higher during pile driving but would not exceed Caltrans standards for vibration, the project would have a less than significant impact.

c) For a project located within the vicinity of a private airstrip or an airport land use plan or, where such a plan has not been adopted, within two miles of a public airport or public use airport, would the project expose people residing or working in the project area to excessive noise levels?

No Impact. The project is not in the vicinity of an airport. There would be no impact.

#### 3.3.14 Population and Housing

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Induce substantial unplanned population growth in an area, either directly (for example, by proposing new homes and businesses) or indirectly (for example, through extension of roads or other infrastructure)?	No	No	No	Yes
b) Displace substantial numbers of existing people or housing, necessitating the construction of replacement housing elsewhere?	No	No	No	Yes

#### 3.3.14.1 CEQA Significance Determinations for Population and Housing

## a) Induce substantial unplanned population growth in an area, either directly (for example, by proposing new homes and businesses) or indirectly (for example, through extension of roads or other infrastructure)?

**No Impact.** The project would not induce substantial population growth, directly (e.g., through construction of new homes or businesses) or indirectly (e.g., through extension of roads or other infrastructure). As noted in Section 2.2.6, the project would provide additional capacity in Caltrans' right-of-way to help alleviate congestion and improve travel time reliability along SR 37. Furthermore, the project would not provide new access to previously undeveloped land. The project would accommodate planned growth but would not result in reasonably foreseeable changes to planned land uses both adjacent to and in the vicinity of the project study area. The project is not expected to induce substantial population growth, either directly or indirectly, and there would be no impact.

### b) Displace substantial numbers of existing people or housing, necessitating the construction of replacement housing elsewhere?

**No Impact.** The project would not require residential or business relocation and would not displace substantial numbers of existing people or housing, necessitating the construction of replacement housing elsewhere. There would be no impact.

#### 3.3.15 Public Services

a) Would the project result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for any of the public services:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
i) Fire protection?	No	No	No	Yes
ii) Police protection?	No	No	No	Yes
iii) Schools?	No	No	No	Yes
iv) Parks?	No	No	No	Yes
v) Other public facilities?	No	No	No	Yes

#### 3.3.15.1 CEQA Significance Determinations for Public Services

a) Result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities, need for new or physically altered governmental facilities, the construction of which could cause significant environmental impacts, in order to maintain acceptable service ratios, response times or other performance objectives for any of the public services:

i) Fire protection?

ii) Police protection?

iii) Schools?

iv) Parks?

v) Other public facilities?

**No Impact.** The project would not involve construction of new housing or other land uses that could increase the local population and demand for governmental facilities and services, such as fire protection, police protection, schools, or parks. Project construction has the potential to increase traffic delays on SR 37 that could affect response times of emergency response vehicles. However, a TMP would be implemented to minimize construction-related delays and ensure accessibility throughout the corridor for emergency service providers. Therefore, the project would not result in substantial adverse physical impacts associated with the provision of new or physically altered governmental facilities. There would be no impact.

#### 3.3.16 Recreation

	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Would the project increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated?	No	No	Yes	No
b) Does the project include recreational facilities or require the construction or expansion of recreational facilities which might have an adverse physical effect on the environment?	No	No	No	Yes

#### 3.3.16.1 CEQA Significance Determinations for Recreation

## a) Would the project increase the use of existing neighborhood and regional parks or other recreational facilities such that substantial physical deterioration of the facility would occur or be accelerated?

**Less than Significant Impact.** As discussed in Section 2.2.4, the project may require TCEs for equipment access and staging at park and recreational facilities during construction. The TCEs would not preclude or substantially impede the use of any parks or recreation facilities during construction. However, implementation of Build Alternatives 3A and 3B would require permanent partial acquisitions from the Refuge and NSMWA. Due to the minimal area of use in the Refuge and NSMWA, and the location of permanent use directly adjacent to SR 37, which provides limited recreational value, Build Alternatives 3A and 3B are not anticipated to adversely affect ongoing recreation and habitat conservation activities. The project would not increase the use of existing neighborhood and regional parks or other recreational facilities in such a way that substantial physical deterioration as result of the temporary or permanent use of the park and recreation areas would occur and impacts would be less significant.

## b) Does the project include recreational facilities or require the construction or expansion of recreational facilities which might have an adverse physical effect on the environment?

**No Impact.** The project would not include the construction of new recreational facilities or require construction or expansion of recreational facilities that would generate an adverse physical effect on the environment. There would be no impact.

#### 3.3.17 Transportation

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Conflict with a program, plan, ordinance, or policy addressing the circulation system, including transit, roadway, bicycle, and pedestrian facilities?	No	No	Yes	No
b) Would the project conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b)?	No	No	Yes	No
c) Substantially increase hazards due to a geometric design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment)?	No	No	No	Yes
d) Result in inadequate emergency access?	No	No	No	Yes

#### 3.3.17.1 CEQA Significance Determinations for Transportation

The existing traffic conditions are described in Section 2.2.11.3. This section uses the No Build Alternative as the CEQA baseline.

### a) Conflict with a program, plan, ordinance, or policy addressing the circulation system, including transit, roadway, bicycle, and pedestrian facilities?

**Less than Significant Impact.** The proposed project traverses Sonoma, Solano, and Napa counties and the city of Vallejo. These jurisdictions have plans and policies regarding local circulation. However, SR 37 is part of the State Highway System and is under Caltrans' jurisdiction. The project is expected to improve traffic conditions along SR 37 by adding HOV lanes, which would improve the person-carrying capacity of the corridor, as well as the traffic flow and travel times in the peak direction. Bicyclists are permitted. The project would not conflict with a program, plan, ordinance, or policy related to transportation, and impacts would be less than significant.

### b) Would the project conflict or be inconsistent with CEQA Guidelines section 15064.3, subdivision (b)?

**Less than Significant Impact.** SB 743 (2013) requires the Governor's Office of Planning and Research to identify new metrics for identifying and mitigating transportation impacts under CEQA. Under SB 743, CEQA Guidelines Section 15064.3(b) was revised to identify VMT as the most appropriate measure of assessing transportation impacts.

**Construction VMT**. During construction, vehicles trips would increase temporarily associated with workers traveling to and from the job sites, construction equipment trips, and transport of materials. Because of the limited accessibility for parking and

staging, vehicle trips would be coordinated by parking off site and consolidating trips to and from the work locations along SR 37. Public outreach and notifications would be used to inform drivers using SR 37 to expect delays, which would likely discourage discretionary trips and reduce VMT during construction (or divert vehicles to alternative routes which may temporarily increase VMT). Vehicle trips associated with construction would be temporary and limited to the construction period only.

**Post-Construction VMT**. As described in Section 2.2.11.3, the daily VMT under all Build Alternatives for 2025 (opening year) and 2045 (design year) is estimated to increase from No Build conditions without tolling. The non-tolled increase in VMT for the alternatives for both study years is summarized below in Table 3-3. Comparing the study years in Table 3-3 also represents how the needed reduction in VMT and number of vehicle trips are tied to the study year, and to offset the VMT increase more reduction in vehicle trips is needed in 2045 than in the near term in 2025.

VMT modeling included the daily trips within the entire Bay Area (all counties). This allowed a comparison between alternatives that captured local and regional trips that divert or change routes associated with each alternative.

Many of the trips in the north Bay Area relying on SR 37 are regional in that they cross multiple counties and tend to be long-distance trips. Based on the MTC regional model, the average vehicular trip length on SR 37 is about 46 miles. This is the approximate distance between Fairfield and San Rafael, or between Fairfield and Petaluma. To illustrate how VMT changes with the project, for every daily change of 10,000 VMT, an average trip length of 46 miles would result in about 217 vehicular trips on this corridor daily.

Scenario	Daily VMT Change
2025 Alternatives 1 and 2	6,346
2025 Alternatives 3A and 3B	9,598
2045 Alternatives 1 and 2	31,729
2045 Alternatives 3A and 3B	47,992

#### Table 3-3 Summary of VMT and Vehicle Trip Increases by Alternative (no tolling)

The proposed project with tolling implemented would reduce VMT compared to a proposed project with no tolling, as described in the following subsections. Performance measures would be used to evaluate or monitor implementation of the proposed actions. The proposed HOV lane in each direction would be enforced as full-time HOV lanes (24 hours per day, 7 days a week). Tolling of the proposed general purpose lane in either one direction or both directions would result in a decrease in VMT compared to the No Build Alternative. Implementing tolling on this section of SR 37 would require approval by the CTC. Tolling is, however, not the only solution available to effectively maintain or reduce VMT, and the following paragraphs describe tolling and other means to maintain VMT at or below the No Build alternative. These actions are not mutually exclusive; one or all could be implemented.

**Implement Tolling and HOV lanes:** Tolling is a proposed project feature. The project description in Section 1.4.1.3 includes tolling equipment that would be installed when the project is constructed and Section 2.2.11.3 describes two tolling methods, either one direction only tolling or tolling in both directions. The tolling equipment includes overhead gantries with toll reading equipment, advance signs notifying drivers of upcoming tolls, cameras to read FasTrak and license plate identification, and the associated power and communications lines.

HOV lanes would not be tolled. Toll fares for the proposed general purpose lanes would be consistent with other Bay Area bridges, with means-based discounts applied. The implementation of tolling would require an application by MTC and approval by the CTC. An SR 37 Toll Authority may be established with oversight by the Bay Area Infrastructure Financing Authority, similar to Bay Area express lanes.

HOV lanes would encourage carpooling and bus use because the lanes provide a faster trip during peak periods in comparison to general-purpose lanes used by SOVs. Additionally, HOV lanes will not be tolled, so there is also a monetary savings for HOV users. Charging a toll on the general purpose-lane would encourage a mode shift from SOV trips to shared rides, or transit, and would result in a reduction in discretionary trips, and charging a higher toll for SOVs would discourage SOV trips. The combination of the designation of HOV lanes and the implementation of tolling on the general-purpose lanes lowers VMT. The MTC regional model used for travel forecasting (see Section 2.2.11) was also used to predict VMT with the proposed HOV lanes and tolling of the general purpose lanes in place, and the result was a reduction in overall VMT (summarized below in Table 3-4, from Table 2-14):

Alternative	Change in VMT with HOV Lanes and Tolling in Westbound Direction Only	Change in VMT with HOV Lanes and Tolling in both Eastbound and Westbound Directions
2025 Daily VMT Difference	-16,668	-11,166
2045 Daily VMT Difference	-83,340	-55,831

Table 3-4 VMT Reduction with the Preferred Alternative (HOV Lanes and Tolling)

Source: VMT values from MTC model, provided by Elite Transportation Group, Inc.

It is important to note that the reduction in VMT with HOV lanes and tolling shown in Table 3-4 is based on the MTC regional model that was applied for all trips in the ninecounty Bay Area. That modeling shows a net reduction in trips for the regional area. Some drivers might choose to use a route other than SR 37 when the HOV lanes and tolling are implemented (i.e., the driver might divert from SR 37 to an alternate route). The regional model was used to account for all trips, including potential diversion trips; a net reduction in trips likely results because of the price sensitivity of drivers when choosing available routes and deciding whether to take discretionary trips. Applying tolling without providing the HOV lane(s) would not effectively reduce the existing congestion where SR 37 drops from two lanes to one lane in each direction and would result in drivers diverting to longer routes to avoid both the congestion and tolls. Therefore, applying tolling alone without the additional travel lane was not considered an effective option to meet the purpose and need of the project, and was not an option that would be funded or constructed by the project sponsors. **Performance Measures.** Achieving a neutral or negative VMT when comparing No Build to the preferred Alternative 3B indicates that VMT is reduced and not increasing with implementation of the preferred alternative. Table 3-4 shows this comparison based on the traffic analysis performed for this project, which used the MTC regional forecast model. This modeling shows that a negative or reduced VMT is achieved in the study years of 2025 and 2045.

The project sponsors and/or Caltrans have committed to use the following measures or equivalent measures to evaluate traffic metrics in the corridor. These actions, or equivalent actions if others are identified in the future, would be applied over time to achieve the metrics indicated in the traffic forecast modeling for the 2025 and 2045 study years.

- Comparison of traffic volumes on SR 37 with one of more similar state highways in the Bay Area would provide a measure of whether traffic on SR 37 has changed disproportionately in comparison with another Bay Area tolled highway. The VMT between 2025 and 2045 modeled for the SR 37 project showed that VMT is expected to increase under No Build conditions, as a result of general growth in traffic in the Bay Area. The performance measure to be used here is a metric that would show whether a change in volume on SR 37 over time was disproportionately different over time than another similar highway, which could be one of the Bay Area bridges. Either annual traffic counts already collected by Caltrans or equivalent traffic counting methods could be used for this comparison.
- If a measured traffic volume increase over the evaluated time period is disproportionately greater for SR 37, this would indicate that VMT may have increased on SR 37 compared to a similar facility. This would justify a change in tolling rates or a change in the qualified users of the HOV lanes on SR 37, or other equivalent measures that would effectively reduce SR 37 traffic volumes (refer to Toll Rates and HOV Enforcement, below).
- Implementation of the proposed SR 37 tolling equipment, which records vehicles passing through the toll readers, would provide accurate counts of trip volumes and vehicle occupancy that can be used to monitor and report vehicle use on SR 37. VMT is projected to increase over time, regardless of the project; therefore, it would be reasonable for the project to mitigate the total VMT increase during the 2025 through 2045 time period that established the VMT modeled reductions used in this report.
- Caltrans and the project partners would establish the interval appropriate for evaluating traffic changes between SR 37 and the comparative highway, which could be conducted on an annual basis to account for seasonal changes to travel patterns.

An equity or means-based discount program is intended to be implemented, which would effectively provide a toll discount for eligible drivers. A means-based discount program could incrementally influence the decision to drive for a portion of SR 37 users (the portion of drivers eligible for a toll discount). The greater the subsidy provided would lower the VMT reduction estimated with tolling in place. Although the reduction in VMT
reported above with tolling would be incrementally less with a discount program, the margin in the VMT reduction estimated in Table 3-4 is still sufficient for this type of program to be implemented and still achieve the VMT reduction goals. This program would be established with oversight and approval by the toll authority.

Based on the travel forecast model and comparing the proposed No Build, preferred alternative with no tolling, and the preferred alternative with tolling, the preferred alternative with tolling and HOV lanes would effectively reduce existing congestion and reduce VMT to levels that are lower than the No Build Alternative. Alternatives 1, 2, 3A, and 3B with tolling would therefore have no net increase (no adverse impact) in VMT and would provide a beneficial reduction in VMT in comparison to the No Build Alternative.

**Toll Rates and HOV Lane Enforcement.** VMT is influenced by tolling rates. A higher toll rate results in a higher cost of driving, which helps influence decisions to carpool, take transit/bus service, or not to drive. If necessary, toll rates will be increased to reduce VMT in the proposed toll lane in either one or both directions. Any decision to change toll rates would be made by the tolling authority for SR 37, with input from Caltrans, MTC, and the project's agency partners, and would be based on evaluation of traffic volume changes using the performance measures described above, or equivalent measures.

To achieve reasonable compliance with the intended use of the lanes for multi-occupant vehicles, management of the HOV lanes will be required on SR 37. Having a tolled lane adjacent to an HOV lane will require enforcement or barriers to minimize HOV lane violations (a violation is the use of an HOV lane by a vehicle that does not qualify to be in the HOV lane based on vehicle occupancy). Section 1.4.1.3 in this Final EIR/EA describes the proposed project features that will be constructed as part of the project, and the highway operations that will be applied. These include the installation of CHP observational areas (where CHP officers can monitor traffic and pursue HOV violations), the requirement that HOV users carry and use a transponder accurately set to the vehicle occupancy, the installation of advance HOV lane signage above and adjacent to the HOV lanes, and the use of pavement markings and striping to identify the lanes.

Caltrans has established a goal to achieve a violation rate of 10 percent or less. Caltrans, the CHP, and the project's sponsoring partners will establish an agreement on a goal for an acceptable violation rate for SR 37, and funding or support for CHP enforcement.

#### VMT-1: Bus Service, Ride Sharing

**Bus Service:** There are no public bus service routes that use SR 37, partially because the existing peak period congestion results in significant delays that affect all vehicles using the route. Bus riders and other HOVs currently gain no time savings advantage or incentive over SOVs because all vehicles must use one lane between Mare Island and SR 121, with no option to pass other vehicles. The following describes the range of options that would take advantage of the proposed HOV lanes and help reduce VMT in the SR 37 corridor, as well as reducing regional trips that connect to the SR 37 corridor.

A transit feasibility study has been completed for the SR 37 corridor (Fehr & Peers 2019). This study identified express bus service as an appropriate and cost-effective route and travel option between Fairfield, Vallejo, and Novato (Figure 3-1). Potential

service stops these routes would serve, using SR 37, were the Fairfield Transit Center, Red Top Road Park-and-ride lot, a (future) Fairgrounds park-and-ride stop, Black Point park-and-ride, SMART Station, and the Novato Hamilton SMART Station. With HOV lanes, these routes would serve existing and future trips between Vallejo and Novato (approximately 25 miles) and Fairfield and Novato (approximately 40 miles). Bus service providers in this regional area include NVTA (which currently operates the Vine Transit system that connects to SR 37 in Vallejo and American Canyon east of the Napa River), Solano Express bus service (whose service area includes SR 37 and Vallejo), Marin Transit (whose service area includes the U.S. 101 corridor in Novato), and Golden Gate Transit (which includes service to Novato in the vicinity of SR 37).

There is an opportunity for bus transit service on SR 37 to connect to the existing SMART rail service at San Marin and Novato Hamilton stations, allowing riders to take advantage of longer trips that parallel the communities along the U.S. 101 corridor. The existing SMART rail service serves the Sonoma County Airport, Santa Rosa, Rohnert Park, Cotati, Petaluma, Novato, San Rafael, and Larkspur, connecting to the Larkspur ferry service. Bus service connecting to SMART would provide an option for use of bus and rail service rather than driving within the North Bay Counties, which would reduce VMT. Similarly, transit service on SR 37 can connect with other transit service providers, such as Golden Gate Transit, to provide transit alternatives for longer trips.

Providing new bus service on SR 37 would require action and funding by local transportation authorities and transit providers. The proposed SR 37 Sears Point to Mare Island Improvement Project is sponsored by MTC, NVTA, STA, and SCTA, who would commit to identifying and assisting in funding regional and local transit providers to expand bus service onto the SR 37 corridor.

**Ride Sharing:** The proposed HOV lanes provide an incentive for individuals to carpool and combine trips, which reduces VMT by reducing the trip length and/or number of SOVs using the SR 37 corridor. These factors include the higher cost of driving due to tolling, the time savings gained by being an HOV, and the savings in mileage and cost by not driving. These incentives are enhanced for some drivers if park-and-ride lots are available. Existing park-and-ride lots in the region and their use are illustrated in Figure 3-2. This figure also shows the estimated occupancy in terms of the average percentage of daily capacity being used. Lots that are over capacity or being used at a high percentage of their capacity indicate the potential opportunity for bus service to effectively serve this corridor and/or the need for more carpool lot availability. The current use of existing car pool lots in the vicinity of SR 37 would be expected to increase with tolling and the attraction of saving travel time with an HOV. The demand for bus and park-and-ride lots could be expected to increase with the proposed project's provision of one or more HOV lanes.

Figure 3-3 shows the potential opportunities for enhanced park-and-ride lot expansion that could help serve SR 37. Expansion of these lots would serve formal and casual carpool and vanpool riders and enhance opportunities for pickup and drop-off locations. This would be another option to help reduce VMT; and, as noted before, implementation of additional park-and-ride capacity serving the SR 37 corridor could be implemented over time by the project sponsors to serve the anticipated increase in demand between the 2025 and 2045 study years.



SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

#### FIGURE 3-1: SR 37 POTENTIAL EXPRESS BUS SERVICE **CONNECTING TRANSIT SERVICES**



SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

Miles

#### FIGURE 3-2: EXISTING PARK-AND-RIDE LOTS IN VICINITY OF SR 37



SR 37 Traffic **Congestion Relief Project** Sonoma, Napa & Solano Counties, CA

#### FIGURE 3-3: PARK-AND-RIDE OPPORTUNITIES IN VICINITY OF SR 37

This page intentionally left blank

Existing and emerging mobility services are also expected to enhance the ability of individuals to combine rides and reduce VMT. Examples of existing services include MTC's 511.org program, which connects riders and aids in finding transit services and setting up vanpools. More recent or emerging technologies, including web-based services, help identify ridesharing and transit opportunities. Park-and-ride lots can help provide a common location to meet and serve trip origins and destinations. These mobility applications are provided by agencies and private services and would be available with or without the proposed project, but the proposed HOV lanes in combination with park-and-ride lots would further enhance or promote the effectiveness of ridesharing using the SR 37 corridor.

In summary, there would be no increase in VMT in either the 2025 or 2045 study years with implementation of the preferred alternative with proposed HOV lanes and toll lanes. Tolling would require CTC approval in addition to the approval of the project. Without tolling, there are other strategies that could reduce VMT, including project sponsor assistance with implementing bus service routes along SR 37, additional park-and-ride availability, and ride sharing services. MTC supports rideshare and transit development along the corridor and would help fund these improvements.

Because VMT can increase over time, as illustrated in the differences in VMT between the 2025 and 2045 study years, the implementation of bus service, increased park-and-ride availability, or other viable options would be implemented over time as VMT increases and demand for these ridesharing services also increases. These are all viable options that without tolling would fully offset the increase in VMT and maintain the project as consistent with CEQA Guidelines Section 15064.3(b). Tolling, bus service, increased park and ride availability, and ride sharing services are not mutually exclusive and any or all of these measures would be used to meet the performance measure of maintaining or reducing VMT below the corresponding No Build VMT levels. The project is sponsored by regional and local transportation authorities who would commit to helping fund bus service and ride sharing facilities and programs that benefit SR 37 VMT reduction.

## c) Substantially increase hazards due to a geometric design feature (e.g., sharp curves or dangerous intersections) or incompatible uses (e.g., farm equipment)?

**No Impact.** Although Build Alternatives 2, 3A, and 3b would involve widening SR 37, the alignment of SR 37 would not change. SR 37 traffic must be maintained during construction, and construction staging areas would be needed along or near the route for equipment and materials. The project would not increase hazards due to design features, and there would be no impact.

#### d) Result in inadequate emergency access?

**Less than Significant Impact.** During construction, access would be maintained within the SR 37 corridor. A TMP would be developed to coordinate with emergency service providers during construction. There would be no impact related to emergency access. None of the Build Alternatives would impair implementation of an emergency response or emergency evacuation plan. During construction, implementation of the TMP would minimize construction-related delays and would include coordination with CHP and local law enforcement agencies. However, even with the implementation of the TMP, there may be slower traffic due to construction activities. While access would

be maintained for emergency response vehicles they may be required to move around traffic, which could result in slight delays.

During project operation, Build Alternatives 2 and 3A would result in the loss of a wider shoulder for emergency vehicle use, but the presence of two lanes in each direction on SR 37 would allow traffic to move over to the other lane, allowing emergency vehicles to pass them. Allowing traffic to move over may result in slight delays. Therefore, the Build Alternatives would have a less than significant impact.

#### 3.3.18 Tribal Cultural Resources

Would the project cause a substantial adverse change in the significance of a tribal cultural resource, defined in Public Resources Code section 21074 as either a site, feature, place, cultural landscape that is geographically defined in terms of the size and scope of the landscape, sacred place, or object with cultural value to a California Native American tribe, and that is:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Listed or eligible for listing in the California Register of Historical Resources, or in a local register of historical resources as defined in Public Resources Code section 5020.1(k), or	No	No	No	Yes
b) A resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of Public Resources Code Section 5024.1. In applying the criteria set forth in subdivision (c) of Public Resource Code Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe.	No	No	No	Yes

#### 3.3.18.1 CEQA Significance Determinations for Tribal Cultural Resources

a, b) Listed or eligible for listing in the California Register of Historical Resources, or in a local register of historical resources as defined in Public Resources Code section 5020.1(k), or a resource determined by the lead agency, in its discretion and supported by substantial evidence, to be significant pursuant to criteria set forth in subdivision (c) of Public Resources Code Section 5024.1. In applying the criteria set forth in subdivision (c) of Public Resource Code Section 5024.1, the lead agency shall consider the significance of the resource to a California Native American tribe.

**No Impact.** The NAHC was contacted on October 2019 to request a search of the Sacred Lands File for sacred lands or other cultural properties of significance to Native Americans in or near the APE. The NAHC replied via email on October 25, 2019, stating that a search of the file had been completed and was negative for cultural resources. Section 2.2.13 discusses the Native American Consultation performed for this project in more detail. As a result of consultation with the NAHC and local Native American tribes, no tribal cultural resources were identified in or near the APE. Therefore, there would be no impact.

#### 3.3.19 Utilities and Service Systems

Would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Require or result in the relocation or construction of new or expanded water, wastewater treatment or storm water drainage, electric power, natural gas, or telecommunications facilities, the construction or relocation of which could cause significant environmental effects?	No	No	Yes	No
b) Have sufficient water supplies available to serve the project and reasonably foreseeable future development during normal, dry, and multiple dry years?	No	No	No	Yes
c) Result in a determination by the wastewater treatment provider which serves or may serve the project that it has adequate capacity to serve the project's projected demand in addition to the provider's existing commitments?	No	No	No	Yes
d) Generate solid waste in excess of State or local standards, or in excess of the capacity of local infrastructure, or otherwise impair the attainment of solid waste reduction goals??	No	No	No	Yes
e) Comply with federal, state, and local management and reduction statutes and regulations related to solid waste?	No	No	No	Yes

#### 3.3.19.1 CEQA Significance Determinations for Utilities and Service Systems

# a) Require or result in the relocation or construction of new or expanded water, wastewater treatment or storm water drainage, electric power, natural gas, or telecommunications facilities, the construction or relocation of which could cause significant environmental effects?

**Less than Significant Impact.** The project would require relocation of some PG&E overhead electrical distribution lines. Seven wooden poles would be relocated due to the construction and widening of the roadway. The relocation of electrical facilities may result in temporary interruptions of service. Final verification of utilities would be performed during the project's detailed design phase, and any needed relocations would be coordinated with the affected utility owner to minimize potential interruptions of service. This impact would be less than significant.

### b) Have sufficient water supplies available to serve the project and reasonably foreseeable future development during normal, dry, and multiple dry years?

**No Impact.** The project would not include new development or uses that would require water supplies. There would be no impact.

#### c) Result in a determination by the wastewater treatment provider which serves or may serve the project that it has adequate capacity to serve the project's projected demand in addition to the provider's existing commitments?

**No Impact.** The project would not generate new wastewater flows or affect public utilities for wastewater treatment. There would be no impact.

## d) Generate solid waste in excess of State or local standards, or in excess of the capacity of local infrastructure, or otherwise impair the attainment of solid waste reduction goals?

No Impact. The project would not generate solid waste, other than during construction.

### e) Comply with federal, state, and local management and reduction statutes and regulations related to solid waste?

**No Impact.** The project would not result in the production of solid waste other than during construction. During construction, the project would not generate or require solid waste disposal that exceeds local standards, or exceeds the capacity of local infrastructure. Construction waste that could not be recycled would be disposed of at a certified facility based on the waste type and is not anticipated to affect landfill capacity. The project would also comply with all federal, state, and local statutes and regulations related to solid waste. There would be no impact.

#### 3.3.20 Wildfire

If located in or near state responsibility areas or lands classified as very high fire hazard severity zones, would the project:	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Substantially impair an adopted emergency response plan or emergency evacuation plan?	No	No	No	Yes
b) Due to slope, prevailing winds, and other factors, exacerbate wildfire risks, and thereby expose project occupants to, pollutant concentrations from a wildfire or the uncontrolled spread of a wildfire?	No	No	No	Yes
c) Require the installation or maintenance of associated infrastructure (such as roads, fuel breaks, emergency water sources, power lines or other utilities) that may exacerbate fire risk or that may result in temporary or ongoing impacts to the environment?	No	No	No	Yes
d) Expose people or structures to significant risks, including downslope or downstream flooding or landslides, as a result of runoff, post- fire slope instability, or drainage changes?	No	No	No	Yes

#### **3.3.20.1 CEQA Significance Determinations for Wildfire**

SB 1241 required the Office of Planning and Research, the Natural Resources Agency, and the California Department of Forestry and Fire Protection to develop

amendments to the "CEQA Checklist" for the inclusion of questions related to fire hazard impacts for projects on lands classified as very high fire hazard severity zones. The 2018 updates to the CEQA Guidelines expanded this to include projects "near" these very high fire hazard severity zones.

### a) Substantially impair an adopted emergency response plan or emergency evacuation plan?

**No Impact.** The project limits traverse Sonoma, Solano, and Napa Counties. These counties have Emergency Operation Plans, which provide guidelines for emergency response planning, preparation, training, and execution throughout their jurisdictions. None of the Build Alternatives would impair implementation of an emergency response or emergency evacuation plan. No potential evacuation routes would be impeded or disrupted during project construction and operation. During project construction, all traffic lanes on SR 37 would remain in operation. A TMP would be implemented to minimize construction-related delays. A substantial reduction in emergency response times is not expected. There would be no impact.

## b) Due to slope, prevailing winds, and other factors, exacerbate wildfire risks, and thereby expose project occupants to, pollutant concentrations from a wildfire or the uncontrolled spread of a wildfire?

**No Impact.** The eastern and central part of the project corridor (from the eastern terminus to Sonoma Creek) passes primarily through publicly owned marshland and agricultural lands. This part of the corridor is generally flat, with little to no slopes, and consists of sloughs, marsh vegetation, and small water channels. Portions of SR 37 in the local responsibility area for Sonoma County are in a moderate severity zone. A large portion of the alignment in the project area is in areas of Solano County that are unzoned.

The project would not change fire risk conditions and it would not change the alignment of SR 37. During construction, most work would occur in Caltrans' right-of-way; areas adjacent to SR 37 would be needed for widening the highway and construction staging. Project features for minimizing fire risks would be incorporated, such as clearing vegetation from the work area; prohibiting the use of highly flammable chemicals; following locally changing meteorological conditions; and maintaining awareness of the possibility of increased fire danger during the time work is in progress (see Table 1-4). All construction activities would follow state and federal fire regulations. The project is not expected to exacerbate wildfire risks or expose project personnel to pollutants from a wildfire or the uncontrolled spread of a wildfire. There would be no impact.

# c) Require the installation or maintenance of associated infrastructure (such as roads, fuel breaks, emergency water sources, power lines or other utilities) that may exacerbate fire risk or that may result in temporary or ongoing impacts to the environment?

**No Impact.** The project involves widening SR 37 and would require relocation of some PG&E overhead electrical distribution lines. Seven wooden poles would be relocated due to the construction and widening of the roadway in certain sections. All project construction would follow state and federal fire regulations during these relocations. Project features for minimizing fire risks would be incorporated, such as clearing vegetation from the work area; prohibiting the use of highly flammable chemicals;

following locally changing meteorological conditions; and maintaining awareness of the possibility of increased fire danger during the time work is in progress. Therefore, the project is not expected to exacerbate wildfire risks or expose project personnel to pollutants from a wildfire or the uncontrolled spread of a wildfire, and there would be no impact.

## d) Expose people or structures to significant risks, including downslope or downstream flooding or landslides, as a result of runoff, post-fire slope instability, or drainage changes?

**No Impact.** No recent fires have occurred in the project vicinity that could result in post-fire slope instability or drainage changes. Furthermore, the project limits are in a relatively flat area with little to no slopes and extensive wetland vegetation that remains green year round. Implementation of standard Caltrans practices for erosion control and other measures would avoid or minimize the project's potential to result in downslope or downstream flooding or landslides. These measures are incorporated into the project design as a matter of Caltrans practice and are not mitigation. The proposed project would not expose the public to a risk of post-fire slope instability or drainage changes. No impact would occur.

	Significant and Unavoidable Impact	Less than Significant with Mitigation Incorporated	Less than Significant Impact	No Impact
a) Does the project have the potential to substantially degrade the quality of the environment, substantially reduce the habitat of a fish or wildlife species, cause a fish or wildlife population to drop below self-sustaining levels, threaten to eliminate a plant or animal community, substantially reduce the number or restrict the range of a rare or endangered plant or animal or eliminate important examples of the major periods of California history or prehistory?	No	No	Yes	No
b) Does the project have impacts that are individually limited, but cumulatively considerable? ("Cumulatively considerable" means that the incremental effects of a project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects)?	No	No	No	Yes
c) Does the project have environmental effects which will cause substantial adverse effects on human beings, either directly or indirectly?	No	No	No	Yes

#### 3.3.21 Mandatory Findings of Significance

#### 3.3.21.1 CEQA Significance Determinations for Mandatory Findings of Significance

a) Does the project have the potential to substantially degrade the quality of the environment, substantially reduce the habitat of a fish or wildlife species, cause a fish or wildlife population to drop below self-sustaining levels, threaten to eliminate a plant or animal community, substantially reduce the number or restrict the range of a rare or endangered plant or animal or eliminate important examples of the major periods of California history or prehistory?

Less than Significant Impact. Though the proposed project would have potential impacts on special-status species and their habitats, impacts would not substantially reduce habitat or wildlife at a population level. Additionally, the project would not eliminate a plant or animal community, or substantially reduce the number or range of any rare or endangered plant or animal. The project would not eliminate any examples of major periods on California history or prehistory. Because the project would have some impacts on special-status species and their habitat in the project area that would be less than substantial at population or community levels, it would have a less than significant impact.

b) Does the project have impacts that are individually limited, but cumulatively considerable? ("Cumulatively considerable" means that the incremental effects of a project are considerable when viewed in connection with the effects of past projects, the effects of other current projects, and the effects of probable future projects)?

Less than Significant Impact with Mitigation Incorporated. The build alternatives without tolling result in an increase in VMT, and measures to reduce VMT below the No Build alternative levels are described in Section 3.3.17, Transportation. One or all of the measures described in that section can be implemented to meet the performance measure of maintaining VMT at the No Build level. If other projects or factors contribute or change VMT, the same strategy of using tolling, bus service, and ride sharing programs can still be used to offset VMT. Additional measures may also be applied, such as changing the toll rate at peak periods. These measures would be effective with respect to a cumulative impact change in VMT, and the impact would be less than significant.

## c) Does the project have environmental effects which will cause substantial adverse effects on human beings, either directly or indirectly?

**No Impact.** The proposed project would include construction impacts that could affect human beings (e.g., construction noise and traffic delays), but these impacts would be short term and not substantially adverse. Therefore, there would be no impact.

#### 3.4 Climate Change

Climate change refers to long-term changes in temperature, precipitation, wind patterns, and other elements of the earth's climate system. An ever-increasing body of scientific research attributes these climatological changes to GHG emissions, particularly those generated from the production and use of fossil fuels.

Although climate change has been a concern for several decades, the establishment of the Intergovernmental Panel on Climate Change (IPCC) by the United Nations and World Meteorological Organization in 1988 led to increased efforts devoted to GHG emissions reduction and climate change research and policy. These efforts are primarily concerned with the emissions of GHGs generated by human activity, including CO<sub>2</sub>, methane (CH<sub>4</sub>), nitrous oxide (N<sub>2</sub>O), tetrafluoromethane, hexafluoroethane, sulfur hexafluoride (SF<sub>6</sub>), and various hydrofluorocarbons (HFCs). CO<sub>2</sub> is the most abundant GHG; while it is a naturally occurring component of Earth's atmosphere, fossil-fuel combustion is the main source of additional, human-generated CO<sub>2</sub>.

Two terms are typically used when discussing how we address the impacts of climate change: "greenhouse gas mitigation" and "adaptation." Greenhouse gas mitigation covers the activities and policies aimed at reducing GHG emissions to limit or "mitigate" the impacts of climate change. Adaptation, on the other hand, is concerned with planning for and responding to impacts resulting from climate change (such as adjusting transportation design standards to withstand more intense storms and higher sea levels). This analysis would include a discussion of both.

#### 3.4.1 Regulatory Setting

This section outlines federal and state efforts to comprehensively reduce GHG emissions from transportation sources.

#### 3.4.1.1 Federal

To date, no national standards have been established for nationwide mobile-source GHG reduction targets, nor have any regulations or legislation been enacted specifically to address climate change and GHG emissions reduction at the project level.

NEPA (42 USC Part 4332) requires federal agencies to assess the environmental effects of their proposed actions prior to deciding on an action or project.

FHWA recognizes the threats that extreme weather, sea-level change, and other changes in environmental conditions pose to valuable transportation infrastructure and those who depend on it. FHWA supports a sustainability approach that assesses vulnerability to climate risks and incorporates resilience into planning, asset management, project development and design, and operations and maintenance practices (FHWA 2019). This approach encourages planning for sustainable highways by addressing climate risks while balancing environmental, economic, and social

values—"the triple bottom line of sustainability" (FHWA n.d.). Program and project elements that foster sustainability and resilience also support economic vitality and global efficiency, increase safety and mobility, enhance the environment, promote energy conservation, and improve the quality of life.

Various efforts have been promulgated at the federal level to improve fuel economy and energy efficiency to address climate change and its associated effects. The most important of these was the Energy Policy and Conservation Act of 1975 (42 USC Section 6201) and Corporate Average Fuel Economy (CAFE) Standards. This act establishes fuel economy standards for on-road motor vehicles sold in the United States. Compliance with federal fuel economy standards is determined through the CAFE program based on each manufacturer's average fuel economy for the portion of its vehicles produced for sale in the United States.

The Energy Policy Act of 2005, 109th Congress H.R.6 (2005–2006): This act sets forth an energy research and development program covering: (1) energy efficiency; (2) renewable energy; (3) oil and gas; (4) coal; (5) the establishment of the Office of Indian Energy Policy and Programs within the Department of Energy; (6) nuclear matters and security; (7) vehicles and motor fuels, including ethanol; (8) hydrogen; (9) electricity; (10) energy tax incentives; (11) hydropower and geothermal energy; and (12) climate change technology.

The U.S. EPA in conjunction with the National Highway Traffic Safety Administration (NHTSA) is responsible for setting GHG emission standards for new cars and lightduty vehicles to significantly increase the fuel economy of all new passenger cars and light trucks sold in the United States. Fuel efficiency standards directly influence GHG emissions.

#### 3.4.1.2 State

California has been innovative and proactive in addressing GHG emissions and climate change by passing multiple Senate and Assembly bills and EOs including, but not limited to, the following:

EO S-3-05 (June 1, 2005): The goal of this EO is to reduce California's GHG emissions to: (1) year 2000 levels by 2010, (2) year 1990 levels by 2020, and (3) 80 percent below year 1990 levels by 2050. This goal was further reinforced with the passage of AB 32 in 2006 and SB 32 in 2016.

AB 32, Chapter 488, 2006, Núñez and Pavley, The Global Warming Solutions Act of 2006: AB 32 codified the 2020 GHG emissions reduction goals outlined in EO S-3-05, while further mandating that CARB create a scoping plan and implement rules to achieve "real, quantifiable, cost-effective reductions of greenhouse gases." The Legislature also intended that the statewide GHG emissions limit continue in existence and be used to maintain and continue reductions in emissions of GHGs beyond 2020 (H&SC Section 38551(b)). The law requires CARB to adopt rules and regulations in an open public process to achieve the maximum technologically feasible and cost-effective GHG reductions.

3-57

EO S-01-07 (January 18, 2007): This order sets forth the low carbon fuel standard (LCFS) for California. Under this EO, the carbon intensity of California's transportation fuels is to be reduced by at least 10 percent by the year 2020. CARB re-adopted the LCFS regulation in September 2015, and the changes went into effect on January 1, 2016. The program establishes a strong framework to promote the low-carbon fuel adoption necessary to achieve the Governor's 2030 and 2050 GHG reduction goals.

SB 375, Chapter 728, 2008, Sustainable Communities and Climate Protection: This bill requires CARB to set regional emissions reduction targets for passenger vehicles. The MPO for each region must then develop a "Sustainable Communities Strategy" (SCS) that integrates transportation, land-use, and housing policies to plan how it would achieve the emissions target for its region.

SB 391, Chapter 585, 2009, California Transportation Plan: This bill requires the State's long-range transportation plan to identify strategies to address California's climate change goals under AB 32.

EO B-16-12 (March 2012) orders State entities under the direction of the Governor, including CARB, the CEC, and the Public Utilities Commission, to support the rapid commercialization of zero-emission vehicles. It directs these entities to achieve various benchmarks related to zero-emission vehicles.

EO B-30-15 (April 2015) establishes an interim statewide GHG emission reduction target of 40 percent below 1990 levels by 2030 to ensure California meets its target of reducing GHG emissions to 80 percent below 1990 levels by 2050. It further orders all state agencies with jurisdiction over sources of GHG emissions to implement measures, pursuant to statutory authority, to achieve reductions of GHG emissions to meet the 2030 and 2050 GHG emissions reductions targets. It also directs CARB to update the Climate Change Scoping Plan to express the 2030 target in terms of million metric tons of carbon dioxide equivalent (MMTCO<sub>2</sub>e).<sup>2</sup> Finally, it requires the Natural Resources Agency to update the state's climate adaptation strategy, Safeguarding California, every 3 years, and to ensure that its provisions are fully implemented.

SB 32, Chapter 249, 2016, codifies the GHG reduction targets established in EO B-30-15 to achieve a mid-range goal of 40 percent below 1990 levels by 2030.

SB 1386, Chapter 545, 2016, declared "it to be the policy of the state that the protection and management of natural and working lands ... is an important strategy in meeting the state's GHG reduction goals, and would require all state agencies, departments, boards, and commissions to consider this policy when revising, adopting, or establishing policies, regulations, expenditures, or grant criteria relating to the protection and management of natural and working lands."

<sup>&</sup>lt;sup>2</sup> GHGs differ in how much heat each trap in the atmosphere (global warming potential). CO<sub>2</sub> is the most important GHG, so amounts of other gases are expressed relative to CO<sub>2</sub>, using a metric called "carbon dioxide equivalent" (CO<sub>2</sub>e). The global warming potential of CO<sub>2</sub> is assigned a value of 1, and the global warming potential of other gases is assessed as multiples of CO<sub>2</sub>.

AB 134, Chapter 254, 2017, allocates GHG Reduction Funds and other sources to various clean vehicle programs, demonstration/pilot projects, clean vehicle rebates and projects, and other emissions-reduction programs statewide.

SB 743, Chapter 386 (September 2013): This bill changes the metric of consideration for transportation impacts pursuant to CEQA from a focus on automobile delay to alternative methods focused on VMT, to promote the state's goals of reducing GHG emissions and traffic related air pollution and promoting multimodal transportation while balancing the needs of congestion management and safety.

SB 150, Chapter 150, 2017, RTPs: This bill requires CARB to prepare a report that assesses progress made by each MPO in meeting their established regional GHG emission reduction targets.

EO B-55-18 (September 2018) sets a new statewide goal to achieve and maintain carbon neutrality no later than 2045. This goal is in addition to existing statewide targets of reducing GHG emissions.

EO N-19-19 (September 2019) advances California's climate goals in part by directing the California State Transportation Agency to leverage annual transportation spending to reverse the trend of increased fuel consumption and reduce GHG emissions from the transportation sector. It orders a focus on transportation investments near housing, managing congestion, and encouraging alternatives to driving. This EO also directs CARB to encourage automakers to produce more clean vehicles, formulate ways to help Californians purchase them, and propose strategies to increase demand for zero-emission vehicles.

#### 3.4.2 Environmental Setting

SR 37 is the main transportation route that connects Solano, Napa, Sonoma, and Marin Counties. Most of the proposed project is in a rural area, which is adjacent to open-space, agricultural, and recreational uses. The project is also adjacent to the Refuge and the NSMWA. The portion of the project alignment in Vallejo is near residential and mixed-use development. Currently, commuters experience significant recurring traffic congestion and delays at the bottlenecks within the traffic study limits during the peak hours. Traffic demands exceed capacity in segments between SR 121 and Mare Island, where the existing two lanes merge into one lane in both the eastbound and westbound directions. Plan Bay Area 2050 guides transportation development in the project area. The BAAQMD's 2017 clean air plan, *Spare the Air, Cool the Climate,* addresses GHGs in the project region.

A GHG emissions inventory estimates the amount of GHGs discharged into the atmosphere by specific sources over a period of time, such as a calendar year. Tracking annual GHG emissions allows countries, states, and smaller jurisdictions to understand how emissions are changing and what actions may be needed to attain emission reduction goals. U.S. EPA is responsible for documenting GHG emissions nationwide, and the CARB does so for the state, as required by H&SC Section 39607.4.

#### **National GHG Inventory**

The U.S. EPA prepares a national GHG inventory every year and submits it to the United Nations in accordance with the Framework Convention on Climate Change. The inventory provides a comprehensive accounting of all human-produced sources of GHGs in the United States, reporting emissions of CO<sub>2</sub>, CH<sub>4</sub>, N<sub>2</sub>O, HFCs, perfluorocarbons, SF<sub>6</sub>, and nitrogen trifluoride. It also accounts for emissions of CO<sub>2</sub> that are removed from the atmosphere by "sinks" such as forests, vegetation, and soils that uptake and store CO<sub>2</sub> (carbon sequestration).

The 1990-2019 inventory found that overall GHG emissions were 6,558 million metric tons in 2019, down 1.7 percent from 2018 but up 1.8 percent from 1990 levels. Of these, 80 percent were  $CO_2$ , 10 percent were  $CH_4$ , and 7 percent were  $N_2O$ ; the balance consisted of fluorinated gases.  $CO_2$  emissions in 2019 were 2.2 percent less than in 2018, but 2.8 percent more than in 1990. As shown on Figure 3-4, the transportation sector accounted for 29 percent of GHG emissions in the United States in 2019 (U.S. EPA 2021a, 2021b).



Figure 3-4 U.S. 2019 Greenhouse Gas Emissions

Source: U.S. EPA 2021c

#### State GHG Inventory

CARB collects GHG emissions data for transportation, electricity, commercial/ residential, industrial, agricultural, and waste management sectors each year. It then summarizes and highlights major annual changes and trends to demonstrate the state's progress in meeting its GHG reduction goals. The 2020 edition of the GHG emissions inventory reported emissions trends from 2000 to 2018. It found total California emissions were 425.3 MMTCO<sub>2</sub>e in 2018, 0.8 MMTCO<sub>2</sub>e higher than 2017 but 6 MMTCO<sub>2</sub>e lower than the statewide 2020 limit of 431 MMTCO<sub>2</sub>e. The transportation sector was responsible for 41 percent of total GHGs. Transportation emissions decreased in 2018 compared to the previous year, which is the first yearover-year decrease since 2013. Overall statewide GHG emissions declined from 2000 to 2018 despite growth in population and state economic output (CARB 2020a).



Figure 3-5 California 2018 Greenhouse Gas Emissions by Economic Sector

Source: CARB 2020b



Figure 3-6 Change in California GDP, Population, and GHG Emissions Since 2000

Source: CARB 2020b

AB 32 required CARB to develop a Scoping Plan that describes the approach California would take to achieve the goal of reducing GHG emissions to 1990 levels by 2020, and to update it every 5 years. CARB adopted the first scoping plan in 2008. The second updated plan, California's 2017 Climate Change Scoping Plan, adopted on December 14, 2017, reflects the 2030 target established in EO B-30-15 and SB 32. The AB 32 Scoping Plan and the subsequent updates contain the main strategies California would use to reduce GHG emissions.

#### **Regional Plans**

CARB sets regional targets for California's 18 MPOs to use in their RTP/SCS to plan future projects that would cumulatively achieve GHG reduction goals. Targets are set at a percent reduction of passenger vehicle GHG emissions per person from 2005 levels. MTC is the MPO and regional transportation planning agency for the project region, with GHG reduction targets of 10 percent by 2020 and 19 percent by 2035. The proposed project would be included in the MTC RTP, Plan Bay Area 2050.

The 2017 clean air plan, Spare the Air, Cool the Climate (BAAQMD 2017), defines strategies for climate protection in the Bay Area that support goals laid out in Plan Bay Area 2050 (ABAG and MTC 2021a). Those goals include transforming the transportation sector to reduce motor vehicle travel, promote zero-emissions vehicles and renewable fuels, adopt fixed- and flexible-route transit services, and support infrastructure and planning that enable a large share of trips by bicycling, walking, and transit. Local climate action plans also offer GHG reduction strategies.

These plans are summarized in Table 3-5. The plans include goals, policies, and strategies for reducing GHG emissions. These strategies are current and future actions that have been or would be implemented and correspond to the proposed project.

Title	GHG Reduction Policies or Strategies
Plan Bay Area 2050	Transportation Demand Management (TDM) strategies, including bicycle and pedestrian networks, transit, targeted transportation alternatives, trip caps, car sharing, carpool and vanpool incentives, and commuter benefits ordinances. Incorporation of regional Climate Action Program that, includes the following:
	<ul> <li>Commuter Benefits Program — use of tax incentives to encourage more commuters to walk, bicycle, take transit, carpool, and vanpool to and from work</li> </ul>
	<ul> <li>Car Sharing — expanding car sharing to more communities and exploring all service models, including round trip, one-way, and peer-to-peer trips; MTC developed a Bay Area Carsharing Implementation Strategy to guide actions over the next few years</li> </ul>
	<ul> <li>Targeted Transportation Alternatives — using campaigns and encouragement programs to change individual travel behavior from driving alone to using sustainable modes, such as walking, biking, riding transit, carpooling, vanpooling and car sharing, for all types of trips</li> </ul>
County of Solano	Commuter and rideshare incentives
Climate Action Plan	Solano Bicycle and Pedestrian Program
	County Fleet Fuel Efficiencies
County of Napa Climate Action Plan	Reduce vehicle trips through consolidation of vehicle trips and nonmotorized trips
	Encourage the use of electric and alternative fuel vehicles
	Reduce VMT through smarter land use planning

Table 3-5 Regional and Local Greenhouse Gas Reduction Plans

#### 3.4.3 Project Analysis

GHG emissions from transportation projects can be divided into those produced during operation of the SHS and those produced during construction. The primary GHGs produced by the transportation sector are CO<sub>2</sub>, CH<sub>4</sub>, N<sub>2</sub>O, and HFCs. CO<sub>2</sub> emissions are a product of the combustion of petroleum-based products, like gasoline, in internal combustion engines. Relatively small amounts of CH<sub>4</sub> and N<sub>2</sub>O are emitted during fuel combustion. In addition, a small amount of HFC emissions are included in the transportation sector.

The CEQA Guidelines generally address GHG emissions as a cumulative impact due to the global nature of climate change (Pub. Resources Code, § 21083(b)(2)). As the California Supreme Court explained, "because of the global scale of climate change, any one project's contribution is unlikely to be significant by itself." (Cleveland National Forest Foundation v. San Diego Assn. of Governments (2017) 3 Cal. 5th 497, 512.) In

assessing cumulative impacts, it must be determined if a project's incremental effect is "cumulatively considerable" (CEQA Guidelines Sections 15064(h)(1) and 15130).

To make this determination, the incremental impacts of the project must be compared with the effects of past, current, and probable future projects. Although climate change is ultimately a cumulative impact, not every individual project that emits GHGs must necessarily be found to contribute to a significant cumulative impact on the environment.

#### 3.4.3.1 Operational Emissions

Nearly 29 percent of GHG emissions in the United States in 2019 came from the transportation sector. CO<sub>2</sub> emissions from fossil fuel combustion accounted for 74.1 percent of all GHG emissions, and transportation activities accounted for about 37.5 percent of CO<sub>2</sub> emissions from fossil fuel combustion in 2019. Most transportation-related GHG emissions are from passenger cars (40.5 percent), freight trucks (23.6 percent), and light-duty trucks (17.2 percent). The remainder of GHG emissions comes from other modes of transportation, including aircraft, ships, boats, and trains, as well as pipelines and lubricants (U.S. EPA 2021a, 2021b). Because CO<sub>2</sub> emissions represent the greatest percentage of GHG emissions it has been selected as a proxy within the following analysis for potential climate change impacts generally expected to occur.

The highest levels of  $CO_2$  from mobile sources such as automobiles occur at stop-andgo speeds (0 to 25 mph) and speeds over 55 mph; the most severe emissions occur from 0 to 25 mph (see Figure 3-7). To the extent that a project relieves congestion by enhancing operations and improving travel times in high-congestion travel corridors, GHG emissions, particularly  $CO_2$ , may be reduced.

Four primary strategies can reduce GHG emissions from transportation sources: (1) improving the transportation system and operational efficiencies, (2) reducing travel activity, (3) transitioning to lower GHG-emitting fuels, and (4) improving vehicle technologies/efficiency. To be most effective, all four strategies should be pursued concurrently.



Figure 3-7 Possible Use of Traffic Operation Strategies in Reducing On-road CO<sub>2</sub> Emissions

(Source: Barth and Boriboonsomsin 2010)

Four primary strategies can reduce GHG emissions from transportation sources: (1) improving the transportation system and operational efficiencies, (2) reducing travel activity, (3) transitioning to lower GHG-emitting fuels, and (4) improving vehicle technologies/efficiency. To be most effective, all four strategies should be pursued concurrently.

The RTP (Plan Bay Area 2050) contains regional strategies for reducing GHG emissions from transportation sources on a regional scale. This project is included in the RTP. *Plan Bay Area 2050* outlines measures to reduce per capita VMT, including but not limited to TDM strategies such as transit, targeted transportation alternatives, car sharing, carpool and vanpool incentives, and commuter benefits ordinances (ABAG and MTC 2021a). Specifically, *Plan Bay Area 2050* incorporated MTC's Climate Initiatives Program, which is designed to reduce GHG emissions from the transportation sector with strategies that include financial tools such as tax incentives to encourage more commuters to take transit, carpool, and vanpool; vanpooling setup support; and campaigns and programs to promote using sustainable travel modes, such as riding transit, carpooling, vanpooling and car sharing. The Counties of Solano and Napa have Climate Action Plans with similar policies and strategies aimed at increasing transit ridership, implementing TDM, and promoting alternatives to solo driving such as ridesharing.

HOV lanes are intended to facilitate transit and increase vehicle occupancy by encouraging ridesharing such as carpools and vanpools.

#### **Quantitative Analysis**

GHG emissions impacts for the No Build and Build Alternatives were computed using CT-EMFAC 2017 for the existing year and future years (2025 Opening Year, 2040 RTP, and 2045 Design Year). The SAFE vehicle emissions adjustment factors developed by CARB for CO<sub>2</sub> were applied to the CT-EMFAC2017 emissions factors, and the adjusted emissions factors were applied to the project area VMT estimates.<sup>3</sup> Table 3-6 shows the estimated CO<sub>2</sub>e emissions and VMT.

Under all Build Alternatives in all study years, the mobile GHG emissions in the region would decrease from baseline levels due to improvements in vehicle technology with or without the project. Modeling shows that Build Alternatives 1 and 2 would have lower annual GHG emissions than the No Build Alternative for all future years. Alternatives 3A and 3B would have 1,187 metric tons of CO<sub>2</sub>e less than the No Build Alternative. In 2040 and 2045 the GHG emissions would be higher than the No Build by 22,657 metric tons and 58,421 metric tons of CO<sub>2</sub>e per year, respectively. These results are without tolling applied. Because the CT-EMFAC2017 model is insensitive to a vehicle's modal events, such as acceleration and deceleration due to traffic congestion, it does not adequately capture CO<sub>2</sub> reductions associated with smoother traffic flow under the build alternatives. As described in Section 3.3.17, the project would improve traffic conditions on SR 37 and intersections. Reducing queues and stop-and-go traffic would reduce idling and GHG emissions due to idling. Factoring these benefits in could result in additional GHG emission reduction benefits compared to the No Build Alternative. Tolling and other proposed VMT reduction measures would also reduce GHG emissions.

Although CT-EMFAC has a rigorous scientific foundation and has been vetted through multiple stakeholder reviews, its GHG emission rates are based on tailpipe emission test data. The model does not account for factors such as the rate of acceleration and vehicle aerodynamics, which influence the amount of emissions generated by a vehicle. GHG emissions quantified using CT-EMFAC are therefore estimates and may not reflect actual physical emissions. Though CT-EMFAC is currently the best available tool for calculating GHG emissions from mobile sources, it is important to note that the GHG results are only useful for a comparison among alternatives.

<sup>&</sup>lt;sup>3</sup> The U.S. National Highway Traffic Safety Administration and Environmental Protection Agency SAFE (Safer Affordable Fuel-Efficient) Vehicles Rule Part One, revoking California's authority to set its own GHG emissions standards, was published on September 27, 2019, and was effective November 26, 2019. The SAFE Vehicles Rule Part Two became effective June 30, 2020. It amended existing CAFE and tailpipe carbon dioxide emissions standards for passenger cars and light trucks and established new standards covering model years 2021 through 2026. The rule retains the model year 2020 standards for both programs through model year 2026. ARB has provided adjustment factors for GHG emissions based on the SAFE Rule, and modeling these estimates with EMFAC2017 or CT-EMFAC2017 remains the most precise means of estimating future GHG emissions.

Alternative	GHG Emissions (MT CO₂e/ year)	Difference Between No Build and Build (MT CO <sub>2</sub> e/ year)	Change Between Existing and Build (MT CO₂e/year)	Daily VMT	Daily Difference in VMT Build and No Build	Annual VMT <sup>1</sup>	Annual Difference in VMT Build and No Build
Baseline (Existing) 2020 Conditions	24,555,199	NA	NA	149,948,925	NA	54,768,844,687	NA
No Build 2025	21,996,764	NA	NA	156,255,326	NA	57,072,258,000	NA
Build Alternatives 1 and 2 2025	21,995,070	-1,694	-2,560,129	156,261,672	6,346	57,074,575,829	2,317,829
Build Alternatives 3A and 3B 2025	21,995,577	-1,187	-2,559,622	156,264,925	9,599	57,075,763,812	3,505,812
With Tolling 2025 westbound only (Tolling Option 1)	21,993,294	-3,470	-2,561,905	156,238,658	-16,668	57,066,170,012	-6,087,987
With Tolling 2025 both Directions (Tolling Option 3)	21,996,633	-132	-2,558,567	156,244,160	-11,166	57,068,179,529	- 4,078,471
No Build 2040	21,394,504	NA	NA	175,174,532	NA	63,982,497,938	NA
Build Alternatives 1 and 2 2040	21,387,079	-7,425	-3,168,120	175,199,916	25,384	63,991,769,254	9,271,316
Build Alternatives 3A and 3B 2040	21,417,161	22,657	-3,138,039	175,212,926	38,394	63,996,521,187	14,023,249
With Tolling 2040 westbound only (Tolling Option 1)	21,387,803	-6,701	-3,167,397	175,107,860	- 66,672	63,958,145,988	-24,351,950
With Tolling 2040 both Directions (Tolling Option 3)	21,389,387	-5,117	-3,165,812	175,129,867	- 44,665	63,966,184,056	-16,313,882
No Build 2045	21,628,584	NA	NA	181,480,934	NA	66,285,911,251	NA
Build Alternatives 1 and 2 2045	21,616,583	-12,001	-2,938,616	181,512,664	31,730	66,297,500,396	11,589,145
Build Alternatives 3A and 3B 2045	21,687,005	58,421	-2,868,195	181,528,926	47,992	66,303,440,312	17,529,061
With Tolling 2045 westbound only (Tolling Option 1)	21,679,909	51,325	-2,875,290	181,397,594	-83,340	66,255,471,314	-30,439,937
With Tolling 2045 both Directions (Tolling Option 3)	21,684,863	56,279	-2,870,336	181,425,103	-55,831	66,265,518,898	-20,392,353

#### Table 3-6 Modeled Annual CO<sub>2</sub>e Emissions and VMT by Alternative

Notes:

Source: Illingworth & Rodkin using CT-EMFAC 2017 version 1.0.2, 2021.

 Annual VMT values from MTC model, provided by Elite Transportation Group, Inc., March 8, 2021, with updated tolling options September 2022.
 <sup>2</sup> Tolling in one direction (westbound only) is also referred to as Tolling Option 1. Tolling in both directions is also referred to as Tolling Option 3. These options are estimated for Alternatives 3A/3B.

#### 3.4.3.2 Construction Emissions

Construction GHG emissions would result from material processing, onsite construction equipment, and traffic delays due to construction. These emissions would be produced at different levels throughout the construction phase; their frequency and occurrence can be reduced by implementing better traffic management during construction.

In addition, with innovations such as longer pavement lives, improved traffic management plans, and changes in materials, the GHG emissions produced during construction can be offset to some degree by longer intervals between maintenance and rehabilitation activities.

Construction emissions were estimated using the latest Sacramento Metropolitan Air Quality Management District's RCEM, version 9.0. Table 2-22 in Section 2.3.6 shows the construction emissions associated with the project. As shown in Table 2-22, CO<sub>2</sub>e emissions from earthmoving activities and use of equipment and vehicles would amount to 5,994 metric tons. These emissions would be similar for all Build Alternatives except Alternative 3B, which involves widening of Sonoma Creek Bridge. Widening of Sonoma Creek Bridge would require additional structures (steel, concrete, and other materials) and construction staging, all of which would result in additional GHG emissions for that alternative.

All construction contracts include Caltrans Standard Specifications Section 7-1.02A and 7-1.02C, Emissions Reduction, which require contractors to comply with all laws applicable to the project and to certify they are aware of and would comply with all CARB emission reduction regulations; and Section 14-9.02, Air Pollution Control, which requires contractors to comply with all air pollution control rules, regulations, ordinances, and statutes. Certain common regulations, such as equipment idling restrictions, that reduce construction vehicle emissions also help reduce GHG emissions.

#### 3.4.3.3 CEQA Conclusion

CEQA Guidelines Section 15064.4 states that, among other factors, a lead agency should consider the extent to which the project may increase or reduce GHG emissions compared to the existing environmental setting when assessing the significance of impacts from GHG emissions on the environment. Although the comparison of future Build to future No Build conditions may be useful in aiding the analysis of significance and in determining the extent of project-level measures to reduce GHG emissions due to the project, CEQA and the CEQA Guidelines remain focused on the comparison of future conditions with the project compared to existing conditions.

Although individual projects are not required to meet the 2050 reduction targets established in EOs and legislation, current professional CEQA practices and important court cases<sup>4</sup> in 2014 and 2015 advocate for demonstrating continued progress toward assisting the state in achieving these goals.

The project would result in a reduction of GHG emissions by 2045 compared to the existing conditions, as shown in Table 3-6. Because the Build Alternatives would not contribute to increases in GHG emissions over existing conditions, and GHG-reduction measures would be implemented during construction, the impact would be less than significant.

#### 3.4.4 Greenhouse Gas Reduction Strategies

#### 3.4.4.1 Statewide Efforts

Major sectors of California's economy, including transportation, would need to reduce emissions to meet the 2030 and 2050 GHG emissions targets. Former Governor Edmund G. Brown promoted GHG reduction goals that involved (1) reducing today's petroleum use in cars and trucks by up to 50 percent; (2) increasing from one-third to 50 percent our electricity derived from renewable sources; (3) doubling the energy efficiency savings achieved at existing buildings and making heating fuels cleaner; (4) reducing the release of methane, black carbon, and other short-lived climate pollutants; (5) managing farms and rangelands, forests, and wetlands so they can store carbon; and (6) periodically updating the state's climate adaptation strategy, Safeguarding California.

The transportation sector is integral to the people and economy of California. To achieve GHG emission reduction goals, it is vital that the state build on past successes in reducing criteria and toxic air pollutants from transportation and goods movement (Figure 3-8). GHG emission reductions would come from cleaner vehicle technologies, lower-carbon fuels, and reduction of VMT. A key state goal for reducing GHG emissions is to reduce today's petroleum use in cars and trucks by up to 40 percent by 2030 (California Environmental Protection Agency 2015).

<sup>&</sup>lt;sup>4</sup> Center for Biological Diversity v. California Department of Fish and Wildlife and Newhall Land and Farming (2015) 224 Cal.App.4th 1105 (CBD vs. CDFW; also known as the "Newhall Ranch" case; Cleveland National Forest Foundation v. San Diego Association of Governments, 180 Cal.App.3d 548 (Cal.App. 2014)



Figure 3-8 California Climate Strategy

In addition, SB 1386 established as state policy the protection and management of natural and working lands and requires state agencies to consider that policy in their own decision making. Trees and vegetation on forests, rangelands, farms, and wetlands remove CO<sub>2</sub> from the atmosphere through biological processes and sequester the carbon in above- and below-ground matter. Subsequently, Governor Gavin Newsom issued EO N-82-20 to combat the crises in climate change and biodiversity. It instructs state agencies to use existing authorities and resources to identify and implement near- and long-term actions to accelerate natural removal of carbon and build climate resilience in our forests, wetlands, urban greenspaces, agricultural soils, and land conservation activities in ways that serve all communities and in particular low-income, disadvantaged, and vulnerable communities. Each agency is to develop a Natural and Working Lands Climate Smart Strategy that serves as a framework to advance the state's carbon neutrality goal and build climate resilience.

#### 3.4.4.2 Caltrans Activities

Caltrans continues to be involved on the Governor's Climate Action Team as the CARB works to implement EOs S-3-05 and S-01-07 and help achieve the targets set forth in AB 32. EO B-30-15, issued in April 2015, and SB 32 (2016), set an interim target to cut GHG emissions to 40 percent below 1990 levels by 2030. The following major initiatives are underway at Caltrans to help meet these targets.

#### **California Transportation Plan**

The *California Transportation Plan* (CTP) is a statewide, long-range transportation plan to meet our future mobility needs and reduce GHG emissions. It serves as an umbrella document for all the other statewide transportation planning documents. The CTP 2050 presents a vision of a safe, resilient, and universally accessible transportation system that supports vibrant communities, advances racial and economic justice, and improves public and environmental health. The plan's climate goal is to achieve statewide GHG emissions reduction targets and increase resilience to climate change. It demonstrates how GHG emissions from the transportation sector can be reduced through advancements in clean fuel technologies; continued shifts toward active travel, transit, and shared mobility; more efficient land use and development practices; and continued shifts to telework (Caltrans 2021a).

SB 391 requires the CTP to meet California's climate change goals under AB 32. Accordingly, the CTP identifies the statewide transportation system needed to achieve maximum feasible GHG emission reductions while meeting the state's transportation needs. Although MPOs have primary responsibility for identifying land use patterns to help reduce GHG emissions, the CTP identifies additional strategies.

#### Caltrans Strategic PLAN

The Caltrans 2020–2024 Strategic Plan includes goals of stewardship, climate action, and equity. Climate action strategies include developing and implementing a Caltrans Climate Action Plan; a robust program of climate action education, training, and outreach; partnership and collaboration; a VMT monitoring and reduction program; and engaging with the most vulnerable communities in developing and implementing Caltrans' climate action activities (Caltrans 2021b).

#### Funding and Technical Assistance Programs

In addition to developing plans and performance targets to reduce GHG emissions, Caltrans also administers several sustainable transportation planning grants. These grants encourage local and regional multimodal transportation, housing, and land use planning that furthers the region's RTP/SCS; contribute to the State's GHG reduction targets and advance transportation-related GHG emission reduction project types/ strategies; and support other climate adaptation goals (e.g., Safeguarding California).

#### **Caltrans Policy Directives and Other Initiatives**

Caltrans Director's Policy 30 (DP-30) Climate Change (June 22, 2012) established a Departmental policy to ensure coordinated efforts to incorporate climate change into

Departmental decisions and activities. Caltrans Activities to Address Climate Change (April 2013) provides a comprehensive overview of Caltrans' statewide activities to reduce GHG emissions resulting from agency operations.

#### **Project-Level GHG Reduction Strategies**

Operation of the proposed project would encourage ridesharing, carpooling, and mass transit use; and reduce recurring during AM and PM peak hours, which would help reduce GHG emissions from idling vehicles. Two tolling strategies are also being considered that would result in a reduction of VMT.

The following measures would also be implemented in the project to reduce GHG emissions and potential climate change impacts from the project.

- 1. A TMP would be prepared during the design phase of the project to minimize traffic disruptions from project construction. Minimizing traffic delays during construction would help reduce GHG emissions from idling vehicles.
- 2. Caltrans Standard Specifications such as Section 14-9.02, Air Pollution Control, require contractors to comply with all federal, state, and local air pollution control rules, regulations, and ordinances. This includes requirements such as idling restrictions and keeping engines properly tuned reduce emissions, including GHG emissions.
- 3. Caltrans would implement all project features described in Table 1-4 in Section 1.5 (see PF-AIR-02), which would reduce GHG emissions from the project during construction activities.

#### 3.4.5 Adaptation

Reducing GHG emissions is only one part of an approach to addressing climate change. Caltrans must plan for the effects of climate change on the state's transportation infrastructure and strengthen or protect the facilities from damage. Climate change is expected to produce increased variability in precipitation, rising temperatures, rising sea levels, variability in storm surges and their intensity, and in the frequency and intensity of wildfires. Flooding and erosion can damage or wash out roads; longer periods of intense heat can buckle pavement and railroad tracks; storm surges combined with a rising sea level can inundate highways. Wildfire can directly burn facilities and indirectly cause damage when rain falls on denuded slopes that landslide after a fire. Effects would vary by location and may, in the most extreme cases, require that a facility be relocated or redesigned. Accordingly, Caltrans must consider these types of climate stressors in how highways are planned, designed, built, operated, and maintained.

#### 3.4.5.1 Federal Efforts

Under NEPA assignment, Caltrans is obligated to comply with all applicable federal environmental laws and FHWA NEPA regulations, policies, and guidance.

The United States Global Change Research Program delivers a report to Congress and the president every 4 years, in accordance with the Global Change Research Act of 1990 (15 USC Chapter 56A § 2921 et seq). The Fourth National Climate Assessment, published in 2018, presents the foundational science and the "human welfare, societal, and environmental elements of climate change and variability for 10 regions and 18 national topics, with particular attention paid to observed and projected risks, impacts, consideration of risk reduction, and implications under different mitigation pathways." Chapter 12, "Transportation," presents a key discussion of vulnerability assessments. It notes that "asset owners and operators have increasingly conducted more focused studies of particular assets that consider multiple climate hazards and scenarios in the context of asset-specific information, such as design lifetime" (USGCRP 2018).

The USDOT Policy Statement on Climate Adaptation in June 2011 committed the federal Department of Transportation to "integrate consideration of climate change impacts and adaptation into the planning, operations, policies, and programs of USDOT to ensure that taxpayer resources are invested wisely, and that transportation infrastructure, services and operations remain effective in current and future climate conditions" (USDOT 2011).

FHWA order 5520 (Transportation System Preparedness and Resilience to Climate Change and Extreme Weather Events, December 15, 2014) established FHWA policy to strive to identify the risks of climate change and extreme weather events to current and planned transportation systems. FHWA has developed guidance and tools for transportation planning that foster resilience to climate effects and sustainability at the federal, state, and local levels (FHWA 2019).

#### 3.4.5.2 State Efforts

Climate change adaptation for transportation infrastructure involves long-term planning and risk management to address vulnerabilities in the transportation system. California's Fourth Climate Change Assessment (State of California 2018) is the state's effort to "translate the state of climate science into useful information for action" in a variety of sectors at both statewide and local scales. It adopts the following key terms used widely in climate change analysis and policy documents:

- *Adaptation* to climate change refers to adjustment in natural or human systems in response to actual or expected climatic stimuli or their effects, which moderates harm or exploits beneficial opportunities.
- Adaptive capacity is the "combination of the strengths, attributes, and resources available to an individual, community, society, or organization that can be used to prepare for and undertake actions to reduce adverse impacts, moderate harm, or exploit beneficial opportunities."

- *Exposure* is the presence of people, infrastructure, natural systems, and economic, cultural, and social resources in areas that are subject to harm.
- *Resilience* is the "capacity of any entity—an individual, a community, an organization, or a natural system—to prepare for disruptions, to recover from shocks and stresses, and to adapt and grow from a disruptive experience." Adaptation actions contribute to increasing resilience, which is a desired outcome or state of being.
- *Sensitivity* is the level to which a species, natural system, or community, government, etc., would be affected by changing climate conditions.
- Vulnerability is the "susceptibility to harm from exposure to stresses associated with environmental and social change and from the absence of capacity to adapt." Vulnerability can increase because of physical (built and environmental), social, political, and/or economic factor(s). These factors include, but are not limited to: ethnicity, class, sexual orientation and identification, national origin, and income inequality. Vulnerability is often defined as the combination of sensitivity and adaptive capacity as affected by the level of exposure to changing climate.

Several key state policies have guided climate change adaptation efforts to date. Recent state publications produced in response to these policies draw on these definitions.

EO S-13-08, issued by then-governor Arnold Schwarzenegger in November 2008, focused on SLR and resulted in the *California Climate Adaptation Strategy* (2009), updated in 2014 as *Safeguarding California: Reducing Climate Risk* (Safeguarding California Plan). The Safeguarding California Plan offers policy principles and recommendations and continues to be revised and augmented with sector-specific adaptation strategies, ongoing actions, and next steps for agencies.

EO S-13-08 also led to the publication of a series of SLR assessment reports and associated guidance and policies. These reports formed the foundation of an interim *State of California Sea-Level Rise Interim Guidance Document* (SLR Guidance) in 2010, with instructions for how state agencies could incorporate "SLR projections into planning and decision making for projects in California" in a consistent way across agencies. The guidance was revised and augmented in 2013. *Rising Seas in California – An Update on Sea-Level Rise Science* was published in 2017 and its updated projections of SLR and new understanding of processes and potential impacts in California were incorporated into the *State of California Sea-Level Rise Guidance Update* in 2018.

EO B-30-15, signed in April 2015, requires state agencies to factor climate change into all planning and investment decisions. This EO recognizes that effects of climate change other than SLR also threaten California's infrastructure. At the direction of EO B-30-15, the Office of Planning and Research published *Planning and Investing for a Resilient California: A Guidebook for State Agencies* in 2017, to encourage a uniform

and systematic approach. Representatives of Caltrans participated in the multiagency, multidisciplinary technical advisory group that developed this guidance on how to integrate climate change into planning and investment.

AB 2800 created the multidisciplinary Climate-Safe Infrastructure Working Group, which in 2018 released its report, *Paying it Forward: The Path Toward Climate-Safe Infrastructure in California*. The report provides guidance to agencies on how to address the challenges of assessing risk in the face of inherent uncertainties still posed by the best available science on climate change. It also examines how state agencies can use infrastructure planning, design, and implementation processes to address the observed and anticipated climate change impacts.

#### 3.4.5.3 Caltrans Adaptation Efforts

#### **Caltrans Vulnerability Assessments**

Caltrans conducted climate change vulnerability assessments to identify segments of the State Highway System vulnerable to climate change effects including precipitation, temperature, wildfire, storm surge, and SLR. The approach to the vulnerability assessments was tailored to the practices of a transportation agency, and involves the following concepts and actions:

- *Exposure* Identify Caltrans assets exposed to damage or reduced service life from expected future conditions.
- *Consequence* Determine what might occur to system assets in terms of loss of use or costs of repair.
- Prioritization Develop a method for making capital programming decisions to address identified risks, including considerations of system use and/or timing of expected exposure.

The climate change data in the assessments were developed in coordination with climate change scientists and experts at federal, state, and regional organizations at the forefront of climate science. The findings of the vulnerability assessments would guide analysis of at-risk assets and development of adaptation plans to reduce the likelihood of damage to the State Highway System, allowing Caltrans to both reduce the costs of storm damage and to provide and maintain transportation that meets the needs of all Californians.

#### **Project Adaptation Analysis**

The January 2018 *Caltrans Climate Change Vulnerability Assessments* for the District 4 region (Caltrans 2018), which covers the nine-county San Francisco Bay Area, was consulted regarding climate stressors in the project area. The report and accompanying Climate Change Vulnerability Assessment map tool (Caltrans 2017a) identified the following climate change conditions for the project area for the analysis years 2025, 2055, and 2085.

#### SR 37 Sea-Level Rise Planning Efforts

In 2018, the California Ocean Protection Council (OPC) released SLR guidance for California that provides probabilistic SLR projections for a range of future scenarios. The SLR projections up to 2100 for San Francisco Bay are shown in Table 3-7. The range in projections represent uncertainties in climate models and scientific understanding of the physical processes associated with climate change. The OPC guidance presents probabilistic projections to capture this uncertainty, and the table shows SLR values with different probabilities of exceedance to allow asset managers to make informed, risk-based decisions on future planning and design.

Modeled Year	Projected SLR Increase with a 66 Percent Likelihood (feet)	Projected SLR Increase with a 5 Percent Likelihood (feet)	Projected SLR Increase with a 0.5 Percent Likelihood (feet)	H++ Scenario Extreme Risk Scenario (feet)
2030	0.5	0.6	0.8	1.0
2040	0.8	1.0	1.3	1.8
2050	1.1	1.4	1.9	2.7
2060	1.5	1.8	2.6	3.9
2070	1.9	2.4	3.5	5.2
2080	2.4	3.0	4.5	6.6
2090	2.9	3.6	5.6	8.3
2100	3.4	4.4	6.9	10.2

Table 3-7	Summary	of State	of California	SLR Pro	ections fo	r San F	- rancisco	Bav
	Cummuny				jeeuono re		141101300	Duy

Source: OPC 2018

Note: Projections assume a high emissions scenario (RCP 8.5). The 66 percent projections assume a low risk aversion scenario, and only upper end range estimates are shown in this table.

OPC guidance recommends that projects with low risk aversion consider the upper end of the projected "likely range" of probabilities (i.e., the 66 percent likelihood), and projects with medium to high risk aversion consider the 0.5 percent scenario. This project considers medium- to-high risk aversion scenarios, with an analysis that focuses on the 0.5 percent likelihood scenario to the assumed design life 2040 because it is a critical transportation asset. Table 3-7 shows the range of projected SLR scenario values from low risk to high risk aversion (66 percent, 5 percent, and 0.5 percent probabilities) for each decade up to 2100. The OPC guidance also includes an extreme risk aversion scenario called the "H++ Scenario." This scenario has an unknown probability and assumes that extreme SLR resulting from the loss of the West Antarctic ice sheet occurs in each projected year. This extreme scenario is typically used for projects with high stakes and long-term decision-making processes. The proposed project is not considered to be high stakes and is proposed as an effort to address congestion in the immediate future. The "H++ Scenario" is presented here for the purpose of illustrating all projected scenarios provided by the OPC. Upon completion in 2025, the project would be protected from frequent tidal inundation by the existing levees and elevated roadway. However, due to the elevation of the shoreline compared to extreme tides, it may still be exposed to temporary flooding during storm conditions or in the event of a levee breach at Tubbs Island. Comparing the elevation of the shoreline protection features with specific amounts of SLR and storm surge can show which portions of the roadway are vulnerable to flooding.

In isolation, SLR would have a limited impact on the project. Under a likely scenario in the relatively near future (0.5 percent probability by 2040), SLR is projected to cause a 1.3-foot increase in the daily high tide, which could lead to permanent inundation of about 600 feet of the roadway near Mare Island if no adaptation measures are taken). In combination with extreme tides, SLR would increase the frequency and magnitude of temporary flooding of the roadway. For example, without estimated SLR, the current 50-year storm is predicted to cause flooding in about 2 miles of the highway, and overtopping in portions of the levees around Tubbs Island.

Based on the shoreline analysis in the project area, the following flood impacts on the SR 37 roadway in the project area are assumed under existing conditions (AECOM 2020; i.e., with no projected SLR):

- The levees around Tubbs Island may be exposed to overtopping during a 1-year (or greater) coastal storm event if they are not raised; portions of the impacted levees are near the highway and may result in flooding to the highway. The likelihood of flooding of the highway increases if the overtopping results in a breach of the levee.
- Approximately 960 feet (0.2 mile) of highway may be exposed to flooding on a frequent basis (i.e., approximately annually or every other year (Figure 3-9).
- A 10-year storm event, which has a 10 percent chance of occurring each year, would expose around 3,680 feet (0.7 mile) of the highway to temporary flooding (Figure 3-10). Frequent flooding would occur at Tolay Creek, and frequent and severe flooding would occur at Mare Island.
- Floods from a 50-year storm event, which has a 2 percent chance of occurring each year, would expose 9,900 feet (1.9 miles) of the highway to temporary flooding (Figure 3-11). Frequent flooding would occur at Tolay Creek, and frequent and severe flooding would occur at Mare Island.
- Floods from a 100-year storm event, which has a 1 percent chance of occurring each year, would expose 14,435 feet (2.7 miles) of the highway to temporary flooding (Figure 3-12). Severe flooding is projected at Tolay Creek, and severe to extreme flooding is projected at Mare Island.



SR 37 Traffic Congestion Relief Project Sonomet, Napa & Solano Counties, CA

#### **AREAS OF FLOOD IMPACT TO HIGHWAY**

1-yr Storm

#### Figure 3-9 Areas of Flood Impact to Highway Under Existing Conditions with a 1-Year Storm Scenario


**AREAS OF FLOOD IMPACT TO HIGHWAY** 

10-yr Storm

Figure 3-10 Areas of Flood Impact to Highway Under Existing Conditions with a 10-Year Storm Scenario



SR 37 Traffic Congestion Relief Project Sonoma, Napa & Solano Counties, CA

## **AREAS OF FLOOD IMPACT TO HIGHWAY**

50-yr Storm

Figure 3-11 Areas of Flood Impact to Highway Under Existing Conditions with a 50-Year Storm Scenario



SR 37 Traffic Congestion Relief Project Sonomat, Napa & Solano Counties, CA

## AREAS OF FLOOD IMPACT TO HIGHWAY

100-yr Storm

## Figure 3-12 Areas of Flood Impact to Highway Under Existing Conditions with a 100-Year Storm Scenario

The following flood impacts on the SR 37 roadway in the project area are assumed under the OPC projected SLR levels with a 0.5 percent likelihood conditions to the year 2050 (i.e., 1.9 feet of SLR) on top of the 1-, 10-, 50-, and 100-year storm events. These scenarios are based on the existing roadway design and elevation, with an SLR increase 1.9 feet of SLR:

- Floods from a 1-year storm event, which has a 100 percent chance of occurring each year, with 1.9 feet of SLR, would expose 9,900 feet (1.9 miles) of the highway to temporary flooding (Figure 3-13). Frequent flooding would occur at Tolay Creek, and frequent and severe flooding would occur at Mare Island.
- Floods from a 10-year storm event, which has a 10 percent chance of occurring each year, with 1.9 feet of SLR, would expose 22,125 feet (4.2 miles) of the highway to temporary flooding (Figure 3-14). Severe flooding is projected at Tolay Creek, and severe to extreme flooding is projected at Mare Island.
- Floods from a 50-year storm event, which has a 2 percent chance of occurring each year, with 1.9 feet of SLR, would expose 34,720 feet (6.6 miles) of the highway to temporary flooding (Figure 3-15). Extreme flooding is projected at Mare Island and at Tolay Creek.

Based on the shoreline analysis, the most vulnerable sections of SR 37 in the project area were identified. Of the highway segments, the areas that are most prone to flooding are:

- A low-lying segment of SR 37 west of the Mare Island Interchange
- A low-lying bayfront segment of SR 37 between Tolay Creek and Tubbs Island Trailhead
- The portion of SR 37 along the interior of Tubbs Island that is protected by perimeter levees

These areas are the most low-lying sections of the highway and the most vulnerable to existing and future flooding from SLR and storm surge. Overtopping of these segments may lead to flooding of the roadway. The segment near the Mare Island interchange first shows overtopping for a 1-year storm; however, flooding along this segment is typically less than what is shown on the maps due to the wide expanse of fronting marsh, channels, and ponded wetland area and the limited duration of high tides. As SLR increases the height and duration of high tides in the Bay, it is expected that Bay waters would more readily inundate the highway.



SR 37 Traffic Congestion Relief Project Sonoma, Napa & Solano Countiles, CA

## **AREAS OF FLOOD IMPACT TO HIGHWAY**

1-yr Storm + 1.9 ft SLR (MHHW + 38 in SLR)

Figure 3-13 Areas of Flood Impact to Highway Under 1.9 Feet of SLR Conditions with a 1-Year Storm Scenario



SR 37 Traffic Congestion Relief Project Sonoma, Napa & Solano Counties, CA

## **AREAS OF FLOOD IMPACT TO HIGHWAY**

10-yr Storm + 1.9 ft SLR (MHHW + 50 in SLR)

Figure 3-14 Areas of Flood Impact to Highway Under 1.9 Feet of SLR Conditions with a 10-Year Storm Scenario



SR 37 Traffic Congestion Relief Project Sonoma, Napa & Solano Counties, CA

#### **AREAS OF FLOOD IMPACT TO HIGHWAY**

50-yr Storm + 1.9 ft SLR (MHHW + 65 in SLR)

Figure 3-15 Areas of Flood Impact to Highway Under 1.9 Feet of SLR Conditions with a 50-Year Storm Scenario

The segment between Tolay Creek and the Tubbs Island Trailhead first shows overtopping with a 1-year storm and may be impacted approximately annually or every other year. The amount of overtopping shows how much the water would exceed the elevation of the highway and can be used to assess the severity of flooding for each event.

Although the primary goal of the project is to relieve traffic congestion, future impacts on the project related to SLR have been considered. The project includes several design features that would make it more resilient to flooding from SLR. These design features include:

- Equipment that may be vulnerable to inundation, such as communications and power equipment, would be relocated and placed on raised pads for its protection.
- Corrosion-resistant construction materials would be required, as appropriate, for utility, power-service connections, foundations, and drainage facilities.
- In consideration of planning responses for inundation or emergency events, an incident management plan would be developed in cooperation with a multi-agency team. The plan would include emergency response procedures, alternative transportation communication protocols, response and enforcement, and recovery procedures.
- Small-scale raising of the road elevation for two segments of SR 37 near the Mare Island Interchange, and between Tolay Creek and the Tubbs Island Trailhead, would be evaluated and addressed during the final design phase for the selected alternative.
- Sheet pile walls along the edge of shoulders would address roadway confinement, may help minimize floodwater percolating into the base and subgrade, and would reduce seepage into the side slopes of the roadway embankment. In addition, the sheet pile walls heights may be increased above finished grade to provide some flood protection.

The segments of the project alignment that are vulnerable to flooding from SLR are near Tolay Creek and West of Mare Island. The segment of SR 37 in the project area is not as vulnerable to flooding from SLR as other parts of the corridor to the west. This portion of SR 37 is part of a larger corridor, and SLR planning efforts are being addressed on a broader scale. Table 2-55 in Section 2.5.2 includes projects and studies related to SLR. These projects and studies look at how SR 37 infrastructure can be designed to be more resilient to SLR and flooding and are described in more detail below.

SR 37 Design Alternative Assessment for the Ultimate Project (SR 121 to Mare Island; completed) and SR 37 Ultimate SLR Resilient Corridor Design Alternatives Assessment (U.S. 101 to SR 121; in progress): These studies provide a high-level evaluation of long-term project alternatives that could be implemented on SR 37 to

address traffic congestion and SLR, while integrating ecosystem enhancements into project design. The purpose of these studies is to improve resiliency of transportation infrastructure to SLR and flooding.

**SR 37 Corridor SLR and Complete Streets (U.S. 101 to SR 29):** This project proposes to reconstruct SR 37 to address SLR and recurring flooding, while including Complete Streets features to address multi-modal bicycle and pedestrian use. The purpose of this project is to address recurring flooding and future SLR impacts to the existing SR 37 in Marin County. Flooding on SR 37 occurs during seasonal rain and high-tide events, causing delays and closures.

SR 37 Corridor Planning and Environmental Linkages Study (U.S. 101 to Interstate 80): The PEL study was a planning process that examined broad transportation, environmental, community, and economic goals in the greater SR 37 corridor. The PEL planning process completed in 2022 considered strategies and adaptation measures for complex corridor issues such as SLR and flooding, and how these could be addressed in the project's being considered for SR 37.

## Floodplains and Precipitation

As described in Section 3.3.10, most of the project is in SFHA Zone AE. This zone represents the base floodplain with areas subject to flooding by the 100-year flood event, where base floodplain elevations are provided. In these areas, the 100-year flood elevation is approximately 10 to 11 feet NAVD88. Other portions of the project area adjacent to Tubbs Island (between Tolay Creek and Sonoma Creek) are in SFHA Zone VE, which are coastal areas subject to coastal high-hazard flooding and to inundation by the 1-percent-annual-chance flood (100-year) event with additional hazards due to storm-induced velocity wave action. In these project areas, the 100-year flood elevation is approximately 11 to 12 feet NAVD88.

In addition to SLR, the project area is exposed to flooding from extreme tides from the Bay. Storm surges caused by storm induced velocity wave action can have an additive impact to SLR and cause temporary flooding of the roadway. Furthermore, climate change can increase the frequency of intense storms. EPA's Climate Change Indicators tracks the frequency of heavy precipitation events in the US and shows a greater than normal portion of total annual precipitation has come from extreme single-day precipitation events (EPA 2021d). These storms can result in increased runoff from creeks into the Bay. Intense storm events that coincide with very high tides increases the risk of flooding events that could affect the most vulnerable (lowest elevation) segments of the highway (Tolay Creek and Mare Island areas). Although the project would add fill and impervious surface adjacent to the Bay, it would be a minimal amount that only incrementally contributes to risk of fluore with SLR. The project includes several design features that would make it more resilient to precipitation, as described in the bullet point list in the SLR Planning Efforts section above.

The Caltrans Climate Change Vulnerability Assessment for District 4 (2018) mapped the potential change in 100-year storm precipitation depths at less than 5 percent

through 2085 under a "business-as-usual" high-emissions scenario. BCDC and other jurisdictions' policies related to flooding are discussed in Section 2.2.2 and Table 2-2. BCDC would be consulted during the project design phase.

#### Wildfire

Most of the project area is surrounded by marshlands that do not contain steep slopes or high vegetation prone to wildfires. Portions of SR 37 in the local responsibility area for Sonoma County are in moderate fire-hazard severity zones. A large portion of the alignment in the project area in Solano County is in unzoned areas. The project would not change fire risk conditions and it would not change the alignment of SR 37. Caltrans' 2018 revised Standard Specification 7-1.02M(2) mandates fire prevention procedures, including a fire prevention plan, to avoid accidental fire starts during construction.

# Chapter 4 Comments and Coordination

Early and continuing coordination with the general public and public agencies is an essential part of the environmental process. It helps planners determine the necessary scope of environmental documentation and the level of analysis required, and to identify potential impacts and avoidance, minimization, and/or mitigation measures and related environmental requirements. Agency and tribal consultation and public participation for this project have been accomplished through a variety of formal and informal methods, including interagency coordination meetings, public meetings, public notices, PDT meetings, and stakeholder meetings. This chapter summarizes the results of the Caltrans' efforts to fully identify, address, and resolve project-related issues through early and continuing coordination.

## 4.1 Corridor Study

Letters were received from the RWQCB on May 18, 2018, and from the U.S. EPA on August 31, 2018, that requested that Caltrans look at the broader corridor and not compartmentalize issues by projects. As discussed in Table 2-55 in Section 2.5.2, Caltrans and its partners are conducting multiple studies and projects that consider the corridor as a whole, including a Planning and Environmental Linkages (PEL) planning process completed in late 2022 that looked at long-term solutions to address or include adaptation for climate change and SLR. Studies and projects being proposed to address SLR in the broader corridor are discussed in Section 2.4.5.

## 4.2 Public Scoping Process

In compliance with CEQA, a NOP for an EIR was filed with the State Clearinghouse on July 9, 2020. The filing of the NOP began a 46-day public scoping period that extended through August 24, 2020. The NOP is included in Appendix G. The public scoping period and virtual public scoping meeting was noticed on the Caltrans website (https://dot.ca.gov/caltrans-near-me/district-4/d4-projects/d4-37-corridor-projects); through newspaper advertisements in the Marin Independent, Napa Valley Register, Santa Rosa Press Democrat, and Vallejo Times Herald; postcard mailers to approximately 3,000 addresses; an email blast to 180 recipients; flyers mailed to approximately 25 federal and state agencies; and a Caltrans District 4 news release.

A public scoping meeting was held on July 22, 2020, from 6:00 PM to 7:30 PM, through an online Zoom meeting. The purpose of the meeting was to present preliminary information on the project and receive early input on the proposed environmental studies and project alternatives. There were approximately 150 attendees at the meeting and 64 questions/comments were submitted during the meeting.

As part of the public scoping process, the public was invited to submit written comments on the scope and content of the environmental document during the public comment period, which began on July 9, 2020, and ended on August 24, 2020. A total of 48 written comments were submitted. Comments received during the public scoping period were reviewed and are summarized in the *Preliminary Summary of Scoping* 

*Comments Received and Scoping Meeting Questions* Report and in Table 4-1. A more detailed summary is included in Appendix G.

Comment Topic	Summary of Comment Topic
General	General comments included the following topics: support of or opposition to the project or a specific design alternative; requests for consideration of climate change impacts; safety differences among each of the alternatives; consistency with other agency policies; and suggestions for technical topic discussions, evaluation of alternatives, and addressing environmental justice in the environmental document.
Project Design/ Operations	Project design/operations comments included questions regarding general lane and HOV lane usage, requests to maintain the size of vehicle pull-outs, noting that the lack of shoulders could be problematic for emergency response, and suggestions for alternative designs such as lengthening of Tolay Creek Bridge.
Project and Agency Coordination Recommendations	Several comments received included recommendations for coordination with agencies such as the NAHC, BCDC, RWQCB, Transportation Authority of Marin (TAM), and other regulatory agencies.
Bicycle and Pedestrian Facilities	Many comments received expressed concern regarding accommodation of bicycle and pedestrian facilities and requests to include bicycle and pedestrian facilities along the SR 37 project corridor.
Mitigation	The topic of mitigation included comments related to suggestions for the project to avoid potential VMT impacts and impacts to San Francisco Bay.
Environmental Issues to Consider	Several comments received included suggestions for the analysis in the environmental document regarding topics such as biological resources, hydrology/water quality, noise, and transportation. Comments included suggestions for mitigation/avoidance measures, technical studies to be included, and potential impacts to be evaluated.
Tolling	Tolling comments included suggestions for incorporating a toll on the Tolay Creek Bridge and a toll gantry west of the Mare Island intersection.

Table 4-1 Overall Topics Raised in Public Scoping Comments

## 4.3 **Project and Stakeholder Coordination**

A PDT was formed at the initiation of this project, consisting of representatives from the many stakeholders involved. The PDT includes representatives of Caltrans, MTC, SCTA, STA, NVTA, and the consultants.

A Technical Advisory Committee (TAC) was established during the early phases of the project as part of the overall SR 37 Corridor Planning team that consists of these same agencies. The TAC guided the development of the project through early conception into the preliminary design and environmental review phase, referred to as project approval and environmental document. The SR 37 Policy Committee, is a multi-county committee with policy makers participating from Marin, Napa, Sonoma, and Solano Counties, also provided guidance. The SR 37 Policy Committee was originally formed in 2015 as part of a Memorandum of Understanding to discuss joint county efforts in improving the SR 37 corridor, addressing issues such as SLR, traffic congestion,

transit options, and recreational activities. The policy committee has continued to meet and provide input to project development.

During development of the plan for evaluating near-term and long-term solutions to SR 37, a series of workshops and working group meetings were held with key environmental stakeholders in the development of alternatives. Attendees of these workshops and meetings include TAC members, and representatives from BCDC, California Coastal Conservancy, SR 37 Baylands Group, Ducks Unlimited, Greenbelt Alliance, Marin Audubon Society, Point Blue Conservation Science, San Francisco Bay RWQCB, San Francisco Bay Joint Venture, San Francisco Bay Trail, San Francisco Estuary Institute, SMART, Sonoma Land Trust, the Nature Conservancy, USACE, USFWS, CDFW, and Vallejo Sanitation and Flood Control District.

Caltrans also led a public engagement process as part of the early alternatives assessment and corridor improvement development. These efforts included:

- In 2017, Caltrans, the MTC, the TAM, SCTA, NVTA, and STA conducted a series of open houses to inform the public about the SR 37 Improvement Plan. The attendance at the open houses ranged from approximately 30 to about 100 members of the public. Staff and management from Caltrans, MTC and the four transportations authorities were in attendance, as well as elected officials from the local counties and cities.
- An online survey was conducted to better understand the travel patterns of regular SR 37 users and to collect feedback about users' major concerns and priorities for improvements along the highway. The survey was open and available for input in 2017 and 2018, and more than 3,750 responses were collected.
- Two rounds of focus group meetings were held in 2018 throughout the four North Bay Counties. Eleven focus groups were conducted. These focus groups were conducted to gain a better understanding of travel patterns on SR 37 from daily commuters in the four-county area; to identify specific locations on the route where travelers have key issues and concerns; to identify improvements along the corridor; and to obtain feedback for alternatives being considered.

## 4.4 Consultation and Coordination with Public Agencies

## 4.4.1 Federal Highway Administration

The project's air quality conformity determination packet was submitted to FHWA for a project-level conformity determination. FHWA concurred with Caltrans' determination that the project conforms with the State Implementation Plan (refer to Volume 2, Appendix D).

## 4.4.2 United States Army Corps of Engineers

The proposed project alternatives have been designed to minimize impacts to wetlands and other waters of the United States, but the various alternatives under consideration would affect these resources as described in this report. A preliminary jurisdictional wetland delineation has been prepared for submittal to USACE. A permit application would be submitted to the USACE during the detailed design phase.

# 4.4.3 United States Fish and Wildlife Service and National Marine Fisheries Service

Endangered species consultation with USFWS and NMFS is necessary when a project has the potential to affect a federally listed species and/or destroy or adversely modify designated critical habitat. Through the NEPA Assignment Memorandum of Agreement with FHWA, Caltrans is authorized to handle FHWA's responsibilities under NEPA and other federal environmental laws, such as FESA.

Caltrans has prepared draft Biological Assessments based on the identified preferred alternative (Alternative 3B) for use in consultation with USFWS and NMFS. Caltrans has made preliminary effects determinations that would be presented to the Services during Section 7 consultation. USFWS and NMFS staff responsible for conducting Section 7 review and consultation have provided technical assistance and preliminary feedback to Caltrans in project meetings and interagency meetings. The following list summarizes meetings and outcomes:

- March 3, 2022, Interagency Meeting with the SR 37 project team, USFWS, NMFS, USACE, RWQCB, CDFW, and BCDC. The project team presented all alternatives under consideration and preliminary effects analyses for regulated resources. NMFS preferred to refrain from comments until a preferred alternative is identified. NMFS noted that the project could provide opportunities to benefit special-status species and their habitat. USFWS noted that they would expect to see temporary impacts beyond just ground disturbance to special-status species discussed in a BA for the project. USFWS also wanted to make sure that Western snowy plover impacts were considered for the project. Caltrans confirmed that this species was reviewed for potential effects from the project in technical studies.
- March 16, 2022, Caltrans met with USFWS for a pre-consultation meeting to discuss potential project effects on USFWS-regulated species and habitat. Material presented assumed the greatest potential extent of impact from the project. USFWS staff noted that consultation for California red-legged frog will be conducted through the Sacramento field office, and the Bay-Delta office typically handles consultations up to the water line. USFWS staff noted that consultation on California red-legged frog, soft bird's beak, Ridgway's rail, and salt marsh harvest mouse are appropriate to consider in the consultation for the project. USFWS staff noted that the project with implementation of the proposed in-water work windows likely would not likely affect Delta Smelt, because the species is not likely to occur in the San Pablo Bay due to high temperatures and

salinities. USFWS did advise that the project consider including a discussion on Longfin Smelt, but also stated that because it is a candidate species that USFWS will not consult on it. USFWS noted that if the project is assuming presence of Ridgway's rail, it must observe the standard avoidance and minimization measures for that species, which include work windows and noise buffers. To demonstrate the species is absent and avoid implementing standard measures, protocol-level surveys would need to be conducted and reported. USFWS noted that the potential loss of marginal habitat for salt marsh harvest mouse along roadsides may provide some refuge during high tide events, but was generally not a major concern. USFWS noted that the mouse seems to be doing well in the Suisun Marsh, San Pablo Bay, and San Francisco Bay habitats. Ridgway's rail and salt marsh harvest mouse are the species and habitats that have the greatest potential for impact for this project, which is anticipated to trigger formal consultation for the project. USFWS noted that compensation for project effects is still not clear, and that there are limited to no mechanisms for compensation through purchasing credits at banks. It is therefore likely that an off-site mitigation approach would need to be pursued, and would be subject to input from other agencies that regulate the same or overlapping resources.

- On March 18, 2022, Caltrans met with NMFS staff for a pre-consultation meeting to discuss any NMFS-regulated species and their habitat that have potential to occur in the project area (Steelhead, Chinook Salmon, and North American Green Sturgeon). NMFS noted that they are concerned with water quality impacts associated with roadway runoff, and will be looking for assurances that roadway runoff would be treated on site, or at least drained to ground before entering the water column. NMFS staff provided no input on the potential impacts or potential compensation opportunities presented.
- On June 8, 2022, the project team met with the USFWS Bay-Delta Office and SPBNWR staff to discuss the project's mitigation needs and Section 4(f) requirements. The project team and USFWS staff agreed that additional follow-up through the separate Section 4(f) process and the Section 7 consultation process will be forthcoming, and would provide formal processes to better understand and evaluate potential project effects to resources.
- On November 16, 2022, USFWS provided Caltrans with a letter of concurrence with the Section 4(f) de minimis finding (included in Volume 2, Appendix B).

Caltrans will continue to coordinate with USFWS and NMFS prior to and throughout Section 7 consultations for the project.

## 4.4.4 Tribal Entities

The NAHC was contacted on October 2019 to request a search of the Sacred Lands File for sacred lands or other cultural properties of significance to Native Americans in or near the APE. The correspondence is included in Appendix D. The NAHC responded that the search had returned negative results; follow-up correspondence was sent to the individuals identified by NAHC. One response was received, indicating that the APE was outside of the UAIC's territory and that no further information was available. An email was received from FIGR stating that the project could have potential effects, and requesting consultation as the project further develops. This information is also summarized in Section 2.2.13.

## 4.4.5 State Agencies

## 4.4.5.1 State Historic Preservation Officer

The project's cultural resource studies were submitted to the SHPO for concurrence of a determination of resources that are not eligible for the NRHP, and notification of Caltrans' finding of No Historic Properties Affected. No comments were received from SHPO, and Caltrans made the determination to move forward consistent with the provisions of the Section 106 PA. This documentation is included in Appendix D.

## 4.4.5.2 California Department of Fish and Wildlife

CESA Consultation protects species listed as threatened or endangered from take unless authorized through an incidental take permit. The project may affect the California black rail. General and specific conservation measures are proposed that would avoid and minimize effects to California black rail to the maximum extent practicable, and mitigation may be necessary if impacts cannot be avoided (mitigation may be proposed in combination with other species habitat mitigation).

CDFW was included in the interagency meeting held on March 3, 2022. CDFW staff noted concerns about SLR along this corridor, including a letter that was submitted during circulation for the project (see Appendix K for those comments and responses).

On November 17, 2022, project team members met with CDFW staff to review and discuss the required temporary construction use and permanent acquisitions within CDFW ownership. CDFW subsequently provided a letter of concurrence with the de minimis Section 4(f) findings (Volume 2, Appendix B).

## 4.4.6 Regional Agencies

## 4.4.6.1 Metropolitan Transportation Commission (Air Quality Conformity)

The project team initiated consultation with the Bay Area AQCTF by submitting a Project Assessment Form for  $PM_{2.5}$  Interagency Consultation. The project was found not to be a POAQC by MTC's AQCTF at their May 27, 2021, consultation meeting.

Public comment was requested regarding the information in the Project Assessment Summary for PM<sub>2.5</sub> Interagency Consultation and the Task Force's determination (see Appendix D). Following the close of the public review and comment period for this EIR/EA, all comments received on the air quality conformity determination were included in an air quality conformity report that was submitted to FHWA. As noted in Section 4.4.1, FHWA concurred with Caltrans' air quality conformity determination.

## 4.4.6.2 San Francisco Bay RWQCB

Project construction could affect waters of the state. Pursuant to Section 401 of the CWA, a Notice of Intent would be submitted to the RWQCB during the detailed design phase. The proposed project would implement any general WDRs issued by the RWQCB.

The RWQCB participated in the March 2, 2022, interagency meeting for the project. RWQCB staff suggested that the project team review the comments submitted in their letter for the Draft EIR (see Appendix K). RWQCB staff noted that they would like to see the project refine its mitigation approach for the entire corridor in the coming years; minimize impacts to wetlands and waters to the extent practicable; and noted that on-site treatment opportunities would be limited for the project, and offsite treatment should be identified before permitting to avoid permitting delays. RWQCB staff voiced their recommendation for replacing the Tolay Creek Bridge with a longer bridge to accommodate future conditions planned for that creek. RWQCB requested Caltrans meet regularly with them to coordinate permitting and avoidance and minimization measures for the project.

## 4.4.6.3 Bay Conservation Development Commission

Project construction would include work within BCDC jurisdiction, as described in Section 2.2.3. An initial consultation meeting was held with BCDC on February 17, 2021. This meeting included an overview of the project alternatives and the preliminary jurisdictional area potentially affected by the project alternatives; and discussion about public shoreline access, bicycle access along SR 37, and design options that should be addressed. An application would be submitted to BCDC during the project design phase.

BCDC staff participated in the March 2, 2022, interagency meeting for the project. BCDC identified concerns regarding SLR and long-term planning for the corridor. BCDC noted its regulatory requirement for Caltrans to provide maximum feasible public access consistent with the proposed project. The project team noted that it has and will continue to coordinate with BCDC, Bay Trail staff, and other stakeholders regarding public access options for the project.

## 4.5 Circulation, Review, and Comments on the Draft Environmental Document

The Draft EIR/EA was circulated to the public for 45 days between January 13 and February 28, 2022. A public on-line meeting was held on February 2, 2022. Comments received during this period are included in Appendix K (under separate cover). Appendix K also includes the responses to each comment.

The availability of the Draft EIR/EA for review and comment was advertised and noticed through a range of outreach methods. Each of the notices and mailers provided information on how to obtain and review the Draft EIR/EA, how to comment, and the deadline for comments; how to participate in the public meeting; and who to contact at Caltrans for more information or assistance, including weblinks, Caltrans

contacts and phone numbers/addresses, and an email box and postal mail address for submitting comments. The notices included contact information in Spanish. A Notice of Completion (NOC) was posted with the California State Clearinghouse on January 13, 2022, identifying the start and end dates of the public review period (SCH #2020070226), and the NOC was distributed through the Clearinghouse to a wide range of State agencies and commissions. Letters to elected and nonelected officials were mailed. Postcards were distributed to local area mailing addresses through the U.S. Postal Service Direct Mail service, and an information flyer was distributed through emails to the project and Caltrans distribution list. Newspaper advertisements were posted on January 13, 2022, in the Vallejo Times Herald, Santa Rosa Press Democrat, Napa Valley Register, and Marin Independent Journal.

After the approval of this Final EIR/EA, a Notice of Determination and Notice of Availability will be filed, and sent to affected units of federal, state, and local governments, and to the State Clearinghouse, in compliance with EO 12372.

# **Chapter 5 List of Preparers**

The preparation of this environmental document and project design involved a team of Caltrans personnel and consultants.

## California Department of Transportation

Rui "Ricky" Gao, Regional Project Manager, Project Management - North

Lindsay Vivian, Office Chief, Office of Environmental Analysis

Christopher Caputo, Acting Environmental Planning and Engineering Division Chief

Maxwell Lammert, Acting Office Chief, Office of Environmental Analysis

Cody Ericksen, Acting Senior Environmental Scientist, Office of Environmental Analysis

Yolanda Rivas, Senior Environmental Planner, Office of Environmental Analysis

Gesse Melaku, Project Manager, Design, Napa County

Ahmed Rahid, Senior Transportation Engineer, Design, Napa and Marin

Solomon Tesfe, Senior Transportation Engineer, Design

Robert Effinger, Supervising Transportation Engineer, Design

Robert Blizard, Branch Chief, North Counties, Biological Sciences and Permits

Stephen Haas, Chief, Office of Design, Napa and Marin

Kathryn Rose, Senior Environmental Planner, Office of Cultural Resources Studies

Helen Blackmore, Architectural History Branch Chief, Office of Cultural Resources Studies

Kristina Montgomery, Associate Environmental Planner, Archaeology, Office of Cultural Resources Studies

Charles Palmer, Principal Architectural Historian, Office of Cultural Resources Studies

Shilpa Mareddy, Air and Noise Branch Chief, Office of Environmental Engineering

Chris Katrak, Transportation Engineer (Civil), Air and Noise

Kevin Krewson, Chief, Environmental Engineering, Air and Noise

Kenny Tsan, Air and Noise Engineer, Office of Environmental Engineering

Daisy Laurino, Transportation Engineer, Office of Environmental Engineering

Sergio Ruiz, Pedestrian and Bicycle Coordinator, Complete Streets Coordinator		
Wesley Bexton, Landscape Architect, Office of Landscape Architecture		
Diana Pink, Landscape Architect, Office of Landscape Architecture		
Brenda Powell-Jones, Sr. Environmental Planner, Headquarters Climate Change Policy		
Chris Wilson, Hazardous Waste Branch Chief, Office of Environmental Engineering		
Joaquin Pedrin, Branch Chief, Marin, Sonoma, Napa, Solano Counties, Office of Landscape Architecture		
Khai Leong, Office Chief, Office of Hydraulic Engineering		
Brian J. Rowley, Branch Chief, Office of Water Quality		
Andrew Chuong, Transportation Engineer, Office of Water Quality		
Christopher Risden, Branch Chief, Office of Geotechnical Design		
James Allen, Engineering Geologist, Office of Geotechnical Design		
Aung Maung, Division Chief, Traffic Operations		
Lester S. Lee, Office Chief, Corridor Management South-West, Traffic Operations		
Kenneth Xu, Office Chief, Project Development Support		
William Woolery, Senior Transportation Engineer, Traffic Operations		
Rick Yeung, Senior Transportation Engineer, Traffic Safety Engineering		
Metropolitan Transportation Commission		
Kevin Chen, Assistant Director, Design and Project Delivery		
Jeanette Weisman, State Route 37 Corridor Program Manager		
Ashley Nguyen, Director, Design and Project Delivery		
Sonoma County Transportation Authority		
James Cameron, Director of Projects and Programming		
David Ripperda, Assistant Director of Projects and Programming		
Solano Transportation Authority		
Janet Adams, Executive Director of Projects		

Nick Burton, Civil Engineer

## Napa Valley Transportation Authority

Sanjay Mishra, Program Manager

Danielle Schmitz, Director Projects and Planning

## AECOM

Jeff Zimmerman, Senior Environmental Manager Dillon Lennebacker, Deputy Environmental Manager Stephanie Osby, Deputy Environmental Manager Catherine Clark, Environmental Manager/Outreach Emily Biro, Environmental Planner/Outreach Joy Villafranca, Senior Project Manager (Engineering) Ramsey Hissen, Senior Manager (Engineering) Saana Deichsel, Senior Biologist Tierra Groff, Biologist/Natural Resources David Pecora, Senior Biologist/Fisheries Kristin Tremain, Biologist Karin G. Beck, Senior Archaeologist Kathleen Kubal, Archaeologist/Geoarchaeologist Jay Rehor, Archaeologist/Geoarchaeologist Annamarie Guerrero, Archaeologist Alex Remar, Senior GIS Analyst Justin Vandeveer, Coastal Engineer/Sea-Level Rise Sarah Kassem, Sea-Level Rise Aswini Rajagopalan, Transportation Engineer Swathi Korpu, Transportation Engineer Robert Horwath, Paleontology Gita Datt, Hazardous Materials

Broden Farazmand, Environmental Planner, Energy

Steven Smith, Senior Planner, Community Impacts/Section 4(f)

Charlotte Hummer, Planner, Community Impacts

Derek McCulloch, Senior Editor

#### Kimley-Horn

Ravi Puttagunta, Senior Traffic Manager

Ha Dao, Transportation Engineer

#### Kittelson & Associates, Incorporated

Damian Stefanakis, Principal Planner

## Illingworth & Rodkin, Inc.

Michael Thill, Principal Consultant

Jay Witt, PE, Staff Consultant

Steve Deines, Staff Consultant

#### **Earthview Science**

MariaElena Conserva (visual analysis)

## **Apex Strategies**

Eileen Goodwin, President, Public Education

## **Chapter 6 Distribution List**

The following agencies, organizations, and individuals received printed or electronic copies of this document. Agency names marked with an asterisk (\*) received copies through the State Clearinghouse.

#### **Federal Agencies**

Branch Chief, Gary Stern National Oceanic and Atmospheric Administration San Francisco Bay Branch 777 Sonoma Avenue, Room 325 Santa Rosa, CA, 95404

Acting Chief, Regulatory Division, James C. Mazza United States Army Corps of Engineers 450 Golden Gate Avenue, 4th Floor San Francisco, CA 94102

Regional Director, Paul Souza United States Fish and Wildlife Service 2800 Cottage Way, Room W-2605 Sacramento, CA 95825

NEPA Reviewer, Zac Appleton United States Environmental Protection Agency, Region 9 75 Hawthorne Street, ENF-4-2 San Francisco, CA 94105

Acting Regional Administrator, Deborah Jordan United States Environmental Protection Agency, Region 9 75 Hawthorne Street San Francisco, CA 94105

Don Brubaker, San Pablo Baylands National Wildlife Refuge 1 Marshlands Rd Fremont, CA 94555

#### **Federal Elected Officials**

Honorable Dianne Feinstein United States Senate 333 Bush Street, Suite 3225, San Francisco, CA 94104

Honorable Alex Padilla United States Senate 333 Bush Street, Suite 3225, San Francisco, CA 94104

Honorable Jared Huffman United States Congress, California's 2nd District 999 Fifth Avenue, Suite 290, San Rafael, CA 94901

Honorable Mike Thompson United States Congress, California's 5th District 2721 Napa Valley Corporate Drive, Napa, CA 94558

Honorable John Garamendi United States Congress, California's 3rd District 698 Emerson Street Palo Alto, CA 94301

#### **State Agencies**

Director of Planning, Jessica Fain \*San Francisco Bay Conservation and Development Commission 375 Beale St., Suite 510 San Francisco, CA 94105

Division Chief, Keith Lichten Watershed Management \*San Francisco Bay Regional Water Quality Control Board 1515 Clay Street, Suite 1400 Oakland, CA 94612

Executive Director, Mitch Weiss California Transportation Commission 1120 N Street, MS-52 Sacramento, CA 95814 Executive Director, Kate Gordon Office of Planning and Research State Clearinghouse 1400 Tenth Street Sacramento, CA 95814

Director, David Shabazian California Department of Conservation 801 K Street, MS 24-01 Sacramento, CA 95814

Bay Delta Regional Manager, Stephanie Fong (Acting Regional Manager) \*California Department of Fish and Wildlife, Region #3 2825 Cordelia Road, Suite 100 Fairfield, CA 94534

State Historic Preservation Officer, Julianne Polanco \*Office of Historic Preservation 1725 23rd Street, Suite 100 Sacramento, CA 95816

Director, Armando Quintero \*California Department of Parks and Recreation Natural Resources Division P.O. Box 942896 Sacramento, CA 94296

Director, Karla Nemeth California Department of Water Resources Division of Environmental Services 3500 Industrial Boulevard West Sacramento, CA 95691

Commissioner, Amanda L. Ray California Highway Patrol Office of Special Representative 601 North Seventh Street Sacramento, CA 95811 \*California Highway Patrol, Golden Gate Division 3050 Travis Boulevard Fairfield, CA 94534

Secretary, Wade Crowfoot \*California Natural Resources Agency 1416 Ninth Street, Suite 1311 Sacramento, CA 95814

Chair, Liane M. Randolph \*California Air Resources Board 1001 I Street P.O. Box 2815 Sacramento, CA 95812

Board Chair, E. Joaquin Esquivel \*State Water Resources Control Board Division of Water Quality P.O. Box 100 Sacramento, CA 95812

Chair, David Hochschild California Energy Commission 1516 Ninth Street, MS-29 Sacramento, CA 95814

Executive Secretary, Christina Snider \*California Native American Heritage Commission 1550 Harbor Blvd, Suite 100 West Sacramento, CA 95691

President, Marybel Batjer \*California Public Utilities Commission 505 Van Ness Avenue San Francisco, CA 94102

Chief, Environmental Planning and Management, Nicole Dobroski California State Lands Commission 100 Howe Avenue, Suite 100 South Sacramento, CA 95825

\*California State Coastal Conservancy, Jessica Davenport, 1515 Clay Street, 10th Floor Oakland, CA 94612-1401 California Fish and Wildlife, Larry Wycoff and Greg Martinelli 2548 Grizzly Island Road Suisun City, CA 94585

\*State Lands Commission 100 Howe Avenue, Suite 100 S. Sacramento, CA 95825

## State Elected Officials

Honorable Marc Levine California State Assembly District 10 50 D Street, Suite 301 Santa Rosa, CA 95404

Honorable Timothy Grayson California State Assembly District 14 420 Virginia Street, Suite 1C Vallejo, CA 94590

Honorable Cecilia Aguiar-Curry California State Assembly District 4 2721 Napa Valley Corporate Drive Napa, CA 94558

Honorable Bill Dodd California State Senate District 3 2721 Napa Valley Corporate Dr., Napa, CA 94558

Honorable Senator Mike McGuire California State Senate District 2 3501 Civic Center Drive Suite 425 San Rafael, CA 94903

## **Regional Agencies**

Executive Director, Therese McMillan Association of Bay Area Governments 375 Beale Street, Suite 800 San Francisco, CA 94105

Executive Officer, Jack Broadbent Bay Area Air Quality Management District 375 Beale Street, Suite 600 San Francisco, CA 94105 Executive Director, Therese McMillan Metropolitan Transportation Commission 375 Beale Street, Suite 700 San Francisco, CA 94105

Executive Officer, Eileen White \*San Francisco Bay Regional Water Quality Control Board, Region #2 1515 Clay Street, Suite 1400 Oakland, CA 94612

Executive Director, Larry Goldzband San Francisco Bay Conservation and Development Commission Bay Area Metro Center 375 Beale St., Suite 510 San Francisco, CA 94105

## Local Agencies

City Manager, Greg Nyhoff City of Vallejo 555 Santa Clara Street Vallejo, CA 94590

Planning Manager, Margaret Kavanaugh-Lynch City of Vallejo 555 Santa Clara Street Vallejo, CA 94590

Director of Public Works, Terrance Davis City of Vallejo 555 Santa Clara Street Vallejo, CA 94590

Housing and Community Development Manager, Judy Shepard-Hall City of Vallejo 200 Georgia Street Vallejo, CA 94590

Planning Director, David Morrison County of Napa 1195 Third Street, 2nd Floor Napa, CA 94559 Director of Public Works, Steven E. Lederer County of Napa 1195 Third Street Suite 101 Napa, CA 94559

Planning Manager, Allan Calder County of Solano Planning Services Division 675 Texas St., Suite 5500 Fairfield, CA 94533-6341

Director of Resource Management, Terry Schmidtbauer County of Solano 675 Texas Street, Suite 5500 Fairfield, CA 94533

Director of Transportation and Public Works, Johannes J. Hoevertsz County of Sonoma 2300 County Center Drive Suite B 100 Santa Rosa, CA 95403

Interim Executive Director, Community Development Commission, Barbie Robinson County of Sonoma 1440 Guerneville Road Santa Rosa, CA 95403

Deputy Planning Director, Scott Orr County of Sonoma 2550 Ventura Avenue Santa Rosa, CA 95403

Executive Director, Daryl K. Halls Solano Transportation Authority One Harbor Center, Suite 130 Suisun City, CA 94585

Executive Director, Kate Miller Napa Valley Transportation Authority 625 Burnell Street Napa, CA 94559-3420 Executive Director, Suzanne Smith Sonoma County Transportation Authority 411 King Street Santa Rosa, CA 95404

## Local Elected Officials

Erin Hannigan, Solano County Board of Supervisors, District 1, Chairwoman 675 Texas Street, Suite 6500 Fairfield, CA 94533-6352

Monica Brown, Solano County Board of Supervisors, District 2, Vice-Chair 675 Texas Street, Suite 6500 Fairfield, CA 94533-6352

Jim Spering, Solano County Board of Supervisors, District 3 675 Texas Street, Suite 6500 Fairfield, CA 94533-6352

John Vasquez, Solano County Board of Supervisors, District 4 675 Texas Street, Suite 6500 Fairfield, CA 94533-6352

Mitch Mashburn, Solano County Board of Supervisors, District 5 675 Texas Street, Suite 6500 Fairfield, CA 94533-6352

Brad Wagenknecht, Napa County Board of Supervisors, District 1 1195 Third Street Napa, CA 94559

Ryan Gregory, Napa County Board of Supervisors, District 2, Chair 1195 Third Street Napa, CA 94559

Diane Dillon, Napa County Board of Supervisors, District 3, Vice Chair 1195 Third Street Napa, CA 94559 Alfredo Pedroza Napa County Board of Supervisors, District 4 1195 Third Street Napa, CA 94559

Belia Ramos, Napa County Board of Supervisors, District 5 1195 Third Street Napa, CA 94559

Susan Gorin, Sonoma County Board of Supervisors, District 1 575 Administration Drive Room 100 A Santa Rosa, CA 95403

David Rabbitt, Sonoma County Board of Supervisors, District 2 575 Administration Drive Room 100 A Santa Rosa, CA 95403

Chris Coursey, Sonoma County Board of Supervisors, District 3 575 Administration Drive Room 100 A Santa Rosa, CA 95403

James Gore, Sonoma County Board of Supervisors, District 4 575 Administration Drive Room 100 A Santa Rosa, CA 95403

Lynda Hopkins, Sonoma County Board of Supervisors, District 5 575 Administration Drive Room 100 A Santa Rosa, CA 95403 This page is intentionally left blank.

# Chapter 7 References

- Association of Bay Area Governments (ABAG). 2020. Priority Development Areas GIS tool. Available online at: http://opendata.mtc.ca.gov/datasets/priority-development-areascurrent?geometry=-122.365%2C38.088%2C-122.198% 2C38.135. Accessed May 2020.
- Association of Bay Area Governments and Metropolitan Transportation Commission (ABAG and MTC). 2021a. Final Plan Bay Area 2050. October. Available online at: https://www.planbayarea.org/finalplan2050.
- ABAG and MTC. 2021b. Draft Program Environmental Impact Report. June 2021. Available online at: https://www.planbayarea.org/sites/default/files/documents/ 2021-06/PBA\_2050\_DEIR.pdf. Accessed November 2021).
- ABAG and MTC. 2021c. Plan Bay Area Projects 2050: Projections. Available online at: Plan\_Bay\_Area\_2050\_October\_2021.pdf (planbayarea.org). Accessed November 2021.
- AECOM. 2020a. Paleontological Identification Report/Paleontological Evaluation Report for the SR 37 Traffic Congestion Relief Project. October 2020.
- AECOM. 2020b. Initial Site Assessment for the SR 37 Traffic Congestion Relief Project. July 2020.
- AECOM. 2021a. Traffic Operations Analysis Report Final. December 2021.
- AECOM. 2021b. Energy Technical Report for the SR 37 Traffic Congestion Relief Project. November 2021.
- AECOM. 2021c. Archeological Survey Report for the SR 37 Traffic Congestion Relief Project. June 2021.
- AECOM. 2021d. Extended Phase I Report for the SR 37 Traffic Congestion Relief Project. June 2021.
- AECOM. 2021e. Historic Property Survey Report for the SR 37 Traffic Congestion Relief Project. June 2021.
- AECOM. 2021f. Community Impact Assessment for the SR 37 Traffic Congestion Relief Project. June 2021.
- AECOM. 2021g. Air Quality Report for the SR 37 Traffic Congestion Relief Project. September 2021
- AECOM. 2021h. Visual Impact Assessment for the SR 37 Traffic Congestion Relief Project. July 2021.

- AECOM. 2021i. Natural Environment Study for the SR 37 Traffic Congestion Relief Project. September 2021.
- AECOM. 2021j. Preliminary Geotechnical Design Report for the SR 37 Traffic Congestion Relief Project. June 2021
- Bay Area Air Quality Management District (BAAQMD). 2017. *Spare the Air, Cool the Climate. Clean Air Plan 2017*. Adopted April 19, 2017. Available online at: https://www.baaqmd.gov/plans-and-climate/air-quality-plans/current-plans. Accessed June 28, 2021.
- BAAQMD. 2020. Air Quality Summary Reports. Annual Bay Area Air Quality Summaries. Available online at: https://www.baaqmd.gov/about-air-quality/airquality-summaries.
- Bias, M.A. and M.L. Morrison. 1994. Ecology of the salt marsh harvest mouse in San Pablo Bay. Ph.D. Dissertation., University of California, Berkeley, CA. 243 pp.
- Barth, Matthew and Kanok Boriboonsomsin. 2010. Real-World Carbon Dioxide Impacts of Traffic Congestion. Berkeley, CA: University of California Transportation Center. UCTC-FR-2010-11. Available online at: https://www.researchgate.net/publication/46438207.
- Bolster, B.C., editor. 1998. Terrestrial Mammal Species of Special Concern in California. Draft Final Report Prepared by P.V. Brylski, P.W. Collins,
  E.D. Pierson, W.E. Rainey, and T.E. Kucera. Report Submitted to California Department of Fish and Game Wildlife Management Division, Nongame Bird and Mammal Conservation Program for Contract No. FG3145WM.
- CALFED. 2000. North of the Delta Offstream Storage Investigations. Integrated Storage Investigations. CALFED Bay-Delta Program.
- California Air Resources Board (CARB). 2008. *Climate Change Scoping Plan Appendices. Volume II: Analysis and Documentation*. Appendix I, p. I-19. December. Available online at: https://ww3.arb.ca.gov/cc/scopingplan/ document/scopingplandocument.htm. Accessed October 31, 2019.
- CARB. 2020a. California Greenhouse Gas Emissions Inventory–2019 Edition. Available online at: https://ww3.arb.ca.gov/cc/inventory/data/data.htm. Accessed August 21, 2019.
- CARB. 2020b. California Greenhouse Gas Emissions for 2018. Trends of Emissions and Other Indicators. Available online at: https://ww3.arb.ca.gov/cc/inventory/ pubs/reports/2000\_2017/ghg\_inventory\_trends\_00-17.pdf. Accessed August 21, 2019.

- CARB. 2020c. EMFAC Off-Model Adjustment Factors for Carbon Dioxide (CO<sub>2</sub>) Emissions to Account for the SAFE Vehicles Rule Part One and the Final SAFE Rule. Available online at: https://ww3.arb.ca.gov/msei/emfac\_off\_model\_co2\_ adjustment\_factors\_06262020-final. pdf. Accessed June 15, 2021.
- CARB. 2020d. Select 8 Summary: Choose Statistics, Years, and Areas. Available online at: https://arb.ca.gov/adam/select8/sc8start.php,
- California Department of Conservation (CDOC). 2010. Sonoma County Important Farmland. Available online at: https://maps.conservation.ca.gov/dlrp/ metadata/ImportantFarmland/sonoma\_meta.htm. Accessed May 2020.
- California Department of Fish and Wildlife (CDFW). 1988. *Pallid Bat Life History Account*. California Wildlife Habitat Relationships System. California Department of Fish and Wildlife. California Interagency Wildlife Task Group. Based on an account for *Antrozous pallidus* as written by J. Harris, reviewed by P. Brown, and edited by D. Alley and R. Duke.
- CDFW. 1998. Report to the Fish and Game Commission: A Status Review of the Spring-Run Chinook Salmon (*Oncorhynchus tshawytscha*) in the Sacramento River Drainage. Candidate Species Status Report 98-01. Sacramento, California: Department of Fish and Game.
- CDFW. 2011. Final Napa-Sonoma Marshes Wildlife Area Land Management Plan. Available online at: https://nrm.dfg.ca.gov/File Handler.ashx?DocumentID=97155&inline. Accessed May 2020.
- CDFW. 2016. Report to the Fish and Game Commission: A Status Review of the Spring-run Chinook Salmon (*Oncorhynchus tshawytscha*) in the Sacramento River Drainage. Candidate Species Status Report 98-01. Sacramento, California: Department of Fish and Game.
- CDFW. 2019. California Natural Diversity Database (CNDDB) Commercial version dated August 1, 2019. Available online at: https://map.dfg.ca.gov/rarefind/view/ RareFind.aspx. Accessed August 20, 2019.
- CDFW. 2020. Napa-Sonoma Marshes Wildlife Area. Available online at: https://wildlife. ca.gov/Lands/Places-to-Visit/Napa-Sonoma-Marshes-WA#1211490-recreation. Accessed May 2020.
- California Department of Transportation (Caltrans). 2014a. Field Guide to Construction Site Dewatering. Available online at: http://website.dot.ca.gov/hq/construc/ stormwater/field-guide-to-construction-site-dewatering.pdf. Accessed June 2021.
- Caltrans. 2014b. Deputy Directive DD-64-R2. Available online at: https://dot.ca.gov/-/ media/dot- media/programs/transportation-planning/documents/dd-64-r2a11y.pdf. Accessed June 2021.

- Caltrans 2016. Guidance for Preparers of Cumulative Impact Assessments, CEQA Guidelines for Cumulative and Indirect Impacts, Available online at: https://dot.ca.gov/programs/environmental-analysis/standard-environmentalreference-ser/cumulative-impact-analysis. Accessed June 2021.
- Caltrans. 2017a. Caltrans Climate Change Vulnerability Assessment Map. Available online at: https://www.arcgis.com/apps/webappviewer/index.html?id= 517eecf1b5a542e5b0e25f337f87f5bb.
- Caltrans. 2017b. 2017 Traffic Volumes: Route 34-43. Available online at: https://dot.ca. gov/programs/traffic-operations/census/traffic-volumes/2017/route-34-43. Accessed May 20, 2020.
- Caltrans. 2017c. *Toward an Active California, State Bicycle and Pedestrian Plan.* State Bicycle and Pedestrian Plan. Available online at: https://dot.ca.gov/-/media/dot-media/programs/transportation-planning/documents/f0020350\_activeca\_final-plan-2017-05-18-a11y.pdf. May.
- Caltrans. 2018. Caltrans Climate Change Vulnerability Assessment. Available online at: https://dot.ca.gov/-/media/dot-media/programs/transportation-planning/ documents/2019-climate-change-vulnerability-assessments/ada-remediated/ d4-summary-report-a11y.pdf
- Caltrans. 2020a. 2018 Standard Plans and Standard Specifications. Available online at: https://dot.ca.gov/programs/design/ccs-standard-plans-and-standardspecifications. Accessed June 2021.
- Caltrans. 2020b. Standard Environmental Reference. Available online at: https://dot.ca.gov/programs/environmental-analysis/standard-environmentalreference-ser. Accessed June 2021.
- Caltrans. 2020c. Transportation Analysis Framework. Evaluating Transportation Impacts of Sate Highway System Projects. First Edition.
- Caltrans. 2020d. Contextual Guidance for Bike Facilities. Memorandum. March 11. Available online at: https://dot.ca.gov/-/media/dot-media/programs/ transportation-planning/documents/office-of-smart-mobility-and-climate-change/ planning-contextual-guidance-memo-03-11-20-a11y.pdf.
- Caltrans 2020e. SR 37 Planning and Environmental Linkages (PEL). Division of Environmental Planning and Engineering – Caltrans Bay Area – District 4. November 16. Available online at: https://dot.ca.gov/-/media/dot-media/district-4/documents/37-corridor-projects/sr-37\_swg\_11\_16\_20\_pdf\_notes-2020-12-16.

Caltrans. 2021a. Traffic Operations Analysis Report (TOAR). October 2021.

Caltrans. 2021b. Draft Aquatic Resources Delineation Report. July 2021.

- Caltrans. 2021c. Caltrans Standard Environmental Reference (SER), Volume 1, Chapter 13, Energy. Accessed March 25, 2021. Available online at: https://dot.ca.gov/programs/environmental-analysis/standard-environmentalreference-ser/volume-1-guidance-for-compliance/ch-13-energy.CALFED. 2000. North of the Delta Offstream Storage Investigations. Integrated Storage Investigations. CALFED Bay-Delta Program.
- California Energy Commission (CEC). 2003. State of California Energy Action Plan. Available online at: http://www.energy.ca.gov/energy\_action\_plan/. Accessed March 25, 2021.
- California Environmental Protection Agency. 2015. California Climate Strategy. Available online at: https://www.climatechange.ca.gov/. Accessed August 21, 2019.
- California Invasive Plant Council (Cal-IPC) 2019. California Invasive Plant Inventory. Available online at: www.cal-ipc.org. Accessed December 2019.
- California Native Plant Society (CNPS). 2020. *A Manual* of California Vegetation. Available online at: http://vegetation.cnps.org/. Accessed 2020.
- CNPS (California Native Plant Society). 2021. *California Native Plant Society Electronic Inventory of Rare and Endangered Plants of California*. Online edition. Website: CNPS Inventory of Rare Plants | California Native Plant Society.
- Center for Biological Diversity, 2003. Petition to the State of California Fish and Game Commission and Supporting Information for Listing the California Population of the Western Burrowing Owl (*Athene cunicularia hypugaea*) as an Endangered or Threatened Species Under the California Endangered Species Act. Available online at: https://www.biologicaldiversity.org/species/birds/western\_burrowing\_ owl/pdfs/petition.pdfAccessed November 12, 2020.
- City of Vallejo. 1999. Amended 2013. Mare Island Specific Plan. Available online at: http://www.ci.vallejo.ca.us/common/pages/DisplayFile.aspx?itemId=91206. Accessed May 2020.
- City of Vallejo. 2013. Mare Island Specific Plan.
- City of Vallejo. 2017. Propel Vallejo General Plan 2040. Available online at: https://www.cityofvallejo.net/common/pages/DisplayFile.aspx? itemId= 12181697. Accessed May 2020.
- City of Vallejo. 2019. City of Vallejo News Release: City of Vallejo Enters New Partnership As Nimitz Group Completes Mare Island Acquisition. Available online at: http://www.ci.vallejo.ca.us/common/pages/DisplayFile. aspx?itemId=16218003. Accessed May 2020.

- City of Vallejo. 2020. Major Development & Specific Plan Docs. Available online at: https://www.cityofvallejo.net/city\_hall/departments\_\_\_divisions/planning\_and\_ development\_services/planning\_division/major\_development\_\_\_specific\_plan\_ docs. Accessed May 2020.
- Coastal Conservancy. 2004. North Point Joint Venture Acquisition. Staff Recommendation. File No. 04-079. December 2. Available online at: http://s3-us-west-2.amazonawscom/ucldc-nuxeo-ref-media/d1cc80d0-60e4-4432-b1ca-f6b2a40dffac.
- County of Napa. 2008. *Napa County General Plan*. Available online at: https://www.countyofnapa.org/DocumentCenter/View/3334/Napa-County-General-Plan---Complete-Document-PDF. Accessed September 2020.
- County of Solano. 2008. Sonoma County General Plan. Chapter 2 Land Use. Available online at: https://www.solanocounty.com/civicax/filebank/blobdload.aspx? BlobID=6492?. Accessed September 2020.
- County of Sonoma. 2010. 2010 Sonoma County Bicycle and Pedestrian Plan. Available online at: https://sonomacounty.ca.gov/WorkArea/DownloadAsset. aspx?id=2147533319. Accessed May 2020.
- County of Sonoma. 2020. Sonoma County General Plan 2020. Land Use Element.
- Data USA. 2020. Solano County and Sonoma County. Accessed May 2020.
- Department of Toxic Substances Control (DTSC). 2013. Concurrence with Finding of No Department of Defense Actions Indicated (NDAI) for Formerly Used Defense Sites in California. March 12.
- Federal Emergency Management Agency (FEMA). 2016. *Flood Insurance Study* for Napa County, California and Incorporated Areas. 060055CV000C; and *Flood Insurance Study* for Solano County, California and Incorporated Areas. 06095CV001D.
- FEMA. 2017. *Flood Insurance Study* for Sonoma County, California, and Incorporated Areas. 06097CV001E.
- Federal Highway Administration (FHWA). 1998a. FHWA Traffic Noise Model, Version 1.0 User's Guide. January. FHWA-PD-96-009. Washington D.C.
- FHWA. 1998b. FHWA Traffic Noise Model, Version 1.0. February. FHWA-PD-96-010. Washington D.C.
- FHWA. 2019. Sustainability. Available online at: https://www.fhwa.dot.gov/environment/ sustainability/resilience/. Last updated February 7, 2019. Accessed August 21, 2019.

- FHWA. No date. Sustainable Highways Initiative. Available online at: https://www.sustainablehighways.dot.gov/overview.aspx. Accessed August 21, 2019.
- Fehr & Peers. 2019. SR 37 Travel Behavior and Transit Feasibility Study, prepared for Napa Valley Transportation Authority, Transportation Authority of Marin, Solano Transportation Authority, and Sonoma County Transportation Authority. May 3.
- Fisher, Frank W. 1994. Past and Present Status of the Central Valley Chinook Salmon. *Conservation Biology* 8:870-873.
- Fisheries Hydroacoustic Working Group (FHWG). 2008. Agreement in Principle for Interim Criteria for Injury to Fish from Pile Driving Activities.
- Fukushima, L., and E.W. Lesh. 1998. Adult and Juvenile Anadromous Salmonid Migration Timing in California Streams. *California Fish and Game* 84:133–145.
- Graymer, R.W., D.L. Jones, and E.E. Brabb. 2002. Geologic Map and Map Database of Northeastern San Francisco Bay Region, California. Most of Solano County and Parts of Napa, Marin, Contra Costa, San Joaquin, Sacramento, Yolo, and Sonoma Counties.
- Hulst, M.D., L.S. Hall, M.L. Morrison, and M.A. Bias. 2001. Assessing Salt Marsh Harvest Mouse Movements during High Tides, San Pablo Bay, California. *Transactions of the Western Section of the Wildlife Society* 37:88-91.
- Humple, D., and G.R. Geupel. 2004. Song Sparrow (*Melospiza melodia*). In The Riparian Bird Conservation Plan: a Strategy for Reversing the Decline of Riparian-Associated Birds in California. Version 2.0. California Partners in Flight. Available online at: http://www.prbo.org/calpif/pdfs/riparian\_v-2.pdf. Accessed: October 11, 2019.
- Illingworth & Rodkin, Inc. 2021. Noise Study Report. SR Traffic Congestion Relief Project. July 2021.
- Jennings, M.R., and M.P. Hayes. 1990. Final Report on the Status of the California Red-legged Frog (*Rana aurora draytonii*) in the Pescadero Marsh Natural Preserve. Prepared for the California Department of Parks and Recreation under contract No. 4-823-9018 with the California Academy of Sciences.
- Jennings and Hayes. 1994. *Amphibian and Reptile Species of Special Concern in California*. California Department of Fish and Game.
- Jones and Stokes. 2004. *Napa River Salt Marsh Restoration Project Environmental Impact Statement*. Prepared for the U.S. Army Corps of Engineers.
- Klimey, P., D. Tu, A. Hearn, W. Brostoff, P. LaCivita, A. Bremner, and T. Keegan. 2010. Juvenile Salmonid Outmigration and Distribution in the San Francisco Estuary: 2006-2008 Interim Draft Report.

- Klute, D.S., L.W. Ayers, M.T. Green, W.H. Howe, S.L. Jones, J.A. Shaffer,
  S.R. Sheffield, and T.S. Zimmerman. 2003. Status Assessment and
  Conservation Plan for the Western Burrowing Owl in the United States.
  U.S. Department of Interior, Fish and Wildlife Service, Biological Technical
  Publication FWS/BTP-R6001-2003, Washington, D.C.
- Leidy, R.A., G.S. Becker, and B.N. Harvey. 2005. Historical Distribution and Current Status of Steelhead/Rainbow Trout (*Oncorhynchus mykiss*) in Streams of the San Francisco Estuary, California. Center for Ecosystem Management and Restoration, Oakland, California.
- Lindsay Transportation Solutions. 2022. Online/virtual meeting with Lindsay Transportation Solutions, MTC, and AECOM, April 22.
- LSA Associates, Inc. 2012. Solano County Multispecies Habitat Conservation Plan. Available online at: http://www.scwa2.com/home/showdocument?id=424. Accessed May 2020.
- Madrone Audubon Society. 2020. Sonoma County Breeding Bird Atlas: Species Maps – Atlas #2 from surveys conducted in Sonoma County, 2011-2016. Available online at: http://www.madroneaudubon.org/pdf/Atlas2\_r2.pdf. Accessed November 12, 2020.
- Marriott, Meg, personal communication. 2020. Meg Marriott, Wildlife Biologist, San Pablo Bay National Wildlife Refuge. Personal communication, September 10, 2020.
- Merz, J.E., S. Hamilton, P.S. Bergman, and B. Cavallo. 2011. Spatial Perspective for Delta Smelt: A Summary of Contemporary Survey Data. *California Fish and Game* 97(4):164-189.
- Merz, J.E., P.S. Bergman, J.F. Melgo, and S. Hamilton. 2013. Longfin Smelt: Spatial Dynamics and Ontogeny in the San Francisco Estuary, California. *California Fish and Game* 99(3):122-148.
- Moyle, P.B. 2002. Inland Fishes of California (revised and expanded). University of California Press, Berkeley, California.
- Moyle, P.B., P.J. Foley, and R.M. Yoshiyama. 1992. Status of Green Sturgeon, *Acipenser medirostris*, in California. Final Report submitted to the National Marine Fisheries Service, University of California, Davis.
- Moyle, P.B., R.D. Baster, T. Sommer, T.C. Foin, and S.A. Matern. 2004. Biology and Population Dynamics of Sacramento splittail (*Pogonichthys macrolepidotus*) in the San Francisco Estuary: a Review. *San Francisco Estuary and Watershed Science* [online serial]. Vol. 2, Issue 2, Article 3. May 2004. Available online at: https://pdfs.semanticscholar.org/5e1d/a18c0649f747088efe36e6a254 feaccf5778.pdf. Date Accessed: October 14, 2019.
- Moyle, P. B. 2002. Inland Fishes of California (revised and expanded). University of California Press, Berkeley, California.
- Murphy, D.D., and S.A. Hamilton. 2013. Eastward Migration or Marshward Dispersal: Exercising Survey Data to Elicit an Understanding of Seasonal Movement of Delta Smelt. *San Francisco Estuary and Watershed Science* 11(3).
- Myers, J.M., R.G. Kope, G.J. Bryant, D. Teel, L.J. Lierheimer, T.C. Wainwright, W.S. Grant, F.W. Waknitz, K. Neely, S.T. Lindley, and R.S. Waples. 1998. Status Review of Chinook Salmon from Washington, Idaho, Oregon, and California. Report No. NMFS-MWFSC-35. NOAA Tech. Memo. U.S. Department of Commerce.
- Napa Valley Transportation Authority (NVTA). 2015. Vision 2040 Moving Napa Forward. Available online at: https://nvtatransportationplan.org/Customer/File/ Full/926bd2f3-5dd3-45d9-85e0-8f18ca097acc. Accessed May 2020.
- National Center for Sustainable Transportation (NCST). 2019. Induced Travel Calculator, published online September 1, 2019, with University of California, Davis, Institute of Transportation Studies. Available online at: https://ncst.ucdavis.edu/research-product/induced-travel-calculator.
- National Marine Fisheries Service (NMFS). 1993. Designated Critical Habitat; Sacramento River Winter-run Chinook Salmon. Federal Register 58(114):33212-33219.
- NMFS. 2005. Federal Register; Endangered and Threatened species: Designation of Critical Habitat for Seven Evolutionarily Significant Units of Pacific Salmon and Steelhead in California; Final Rule. 50 CFR Part 17, Vol. 70 (170):52488– 52627.
- NMFS. 2009. Endangered and Threatened Wildlife and Plants: Final Rulemaking to Designate Critical Habitat for the Threatened Southern Distinct Population Segment of North American Green Sturgeon. 50 CFR Part 226 Vol. 74 (195): 52299-52351.
- NMFS. 2014. Recovery Plan for the Evolutionarily Significant Units of Sacramento River Winter-run Chinook Salmon and Central Valley Spring-Run Chinook Salmon and the Distinct Population Segment of California Central Valley Steelhead. California Central Valley Area Office.
- NMFS. 2016a. 5-Year Review: Summary and Evaluation of Central California Coast Steelhead. Available online at: https://repository.library.noaa.gov/view/ noaa/17017.
- NMFS. 2016b. Final Coastal Multispecies Recovery Plan. National Marine Fisheries Service, West Coast Region, Santa Rosa, California.

- National Oceanic and Atmospheric Administration (NOAA). 2021. General Tsunami Information. Available online at: https://www.tsunami.gov/?page=tsunamiFAQ.
- Office of Emergency Services (OES). 2007. Local Planning Guidance on Tsunami Response. Available online at: https://mitigation.eeri.org/files/TsunamResponse Guidance.pdf.
- OES. 1999. Legal Guidelines for Controlling Movement of People and Property During an Emergency. Available online at: https://www.caloes.ca.gov/Planning PreparednessSite/Documents/Legal%20Guidelines%20for%20Controlling% 20Movement%20People%20Property%20(FEAT%20doc).pdf.
- Parsons. 2008. Site-Specific Work Plan Addendum to the Programmatic Work Plan, Tubbs Island Bombing (Gunnery) Range, Sonoma County, California. June.
- San Francisco Bay Regional Water Quality Control Board (SFRWQCB). 2008. Memorandum of Caltrans Post-Construction Stormwater and Hydromodification Standards. (CIWQS Place No. 212806 [BT])
- SFRWQCB. 2019. Water Quality Control Plan (Basin Plan) for the San Francisco Bay Basin. Available online at: https://www.waterboards.ca.gov/sanfranciscobay/ water\_issues/programs/planningtmdls/basinplan/web/docs/BP\_all\_chapters.pdf. Accessed September 4, 2020.
- San Francisco Estuary Institute (SFEI) 2011. Bay Area Aquatic Resources Inventory (BAARI). Standards and methodology for stream network, wetland, and riparian mapping. Wetland Regional Monitoring Program. Revised August 9, 2011. Provided directly from SFEI, September 2019.
- Shuford, D.W., and T. Gardali. 2008. California Bird Species of Special Concern: A Ranked Assessment of Species, Subspecies, and Distinct Populations of Birds of Immediate Conservation Concern in California. In *Studies of Western Birds 1*. Western Field Ornithologists, Camarillo, California and California Department of Fish and Game, Sacramento.
- Smith, K. 2019. Ecology and Conservation of the Salt Marsh Harvest Mouse in the Modern San Francisco Estuary.
- Smith, K., and D. Kelt. 2019. Waterfowl Management and Diet of the Salt Marsh Harvest Mouse. *The Journal of Wildlife Management* 83(8).
- Smith, K.R., M.K. Riley, L. Barthman–Thompson, M.J. Statham, S. Estrella, and D.A. Kelt. 2018. Towards Salt Marsh Harvest Mouse Recovery: Research Priorities. *San Francisco Estuary and Watershed Science* 16.
- Society of Vertebrate Paleontology. 1996. Conditions of Receivership for Paleontological Salvage Collections. Society of Vertebrate Paleontology News Bulletin, no. 166, pp. 31-32.

- Society of Vertebrate Paleontology, 2010. Standard Procedures for the Assessment and Mitigation of Adverse Impacts to Paleontological Resources. Electronic document.
- Solano County Environmental Health Services. 2002. Request for Well Destruction, Riverview Estates, Corner of Wilson Avenue and Lighthouse Drive, Vallejo, California 94590, Solano County File # 10803. February 20.
- Sonoma County Transportation Authority (SCTA). 2016. Moving Forward 2040: Sonoma County's Comprehensive Transportation Plan. Available online at: https://scta.ca.gov/wpcontent/uploads/2016/09/CTP16\_090616.pdf. Accessed May 2020.
- Sonoma Land Trust. 2021. Sears Point Ranch. Available online at: https://sonomaland trust.org/our-preserve-system/anchor-preserves/sears-point-ranch/.
- Sonoma Land Trust and San Francisco Bay Restoration Authority. 2020. Sonoma Creek Baylands Strategy, Final Report – May 2020. Available online at: https://www.sfei.org/sites/default/files/biblio\_files/Sonoma-Creek-Baylands-Strategy\_May-2020\_1.pdf.
- Sonoma Resource Conservation District (SRCD). 2015. Draft Petaluma Watershed Enhancement Plan.
- Solano Transportation Authority. 2011. Solano Countywide Bicycle Transportation Plan. Available online at: https://sta.ca.gov/wpcontent/uploads/2019/01/Solano\_ BikeTransPlan\_Final-12-14-11.pdf. Accessed May 2020.
- Solano Transportation Authority. 2020. Solano County Comprehensive Transportation Plan 2040. Available online at: https://sta.ca.gov/wpcontent/uploads/2018/06/ CTP\_2020\_Final-updated.pdf. Accessed May 2020.
- Sonoma Index Tribune. 2019. NASCAR to cause Sonoma Raceway area delays. June 20. Available online at: https://www.sonomanews.com/news/9723147-181/nascar-to-cause-sonoma-raceway?sba=AAS&artslide=0. Accessed May 28, 2020.
- Speedway Motorsports, LLC. 2020. Sonoma Raceway: Track Facts & History. Available online at: https://www.sonomaraceway.com/track/track-factshistory/. Accessed May 28, 2020.
- State of California. 2018. California's Fourth Climate Change Assessment. Available online at: http://www.climateassessment.ca.gov/. Accessed August 21, 2019.
- State Water Resources Control Board (SWRCB) 2013. NPDES No. CAS000003, SWRCB Order No. 2012-0011-DWQ. Adopted on September 19, 2012, and effective on July 1, 2013. As amended by Order No. 2014-0006-EXEC (effective January 17, 2014), Order No. 2014-0077-DWQ (effective May 20, 2014), and Order No. 2015-0036-EXEC (conformed and effective April 7, 2015).

- SWRCB. 2017. Final 2014/2016 California Integrated Report (Clean Water Act Section 303(d) List/305(b) Report). Available online at: https://www.water boards.ca.gov/water\_issues/programs/tmdl/integrated2014\_2016.shtml. Accessed September 18, 2020.
- Tatarian, P. 2008. Movement Patterns of California Red-Legged Frogs (*Rana draytonii*) in an Inland California Environment. *Herpetological Conservation and Biology* 3(2):155–169.
- Taylor, R.N., and M. Love. 2003. California Salmonid Stream Habitat Restoration Manual, Part IX: Fish Passage Evaluation at Stream Crossings. California Department of Fish and Game, California, USA.
- Tom, Challenger. 2020. California State Route 37; the Blood Alley of San Pablo Bay. Available online at: http://www.gribblenation.org/2020/01/california-state-route-37-blood-alley.html. Accessed June 2021.
- United States Army Corps of Engineers (USACE). 2018. U.S. Army Corps of Engineers Proposed Additional Procedures and Criteria for Permitting Projects under a Programmatic Determination of Not Likely to Adversely Affect Select Listed Species in California (the 2018 NLAA Program). Available online at: https://www.spn.usace.army.mil/Portals/68/docs/regulatory/BOs/Prog/ NMFS\_PBO\_NLAA\_2018.pdf.
- United States Census Bureau. 2017. United States Census Bureau's 2013-2017 American Community Survey 5-Year Estimates.
- United States Department of Energy. 2020. Fuel Economy at Various Driving Speeds. Accessed March 25, 2021. Available online at: https://afdc.energy.gov/data/ 10312U.S.
- United States Environmental Protection Agency (U.S. EPA). 2006. Federal Register, Vol. 71, No. 47, PM<sub>2.5</sub> and PM<sub>10</sub> Hot-Spot Analyses in Project-Level Transportation Conformity Determinations for the New PM<sub>2.5</sub> and Existing PM10 National Ambient Air Quality Standards. Available online at: https://www.govinfo. gov/content/pkg/FR-2006-03-10/pdf/06-2178.pdf.
- U.S. EPA. 2009. Endangerment and Cause or Contribute Findings for Greenhouse Gases under the Section 202(a) of the Clean Air Act. Available online at: https://www.epa.gov/ghgemissions/endangerment-and-cause-or-contributefindings-greenhouse-gases-under-section-202a-clean. Accessed August 21, 2019.
- U.S. EPA. 2021a. Greenhouse Gases Equivalencies Calculator Calculations and References. Available online at: https://www.epa.gov/energy/greenhousegases-equivalencies-calculator-calculations-and-references. Accessed June 15, 2021.

- U.S. EPA. 2021b. U.S. EPA Greenbook. Carbon Monoxide (1971) Designated Area/State Information. Available online at: https://www3.epa.gov/airquality/ greenbook/cbtc.html.
- U.S. EPA. 2021c. Sources of Greenhouse Gas Emissions Overview. Available online at: https://www.epa.gov/ghgemissions/sources-greenhouse-gas-emissions. Accessed July 30, 2021.
- U.S. EPA. 2021d. Climate Change Indicators: Heavy Precipitation. Available online at: https://www.epa.gov/climate-indicators/climate-change-indicators-heavyprecipitation. Accessed November 18, 2021.
- United States Energy Information Administration (EIA). 2018. State Energy Consumption Estimates, 2018. Available online at: https://www.eia.gov/state/ seds/seds-data-complete.php?sid=US#Consumption. Accessed March 15, 2021.
- United States Fish and Wildlife Service (USFWS). 1994. Endangered and threatened wildlife and plants: Critical Habitat determination for delta smelt. December 19, 1994. Federal Register 58(42):12854-12864.
- USFWS. 1996. Sacramento-San Joaquin Delta Native Fishes Recovery Plan. U.S. Fish and Wildlife Service, Portland, Oregon. Available online at: https://ecos.fws.gov/docs/recovery\_plan/961126.pdf.
- USFWS. 2002. Recovery Plan for the California Red-legged Frog (Rana aurora draytonii). U.S. Fish and Wildlife Service, Portland, Oregon. Available online at: https://www.fws.gov/arcata/es/amphibians/crlf/documents/020528.pdf.
- USFWS. 2005. Revised Guidance on Site Assessments and Field Surveys for the California Red-Legged Frog.
- USFWS. 2007. Federal Register; Endangered and Threatened Wildlife and Plants; Designation of Critical Habitat for *Cirsium hydrophilum* var. *hydrophilum* (Suisun thistle) and *Cordylanthus mollis* ssp. *mollis* (soft bird's-beak); Final Rule. 50 CFR Part 17, Vol. 72 (70):18518-18553. April 12, 2007.
- USFWS. 2010. Federal Register; Endangered and Threatened Wildlife and Plants; Revised Designation of Critical Habitat for the California Red-Legged Frog; Final Rule. 50 CFR Part 17, Vol. 75 (51):12816-12959. March 17.
- USFWS. 2013a. *Recovery Plan for Tidal marsh Ecosystems of Northern and Central California*. Sacramento, CA. Available online at: https://www.fws.gov/sfbaydelta/documents/tidal\_marsh\_recovery\_plan\_v1.pdf. Accessed October 7, 2019.
- USFWS. 2013b. *California Clapper Rail 5-Year Review*, Sacramento CA. Available online at: https://www.fws.gov/sfbaydelta/documents/CaClapperRail5Yr Review.pdf. Accessed October 16, 2019.

- USFWS. 2016. San Pablo National Wildlife Refuge. Available online at: https://www.fws.gov/uploadedFiles/San%20Pablo%20Map%20\_Trails.pdf#a. Accessed May 2020.
- USFWS. 2017a. Pacific Gas and Electric Company Bay Area Operations & Maintenance Habitat Conservation Plan. Available online at: https://www.fws. gov/sacramento/outreach/2017/11-22/docs/PGE\_Bay\_Area\_HCP\_Final.pdf. Accessed May 2020.
- USFWS. 2017b. Plan Your Visit. Available online at: https://www.fws.gov/refuge/San\_ Pablo\_Bay/plan\_your\_visit.html. Accessed May 2020.
- USFWS. 2021. Visitor Activities. Available online at: https://www.fws.gov/refuge/San\_ Pablo\_Bay/activities.html. Accessed May 2020.
- USFWS. 2009. Sears Point Wetland and Watershed Restoration Project Draft Environmental Impact Report/Environmental Impact Statement.
- USFWS and CDFW. 2009. Sears Point Wetland and Watershed Restoration Project Environmental Impact Report/Environmental Impact Statement. Draft. Prepared with the support of ICF Jones & Stokes. Sacramento, California.
- United States Department of Transportation (USDOT). 2011. Policy Statement on Climate Change Adaptation. June. Available online at: https://www.fhwa.dot.gov/ environment/sustainability/resilience/policy\_and\_guidance/usdot.cfm. Accessed August 21, 2019.
- USDOT. 2018. National Highway Traffic Safety Administration Corporate Average Fuel *Economy*. Available online at: https://www.nhtsa.gov/laws-regulations/corporateaverage-fuel-economy. Accessed August 21,2019.
- United States Global Change Research Program (USGCRP). 2018. Fourth National Climate Assessment. Available online at: https://nca2018.globalchange.gov/. Accessed August 21, 2019.
- Ward, P., T. McReynolds, and C. Garman. 2003. Butte and Big Chico Creeks Springrun Chinook Salmon, *Oncorhynchus tshawytscha*, Life History Investigations 2001-2202. Prepared for CDFW.
- WRECO. 2021. Draft Water Quality Assessment Report. May 2021.
- Yoshiyama, R.M., F.W. Fisher, and P.B. Moyle. 1998. Historical Abundance and Decline of Chinook Salmon in the Central Valley Region of California. *North American Journal of Fisheries Management* 18: 487-521.